

Higher Education Institutions and Sustainable Development Implementing a Whole-Institution Approach

Edited by Marco Rieckmann and Inka Bormann Printed Edition of the Special Issue Published in Sustainability



www.mdpi.com/journal/sustainability

Higher Education Institutions and Sustainable Development

Higher Education Institutions and Sustainable Development—Implementing a Whole-Institution Approach

Editors

Marco Rieckmann Inka Bormann

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade • Manchester • Tokyo • Cluj • Tianjin



Editors Marco Rieckmann University of Vechta Germany

Inka Bormann Freie Universität Berlin Germany

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Sustainability* (ISSN 2071-1050) (available at: https://www.mdpi.com/journal/sustainability/special_issues/Higher_Education_Institutions_and_Sustainable_Development).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, Article Number, Page Range.

ISBN 978-3-03936-988-1 (Hbk) ISBN 978-3-03936-989-8 (PDF)

© 2020 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

Contents

About the Editors
Preface to "Higher Education Institutions and Sustainable Development—Implementing a Whole-Institution Approach"ix
Markus Vogt and Christoph Weber The Role of Universities in a Sustainable Society. Why Value-Free Research is Neither Possible nor Desirable
Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 2811, doi:10.3390/su12072811
Bror Giesenbauer and Georg Müller-ChristUniversity 4.0: Promoting the Transformation of Higher Education Institutions towardSustainable DevelopmentReprinted from: Sustainability 2020, 12, 3371, doi:10.3390/su1208337123
Isabel Ruiz-Mallén and María HerasWhat Sustainability? Higher Education Institutions' Pathways to Reach the Agenda 2030 GoalsReprinted from: Sustainability 2020, 12, 1290, doi:10.3390/su1204129051
Lisa Bohunovsky, Verena Radinger-Peer and Marianne Penker Alliances of Change Pushing Organizational Transformation Towards Sustainability across 13 Universities
Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 2853, doi:10.3390/su12072853
Sebastian Mehling and Nina KolleckCross-Sector Collaboration in Higher Education Institutions (HEIs): A Critical Analysis of anUrban Sustainability Development ProgramReprinted from: Sustainability 2019, 11, 4982, doi:10.3390/su1118498289
Sebastian Niedlich, Mara Bauer, Margarita Doneliene, Larissa Jaeger, Marco Rieckmann and
Inka Bormann Assessment of Sustainability Governance in Higher Education Institutions—A Systemic Tool Using a Governance Equalizer Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 1816, doi:10.3390/su12051816
Benjamin Nölting, Heike Molitor, Julian Reimann, Jan-Hendrik Skroblin and Nadine Dembski
Transfer for Sustainable Development at Higher Education Institutions—Untapped Potential for Education for Sustainable Development and for Societal Transformation Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 2925, doi:10.3390/su12072925
Mara Bauer, Sebastian Niedlich, Marco Rieckmann, Inka Bormann and Larissa Jaeger Interdependencies of Culture and Functions of Sustainability Governance at Higher Education Institutions
Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 2780, doi:10.3390/su12072780
Kerstin Schopp, Matthias Bornemann and Thomas Potthast The Whole-Institution Approach at the University of Tübingen: Sustainable Development Set in Practice Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 861, doi:10.3390/su12030861

Nicolas Roos, Xaver Heinicke, Edeltraud Guenther and Thomas W. Guenther The Role of Environmental Management Performance in Higher Education Institutions Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 655, doi:10.3390/su12020655
Eugénia de Matos Pedro, João Leitão and Helena Alves
Bridging Intellectual Capital, Sustainable Development and Quality of Life in Higher
Education Institutions
Reprinted from: <i>Sustainability</i> 2020 , <i>12</i> , 479, doi:10.3390/su12020479
Martin Barrett, Kyle S. Bunds, Jonathan M. Casper, Michael B. Edwards, D. Scott Showalter
and Gareth J. Jones
'A Nut We Have Officially yet to Crack': Forcing the Attention of Athletic Departments Toward
Sustainability Through Shared Governance
Reprinted from: <i>Sustainability</i> 2019 , <i>11</i> , 5198, doi:10.3390/su11195198
Bankole Osita Awuzie and Amal Abuzeinab
Modelling Organisational Factors Influencing Sustainable Development Implementation
Performance in Higher Education Institutions: An Interpretative Structural Modelling
(ISM) Approach
Reprinted from: <i>Sustainability</i> 2019 , <i>11</i> , 4312, doi:10.3390/su11164312

About the Editors

Marco Rieckmann is Professor of Higher Education Development in the Department of Education of Faculty I—Education and Social Sciences at the University of Vechta. He is Presidential Advisor on Sustainability of the University of Vechta, representative of the German Educational Research Association (GERA) in the Council of the European Educational Research Association (EERA) and Speaker of the German-speaking network "Teacher Education for Sustainable Development" (LeNa). His main research interests are higher education development, (higher) education for sustainable development, and sustainable university development.

Inka Bormann is Professor of General Education at the Free University of Berlin. She is a member of the Ad Hoc Group on Indicators for Education for Sustainable Development of the United Nations Economic Commission for Europe (2019–2020) and an elected member of the German Commission for UNESCO. Her main research interests are trust in and towards educational institutions, educational governance, transformative learning through engagement, education for sustainable development, and sustainable university development.

Preface to "Higher Education Institutions and Sustainable Development—Implementing a Whole-Institution Approach"

Sustainability is an urgent developmental task for our society and is attracting increasing attention. Therefore, higher education institutions (HEIs) are also called upon to deal theoretically, conceptually, methodically, critically, and reflectively with the associated challenges and the processes and conditions of transformation in order to contribute to sustainable development. But how can complex organisations, such as HEIs, succeed in initiating and maintaining the process of sustainable development within their own institutions and make it a permanent responsibility? How can as many protagonists as possible be persuaded to get involved in sustainable development? For these questions, there is no patent recipe, no guidelines to action, no checklist that would be equally helpful for all universities or that could be applied across the board by all. HEIs are too different, for example, in terms of their legal form (private or public), their location (rural or metropolitan), or size (small and specialised or large universities). In addition, HEIs are influenced by external framework conditions that promote aspects of sustainability to varying degrees, depending on national or regional policies. While the higher education landscape is currently still characterised more by individual sustainability projects and the addressing of sustainability issues in individual courses, the focus should be more on implementing sustainability in the structures - understood as a holistic transformation of learning and teaching environments. HEIs should see themselves as places of learning and experience for sustainable development and should therefore orient all their processes towards principles of sustainability. For Education for Sustainable Development to be more effective, each HEI must be transformed as a whole. Such a whole-institution approach aims to integrate sustainability into all aspects of each HEI. It involves rethinking the curriculum, operations, organisational culture, learner participation, leadership and management, community relations, and research. In this way, the institution itself acts as a role model for the learners. This book deals with the promotion of sustainable university development and provides an overview of how universities can be organised sustainably and how sustainable development can be implemented in their various functional areas. In the sense of a "whole-institution approach", which encompasses entire HEIs, the focus is not only on the core areas of teaching (higher education for sustainable development) and research (sustainability in research) but, also, on the operational management of HEIs. In addition, this book focuses on sustainability governance and transfer for sustainable development at HEIs as cross-disciplinary issues.

> Marco Rieckmann, Inka Bormann Editors



Article



The Role of Universities in a Sustainable Society. Why Value-Free Research is Neither Possible nor Desirable

Markus Vogt and Christoph Weber *

Chair of Christian Social Ethics, Ludwig-Maximilians-Universität Munich (LMU), 80539 Munich, Germany; m.vogt@lmu.de

* Correspondence: ch.weber@lmu.de; Tel.: +49-89-2180-2296

Received: 26 December 2019; Accepted: 31 March 2020; Published: 2 April 2020

Abstract: The current climate crisis confronts us with a deep discrepancy between knowledge and action. Therefore, this article is looking for a readjustment of the relationship between science and society. The positivist self-understanding of science and its fragmented organizational form lead to a marginalization of ethical questions. Instead, sustainability calls for a re-examination of the preconditions and embedding contexts of supposedly value-free research. Faced with the increasing complexity of the modern world, ethics must spell out a new "grammar of responsibility" that addresses the prevalent "declamatory overload of responsibility". Ethicists can fulfil this role by uncovering and regulating conflicting goals and dilemmas. Instead of playing the role of "marginal echo chambers", universities ought to assume their social responsibility as structural policy actors. This article suggests a methodology of responsible research as a specific ethical contribution to the model of "transformative" and "catalytic" science for a "post-normal age". True to their founding mission, academia should herald a "New Enlightenment" that is more self-reflexive regarding its own practical and ethical preconditions, foundations, and consequences. This article presents a possible practical method for fostering the dialogue between the natural sciences and the humanities and to link research, education, practice, and social communication in new ways. It is concluded that a foundation of a whole-rationality approach with a multidimensional understanding of wisdom and, respectively, rationality and sagacity is necessary for sustainable universities.

Keywords: sustainability in science; transformative science; grammar of responsibility; ethics of knowledge; universities as echo chambers of society; catalytic science; whole-institution approach

1. Introduction—Humanity is Running out of Time

The current situation of global society in the upheaval of modernity is marked by a discrepancy between knowledge of probable future disasters and a lack of adequate reaction of today's society. Theoretically, available technology could ensure proactive protection for the environment and climate that respects the "planetary boundaries" [1] (p. 1) of a safe and fair space for the development of human civilization. However, there is a lack of social will and of binding political framework conditions to shift from an irresponsible logic of competitive pressure to ecological and social foresight.

The development of solutions to scientific-technical problems requires great trust in an open future and in the complex interaction between self-dependent actors, with such interaction being conducted freely and reasonably. By contrast, deep mistrust is spreading regarding democracy, reason, and the concepts of progress with which science is closely interwoven. Ethically and politically, particularly precarious is the increasing distrust of international cooperation and of the ethical universalism of human-rights-based political liberalism [2–4]. Against this background, the role, communication conditions, and tasks of science in political discourse are changing.

This article focuses on a definition of the relationship between science and society, based on the "whole-institution approach" [5] (p. 19). Instead of entrusting the issue of sustainability just to

a single academic department, the "whole-institution approach" encompasses the university as a whole. On all levels and all fields, sustainability needs to be established as a core value and common goal that all stakeholders actively seek to put into practice. For instance, sustainability "involves rethinking the curriculum, campus operations, organizational culture, student participation, leadership and management, community relationships, and research" [6] (p. 46). This process can be fostered by offering financial and administrative support, training schemes, guidelines, and best practice models. The central task is to establish viable inter-institutional, interdisciplinary, and transdisciplinary networks. Thus, the university itself becomes a beacon of sustainability by linking research and practice on its own campus [6,7] (pp. 39–59, pp. 113f.).

The main thesis of this paper is that ineffective ethical knowledge relates to a deeper deficit of the current self-understanding and of the organizational form of science, namely a shortened understanding of rationality, which subsequently leads to the marginalization of ethical questions. Therefore, this article stimulates a debate with an understanding of ethics as "philosophy of science" and a norm–theoretical analysis of different ways by using the terms "responsibility" and "freedom". It explores, in a theoretical and practical way, whether and how universities can contribute to a sustainable society under changed conditions of communication.

On the one hand, the "social grammar" of responsibility in the field of tension between actor, object, and controlling authority, and the handling of highly complex risks, is examined. On the other hand, the understanding of rationality and the associated current debate about the relationship between science and society in times of climate change as well as the "post-factual" weakness of trust in reason and democracy are analysed. The aim of this paper is to outline an ethics of knowledge that understands research, freedom, and responsibility as a unit and plumbs the academic discourse space anew.

Owing to the above-mentioned reason, the considerations are divided into five parts (a–e): (a) The current situation of science between the role of observer and actor is presented at first in Section 2. (b) Section 3 will describe the need to change the cultural patterns and guiding values of society, so that universities can become driving forces for a cultural revolution. (c) The methodology of how this change can take place follows in Section 4, and (d) the vision of a "New Enlightenment" ends the line of argument in Section 5. (e) In the final section, Section 6, all arguments are summarized together with discussion from a different point of view from that of the authors, as well as other issues not considered in this article.

2. Science between the Role of Observer and Actor

Numerous scientists around the world have raised their voices because they do not want to witness society fall into the climate trap. Inspired by the vigor of striking students with "Fridays for Future", the "Scientists for Future" initiative has developed into a strong international network in 2019. At the level of higher education policy, there are various initiatives to institutionalize climate responsibility. Examples include the Germany-wide joint project "HOCH^N" (www.hoch-n.org), the "Network University and Sustainability in Bavaria" (www.nachhaltigehochschule.de/), and the German "Science Platform Sustainability 2030" (www.wpn2030.de).

2.1. Re-Examination of the Relationships Between Science and Society

Thus, the relationships between science and society, knowledge and responsibility, and freedom and autonomy must be re-examined. The debate is conducted under various headings, but none of these approaches have received proper attention so far. They include the terms "socially responsible research" [8] (p. 38), "transdisciplinary" [9] (p. 68), "catalytic science" [10] (p. 44), "knowledge hierarchy (expert vs. lay)" [11] (pp. 86f.; 89f.), "citizen science" [12] (pp. 13f.), "dialogical" and "integral higher education system" [13] (pp. 166f.), "transformative science" [14] (p. 17), "oppositional and emancipatory science" [15] (pp. 101f.), or "science for a post-normal age" [16] (p. 739). All of these approaches urge the scientific community to play an active role in orientation and conflict resolution in

the multi-layered field of tension of the current processes of change. They establish a robust knowledge base for a just and sustainable design in a transdisciplinary, dialogical, and context-sensitive way.

In our view, Ortwin Renn's concept of "catalytic science" most succinctly summarizes the various aspects of the relationship between science and society. Renn specifies the role of science in social transformation processes by describing science as a catalyst. Scientists should not see themselves as moderators, but should contribute their knowledge as an indispensable energy to dissolve blockages in thought and processes, and to activate desirable transformations. Among the scientific public, however, the term "transformative science" has become established as a guiding concept in debate, perhaps because the term is controversial and has stimulated conceptual discussions [10].

Ultimately, the pragmatic challenges of climate change call into question the current fundamentals of science. For example, the French sociologist Geoffroy de Lagasnerie argued that scientists are already involved in social change—as soon as they begin to produce ideas and discourses [15] (p. 14):

"If the concept of science (and especially the intellectual field) is to have relevance as a sphere of discussion, then the questioning of certain structures in which knowledge is produced (and also the question of what knowledge production actually means) is not an 'attack on science', but, on the contrary, a form of the use of scientific reason, which, in this case, chooses itself as its object, a kind of academic practice which remains loyal to its concept and definition" (translation from German) [15] (pp. 81f.).

Truth is not a neutral descriptive perspective, but an "oppositional concept" that shows how and why a practice or an institution is incorrect [15] (p. 55). If one locates usefulness in this original linkage of theory and practice, then it is not an externally applied measure to evaluate utilitarian consequence. Instead, usefulness represents an inherent moment of the practice of knowledge. It is a counterpart to the self-referential nature of science. Especially in the humanities, such self-referencing increasingly refers to itself through a flood of footnotes and thus forms a closed system that seems to decouple itself from the outside world [17].

2.2. Three Dimensions of Sustainability Research

The best politically established term in the search of responsible, transformative, catalytic, or public science is "sustainable research". However, there are three quite different ways of understanding the concept:

- (1) Sustainability research in a broad sense focuses on particular questions of sustainability, such as climate change, renewable energy, and biodiversity.
- (2) Research procedures respect guidelines about sustainability, for example, with regard to the use of natural resources, animal welfare, and social compatibility
- (3) Sustainability research in a narrow sense examines the coherence of the concept and its normative logic. This is essentially a logic of integration, inclusion, and balance, which seeks to harmonize heterogeneous and conflicting goals and to establish strategic networks between different fields and levels of action. Ethical reflection should not stop at asserting synergies. It is also necessary to analyze conflicts and trade-offs, to reflect on priorities, to define criteria for appropriate decisions in different contexts, and to establish procedures for dealing with dissent. Of central importance here is to mediate between the different logics of the social subsystems.

The concept "research in a socially responsible way" (translation from German) [8] (p. 38) focuses on the second area: It primarily develops standards for the process of research. However, the other two dimensions are by no means excluded. Academic responsibility cannot be limited to a few formal criteria of research, but must rather deal with the grand challenges.

2.3. The Conflict between the Normative Claim of Transformative Science and the Positivist Theory of Science

The normative claim of sustainable and transformative science is an attack on the positivist theory of science. In terms of moral theory, this is associated with a profound dilemma. By reducing the

understanding of "reason" in a positivist manner to its knowledge function, it necessarily surrenders itself to a morality of subjective decisions and purposes released to arbitrariness: It declines to the means for goals about which it itself ultimately does not decide [18] (p. 33). In positivism, morality is understood as a question of subjective preferences that cannot be further justified and is therefore excluded from the concept of science [15] (pp. 17–27).

Much of what Max Weber wrote 100 years ago in his two lectures "Science as a Vocation" [19] and "Politics as a Vocation" [20] is still valid and ground-breaking. This is relevant not only in ethics, but also for the self-understanding of science and politics. Nevertheless, there are several methodological problems in his model of ethics of responsibility. Weber proposed a method of weighing consequences and strict separation between a science that ascertains and analyzes facts versus a policy that negotiates compromises between diverse interests, preferences, and value convictions [21] (pp. 97–121). The model needs critical further development, because reflecting on the rationality of the goals and ideas of a good and meaningful life are excluded from the thinking space of science. The concept of the unity of analytical and normative reason becomes fragile. This is especially true with regard to its perception by ancient and medieval traditions, especially the concept of wisdom (phronesis or prudentia) [22,23]. Weber's theory of responsibility as consequentialism is subjected to a calculation of purposeful rationality, in which essential dimensions of practical reason are ignored [24].

With regard to methods of ethical decision making, the concept of responsibility needs to be supplemented [18] (pp. 17–128). For Horkheimer and Adorno, Weber's separation of ethics from research and action from knowledge in conventional social-scientific thinking neglects the practical use of the conceptual systems and one's own public role. This tendency hides the structural preconditions and consequences as well as the perspective of scientific positions behind the appearance of neutrality instead of making them transparent [25]. It also undermines the necessary distance from the system of rules of society that has coagulated in science [26].

The positivist understanding of science needs to be critically revised [15] (pp. 17–27) and relativized regarding the preconditions and embedding contexts of supposedly value-free research [27] (pp. 201–240). A multidimensional understanding of wisdom, prudence, intelligence, knowledgeability, judiciousness, and sagacity has to be established as the foundation for a "whole-rationality approach" of understanding sustainability sciences and sustainability education. In addition, spirituality might be a crucial part of "sustainable wisdom" [28] (pp. 279–290). It is a kind of rationality which is open to the ambiguity of the world [29]. At the same time, it is a form of "practical wisdom" [22].

Thus, with new urgency, the old question arises as to whether science can be content with analyzing the world, or whether it should also immediately strive to change it. Is the role of science mainly that of an observer or an actor? What role do universities play in the society?

Universities are not only observers. They can also be seen as "change agents" by having an ethical scientific theoretical basis in the concept of sustainability. This core has established itself above all in the sphere of politics and was initially a socio-political and not a scientific concept. It is a discourse of responsibility whose strong normative charge in its deep structure does not fit with current ideas of freedom, autonomy, and scientific excellence at universities [30]. Some scientists fear that the freedom of science will be used for ethical and political purposes and thus sacrificed.

Against the background of this unresolvable tension between different models of scientific theory, the claim of responsible and transformative science should not be interpreted primarily as a moral appeal. Such a claim should first be reflected in theories of science and norms. Science in the era of climate change should not lose itself in activism under the pressure of supposedly urgent political goals. It will only succeed if it methodically and structurally reflects the search process in the field of tension between empirical research, normative demands, and social transformation in relation to the self-conception and organization of science. The reflections of de Lagasnerie [15], Schneidewind [31], Grunwald [14], Müller-Christ [13], and others support this idea, but require greater ethical depth.

The following part combines a reflection about the role of universities in society with some considerations about the underlining concept of rationality and its normative implications. The thesis is that we need a new discovering of the traditional idea of wisdom. The aim of the argumentation is a contribution to overcoming the dualism between science and society by connecting theoretical and practical competences as well as empirical and normative approaches. Universities are understood as "structural policy actors" that actively reflect upon and shape their own working conditions. This leads to a shift in understanding progress which underlines the need to change the cultural patterns and guiding values of society.

3. Universities as Driving Forces for a Cultural Revolution

3.1. The Crisis of the Wise—Universities as Marginal Echo Chambers?

The science journalist Manuel Hartung became known for his "College Novel", published in 2007, about the everyday life of modern students. He diagnosed a "crisis of the wise" that is ignited by the question of whether universities want to remain in their "marginal echo chamber" or become "centers of social certainty" (translation from German) [32].

This harsh criticism can be read as a call for transformative science. The current highly differentiated scientific world produces heaps of detailed data-fed studies and reflections; these mostly generate neither existential knowledge nor the will to act. Not least due to the vast increase in scientific publications that are produced with great effort but seldom read, many sectors of science are threatened with becoming self-referentially closed systems. This would mean that "scientific impact" is insufficient as a criterion for excellence, and should be accompanied by "societal impact" to ensure quality [33] (p. 27). This complaint is not new (cf., e.g., the educational theoretical analyses of Alfred North Whitehead dating back to a lecture cycle held at Oxford University in 1912 [34]). In the context of the challenges of climate change and digitization, however, such a sentiment is topical and urgent [35] (pp. 150–163). In brief, people do not know what to believe, and do not believe what they know. We are displacing ecological knowledge because we do not want to admit it. The announced catastrophe reaches our consciousness superficially because we are trapped in the comforts of everyday life and cannot imagine its loss. We compensate with a bad conscience and moral appeals to third parties. There is a lack of a sense of reality, because the uncomfortable facts of climate change and global poverty remain abstract for most people. They are of little immediate sensual significance.

Science is faced with a communicative dilemma. If it points out its unavoidable fuzziness, "then it develops no appellative force and fails to make the point that the world could very soon become very similar to the fictional scenarios of doom" (translation from German) [36] (p. 4). An apocalypse of global warming and extreme events seems to be perceived as a myth, and "any resembling prognosis seems to be untrustworthy precisely because of this resemblance. That is where the effect we are experiencing right now comes from. If the present actually bears (pre-)apocalyptic traits, this is not perceptible, or can easily be repressed or rationalized on a small scale" (translation from German) [36] (p. 4).

Against this background, the crisis of the wise is not simply a failure of individual intellectuals, but is deeply rooted in the modern concept of rationality and science. This diagnosis calls for the transformation of science. Climate and environmental change, which are increasingly harsh real experiences, challenge scientists to question the current patterns of thought and action. Under great time pressure, they need to make the knowledge base available for comprehensive transformation of the economy and modern ways of life. This undertaking will not succeed without a revision of our human self-understanding. The Anthropocene takes philosophical anthropology to new horizons of reflection. This period can be described as a new phase of enlightenment with a changed thrust [37] (pp. 92f.), focusing on the integration of fragmented knowledge.

Sustainability science as an educational method can be measured by whether it enables an ethically founded reorientation in the Anthropocene and links diverse segments of dissociated knowledge landscapes. It aims to enable students to acquire the knowledge to generate judgement and the will to act. This can also be described as "emancipatory science" (translation from German) [15] (pp. 101f.).

3.2. Transforming the Concept of Rationality into a Reliable Normative Compass

The controversial debate about the role of universities in society in times of climate change challenges the self-understanding of science, which also has to answer to the contempt of rationality in parts of the public debates. The "crisis of the wise" can also be interpreted as the consequence of a limited understanding of rationality through loss of the classical concept of wisdom [22]. This includes analytical as well as normative and everyday practical skills. It combines a precise perception of the situation with qualitative standards of value resulting from wishes, interests, and convictions in order to compare different alternatives for action and to make goal-oriented decisions. Wisdom enables individuals and collectives not simply to blindly follow their own preferences, but to judge, weigh, coordinate, and implement those preferences according to the situation. Wisdom establishes a concept of rationality that integrates the discourse about a successful lifestyle and enables decisive action as the central moment of all virtues. It constitutes "our personal grammar of importance, preference, and desirability" [22] (p. 10) and self-esteem as an "evaluative gravitational field of personal identity" [22] (p. 11).

In modern times, wisdom is "moved from its ethical center. Wisdom is losing its ethical impregnation and has no longer the character of a life-management authority" [22] (p. 7). Wisdom becomes a "contingency management technique" that serves interests. It is focused on an "ethically neutral optimization of consequences of action" and "loses the dimension of supervision over one's own quality of life" (all translations from German) [22] (p. 7). The common good can thus only be understood as the intersection of private interests. However, this remains a highly fragile construct, which—given the complex challenges of the Anthropocene—cannot generate sufficient stability for long-term global cooperation.

Science has become the catalyst for such profound change in society that ethical reflections are scarcely able to follow. The German Advisory Council on Global Change (Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen) therefore proposes a "normative compass"; at its center is the concept of dignity [38] (pp. 2f.). This is flanked by the maxims of participation, the protection of individual character and diversity, and the conservation of natural resources. However, these concepts remain abstract and powerless unless they prompt science to shape the digital future and promote ethical–political development. If humanity, as "Homo Deus" [39], with our increased use of technology, is seen godlike, the inversion of all values in favor of a transhumanism can hardly be stopped. Then, the humanistic foundation of the universities would be a leftover of the past [40].

3.3. Redesigning Discourse Spaces by Overcoming the Dualism between Science and Society

Transformative science relativizes the significance of the disciplinary professional community as a delimitation of discursive spaces and affiliations. According to de Lagasnerie, the supposed autonomy and neutrality through subject-related membership generates a dualistic two-world constellation. The first constellation is that of academic discourses and the second is the public in media, politics, and society [15] (p. 94). Emancipatory science overcomes this dualism by creating situational spaces for discussion in the common struggle for justice. It establishes an inclusive relationship between intellectuals, politics, and the public sphere [15] (p. 102).

This adjustment in the relationship between science and society can be described as overcoming a dual chronology. The old model initially envisaged an "internal" discussion and knowledge generation reserved for specific disciplines, which were then confronted—in a second step—in their encounter with the "outside world" through some kind of dialogue [15] (p. 102). Models of "citizen science" [12] (pp. 13f.) and "dialogical science" [13] (pp. 166f.) understand the public as well as every day and practical knowledge of supposed laymen as always present in the process of knowledge acquisition. This results in a "heterogeneous intellectual space, [...] to which activists, artists, writers, and authors with the most diverse horizons belong" (translation from German) [15] (p. 103). In this space, thinking occurs; it represents a public exterior space to which science refers before the practice of knowledge emerges.

The structural change of the public through digital media is also changing the way science is conducted and communicated. This raises fundamental questions, both practical and philosophical. The practical aspect concerns data protection, transparency, and the suggestive power of digital algorithms. The philosophical aspect concerns human self-understanding in view of the increasingly superior functional performance of artificial intelligence in certain areas [35]. Fair, inclusive, and humane coping with the complex radical processes of change associated with digitization requires new discourse spaces in the dialogue between scientific, social, entrepreneurial, and political practice. This is the only way to involve complex expert knowledge without installing the exclusive rule of experts that Strohschneider rightly rejects [41] (p. 190). Digitization offers many opportunities for innovation, without which adaptation to the challenges of climate change would be slow or impossible [42].

Citizen Science does not mean that the difference between spontaneous, civic, and scientific knowledge is leveled. Scientists always seek what others cannot see, even in practical social contexts. Through their scientific perspective, they discover new and uncomfortable questions [15] (pp. 104–106). If the people abandon the idea of academic discipline as a community and commit themselves to ethical spaces of discussion, society does not abandon research and science at all, but affirms both, because, in this way, the citizens address both of them [15] (p. 106). Since the judgement of everyday experience plays an important role in ethical questions, and questions of sustainability have an ethical dimension, the various models of civic science are of particular importance here.

The most methodologically differentiated and long-established guiding concept for redesigning scientific discourse spaces is transdisciplinarity [9]. Three aspects are guiding here [10] (p. 46): Research practices that adapt their objects of investigation, methods, and questions to social problems; cross-disciplinary development of analyses and options for action, considering empirical knowledge of contextual conditions and areas of application in practice; integration of knowledge carriers outside of science, throughout all phases of the knowledge and research process, to develop practical and socially accepted solutions.

Transdisciplinarity is a hybrid of three concepts of science. These are 1) the classical, which offers orientation aid with the help of systematic analyses of complex contexts, 2) the instrumental, which formulates problem-solving options for action with the help of scenarios, and 3) the catalytic, which contributes co-creatively to controlling decision-making and communication processes on the basis of a procedural design [10] (pp. 49f.). The success of scientific social and policy advice regarding the current transformation processes and governance problems in the context of climate change, digitization, and sustainability depends to a large extent on whether a synthesis of these three strategies succeeds at the level of "high-quality knowledge" [43] and whether science establishes itself as an "honest broker" [44,45] (p. 113, pp. 2f.) of mediation in complex conflict situations. The prerequisite is that science should cultivate and promote communicative intelligence in heterogeneous discourse spaces as an integral part of its self-image and working method.

To adequately fulfil this demanding role, transformative science needs places and arenas in which its socially productive impact can be experienced and tested [33] (pp. 27f.). Institutional innovations and resources are also important to overcome the strong inertia in the science system and to open up transdisciplinary discourse spaces more than just selectively—or with a loss of quality [46] (p. 64).

What the concepts of transdisciplinary research and transformative science have in common is the goal of "reviving the relationship between science and society and supporting it with proposals for political innovations to promote sustainability research" (translation from German) [33] (p. 26). It deals with the self-image of the scientists as well as with institutional questions of change and the distribution of resources. The concept is about the "scientific impact" in connection with the "societal impact" [33] (p. 27), regarding the role of science in deliberative democracies based on reason [44]. The various concepts can also be summarized as science in the service of society.

3.4. Universities As "Structural Policy Actors"

Against the background of the complex debates briefly outlined here, what has been said is applied in this final section on the role of universities and a new concept of transformative science. That is, how can sustainability—in the sense of a "whole-institution approach"—be anchored in the depth structure of the university so that it becomes an influential structural policy actor?

Universities face heterogeneous role expectations: Education of a growing number of students with a simultaneous integration of all learners; producers of innovative knowledge for the economy; a place of critical thinking. The multitude of expectations creates considerable pressure. Autonomy helps to mobilize productivity and efficiency potentials. Often, however, this is a sham autonomy, since it merely addresses the resource constraints with a range of management instruments, which usually follow a business rationality model [13,15,47] (p. 171, pp. 24–27, 33–36, p. 84). The increase in university autonomy through liberation from state and political influence often leads to appropriation by other subsystems (e.g., by industry as a third-party donor).

Securing autonomy requires a critical examination of one's own institutional conditions for action. Universities are structural policy actors; they are institutions whose actions do not simply react to social framework conditions, but actively influence those social conditions with their strategies [15,47] (p. 82, p. 102). To this end, universities must perceive the multiple demands placed on them as an opportunity for further development, to become social bridge-builders between the various social subsystems. In the sense of a "whole-institution approach", holistic approaches to the design of higher education institutions should aim to shape the curriculum and the management of the organization. Cooperation with other local institutions, initiatives, and actors in the sense of sustainable development are also crucial. Autonomy means moving freely between systems in the field of plural demands. This can succeed only if the areas of tension in the sense of an intelligent dilemma management, a high tolerance of ambiguity, and transparency and credibility are mastered and placed into a productive relationship [6,47] (pp. 46–48; pp. 86f.).

Sustainability is not an externally defined goal, but an open search process with heterogeneous target components, which are thus plural and culturally variable [28] (pp. 134–179, 369–372). It does not replace normative debates, but challenges them. A precondition for this challenge is that the adjective "sustainable" is not used synonymously with "good". Instead, the relevant ethical and cultural guiding values—such as freedom, human dignity, justice, or prosperity—must be reflected. The conflicting goals must clearly be named. Transformative science addresses the conditions and goals of research. It does not aim at softening quality standards in favor of practical purposes, but at a self-reflexive and pluralistic value debate about good education and excellent research. It understands the ability of science not only to collect data, but also to evaluate them and to recommend actions as a criterion for excellence. In such an understanding of science, "ethics" is not a single criterion among many, but provides a framework for reflection and a structure for argumentation of the whole. It aims at "excellence of responsibility". This has proven its worth in the ability to identify the epistemic and structural causes of conflicts of values and justice and to identify dilemmas, while representing clear options and priorities.

3.5. A Paradigm Shift in the Understanding of Progress

The scientific system reflects information highly selectively, using specific codes that reduce complexity and guarantee efficiency, effectiveness, and verifiability. These features enable specialization and dynamism, but at the same time create limited perception [47] (p. 88). Non-system subordinate sequences are produced, but are usually not recognized and therefore not adequately processed. In a reflexive modernity [11,48], however, it will be important to think systemically about the social side effects of technical and socio-economic innovations from the outset. To achieve this, rapidly changing knowledge societies depend on a prospective responsibility among scientific institutions. Ultimately, it is about a "cultural revolution" [49] (p. 3), a "World in Transition—A Social Contract for Sustainability" [50]. This is not only the postulate of some scientists, but is already a political decision:

With the Sustainable Development Goals (SDGs) and the Climate Treaty of Paris, the international community has committed itself to such a revolution. By addressing old and new industrialized countries as well as countries of the Global South, these documents bid farewell to the development concept that the United Nations previously promoted [51]. However, this farewell has not been consistent, and many contradictions arise in the tension between ecological and socio-economic goals [52] (pp. 13–20, 55–67). Awareness of the consequences of the SDGs for everyday politics, economy, and society seems limited. Thus, the potential for conflict associated with the SDGs is considerably underestimated.

The implementation of the SDGs cannot be achieved without a change in the cultural patterns and guiding values of society. These cannot be changed by decisions from above, but must change gradually in a complex interplay between value change, institutional design of framework conditions, and pioneers of transformative practice. So-called Leaderships are indispensable at all levels. One of their core tasks is to illustrate new ways of understanding progress. In this way, universities can change their role from marginal echo chambers to driving forces for a revolution of the society in favor of sustainability. In order to grow up in this role, they have to provide a better understanding of transformation and the challenges of responsibility not as a moralistic proclamation, but as a scientifically reflected concept. Some considerations about this are the content of the next paragraph.

4. Transformation and the "Social Grammar of Responsibility" as Methodology

The claim of a "World in Transition" [50] derived from the moral principle of sustainability aims for a global, intergenerational, and ecologically unbounded responsibility. Responsibility can be defined as a relationship that contains three elements. It is a competence that (1) lies with someone, (2) for something, (3) towards someone.

All questions of accountability and liability occur in such a three-dimensional field of tension [53] (p. 23). The relationship between subject, object, and addressee constitutes the "social grammar of responsibility" [21] (p. 11). The thesis is that all three dimensions are today specifically unsettled. Our legal system, international institutions, and sense of morality are insufficiently prepared to identify and attribute responsibility in dealing with ecological collective goods or diverse socio-cultural values. In addition, the universities are overstrained with that kind of unbounded claim of responsibility" in such a way that subjects for action, accountability, and control become tangible in the complex processes of delimited development. The goal cannot be absolute control, but it can be counteraction in the most conspicuous fields of irresponsibility. The task of universities is also to enable interdisciplinary and methodological reflected research about ethical questions, especially about moral dilemmas and not trivial ethical conflicts.

Beyond the orientation for such a system of rules, an ethical–systematic reflection on the concept of responsibility can provide access to an existential understanding of ethics. Ethics is neither mere optimization of consequences, nor simply the deductive application of norms and principles. It is more than a benevolent attitude or an altruistic willingness to renounce advantages in favor of others. Ethics must address the challenge of people living together in a given situation. This responsive character is directly expressed in the concept of responsibility. Ethics refers to attentive care in dealing with people as well as complex technical and social challenges. Responsibility as a virtue requires active planning and the willingness to learn more about life's creative possibilities. It is a basic attitude that can be enforced or calculated only to a limited extent from the outside. Responsibility manifests itself in the willingness to account for one's actions, to both oneself and others. When applied to universities, that means that the claim of responsibility cannot be handled adequately just by delegating it to a special discipline, but it has to be implemented in the habitus of responsible scientists and a sustainable reorganization of universities regarding their social and ecological impact.

4.1. The Declamatory Overload of Responsibility

Instead of harmonizing and thus resolving conflicting goals, the more convenient path is often chosen: Many different goals are set up without taking into account that some of them contradict each other and thus cannot be easily fulfilled at the same time. The result is an excessive demand [54] and "declamatory overload of responsibility" [55] (p. 298). For example, the SDGs seem to initiate a renaissance of utopian thinking [56] (pp. 13–109). This results from the unresolved tension between their developmental policy and ecological goals. However, they simply lack conceptual coherence. Unfortunately, the Paris Agreement in its present state that wants the countries to pursue efforts to limit the increase of climate warming to 1.5 °C must now also be regarded as a largely utopian target [57].

The commitment of all governments to implementing the "2030 Development Agenda" is accompanied by a blatant contradiction in the form of measures not taken [51] (pp. 247f.) This could be called a mode of "Symbolic Politics and the Politics of Simulation" [58], or perhaps a dilemma of politics in the field of tension between national and international challenges. Preserving the claim to rationality of deliberative democracy, however, especially in post-factual times, depends on an active role of science in the complex and conflictual transformation processes of the present. These overtax the traditional arenas and forms of communication of politics [44].

From an ethical perspective, the arenas, limits, and conditions of responsibility need to be reassessed. Without dealing with the structural political prerequisites of their active perception and implementation, the full-bodied rhetoric of responsibility is just an "uncovered check" [53] (p. 187). The limits of attributing responsibility, planning sustainability, and controlling political processes must be remeasured [59] (pp. 1–6). The ethical critique of utopian thinking thus does not primarily aim at reducing the goals to an extent that is supposedly achievable in realpolitik. Rather, it aims at reflecting on the conditions for action, resistance, and coherence problems in which the serious will must prove itself. Such a critique aims at dilemma management regarding the difficulty of acting correctly under wrong systems or conditions. It is also concerned with transformation and governance knowledge regarding the change of regulatory structures.

The SDGs have established themselves as a ground-breaking document in the struggle for sustainable development. There are now quite demanding reflections on ways and conditions of implementation. However, the dilemma of the great promises of responsibility associated with the global, intergenerational, and ecological expansion of responsibility threatens to lead to excessive demands. This would result in a permanent crisis of legitimacy.

The problem, however, goes even deeper. The promises of humanity that have been promoted in modern times [53] (pp. 49–72) are being tested in new ways. The great success of the development idea is bought by "wear and tear of the biosphere" (translation from German) [51] (p. 251). Wolfgang Haber also saw a fundamental contradiction between humanitarian development goals and the "uncomfortable truths of ecology" (translation from German) [60] (pp. 45f.). This contradiction is not solved by SDGs, but is hidden, and it renders the full-bodied speech of responsibility in the anthropoid debate a short-sighted suppression of planetary boundaries [61]. Promises of prosperity and responsibility are unsustainable in the age of the Anthropocene. That dilemma situation is also a crucial problem of universities because they are driving forces of the traditional promise of progress. The challenge is to define a new version of a humanism which is aware of ecological interconnectedness and the dialectic tension between progress and risk. To establish this new understanding of progress and humanism, there is a need for deep inter- and transdisciplinary discourses. They have to be pluralistic and intercultural, but not without engagement and obligation. Universities have to become "transformation labs" in search of a culture of sustainability and responsibility.

One of the crucial challenges for a sustainable society is that responsibility is eroded by the anonymous system logic of the modern age. In the confusion of late modern societies, it is often impossible to determine with certainty the subjects of action, the objects, and the addressees of responsibility [55] (p. 299). Familiar models of addressing, delimiting, and monitoring responsibility are often no longer applicable to the anonymous, widely ramified, and confusing networks of actors of

late modern societies [21]. The dynamics of functionally differentiated system logics with partially incalculable effects, such as the highly complex financial system, do not seem to be effectively managed within the framework of the traditional concept of responsibility. In view of the side effects of action that no one wants or can calculate, virtue ethics runs into nowhere. Consequentialist concepts of responsibility—that is, concepts related to weighing consequences—fail because of the unpredictability of the non-linear results of action in complex chains of effects [21] (pp. 23–66).

4.2. Promoting a Culture of Risk-Taking and Innovation

In assessing technological innovation, "the heuristics of fear" [62] (p. 27) emphasizes the prognosis of calamities. However, this may paralyze the capacity for action and innovation in many areas, thus possibly generating more dangers than limiting them [63]. The contribution of research to a sustainable society consists essentially in innovation. The evaluation of innovation cannot be sufficiently performed within the framework of classical concepts of technology assessment.

Often, the effects of technological innovation cannot be predicted in advance, but become visible in an open research process. The concept of responsible innovation thus needs philosophical–ethical as well as socio-theoretical and legal basic research on how to deal with ignorance and uncertainty [64]. In recent years, a process-accompanying resilience concept has established itself as a normatively rich and open category, neither purely defensive nor risk-blind [65,66]. The systematic core of an ethic of innovation is the theoretical reflection of action and institutions on how to deal with systemic risks and the different types of ignorance with which the search for responsible decisions is confronted [67]. It needs "risk maturity" [28] (pp. 347–372) in the sense of a culture-conscious and self-reflexive handling of systemic risks. It also requires an ethical foundation that goes beyond a consistent weighing of consequences. Innovations are always risky and difficult to control. Without openness to innovation, however, the specific problem-solving potential of science can hardly unfold. Up to now, universities missed the development and practice of a differentiated model of freedom and autonomy in the context of climate change. This has to include an adequate degree of ecological responsibility and services for the public in order to be clearly different from arbitrariness.

A theory of responsibility in and through science must distinguish between the necessary freedom for desirable innovations and the equally necessary restrictions to avoid the escalation of systemic risks. Humanity knows too little about social transformation processes in many areas, and often fails because of path dependencies of technological developments. Responsible science needs an intelligent combination of scientific-technical, entrepreneurial, political-institutional, cultural, and social innovation. The EU concept of "Responsible Research and Innovation" could be a pioneer in this respect. Due to the complexity of cause-effect relationships, it is insufficient from an ethical-systematic viewpoint to interpret "responsibility" primarily in the space of a utilitarian-consequentialist-oriented technology assessment and linear intervention models. Rather, these models must be fundamentally extended by participatory, democratic-theoretical, systemic, and cultural aspects [42,63,68]. To date, "Responsible Research and Innovation" has been mainly a buzzword that points to a field of research that is still underdeveloped in ethical-systematic terms. It is not yet a normative, clearly structured concept for the political governance of responsible research [69] (pp. 753; 757).

A crucial strategy to counter uncomfortable environmental–ethical claims in the context of climate change is to try to get rid of those claims by questioning the truth or precision of the scientific analysis of reality. In the worldwide movements of the "Science March" and "Scientists for Future", scientists have protested against the fact that scientific findings are not taken seriously enough in many areas of shaping society's future.

Julian Nida-Rümelin diagnosed a connection between the refusal to take note of scientific facts relevant to action and the "ideology of anti-realism" [70] (p. 33). Political powers may insinuate that there is no reality, but rather only opinions conveyed by the media. Digital media are being cleverly used to reinforce this impression. Many people nowadays seem to form their opinions in "digital bubbles" that immunize against criticism because they select and reinforce predetermined opinions. This

dynamic is politically dangerous and explains the deep uncertainty behind the superficial political excitement about the rhetorical phenomena of the post-factual. The uncertainty of communication in the age of digital manipulation of opinions demands active effort from the sciences to provide solid information to the public and to maintain standards of rationality.

As the binary logic of the digitally comprehensible and operationalizable forms of rationality becomes dominant, the fiction of measurability and quantitative comparability within linear logics leads to a unification of the world [29]. This allows the old development model to be revived in the operationalization of SDGs because it corresponds to the logic of comparability and optimization [51]. There is no room in the model of digital algorithms for the perception of cultural peculiarities and identity conflicts. In relation to ethics, the question arises whether normative reason is mere convention and a question of subjective preferences; or is content that is scientifically accessible and true suitable for such reason? [70] (p. 82). If it is assumed that ethics has a scientifically founded truth content, then it can and must be uncomfortable and must not depend on public opinion. The contempt for expert knowledge and reason undermines the foundations of our culture, as do self-generated systemic constraints that prevent politics, business, and society from doing what is reasonable [21] (p. 10). These questions are not just abstract problems of the philosophy of science, but affect the heart of self-understanding of universities and their role in the society.

4.3. The Grammar of Responsibility—Uncovering and Overcoming Conflicting Goals and Dilemmas

The decisive task is therefore to spell out the "grammar of responsibility" in such a way that subjects for action, accountability, and control become tangible again in the complex processes of delimited development. The art of responsibility is the distinction between primary and secondary issues as well as between different levels of responsibility and degrees of commitment. It aims at empowerment justice in the sense of a subsidiary strengthening of autonomy, personal potential, and participation. In this context, freedom is understood not merely as freedom of choice or arbitrariness, but refers to the propensity for self-determination that creates identity as a moral subject in conflict situations.

Such conflicts should not simply be ignored in the sustainability discourse. Although the discourse is ethically and systematically geared towards integration, it is necessary to distinguish between objectively appropriate, productive compromises versus lazy and short-sighted compromises. Such questions are not only, as Weber suggests, the tasks of politics, but also of ethics as method-guided scientific reflection of normative reason [18]. The transformation of SDGs into a scientific concept depends on whether the complex problems of their conceptual coherence and the associated attributions of responsibility are reflected in an interdisciplinary manner.

Global society is losing sight of the moral and cultural foundations of social cohesion, because there is no consensus on the changed conditions of the interplay of freedom and responsibility in the processes of disruptive change in an increasingly polycentric world. Climate change poses enormous challenges to the idea of freedom that Ghosh classifies as the most important political concept of modern times [71]. In a society that has banned the idea of collectives from politics, business, and literature in equal measure, it is difficult even to think about the collective structure of climate responsibility. An individualistic narrow concept of freedom and responsibility, devoid of its cultural and institutional embedding contexts, fails because of the complex chains of action and causality in climate change. The purely formal understanding of freedom as the maximization of options is empty because it leads to an ultimately uncontrollable indeterminacy.

Ethics should not surrender to the apparent autonomy of system imperatives, but must analyze actor constellations in complex networks in a new way. In doing so, it can negotiate moral and legal attributions of accountability on the basis of social–anthropological, institutional, and action– and property–theoretical approaches [72]. This will in turn require a critique of the "neoliberal revolution" that has "elevated the market to the center of society" and "weakened public institutions" (translation from German) [73] (p. 16).

According to Patrick Deneen's diagnosis, the transnational universalism of ethics has ended because the elites in politics, business, and science are unilaterally exploiting their opportunities for freedom for private gain [2] (pp. 15–27). The normative basis of an open democratic society, namely the interplay of responsibility and freedom, had been abused and systemically eroded. Thus, the social embedding and legitimation of modern science seems fragile. Universities in some countries are already experiencing this fragility on a large scale. Deep distrust against the academic elite is breaking ground worldwide in populism and is becoming a political force [4].

This accusation hits science hard, which can only develop its potential under the condition of fundamental trust in the reason of freedom. Science is challenged to defend the trust in reason and freedom as the basis of democracy by proactively assuming responsibility for overcoming the central social challenges. Instead of hiding behind the non-binding nature of a systemic, ultimately questionable concept of freedom and idly watching its structural incapacitation in many countries, it must become more political and transformative. The necessary knowledge for a sustainable model of global development requires a transdisciplinary approach that can be generated and communicated to key actors in a broadly effective and comprehensible manner. Such methodological, practical, and scientific-political reflection is the necessary first step to a science of sustainability that is both aware of its responsibility and respects the freedom of research. It is a prerequisite for science not simply to adopt declamatory large goals from the outside, from the political realm, but to adopt goals from the inside. The competencies of science in analyzing problems, opportunities, and strategies for action should be methodically reflected upon. Strengthening the voice of science in the context of political decision-making processes improves the chances of responsibility. This would not, as the critique of transformative science proposes, imply an "atrophy" of the political [41] (p. 190). Exactly this is an indispensable task of ethics. If ethics is to be more than a trivializing Sunday rhetoric, it must not stop at conjuring up desirable things, but must uncover and overcome conflicting goals and dilemmas. Responsible science's task is to be the voice of those who have no voice in the arena of power.

After this norm-theoretical approach, the following reflections will focus on the relation of scientific research to society and the societal impact of research and education in the context of the actual discussion about climate change and the challenges of sustainability.

4.4. Research and Its Social Responsibility

Transformative science should help students to reflect on key problems in depth and to develop action knowledge suitable for their era [13] (p. 162). How this can be achieved is a topic of its own, to which reference can only be made in the context of the reflection focused here on research ethics. Such a reference, however, is necessary and integral if one adheres to Humboldt's idea of the university as a unity of research and teaching. Clever and responsible graduates are an indispensable "resource" of modern knowledge societies. Since students are the first addressees of the knowledge generated at universities, models of transformative and emancipatory science must always prove themselves. This can only succeed if students are not seen as passive recipients of knowledge, but are increasingly involved in the generation and communication of knowledge. Transformative science and transformative education belong together [6,7,74] (pp. 46f.; pp. 113f.; pp. 51f., 69). With regard to the "whole-institution approach", even "the institution itself functions as a role model for the learners. Sustainable learning environments, such as eco-schools or green campuses, allow educators and learners to integrate sustainability principles into their daily practices and facilitate capacity-building and competence development, and value education in a comprehensive manner" [6] (p. 4).

If thinking is interpreted as the capacity for critical reflection, then thinking will always be questioning and uncomfortable. The debate on sustainability has often been oriented towards compromises. Now, the mood seems to be turning to radical criticism of culture and the system. In this situation, the voice of a science that weighs complex arguments and remains objective without trivializing conflicts is essential.

5. Towards a New Enlightenment

5.1. A Methodically Controlled Reflection on the Epistemic and Normatively Rich Premises of Every Science

The discourse on sustainability leads to a philosophical–scientific reflection on the epistemic and ethical–political foundations of the project of modernity, which have become fragile and need further development. It is important not to give up on human rights universalism, but at the same time to become more sensitive to cultural contexts and ecological preconditions, which are often decisive for its concrete perception. Equally important is a critical revision of notions of rationality, space, and time, as well as the freedom and political control that underlie the project of modernity.

Ernst Ulrich von Weizsäcker and Anders Wijkman postulated a "New Enlightenment" [37] (p. 92), which they characterized as Enlightenment "that is fitting for the 'full world' and for *sustainable* development" [37] (pp. vii). Its core, according to the authors, is a methodically controlled reflection on the epistemic and normatively rich premises of every science, even supposedly value-free fields. Fundamental to the success of a New Enlightenment is a new quality of dialogue between the natural sciences and the humanities, which for decades have been divided into "two cultures" [75]. An explanation of the prerequisites and limits of the different models of rationality in the sciences is the decisive impulse for interdisciplinary discourse. The predominant model of enlightenment and rationalization as linearly increasing secularization and "disenchantment" is questioned [27]. The New Enlightenment strives to regain the unity of reason and faith that was a central guiding idea in the founding of European universities; however, it does so in a changed context, as an impulse for a comprehensive understanding of rationality [37] (pp. 92–98).

The current challenges of global development are concerned with fundamental questions as well as human and world views. Hence, theological and religious aspects play a constitutive role-for example, regarding the ideas of progress, prosperity, quality of life, and social cohesion. Notions about nature, the meaning of history, and the sources of responsibility and morality are also addressed. The significance of enlightened theology in a pluralistic public sphere and scientific culture is not limited to updating and revising religiously orientated knowledge. In the interest of enlightened reason, theology must also counter the false public use of religion, above all religious self-exaltation of the political, ecological, or economic. Often, the religious factor expresses itself in "post-secular societies" [76] (p. 328) by granting a quasi-religious status to the ecology and linking "green religion" with sense-making [77]. Here, the task of interdisciplinary dialogue among theology, philosophy, and the humanities or social sciences is to distinguish between "neomythic" ways of thinking and a critical awareness of the mutual referentiality of faith and knowledge [76] (pp. 328–338). The often-subliminal equation of freedom with market freedom, of the model "homo economicus" with an image of man, or of profit maximization with claims to the creation of meaning also challenges the sciences to a New Enlightenment. Such demarcations are as much a scientific-theoretical task as a cultural one. They require new forms and forums for transdisciplinary dialogue.

A New Enlightenment uncovers ambivalences of modernity. For example, human identity cannot be construed as an isolated point-shaped self. It is rather formed in relationships with others and the other, with self-understanding depending on the absolute other [78] (pp. 6f.; 321–324). Critical examination of the "anthropic principle" of modern idealism [79] is an indispensable deep dimension of spiritual and moral reorientation in the upheaval of late modernity.

The classification of empirical, normative, and transformative aspects of knowledge must be reflected in a new way if science is to promote not only factual knowledge, but also judgement and action competence [21] (pp. 436–456). What is needed is an ethics of knowledge that fosters conversation among the various forms of rationality. This is the only way in which universities can guarantee comprehensive education in the claim to genuine, cosmopolitan "universitas" and to develop problem-oriented interdisciplinary approaches to solve the increasingly complex questions of the future concerning society, the environment, and technology. A self-reflexive enlightened science focuses on the institutional embedding of knowledge generation and reception. In turn, this focus is associated

with contingent perspectives, interests, and prerequisites. It is up to such a science to give more space to the old concept of phronesis (wisdom) as a value-led and context-sensitive judgement [80] (p. 65). In addition, a rediscovery of the spiritual dimension of ethics and of sustainability could be part of such a new phase of upgrading Enlightenment [28] (pp. 279–282, 482–494).

One possibility of implementing a New Enlightenment—including critical awareness of values and history—would be a philosophical and theoretical examination of the fundamental models of the respective subject at the beginning of all courses of study. Each course could begin with conceptual training, learning to think about and understand the normative premises of the subject-specific guiding models. A general philosophical–ecological study program could also be implemented for first-year students at universities to introduce sustainability knowledge in the fields of general education, environmental sciences, and social sciences.

5.2. A New Understanding of the Relationship between Sustainability and Freedom

A crucial part of that enlightenment in the age of climate change is a new understanding of the relationship between sustainability and freedom. Sustainability has to be understood as empowerment to freedom. People are not only entitled to have freedom and human rights, but should also be empowered to exercise them by securing the necessary ecological, socio-cultural, and economic conditions for all. This includes future generations and people in the Global South. By referring to the basic capabilities, freedom becomes concrete. This understanding of sustainability can be inspired by the capability approach of Amartya Sen and Martha Nussbaum. Following Amartya Sen, one could spell out "sustainability as freedom": Sustainability promotes freedom by securing the increasingly decisive preconditions. This also applies to universities; that is, commitment to sustainability requires enforcing the freedom of research and teaching through securing their essential prerequisites [74] (pp. 84f.).

Freedom arises by assuming responsibility [81] and through the practice of independent and resistant thinking. Hannah Arendt did not define "The Freedom to Be Free" [82] as the absence of fear and limitations. In the sense of the Aristotelian understanding of man as "zoon politikon", she defined freedom as participation in the political process. This is also a task of higher education, not only individually, but institutionally. Universities can only secure freedom and autonomy if they actively shape their own conditions for action as structural policy actors. The striking discrepancy between knowledge and action in times of climate change challenges educational institutions to link research, education, practice, and social communication in new ways. These are aimed at enabling independent analysis and catalytic participation in shaping social change. Knowledge that takes itself seriously seeks to reduce the discrepancy between critical thinking and everyday action; therefore, it has an ethical dimension. Creating such a transdisciplinary ethics of knowledge is a crucial task for universities during this time of climate change [74] (pp. 87f.). Shifting from the abstract idea of freedom to an awareness of socioecological preconditions of freedom to be free lies at the core of the new enlightenment.

5.3. Overcoming the Fragmentation of Knowledge

The central thesis of this article is that value-free research is neither possible nor desirable. Therefore, the following steps of argumentation are crucial: Science in a knowledge-driven society is one of the challenges we face in the context of sustainability, just as it is part of the solution. Because of this ambivalence, science has to be self-reflexive regarding its own practical and ethical preconditions, foundations, and consequences. This could be the starting point of a new phase of enlightenment. Therefore, this article proposed a philosophy of science as a methodological frame for such a "New Enlightenment" through inter- and transdisciplinary research and a transformative role of science under the demands of sustainability. This role can be described as catalytic because it is not the main force for transformation, but just a specific element of it. Philosophy of science can help to overcome the fragmentation of knowledge and be an enabler for the dialogue between the

natural sciences, humanities, ethics, and society. Finally, philosophy of science resists the positivist self-understanding of science which leads to a marginalization of ethical questions [15,21]. Instead, sustainability calls for a new emancipation of science as a structural policy actor.

Thus, this essay proposed a multidimensional understanding of wisdom, rationality, and sagacity as a foundation of a "whole-rationality approach" for a sustainable university. The challenges of sustainable, transformative, or responsible science are grounded in the current self-understanding of the organizational form of science, namely a shortened understanding of rationality. Therefore, it needs a philosophy of science and a norm-theoretical analysis of the different models of understanding science. Only on this basis can universities contribute to a sustainable and responsible society under the conditions of the age of Anthropocene.

Faced with the increasing complexity of the modern world, ethics must spell out a new "grammar of responsibility" which organizes accountability in a complex world [74] (pp. 29–47). Ethicists can fulfil this role by methodologically reflecting, uncovering, and regulating conflicting goals and dilemmas. The morally overloaded avowal to sustainability needs to be transformed into a rational discourse about common but differentiated duties in a pluralistic world. The model of transformative and catalytic "science for a post-normal age" [16] (p. 739) requires a foundation in ethical reflection.

The acceptance of sustainability depends on whether it is thought of as a limitation of what humanity is not allowed to do, or understood as an authorization to act through long-term reason, cooperation, and fairness. The problem facing the global sustainability debate does not seem to be different definitions of objectives, but rather the question of how moral claims can be enforced given the dominance of particular interests. Focused on content, the sustainability and transformation discourse lacks reflection about what norms are and how they work [21] (pp. 9–20). The function of norms is not to indicate moral reasons when there are no obvious material reasons and interests for certain actions. Norms rather serve to open up an area of freedom by creating a certain distance from the world of the factual. In the modern knowledge society, solutions to the challenges of climate change and sustainable development can only be found on a scientific basis. Science can be only productive under conditions of self-dependence. It needs anticipation of trust. To justify and vindicate this, the scientists have to develop methods of self-control regarding criteria of responsibility. Sustainability could be a frame for the search for new models of ethical universalism in a broken world. The scientific frame of this is an understanding of responsibility as accountability in the field of tension between actor, object, and controlling authority, and the handling of highly complex risks. It has to be seen as an integral element of understanding rationality and the associated concept of science and society in times of climate change. The aim is to outline an ethics of knowledge that understands research, freedom, and responsibility as a unit, and plumbs the academic discourse space anew. Only the reunification of freedom and responsibility can build up a strong resistance against the "post-factual" weakness of trust in reason and democracy.

6. Conclusions

The argumentation of this article was developed in five steps:

- (1) In the present situation of climate change and the great acceleration in the epoch of the Anthropocene, humanity is running out of time. Therefore, the role, the conditions of communication, the perceptions, and the tasks of universities in the present public discourses are changing. Universities are no longer only observers, but also become "change agents" and have to deal with the request of a proactive relationship between science and society.
- (2) There is a wide call for a new role of science in society which is discussed under different headlines, e.g., transformative, responsible, public, sustainable, or catalytic science. All of these models are in a deep conflict with the positivistic theory of science, which is still dominant. Scientists are involved from the beginning in social change—as soon as they begin to produce ideas and discourses. It is their task to reflect upon this situation.

- (3) Against this background, universities are part of the problem and part of the solution as well. If they shall become driving forces for a sustainable society, they have to undergo a cultural revolution regarding the concepts of rationality, freedom, wealth, and progress. Therefore, the universities have to overcome a shortened understanding of rationality that leads to the marginalization of ethical questions. There is often a lack of scientific research about moral dilemmas on the way to a sustainable society. To deal with this adequately, sustainability science in the Anthropocene that recognizes the claim of the SDGs needs a comprehensive ethically founded reorientation as well as a transdisciplinary approach that establishes an inclusive relationship between intellectuals, politics, and the public sphere. Sustainable science includes wisdom, which constitutes our personal grammar of importance, preference, desirability, and identity. By beginning to act as "transformation labs", the universities become self-reflective "structural policy actors" that see sustainability not as an externally defined goal, but as an open search process with heterogeneous target components that have to be clearly specified. Hence, the normative claim of sustainable science is an attack on the positivist theory of science.
- (4) A responsible shaping of the accelerated change in society needs a "normative compass" with the concept of dignity at its heart. The centerpiece of this cultural and scientific revolution is a differentiated understanding of the concept of responsibility. The art of responsibility is the distinction between primary and secondary issues as well as between different levels and degrees of commitment. Needed is an analytical approach to the "grammar of responsibility" in its three dimensions. The relationship between subject, object, and addressee that constitutes the "social grammar of responsibility" helps to overcome the idleness of declamatory overloaded concepts of responsibility in favor of a better understanding of conflicts between ecological and social demands and a down-to-the-earth distance to utopian thinking. The challenge is to define a new version of a humanism which is aware of ecological interconnectedness and the dialectical tension between progress and risk. Responsible science's task in this context is to strengthen risk maturity that opts for innovations and avoids system risks. Scientists always have to be the voice of those who have no voice in the arena of power.
- (5) These considerations lead to the program for a "New Enlightenment" with three dimensions: (a) The autonomy of research requires a critical examination of one's own institutional conditions for action. So, science needs a methodically controlled reflection on the epistemic and normatively rich premises of every science. (b) The lack of consensus regarding the interplay of freedom and responsibility imperils the social cohesion. There is a need for a new understanding of the relationship between sustainability and freedom. (c) Regarding a whole-institution approach, transformative science and transformative education belong together. In order to promote a sustainable society in the age of Anthropocene, science needs to overcome the fragmentation of knowledge, the dualism between natural and social sciences. Thus, universities need a "whole-rationality approach" that is aware that value-free research is neither possible nor desirable. In this manner, sustainability enables the "freedom to be free".

Author Contributions: Conceptualization, M.V.; methodology, M.V.; writing—original draft preparation, M.V. and C.W.; writing—review and editing, M.V. and C.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by HOCH^N—Sustainability at Higher Education Institutions: Develop—network—report, grant number 13NKE007E.

Conflicts of Interest: The authors declare no conflict of interest.

References

 Rockström, J.; Steffen, W.; Noone, K.; Persson, Å.; Chapin, F.S., III; Lambin, E.; Lenton, T.M.; Scheffer, M.; Folke, C.; Schellnhuber, H.J. Planetary boundaries. Exploring the safe operating space for humanity. *Ecol. Soc.* 2009, *14*, 1–33. [CrossRef]

- 2. Deneen, P. Why Liberalism Failed; Yale University Press: New Haven, CT, USA, 2018.
- 3. Fukuyama, F. Identity: The Demand for Dignity and the Politics of Resentment; Macmillan: New York, NY, USA, 2018.
- Vogt, M. Politische Emotionen als moraltheoretische Herausforderung. Münchner Theologische Zeitschrift 2017, 68, 292–305.
- 5. Mehling, S.; Kolleck, N. Cross-Sector Collaboration in Higher Education Institutions (HEIs): A Critical Analysis of an Urban Sustainability Development Program. *Sustainability* **2019**, *11*, 4982. [CrossRef]
- Rieckmann, M. Learning to transform the world. Key competencies in Education for Sustainable Development. In *Issues and Trends in Education for Sustainable Development;* Leicht, A., Heiss, J., Byun, W.J., Eds.; United Nations Educational, Scientific and Cultural Organization: Paris, France, 2018; pp. 39–59.
- O'Donoghue, R.; Taylor, J.; Venter, V. How are learning and training environments transforming with ESD? In *Issues and Trends in Education for Sustainable Development*; Leicht, A., Heiss, J., Byun, W.J., Eds.; United Nations Educational, Scientific and Cultural Organization: Paris, France, 2018; pp. 111–131.
- Fraunhofer-Gesellschaft; Helmholtz-Gemeinschaft; Leibniz-Gemeinschaft. Forschen in gesellschaftlicher Verantwortung. In Ein Leitfaden zum Nachhaltigkeitsmanagement in Außeruniversitären Forschungsorganisationen (LeNa); Fraunhofer-Gesellschaft, Helmholtz-Gemeinschaft, Leibniz-Gemeinschaft: Munich, Germany, 2016.
- Mittelstrass, J. The Order of Knowledge. From Disciplinarity to Transdisciplinarity and Back. *Eur. Rev.* 2018, 26, 68–75. [CrossRef]
- 10. Renn, O. Die Rolle(n) transdisziplinärer Wissenschaft bei konfliktgeladenen Transformationsprozessen. *GAIA* 2019, 28, 44–51.
- 11. Beck, U. World at Risk; Polity Press: Cambridge, UK, 2009.
- 12. Finke, P. Citizen Science; Das unterschätzte Wissen der Laien Oekom: Munich, Germany, 2014.
- Müller-Christ, G. Nachhaltigkeitsforschung in einer transzendenten Entwicklung des Hochschulsystems-ein Ordnungsangebot für Innovation. In *Innovation in der Nachhaltigkeitsforschung*. Theorie und Praxis der Nachhaltigkeit; Leal Filho, W., Ed.; Springer: Berlin, Germany, 2017; pp. 161–180.
- 14. Grunwald, A. Transformative Wissenschaft-eine neue Ordnung im Wissenschaftsbetrieb? *GAIA* 2015, 24, 17–20. [CrossRef]
- 15. de Lagasnerie, G. Denken in Einer Schlechten Welt; Matthes & Seitz: Berlin, Germany, 2018.
- 16. Funtowicz, S.O.; Rawetz, J.R. Science for the post-normal age. Futures 1993, 25, 739–755. [CrossRef]
- 17. Rieß, P.; Fisch, S.; Strohschneider, P. Prolegommena zu Einer Theorie der Fußnote; Lit: Münster, Germany, 1995.
- Korff, W. Norm und Sittlichkeit. Untersuchungen zur Logik Normativer Vernunft, 2nd ed.; Alber: Freiburg im Breisgau, Germany, 1985.
- Weber, M. Science as a Vocation. In *The Vocation Lectures*; Owen, D., Strong, T., Eds.; Hacket Publishing: Indianapolis, IN, USA, 2004; (first publication in 1919).
- Weber, M. Politics as a Vocation. In *The Vocation Lectures*; Owen, D., Strong, T., Eds.; Hacket Publishing: Indianapolis, IN, USA, 2004; (first publication in 1919).
- 21. Möllers, C. Die Möglichkeit der Normen. Über Eine Praxis Jenseits von Moralität und Kausalität; Suhrkamp: Berlin, Germany, 2018.
- 22. Kersting, W. Rehabilitierung der Klugheit. In *Klugheit;* Kersting, W., Ed.; Verlbrück Wissenschaft Verlag: Weilerswist, Germany, 2005; pp. 7–14.
- 23. Roos, J. Practical wisdom: Making and teaching the governance case for sustainability. J. Clean. Prod. 2017, 140, 217–224. [CrossRef]
- Luckner, A. Klugheit und Orientierung. Historisch-systematische Ortsbestimmung. In Klugheit. Begriff—Konzept—Anwendungen; Scherzberg, A., Ed.; Mohr Siebeck: Tübingen, Germany, 2008; pp. 3–23.
- Horkheimer, M. Traditional and Critical Theory. In Subject and Object. Frankfurt School Writings on Epistemology, Ontology, and Method; Groff, R., Ed.; Bloomsbury Academic: New York, NY, USA, 2014; pp. 185–232.
- 26. Adorno, T.W. The Positivist Dispute in German Sociology; Heinemann: London, UK, 1976.
- 27. Joas, H. Die Macht des Heiligen. Eine Alternative zur Geschichte der Entzauberung; Suhrkamp: Berlin, Germany, 2017.
- Vogt, M. Prinzip Nachhaltigkeit. Ein Entwurf aus Theologisch-Ethischer Perspektive, 3rd ed.; Oekom: Munich, Germany, 2013.
- 29. Bauer, T. Die Vereindeutigung der Welt. Über den Verlust von Mehrdeutigkeit und Vielfalt, 9th ed.; Reclam: Stuttgart, Germany, 2018.

- Mittelstraß, J. Der Philosophische Blick. Elf Studien über Wissen und Denken; Berlin University Press: Berlin, Germany, 2015.
- 31. Schneidewind, U. Transformative Wissenschaft—Motor für Gute Wissenschaft und Lebendige Demokratie. *GAIA* 2015, 24, 17–20. [CrossRef]
- Hartung, M. Krise der Klugen. In Die Zeit; Zeitverlag: Hamburg, Germany, 2017; Available online: https://www.zeit.de/2017/09/demokratie-gefahr-universitaeten-wissenschaft (accessed on 15 October 2019).
- Singer-Brodowski, M.; Schneidewind, U. Transformative Wissenschaft: Zurück ins Labor. GAIA 2019, 28, 26–28. [CrossRef]
- 34. Whitehead, A.N. The Aims of Education and Other Essays; Macmillan: New-York, NY, USA, 1929.
- 35. Nida-Rümelin, J.; Weidenfeld, N. *Digitaler Humanismus. Eine Ethik für das Zeitalter der Künstlichen Intelligenz;* Piper: Munich, Germany, 2018.
- Schneider, J. Klimakatastrophe: Die Apokalypse ist leider auserzählt. In *Die Zeit;* Zeitverlag: Hamburg, Germany, 2019; Available online: https://www.zeit.de/kultur/2019-07/klimakatastrophe-apokalypseweltuntergang-hysterie-erderwaermung/komplettansicht (accessed on 15 October 2019).
- 37. von Weizsäcker, E.U.; Wijkman, A. Come on! Capitalism, Short-termism, Population and the Destruction of the Planet—A Report to the Club of Rome; Springer: New York, NY, USA, 2018.
- WBGU—Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen. Towards our Common Digital Future; WBGU: Berlin, Germany, 2019.
- 39. Harari, Y. Homo Deus. A Brief History of Tomorrow; Harvill Secker: London, UK, 2016.
- 40. Vogt, M.; Weber, C. Verschmelzung von Mensch und Technologie als lohnenswertes Abenteuer? Ethische Spannungen angesichts des Transhumanismus. In *Digitalisierung und Nachhaltigkeit*; von Hauff, M., Renner, A., Eds.; (in press 2020).
- Strohschneider, P. Zur Politik der Transformativen Wissenschaft. In *Die Verfassung des Politischen*; Brodocz, A., Herrmann, D., Schmidt, R., Schulz, D., Schulze Wessel, J., Eds.; Springer: Wiesbaden, Germany, 2014; pp. 175–192.
- Lindner, K.; Goos, S.; Güth, O. Responsible Research and Innovation als Ansatz für die Forschungs-, Technologie- und Innovationspolitik—Hintergründe und Entwicklungen; Büro für Technikfolgen-Abschätzung beim Deutschen Bundestag: Berlin, Germany, 2016.
- 43. Bremer, S. Mobilising high-quality knowledge through dialogic environmental governance: A comparison of approaches and their institutional settings. *Int. J. Sustain. Dev.* **2013**, *16*, 66–90. [CrossRef]
- 44. Grunwald, A. Transformative Wissenschaft als honest broker? Das passt! GAIA 2018, 27, 113–116. [CrossRef]
- Pielke, R.A. The Honest Broker—Making Sense of Science in Policy and Politics; Cambridge University Press: Cambridge, UK, 2007.
- Bohunovsky, L.; Kernegger, M.; Kromp-Kolb, H.; Esca-Scheuringer, H.; Höltl, A.; Langthaler, M.; Stinnig, E.; Luks, F. Wissenschaft im Wandel. Hochschulen und die Sustainable Development Goals. *GAIA* 2019, 28, 63–65. [CrossRef]
- Schneidewind, U.; Singer-Brodowski, M. Transformative Wissenschaft. Klimawandel im deutschen Wissenschaftsund Hochschulsystem; Metropolis: Marburg, Germany, 2014.
- Beck, U. The Reinvention of Politics. Rethinking Modernity in the Global Social Order; Polity Press: Cambridge, UK, 1997.
- Pope Francis. Veritatis Gaudium. Apostolic Constitution on the Ecclesiastical Universities and Faculties. Available online: https://w2.vatican.va/content/francesco/en/apost_constitutions/documents/papa-francesco_ costituzione-ap_20171208_veritatis-gaudium.html (accessed on 15 October 2019).
- 50. WBGU—Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen. In World in Transition—A Social Contract for Sustainability; WBGU: Berlin, Germany, 2011.
- Sachs, W. Papst vs. UNO. Sustainable Development Goals und Laudato si'. Abgesang auf das Entwicklungszeitalter? *Peripherie* 2018, 150–151, 245–260. [CrossRef]
- 52. Maxton, G. Change! Warum Wir Eine Radikale Wende Brauchen; Komplett-Verlag: Munich, Germany, 2018.
- Höffe, O. Moral als Preis der Moderne. Ein Versuch über Wissenschaft, Technik und Umwelt; Suhrkamp: Frankfurt am Main, Germany, 1993.
- 54. Miller, D. National Responsibility and Global Justice; Oxford University Press: Oxford, UK, 2007.

- Lübbe, H. Moralismus oder fingierte Handlungssubjektivität in komplexen historischen Prozessen. In Kausalität und Zurechnung. Über Verantwortung in komplexen kulturellen Prozessen; Lübbe, W., Ed.; Springer: Berlin, Germany, 1994; pp. 289–301.
- 56. Hinkelammert, F. Kritik der Utopischen Vernunft. Eine Auseinandersetzung mit den Hauptströmungen der modernen Gesellschaftstheorie; Edition Exodus: Luzern, Switzerland, 1994.
- 57. Raftery, A.E.; Zimmer, A.; Frierson, D.M.; Startz, R.; Liu, P. Less than 2 °C warming by 2100 unlikely. *Nat. Clim. Chang.* 2017, 7, 637–641. [CrossRef] [PubMed]
- Blühdorn, I. Sustaining the Unsustainable: Symbolic Politics and the Politics of Simulation. *Environ. Politics* 2007, 16, 251–275. [CrossRef]
- 59. Vogt, M.; Weber, C. Current challenges to the concept of sustainability. Glob. Sustain. 2019, 2, 1-6. [CrossRef]
- 60. Haber, W. Die unbequemen Wahrheiten der Ökologie. Eine Nachhaltigkeitsperspektive für das 21. Jahrhundert; Oekom: Munich, Germany, 2010.
- 61. Haber, W.; Held, M.; Vogt, M. Die Welt im Anthropozän. Erkundungen im Spannungsfeld zwischen Ökologie und Humanität; Oekom: Munich, Germany, 2016.
- 62. Jonas, H. The Imperative of Responsibility. In *Search of an Ethics for the Technological Age;* University of Chicago Press: Chicago, IL, USA, 1984.
- 63. Vogt, M. Wandel als Chance oder Katastrophe; KomplettMedia: Munich, Germany, 2018.
- Böschen, S.; Schneider, M.; Lerf, A. Handeln trotz Nichtwissen. Vom Umgang mit Chaos und Risiko in Politik, Industrie und Wissenschaft; Campus: Frankfurt am Main, Germany, 2004.
- Böschen, S.; Vogt, M.; Binder, C.; Rathgeber, A. Resilienz—Analysetool sozialer Transformationen? *GAIA* 2017, 26, 164–165. [CrossRef]
- Schneider, M.; Vogt, M. Responsive Ethik. Reflexionen zum Theorie-Praxis-Verhältnis am Beispiel von Resilienz und sozialem Wandel. In *Christliche Sozialethik—Orientierung Welcher Praxis*; Emunds, B., Ed.; Nomos: Baden-Baden, Germany, 2018; pp. 179–199.
- 67. Renn, O. Das Risikoparadox: Warum wir uns vor dem Falschen Fürchten; Fischer: Frankfurt am Main, Germany, 2014.
- Richard, O.; Macnaghten, P.; Stilgoe, J. Responsible research and innovation. From science in society to science for society, with society. *Sci. Public Policy* 2012, *39*, 751–760.
- 70. Nida-Rümelin, J. Unaufgeregter Realismus: Eine philosophische Streitschrift; Mentis: Münster, Germany, 2018.
- Ghosh, A. The Great Derangement: Climate Change and the Unthinkable; University of Chicago Press: Chicago, IA, USA, 2017.
- Eckert, J. Tugendethik und Verantwortung-eine sozialanthropologische Perspektive. In Anthropologie und Ethik. Humanprojekt; Nida-Rümelin, J., Heilinger, J.-C., Eds.; de Gruyter: Berlin, Germany, 2015; pp. 151–170.
- 73. Chomsky, N. Kampf oder Untergang! Warum wir Gegen die Herren der Menschheit Aufstehen Müssen; Westend: Frankfurt am Main, Germany, 2018.
- 74. Vogt, M. Ethik des Wissens. Freiheit und Verantwortung der Wissenschaft in Zeiten des Klimawandels; Oekom: Munich, Germany, 2019.
- 75. Snow, P.-C. The Two Cultures; Cambridge University Press: Cambridge, UK, 1993; (first publication in 1959).
- Habermas, J. Faith and Knowledge. In *Frankfurt School on Religion*; Mendieta, E., Ed.; Routledge: New York, NY, USA, 2012; pp. 328–338.
- 77. Bolz, N. Die grüne Ersatzreligion. Leb. Seelsorge 2019, 1, 10–15.
- Taylor, C. Sources of the Self. the Making of the Modern Identity; Cambridge University Press: Cambridge, UK, 2012.
- 79. Welsch, W. Homo Mundanus—Jenseits der anthropischen Denkform der Moderne; Velbrück Wissenschaft: Weilerwist, Germany, 2012.
- Horn, C. Klugheit bei Thomas von Aquin. In *Klugheit;* Kersting, W., Ed.; Verlbrück Wissenschaft Verlag: Weilerswist, Germany, 2005; pp. 42–67.
- Vogt, M. Die Freiheit der Verantwortung. In Verantwortung—Freiheit und Grenzen. Interdisziplinäre Veranstaltungen der Aeneus-Silivius-Stiftung; Breidenstein, U., Ed.; Schwabe Verlag: Basel, Switzerland, 2016; pp. 7–38.
- 82. Arendt, H. The Freedom to Be Free. From Thinking Without a Banister; Vintage Books: New York, NY, USA, 2018.

Sustainability 2020, 12, 2811



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article University 4.0: Promoting the Transformation of Higher Education Institutions toward Sustainable Development

Bror Giesenbauer * and Georg Müller-Christ

Faculty of Business Studies and Economics: Sustainable Management, University of Bremen, Enrique-Schmidt-Str. 1, 28359 Bremen, Germany; gmc@uni-bremen.de * Correspondence: giesenbauer@uni-bremen.de

Received: 17 February 2020; Accepted: 16 April 2020; Published: 21 April 2020

Abstract: Higher education institutions (HEIs) could act as pivotal change agents for sustainable development (SD) in times of global climate action. However, HEIs have to respond to increasingly complex demands simultaneously, such as massification, globalization, marketization, and digitalization. Based on Graves' model of systemic development, this paper discusses two main strategies to deal with increased complexity in order to meet the challenge of SD: (a) Promoting general systemic development of a given HEI, progressively opening up to various stakeholders and focusing on co-creative collaboration, and (b) participating in inter-organizational networks to find inspiration for dealing with challenging trends. Four distinct phases of higher education development are presented. It is argued that transdisciplinary research and research-based learning will increasingly be needed for tackling societal issues and that HEIs should address different organizational subsystems individually. Furthermore, four types of inter-organizational networks are proposed and implications for network management are discussed. A case study of the HOCH-N network illustrates the practical application of the presented ideas. Finally, adopting a multi-dimensional and networked organizational model as an integrative University 4.0 is argued to be suitable for increasing the capacity to deal with complexity, thus meeting the challenge of sustainable development.

Keywords: higher education development; sustainable university development; systemic development; inter-organizational networks; sustainable development; worldviews; societal transformation; systemic transformation

1. Introduction

In times of climate change and massive societal change, universities and other higher education institutions (HEIs) are called to become change agents for societal transformation [1]. However, HEIs are facing complex times, juggling very disparate roles and demands [2]. From educating increasing numbers of students to conducting internationally competitive research, from acquiring grants and even making profits to serving society, from managing large physical campuses to integrating digitalization, from serving the local while keeping an eye on the global, and from supporting excellence in research in disciplinary niches to championing interdisciplinary topics such as sustainable development (SD): HEIs have to react to global trends that are difficult to chase at the same time and that indeed often impose conflicting goals. As the success of HEIs is mostly measured by international rankings and leadership in specialized disciplines, the advancement of cross-cutting topics such as sustainable development is often compromised [3].

Being part of the German research consortium on 'Sustainability at Higher Education Institutions' (*Nachhaltigkeit an Hochschulen*) HOCH-N [4], we wondered how HEIs could possibly come to integrate the pressing challenge of SD into their actions while also having to manage the aforementioned

challenges. Two blog articles by Massachusetts Institute of Technology (MIT) scientist and consultant Otto Scharmer on the systemic development of HEIs toward SD sparked our interest and seemed promising [5,6]. Therein, Scharmer proposes that HEIs should focus on "systemic upgrades" and increased inter-organizational collaboration in order to be able to deal with the challenge of SD. However, as the intersection between HEI development and SD is a relatively new one, the underlying systemic framework of Scharmer's articles has not yet been linked to sustainable higher education development in a scientific context.

The main research question we set out to answer is therefore: (A) Which general systemic strategies would help HEIs to deal with the complex challenges of the 21st century in order to live up to the challenge of sustainable development? Two secondary research questions were derived from our practical application of the presented ideas in the HOCH-N project: (B1) How can models of systemic development be applied to the development of HEIs? (B2) How can inter-organizational networks contribute to the integration of SD into HEIs and to the general systemic development of HEIs?

Having a background in the research on corporate sustainability management and systemic development in particular, we chose to focus our research on conceptually applying scientific models of systemic development that have proven to be of value to the analysis of corporate sustainability to the field of HEI development and SD and that seem to inform Scharmer's ideas scientifically. It is our understanding that all sound empirical research needs to be built on sound conceptual models and sound operationalization thereof. Thus, this contribution is intended to advance theory building on the transformation of higher education by systematically applying Clare W. Graves' model of systemic evolution to sustainable higher education development. This novel conception of higher education development will help to differentiate existing broader conceptions (cf. [7–9]), like the idea of a continuum of sustainability governance by Niedlich et al. [9].

As the overarching theme of this paper is the transformation of HEIs toward meeting the challenges of sustainable development, the latter is shortly introduced in Section 2.1. In order to set the stage for introducing two systemic strategies for managing complexity at HEIs, an encompassing theoretical framework is introduced in Section 2.2 and later applied to the advancement of sustainable development in higher education: Clare W. Graves' [10,11] theory of systemic evolution, originally derived from empirical data in the 1950s and 1960s [12]. Four distinct developmental phases of worldviews are presented and applied to HEIs in Section 3 (answering research question A and B1) and to inter-organizational networks in Section 4 (answering research question B2). The latter application is illustrated with examples from the German network for sustainability at HEIs "HOCH-N". Finally, the role of networks in promoting systemic change is discussed and further implications for change agents are inferred.

Thus, based on our practical work and the ideas of Scharmer [5,6], this paper introduces two strategies of how HEIs and their members are managing the complex demands of the 21st century: the first strategy is to promote the transformation of higher education in general, resulting in HEIs that are able to handle more complexity, requiring an update of a given HEI's self-concept and worldview (see Section 3). This strategy will lead to a multi-dimensional and networked organizational model, described as HEI 4.0 or University 4.0. The second, more specialized strategy is to engage in inter-organizational networks in order to connect with members of other HEIs dealing with similar challenges (see Section 4). The second strategy is essentially already implied by the first strategy. Conversely, the purpose of networks is often to nudge systemic transformation.

A case study of the German HOCH-N network is included in order to illustrate the proposed ideas on network management (Section 4.2). This case study focuses on the key structures, services, and products of the HOCH-N network, linking them to the presented systemic ideas on network management. It is not intended to "prove" the preceding conceptual framework as case studies are not suited to falsify conceptualizations in a classic way. However, the main scientific contribution of a case study can be described as "case-inspired self-reflection" [13] (p. 347) (cf. [14]). In this sense, the case

study on HOCH-N is intended to illustrate the practical application of the systemic theories at hand, to validate their subjective *usefulness*, and to inspire ideas for application in the reader.

Combining Graves' theory with the development of higher education and inter-organizational networks is the main contribution of this paper to the existing body of research, substantiating the ideas of Otto Scharmer on sustainable higher education development. It is further intended clarify the role of networks in promoting the systemic change of HEIs toward SD.

2. Background: The Challenge of Sustainable Development and Systemic Development

How can higher education institutions (HEIs) live up to the challenge of sustainable development (SD) while also having to deal with complex issues such as internationalization, massification, or marketization [2,15]? Following the reasoning of Otto Scharmer, HEIs would need to upgrade their "operating system" (cf. [6]) and increase their network activities in order to integrate SD as a whole-institutional approach [5,6]. But what is meant by sustainable development in higher education? And how can the upgrading of a HEI's operating system be understood?

In the following section, the topic of sustainable development is thus shortly introduced and linked to HEIs. Thereafter follows an overview of Clare W. Graves' systemic theories which have been applied successfully to the field of corporate sustainability and are linked to managing increasingly complex challenges [16,17]. Section 2 closes with a reflection of principles of systemic development to countervail the popular misconception that systemic development models are overly deterministic and judgmental models (see e.g., [18]).

2.1. Sustainable Development in Higher Education

Education for sustainable development has become increasingly important for HEIs and societies in general during the past three decades and thus, has created a more complex environment for operating a HEI. While this paper deals specifically with the challenge of integrating sustainable development (SD) into HEIs, the issue of SD is also used to illustrate complex systemic demands in general.

The publication of "Our Common Future" in 1987 by the world commission on environment and development [19] promoted the term "sustainable development" and gave birth to large global efforts to spread and develop the concept. Therein, SD is defined as "development that meets the needs of the present without compromising the ability of future generations to meet their own needs." [19] (p. 41). From the mid-1990s on, researchers have conceptualized ways of implementing SD both as a concept and as a way of thinking at universities (see e.g., [1,20]). Since then, the research on universities' role in supporting sustainable development has matured, especially in the context of competencies for future challenges (see e.g., [21–23]).

In September 2015, the United Nations [24] General Assembly adopted the Agenda 2030 resolution and thereby amplified the commitment to SD with the 17 sustainable development goals (SDGs) [25]. This framework aids in breaking down the complexity of SD into more practicable fields of action. Education for sustainable development (ESD) is part of SDG 4: High quality education.

Along those developments, the cause of sustainable development has received widespread attention and a lot of effort has been put into promoting it at higher education institutions [26,27]. In particular, the adoption of the SDGs and an increasing societal awareness of climate change urges HEIs to integrate SD into their actions.

But even though the idea of sustainability has reached the general public—for example, through Greta Thunberg and the *Fridays for Future* movement—the concept of sustainability itself has not yet made it to mainstream academia [8,28,29]. For example, the SDGs are rarely integrated into existing curricula or campus operations at German HEIs in spite of the fact that Germany has strong political initiatives for SD and ESD such as the German Council for Sustainable Development of the German Federal Government (*Rat für Nachhaltige Entwicklung*), the German Advisory Council on Global Change (*WGBU*), and the German National Action Plan for ESD (*Nationaler Aktionsplan BNE*) [30]. Moreover, the term sustainability is subject to several misconceptions which impedes the advancement of SD at HEIs [31].
One of the reasons that SD has not made its way into mainstream academia and management of HEIs is likely to be found in its inherent complexity, as SD requires systemic transformation and not only adaptation [7,32]. In other words, taking the challenge of SD seriously will lead to more tensions and dilemmas and therefore, to increased complexity [33]. Consequently, Bauer et al. [7] argue that HEIs should implement SD as a whole-institution approach in order to encourage transformative practices at all levels. If a given HEI wants to get serious about integrating the idea of sustainability, it should therefore aim to build its capacity to deal with complexity and ambiguity and lean into a more integrative worldview, which will be explained and discussed below.

2.2. Theoretical Background: Worldviews and Systemic Development

To understand complex systems such as universities and the field of higher education in general, it is helpful to build on a framework that is able to structure and explain the evolution of systems in general—the "upgrading of operating systems", as Scharmer coins it (cf. [6]). The psychologist Clare W. Graves developed such a systemic theory from the 1950s to the 1970s, trying to explain adult human development based on his own empirical data [10–12]. Graves distinguished eight distinct worldviews, also described as value systems, phases, or levels, and later, a ninth worldview was proposed. His teachings were made popular in the 1990s by his followers Don Beck and Christopher Cowan under the name of *Spiral Dynamics* [10]. In this paper, we rely mostly on the adaptations by the Dutch researchers Annick Hedlund-de Witt [34–36] and Marcel van Marrewijk [37,38], as well as on our own works [3,16,17,30,39] and on a few popular science works that shed light on practical nuances of the model [5,6,10,40,41].

According to Graves, the development of systems and adults oscillates between worldviews that focus on the individual and worldviews that focus on the collective [10,38]. In simplified terms, whenever societies or smaller groups start focusing almost exclusively on the needs of the collective while suppressing individual interests, individuals tend to fight for their right of self-expression. This will usually lead to more individual freedom which in turn can develop into an exaggerated individualism and a loss of social cohesion, resulting in a renewed call for more considerate behavior and a stronger focus on the needs of the group [10].

This oscillation between the collective and the individual in reaction to environmental challenges is described as an upward spiral development, similar to the double-helix model of DNA [11]. The idea of a spiral development also implies that certain phases or worldviews cannot be skipped—they have to be run through sequentially, though different speeds and depths of development are possible. For example, universities are expected to start from a solid infrastructure and orderly formal settings for teaching and studying, to then go on to building capacities for high quality research and only later to champion transdisciplinary research focusing on solving relevant societal problems. However, when new institutions are founded, they can of course build on the lessons from older institutions and speed their own development—especially compared to the systemic development of universities that were founded in medieval times with a focus on dissemination of factual knowledge.

2.2.1. Four Worldviews: From Traditional to Integrative

Not all of the worldviews or phases of Graves' model are relevant for the discussion of this paper. Following the reasoning of Hedlund-de Witt [35], only four worldviews from Graves' model will be discussed in detail as the preceding three worldviews are more applicable to e.g., tribes, early empires, and mafia-type organizational forms and later worldviews are currently mostly expressed by individuals or in spiritual contexts. The four worldviews in question can theoretically be applied to the development of both individuals, organizations, and societies as a whole. They are, therefore, first introduced in general terms before being applied to the development of HEIs in particular (Section 3) and to different types of inter-organizational networks (Section 4). Table 1 contrasts the four relevant worldviews with respect to their central values and foreshadows their application to HEIs and SD.

Table 1. Overview of four different worldviews 1 .

	Traditional Worldview	Modern Worldview	Postmodern Worldview	Integrative Worldview
guiding principles and values; keywords	order; security; absolute truths; correctness	success; goals; optimization; professionalism; materialism; efficiency; finding good deals	taking care; inclusion; community; individual expression; humanism; awareness; responsibility	finding non-dogmatic solutions; effectivity; relativity; acting in spite of complexity and tensions
purpose/self-concept	creating structure in a chaotic world	optimizing an overly rigid world; designing a functional world	slowing down an exploitative world; creating safe spaces	mending a broken world; integrating all levels
common societal examples	bureaucracy; formalized religion; classic family-owned corporations	shareholder owned corporations; evidence-based medicine	NGOs; stakeholder-oriented organizations (esp. employees and customers)	purpose-driven organizations; self-managed teams
basis for solutions	authority (moral, legal, or patriarchal); rules; goodwill	optimal allocation of resources; process analysis; competition; calculations	collaboration; common ground; community; purpose	competence; synergy; co-creation; purpose
possible negative expression	unfair; bureaucratic; ideologist; unengaged individuals	too egocentric; disrupting cooperation by competing; lost in detail; too opportunistic	ineffective; overly political correct, complacent; allowing only certain aspects of individual expression	too unbound and detached; not empathetic; arrogant
typical topics related to sustainable development	waste management, hygiene and basic health; pasic infrastruture; public policies; understanding and documenting changes in ecosystems; nature protection	resource efficiency and efficient resource allocation; green tech; renewable energies; health; modeling future scenarios	education for sustainable development: externalities; health; agriculture; sustainable economic processes (production and consumption); peace	systems transformation; climate change; green infrastructure; spirituality; enabling and sustaining capacity to act in times of complexity; co-creative democracy
prototypical scientist	objective observer	professional, objective user of scientific method	developer of new methods; hearing the unheard; highlighting pressing global problems	co-creator and broker of effective methods and solutions
		¹ Based on [12,16,36,38	3].	

Sustainability 2020, 12, 3371

The traditional worldview is marked by a focus on absolute truths, rule obedience, and a trust in authorities [38]. This kind of value system laid the foundation for very stable empires and organizations that were able to last for hundreds of years, e.g., the Catholic church and some universities like those located in Oxford, Salamanca, or Heidelberg. It is not by chance that these universities were indeed founded to disseminate absolute truths from a Christian perspective [42]. The traditional worldview enabled societies to adopt a long term perspective, going beyond patterns of instant gratification found in earlier empires, and thus gave birth to enormous projects such as the construction of the Gothic cathedrals that often took several hundred years to be finished. However, its focus on one right way and bureaucratic regulations makes the traditional worldview less able to deal with unplanned incidents and oftentimes prevents people from thinking on their own [10].

The modern worldview can be seen as a reaction to the inherent rigidity of the traditional worldview. Modern science, technology, and neoliberalism are expressions of this way of thinking, which highlights individual effort and the pursuit of success [3]. Indeed, the modern worldview has led to enormous breakthroughs for civilization [43], exemplified by the material wealth of industrial nations and the invention of airplanes, satellites, computers, and advanced medical treatments. However, the rise of the modern worldview has also brought about massive negative ecological and social side effects. When both people and the earth are merely seen as means to an end, ecological and social systems are often subject to abuse and deterioration.

The postmodern worldview can in turn be seen as a reaction to the side effects of phenomena such as neoliberalism, materialism, and consumerism and is often expressed in an attempt to go "back to the roots" and to find more "natural" solutions [10,38]. Trying to make minorities and underserved groups heard, postmodern activists will often take a critical stance toward neoclassical economics and quantitative science. More moderate and forgiving expressions of the postmodern worldview can be seen in the rise of team work and in the focus on customer and employee satisfaction [17]. This type of worldview has therefore brought about more humane, ecological, and considerate procedures and championed topics such as gender equality, diversity, and sustainability [38]. In general, it will favor community building over hierarchy and consensus over individual competence, sometimes leading to inefficient and even ineffective ways of responding to challenges.

Finally, one of the more current developments of society has brought about the integrative worldview (also called synergistic, integral, or systemic) [17,38,44]. The integrative worldview can be framed as a reaction to ever-increasing societal complexity: Issues such as globalization, global migration, and climate change challenge traditional ways of decision making and problem solving and highlight the interconnectedness of various systemic phenomena. As a consequence, individuals and organizations alike are often called to take action in spite of not being able to foresee the probable main and side effects of each action. A certain tolerance of ambiguity is therefore needed in order to maintain the ability to take action [17,38]. Consequently, the management of tensions has been made the center-piece of the integrative framework for corporate sustainability by Hahn et al. [45,46], which was recently applied to the development of HEIs by Annina Lattu and Yuzhuo Cai [33].

The historic breakthrough of the integrative worldview is thus the ability to see both the relative truths and pitfalls of each preceding worldview and therefore being able to act as a broker in times of extremism and societal separation [3,44,47]. Instead of taking a fixed position, the integrative worldview facilitates effective problem solving based on co-creative and competency based methods of innovation, thereby challenging conventional ways of operating businesses, politics, or HEIs [48]. However, when the integrative worldview is only developed in the cognitive line of development and does not include emotional development of the same magnitude, it might lack empathy and thus lead to aloofness and arrogance. Therefore, intention setting and mindfulness are important parts of integrative practices, shifting the focus from "ego" to "eco", as Otto Scharmer puts it [41] (cf. p. 3).

While the descriptions above seem to imply a hierarchy of development and therefore dismiss the modern and especially the traditional worldview, all worldviews are inherently equally important. However, they can be more or less suited to handle different environmental conditions. Furthermore,

worldviews are often combined and can be temporarily adopted depending on the situational context. As these finer points are important for the application of the model, various systemic principles are compiled in the following section.

2.2.2. Principles of Systemic Development

Before the four worldviews are applied to HEIs and inter-organizational networks, we would like to highlight a couple of important principles of systemic development to provide a more differentiated view and to countervail possible misconceptions.

- 1. *Each subsequent worldview is able to handle more complexity than the preceding ones* [11]. Worldviews are adapted to changes in the system's environments. When a system is facing increasingly complex challenges, as HEIs currently do, an upward movement to a more complex worldview might be needed to effectively deal with the novel threats and demands.
- 2. No worldview is inherently "better" than another—different worldviews are simply more or less suited to handle environmental conditions [3]. More complex worldviews enable individuals and organizations to deal with more complexity—which can sometimes be "better", as stated above. However, a person just coming from drug rehabilitation might, for example, require a setting with a more structured and rule-focused worldview (traditionalist worldview) in order to tame a chaotic life and might actually be overwhelmed with the requirements of self-management (integrative worldview). Similarly, a country without a stable educational system might first need to implement more traditionalist forms of education before it starts implementing more integrative practices.
- 3. Subsequent worldviews transcend and include preceding ones [38,40]. For the sake of simplicity, each worldview is illustrated by its prototypical and novel characteristics. However, the lessons and principles from preceding worldviews are not annihilated. For example, when societies moved from more hierarchical and feudalistic structures (pre-traditionalist and traditionalist worldview) to more fluid capitalist democracies (modern worldview), the use of common rules was not suspended but merely reframed: rules were still regarded as important, but more as means to an end (like material success) than as ends in themselves.
- 4. Worldviews might be dominant but not necessarily exclusive [16]. Certain worldviews are often conjoined. For example, leaders of multinational corporations often follow strict metric-based strategies of profit maximization (modern worldview) while promoting team work and focusing on customer-satisfaction on a more operative level (postmodern worldview). Likewise, HEIs can simultaneously be places of more traditionalist teaching (e.g., massive lectures in general undergraduate education) and both modern and postmodern research practices (e.g., stakeholder-oriented mixed-methods approaches).
- 5. Systems are made of subsystems [16]. Organizations are in effect stratified and usually include several worldviews at once [49]. In HEIs, technical, administrative, and scientific staff tend to have their own distinct (world-)view on things. While a more traditionalist rule-focused worldview often suits administrative work, scientists might often gravitate more toward process optimization with regards to their career (modern worldview) and toward a stakeholder-perspective with regards to their students (postmodern worldview). However, the integrative worldview is the first one to not only be aware of different worldviews, but to explicitly acknowledge their respective strengths and shortcomings.
- 6. *Worldviews can be adopted temporarily* [10,40]. While each individual, subsystem, and system is usually centered around one or two dominant worldviews, other worldviews can temporarily become dominant. Indeed, it is common to operate from different worldviews or states of consciousness in different contexts—for example, when talking to a police officer, attending a sport event, making a sales call, or during meditation.

- 7. *Development can go both ways* [10]. When being confronted with changed environmental conditions and threats, systems might react by opening up and embracing complexity or by shutting down and regressing to a perceived "tried-and-true" way. The latter is illustrated by the rise of populistic political parties in Western societies in the face of globalization.
- 8. *A novel worldview will usually contest its predecessor.* When a certain worldview is emerging, its proponents will, from our observations, usually focus on the shortcomings of the preceding worldview to underline why the novel one is more suited to the current environmental conditions. However, both worldviews might later be reconciled and form a functional alliance. As an example, the postmodern worldview found an early expression in a certain kind of environmentalism that not only highlighted the negative side effects of modern market economies (modern worldview), but that also rejected business endeavors in general; whereas today, both worldviews are often combined in ideas such as social entrepreneurship.

These principles are important to keep in mind because models of systemic development are otherwise too easily misinterpreted as judgmental models, separating "the highly evolved" from "the primitive". On the contrary, each worldview represents an important contribution to the development of individuals and societies. The main question is, therefore, not "Which worldview is better?", but "Which feature of which worldview is best suited for the current environmental conditions?".

3. Systemic Development of Higher Education Institutions

As pointed out in the introduction, HEIs and especially research universities are placed under tremendous pressure by a variety of societal trends, creating an environment of enormous complexity [50]. Following the rationale of systemic development and Clare W. Graves' [10,11,51] assumptions, this should lead to minor and major updates to the prototypical worldview of a given HEI. In order to deal with the increasing complexity, HEIs are likely to adopt more and more postmodern and integrative values and practices. But what does that mean for the character of a HEI? In the following, four types of worldviews in the systemic development of HEIs are shortly introduced, following the ideas of Georg Müller-Christ [39], Otto Scharmer [5,6], and Bror Giesenbauer and Merle Tegeler [3]. As these four HEI specific worldviews have historically evolved in a sequential manner, they are also described as phases of HEI development. Furthermore, as these phases can be described as major upgrades to the "operating system" of a HEI [5], they are labelled with numbering from 1.0 to 4.0, following the nomenclature of software development. When abstracted from their historic occurrence and translated into a conception of sustainability governance, they can be framed as representing four distinct conditions along the continuum of sustainability governance proposed by Niedlich et al. [9].

The major characteristics of each phase of HEI development are summarized in Table 2 and described in Section 3.1—with a focus on the conception of an integrative HEI or University 4.0, which is argued to be needed in times of complexity. Section 3.2 will then explore implications for management of sustainable higher education development based on the presented phases of HEI development in Section 3.1 and the principles of systemic development laid out in Section 2.2.2. In order to make these theoretical implications more tangible, Section 3.2.1 provides examples for the central fields of education, research, and governance.

n a worldview perspective. ¹
ron
development f.
EI)
E
institution
ducation j
higher e
of of
ır phases
Tou
5. I
le
ab
H

		Traditional HEI 1.0		Modern HEI 2.0		Postmodern HEI 3.0		Integrative HEI 4.0
general focus on	•	input, authority and hierarchy	•	output, efficiency and competition	•	dialogue with stakeholders and learners	•	systemic solutions, co-creativity and sustainability
education	••••	teacher-centric; the scientist reads his books; memorizing standardized knowledge; learning for recognition and a cademic titles	•••••	test-centric; disseminating factual knowledge, analytic strategies and sound methods; modules and projects; learning for the test, learning as a competitive game for future success	•••••	learner-centric; competencies-oriented transfer of self-reflective knowledge; focus on dialogical seminars and project-based learning; blended learning; learning as personal growth	•••••	system-centric, holistic; whole-person approach; dynamic balance between subject matter, group, individual learners and context; research-based learning; co-creative and mindful learning
research	••••	search for absolute truths; self-concept: observing universal natural laws; focus on strong theories based on both deduction and induction; construction of disciplines	• • • • •	standardization of research; processes and peer-review self-concept: testing and applying natural laws; competition for grants; measurement of success with rankings, impact factors etc.; focus on quantitative methods	•••••	inter- and transdisciplinarity; action research; self-concept: understanding social dynamics; dialogical research processes dealing with societal issues; integration of qualitative research methods	•••	transdisciplinarity; co-creative research; self-concept: co-creating systemic transformation; global action university; living lab approach; focus on real-life solutions; idea of open science
governance, operations and culture	••••	focused on teaching, basic research and reternological transfer; building palaces of knowledge: impressive buildings and extensive libraries; extensive libraries; compliant to regulation, e.g. waste management and safety; one-dimensional approach to sustainability	•••••	focused on quantitative growth; rapid growth in functional buildings with little energy awareness; control of cash flows and process management; entrepreneurial activity; science parks; sustainable development as a management task	• • • • • •	HEI as a place for meeting diverses yet like-minded people; facilitating community and individual expression; diversity management; diversity management goal of climate neutrality; sustainable development as a community task and third mission content	• • • •	HEI as a space for encounter, reflection and inspiration; physical and virtual integration of different societal and ecological systems; whole-institution approach to sustainability; additional fourth mission: co-creation for sustainability
				¹ Based on [3,5,6,9,39].				

Sustainability 2020, 12, 3371

3.1. Four Phases of HEI Development and Their Respective Worldviews

Traditional HEI 1.0. Universities were historically invented from the catholic idea of preserving and teaching universal truths in medieval times [42], which can be framed as an expression of the traditionalist worldview. A supposedly "all-knowing" scholar dressed in academic gown would then read his teachings to relatively passive students, separated by strong disciplines. The impressiveness of large classicist university buildings as palaces of knowledge reflects this kind of focus on authority, stability, and persisting truths [39]. The prototypical HEI 1.0 might seem outdated and yet has succeeded in preserving academic education for several centuries. Indeed, its worldview still influences the ethos of modern universities—based on the systemic principle of *transcend and include* (see Section 2.2.2). By itself, a traditional HEI or University 1.0 is not likely to integrate fast-paced societal change and address cross-cutting topics such as sustainable development (SD) as a whole as these topics demand more openness, dialogue, and at least some degree of interdisciplinarity. However, even traditional universities had to adapt to societal change following the Age of Enlightment and the rise of modern democracy.

Modern HEI 2.0. Universities were reinvented in Germany in the 19th century by Wilhelm von Humboldt and others as research universities [52], which were later adapted into the American model, combining the German research idea with the English collegiate tradition and the American idea of service to society [2] (cf. p. 8). As an early expression of the modern worldview (see Section 2.2.1), it focused on the research *process*, allowing for more fluidity and leading to the idea of continuous improvement and process optimization. The rise of the research university model enabled massive breakthroughs in terms of research methods, standards of publication, and historic innovations for civilization in fields such as technology, engineering, and medicine.

Current academia is mainly shaped by this reinvention of higher education in light of the modern worldview. Quantification, professional specialization, and competition form the basis for most endeavors of HEIs. HEIs 2.0 compete for grants, students, and placements in rankings and thus, in short, for quantitative success [15,53]. This orientation toward quantitative success is amplified by the trends of massification (as participation in higher education growths on national and international levels), globalization, and internationalization and leads to increased marketization and privatization [2,50]. Consequently, teaching has become test-centric and modularized and HEIs have come to adopt entrepreneurial activities. Furthermore, scientific careers can nearly exclusively be advanced within disciplinary niches based on metrics such as numbers of publications and impact factors, hindering the advancement of interdisciplinary fields and cross-cutting topics such as SD [3]. These trends seem to intensify at the moment even though there are parallel lines of developments.

Postmodern HEI 3.0. The main alternative development of higher education is currently shaped by the postmodern worldview, especially in the social sciences and humanities. Dismissing positivism and objectivism, the subjective viewpoints from both research participants and students stand at the center of research and education at postmodern HEIs. Seminars, project work, and qualitative research methods have been developed in the spirit of this HEI 3.0. Learning arrangements are then focused on competencies rather than on knowledge accumulation only [23]. Moreover, forerunners such as Kurt Lewin have introduced alternative approaches to research such as action research [3,54]. These developments occurred together with student movements from about the 1950s, protesting against patriarchal hierarchies of HEI 1.0 and the somewhat mechanical teaching styles of HEI 1.0 and 2.0. This type of postmodern HEI or University 3.0 has brought about a focus on societal issues such as SD and led to the rise of interdisciplinary research. Researchers operating from a postmodern worldview will usually try to make everyone heard and to include regional and international stakeholders [22]. However, researchers often have to play by the rules of the modern worldview of HEI 2.0 in order to advance their careers, leading to trade-offs and tensions on a personal level [3] (cf. p. 645).

Integrative HEI 4.0. As conventional ways of decision making and education (including postmodern dialogical practices) are put under pressure by increasingly fast-paced and complex societal changes in times of globalization and digitalization, some parts of HEIs have come to adopt novel practices in line with the integrative worldview. So far, there are few pure examples for these

Universities or HEIs 4.0. However, experiences from smaller academic projects and other organizational forms (e.g., businesses) operating from an integrative worldview allow for preliminary descriptions of this emerging type of HEI.

Integrative HEIS 4.0 will likely exhibit a focus on self-management, a strive for wholeness, as well as an awareness of their evolutionary purpose, taking responsibility and trying to actively participate in societal change [3,43]. Taking systems as a whole into perspective, the co-creation of effective solutions for pressing societal issues such as SD will be emphasized in HEIS 4.0. Based on Graves' model of systemic development, HEIs or Universities 4.0 are bound to act as brokers for integrative processes, facilitating synergies between different societal sectors.

Building almost exclusively on Russian research and philosophical works, Alla Lapteva and Varlerii Efimov come to very similar conclusions and conceptualize a University 4.0 as "an infrastructure platform" for a variety of activities [55] (p. 2691). Focusing on the technical basis for HEI development, analogous to the waves of industrial revolution, Lapteva and Efimov stress the importance of telecommunication technologies for Universities 4.0. Consequently, HEIs and Universities 4.0 would be expressions of a "cognitive society" [55] (p. 2690), making use of hybrid technologies and collective intelligence [56]. HEIs become, in short, "a very open environment – a hub for a variety of communications, a node at the intersection of multiple networks (...). These communications, research works and development projects involve not only professors and students, but also a wide range of external participants." [55] (p. 2691).

Consequently, new concepts for higher education such as the living lab approach [57] are built around the idea of inclusive and dynamic research processes (see Section 3.2.1). The inclusion of students and citizens in the research process is intended to facilitate deep learning and to link research with education, knowledge transfer, and real-life application [58]. And even without direct field testing of ideas, research-based learning and co-creative innovation methods can be applied in courses. At the very least, learners should be encouraged to personally engage with sustainability and to learn by *experiencing* and *becoming aware*, going beyond mere cognitive processing [59].

Emphasizing the transformative aspect of HEI development, Otto Scharmer and Katrin Kaufer propose that learning at integrative HEIs 4.0 will be shaped by action learning, global classrooms, innovation hubs, and individualized lifelong learning journeys [60]. Scharmer goes on to argue in two blog posts that the university of the 21st century should in essence focus on providing *vertical development literacy*, i.e., the ability to understand systems and their respective worldviews and to guide these systems through a systemic upgrade, if needed [6]. According to Scharmer, this leading of transformative change requires the skill of "deep listening", self-awareness, and compassion [5].

Similarly, Uwe Schneidewind [61] proposes that HEIs should focus on facilitating *transformative literacy*, i.e., the ability to understand and participate in social transformation, including the technological, economic, institutional, and cultural dimensions of transformation. Sustainable development will then supposedly not be a special topic to deal with, but an integral part of a HEIs' DNA and governance [3,62]. Interestingly, the qualitative data from a multi-case study by Niedlich et al. [9] suggest a linear relationship between the orientation toward organizational learning of a given HEI and the degree of holistic orientation of its sustainability governance, supporting the general assumption that the development of HEIs from 1.0 to 4.0 equals a general systemic upgrade, going beyond incremental and isolated updates.

3.2. Leading Multi-Level Development of HEIs

The presented four phases of systemic HEI development are intended to provide a map for navigating the transformation of HEIs in the 21st century. In times of increasing complexity and a "knowing-doing gap", as Scharmer [5] coins it, HEIs would do well to prepare for a systemic upgrade in order to keep up with societal demands, specifically the challenge of sustainable development. Currently HEIs around the world are at very different stages of development, described as stratification of higher education [63]. We would go one step further and argue that each HEI in itself is stratified as

is, as different organizational subsystems emphasize different worldviews and exhibit different levels of maturity within a given worldview. Transformation would thus necessitate a consciously chosen multi-level approach.

However, this systemic upgrading is no easy task, especially with the principles of systemic development in mind (see Section 2.2.2). For example, how does one ensure and maintain the quality of teaching and testing while experimenting with new forms of education? How does one decide what to keep from the "old" system and how to transcend it in light of newly adopted core principles? And how should one address different subsystems of a given HEI? As these questions do not have a fixed answer, change agents might profit from engaging in peer networks (see Section 4).

At the very least, HEIs should practice self-reflection, open up to societal discourse, and prepare for necessary change, if judged to be appropriate. This includes that not all subsystems have to embrace complexity and transdisciplinary research. If, for example, a molecular biologist finds that robust quantitative methods (based on the idea of falsification and statistical inference) and exclusive discourse within their disciplinary niche are still the most fitting approaches, then change agents would do well to support them in continuing that type of research. At the same time, the communication of research results and processes might be adjusted by publishing in open access journals and including students in the research process to facilitate research-based learning. Similarly, research questions connected to hot topics such as climate change might be favored.

As most HEIs are expected to be centered around the traditional 1.0 or modern 2.0 worldviews, the next step would likely be one of strengthening the ideas of quality control (traditional worldview) and process optimization (modern worldview), while also championing dialogical forms of research and education (postmodern worldview). Inspiration for taking the predominant HEI 2.0 one step further can be found in J.G. Wissema's *3GU* model [64], mainly with regards to restructuring HEIs—enabling more interdisciplinary research in institutes, professionalizing HEI management, and promoting entrepreneurial activities and outreach.

In sum, the integration of more participatory, open, and transdisciplinary practices should be encouraged at all levels while simultaneously consolidating methodological rigor and effective process management—rethinking prevalent ways of HEI organization. In this way, a University or HEI 4.0 will become an infrastructure platform for cross-sectoral communication, facilitating open science and co-creative problem solving.

3.2.1. University 4.0 Practices for Education, Research, and Governance

Even though the systemic development of a given HEI can hardly be prescribed generically, some examples for different fields of action might be helpful to inspire customized measures. Thus, we will present a couple of loosely chosen best practices for the areas of education, research, and governance.

Education. As HEIs will move toward a larger proportion of postmodern and integrative practices, education will likely be focused on a whole-person approach, developing (personal) competencies in co-creative settings, rather than on knowledge and methods only [65]. One didactic approach that has been proven vital for leading co-creative seminars with learning groups of all levels is Ruth Cohn's *Theme-Centered Interaction* (TCI). TCI offers a clear framework for dynamically balancing different factors of group learning—namely, the subject matter ("it") at hand, the needs of the group ("we"), the needs of each individual ("I"), as well as the demands of the specific context (the "globe") [66] (see Figure 1 for a display of the TCI model). Although it was originally developed for facilitating group therapy sessions, it has successfully been applied to classrooms in both primary, secondary, and higher education (see the work of Sylke Meyerhuber et al. [67] for its application to higher education). Its goal is to promote what Ruth Cohn called "living learning", a kind of learning which is based on the humanistic ideal of meeting the psychological needs of the group members in order to facilitate personal growth [66] (cf. p. 19). Living learning is further described as being "emancipatory in nature" and thus, supporting self-leadership; as viewing the group as a "source of learning"; as being led by teachers that are also co-learning; and as being oriented toward meaningful experiences

and behavior [68] (p. 142). Due to its holistic nature, this approach seems well suited for facilitating transformative learning, especially education for sustainable development.



Figure 1. The four-factor model of Theme-Centered Interaction (TCI): group facilitation in dynamic balance (source: [66] p. 20).

From our experience, the Theme-Centered Interaction (TCI) approach is especially helpful for dealing with emerging topics and for integrating all kinds of disturbances (e.g., personal irritation, side talks, group conflicts, or arising needs) that would otherwise negatively impact the learning process [69]. Moreover, it accounts for both the relative autonomy and interdependence of human beings [70]—facilitating mutual respect and self-responsibility at the same time—and thus, sets the stage for co-creative learning and working. As a principles-based approach, it can theoretically be used with all kinds of group settings, regardless of the underlying worldview, and seems to be a promising approach for gently paving the way toward integrative education of HEIs 4.0. HEI leaders could therefore offer personnel development courses to lecturers of all levels on the TCI approach to facilitate a systemic upgrade of teaching methods. Furthermore, HEIs could initiate courses or information material on other didactical topics such as competency-based learning, blended learning, whole person education, research-based learning, and education for sustainable development (see e.g., [71,72]).

Research. The transdisciplinary *living lab* approach illustrates how a University or HEI 4.0 can attempt to contribute to sustainable development through novel research practices. The basic idea is to "leverage the campus as a test bed for sustainability" [73], integrating faculty, staff, researchers, and students into the process and using rapid prototyping methodologies for finding local solutions for global sustainability challenges. Living labs are moreover described as "open innovation ecosystems based on a systematic user co-creation approach" and have similarities to approaches such as real world laboratories, urban transition labs, and transformation labs [57] (cf. 32). In essence, they are focused on real-world application and effective problem solving through collaborative and open research. As such, living labs are not confined to the research category, but strongly interlink research with education, campus operations, and outreach, mirroring the cross-sector collaboration that is needed for promoting sustainable development. However, while the living lab approach is still emerging and preparing the ground for the integrative HEI 4.0, HEIs should not forget to also strengthen sound research practices on all levels, for example, by supporting open access publication or by prescribing study pre-registration to fight *p-hacking* [74,75].

Governance. As the example of living labs goes to show, governance for sustainable development will increasingly be framed as a community task with the advent of postmodern HEIs 3.0 and integrative HEIs 4.0. According to Niedlich et al. [9] (p. 11), HEIs will then describe their purpose as that of being a change agent and not merely a knowledge producer, closely tied to civil society and other external actors. In this way, HEI management becomes a part of the community. Furthermore, sustainability governance will increasingly be based on a whole-institution approach and facilitate cross-sectional and inter-organizational decision-making processes. This includes pursuing multiple dimensions of

sustainability and other cross-cutting issues simultaneously in an ambidextrous approach [9] (cf. p. 14). Living labs are one possible expression of integrative sustainability governance at HEIs 4.0, especially at older institutions. However, smaller and more rural institutions might be better able to make the transition toward postmodern and integrative practices of sustainability governance than larger institutions with a long history of traditional and modern practices.

In essence, the development of higher education can be described as a process of progressively opening up to stakeholders of all kinds and of integrating participatory methods. This movement of opening up enables HEIs to keep up with societal change and to provide learning arrangements that not only address cognitive development, but also emotional and even moral and spiritual development.

Thus, the main strategy to deal with increasing complexity is argued to be found in the facilitation of systemic development toward more integrative approaches, while trying to consolidate lessons and standards from preceding phases. As HEIs are invariably stratified organizations including a multitude of subsystems and worldviews, this strategy should be consciously adopted as a stratified or multi-dimensional approach, enabling each subsystem of a given HEI to take the next necessary step or to strengthen what is working well. In other words, HEIs should be wary of a one-size-fits-all approach and address each subsystem independently, allowing for stratified, parallel development with a general tendency toward integrative practices, as reflected in the whole-institution approach for sustainability governance and the broader model of University or HEI 4.0.

As the complexity of this task could easily overwhelm local change agents, we have found that this main strategy is often supported by inter-organizational networking among peers. Moreover, inter-organizational collaboration can be seen as a central organizational element of the University 4.0 model. Therefore, inter-organizational networking will be discussed in the following section as the second, more specialized strategy for dealing with increased complexity.

4. The Role of Inter-Organizational Networks for Higher Education Development

To reiterate, if higher education institutions (HEIs) are willing to meet the complex challenge of sustainable development (SD), they would do well to raise their general capacity to deal with complexity. This strategy of living up to SD by upgrading the metaphorical operating system of a given HEI has been argued to be the primary strategy of choice for sustainable HEI development. However, if dealing with evermore complexity can be described as a process of opening up to various stakeholders and thus increasing inter-organizational collaboration, how can this process be designed and formally supported?

One of the more tangible ways of inter-organizational communication and collaboration is that of participating in specialized networks, especially for cross-cutting topics such as SD. The second strategy to deal with increasingly complex demands is therefore described in the following as the strategy of inter-organizational networking. As Kurt de Wit [76] and Jeffrey Selingo [77] point out, successful HEIs are increasingly networked and work in alliances to tackle common challenges. Their networking activities include both internal and external stakeholders. In this paper, we are going to focus on the latter.

Scientific conceptions of networks come in many shapes and sizes. One dominant line of research views basically all relationships as part of a bigger network of relationships that make up society as a whole [78]. This perspective is built on the idea that "networks do not have boundaries" [79] (p. 1039) and forms the basis for quantitative network analysis and the famous "small worlds" theorem.

However, another line of research uses the term network to describe informal patterns of coordination that are based on personal relations and that bypass formal ways of decision making [80]. Thus, from this point of view, networks are seen as a *special* form of social coordination and are usually linked to phenomena such as cliques, corruption, and insider relationships [78].

Both lines of research have in common that they highlight the power of personal relationships for the distribution of information and resources. In this paper, we are assuming a middle ground position: on the basis of the works of Heiko Kleve [81,82], we regard networks as both formal and informal patterns of coordination that cut across other forms of organizations in highly specialized

societies, such as higher education institutions or enterprises, or even sectors such as politics and higher education in general. Networks thus fulfill the function of bridging units that are functioning separately from one another in order to solve problems that are hard to solve for each unit separately. In this sense, phenomena such as public-private partnerships can be seen as networks [83].

According to Kleve, networks are formed when highly differentiated expert organizations (e.g., hospitals, enterprises, or HEIs) are not able to solve current issues on their own or when the contribution of formal ways to problem solving is doubted [82] (cf. p. 354). In line with the sociological systems theory of Niklas Luhmann [84], networks are characterized by the principle of personal reciprocity, i.e., by the binary code of giving and taking [82] (p. 360). In other words, the power of networks relies on trust, the principle of mutual exchange, as well as on mutual expectations.

In higher education, inter-organizational networks are formed to enable flow of information and problem solving across organizations, usually with respect to specialized administrative and operative topics like that of digitalization, international recruitment, legal affairs, or online education [77]. Furthermore, cross-cutting research topics such as sustainable development or advancement of research methods can also be the focal point of inter-organizational networks, connecting professionals experiencing similar challenges.

Building on Kleve's conception, the success of inter-organizational networks should be dependent on the degree of exchange (and thus, giving and taking) within the network and on the degree of perceived reduction of complexity among its members. Ideally, both factors would help members with solving issues—both complex and trivial—at their respective organizations. However, we propose that the prototypical conception of a network differs greatly depending on the worldviews of the participating members, be they individuals or organizations. Hence, if network managers want to initiate lively exchange and to reduce the perceived complexity of their network's members, they should account for different expectations and worldviews.

In Section 4.1, we will thus bring forward different conceptions of networks on the basis of Graves' worldview model (see Section 2.2) and discuss implications for network management, especially in the context of higher education and sustainable development. The theoretical exploration is followed by a short case study in Section 4.2 on how the presented ideas were applied in the network "HOCH-N". The project name HOCH-N stands for "Sustainability at Higher Education Institutions". The project was funded by the Federal Ministry of Education and Research (BMBF), Germany to conduct research on sustainability at HEIs from the view of a whole-institutional approach and to build a network of HEI members that engage in SD [4,85]. This project serves as an empirical back-drop for the presented ideas. The authors are part of the HOCH-N network management team and thus have had the opportunity to compare theoretical ideas on networks with practical experiences.

4.1. Network Management from a Worldview Perspective

When HEIs are increasingly seeking out inter-organizational networks in order to tackle complex cross-cutting issues, the question has to be answered how these networks should be managed. As pointed out above, the term network is used to describe social phenomena of very different scope and sizes—a circumstance that complicates the conscious initiation and coordination of networks. If all relationships are part of a large invisible network, how could networks possibly be managed? Or if networks are seen as informal corrupt cliques, is their use not immoral *per se*?

However, combining the idea of networks as inter-organizational patterns of giving and taking with the idea of systemic worldviews, the following table and paragraphs are intended to highlight different conceptions of networks and principles of network management. From a worldview perspective, four types of networks are relevant for the advancement of HEIs or other expert organizations (see Table 3). These four types could be preceded by a fifth type of network, but as this one would indeed be marked by the aforementioned bypassing of formal decision making (and therefore, disposed to corruption), it is not included as a viable or advisable basis for deliberate network management.

worldview perspective.
a
from
of networks
types
Four
able 3.
Г

	Network Type 1.0	Network Type 2.0	Network Type 3.0	Network Type 4.0
label	traditionalist	modern	postmodern	integrative
motivation to participate in a network	staying in synch with the herd; standing and fighting together	staying in touch with the best; finding an audience; bartering of ideas with peers; reaching influential people	meeting like-minded people; finding support and supporting others; sharing challenges with peers	sparking and finding inspiration; co-creating solutions for common challenges
conception of giving	administering; providing; offering; depositing	investing (ROI)	tending needs; paying it forward	sharing gifts; helping to build a thriving system
taking is seen as	justified (based on principles and liability)	a reward and the point of the exchange	selfish or as a cause for feeling indebted	an opportunity for growth; a means for building something bigger
character	brotherhood/lobby	competitive game/market	campfire	jazz session
types and characteristics of meetings and events	official, regular, structured, and orchestrated	big, output oriented, scientific conference with shorter sessions and market of ideas	conference or camp with workshops; organic and plant-based food; cosy	hubs, labs, and camps; leaving space for experimentation
implications for network management	focus on stipulating standards; political work; adopting resolutions; strict admission criteria; clear network structure with committees, boards etc.; communication with authorities in the field	focus on high quality input and visibility, highling best practices, organizing conferences; enabling exchange and opportunities for presentations;	focus on creating a safe space to discuss common problems; strengthening bonds; organizing workshops; sustainable catering; providing resources	focus on enabling co-creation and innovation; holding the space; highlighting creativity methods and inspiring solutions; open source and sharing culture
examples from HOCH-N	three distinct levels of membership (from individuals to institutions, criteria based), practitioner's guides; support by federal ministry; consulting committee with experts	practitioner's guides with toolboxes and best practices; collection of more than 700 best practices on digital map; scientific and networking conferences	stakeholder-oriented interdisciplinary research; practitioner's guides; regional networking hubs; digital map of sustainability researchers in Germany; free resources	co-creation hubs with innovative methods (LEGO Serious Play®), systemic constellations, Theory U) focused on systemic transformation and whole-institutional approach

Network type 1.0. When a network is characterized by more traditional traits (see Section 2.2), it will likely be focused around the idea of finding and voicing common standards and concerns. Likewise, traditionalist network members might participate in a network in order to stay in synch with their peers and to comply with authorities in the field. If a network manager considers her network to be a more traditional network type 1.0, she could thus focus on more formal rules for admission and meetings, on providing orientation for its members, and on enabling communication with experts and authorities in the field.

Network type 2.0. A network type 2.0, based on the modern worldview, would in contrast be geared toward friendly competition among its members, seeking to learn from the best, and to find an audience themselves. Giving and taking are then seen more as a transaction and networking itself as an opportunity for marketing oneself or ideas. In order to fulfill the expectations of a network that is best characterized in terms of the modern worldview, network managers should then focus on providing plenty of opportunities for the presentation of ideas and personal exchange. Furthermore, it could be well-advised to champion best practices and to favor shorter and frequent sessions during events. With respect to higher education development, typical themes might include resource efficiency, quantitative methods, mainstream quantitative science, and digitalization.

Network type 3.0. When networks are formed around topics of social or environmental issues, they tend to focus much more on community building and workshops than on career advancement or efficient presentation of research results. A network type 3.0 is in essence built around the idea of mutual support and sharing, and giving and taking are likewise seen as a matter of tending needs and humble receiving. Its members might, therefore, expect more workshop-like formats during events, freely available resources, and low admission criteria. The topics of sustainable development or advancement of qualitative and transdisciplinary research lend themselves quite naturally to these postmodern network types 3.0. Network managers of more postmodern networks would therefore be well-advised to focus on dialogue, transparency, and responsible event hosting (especially with regards to catering).

Network type 4.0. When networks are more centered around the integrative worldview, they tend to be less predictable in their choice of tools and structures as their main focus lies on flexible problem solving and co-creative work with open outcomes. The main driver of a network type 4.0 seems to be the challenge of complexity and the desire to go beyond dogmatic and pre-defined solutions. While integrative networks might commit to a certain type of method (such as systemic constellations, *LEGO Serious Play*®, or methods from *Theory U* or the *Art of Hosting*), their key feature is that of co-creative work in the present moment—which can be likened to a jazz session, building both on a strong base of competence and on common standards while striving on present-moment interaction and improvisation.

In that regard, tools might be important for a network type 4.0, but only on the basis of mindful application and benevolent intentions as giving and taking are seen as a matter of sharing gifts. To set the stage for mindfulness, integrative networks often include relaxation techniques, meditation, or other practices from spiritual traditions. A network manager of an integrative network type 4.0 thus needs to fill the role of a facilitator, setting and holding the space for safe sharing of ideas. Because this type of interaction is rather atypical in mainstream scientific communities, integrative network events are usually either smaller-scale events (often labeled with alternative terms such as hubs, labs, or camps) or sub-tracks at larger, more typical scientific events.

Taking everything together, several distinct types of networks can be described through the lens of Clare W. Graves' worldview model. On the basis of the principles of systemic development laid out in Section 2.2.1, it is, however, unlikely that a network or its members will be a pure representation of one of the four network types. Nonetheless, network managers should be aware of the values and worldviews that are most prevalent in their respective field in order to promote coherent structures, facilitating the problem solving capabilities of networks. When the character of a network is experienced as incoherent to its contents, it puts the integrity and ultimately the liveliness and effectivity of the network at risk.

All in all, we propose that the design of a network should be intended to roughly match the worldview of a network's members and, in short, meet the network's participants where they are. This will most likely lead to a multi-level approach, serving multiple worldviews at once, while stressing one or two in particular. A bias toward inclusion of integrative approaches (network type 4.0) might furthermore be called for when a network is dealing with a particularly challenging issue in order to facilitate effective and co-creative problem solving.

4.2. Case Study: Managing the HOCH-N Network

Some examples from the HOCH-N network on sustainability in higher education in Germany are to illustrate the application of the presented ideas and to find the first answers to research question B2: How can inter-organizational networks contribute to the integration of SD into HEIs and to the general systemic development of HEIs? The case study is intended to explore practical application of the aforementioned conceptions (Section 4.1), thereby inspiring self-reflection in the reader (cf. [13,14], see Section 1). Hence, it demonstrates how key structures, products, and services of the HOCH-N network are set up to serve different worldviews and to promote systemic transformation.

Being part of the HOCH-N network management team, we could build our analysis on both publicly available resources and on internal documents and discussions. As the first step, we identified the most pivotal aspects of the network's design and outputs. Thereafter, we analyzed both the conceptualization and realization of those aspects on the basis of Graves' model (see Sections 2.2.1 and 3.1) and its application to networks (see Section 4.1), identifying (a) main drivers of participation in the HOCH-N network and (b) the respective worldviews that are predominantly served.

Case Study Results

The general design of the HOCH-N network is mostly based around postmodern ideas (network type 3.0) as the HOCH-N network is intended to promote the inclusion of sustainable development into HEIs' practices and curricula and the topic itself is typically associated with a postmodern worldview. This includes free membership and resources, low admission criteria, stakeholder inclusion, and a focus on smaller regional events (in order to facilitate personal exchange and to minimize travel costs and carbon footprint). However, in order to make the network more resilient and agile, the core structures, tools, and products of HOCH-N are designed to serve multiple worldviews at once, building on our own ideas and on empirical research on networks of HOCH-N network members [86,87].

To begin with, as HOCH-N is touching issues of both higher education politics and HEI governance, it has come to adopt certain traits of a more traditional network type 1.0. For example, three distinct levels of membership were defined with different barriers of entry. On the lowest level, each interested individual can participate. On the second level, members of HEIs can join HOCH-N individually, as long as they have attended a HOCH-N event. Moreover, level two members appear on the digital map (see below) and have to provide three best-practice examples concerning SD from their respective HEI as well as a short motivational statement for their membership. On the third level, HEIs as a whole can join HOCH-N as long as they demonstrate their engagement for SD with examples in six different fields of action, have the explicit support of leadership, and fulfill further structural criteria (for example, publication of a sustainability report). This three-level membership structure is intended to serve both the need for hierarchy and structure (traditional network type 1.0) as well as the disposition toward community and inclusion (postmodern network type 3.0).

The most engaging feature of the HOCH-N network is probably to be found in the networking events for sustainability at HEIs, enabling personal networking, exchange of concepts and ideas, and collaborative work on common issues. As the project evolved, several types of events have proven to be vital. As we identified community building (network type 3.0) and innovative problem solving (network type 4.0) to be the main drivers for HOCH-N members to participate, the typical HOCH-N events have turned more and more into so called hubs with either a focus on regional networking or

innovative methods. The latter are usually smaller-scale events or sub-tracks with a focus on systemic methods (like LEGO Serious Play®, systemic constellations [88], or methods from Theory U [41]) in order to build the needed capacity to deal with complexity (see Sections 2 and 3). This does not mean that more standard event formats and presentations of scientific results are not valued—these formats merely play a lesser role as the topic of SD is less linked to mainstream scientific careers.

Going beyond physical events, one of the key features of HOCH-N's website is the digital map of sustainable development at German HEIs (see Figure 2), programmed and maintained at the University of Hamburg [89,90]. As all level two members have to contribute a personal picture, a motivational statement, and have to highlight three examples of engagement for SD at their HEI, the map is able to fulfill several functions at once. First of all, it is intended to facilitate bilateral exchange via personal profiles as a kind of who-is-who of sustainability champions at German HEI (fitting network types 2.0 and 3.0). Moreover, it is also a rich collection of best-practices, with currently more than 700 entries from 250 individuals from 133 different German HEIs (again fitting network types 2.0 and 3.0), representing roughly a third of all German HEIs. The map is currently being re-programmed to include a novel search and database feature, enabling users to display, for example, all best-practices dealing with education for SD or SD governance (see [90] for the URL).



Figure 2. The HOCH-N digital map of sustainable development at German higher education institutions. Combs represent network members and link to individual profiles [90].

As HOCH-N was not only set up as a network but also as a research alliance, the project has also produced substantial results in six different fields of action: reporting [91], governance [7,9,92,93], teaching & education [94,95], research [62,96], operations [97,98], and transfer [99,100]. The six corresponding work packages have published practitioner's guides (see references above), available free of charge as print versions and PDFs (currently, only the guide on sustainability governance is available in English). These guides represent a major milestone in advancing SD at German HEIs as they help to provide interested but inexperienced members of HEIs with overviews, checklists, guidelines,

and inspiration for SD at HEIs. The practitioner's guides lend themselves to all four network types and thus help to ensure the success of the network. All guides will also be turned into short online video courses and merged on a common homepage to reflect the whole-institution approach, in line with the integrative worldview.

All in all, the HOCH-N network is intended to facilitate lively giving and taking, for example, by providing free resources and practitioner's guides, networking events, and by involving members and other stakeholders in the development of said resources. Its main focus is on serving the postmodern and integrative worldviews with networking events, innovation hubs, the digital map, and with practitioner's guides, while at the same time also serving and respecting both the traditional and modern worldview. Most key features were therefore designed to reflect a wide variety of needs. In this way, the HOCH-N networks tries to strengthen local change agents at HEIs and to provide them with inspiration through best practices, handbooks, and innovative methods, gently nudging systemic transformation toward a whole-institutional approach of SD.

5. Discussion

As members of the HOCH-N project, our work is focused on supporting HEIs in integrating sustainability into their education, research, governance, operations, reporting, and transfer, implementing the whole-institution approach. However, HEIs around the world are also facing the difficult task of dealing with multiple complex trends simultaneously while trying to maintain the ability to take action in times of uncertainty. Consequently, the implementation of measures for issues such as sustainable development often falls short. Otto Scharmer summarizes the challenge of implementing sustainable development in HEIs and society in general:

"The difficulties we have in meeting today's global challenges, such as implementing the 17 Sustainable Development Goals (SDGs) worldwide, are not caused by a knowledge gap. We have all the knowledge we need. The problem is a knowing-doing gap: a disconnect between our collective consciousness and our collective actions. (. . .) To address these issues at their root requires two things: new platforms for cross-sector co-creation and an upgrade in the operating system that people use to collaborate—practices that facilitate a shift from ego-system to eco-system awareness." [6]

Similarly, we have proposed two main strategies for HEIs for dealing with the challenge of complexity: (a) promoting systemic development of a HEI by updating its self-concept and worldview, enabling the HEI to deal with more complexity, and (b) participating in inter-organizational networks in order to promote transformative capabilities and problem solving across organizations. Both strategies combined are argued to enable HEIs to deal with increasing levels of complexity in general and the challenge of sustainable development in particular.

In essence, the systemic development of higher education is laid out as one of progressively opening up to internal and external stakeholders and thus, embracing interconnected and open environments. Ideally, this enables HEIs to keep up with societal change and to deal with increasing complexity, gradually leaving professional silos behind and allowing co-creative problem solving and collaboration to occur.

As HEIs are stratified organizations consisting of diverse subsystems, this strategy should be consciously adopted as a multi-level approach, enabling each subsystem of a given HEI to take the next necessary step—or alternatively, to strengthen best practices at the current level of development. Consciously developing a stratified university or HEI following the integrative University 4.0 model requires the ability of change agents and leaders to tolerate ambiguity and tension, appreciating the value of each distinct worldview of each subsystem while still trying to plant the seed of systemic growth. It is thus dependent on the personal development of leaders and their ability to embrace uncertainty, as qualitative research by Barret C. Brown suggests [47]. Brown further concludes that in order for global sustainability to be achieved, leaders would need to build their decisions (a) on a "deep inner foundation"; (b) on the conscious application of systems theory, complexity theory,

or integral theory; and (c) on mindfully responding to emerging topics by following an "adaptive design management" approach [47] (cf. 566).

As HEIs have to manage several complex issues at once, it might be argued that the simultaneity of trends would hinder general systemic development. However, some of these trends could actually accelerate systemic development by challenging predominant worldviews and processes. Specifically the trend of digitalization might indeed create windows of opportunity for systemic upgrades, as Giesenbauer [101] argues. For example, digital administration tools might facilitate the development from HEI 1.0 to 2.0 by promoting transparent process management; interactive tools might moreover facilitate the development from HEI 2.0 to 3.0 by increasing participation, stakeholder engagement, and transparency; and new tools for networking and collaboration might finally facilitate the development from HEI 3.0 to 4.0 by enabling inter-organizational and co-creative work processes.

The latter aspect hints at a particular feature of integrative HEIs 4.0 as those are expected to be highly networked. Organizational psychologist Peter Kruse [102] proposes that the challenge of rising complexity can only be successfully managed when the density of network connections in a given organization are increased significantly in order to tap into the organization's inherent problem solving capacities and intelligence. Analogous to neural networks, inter- and intra-organizational networks thrive on resonance and exchange of information in order to create viable solutions. Once different members are connected and resonate with each other, co-creative solutions can potentially emerge. Thus, a higher network density increases the chance of effective problem solving.

In this sense, inter-organizational networks play an important role in enabling HEIs to promote systemic development. At the very least, network managers should reflect the worldviews and expectations of their networks' members in the design of their network in order to assist fruitful exchange. Ideally, they would additionally reflect how the systemic development of their members' organizations is likely going to unfold and design their network in a way that facilitates and even accelerates further systemic upgrades. The case study of the HOCH-N network illustrates that different structural features, events, and products can be used to serve different worldviews in order to build an agile network.

The case study is not without limitations. First and foremost, the authors are part of the network management team themselves and thus, are possibly biased. Furthermore, a single case delivers mainly anecdotal evidence. At the same time, the case study goes to show that accounting for different worldviews can help to inform the conception of a network, strengthening its strategic flexibility. Theoretically, this should improve a network's impact and help its members with the task of systemic transformation. However, while we are confident that HOCH-N supports the systemic transformation of HEIs in Germany toward a whole-institution approach and a University 4.0 model, these effects are difficult to operationalize and quantify. Further research, both qualitative and quantitative, would be needed to substantiate our findings and convictions.

Moreover, even while cross-sector collaboration might be needed for effective societal solutions, its implementation can be challenging, especially with regards to different narratives and organizing principles across sectors [103]. Thus, the diversity of inter-organizational networks does not only represent an asset for co-creativity, but also a liability for effective coordination. Network managers should therefore keep an eye on communication practices and trust levels within their network as co-creative collaboration is dependent on both.

6. Conclusions

How can higher education institutions become change agents for sustainable development while also having to respond to increasingly complex demands simultaneously? These demands include massification, globalization, marketization, and digitalization. The pressure is further intensified by increased competition for grants and placements in rankings. In order to deal with the increased complexity, two main strategies can be identified: (a) promoting general systemic development of a given HEI (upgrading the HEI's "operating system"), progressively opening up to various stakeholders and focusing on co-creative collaboration, and (b) participating in inter-organizational networks in order to find inspiration for dealing with challenging trends. The multi-dimensional and networked organizational model of an integrative University 4.0 is argued to be suitable for following both strategies and consequently managing complexity.

Building on Clare. W. Graves model of systemic development, both general systemic development as well as the development of HEIs and inter-organizational networks can be described with a single coherent framework, supplying change agents with a navigational tool for systemic development. Further research is needed to elaborate on the presented ideas and to test their practical application.

All in all, our research suggests that it would be advisable for HEI management to build on the idea of a University 4.0, focusing on increased collaboration, co-creativity, and general systemic development. If higher education institutions are to contribute to mastery of pressing societal issues such as climate change and global migration, they should therefore reflect their dominant worldviews, making sure that they are not unnecessarily stuck in practices that were developed during very different times—e.g., to disseminate absolute truths in pre-democratic societies (HEI 1.0) or to standardize the scientific process in the Modern Age (HEI 2.0). While these phases were necessary for the development of higher education and their best practices still hold value, HEIs should progressively increase collaboration with stakeholders of all kinds in the generation of both knowledge and solutions, requiring increased intra- and inter-organizational network density.

In this way, adopting the concept of an integrative HEI or University 4.0 could enable HEIs to deal with the complex demands of the 21st century. Moreover, network managers could support the systemic transformation of HEIs by designing integrative inter-organizational networks that are focused on reducing perceived complexity and inspiring lively exchange. Both strategies combined might help HEIs to live up to the challenge of sustainable development and to become change agents for a sustainable future.

Author Contributions: Conceptualization, B.G. and G.M.-C.; writing—original draft preparation, review and editing, B.G.; supervision, G.M.-C. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the German Federal Ministry of Education and Research (BMBF), grant number 13NKE007A.

Acknowledgments: We would like to thank our reviewers for their considerate and helpful comments.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Barth, M.; Rieckmann, M. State of the Art in Research on Higher Education for Sustainable Development. In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge; Earthscan from Routledge: London, UK, 2016; pp. 100–113.
- Altbach, P.G. The complex roles of universities in the period of globalization. In *Higher Education in the World* 3–Higher Education: New Challenges and Emerging Roles For Human and Social Development; Global University Network for Innovation, Ed.; Palgrave Macmillan: London, UK, 2008; pp. 5–14.
- Giesenbauer, B.; Tegeler, M. The Transformation of Higher Education Institutions Towards Sustainability from a Systemic Perspective. In *Universities as Living Labs for Sustainable Development*; Leal Filho, W., Salvia, A.L., Pretorius, R.W., Brandli, L.L., Manolas, E., Alves, F., Azeiteiro, U., Rogers, J., Shiel, C., Do Paco, A., Eds.; Springer International Publishing: Cham, Germany, 2020; pp. 637–650. ISBN 978-3-030-15603-9.
- Bassen, A.; Schmitt, C.T.; Stecker, C. Nachhaltigkeit an Hochschulen: Entwickeln–vernetzen–berichten (HOCH-N). uwf UmweltWirtschaftsForum 2017, 25, 139–146. [CrossRef]
- Scharmer, C.O. Vertical Literacy: Reimagining the 21st-Century University. Available online: https://medium.com/presencing-institute-blog/vertical-literacy-12-principles-for-reinventing-the-21stcentury-university-39c2948192ee (accessed on 2 May 2019).
- Scharmer, C.O. Education is the kindling of a flame: How to reinvent the 21st-century university. Available online: https://www.huffingtonpost.com/entry/education-is-the-kindling-of-a-flame-how-to-reinvent_us_ 5a4ffec5e4b0ee59d41c0a9f (accessed on 25 March 2019).

- Bauer, M.; Bormann, I.; Kummer, B.; Niedlich, S.; Rieckmann, M. Sustainability Governance at Universities: Using a Governance Equalizer as a Research Heuristic. *High. Educ. Policy* 2018, 31, 491–511. [CrossRef]
- 8. Mader, C.; Scott, G.; Abdul Razak, D. Effective change management, governance and policy for sustainability transformation in higher education. *Sustainability* **2013**, *4*, 264–284. [CrossRef]
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *Higher Educ. Q* 2019, 1–18. [CrossRef]
- Beck, D.; Cowan, C.C. Spiral dynamics. In Mastering Values, Leadership and Change: Exploring the new Science of Memetics; Blackwell: Malden, MA, USA, 2006; ISBN 9781405133562.
- 11. Graves, C.W. Human nature prepares for a momentous leap. Futurist 1974, 8, 72-87.
- 12. Graves, C.W. Levels of Human Existence. Transcription of a Seminar at the Washington School of Psychiatry, October 16, 1971; Lee William, R., Ed.; ECLET Publishing: Santa Barbara, CA, USA, 2002; ISBN 097247420X.
- Wals, A.E.J.; Walker, K.E.; Corcoran, P.B. The Practice of Sustainability in Higher Education: A Synthesis. In *Higher Education and the Challenge of Sustainability: Problematics Promise, and Practice*; Corcoran, P.B., Ed.; Kluwer, Acad. Publ: Dordrecht, The Netherlands, 2009; pp. 347–348. ISBN 1-4020-2026-0.
- 14. Wals, A.E.J.; Alblas, A.H. School-based Research and Development of Environmental Education: A case study. *Environ. Educ. Research* **1997**, *3*, 253–267. [CrossRef]
- The World-Class University and Ranking: Aiming Beyond Status; Sadlak, J.; Liu, N.C. (Eds.) UNESCO-CEPES: Bucharest, Romania, 2007.
- Giesenbauer, B.; Müller-Christ, G. Mit den Sustainable Development Goals zu einer sinnhaften und nachhaltigen Unternehmensführung?: Systemische Evolutionsstufen als Unterscheidungsmerkmal für unterschiedliche Zugänge von Unternehmen. In *Im Brennpunkt: Zukunft des nachhaltigen Wirtschaftens in der Digitalen Welt*; Rogall, H., Binswanger, H.C., Ekardt, F., Grothe, A., Hasenclever, W.-D., Hauchler, I., Jänicke, M., Kollmann, K., Michaelis, N.V., Nutzinger, H.G., et al., Eds.; Metropolis Verlag: Marburg, Germany, 2018; pp. 281–294. ISBN 978-3-7316-1339-8.
- Müller-Christ, G.; Giesenbauer, B. Integrales Ressourcenmanagement: Leitplanken einer nachhaltigkeitsbezogenen Möglichkeitswissenschaft. In *Möglichkeitswissenschaften: Ökonomie mit Möglichkeitssinn*; Hochmann, L., Graupe, S., Korbun, T., Panther, S., Schneidewind, U., Eds.; Metropolis Verlag: Marburg, Germany, 2019; pp. 307–332.
- Brys, Z.; Bokor, P. Evaluation of Ken Wilber's Integral Psychology From a Scientific Perspective. J. Spiritual. Mental Health 2013, 15, 19–33. [CrossRef]
- World Commission on Environment and Development. Our Common Future. A/42/427. 1987. Available online: https://sustainabledevelopment.un.org/milestones/wced (accessed on 25 March 2019).
- Implementing Sustainable Development at University Level. A Manual of Good Practice; Leal Filho, W., MacDermott, F., Padgham, J., Eds.; European Research and Training Centre on Environmental Education: Bradford, UK, 1996; ISBN 1851431497.
- Lozano, R. Incorporation and institutionalization of SD into universities: Breaking through barriers to change. J. Clin. Prod. 2006, 14, 787–796. [CrossRef]
- 22. Leal Filho, W. Teaching sustainable development at university level: Current trends and future needs. *J. Balt. Sci. Educ.* 2010, *9*, 273–284.
- 23. Rieckmann, M. Future-oriented higher education: Which key competencies should be fostered through university teaching and learning? *Futures* **2012**, *44*, 127–135. [CrossRef]
- United Nations. Transforming our world: The 2030 agenda for sustainable development. A/RES/70/1. 2015. Available online: https://sustainabledevelopment.un.org/post2015/transformingourworld (accessed on 25 March 2019).
- Hugé, J.; Block, T.; Waas, T.; Wright, T.; Dahdouh-Guebas, F. How to walk the talk? Developing actions for sustainability in academic research. J. Clean. Prod. 2016, 137, 83–92. [CrossRef]
- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Kovačič Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]
- 27. Peer, V.; Stoeglehner, G. Universities as change agents for sustainability–framing the role of knowledge transfer and generation in regional development processes. *J. Clean. Prod.* **2013**, *44*, 85–95. [CrossRef]

- Thomas, I. Sustainability in tertiary curricula: What is stopping it happening? Int. J. Sustain. Higher Educ. 2004, 5, 33–47. [CrossRef]
- Blanco-Portela, N.; Benayas, J.; Pertierra, L.R.; Lozano, R. The Integration of Sustainability in Higher Education Institutions: A Review of Drivers of and Barriers to Organisational Change and Their Comparison Against Those Found of Companies. J. Clean. Prod. 2017, 166, 563–578. [CrossRef]
- Müller-Christ, G.; Giesenbauer, B.; Tegeler, M.K. Die Umsetzung der SDGs im deutschen Bildungssystem –Studie im Auftrag des Rats für Nachhaltige Entwicklung der Bundesregierung. Zeitschrift Int. Bildungsforschung Entwicklungspädagogik 2018, 41, 19–26.
- 31. Leal Filho, W. Dealing with misconceptions on the concept of sustainability. *Int. J. Sustain. Higher Educ.* 2000, 1, 9–19. [CrossRef]
- 32. Moore, J.; Pagani, F.; Quayle, M.; Robinson, J.; Sawada, B.; Spiegelman, G.; van Wynsberghe, R. Recreating the university from within. *Int. J. Sustain. Higher Educ.* 2005, *6*, 65–80. [CrossRef]
- Lattu, A.; Cai, Y. Tensions in the Sustainability of Higher Education—The Case of Finnish Universities. Sustainability 2020, 12, 1941. [CrossRef]
- 34. Hedlund-de Witt, A. Worldviews and the Transformation to Sustainable Societies. An Exploration of the Cultural and Psychological Dimensions of our Global Environmental Challenges. Ph.D. Thesis, VU University Amsterdam, Amsterdam, The Netherlands, 2013.
- 35. Hedlund-de Witt, A. Rethinking Sustainable Development: Considering How Different Worldviews Envision "Development" and "Quality of Life". *Sustainability* **2014**, *6*, 8310–8328. [CrossRef]
- Hedlund-de Witt, A. The Integrative Worldview and its Potential for Sustainable Societies. Worldviews 2014, 18, 191–229. [CrossRef]
- 37. Van Marrewijk, M. The Cubrix, an Integral Framework for Managing Performance Improvement and Organisational Development. *Technol. Invest.* **2010**, *1*, 1–13. [CrossRef]
- Van Marrewijk, M.; Werre, M. Multiple levels of corporate sustainability. J. Business Ethics 2003, 44, 107–119. [CrossRef]
- Müller-Christ, G. Nachhaltigkeitsforschung in einer transzendenten Entwicklung des Hochschulsystems—Ein Ordnungsangebot für Innovativität. In *Innovation in der Nachhaltigkeitsforschung: Ein Beitrag zur Umsetzung der UNO Nachhaltigkeitsziele;* Leal Filho, W., Ed.; Springer: Berlin/Heidelberg, Germany, 2017; pp. 161–180. ISBN 978-3-662-54358-0.
- 40. Wilber, K. A Theory of Everything. An Integral Vision for Business, Politics, Science and Spirituality; Shambhala: Boston, MA, USA, 2001; ISBN 9781570628559.
- Scharmer, C.O. The Essentials of Theory, U. Core Principles and Applications; Berrett-Koehler: Oakland, CA, USA, 2018; ISBN 978-1523094400.
- 42. Fallis, G. *Multiversities, Ideas, and Democracy,* 2nd ed.; University of Toronto Press: Toronto, ON, Canada, 2007; ISBN 9780802092403.
- 43. Laloux, F. Reinventing Organizations. A Guide to Creating Organizations Inspired by the Next Stage of Human Consciousness, 5th ed.; Nelson Parker: Brussels, Belgium, 2014; ISBN 2960133501.
- Van Egmond, N.D.; de Vries, H.J.M. Sustainability: The search for the integral worldview. *Futures* 2011, 43, 853–867. [CrossRef]
- Hahn, T.; Pinkse, J.; Preuss, L.; Figge, F. Tensions in Corporate Sustainability: Towards an Integrative Framework. J. Bus. Ethics 2015, 127, 297–316. [CrossRef]
- Hahn, T.; Figge, F.; Pinkse, J.; Preuss, L. A Paradox Perspective on Corporate Sustainability: Descriptive, Instrumental, and Normative Aspects. J. Bus. Ethics 2018, 148, 235–248. [CrossRef]
- Brown, B.C. Leading complex change with post-conventional consciousness. J. Organ. Change Manag. 2012, 25, 560–577. [CrossRef]
- Scharmer, C.O.; Theory, U. Leading from the future as it emerges: The social technology of presencing, 2nd ed.; Berrett-Koehler: San Francisco, CA, USA, 2016; ISBN 9781626567986.
- 49. Shapiro, A.; Permuth, S. Stratification, in Organizations. In *Encyclopedia of Educational Leadership and Administration*; English, F.W., Ed.; SAGE: Thousand Oaks, CA, USA, 2011; ISBN 9780761930877.
- Altbach, P.G.; Reisberg, L.; Rumbley, L. Trends in Global Higher Education. Tracking an Academic Revolution; Sense Publishers; UNESCO Publishing: Rotterdam, The Netherlands, 2010; ISBN 9460913393.
- Graves, C.W. Levels of Existence Related to Learning Systems. In Proceedings of the Ninth Annual Conference of the National Society for Programmed Instruction, Rochester, NY, USA, 31 March 1971.

- 52. Kerr, C. *The Uses of the University. With a New Chapter and Preface*, 5th ed.; Harvard University Press: Cambridge, MA, USA, 2001; ISBN 0674005325.
- 53. Private Higher Education. A Global Revolution; Altbach, P.G., Levy, D.C., Eds.; Sense Publishers: Rotterdam, The Netherlands, 2005; ISBN 9789077874080.
- 54. Lewin, K. Feldtheorie; Klett-Cotta: Bern, Germany, 1982; ISBN 3-12-935140-X.
- 55. Laptev, A.V.; Efimov, V.S. New Generation of Universities. University 4.0. Journal of Siberian Federal University. *Hum. Soc. Sci.* 2016, *11*, 2681–2696. [CrossRef]
- Efimov, V.; Lapteva, A. University 4.0: Philosophical Analysis. In Proceedings of the 10th Annual International Conference of Education, Research and Innovation, Seville, Spain, 16–18 November 2017; Gómez Chova, L., López Martínez, A., Candel Torres, I., Eds.; IATED, 2017; pp. 589–596.
- Verhoef, L.; Bossert, M. The University Campus as a Living Lab for Sustainability. A Practitionier's Guide and Handbook; Delft University of Technology: Delft, Germany; Hochschule f
 ür Technik Stuttgart: Stuttgart, Germany, 2019.
- Schneidewind, U. Von der nachhaltigen zur transformativen Hochschule: Perspektiven einer "True University Sustainability". UmweltWirtschaftsForum 2014, 22, 221–225. [CrossRef]
- 59. Murray, P.; Goodhew, J.; Murray, S. The heart of ESD: Personally engaging learners with sustainability. *Environ. Educ. Res.* **2014**, *20*, 718–734. [CrossRef]
- Scharmer, C.O.; Kaufer, K. Leading from the emerging future. From ego-system to eco-system economies, 1st ed.; Berrett-Koehler: San Francisco, CA, USA, 2013; ISBN 1605099260.
- Schneidewind, U. Transformative Literacy. Gesellschaftliche Veränderungsprozesse verstehen und gestalten. GAIA Ecol. Perspect. Sci. Soc. 2013, 22, 82–86.
- 62. Vogt, M.; Weber, C. The Role of Universities in a Sustainable Society. Why Value-Free Research is Neither Possible nor Desirable. *Sustainability* **2020**, *12*, 2811. [CrossRef]
- Stensaker, B.; Lee, J.J.; Rhoades, G.; Ghosh, S.; Castiello-Gutiérrez, S.; Vance, H.; Çalıkoğlu, A.; Kramer, V.; Liu, S.; Marei, M.S.; et al. Stratified University Strategies: The Shaping of Institutional Legitimacy in a Global Perspective. J. Higher Educ. 2019, 90, 539–562. [CrossRef]
- 64. Wissema, J. *Towards the Third Generation University*; Edward Elgar Publishing: Cheltenham, UK, 2009; ISBN 9781848446182.
- Podger, D.M.; Mustakova-Possardt, E.; Reid, A. A whole-person approach to educating for sustainability. Int. J. Sustain. Higher Educ. 2010, 11, 339–352. [CrossRef]
- Meyerhuber, S. Theme-Centered Interaction for Educating Future Leaders: Applied Social Psychology for Teaching Academics to Act Socially Sustainably in Organisations. In *Theme Centered Interaction (TCI) in Higher Education: A Didactic Approach for Sustainable and Living Learning*; Meyerhuber, S., Reiser, H., Scharer, M., Eds.; Springer: Cham, Germany, 2019; pp. 3–56. ISBN 978-3-030-01047-8.
- Theme Centered Interaction (TCI) in Higher Education. A Didactic Approach for Sustainable and Living Learning; Meyerhuber, S.; Reiser, H.; Scharer, M. (Eds.) Springer: Cham, Germany, 2019; ISBN 978-3-030-01047-8.
- Stollberg, D.; Schneider-Landolf, M. Living Learning. In *Handbook of Theme-Centered Interaction (TCI)*; Schneider-Landolf, M., Spielmann, J., Zitterbarth, W., Eds.; Vandenhoeck & Ruprecht: Göttingen, Germany, 2017; pp. 137–146. ISBN 9783525451908.
- Hoffmann, S.G. The Disturbance Postulate. In *Handbook of Theme-Centered Interaction (TCI)*; Schneider-Landolf, M., Spielmann, J., Zitterbarth, W., Eds.; Vandenhoeck & Ruprecht: Göttingen, Germany, 2017; pp. 95–100. ISBN 9783525451908.
- Faßhauer, U. The Existential-Anthropological Axiom. In *Handbook of Theme-Centered Interaction (TCI)*; Schneider-Landolf, M., Spielmann, J., Zitterbarth, W., Eds.; Vandenhoeck & Ruprecht: Göttingen, Germany, 2017; pp. 75–79. ISBN 9783525451908.
- Barth, M.; Godemann, J.; Rieckmann, M.; Stoltenberg, U. Developing key competencies for sustainable development in higher education. *Int. J. Sustain. Higher Educ.* 2007, *8*, 416–430. [CrossRef]
- 72. Rieckmann, M. Education for sustainable development goals. Learning objectives; UNESCO: Paris, France, 2017; ISBN 9231002090.
- 73. MIT Sustainability Office. Discover Living Labs. Available online: https://sustainability.mit.edu/living-labs (accessed on 16 March 2020).

- Simonsohn, U.; Simmons, J.P.; Nelson, L.D. Better P-curves: Making P-curve analysis more robust to errors, fraud, and ambitious P-hacking, a Reply to Ulrich and Miller (2015). J. Exp. Psychol. Gen. 2015, 144, 1146–1152. [CrossRef]
- 75. Yamada, Y. How to Crack Pre-registration: Toward Transparent and Open Science. *Front. Psychol.* 2018, *9*, 1831. [CrossRef]
- De Wit, K. The Networked University: The structure, culture, and policy of universities in a changing environment. *Tert. Educ. Manag.* 2010, *16*, 1–14. [CrossRef]
- 77. Selingo, J.J. The Networked University: Building Alliances for Innovation in Higher Education; Pearson: London, UK, 2017.
- 78. Holzer, B. Netzwerke; Transcript Verlag: Bielefeld, Germany, 2006; ISBN 9783839403655.
- White, H.C. Network Switchings and Bayesian Forks: Reconstructing the Social and Behavioral Sciences. Soc. Res. 1995, 62, 1035–1063.
- Tacke, V. Systeme und Netzwerke-oder: Was man an sozialen Netzwerken zu sehen bekommt, wenn man sie systemtheoretisch beschreibt. In *Netzwerke und soziale Arbeit: Theorien, Methoden, Anwendungen*; Fischer, J., Kosellek, T., Eds.; Beltz Juventa: Weinheim, Germany, 2013; pp. 143–162. ISBN 9783779928263.
- Kleve, H. Die Wechselseitigkeit von Geben und Nehmen: Netzwerke als soziale Systeme. Systhema 2017, 31, 110–121.
- 82. Kleve, H. Reziprozität ermöglichen. Kontext 2017, 48, 353-367. [CrossRef]
- Vangen, S.; Hayes, J.P.; Cornforth, C. Governing Cross-Sector, Inter-Organizational Collaborations. Pub. Manag. Rev. 2015, 17, 1237–1260. [CrossRef]
- 84. Luhmann, N. Die Gesellschaft der Gesellschaft; Suhrkamp: Frankfurt am Main, Germany, 1997; ISBN 351858247X.
- Deutscher Bundestag. Bericht der Bundesregierung zur Bildung für nachhaltige Entwicklung. Unterrichtung durch die Bundesregierung (18. Wahlperiode); Bundestagsdrucksache 18/13665. 2017. Available online: https://www.bmbf.de/files/Drucksache_1813665_BT-Bericht%20BNE.pdf (accessed on 20 March 2019).
- Kiefhaber, E.; Spraul, K. Erfolgsfaktoren von Nachhaltigkeitsnetzwerken: Eine Sondierungsstudie für zukünftige Netzwerkforschung–Zusammenfassung der Ergebnisse für Studienteilnehmer; Technische Universitat Kaiserslautern: Kaiserslautern, Germant, 2016.
- 87. Spraul, K. Erfolgsfaktoren von Nachhaltigkeitsnetzwerken. In Proceedings of the 4. Symposium "Nachhaltigkeit in der Wissenschaft" (SISI), München, Germany, 17 July 2018.
- Müller-Christ, G.; Liebscher, A.K. Advanced Training for Sustainability Change Agents—Insights and Experiences from a Seminar Series Using the Method of Systemic Constellations. In *Integrative approaches to* sustainable development al university level: Making the links; Leal Filho, W., Ed.; Springer: Heidelberg, Germany, 2015; pp. 451–466. ISBN 978-3-319-10689-2.
- Denzler, W.; Schmitt, C.T. Digitalisierung und Nachhaltige Entwicklung an Hochschulen: Synergien und Spannungsfelder: Digitalisierung—Werkzeug und Thema im Hochschulnetzwerk HOCHN. Synergie Fachmagazin Digitalisierung Lehre 2019, 7, 30–33. [CrossRef]
- 90. Die HOCH-N-Nachhaltigkeitslandkarte. Available online: https://www.nachhaltige-hochschullandschaft.de (accessed on 10 January 2020).
- Bassen, A.; Sassen, R.; de Haan, G.; Klußmann, C.; Niemann, A.; Gansel, E. Anwendung des hochschulspezifischen Nachhaltigkeitskodex—Ein Weg zur Nachhaltigkeitsberichterstattung an Hochschulen (Betaversion); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Hamburg, Germany, 2018.
- Bormann, I.; Rieckmann, M.; Bauer, M.; Kummer, B.; Niedlich, S. Nachhaltigkeitsgovernance an Hochschulen (Betaversion); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Berlin, Germany, 2018; Available online: https://www.hochn.uni-hamburg.de/-downloads/handlungsfelder/ governance/hoch-n-leitfaden-nachhaltigkeitsgovernance-an-hochschulen.pdf (accessed on 9 January 2020).
- Bauer, M.; Niedlich, S.; Rieckmann, M.; Bormann, I.; Jaeger, L. Interdependencies of Culture and Functions of Sustainability Governance at Higher Education Institutions. *Sustainability* 2020, 12, 2780. [CrossRef]
- Bellina, L.; Tegeler, M.K.; Müller-Christ, G.; Potthast, T. Bildung für Nachhaltige Entwicklung (BNE) in der Hochschullehre (Betaversion); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Bremen, Germany, 2018; Available online: https://www.hochn.uni-hamburg.de/-downloads/ handlungsfelder/lehre/hoch-n-leitfaden-bne-in-der-hochschullehre.pdf (accessed on 9 January 2020).
- 95. Schopp, K.; Bornemann, M.; Potthast, T. The Whole-Institution Approach at the University of Tübingen: Sustainable Development Set in Practice. *Sustainability* **2020**, *12*, 861. [CrossRef]

- Kahle, J.; Jahn, S.; Lang, D.J.; Vogt, M.; Weber, C.F.; Lütke-Spatz, L.; Winkler, J. Nachhaltigkeit in der Hochschulforschung (Betaversion); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Lüneburg, Germany, 2018; Available online: https://www.hochn.uni-hamburg.de/-downloads/handlungsfelder/forschung/hoch-n-leitfadennachhaltigkeit-in-der-hochschulforschung.pdf (accessed on 9 January 2020).
- 97. Delakowitz, B.; Schön, E.; Bulcsu, A.; Günther, E.; Hüske, A.-K.; Roos, N.; Brauweiler, J.; Will, M.; Zenker-Hoffmann, A. Nachhaltigkeit im Hochschulbetrieb (Betaversion); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Zittau, Germany, 2018; Available online: https://www.hochn.uni-hamburg.de/-downloads/handlungsfelder/betrieb/hochn-leitfaden-nachhaltiger-hochschulbetrieb.pdf (accessed on 9 January 2020).
- Roos, N.; Heinicke, X.; Guenther, E.; Guenther, T.W. The Role of Environmental Management Performance in Higher Education Institutions. *Sustainability* 2020, 12, 655. [CrossRef]
- 99. Nölting, B.; Dembski, N.; Kräusche, K.; Lehmann, K.; Molitor, H.; Pape, J.; Pfriem, A.; Reimann, J.; Skroblin, J.-H.; Walk, H. *Transfer für nachhaltige Entwicklung an Hochschulen (Beta-Version)*; BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln-vernetzen-berichten (HOCH-N)": Eberswalde, Germany, 2018; Available online: https://www.hochn.uni-hamburg.de/-downloads/handlungsfelder/transfer/hoch-nleitfaden-transfer-fuer-nachhaltige-entwicklung-an-hochschulen.pdf (accessed on 9 January 2020).
- Nölting, B.; Molitor, H.; Reimann, J.; Skroblin, J.-H.; Dembski, N. Transfer for Sustainable Development at Higher Education Institutions—Untapped Potential for Education for Sustainable Development and for Societal Transformation. *Sustainability* 2020, *12*, 2925. [CrossRef]
- 101. Giesenbauer, B. Veränderung durch Veränderung: Nachhaltige Entwicklung von Hochschulen im Huckepack der Digitalisierung. In Digitalisierung und Nachhaltigkeit: Chancen und Perspektiven für deutsche Hochschulen; Leal, W., Ed.; Springer: Heidelberg, Germany, 2020; in press.
- 102. Kruse, P. Next practice. In Erfolgreiches Management von Instabilität; GABAL: Offenbach, Germany, 2004.
- 103. Mehling, S.; Kolleck, N. Cross-Sector Collaboration in Higher Education Institutions (HEIs): A Critical Analysis of an Urban Sustainability Development Program. *Sustainability* **2019**, *11*, 4982. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article What Sustainability? Higher Education Institutions' Pathways to Reach the Agenda 2030 Goals

Isabel Ruiz-Mallén * and María Heras

Internet Interdisciplinary Institute (IN3), Universitat Oberta de Catalunya, Barcelona 08860, Spain; mheras0@uoc.edu

* Correspondence: iruiz_mallen@uoc.edu

Received: 29 December 2019; Accepted: 8 February 2020; Published: 11 February 2020

Abstract: Higher Education Institutions (HEIs) have the mandate of promoting sustainability through addressing the Agenda 2030. However, how this is being understood and framed in both discourse and practice by HEIs remains an underexplored issue. This article interrogates the concept of sustainability embraced by ten key HEIs networks at global and regional levels while identifying and discussing the main pathways for action displayed. We rely on HEIs networks' data from available online documents related to the Agenda 2030. "Greening" is the dominant sustainability discourse among the global and many regional HEIs networks, that is, the one that refers to the links between people, planet and profit. Two other discourses are minor and regional, "resilience" and "alternative". The "alternative" discourse is the only one entailing a critical approach to the Agenda 2030 goals. All networks promote changes in HEIs organizational culture to embed sustainability values in strategic planning, academic and managerial work. Yet there is a need for further engagement with society to readdress HEIs societal role. Deep and critical reflection of the worldviews, contradictions and tensions in the discourses and practices proposed by HEIs networks at global and regional scales is also needed to build common pathways toward sustainability.

Keywords: discourse analysis; environment; Global South; Sustainable Development Goals; universities' transformation

1. Introduction

In the last century, human activities have dramatically modified natural processes while significantly affecting social-ecological systems, leading to the current environmental crisis. In the coming decades, a "point of no return" could be reached unless greenhouse gas emissions will be reduced, and responsible environmental stewardship will be promoted at a global scale [1]. Climate action is one of the 17 Sustainable Development Goals (SDGs) of the Agenda 2030, a plan of action adopted by all United Nations member states in 2015 "to stimulate action over the next 15 years in areas of critical importance for humanity and the planet" [2]. Meeting the Agenda 2030 goals requires a political willingness to build pathways toward sustainable futures by changing the current development trends. In this regard, society can exert pressure on the governments and counteract those corporate interests defending the status quo. It becomes of paramount importance to raise public awareness and reflection on the causes and consequences of the environmental crisis, as well as to build capacities to responsibly and creatively deal with related challenges. Universities and other Higher Education Institutions (HEIs) can play a crucial role in this endeavor. In particular, HEIs can prevent students from being overwhelmed by the nihilism and hopelessness of the current dramatic situation while promoting effective skills acquisition and values of connectedness between humans and nature [3].

Indeed, HEIs have been formally working on sustainability issues since Rio's Summit in 1992 through the implementation of the Agenda 21. Its Chapters 35 and 36 already called universities to (i)

improve long-term scientific assessment, (ii) build up scientific capacity and capability, (iii) reorient education towards sustainable development, (iv) increase public awareness of the interrelated nature of human activities and the environment, and (v) promote training to develop human resources and facilitate the transition to a more sustainable world [4]. Later on, in 2002, and to coordinate educational efforts of HEIs and other entities toward sustainable development, the United Nations (UN) established the UN Decade of Education for Sustainable Development 2005–2014. Overall, the Decade of Education curriculum and quality systems. It also increased recognition on the value of outreach activities and attracted funding to lowering universities' ecological footprint [5]. However, progress in this regard seemed to have remained slower than desirable due to universities' resistance to adopt a whole-institution approach that can lead them to move from reductionist to more holistic and transdisciplinary perspectives [6–8]. Such transition involves moving toward collaborative work between different disciplines and proactive engagement with the society toward transformative changes [6,7].

The current Agenda 2030, through its emphasis in education, can provide further opportunities for transformative change toward sustainability for HEIs. The Agenda 2030 includes an SDG devoted explicitly to education (SDG #4: Quality Education) and a target that addresses education for sustainable development (target 4.7) while highlighting the essential contribution of sustainability education to the other 16 SDGs [9]. Moreover, the UN Global Action Program (GAP), launched at the UNESCO Conference on Education for Sustainable Development in 2014 to continue the legacy of the Decade of Education for Sustainable Development until 2019, contributed to this target through scaling-up best practices and actions of education institutions including HEIs [10]. Further, many HEIs worldwide are implementing strategies for the adoption of the Agenda 2030 and its 17 SDGs. There seems to be, however, divergent and contrasting views on the adjustments they need to do toward more sustainable futures, which are also inherent to the Agenda 2030, such as the debate between the proponents of greening the economy and those advocating for alternatives to economic growth [11]. There is evidence that these different visions also permeate the way local universities are responding to the sustainability call [12]. How the sustainability mandate of the Agenda 2030 is being understood and framed in both discourse and proposed practice by HEIs at global and regional scales remains an underexplored issue. Is there a common understanding among HEIs of what sustainability means in higher education discourse and practice at global and regional levels? Are their efforts going in the same direction? Is there anything missing? This article sheds light over these issues by comparatively examining the sustainability discourses of a sample of key global and regional HEIs networks. It also identifies the main action points these HEIs are advocating for and discusses the main trends and most fundamental tensions and gaps in promoting sustainability within higher education.

In what follows, we review frameworks that are nourishing the concept of sustainability through compiling similar and contrasting visions within and beyond the educational realm, as well as methodological approaches analyzing universities' efforts in achieving sustainability goals. Based on this review, we present the analytical framework that we used in our analysis. We then explain the selection of the main HEIs networks at global and regional levels that are leading and guiding universities toward the Agenda 2030. We describe how we analyzed sustainability discourses and practices of selected HEIs networks by relying on the analytical framework previously presented. We report our findings on HEIs in this regard while revealing and discussing general trends as well as missing issues that would need to be further addressed by HEIs to improve coherence and find common pathways toward sustainability. We also highlight further research lines in this regard.

2. An Analytical Framework on Sustainability Discourse and Practice

There is no questioning about the ubiquity and ambiguity of the term sustainability as a critical concept for social change across disciplines and institutions. Previous work has identified three main trajectories of sustainability that rely on different values, processes and understandings of

the changes and transformations required for sustainability [11]. A first trajectory, based on the eco-modern paradigm, advocates for green economy supported by technological progress as the primary strategy to build sustainable futures. A second trajectory, relying on social transformation, which, in contrast to the former, challenges the current economic system by advocating for a radical change such as in the case of the degrowth movement. And, a third one, based on the resilience paradigm, promotes anticipating and controlling risks while finding solutions through socio-technical mechanisms. These main trajectories are, in turn, related to how the relationship between humans and nature is conceived within sustainability discourses. Epistemologically, three ways of approaching such a relationship can be identified. One is based on the reciprocal relationship between the environment and the tandem society-economy that supports economic growth. A second approach focuses on intergenerational equity and fairness that questions the current economic system. A third approach relies on forward-thinking, technology and innovation to find solutions [12]. These different discourses are also being projected in a variety of sustainability educational approaches within HEIs, which reproduce similar debates on the economic models and human-nature relationships that lead to sustainable futures. While some sectors advocate for market-driven, outcomes-oriented and standardized models of education mainly focused on competition, knowledge acquisition and technical skills, other educators and practitioners offer alternatives based on collaboration, emotions and values such as solidarity [13]. By taking positions on these discussions, HEIs navigate sustainability discursively but also practically.

Previous studies have reviewed and analyzed how universities are defining and implementing sustainability locally and discussed the links with sustainability discourses [6,14,15]. HEIs are doing multiple actions to engage in sustainability, such as integrating sustainability issues in the curricula, research, outreach and campus operations. For instance, a study surveyed 167 universities across five continents on the integration of the Agenda 2030 SGDs with sustainability teaching and found that lectures are the most common way they use to incorporating these issues [14]. However, other actions that could have a direct impact in society seem to be less common, such as those oriented to improving teachers' capacities to educate and empower students toward building sustainable futures and supporting inter- and transdisciplinary research to deal with complex challenges [6]. Researchers also note that efforts to reach sustainability in the university context are mostly focused on technological solutions and operational activities, such as the greening of university campuses. By contrast, actions addressed to promote HEIs reflection on behavioral and cultural issues within the organization itself are often absent but posed as crucial for a transition to sustainability. These authors frame such a transition in three stages that universities can navigate. In the first stage, called "operational optimization," HEIs increase the efficiency of the technical solutions they apply to deal with sustainability challenges and to comply with legal requirements. In the second stage, named "organizational transformation," HEIs actions keep a focus on infusing sustainability within the organization but also prioritize engaging with the behaviors and attitudes of students, teachers and other immediate actors. Finally, the third stage, called "systems building," involves a change in the vision and values of HEIs to create a sustainability culture. In doing this, HEIs reflect collaboratively with other actors on improving their role in society and extend actions beyond the limits of their organizations [15].

The transformative potential of the Agenda 2030 mandate in HEIs is thus subjected to these different understandings and implementation traits, which can potentially orient HEIs within contrasting transition navigation processes. Within this context, understanding how global and regional HEIs networks are building sustainability discourses and promoting their implementation in practice becomes crucial for assessing and orienting HEIs efforts in this regard. By reviewing online documents that state and describe the visions, goals and actions of ten key global and regional HEIs networks, we examine what type of discourses and actions are promoted by these networks globally as well as at the level of each continent (except for Antarctica). To do that, and by relying on the above-described frameworks approaching sustainability discursively and in practice, we elaborate an analytical framework for the characterization of global and regional HEIs discourses and action (Table 1). Our analytical framework includes, on the one hand, the categories of analysis characterizing HEIs

discourses. These categories are pre-defined based on the main traits illustrating the three sustainability trajectories and understandings previously described [11,12]. We also add a fourth discourse trait specifically addressed to capture HEIs views on the Agenda 2030. On the other hand, the analytical framework includes another set of pre-defined categories characterizing HEIs proposed practices. These categories correspond to the main traits of these practices in each of the three stages defining HEIs transition toward sustainability [15] as well as those previously identified actions in this regard [6,14].

Table 1. The analytical framework for the analysis of higher education institutions (HEIs) discourses and proposed practices to reach the Agenda 2030. Categories of analysis are marked in italics.

	Catego	ries of Analysis	
Discourse Traits	Wh	at Does Sustainability Mean? [1	1,12]
	Ecological Modernization	Resilience	Society Transformation
Role of development and technology in the envisioned solutions to sustainability challenges	<i>Make it green</i> : becoming more active about protecting the environment with the support of technology while keeping economic growth	Increase resilience: confronting the vulnerabilities of the system through relying on technological progress and technocratic decision-making	Search for alternatives: supporting alternatives to economic growth and the global economic system while questioning the role of technology
Main strategy pushed forward to respond to risks	Coping or adaptation: mainly based on technical measures aiming to improve current practices	Anticipation and control: relying on technological solutions but focusing more on anticipating future challenges and controlling trade-offs	Transformation: implying a broad reflection on the organization's inherent behavioral and cultural aspects to redefine and build new practices
Nature-people relationship	<i>Utilitarianism:</i> nature understood as a resource that is at the service of people	<i>Objectivism</i> : nature seen as an object that can be controlled and shaped by people	Spiritualism: nature perceived at the same level than people so a convivial relationship can be established
Stance on the Agenda 2030	Pro-Agenda: embracing the and/or focusing on	Agenda 2030 in general terms, one or several SDGs	<i>Debate</i> : questioning the Agenda 2030 vision of sustainability
Practice Traits	How Is Sustainability Promoted in Practice? [6,14,15]		
	Technological Optimization	Organizational Transformation	Systems Building
Focus of the practical actions promoted	Organization improvement: efficiency and the compliance of regulation	Organizational culture: changes of attitude and development of a new set of values and behaviors	Systems' shift: targeted aims beyond the organization
Types of collaboration fostered	Isolated: no collaborations fostered beyond the institution	Academic partnerships: networking and advocacy with other HEIs	All types of partnerships: academic, government, private, general public
Main actions proposed for implementation	<i>Technical</i> : curricula, research, operations, campus experiences	Behavioral: assessment and reporting, educators training, academic collaboration	<i>Systemic:</i> transdisciplinary, outreach and collaboration beyond HEIs, advocacy

3. Materials and Methods

3.1. HEIs Networks Leading the Agenda 2030 Mandate

We strategically focused our review on HEIs networks leading the implementation of the Agenda 2030 worldwide to ensure coherence with our research purpose and limit the scope within the vast field of sustainability in higher education. We studied HEIs networks instead of other HEIs such as single universities or research centers because networks accelerate the dissemination of discourses and practices.

We used a snowball sampling strategy, starting from the Global University Network for Innovation (GUNi). This global network was chosen in the first place for being the direct holder of the UN Mandate for the implementation of the Agenda 2030 across HEIs. We conducted the snowball sampling in two consecutive stages, which included HEIs at the global and regional levels, respectively. First, and from GUNi's reviewed documents, we identified a set of seven HEIs networks of relevance at the global scale (see Figure 1 for further details). We then applied the following selection criteria: (i) being currently active; (ii) having a clear focus on sustainability, (iii) being mainly addressed at universities; and (iv) having available and balanced information about the two dimensions addressed in our review (i.e., discursive and practical). As a result, we included the following three global HEIs networks in our sample: the International Association of Universities (IAU), the Global Universities Partnership on Environment for Sustainability (GUPES) and Higher Education Sustainability Initiative (HESI). Together with GUNi, these four global networks represent a set of consolidated HEIs networks with a solid trajectory, born from different branches of the UN (i.e. GUNi and IAU from UNESCO, and GUPES and HESI from UNEP) and representing more than 800 HEIs distributed across 130 countries worldwide.



Figure 1. Global and regional networks identified in the snowball sampling. Source: own elaboration using image retrieved from Pixabay.com.

Second, from the review of the global networks, we identified 13 HEIs at the regional level. Such HEIs were distributed throughout all continents, except for Antarctica (see Figure 1). We then conducted a second screening applying the same selection criteria as in the case of the global networks and to select one network from each of the represented continents. As a result, we included six HEIs operating at the regional scale in the sample: the Association of African Universities (AAU, Africa), the Sustainable Development Solutions Network (SDSN, Australia/Pacific), the Alliance of Ibero-American Networks of Universities for Sustainability and the Environment (ARIUSA, South America), the Association for the Advancement of Sustainability in Higher Education (ASHEE, North America), the COPERNICUS Alliance (Europe) and the Network for the Promotion of Sustainability in Postgraduate Education and Research (PROSPER.Net, Asia).

Despite not being representative of all the HEIs worldwide, the resulting final set of ten HEIs represents a coherent and geographically balanced sample of acknowledged networks, both at the

continental and global scales. This sample can illustrate the main pathways currently promoted around the conceptualization and implementation of sustainability within higher education.

3.2. Data Collection and Analysis

For each of the ten selected HEIs networks, we reviewed two kinds of sources: (i) their official webpages, and most specifically, sections related to their mission and vision and their understanding of sustainability; and (ii) online documents about their links to the Agenda 2030 and/or the accomplishment of the SDGs, such as reports, declarations, charts and newsletters. Appendix A provides a list with the document types and sources reviewed for each network.

We analyzed these documents through content analysis to examine the main traits of sustainability discourses and practices promoted by the selected HEIs networks. We coded these data into the corresponding pre-defined categories of our analytical framework (see Section 2) in two ways. First, we coded data as 1=presence and 0=absence into each category. Second, we also coded data as key quotes when these reflected the meaning of the category. Even though our analytical framework was guided by the previously identified sustainability trajectories [11] and transition stages [15], this did not imply that an analyzed HEIs network should follow only one trajectory or be in a single stage. We codified the content of the reviewed documents independently by each discourse and practice trait and then grouped HEIs according to similar combinations of presence and absence into each trait to further identify the main discourses and proposed ways of action.

The two authors of this article conducted data collection and analysis. To ensure consistency of the analysis, we first analyzed two selected global networks and compared the consistency of coding among us, and then proceeded with the rest of the review.

4. Results

4.1. Main Commonalities and Differences in Global and Regional HEIs Networks' Sustainability Discourses

Three main discourses along which the analyzed HEIs networks navigate sustainability emerged from our analysis, which we have called: "resilience," "greening" and "alternative." These discourses are mainly shaped by HEIs networks' different understandings of the role of development in their envisioned solutions to face sustainability challenges and the strategies HEIs push forward to deal with risks. HEIs networks' stances on the Agenda 2030 are also crucial in defining their discourses as well as the visions on the nature–people relationship. Figure 2 includes the definitions of the identified sustainability discourses and places each network over the embraced discourse or discourses by global and regional levels.

Our results reveal that there is a major trend among the reviewed HEIs networks to concur with the "greening" discourse. In the visions, missions or reports analyzed, the four global HEIs and those regional networks in Europe, North America, Asia and Australia/Pacific refer to sustainability by highlighting the links between the dimensions of people, planet and profit. These HEIs understand that sustainability is achieved by guaranteeing economic growth while taking care of nature and people, which are interlinked challenges that need to be solved. As an example, IAU states the following in one of its publications about the Agenda 2030:

"Future well-being of humanity and the planet depends on successful resolution of the interconnected challenges of economic, social, cultural, and environmental sustainability" [16] (p. 10)



Figure 2. Main sustainability discourses identified among global and regional HEIs.

HEIs networks supporting the "greening" discourse aim to strengthen the role of universities in society as the leaders of adaptive or transformative processes toward such "green" futures through reinforcing collaboration and collective action. This approach often implies an organizational transformation within the HEIs, as we will explain in the next section. Interestingly, positions adopted on the role of technology for these sustainable futures differ among HEIs. On the one hand, IAU calls to embrace technological opportunities, ICT in particular, but also asks to analyze the potential trade-offs of these technologies. On the other hand, HESI, GUPES and COPERNICUS seem to omit this issue in their sustainability discourses. Finally, GUNi, AASHE, PROSPER.Net and SDSN emphasize the role of environmentally sound technologies in supporting sustainable development, especially in the Global South, as this quotation from GUNi shows:

"Enhance North-South, South-South and triangular regional and international cooperation on and access to science, technology and innovation and enhance knowledge sharing on mutually agreed terms, including through improved coordination among existing mechanisms, in particular at the United Nations level, and through a global technology facilitation mechanism" [17] (p. 13)

The other two discourses, "resilience" and "alternative" are mainly held by two regional HEIs networks: AAU in Africa and ARIUSA in Latin America and the Caribbean, respectively. AAU supports a sustainable development approach based on economic growth and effective natural resource management to prevent and control environmental risks by relying on technocratic decision-making processes. Indeed, its Strategic Plan 2016–2020 directly calls for a "resilient" higher education system:

"The onus to guide Africa on a sustainable path lies with resilient institutions like her higher education system to develop, train and retrain the human capital available on the continent; to efficiently manage her natural resources for the benefit of current and future generations; to conduct relevant research to stimulate her industries; and to build effective partnerships with multi-stakeholders (governments, civil society, industry, donors, etc.) to facilitate the development of strong institutions that are credentials of good governance." [18] (p. 23)

ARIUSA, in turn, poses alternative views to growth and development by putting the convivial relationship between nature and people above the interests of the global economic system. Moreover, and differently from the other HEIs networks that are embracing the Agenda 2030, ARIUSA questions

the agenda's SGDs. That is why we call this discourse "alternative." ARIUSA raises the concern about the economic growth connotations of the concept of sustainable development and the SDGs and advocates for putting the focus on taking care of the environment instead:

"When ARIUSA was created, almost two decades had already passed since the concept of sustainable development had been coined (...). This concept has been the object of much criticism and resistance from major sectors of the academic community in the region that, since the seventies, has been in charge of environmental matters, particularly that which is identified with the so-called "Latin American Environmental Thinking" (Ángel, 1997 and Leff, 2009). Sharing or taking into consideration these positions, when deciding on the name of its alliance of university environmental networks, the founders of ARIUSA opted for the term 'sustainability' as opposed to 'sustainable development'." [19] (p. 68)

Finally, it is relevant to highlight that some HEIs, both at global and regional scales, move along two sustainability discourses. IAU, AASHE and COPERNICUS mainly emphasize the "greening" discourse. Still, the documents reviewed also highlight elements from the "alternative" discourse, such as the importance given to having a balanced relationship between nature and people. COPERNICUS, for example, clearly states this issue in its Chart 2.0:

"We pledge that the signatories, all universities and other higher education institutions, are firmly committed to playing the central role they 'noblesse oblige' are obligated to in contributing to our successful transition towards a sustainable society, which is free, just, equal, solidary and tolerant. A society which is characterized by respect for nature and our fellow humans and by shared responsibility." [20] (p. 1)

The sustainability discourses of the global HEIS GUPES and the regional HEIS PROSPER.Net and SDSN are also twofold: "greening" and "resilience." In the case of GUPES, this is shown by two of its objectives that call for a "green" economy and refer to the prevention of risks, respectively:

"To optimize development opportunities provided by ecosystem services in a sustainable manner in line with the principles of "Green Economy" and in the context of sustainable development.

To help prepare the world for the projected impacts of global climate change, disasters and conflicts, harmful substances and hazardous wastes, as well as to assist in reversing and mitigating these and other negative environmental and sustainability trends." [21]

4.2. Main Trends and Gaps in the Promotion of Sustainability Practices by Global and Regional HEIs Networks

This section presents the results of our analysis concerning the characterization of sustainability practices promoted by the selected HEIs networks. We identified a clear pattern followed by all these networks: they mainly focus their efforts on changing the institutions' organizational culture and behavior through the integration of sustainability values and environmental concerns in strategic planning, academic and organizational work. Such focus can be seen throughout a continuum of promoted practices: from actions aimed at the creative development and exchange of best practices (e.g., COPERNICUS, ASHEE, HESI) to the institutionalization or mainstreaming of sustainability concerns within university systems (e.g., GUPES, ARIUSA), and the promotion of HEIs responsiveness to societal needs in order to become active agents of change (e.g., GUNi, IAU, AAU, SDSN, PROSPER.Net). See, for instance, two examples from ARIUSA and SDSN:

"ARIUSA has been constructed as a framework for communication, coordination, cooperation and co-managerial relations between different types of university environmental networks working together to foster the institutionalization of the environmental commitments of universities and other HEIs in Ibero-America." [19] (p. 65)

"Universities are increasingly re-thinking their role in the 21st century and looking to be both more responsive to societal needs and to become agents of change towards solving global challenges. (....) Furthermore, given the critical role universities have in ensuring the success of the SDGs, universities have a moral imperative to embody support for the SDGs as part of their social missions and core functions." [22] (p. 9)

The last quotation from SDSN points to questioning HEIs purpose and role in society. It emphasizes HEIs public service obligations and ethical imperative to contribute to societal changes fostering transitions toward sustainability, which denotes a shy transition from the stage of organizational transformation to the system building stage [15]. In the same line, other HEIs promote actions aligned with values such as social responsibility, commitment, collaboration, equity and inclusion (e.g., GUNi, COPERNICUS, ASHEE), platforms for multi- and interdisciplinary dialogue (PROSPER.Net), measures aimed at universities' systemic change (IAU), or efforts oriented toward ensuring the organizational capacity to deliver transformational change (AAU). All these actions are expected to benefit not only society but also HEIs by guaranteeing their own sustainability. Such an opportunity is also directly linked to the implementation of the Agenda 2030 mandate, as pointed out by GUNi:

"There is no doubt that HEIs have realized the importance of integrating sustainability in their strategies, both for the benefit of society and for the benefit that integrating it appears to have on the institution itself—according to SDSN, Universities benefit because they can demonstrate impact, capture more demand for SDG-related education, build new partnerships, access new funding streams, and make comparisons with other institutions via an agreed definition of a responsible university (SDSN, 2017:9). The latter is a very interesting aspect of the relationship between SDGs and HEIs." [23] (p. 11)

HEIs networks also promote similar practices when looking at the types of collaboration they fostered. The most emphasized types of collaboration are among HEIs and commonly oriented toward academic cooperation for joint-curricula development, interuniversity training and research (e.g., joint research projects, shared capacity building, organization of conferences and meetings in a network, joint publication of journals), and exchange of best practices and resources. Collaborations with external actors are also often expressed through the establishment of partnerships, networking and advocacy. While networks often mention these collaborations in general terms, some HEIs emphasize alliances with specific actors. For instance, government organizations and development institutions (GUNi, HESI, SDSN, AAU, COPERNICUS, GUPES, PROSPER.Net), industry and private sector (SDSN, AAU, PROSPER.Net) or local communities (SDSN, AAU, COPERNICUS, PROSPER.Net). In this regard, the Agenda 2030 is identified by these networks as a strategic framework to fostering partnerships within and beyond the universities, such as in the case of SDSN:

"One of the strengths of the SDG agenda is that it provides a common framework for different sectors and organizations to connect and work together on shared interests. This will give universities opportunities to form new collaborations with government, industry, and the community in both research and education. Equally, the framework can help identify common interests across different areas of the university, helping to drive cross-disciplinary partnerships, collaboration, and innovation." [22] (p. 9)

Our analysis shows, however, that engagement and critical self-reflection actions with actors beyond the academia aimed to readdress HEIs societal role are only partially mentioned by HEIs networks, or not mentioned at all, such as in the case of GUPES, HESI and ARIUSA. Although seven out of the ten networks explicitly emphasize collaborations with non-academic stakeholders (GUNi, IAU, COPERNICUS, ASHEE, AAU, SDSN, PROSPER.Net), these are often promoted in a relatively unidirectional way (e.g., HEIs as providers of input and knowledge). For instance, the GUNi network refers to the unique opportunity for HEIs, as trusted and educational institutions, to provide the skills and attitudes needed for active citizen participation but does not mention how HEIs can benefit from

citizen engagement [23]. Similarly, the HESI network highlights HEIs contributions at the policy level but not the other way around:

"HESI provides a unique opportunity for higher education institutions to provide input during the United Nations annual High-Level Political Forum on Sustainable Development, where Member States review progress made towards the SDGs." [24]

Regarding the types of implementation actions proposed by HEIs networks, these include in all cases technical actions. At the same time, we find differences in some of the specific activities related to behavioral and systemic change. Concerning technical actions, all the networks reviewed include actions oriented toward learning and teaching. In this regard, all HEIs encourage curriculum change and innovation, mostly through the integration of sustainability related issues and the teaching of competences, skills and motivation to understand and address sustainability goals. Furthermore, seven out of the ten networks analyzed propose actions around campus operations and/or governance (exceptions are GUPES, AAU, PROSPER.Net). The role of research actions is also emphasized by most HEIs networks (except IAU and HESI) as key to advance knowledge and evidence-based solutions and innovations. Some networks further mention strategic actions in this regard, such as the inclusion of Responsible Research and Innovation (RRI) guidelines in research methods (GUNi).

Also related to research, some HEIs networks go beyond technical actions to promote behavioral initiatives oriented to change HEIs organization, such as the combination of traditional disciplinary approaches and newer interdisciplinary and transdisciplinary ones (SDSN, PROSPER.Net). Within such behavioral approaches, assessment and monitoring of HEIs progress are proposed by six of the analyzed networks (ARIUSA, GUNi, SDSN, IAU, CORPERNICUS, ASHEE). They acknowledge these practices as critical in any strategic action plan that expects to have an impact on the achievement of the SDGs, as ARISUA highlights:

"One of the first steps needs to be the establishment of a baseline or assessment of the initial status of the process of associating higher education institutions to the achievement of the Sustainable Development Goals. Knowledge of this process is even more nascent and differentiated for the different aspects of economic, social and environmental sustainability to which universities contribute. To overcome this situation, there is a need to construct a basic system of indicators to be able to assess the contribution made by HEIs to the goals of the 2030 Agenda." [19] (p. 72)

Steaming from this need, some networks at both global and regional scales propose specific monitoring systems and research projects. HESI's online self-evaluation tool (i.e., the Sustainability Test) focuses on sustainability literacy issues and is addressed to individuals, universities and organizations. ASHEE's self-reporting tool, named STARS, measures sustainability performance of universities. SDSN proposes a monitoring framework with 100 indicators for the SGDs. ARIUSA's research project, called RISU, develops indicators to assess the implementation of sustainability policies in Latin American universities. Other behavioral actions are oriented toward supporting capacity building to contribute to educators' development of leadership skills and further abilities required to teach sustainability issues and change HEIs learning environments. This is the case, for instance, of the PROSPER.Net "Leadership Programme," a training addressed to early career researchers and young professionals from diverse sectors (e.g., local communities, public officials, private sector) to explore together how partnerships can be fostered toward more sustainable practices across a variety of fields. Reviewed HEIs also include behavioral actions related to academic collaborations, such as the development of conferences and seminars for sharing practices and knowledge and to foster debate and collaboration, like the International Conference on Sustainable Development Goals: Actors and Implementation, organized by GUNi, or the National and Latin American Forums of Universities and Sustainability organized by ARIUSA.

Finally, systemic actions are proposed by most reviewed networks (all except ARIUSA) but are less present in emphasis. AAU, HESI, COPERNICUS, IAU, SDSN and PROSPER.Net explicitly include advocacy-related actions, emphasizing the political dimension of SDGs and the interaction with actors beyond academic ones. For instance, advocacy for achieving sustainable outcomes is outlined by AAU as one of the four core elements of its programmatic approach:

"Our programmatic approach consists of a coherent package of service delivery and advocacy activities delivered in partnership at all levels—from community to national, regional and international levels. In the context of AAU, this is based on a critical analysis of HEIs in the specific context of each country and across countries." [18] (p. 17)

Actions proposed within the systemic approach also include applied research collaborations with private companies and development institutions working on SDGs (AAU, SDSN, GUPES, PROSPER.Net) and the evaluation and follow-up of broader policies (i.e., beyond educational ones) created by policy makers to address SDGs (e.g., GUNi).

All in all, through this diversity of proposed actions (teaching, research, assessment, dissemination, advocacy, etc.), all the reviewed networks go beyond SDG#4, Quality Education, perceiving the Agenda 2030 as a strategic and holistic framework for broad transversal action. Some networks are also emphasizing other SDGs in the reviewed documents. This is the case, for instance, of GUNi that has created a line of strategic work steaming from SDG#17. This line of action aims at sharing and building on expertise to reflect about opportunities and obstacles to foster effective and inclusive multi-stakeholder partnerships as a keystone for the achievement of SDGs within higher education [17]. Similarly, IAU addressed SDG#5 and the mandate of HEIs to bring women into higher levels of institutional leadership worldwide. IAU advocates for the anchoring of gender equity within the whole institution through the implementation of strategies that value diversity and contribute to the opportunities for women to access the highest positions and to move beyond the glass ceiling [16].

Table 2 summarizes the main traits of each network according to the focus of the promoted actions, the types of collaborations fostered and the main actions proposed.
	•		8
HEIs Network	Focus of the Practical Actions Promoted	Types of Collaborations Fostered	Main Actions Proposed
GUNi	Integration of sustainability and social values in HEIs strategies to strengthen their critical role within society and academic diplomacy to achieve partnerships and collaborations	Networks, resources exchange, values promotion, knowledge co-production and capacity building among HEIs and with other stakeholders	Inclusion of the concept of sustainability within HEIs research, education and training; Campus operations & governance; Sustainability assessment and follow-up of policies
IAU	HEIs systemic change to embed SDCs in strategic planning, academic and organizational work	Interdisciplinary research among HEIs and transdisciplinary work with other stakeholders	Curricula development and training: Outreach and networking (women leaders); Assessment and monitoring
GUPES	Mainstreaming of environment and sustainability concerns into university systems; interaction between UNEP and universities	Networks and resources exchange	Education and training: Applied research; Networking
HESI	Provision of a platform for HEIs to engage and contribute to the SDGs and exchange best practices	Exchange platforms and advocacy	Integrating SDGs within teaching, research and dissemination; Greening campuses; Support local sustainability efforts and explore innovative practices from other sectors; Engage with international networks; Outline an advocacy agenda
COPERNICUS	HEIs improvement by creatively developing and implementing comprehensive and integrated sustainability actions	Networks, joint knowledge production, and active engagement among HEIs and with other stakeholders	Curriculum change and capacity building within HEls; Change quality assessments and assurance systems; Outreach and dissemination actions; Advocacy for HEls for SD in Europe
AASHE	Improvement of HEIs practices toward the integration of sustainability	Networks among HEIs, partnerships with other actors	Training to teachers and students; Campus sustainability hub; Outreach and networking; Self-assessment; Partnerships with private, public and civic sectors
ARIUSA	Institutionalization of HEIs engagement toward environmental and sustainability issues and cooperation and coordination of actions between HEIs	Networks, academic cooperation and co-management among HEIs	Outreach events; Development of educational programs and research projects; Diagnosis and assessment
AAU	Capacity building of member institutions to address societal needs and to deliver transformational change	Partnerships with national and international actors in development and engagement with local communities	Training & research; Partnerships with other stakeholders; Community action
NSDS	HEIs responsiveness to societal needs to become agents of change toward solving global challenges	Inter and cross-disciplinary work at universities, partnerships with other actors	Learning and teaching; Research; Organizational governance, culture and operations of the university; External leadership
PROSPER.Net	Contributing to societal transformation for sustainable development, through transforming knowledge institutions and training future leaders	Sharing of resources and expertise among HEIs and similar networks, partnerships with other actors such as public officials, the private sector and local communities	Integration of sustainable development into curricula and research; Promotion of sustainability-oriented experiences between researchers and practitioners; Policy advocacy in higher education

Table 2. Main traits of sustainability practices promoted by selected global and regional HEIs.

5. Discussion and Conclusions

This research has explored how the sustainability mandate of the Agenda 2030 is being understood and framed in both discourse and proposed practice by HEIs networks at global and regional levels. Our findings show that in most cases HEIs networks' pathways toward sustainability are framed within a "greening" discourse that aims at improving nature and people's wellbeing through relying on less environmentally damaging forms of economic growth, as being in line with the Agenda 2030 goals. In doing this, these networks promote the integration of sustainability values in HEIs strategic planning, academic and organizational work, emphasize the need for partnerships and support actions mainly oriented to learning and teaching. This "greening" discourse is embraced by the four global networks reviewed (GUNi, IAU, HESI, GUPES) and four out of the six analyzed HEIs networks at the regional level (COPERNICUS, ASHEE, SDSN, PROSPER.Net). Our findings also show two minority sustainable pathways that only a couple of HEIs networks are following. On the one hand, the understanding of sustainability by the African network (AAU) relates to a "resilient" discourse that is based on increased control over nature and the use of technological solutions to reach the Agenda 2030 goals. Through promoting HEIs organizational change, this network aims to make them become active agents of change and provide sound responses to societal needs, with community action being one of its main supported activities. On the other hand, the Latin America network (ARIUSA) builds its understanding of sustainability on an "alternative" discourse that calls for prioritizing nature and human well-being over economic growth and, in doing this, questions the sustainable development approach of the Agenda 2030. Differently than the other cases, the main focus of this network is to foster HEIs organizational change through the institutionalization of sustainability concerns within university systems, with academic collaborations being a pivotal action to be enforced.

Before going into the discussion of relevant results on current potentials and gaps in the promotion of sustainability within higher education, we note some methodological limitations of our study. Findings described in the previous section need to be interpreted with caution, and broad generalizations on the identified sustainability discourses and practices among HEIs should be avoided because of two main reasons. First, it was not our intention to provide a single snapshot of the whole community of universities and other HEIs in terms of their approaches and actions toward sustainability and the SDGs. Other literature is already taking care of some of these issues [14]. Instead, we aimed at identifying commonalities and divergences across sustainability understandings and practices within the higher education system at both global and regional levels, as well as finding out the main gaps in this endeavor. To do that, we focused on HEIs networks as accelerators in the dissemination of discourses and practices, analyzing only those identified as actively working in promoting sustainability among universities and other HEIs worldwide or regionally. Therefore, our findings on identified HEIs efforts and gaps in navigating sustainability should be confined to these cases. Second, in general, we have found limited data on HEIs conceptual approaches to sustainability and the Agenda 2030. Our analysis has been based on the information and documents available at HEIs websites, so we may have missed other approaches to sustainability that are not public, are uploaded at other platforms or are still a work in progress. In this last regard, it is also possible that the reviewed HEIs are currently modifying their discourses or adopting new practices in the way they are embracing the different SDGs, which go beyond our analysis.

This said, our findings are valuable for both addressing the aim of the study and reflecting on the main points of discussion derived from relevant insights. On the one side, as mentioned above, results show that the sustainability discourses of the global HEIs networks are mainly placed within the "greening" discourse. In contrast, there is more diversity at the continental scale. It seems that the analyzed regional HEIs networks move along a gradient of acceptance of the sustainable development paradigm, being AAU in Africa and ARIUSA in Latin America and the Caribbean in the two opposite poles by embracing "resilient" and "alternative" discourses, respectively. While AAU supports technocratic decision-making processes to increase the effectiveness of current development approaches, ARIUSA poses alternative views to economic growth and the global economic system.

Differences in approaching sustainability between these regional HEIs could be related to the historical development of the field within each context. Differently than in Africa, in Latin America and the Caribbean the integration of sustainability within the higher education system is a result of an active tradition in environmental education that began with universities offering technical and vocational training on natural resource use and conservation in the 1950s [25]. Besides, there is another tradition in this field, that of questioning the term "sustainable development." It is argued that sustainable development is directly related to continuous growth that implies the accumulation of wealth by the rich in detriment of the poor in a context of limited natural resources, increasing social inequalities between the Global North and South. It is also argued that the concept of sustainable development omits the historical dominance and exploitation of both natural resources and human labor suffered by Latin America, and the Global South in general. Consequently, this concept cannot be transferred to new generations without a critical approach [26]. As our findings suggest, this might have implications for the adoption of the Agenda 2030 and the SDGs. Taking into account the links between environmental degradation and political and socio-economic backgrounds, as well as relevant socio-cultural issues in each region when promoting HEIs efforts toward sustainability, this could be one of the steps to resolve such a deadlock. This has been, so far, often neglected by universities [12] and it also seems to be omitted in the analyzed global HEIs discourses.

On the other side, and similarly to discourses, actions promoted by HEIs also follow a gradient of practice traits. Our findings suggest that all reviewed networks commonly assume technical solutions and compliance with the law. Moreover, there seems to be an emphasis in the promotion of actions aimed at changing the values, behaviors and attitudes of the organizations to create a sustainability culture within HEIs (e.g., organizational transformation stage). Many of the reviewed HEIs networks also propose collaborations beyond the educational sector to shift change to the whole system, in line with systems building, which is the last stage of the continuum of sustainability transition stages [15]. This emphasis on changing values and fostering non-academic collaborations highlighted by HEIs networks suggest certain progress in comparison to previous reviews in which operational efficiency was promoted in detriment of an organizational change [6]. Further research should analyze and assess the impact of HEIs actions in terms of operational transformation and systems building during the mandate of the Agenda 2030.

Moreover, to what extent these projected actions are being implemented remains an issue to be explored in further research. However, our findings already suggest tensions in this regard. As prior research points to, despite years of international agreements, calls for action and guides for good practices, the implementation of sustainability is still not mainstream within academia [27,28]. Indeed, transforming HEIs toward sustainability has encountered resistance to change pre-established systems due to universities' complex bureaucracy and rigid structures, among others [28]. Furthermore, the dominance of competition and disciplinary based culture within academia limits and slows efforts in this direction [27,29]. It is argued that HEIs should question these constraints to be able to transform their value systems and worldviews. Such questioning should be done within the institutions by reflecting on the values that are deterring academics from redesigning their disciplines and appreciating the epistemology and multicultural vision of sustainability [7]. In this regard, being aware of the institution's cultural predispositions and willingness to transform is critical for enhancing HEIs sustainability governance [30]. These studies already point to the crucial action of the human factor and the underlying worldviews and value systems within HEIs as barriers to change in some cases, or drivers of change in others [7,30]. Further research is needed to better understand the relation between HEIs institutional cultures and sustainability transformations. As the Rio+20 Treaty on Higher Education (2014) highlights, "to be transformative, higher education must transform itself" [31].

In this context, the efforts to embrace change posed by the reviewed HEIs networks should be interpreted positively as a step required for transformation. At the same time, it calls for caution as, to have a real impact, this change should be accompanied with an aligned practical implementation aimed at truly transforming HEIs organizational values by adopting a whole-institution approach [8].

Some of our findings point to gaps in this regard. First, although almost all HEIs networks mentioned collaborations with external actors, these collaborations are expressed mostly through providing input to change other sectors (e.g., advocacy, partnerships with private companies, or assessment of public policies). By contrast, allowing different actors' practices to permeate universities and engage in collaborations to actively rethink their role in society are actions poorly endorsed in the documents reviewed. This absence might also reveal a lack of reflective practices to go further in the transformation of HEIs themselves. Second, and connected to this, our findings show that most of the external collaborations are called to take place with institutional and governmental actors. Among other reasons, this trend might be due to the prominent role that governmental support has in encouraging the integration of sustainable practices within HEIs [28]. Such a focus contrasts, however, with the recognition for local partnerships as a prime way to tackle the need of interconnected governance responses demanded by the Agenda 2030 multidisciplinary approach [32,33]. Many universities have been compelled for more than 25 years now to develop community engagement work toward sustainable development (see for instance the COPERNICUS Charter, signed by more than 300 HEIs in Europe) [34]. Yet, our findings show that partnerships with local actors still represent a gap in HEIs external collaborations. Given that HEIs work and decisions may affect the economic, social and environmental aspects of local communities and regions [35], it should be highlighted the urgency of boosting the inclusion of local actors in HEIs actions. By establishing bidirectional communication channels and mechanisms for mutual learning, HEIs could also benefit from local actors' knowledge and ideas to navigate sustainability locally.

To conclude, our findings suggest that linking values and ontologies behind HEIs networks' sustainability discourses with their promoted actions might not always be taking place, although it is a required exercise for guaranteeing coherence. The development of robust and reflective assessment approaches, which is a line of action already promoted by several of the HEIs networks reviewed, seems an opportunity to work in this direction. The inclusion of external actors, and especially local communities, in processes of critical self-reflection could bring the chance to assess whether discourses and actions are aligned or not and how to improve progress in this regard. There is a perceived need to foster more integrative forms of societal and academic collaboration in the approach of HEIs to sustainability, whereby the combination of inputs from diverse fields may contribute to a better understanding of what sustainability is and means to people [27,32]. This progress might imply a turn of the dominant sustainability paradigm (i.e., "greening") toward a more humanistic approach, which is framed around convivial relationships between nature and humans and sees sustainability as a process rather than just an outcome [36]. Such an alignment will not be possible without a deep and critical reflection of the worldviews, contradictions and tensions in the discourses and practices proposed by HEIs networks at global and regional scales so to build common pathways. Further efforts in higher education research and policy development could take these elements into account to boost the envisioned societal transformations.

Author Contributions: Conceptualization and funding acquisition, I.R.-M.; methodology, formal analysis, and writing, I.R.-M. and M.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by European's Union Erasmus+ program under grant agreement No. 2018-1-UK01-KA203-048121 "RSP II: A Rounder Sense of Purpose II (RSPII)". I. Ruiz-Mallén gratefully acknowledges the financial support of the Spanish government's Research Agency through a 'Ramón y Cajal' research fellowship (RYC-2015-17676).

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

This appendix includes a table compiling the reviewed online documentation of each selected HEIs network for the analysis.

HEIs Network	Reviewed documents
GUNi	Website http://www.guninetwork.org/mission-and-objectives (Vision/Mission) Implementing the 2030 Agenda at Higher Education Institutions: Challenges and Responses. 2019. http://www.guninetwork.org/publication/implementing-2030-agenda -higher-education-institutions-challenges-and-responses Approaches to SDG17: Partnerships for the Sustainable Development Goals (SDGs) http://www.guninetwork.org/publication/approaches-sdg-17-partnerships-sustainable-d evelopment-goals-sdgs Sustainable Development Goals: Actors and Implementation. A Report from the International Conference http://www.guninetwork.org/files/guni_sdgs_report_0.pdf
IAU	Website http://www.iau-hesd.net/en/contenu/139-iau-action.html (Mission and HESD strategy) IAU Horizons, 24(1), April 2019. In focus: Universities and Agenda 2030: Engaging with the SDGs https://iau-aiu.net/IMG/pdf/iau_horizons_vol.24.1_en_lightpdf
GUPES	Website http://gupes.org/index.php?classid=3244 (Overall goal & objectives, pillars)
HESI	Website https://sustainabledevelopment.un.org/sdinaction/hesi Brochure https://sustainabledevelopment.un.org/content/documents/16065HESI_info_July3_v2.pdf
COPERNICUS	Website, Charts, Action Plan https://www.copernicus-alliance.org/about
AASHE	Website and Annual reports 2018, 2017, 2016, 2015 https://www.aashe.org/
ARIUSA	Website https://ariusa.net/es/ariusa Primera década de la Alianza de Redes Iberoamericanas de Universidades por la Sustentabilidad y el Ambiente (ARIUSA 2017) https://ariusa.net/es/primera-decada-de-la-alianza-de-redes-iberoamericanas-de-univers idades-por-la-sustentabilidad-y-el-ambiente-ariusa-2 Alliance of Networks for the Environmental Sustainability of Higher Education Institutions in Ibero-America (2018, pp. 60–74) http://www.guninetwork.org/publication/approaches-sdg-17-partnerships-sustainable-d evelopment-goals-sdgs
AAU	Website and AAU Strategic Plan 2016-2020 https://www.aau.org/about/
SDSN	Website Regional Network Australia, New Zealand and the Pacific https://www.unsdsn.org/newpageca2aed64 Getting started with the SDGs in universities: A guide for universities, higher education institutions, and the academic sector http://ap-unsdsn.org/about/
PROSPER.Net	Website https://prospernet.ias.unu.edu/ ProSPER.Net Strategies and Roadmap https://prospernet.ias.unu.edu/wp-content/uploads/ 2016/02/ProSPER.Net-Strategies-and-Roadmap_revised.pdf

References

- Steffen, W.; Rockström, J.; Richardson, K.; Lenton, T.M.; Folke, C.; Liverman, D.; Summerhayes, C.P.; Barnosky, A.D.; Cornell, S.E.; Crucifix, M.; et al. Trajectories of the Earth System in the Anthropocene. *Proc. Natl. Acad. Sci. USA* 2018, 115, 8252–8259. [CrossRef] [PubMed]
- UN General Assembly. Resolution adopted on 25 September 2015. Transforming our World: The 2030 Agenda for Sustainable Development. Available online: https://www.un.org/ga/search/view_doc.asp?symb ol=A/RES/70/1&Lang=E (accessed on 18 January 2020).

- 3. Orr, D.W. Foreword. In *Post-Sustainability and Environmental Education;* Jickling, B., Sterling, S., Eds.; Palgrave Studies in Education and the Environment; Springer Nature: Cham, Switzerland, 2017; pp. vi–xi.
- Agenda 21. United Nations Sustainable Development. In Proceedings of the United Nations Conference on Environment & Development, Rio de Janeiro, Brazil, 3–14 June 1992. Available online: https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcom pact/A_CONF.151_26_Rev.l_Vol.%20l_Agenda.pdf (accessed on 25 November 2019).
- 5. Tilbury, D. Ten Years of Education for Sustainability in Higher Education: UNESCO Commissioned Review for the Decade in Education for Sustainable Development (DESD); UNESCO: Paris, France, 2014.
- Lozano, R.; Lukman, R.; Lozano, F.J.; Huisingh, D.; Lambrechts, W. Declarations for sustainability in higher education: Becoming better leaders, through addressing the university system. J. Clean. Prod. 2013, 48, 10–19. [CrossRef]
- Leal Filho, W.; Raath, S.; Lazzarini, B.; Vargas, V.R.; de Souza, L.; Anholon, R.; Quelhas, O.L.G.; Haddad, R.; Klavins, M.; Orlovic, V.L. The role of transformation in learning and education for sustainability. *J. Clean. Prod.* 2018, 199, 286–295. [CrossRef]
- Rammel, C.; Velázquez, L.; Mader, C. Sustainability assessment in higher education institutions. In *Routledge Handbook of Higher Education for Sustainable Development*; Batt, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge: Oxon, UK, 2016; pp. 331–346.
- 9. UNESCO. Education for Sustainable Development Goals: Learning Objectives; UNESCO: Paris, France, 2017.
- 10. Leicht, A.; Heiss, J.; Byun, W.J. (Eds.) *Issues and Trends in Education for Sustainable Development;* UNESCO: Paris, France, 2018.
- 11. Adloff, F.; Neckel, S. Futures of sustainability as modernization, transformation, and control: A conceptual framework. *Sustain. Sci.* **2019**, *14*, 1015–1025. [CrossRef]
- 12. Weisser, C.R. Defining sustainability in higher education institutions: A rhetorical analysis. *Int. J. Sustain. High. Educ.* 2017, *18*, 1076–1089. [CrossRef]
- Jickling, B. Education Revisited: Creating Educational Experiences that are held, felt, and disruptive. In *Post-Sustainability and Environmental Education;* Jickling, B., Sterling, S., Eds.; Palgrave Studies in Education and the Environment; Springer Nature: Cham, Switzerland, 2017; pp. 15–30.
- Leal Filho, W.; Shiel, C.; Paço, A.; Mifsud, M.; Veiga Ávila, L.; Londero Brandli, L.; Molthan-Hill, P.; Pace, P.; Azeiteiro, U.M.; Ruiz Vargas, V.; et al. Sustainable Development Goals and sustainability teaching at universities: Falling behind or getting ahead of the pack? J. Clean. Prod. 2019, 232, 285–294. [CrossRef]
- 15. Adams, R.; Stephen, M.; Boom, K. University culture and sustainability: Designing and implementing an enabling framework. *J. Clean. Prod.* **2018**, *171*, 434–445. [CrossRef]
- IAU Horizons. April 2019; Volume 24. In focus: Universities and Agenda 2030: Engaging with the SDGs. Available online: https://iau-aiu.net/IMG/pdf/iau_horizons_vol.24.1_en_light_.pdf (accessed on 7 November 2019).
- Vilalta, J.M.; Betts, A.; Gómez, V.; Cayetano, M. (Eds.) Approaches to SDG 17. Partnerships for the Sustainable Development Goals; GUNi: Barcelona, Spain, 2018. Available online: http://www.guninetwork.org/publicatio n/approaches-sdg-17-partnerships-sustainable-development-goals-sdgs (accessed on 7 November 2019).
- Association of African Universities. AAU Strategic Plan 2016–2020 (August 2016). Available online: https:// www.aau.org/wp-content/uploads/sites/9/2016/10/AAU-STRATEGIC-PLAN-2016-2020-FINAL.pdf (accessed on 5 November 2019).
- Sáenz, O. Alliance of Networks for the Environmental Sustainability of Higher Education Institutions in Ibero-America. In *Approaches to SDG 17 Partnerships for the Sustainable Development Goals (SDGs)*; Vilalta, J.M., Betts, A., Gómez, V., Cayetano, M., Eds.; GUNi: Barcelona, Spain, 2018; pp. 60–74.
- COPERNICUS. CHARTA 2.0/2011. European Commitment to Higher Education for Sustainable Development. Available online: https://www.copernicus-alliance.org/images/Downloads/COPERNICUSCharta_2.0.pdf (accessed on 7 November 2019).
- 21. Global Universities Partnership on Environment for Sustainability. GUPES Objectives. Available online: https://www.unenvironment.org/es/node/10655 (accessed on 5 November 2019).

- SDSN Australia/Pacific. Sustainable Development Solutions Network—Australia/Pacific. In Getting Started with the SDGs in Universities: A Guide for Universities, Higher Education Institutions, and the Academic Sector; Australia, New Zealand and Pacific Edition; SDSN Australia/Pacific: Melbourne, Australia, 2017. Available online: http://ap-unsdsn.org/wp-content/uploads/University-SDG-Guide_web.pdf (accessed on 8 November 2019).
- Vilalta, J.M.; Betts, A.; Gómez, V. Higher Education's role in the 2030 agenda: The why and how of GUNi's commitment to the SDGs. In *Sustainable Development Goals: Actors and Implementation. A Report from the International Conference*; Vilalta, J.M., Betts, A., Gómez, V., Eds.; GUNi: Barcelona, Spain, 2018; pp. 10–14.
- 24. HESI. Higher Education Sustainability Initiative. Available online: https://sustainabledevelopment.un.org/ content/documents/16065HESI_info_July3_v2.pdf (accessed on 5 November 2019).
- 25. Sáenz, O.; Benayas, J. Higher Education, Environment and Sustainability in Latin America and the Caribbean. Available online: http://www.guninetwork.org/articles/higher-education-environment-and-sustainability-l atin-america-and-caribbean (accessed on 26 November 2019).
- Benayas, J.; Blanco-Portela, N. Evolution of the actions of Latin American universities to move toward sustainability and SDGs. In *Higher Education and Sustainability: Opportunities and Challenges for Achieving Sustainable Development Goals*; Azeiteiro, U.M.M., Davim, J.P., Eds.; CRC Press Taylor & Francis Group: Boca Raton, FL, USA, 2019; Chapter 2.
- 27. Giesenbauer, B.; Tegeler, M. The Transformation of Higher Education Institutions Towards Sustainability from a Systemic Perspective. In *Universities as Living Labs for Sustainable Development*; Leal Filho, W., Lange Salvia, A., Pretorius, R.W., Londero Brandli, L., Manolas, E., Alves, F., Azeiteiro, U., Rogers, J., Shiel, C., do Paco, A., Eds.; World Sustainability Series; Springer: Cham, Switzerland, 2020; pp. 637–650.
- Blanco-Portela, N.; R-Pertierra, L.; Benayas, J.; Lozano, R. Sustainability leaders' perceptions on the drivers for and the barriers to the integration of sustainability in Latin American Higher Education Institutions. *Sustainability* 2018, 10, 2954. [CrossRef]
- Van Opstal, M.; Hugé, J. Knowledge for sustainable development: A worldviews perspective. *Environ. Dev. Sustain.* 2013, 15, 687–709. [CrossRef]
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, 1–18. [CrossRef]
- Eauc. Rio+20 Treaty on Higher Education. 2014. Available online: http://www.eauc.org.uk/higher_educati on_treaty_for_rio20 (accessed on 26 November 2019).
- Leal Filho, W.; Lange Salvia, A.; Pretorius, R.W.; Londero Brandli, L.; Manolas, E.; Alves, F.; Azeiteiro, U.; Rogers, J.; Shiel, C.; do Paco, A. (Eds.) Universities as Living Labs for Sustainable Development; World Sustainability Series; Springer: Cham, Switzerland, 2020.
- Stevens, C.; Kanie, N. The transformative potential of the Sustainable Development Goals (SDGs). Int. Environ. Agreem. Polit. Law Econ. 2016, 16, 393–396. [CrossRef]
- COPERNICUS. CRE-COPERNICUS Charta. Available online: https://www.copernicus-alliance.org/images /Downloads/CRE_COPERNICUS_University_Charta.pdf (accessed on 7 November 2019).
- 35. Katiliute, E.; Daunoriene, A.; Katkute, J. Communicating the sustainability issues in higher education institutions World Wide Webs. *Procedia-Soc. Behav. Sci.* **2014**, *156*, 106–110. [CrossRef]
- Maggs, D.; Robinson, J. Recalibrating the Anthropocene. Sustainability in an Imaginary World. *Environ. Philos.* 2016, 13, 175–194. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article Alliances of Change Pushing Organizational Transformation Towards Sustainability across 13 Universities

Lisa Bohunovsky¹, Verena Radinger-Peer^{2,*} and Marianne Penker²

- ¹ Center for Global Change and Sustainability, BOKU University of Natural Resources and Life Science, 1180 Vienna, Austria; lisa.bohunovsky@boku.ac.at
- ² Institute for Sustainable Economic Development, BOKU University of Natural Resources and Life Science, 1180 Vienna, Austria; marianne.penker@boku.ac.at
- Correspondence: verena.radinger-peer@boku.ac.at

Received: 25 February 2020; Accepted: 27 March 2020; Published: 3 April 2020

Abstract: Universities are expected to play a leading role in developing and maintaining sustainability. To contribute to a systemic and dynamic understanding of organizational change that is necessary in order to play such a role, we comparatively analyzed processes of organizational changes towards sustainability across thirteen universities in Austria. This comparative analysis is based on data from guided interviews and document analysis and on validation of preliminary results via group discussion and individual comments. The results show that all universities embedded sustainability in most of their areas of activity (research, teaching, operations, organizational culture, societal engagement), but the depth of integration and the type of structural embedding varies. Especially for early changes dating back to the 1990s, academics working in the broader field of sustainability studies were those agents of change, who-without formal mandate-skillfully and proactively initiated and drove organizational transformations following an idealistic and intrinsic motivation. A timeline analysis illustrates peaks of sustainability-related changes in the years of the foundation of inter-university networks in 2011 and 2017, which acted as alliances of change. Ministry intervention in 2015 helped to bring sustainability on the agenda of those universities with less change agency. In summary, sustainability transformations across the fields of teaching, research, operations, organizational culture, and societal engagement were driven by a fruitful interplay of change agency and change alliances and to a minor degree by top-down interventions.

Keywords: universities; organizational change; higher education for sustainable development (HESD); sustainability transitions; SD; alliances

1. Introduction

The important role of universities as active stakeholders to support a paradigm shift towards sustainable development (SD) has been stressed both on the policy level and in the scientific discourse (e.g., [1]).

The contribution of universities to sustainable development is seen, for example, in developing strategic long-term visions and goals [2], in bridging different types of knowledge through inter- and transdisciplinary approaches [3,4], as well as assuming a boundary spanning role between science and society [5]. The role as 'change agent' [6,7] comes not without challenges. In order to fulfil this role, universities need to rethink their current organizational structures and societal purposes, leading to a structural transformation towards sustainability [8,9]. These processes of organizational change are expected to contribute to overarching political agendas (e.g., Education for Sustainable Development,

Sustainable Development Goals) and involve all areas of activity: learning and teaching, operations, external societal engagement, and research.

Much research has been undertaken regarding the *what* of organizational transformation, including (1) the integration of sustainability in university management practices [10,11]; (2) corporate social responsibility, sustainability reporting, and accounting [12–14]; (3) teaching and education for sustainable development [15–17]; (4) generation and advancement of scientific knowledge and other forms of knowledge [18,19]; (5) boundary spanning, participation in regional sustainability initiatives, and networking [2,5,20]; and (6) applying sustainability as an overall concept for universities as in, for example, the 'sustainable university' [4,9].

At the same time relatively little is known about *how* processes of organizational change towards SD take place [21,22]. Insights into the *how* of organizational change towards SD support the understanding on how to orchestrate, push, and/or support organizational change processes in universities. Furthermore, it supports the understanding on how, why, and when diverse internal and external factors influence these change processes towards SD. Understanding the *how* of organizational change means taking into account the deep structure and inter-personality of a university, its sub-systems, facilities, units, and departments, including their interdependencies in a systemic and dynamic understanding [23] as well as the universities' cultural orientations [24].

This makes the process of transformation particularly complex, and those who have researched or engaged in sustainability initiatives or change processes often characterize these as long, progressive, and challenging, and characterized by resistance, barriers, and contradictions [11,25,26].

The subsequent aspects have been derived from the literature to enlighten the *how* of universities' organizational change towards SD [8,9,21,22,26–28]:

- Structural transformation and entrance of SD into universities organizational structures;
- Decision making processes, leadership strategies, and strategic planning dynamics;
- Role of internal factors (e.g., institutional culture, strategic agency, relationships and power on campus);
- Role of external factors (e.g., funding/regulative bodies, networks, other higher education institutions);
- Focus on organizational learning, to explicitly investigate the process of change.

Apart from these content wise perspectives, there is a need to investigate organizational change from a systemic and dynamic perspective [23]. Thus far, only few scholars have looked at all fields of activity from an integrative perspective and on *how* questions of university change processes towards SD (e.g., [27,28]) besides more theoretical work to conceptualize organizational change (e.g., [21,29]). These seminal works enfold that there is no common path for universities towards SD, but different paths driven by diverging factors. Whereby some attention has been paid to internal and external drivers of change, there is yet a lack of (a) understanding how SD entered the respective universities and what are the influencing factors behind, (b) a detailed conceptualization of what is meant by organizational change in the context of universities (an exception states [28]), and (c) a dynamic perspective on organizational change.

The present paper aims to contribute to these research gaps by investigating the *how* question of organizational change processes towards sustainability by scrutinizing the organizational change process of 13 Austrian public universities since the 1990s. In doing so, we explicitly focus on processes of structural transformation and embedding. The authors take a dynamic perspective to address two research questions: (a) How have the 13 universities implemented SD into the universities' organizational structures? (b) How have these changes been driven by internal and external factors? The paper is empirically based on in-depth interviews with selected stakeholders form each university as well as a complementary analysis of strategic papers, visions, and guidelines. We present long-term cross-organizational evidence from 13 universities in Austria and operationalize organizational change as the depth of integration as well as the type of structural embedding. We further enlighten the question

on internal and external factors spurring organizational change processes. In doing so, the valuable contribution of the present paper is to (a) clearly define and operationalize organizational change in the context of universities and sustainable development, (b) take a dynamic perspective which enlightens change processes over the last decades, and (c) present data from multiple case studies which offer potential for generalization.

Section 2 presents the main elements of the analytical framework. Section 3 focuses on the methodology and the research context. Section 4 presents the results structured along the analytical framework outlined in Section 2. Section 5 discusses the results in light of the current academic debate and leads to the conclusions in Section 6.

2. Analytical Framework

The process of embedding sustainability into universities is discussed as a multi-faceted process of organizational learning and change, which leads to a structural transformation towards sustainability and asks for a whole-institution approach [8,29,30]. To ensure consistency and credibility, this structural change must involve all the areas of a university including research, teaching and learning, societal engagement, campus estates, and operations [28,31]. Our analytic framework—in line with the two research questions of the paper—provides (a) an operationalization and definition of organizational change, taking into account the depth of integration and types of change and (b) a conceptualization of the internal and external factors for spurring these organizational change processes.

2.1. Organizational Changes—Depth of Integration and Types of Change

When talking about organizational change, we focus on structural change, i.e., change that is embedded in universities' practices and institutions. By doing so, we set aside projects or activities that are run by an individual or a very limited number of persons and have no major organizational impact after their end. According to Ferrer-Balas et al. [28], the question of how far reaching an organizational change is, can be seen as a question

- (a) of how many different persons (groups) carry or are involved in the change process and
- (b) if the change is an optimization, improvement, or a real renewal of the system and
- (c) if changes are only initiated or also mainstreamed within the organization.

We refer to these aspects as depth of integration and operationalize them as outlined in Table 1. The scales of the depth of organizational change thus reflect the number of persons or groups that are involved in the processes described and their distribution within the universities, but also signs of mainstreaming or system-improvement/renewal, such as e.g., university-wide integration, uptake in strategic papers, setting of new standards. Thus, changes by single actors that only affect a small group of persons are ranked lower than changes carried by a variety of actors in different entities of the university or even changes mainstreamed within universities and thus affecting all its members. Moreover, the size of the 'target groups' was taken into account, e.g., changes that affect all students were ranked higher than changes that affect only a particular group of students.

Area of Activity	Depth of Integration	Ту	pe of Structural Embedding
Teaching	0 = no SD relevant classes, 1 = single SD related courses for students, 2 = SD relevant study programs, 3 = SD relevant programs plus obligatory SD relevant classes for several/all study programs	1.	Institutional changes (change of values, norms, formal regulation,
Research	 0 = no SD-specific research activities 1 = single SD research activities 2 = various, but isolated activities 3 = broad integration and/or strategic focus on SD in research activities 	2. 3. 4.	voluntary standards) Membership New organizational units Working groups
Operations	0 = no/hardly any activities 1 = single activities 2 = activities plus some kind of certification 3 = EMAS certification plus other activities 1 = low (1 of 4 types)	5.	Project/temporary events (e.g., SD days)
Organizational culture	2 = middle (2–3 of 4 types) 3 = high (all 4 types) Types: (a) integration of SD into the scope of functions of the rectorate or strong support from university management, (b) integration of SD into strategic papers, (c) establishment of a SD board or center, (d) implementation of a broad, participatory SD process * Student activities as a specific form of cultural embedding are analyzed separately.		
Societal engagement	0 = no focus on SD related societal engagement 1 = SD-service-learning projects in cooperation with government, public administration, schools 2 = specific SD-related activities such as SD day, SD report, SD public event series		

Table 1. Operationalization and definition of organizational change (own illustration).

A further aspect which has to be considered under the frame of organizational change within universities is the type of structural embedding. In our work, we differentiate five sub-categories of structural embedding (adapted and extended from Pflitsch and Radinger-Peer [32]) that also relate to the question of how far reaching the change is:

- 1. Changes in the institutional framework, which reflect changes in rules, norms, and cognition related to SD, such as the implementation of a new formal regulation or the official announcement of new voluntary standards, which legitimize new social practices in favor of sustainability or delegitimize unsustainable behavior.
- 2. Memberships in a (new) organization mean the commitment and support of a network or organization and its goals. The membership in an organization does not necessarily mean structural changes within a university, but in its best sense can lead to learning processes initiated by this membership.
- 3. New organizations understood as establishment of new independent organizations, new study programs, or departments within a university. They are characterized by assigned responsibilities, competencies, and/or rule systems and have their own administration, technical, and/or financial resources. Thus, they are seen as relatively stable and deeply embedded in universities' structures.
- 4. Working groups, which are loosely coupled groups of independent actors with a common interest, usually have a non-hierarchical form and no own resources. They are thus more fluid than formal organizations, but can initiate learning processes, especially if their members come from different parts of the university who spread the ideas.

5. Projects are institutionalized temporary events, which follow a specific purpose over a limited time, but might lead to a new structure (organization/working group). Due to the large number of SD-related activities and actors, the study does not claim to give a comprehensive overview but focusses on those activities that were reported as having led to structural changes. Especially, it does not include single classes, single research projects, or other projects that were started and ended without a structural effect.

In order to understand how these changes are initiated, the likewise influence of internal as well as external factors has to be taken into account [27,29]. Among the university-internal factors, leadership [27,33] and champions as agents of change [28,34] are ranked first in various studies. Regarding external factors, the influence of funding and regulatory organizations but also inter-organizational networks for initiating organizational change of universities towards SD are pointed out.

2.2. Internal Factors—Institutional Agency and Leadership

Institutional agency defines an actor's ability to make an impact on the social order, changing the rules, relational ties, or allocation of capital within an organization. Such actors serve as agents of legitimacy who support the creation of new institutions and reform existing institutions in ways that they deem to be appropriate and aligned with their interests. So-called 'champions' and 'frontrunners' are deemed essential for any transition to sustainability [35]. They are often characterized by holding key positions at the university (e.g., rector, study program manager, head of institute) which allow them to take agency and precipitate organizational and institutional change within the university [32,36]). Commitment, leadership, and support by those 'higher-up' in universities are crucial to progress and embed sustainability [37] because it means that new structures, incentives, and funding are put in place [38]. The role of leadership is seen in guaranteeing the mainstreaming of SD-related organizational changes in contrast to mere 'cosmetic reforms' [27].

At the same time, literature on leadership argues that leadership is not just centralized but is also dispersed in the sense that leadership opportunities are available to any member of an organization, no matter what the rank [39]. Thus, leadership can come from any corner of an organization, from senior officials to students, from academia to administrative staff [22,40]. Thereby top-down and bottom-up activities interact with and reinforce each other [22]. Hoover and Harder [26] detect in their cross case synthesis that especially in sustainability initiatives 'pointing at power' is a frequent phenomenon, that means, people tend to perceive sustainability work as someone else's responsibility. "Opportunities for leadership and possibilities for change have more to do with how one sees oneself within an institution and in relation to others there than the position one holds" [25] (page 5). Therewith, the presence and the perception of power has an important role in constructing how change takes place, and who gets involved. For analyzing the university internal factors, each organizational change reported in the interviews is attributed to agency and/or leadership of one of four groups of actors in the analysis: academia, administration, management, or students, as shown in Table 1.

2.3. External Factors—The Universities' Environment

Modern universities may be referred to as open systems, where complex and dynamic reactions between the organization and its environment mediate through its structural operations [41]. For universities, the following elements are part of their environment: (a) government as funding/subsidizing organization, (b) other higher education institutions (and the cooperation or concurrence with them as well as a role model effect) [41,42], as well as inter-organizational networks.

Universities' activities are often driven by their sources of funding [28], therefore the government as the main funding organization but also the focus of third-party funds exerts influence on universities' organizational change processes, especially also with regard to SD.

Apart from dependence-relations, which exist between university and their funding organizations, relationships and networks are highlighted rather often in the literature as having an influence on

organizational change processes of the university towards SD [26]. Interpersonal relationships and networks both appear to have a strong influence on the nature and development of initiatives on campus [43]. Networks fulfil different functions: they transmit data, information, and knowledge, facilitate the coordination of decisions [44], they support innovation [45], and can contribute to meta-effects through steering and self-organizing processes [46]. Networks can furthermore enable individuals without specific interpersonal connections to get involved, and bridge boundaries within and between institutions. On the other hand, they also have the potential to exclude people or groups and therewith influence who has access to power in the change process [26]. Based on their international study, Ruiz-Mallen and Heras [47] point out that university networks influence the sustainability discourse as well as practice at universities. In our analysis of external factors, we focus on incentives or pressures from government/funding organizations on the one hand and other universities and networks.

The presented analytical framework comprising organizational changes as well as internal and external factors of change built the basis for the further analysis of the empirical findings.

3. Methods and the Research Context

The study follows a case study approach which is recommended if the research focusses on *why* and *how* questions, if the researcher has only little control over participants' behavior, if the focused phenomenon is relevant to present time and if there is only a poorly developed data basis available [48–50]. They provide answers to *how* and *why* questions by conducting a detailed contextual analysis of underlying conditions, behavior patterns of individuals, and their relationships [51]. As single-case studies are often criticized regarding their unique conditions surrounding the specific case, multiple case studies are considered to provide more compelling and more robust evidence and more substantial analytic results, particularly if they are based on the triangulation of evidence from different sources [48] (pp. 103–119). Therefore, we opted for a multiple case and mixed methods design combining the analysis of documents, transcripts of semi-structured interviews with feedback loops of interviewees, and additional key-informants.

The case study covers 13 of 22 Austrian public universities, all of them members of the Alliance of Sustainable Universities in Austria ('Alliance'). By February 2020, 16 universities are members of the Alliance. The case study covers 13 of the 14 universities that were members in 2018. The 14th university did not respond to our request for interview. The Alliance is an informal network of universities that was founded in the beginning of 2012 and aims at promoting sustainability issues in Austrian universities. The member universities cover a spectrum of different types of universities. Table 2 gives an overview and lists the short names used in the following text. BOKU university hosts a coordination position for the Alliance, which is held by the first author. The main body of exchange and collaboration is the so-called expert group, which consists of persons who are nominated by the rectorates and usually have a major role in SD activities in their universities. The lead for joint activities and working groups is distributed among the members.

All case universities are subject to the Austrian university law of 2002 and are publicly funded. A slightly modified legal situation applies to DUK as a university of continuing education with an additional legal basis, the state of Lower Austria as a co-owner, and fee-based study programs. Studying at all other Austrian public universities is largely free of charge and for most study programs without restrictions to admission. The budget for universities is negotiated with the Austrian Federal Ministry of Education, Science and Research (BMWFW) every three years and is based on the so-called performance agreements. Public funding covers on average across the 22 public universities about 74% of the total budget [52], the largest part of the rest comes from external project funds.

ID	English Name	Short Names	Characteristics	Alliance Member Since	Interview with Member of ¹
1	University of Klagenfurt	AAU	4 faculties: technical, economics, humanities, interdisciplinary faculty (IFF)	2012	academia
2	University of Natural Resources and Life Sciences	BOKU	life science university	2012	academia
3	Danube University Krems	DUK	university for continuing education	2017	management *
4	Johannes Kepler University Linz	JKU	4 faculties: engineering & natural sci., business & social. law, medicine	2018	academia (2 IP)
5	University of Graz	KFU	full university	2012	management
6	University of Music and Performing Arts Graz	KUG	art university	2012	management
7	University of Music and Performing Arts Vienna	MDW	art university	2017	non-scientific
8	Mozarteum University	MOZ	art university	2018	management
9	Montanuniversität Leoben	MUL	university of mining 4 faculties: catholic	2018	management *
10	University of Salzburg	PLUS	theology, cultural/social, natural sciences, law	2012	academia
11	Graz University of Technology	TUG	technical university	2012	academia (2 IP)
12	University of Innsbruck	UIBK	full university	2012	academia
13	Vienna University of economics and business	WU	economics	2012	academia ²

Table 2. Universities covered by the study, background of interview partners.

¹ Academia = scientific staff; management = (vice-) rectors and staff in university management; non-scientific = persons working within administration or other non-scientific position; all interview partners except those marked with * are member of the Alliance's expert group; (2 IP) = interview with 2 persons in parallel; ² the interview partner has in the meantime left WU.

Guided interviews with at least one representative of each of the above-mentioned universities were held between March and September 2018, with a duration of 50 to 90 min each. The interview partners were mainly members of the Alliance's expert group and were thus supposed to have a good overview over SD activities as well as the SD history of their university.

The interviews were led by the first author, partly together with the second author. The former, as coordinator of the Alliance, has good insights into the activities of each university and good relations to most of the interview partners, thus the interviews could build on a certain basis of trust and mutual knowledge and understanding. The interviews focused on the process of each university (and not the Alliance), which allowed the interviewer to remain in a neutral position, as she was not involved in these internal processes. BOKU university is an exception, as all three authors are affiliated there and are to different extents also involved in SD activities. Nevertheless, it is not possible to rule out the possibility that Alliance-related activities were mentioned to a higher extent as they represented the common point of reference for both the interviewer and the interviewees.

The interviews focused on the causes, processes, and main actors behind SD at the university and the question of how it was (historically) integrated at the university.

The interviews were recorded, transcribed, and analyzed with Atlas.ti, as shown in Figure 1) (data analysis). The aim of the qualitative content analysis [53] was to extract organizational changes along the categories operationalized in the analytic framework, as shown in Table 1, and list them according to the

Area of activity;

- Year of implementation;
- Actors driving these changes.

As the background and institutional position of the interviewees differed, a certain bias regarding the change process reported cannot be precluded. Validation of preliminary results was sought by (a) presenting and discussing them in one of the Alliance's expert group meetings; (b) sending them for comments to the interview partners; (c) sending them to the student unions of each university. Moreover, the results were triangulated with information from documents and online resources. The last step of data analysis included the attribution of each change to one of the areas of activity, the year of implementation, and the persons or groups that were involved.



Figure 1. Process of data collection, analysis, and interpretation.

The interpretation of the empirical findings focused on comparatively analyzing the depth of integration and the type of structural embedding according to the analytic framework, as shown in Table 1. Addressing the second research question on factors of change, each organizational change reported in the interviews was attributed to agency and/or leadership of one of four university-internal groups of actors (academia, administration, management, students) and the timeline of changes is interpreted against events in the organizational field (external factors), as shown in Table 1.

4. Results

The first part of the results, Section 4.1, presents results on the how questions, i.e., we present evidence showing that all 13 universities embedded sustainability in research, teaching, operations, organizational culture, and/or societal engagement, and we interpret these changes in regard to their depth (Section 4.1.1), as well as in regard to types and structural effects of these changes (Section 4.1.2). The second part (Section 4.2) analyzes the second research question on internal and external factors. Moreover, the timeline of changes is interpreted along the appearance of influencing factors.

4.1. Organizational Changes in the Case Study Universities

The results show that universities made changes towards SD in at least three out of five areas of activities, thus SD is embedded in the whole institution. Nevertheless, changes vary in regard to the question of how deep these changes are structurally embedded. About half of the universities

show deep or quite deep integration in all areas (AAU, BOKU, KFU, WU, PLUS, TUG, UIBK – short names according to Table 2), i.e., changes that are built on a broad basis of involved or affected persons. Others show lower levels of integration, often combined with one or two areas of activities with no integration—usually universities that only recently became members of the Alliance or also art universities. Furthermore, in regard to types of change, i.e., changes in the institutional framework, memberships, new organizations, working groups or projects, it can be observed that most of them play a role in the universities, but their relevance in regard to structural embedding varies.

4.1.1. Depth of Integration

The analysis addressed the question on how and how deeply the 13 universities have organizationally embedded SD into their five areas of activity, i.e., research, teaching, operations, and also activities in regard to organizational culture and societal engagement. Many case universities embed SD in most of these areas and therewith follow integral approaches. Nevertheless, these organizational changes differ in regard to the depth of structural embedding. Aggregated results according the 0–3 scale of the analytical framework, as shown in Table 1, are shown in Table 3. In the following, we argue the results by giving evidence on integration within the 13 universities.

ID	University	Teaching	Research	Operations	Organizational Culture ¹	Societal Engagement
1	AAU	2	3	3	2 + s	1
2	BOKU	3 *	3	3	3 + s	2
3	DUK	2	2	0	2	1
4	JKU	2	3	0	1	0
5	KFU	2	3	3	2 + s	2
6	KUG	1	1	2	2	0
7	MDW	0	1	2	2	0
8	MOZ	0	1	1	2	0
9	MUL	2	2	0	2	0
10	PLUS	1	2	3	2 + s	1
11	TUG	1	2	2	2 + s	2
12	UIBK	2	2	1	2 + s	1
13	WU	3	2	3	3	2

 Table 3. Embedding of sustainable development (SD) in different areas of activity of the case universities.

 The numbers represent the depth of integration—for details see Table 1.

* BOKU moreover has an ESD working group, ¹ 's' means additional student activities.

The results in the area of teaching show that only two universities (BOKU, WU) deeply integrated changes towards SD in their teaching area, whereas other universities show only limited changes in this field.

By deep integration, we mean e.g., that WU established a SD class which is obligatory for (almost) all students in their first years ('Sustainable Economics'). Thus, students are at least confronted once in their studies with SD. Or BOKU, that has SD inputs in obligatory introductory classes for three study programs and established a working group on Education for Sustainable Development (ESD) in order to promote sustainability issues in teaching. Therewith, these two universities have undergone the deepest structural change, although the depicted changes are to a large extent on a project or working group level only and therefore can be considered as improvement of the current system rather than a renewal. Although a single course during a study program cannot come up to profound education for sustainable development (ESD), it can be considered as an important step in addressing

all students regardless of their program. Both universities also offer SD study programs (e.g., WU on Socio-Ecological Economics and Policy, BOKU on Environment and Bio-Resources Management), as well as single SD-relevant learning opportunities for students. Less deep forms of change are voluntary extension curricula on sustainability or student-organized classes for students of all study programs (e.g., AAU) or study programs for a limited number of students. The latter can be differentiated in study programs with a system approach on SD (AAU, KFU, and UIBK) and more focused study programs with an SD relation (DUK on sustainable buildings, JKU and MUL on energy). As these changes affect only a small part of the students, they are assigned a lower depth of organizational integration.

Changes in the area of research (such as the establishment of SD-relevant institutes, departments, or professorial chairs, internal sustainability research networks, research activities related to SD) exist in all research-oriented universities but they remain isolated in most universities, i.e., they are based on the commitment of few actors and have no specific support from university management. Art universities form an exception, as they participate in the UniNEtZ project (see below), but have no further research activities in regard to SD. Only four universities (BOKU, KFU, AAU, JKU) have a broader and more strategic SD focus in this area of activity. They name sustainability as one of their research priorities in their strategic papers—reflected by a higher rating in regard to the depth of integration. At BOKU, KFU, and AAU this strategic focus is moreover reflected in internal networks and cross-sectoral organizational entities covering various disciplines and approaches, which might be seen as another sign of relatively deep integration, but a fundamental change in structures and cultures still has to be proven.

One research activity that unites all except one university is the participation in the UniNEtZ project [54], a joint project of the Alliance universities, which officially started in 2019, but intensive preparatory work dates back to 2017. All case study universities except WU participate in this project. The project's aim is to elaborate an option report which supports the Austrian government to implement and reach the sustainable development goals. Although started as an informal activity of some Alliance universities, it quickly gained momentum and (a) led to the integration of 16 universities and more than 200 researchers, (b) achieved the embedding of SD into the performance agreements of all participating universities with the ministry, and (c) encouraged many universities to join the Alliance. The global initiative of the Sustainable Development Goals (SDGs) and the corresponding UniNEtZ project led to changes in the universities' environment. At the same time, it also has the potential to lead to system renewal as the project follows a fundamental shift towards "research guided to the 2030 Agenda" [55] (p. 113) and includes a great number of researchers who have the potential to spread this shift in their universities.

Changes in 'operations' refer to activities with regard to operational management, such as procurement, mobility, organization of meetings, etc. Changes in operations are not necessarily linked to teaching and research but are often seen as important in order to achieve consistency between universities' fields of action. In regard to the depth of integration, the existence of some kind of certification was considered as an important indicator. This contributes to a deeper structural effect as it is based on official standards and monitoring and thus affects the university's everyday life and therewith all university members. If taken seriously, implementing such schemes requires revision and redesign of routines, leading to substantial change towards SD. Moreover, it was found that changes in this area can either be the entrance point for a wider orientation towards sustainability (e.g., KUG and MDW) or are the logical consequence of changes in other areas, like research or teaching. The latter holds true for five universities that reported no or very little focus on sustainable operations before their membership in the Alliance but are strengthening such initiatives in the meantime.

In detail, five universities (AAU, BOKU, KFU, PLUS, and WU) undergo external certification of activities in this field according to the European Eco-Management and Audit Scheme (EMAS). Although usually focusing on operational issues, e.g., AAU and KFU also report on their SD research and teaching activities within EMAS, which gives the certification an even wider scope. EMAS and its international counterpart ISO 14001 are interpreted as the deepest form of integration as they require

reporting on and improvements in a wide range of topics covering energy, material resources, etc. Other universities follow more regional schemes like Ökoprofit (KUG) or the Viennese Eco-Business Plan (MDW), or schemes on a specific topic (e.g., TUG with ISO 50001 on energy management).

Changes within 'organizational culture' were categorized in four types, as shown in Table 1. As it is difficult to rank these different types of changes, the 4-degree scale reflects the assumption that the depth of integration depends on the amount of cultural changes. The more forms were observed in a university, the higher the depth of integration as well as the effect on university members. Again, BOKU and WU show the deepest integration of SD in their organizational culture as they implemented all types of organizational change in this area. Most other universities show at least two different types of changes and were ranked in the middle scale. Students' initiatives were analyzed separately, as they are often limited in time and it is difficult to say whether they have a long-lasting effect, although they definitely have a huge potential for inducing change. At about half of the universities, SD-related student initiatives were established.

In particular, one change that applies to almost all universities is the integration of SD into the scope of functions of the rectorate or strong support from university management (type a). This underlines the finding that support from actors of the university management is important at some point of the process (see below, Section 4.2.1). By integration of SD in strategic papers (type b), we mean e.g., the formulation of SD research priorities or stressing SD in the development plan (e.g., AAU, DUK, JKU, KFU, MDW). Type c covers the establishment of SD centers (i.e., new organizations with financial and personnel resources, with the explicit task to strengthen SD in the university, applicable for BOKU and WU) or SD boards (i.e., persons from different university entities are assigned to these boards, but no special financial and personnel resources, applicable for KFU, MOZ, PLUS, TUG). The latter is a more decentralized form of organizational change in this field. BOKU and WU also initiated a broad participatory SD process (type d), which was as well interpreted as a change of the organizational culture, as these lead to a stronger perception and acceptance of SD in the whole university. At JKU, it is mainly the research strategy with SD as one of three research priorities that gives a hint to a change in the organizational culture.

The area of 'societal engagement' includes changes that promote the interaction with society in regard to SD. Within the expert interviews, this area of activity was reported only fragmentary and these activities are difficult to research online, as societal engagement often is a side-product of other activities and thus difficult to differentiate from societal engagement without relation to SD. Therefore, the results in this area are least comprehensive and a simple scale to capture the depth of integration was applied.

Based on this limited information, four universities were ranked relatively high ('2') as they have specific SD-related communication formats that address a larger target group—mainly annual SD days, where students, staff, and the public are invited to learn about and celebrate SD at the universities; but also SD reporting and several public SD events were reported. Another four universities were ranked with '1', as they gave accounts of projects that are conducted by single research groups with a limited range within the universities, e.g., SD service projects, i.e., projects in cooperation with the government, public administration (ministry, country administration), or schools with a strong service component for these partners. Moreover, DUK was subsumed in this category where societal engagement is a central element of their work as their students usually are in employment and directly apply their knowledge at work—but a specific SD focus is missing.

4.1.2. Types and Structural Effects of Changes

Five types of change were differentiated in the analysis (see conceptual framework) to better understand the how-question of organizational change towards sustainability. In the following, we present examples for each of these types and draw conclusions on their relevance in regard to structural embedding of SD. Changes in the institutional framework, such as the integration of sustainability in strategic papers, can be observed in all universities. They are either implemented top-down from the university management, or they are a consequence of bottom-up engagement. Especially if implemented top-down, they might, but do not necessarily lead to changes in real-life [56]. This can, for example, be observed in the case of DUK, where SD is a focus of strategic papers, but a broad integration into the university's structures and university-members' activities is still missing. Nevertheless, these changes in the institutional framework can be the starting point for further changes and are important steps towards deep and lasting institutionalization of bottom-up initiatives.

Like changes in the institutional framework, memberships in organizations do not necessarily result in real-life changes, but in some cases they did. Especially the membership in the Alliance of Sustainable Universities and the participation in its UniNEtZ project often were reported as a starting point for further organizational changes. Besides the Alliance membership, the Graz-wide network 'Sustainability4U' and the Climate Change Centre Austria (CCCA) play important roles. The latter is a research network that promotes climate research and climate impact research and fosters collaboration in and among these fields, as well as provides society and policymakers with scientifically sound information and advice on climate-relevant topics. Furthermore, the universities take part in more specific networks like the 'Responsible Science initiative' or the 'Principles for Responsible Management Education' network. As SD-relevant memberships were not systematically researched, the list is not comprehensive and only relies on these memberships reported by interviewees during the interview.

The foundation of new organizations usually is reflected in formal documents, i.e., the institutional framework. Moreover, they are mostly equipped with financial and/or personnel resources. New organizations comprise the foundation of research institutes or departments, study programs, EMAS, but also students' organizations. All universities, except two art universities, have founded some kind of new organization. TU Graz has founded a new SD organization recently. In two cases, special organizational SD entities at the university, i.e., SD competence centers, were founded to deal with SD at and beyond the university. Such competence centers play a specific role in embedding SD in universities. The two competence centers of the case study (BOKU and WU) started broad sustainability processes and took a leading and coordinating role. Therefore, they are perceived as role models by other universities.

Despite the fact, that organizations are usually equipped with financial and personnel resources and are usually reflected in the institutional framework, changes in strategic orientations of the university management can also destabilize such structures. In one case (AAU) institutes of an interdisciplinary, SD-oriented faculty were transferred to other universities following a decision from the rectorate. Due to the same reason, AAU left the Climate Change Centre Austria. Furthermore, due to changes in strategic decisions of the rectorate, the competence center for sustainability at WU shifted the focus of the center to sustainability-related research.

In contrast to new organizations, working groups and/or projects, as more short-term and informal forms of SD integration, can also be found at all universities. Interestingly, there are no SD-related working groups at those three universities (DUK, JKU, MUL) that only recently started with their SD activities and where this process was initiated rather top-down.

4.2. Internal and External Factors of Organizational Change

In order to answer the second research question on internal and external factors, (1) each organizational change was attributed to agency and/or leadership of one of four groups of actors in the analysis: academia, administration, management, and students, and (2) the timeline of changes is interpreted against changes in the organizational environment and the therewith presumed influences. Here, we show that academics working in the broader field of sustainability studies often were agents of change, who skillfully initiated and drove organizational changes. They mainly acted proactively without formal mandate, but out of personal interest and conviction. Furthermore, a timeline analysis (in Section 4.2.2) illustrates peaks of sustainability-related changes in the years of the foundation of

inter-university networks, which acted as alliances of change, and ministry intervention, which helped to bring sustainability on the agenda of those universities with less change agency.

4.2.1. Internal Factors—Agency and Leadership by Various Actor Groups

The results show that all four actor groups play a decisive role for organizational change processes at universities, although their power and agency differ. The most prominent group in all but art universities is academic actors—numerous individual, fragmented activities from scientists in the area of environmental science and sustainability in teaching and research date back to the early 1990s and form the nucleus for later changes. The key academic institutional actors are professors, department heads, or study program managers who work in disciplines that have a proximity to SD (e.g., geography, meteorology, biology, system science) and seem to have the necessary agency and leadership abilities to initiate the alignment towards sustainability via the foundation of new institutes (e.g., AAU, JKU, UIBK), the foundation of competence centers (BOKU, WU), as well as new study programs or new specialization areas (KFU, TUG, UIBK). Based on the empirical material, the motivation of these academic actors seems to be a mixture of personal motivation and conviction, as well as their understanding of their position and power.

Nevertheless, at some point bottom-up initiatives from researchers need the support from management. At two universities (BOKU, WU), the interplay and reinforcing of academia and management was key to the initiation of comprehensive organizational change processes. Highly motivated and engaged academics, as well as a rectorate that fully supported these initiatives via new structures and additional funding, characterize these processes. Although the management's commitment and support are needed in order to successfully embed SD structures in universities, it shows that only few changes can directly be attributed to this actor group (KUG, MOZ, DUK).

The art universities show a different pattern. There, actors from management and administration have taken the lead. As there are no faculty members with a disciplinary proximity to SD at art universities, members of administration who have a strong personal connection to SD issues either take the lead and motivate the management (MDW) or they closely collaborate with—also personally motivated—members of the management.

Differences can also be detected in regard to students as a further internal actor group. At BOKU, KFU, AAU, and UIBK, they form a kind of sparring partner for institutional change agents from academia, by mutually supporting activities and by collaborating for change. On the contrary, art universities, the DUK as a continuing education university, as well as MUL and JKU, do not show strong student engagement—which might be due to a tighter timeframe of studying and/or less affinity to the topic. The lack of student support reported for MUL and JKU might be a bias from interviewees selected from faculty and management. However, students' unions were also asked for feedback to the results and only took the opportunity in a few cases (AAU, KFU, TUG, UIBK)—which might be another indicator for less engagement.

4.2.2. External Driving Factors and Their Influence on the Timeline of Changes

The analysis of the coincidence of organization change and external factors shows interesting patterns when looking at the timeline of changes, as shown in Figure 2. Changes were only singular and few until the mid-2000s. Then, a steady increase of changes can be observed until 2011, when a first peak of activities can be detected for a period of 2 years (2011–2012). This peak of changes is due to activities of seven of the universities and correlates with the foundation of the Climate Change Centre Austria (CCCA) in 2011 and the foundation of the Alliance of Sustainable Universities in Austria in the beginning of 2012—going hand-in-hand with the raising importance of SD topics in the context of universities. Especially, the Alliance was often reported as a driver of change. One of the main purposes of this network is to exchange good and best practice experiences and to support SD-related changes in its member universities.



Figure 2. Number of reported changes per year excluding memberships in Alliance and Climate Change Centre Austria (CCCA), and contributions to UniNEtZ; total for all case universities.

A second peak of organizational changes towards SD in the investigated universities occurs in 2015. It correlates with the timeframe for the negotiations of the performance agreements with the ministry. The therewith template prepared by the ministry included a new aspect of the universities' contributions towards SD. Although the ministry cannot directly impose SD on a university, it can stress the topic's importance by including it in the reporting forms that universities use. This happened in 2015, and although not explicitly mentioned in the interviews, the increasing number of changes from 2015 on might be attributed to this.

A further external driver for changes was the Agenda 2030, published and agreed on by all UN member states in 2015. By signing the Agenda, Austria obliged itself to accomplish the Sustainable Development Goals and called on its institutions to support this endeavor. The call was taken up by scientists from the Alliance. They started the UniNEtZ project, which again boosted many SD-relevant changes at the universities and led to the accession of four further universities to the Alliance.

5. Discussion

Based on the detected research gaps, the present paper aims to investigate how universities implemented organizational change towards sustainability and how these organizational changes have been driven by external and internal factors. The scientific insights gained are compared with the scientific state-of-the-art in this section.

The results show that all 13 analyzed universities have implemented SD in most or even all areas of activities, i.e., research, teaching, operations, organizational culture, and societal engagement, and thus come up to the call for an integrative approach (e.g., [21,27,28,31]), although the depth of integration varies. While Lozano [57] stated in 2015 that "in general, the implementation of SD in HEIs [Higher Education Institutions] has been compartmentalized and not holistically integrated throughout the institutions" (p. 14), most of the analyzed universities follow an integrative approach. Some of the reported changes only affect small aspects of a university, but the fact that the universities show efforts in most areas of activity indicate that the ideas are integrated holistically. Moreover, the depth of integration was taken into account—operationalized by the number of persons involved, signs of mainstreaming, or system renewal—building on Ferrer-Balas [28]. Some universities show deep changes in all areas of activity, such as AAU, BOKU, KFU, UIBK, and WU. Those universities that show lower levels of intensity and a stronger focus on single areas are, on the one hand, art universities that lack research with a proximity to the topic, or on the other hand, universities that have started the process recently and rather top-down from the university management. Nevertheless, all universities show structural embedding in at least three of the five analyzed areas of activity.

The driving forces behind these changes can be observed in many internal and external factors. Our study reveals academic staff as a major driver for change in about half of the cases—single actors who initiated changes based on their personal motivation and engagement. This adds an important aspect to Barth [27], who differentiated between student-, operations-, and unique selling-point-driven implementation processes of SD and saw the role of academic staff rather as transmitters than as factors of change. Through the intrinsic personal motivation of single researchers, in combination with their position and decision-making power, processes of structural embedding were initiated—not only in the area of research. Thus, they acted as institutional change agents [36] who make the university management support their endeavors and not vice versa [26,32]. In many cases, a certain awareness for environmental and sustainability issues was already established, though fragmented, which formed the seedbed for organizational change (foundation of new institutes, centers, new university courses or study programs).

Nevertheless, it needs a good interplay between bottom-up engagement and the university management—at least at some point of the process. Ideally, the management's endeavors are backed and supported by incentives from funding organizations. On the downside, missing or withdrawn support can end changes quickly. In this sense, our results support the findings of Heck [40] and Lozano [11] that a strong interplay of bottom-up engagement with top-down support leads to a broad embedding in various areas of activity of the university. Within all universities the university management and rectorate played a decisive role to bring SD on the agenda at some point of the process. We interpret these developments in the Austrian case as consequences of the organizations' environmental influence—in 2015 the ministry included SD into the template of the performance agreement and from 2017 reinforced its intention to extend universities' participation in the Alliance and the UniNEtZ project. This gentle pressure changed the regulative framework and induced organizational change processes in the mentioned cases. The support of the rectorate/leadership is essential to ensure continued existence of new organizational structures. Several times new developments (towards or away from) SD go hand in hand with changes of leadership on the top-management level. On the contrary, the loss of leadership support or changing strategic orientation in the university management also led to a weakening of established structures (see the example of the AAU SD faculty).

Students have a very specific role in SD processes [22,58]. First, they as a main 'target/stakeholder group' of universities have the legitimacy to call for changes towards sustainability. It is their generation that will be affected most by an un-sustainable development, which adds another form of legitimacy and an intrinsic motivation to call for changes. Nevertheless, not all students seem to have a personal motivation and interest in the topic, which seems also to depend on the study program in which they are enrolled. Second, they are very fluent groups as they usually only stay for a couple of years, which makes it difficult to provide continuity to their actions. Third, they are "easy to motivate, but difficult to organize" [58] (page 112), as their calendars and time management differs from university staff. These differences can also be seen in the results—universities with study programs related to SD tend to have more active students. In the case of PLUS, it was even them who initiated changes by proposing them to the rectorate—but then the group of students dispersed quickly.

Usually, universities started their process from one area of activity—with a further diffusion to other areas. Although it is difficult to attribute these broadening patterns simply to the membership in the Alliance, the exchange between actors from different universities, continuous confrontation with new ideas, and a certain normative pressure from within the network seem to have supported these developments. Those universities that started from engagement in operations, show similar patterns as observed by Barth [27], by focusing on changing daily routines. The initial factor varies, from single actors with or without a leading position to isomorphism [59], trying to copy a path that was taken by similar universities before. The latter holds true for art universities. Nevertheless, these universities broadened their activities incrementally (broader embedding of SD in strategic papers or in research activities, etc.). Universities starting from research activities also show a pattern towards broadening,

as they usually started with single actors, lectures, research topics, and later started with activities in operations, teaching, etc.

Apart from institutional agents and leadership, we further detect the following influencing factors in the universities' environment, which initiated structural changes towards SD: networks and alliances, and the ministry (regulative environment). The reported and observed effects (increase in changes towards SD) support the assumption that the Alliance and similar networks strengthen institutional change agents by making SD a relevant topic at the university and by enabling the learning from others of how to initiate, accelerate, and mainstream change. These findings are in line with Ruiz-Mallen and Heras [47] who emphasize that these type of (university) networks influence the sustainability discourse as well as practice at universities. Accordingly, our results reveal a correlation between the foundation of the Alliance, the CCCA, and the start of the UniNEtZ project and the number of SD-related changes at the member universities (see also Section 4.2.2). The functions of networks in transmitting information, knowledge, support of innovation, and steering processes described in the literature [44-46] is in accordance with the observed reasons for universities to join the Alliance-interest to support this topic combined with a search for support and guidance. For some universities, the membership in the Alliance stood in the beginning of their endeavors (like the participation in the UniNEtZ project), for other universities it was a logical consequence of already ongoing SD activities—but still seeking to increase effectiveness through networking.

The UniNEtZ project functioned as another re-enforcing mechanism. Initiated by the Alliance, it soon gained strong support from the ministry, which then called on other universities to join the project before and during negotiations of the performance agreements. The participation in the project made many universities also join the Alliance.

While the network of the Alliance exerts mimetic and normative influence [59] on universities, the Austrian Federal Ministry of Education, Science and Research exerts coercive power via the performance agreements, which are the basis for the negotiations between the university and the ministry. Other influencing factors with a normative effect are international/global developments such as Agenda 2030. Via this international discourse, the topic of SD got extra attention and momentum.

Before summarizing the general implications of our research, the following limitations have to be taken into account. The study focuses on public universities in Austria, as the Alliance network until now is limited to this type of higher education institution. Furthermore, the background and institutional position of the interviewees differed, therewith a certain bias regarding the change process reported cannot be precluded. Although our investigations have a dynamic perspective, even more in-depth interviews with actors from various backgrounds, inside and outside the university would be necessary to explore the processes of change in detail at each university.

Nevertheless, the following findings can be generalized to other (semi-)autonomous public universities. First, organizational change processes towards SD in universities are complex and characterized by numerous parallel developments, a diverse range of actors involved, and are influenced differently by internal and external factors. It became also evident that various development paths can be taken leading to organizational transformations towards SD. Second, the important role of individual (often, but not only academic) actors, which due to their motivation and position, enact change towards SD has to be pointed out. If single actors or single actor groups manage to act skillfully within the opportunities given by networks, university management, and, ideally, funding organizations, they can exert a huge changing power. Third, the university management and rectorate have an important enabling role, providing windows of opportunity, funding, or other types of support (e.g., foundation of new organizational units, personnel, or financial resources, incorporation into university strategies). The universities with the most comprehensive and advanced organizational transformations towards SD are those where a fruitful interplay between the rectorate and the individual (academic) actor has taken place. Fourth, the role of inter-university networks is crucial to start the dialogue on certain (SD) topics, to provide room for discussion, as well as joint-action. These SD-focused networks initiate mimetic and normative influence and state important

platforms for change agents. We hope that our findings inspire more scientific investigations on whole-institution organizational change processes towards SD, especially focusing on other national and international settings. While we operationalized organizational change as depth of integration and type of structural embedding, further research could pay more attention to how these changes are legitimized and institutionalized. Moreover, experiences from other countries and also diverse higher education institutions could further enrich the current academic discourse.

6. Conclusions

The comparative analysis illustrates thirteen Austrian universities' organizational change processes towards sustainability in the areas of teaching, research, operations, organizational culture, and societal engagement. Academics working in the broader field of sustainable studies were change agents, who skillfully and proactively initiated and drove organizational change. A timeline analysis illustrates peaks of sustainability-related changes related to the years of the foundation of three inter-university networks. These alliances of change, in a fruitful interplay with academic change agents, have further pushed organizational change towards sustainability. Management and rectorates, as well as ministry, played an important role in bringing sustainability on the agenda of those universities following later. In a nutshell, the thirteen universities have implemented organizational change towards sustainability by an interplay of intra-university change agency and inter-university change alliances, complemented by more coercive management and ministry interventions pushing sustainability transformations in the group of late followers.

Author Contributions: Conceptualization, V.R.-P., L.B., M.P.; methodology, L.B., V.R.-P.; investigation, L.B., V.R.-P.; data curation, L.B.; writing—original draft preparation, V.R.-P., L.B.; writing—review and editing, M.P., V.R.-P. All authors have read and agreed to the published version of the manuscript.

Funding: This research was partially funded by the Austrian Science Fund, grant number T761-G27.

Acknowledgments: Open Access Funding by the Austrian Science Fund (FWF). The authors grant special thanks to the three anonymous reviewers for their valuable feedback and input, which helped us to reconsider and elaborate further on main parts of the paper.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Adomssent, M.; Michelsen, G. German Academia heading for sustainability? Reflections on policy and practice in teaching, research and institutional innovations. *Environ. Educ. Res.* 2006, 12, 85–99. [CrossRef]
- Zilahy, G.; Huisingh, D. Roles of academia in Regional Sustainability Initiatives. J. Clean. Prod. 2009, 17, 1053–1056. [CrossRef]
- 3. Caniëls, M.C.J.; van den Bosch, H. The role of Higher Education Institutions in building regional innovation systems. *Pap. Reg. Sci.* 2011, *90*, 271–286. [CrossRef]
- 4. Adomßent, M.; Godemann, J.; Michelsen, G. Transferability of Approaches to Sustainable Development at Universities as a Challenge. *Int. J. Sustain. High. Educ.* **2007**, *8*, 385–402. [CrossRef]
- Pflitsch, G.; Radinger-Peer, V. Developing Boundary-Spanning Capacity for Regional Sustainability Transitions—A Comparative Case Study of the Universities of Augsburg (Germany) and Linz (Austria). Sustainability 2018, 10, 918. [CrossRef]
- Peer, V.; Stoeglehner, G. Universities as change agents for sustainability–Framing the role of knowledge transfer and generation in regional development processes. J. Clean. Prod. 2013, 44, 85–95. [CrossRef]
- Stephens, J.C.; Hernandez, M.E.; Román, M.; Graham, A.C.; Scholz, R.W. Higher education as a change agent for sustainability in different cultures and contexts. *Int. J. Sustain. High. Educ.* 2008, 9, 317–338. [CrossRef]
- 8. Tilbury, D. Education for Sustainable Development: An Expert Review of Processes and Learning; Unesco: Paris, France, 2011.
- 9. Velazquez, L.; Munguia, N.; Platt, A.; Taddei, J. Sustainable university: What can be the matter? *J. Clean. Prod.* **2006**, *14*, 810–819. [CrossRef]

- 10. Arbo, P.; Benneworth, P. Understanding the Regional Contribution of Higher Education Institutions: A Literature Review. In *OECD Education Working Papers*; OECD Publishing: Paris, France, 2007; Volume 9.
- Lozano, R. Incorporation and institutionalization of SD into universities: Breaking through barriers to change. J. Clean. Prod. 2006, 14, 787–796. [CrossRef]
- Zadek, S.; Radovich, S. Governing Collaborative Governance. Enhancing Development Outcomes by Improving Partnership Governance and Accountability. In *Working Paper 23*; John, F., Kennedy School of Government, Harvard Kennedy School, Eds.; AccountAbility and the Corporate Social Responsibility Initiative: Cambridge, MA, USA, 2006.
- 13. Lozano, R. The state of sustainability reporting in universities. *Int. J. Sustain. High. Educ.* 2011, 12, 67–78. [CrossRef]
- 14. Aras, G.; Crowther, D. Governance and sustainability: An investigation into the relationship between corporate governance and corporate sustainability. *Manag. Decis.* **2008**, *46*, 433–448. [CrossRef]
- 15. Cortese, A.D. The critical role of higher education in creating a sustainable future. *Plan. High. Educ.* **2003**, *31*, 15–22.
- Dlouhá, J.; Huisingh, D.; Barton, A. Learning networks in higher education: Universities in search of making effective regional impacts. J. Clean. Prod. 2013, 49, 5–10. [CrossRef]
- Lee, Y.S.; Schottenfeld, M.A. Internationalising Experiential Learning for Sustainable Development Education. J. Educ. Sustain. Dev. 2012, 6, 341–354. [CrossRef]
- Waas, T.; Verbruggen, A.; Wright, T. University research for sustainable development: Definition and characteristics explored. J. Clean. Prod. 2010, 18, 629–636. [CrossRef]
- Filho, W.L. About the Role of Universities and Their Contribution to Sustainable Development. *High. Educ. Policy* 2011, 24, 427–438. [CrossRef]
- 20. Dlouhá, J.; Barton, A.; Huisingh, D.; Adomssent, M. Learning for sustainable development in regional networks. J. Clean. Prod. 2013, 49, 1–4. [CrossRef]
- Baker-Shelley, A.; van Zeijl-Rozema, A.; Martens, P. A conceptual synthesis of organisational transformation: How to diagnose, and navigate, pathways for sustainability at universities? J. Clean. Prod. 2017, 145, 262–276. [CrossRef]
- 22. Stephens, J.C.; Graham, A.C. Toward an empirical research agenda for sustainability in higher education: Exploring the transition management framework. *J. Clean. Prod.* **2010**, *18*, 611–618. [CrossRef]
- Yarime, M.; Trencher, G.; Mino, T.; Scholz, R.W.; Olsson, L.; Ness, B.; Frantzeskaki, N.; Rotmans, J. Establishing sustainability science in higher education institutions: Towards an integration of academic development, institutionalization, and stakeholder collaborations. *Sustain. Sci.* 2012, *7*, 101–113. [CrossRef]
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, 1–18. [CrossRef]
- Barlett, P.F.; Chase, G.W. Sustainability on Campus: Stories and Strategies for Change; MIT Press: Cambridge, MA, USA, 2004.
- Hoover, E.; Harder, M.K. What lies beneath the surface? The hidden complexities of organizational change for sustainability in higher education. J. Clean. Prod. 2015, 106, 175–188. [CrossRef]
- Barth, M. Many roads lead to sustainability: A process-oriented analysis of change in higher education. *Int. J. Sustain. High. Educ.* 2013, 14, 160–175. [CrossRef]
- Ferrer-Balas, D.; Adachi, J.; Banas, S.; Davidson, C.I.; Hoshikoshi, A.; Mishra, A.; Motodoa, Y.; Onga, M.; Ostwald, M. An international comparative analysis of sustainability transformation across seven universities. *Int. J. Sustain. High. Educ.* 2008, *9*, 295–316. [CrossRef]
- Cebrián, G.; Grace, M.; Humphris, D. Organisational learning towards sustainability in higher education. Sustain. Account. Manag. Policy J. 2013, 4, 285–306. [CrossRef]
- Sterling, S. Higher education, sustainability, and the role of systemic learning. In *Higher Education and the Challenge of Sustainability;* Corcoran, P.B., Wals, A.E., Eds.; Kluwer Academic Publishers: Dordrecht, The Netherlands, 2004; pp. 49–70.
- Ferrer-Balas, D.; Lozano, R.; Huisingh, D.; Buckland, H.; Ysern, P.; Zilahy, G. Going beyond the rhetoric: System-wide changes in universities for sustainable societies. J. Clean. Prod. 2010, 18, 607–610. [CrossRef]
- 32. Radinger-Peer, V.; Pflitsch, G. The role of higher education institutions in regional transition paths towards sustainability. *Rev. Reg. Res.* 2017, *37*, 161–187. [CrossRef]

- 33. Fullan, M.; Scott, G. Turnaround Leadership for Higher Education; Jossey-Bass: San Francisco, CA, USA, 2009.
- 34. Ryan, A. Education for Sustainable Development and Holistic Curriculum Change. A Review and Guide; The Higher Education Academy: York, UK, 2011.
- 35. Loorbach, D.; Rotmans, J. The practice of transition management: Examples and lessons from four distinct cases. *Futures* **2010**, *42*, 237–246. [CrossRef]
- Haunschild, P.; Chandler, D. Institutional-level learning: Learning as a source of institutional change. In *The Sage Handbook of Organizational Institutionalism*; SAGE Publications: Thousand Oaks, CA, USA, 2008; pp. 624–649.
- 37. Lozano-García, F.J.; Huisingh, D.; Delgado-Fabián, M. An interconnected approach to incorporate sustainable development at Tecnológico de Monterrey. *Int. J. Sustain. High. Educ.* **2009**, *10*, 318–333. [CrossRef]
- Holmberg, J.; Svanström, M.; Lundqvist, U.; Svanström, M.; Arehag, M. The university and transformation towards sustainability. *Int. J. Sustain. High. Educ.* 2012, 13, 219–231. [CrossRef]
- 39. Ancona, D. Leadership in an Age of Uncertainty. MIT Cent. Bus. Res. Brief 2005, 6, 1-4.
- Heck, R.H.; Hallinger, P. The Study of Educational Leadership and Management. Where Does the Field Stand Today? *Educ. Manag. Adm. Leadersh.* 2005, 33, 229–244. [CrossRef]
- 41. Scott, W.R. Instituitions and Organizations; Sage: Thousande Oakes, CA, USA, 2001.
- 42. Pinheiro, R. In the Region, for the Region? A Comparative Study of the Institutionalisation of the Regional Mission of Universities; University of Oslo: Oslo, Norway, 2011.
- Bardati, D.R. The integrative role of the campus environmental audit: Experiences at Bishop's University, Canada. Int. J. Sustain. High. Educ. 2006, 7, 57–68. [CrossRef]
- 44. Lambooy, J. The transmission of knowledge, emerging networks, and the role of universities: An evolutionary approach. *Eur. Plan. Stud.* **2004**, *12*, 643–657. [CrossRef]
- 45. Mader, C. How to assess transformative performance towards sustainable development in higher education institutions. *J. Educ. Sustain. Dev.* **2012**, *6*, 79–89. [CrossRef]
- Mayntz, R.; Scharpf, F.W. Der Ansatz des aktuerzentrierten Institutionalismus. In Gesellschaftliche Selbstregelung und Steuerung; Mayntz, R., Scharpf, F.W., Eds.; Campus: Frankfurt am Main, Germany, 1995.
- 47. Ruiz-Mallén, I.; Heras, M. What Sustainability? Higher Education Institutions' Pathways to Reach the Agenda 2030 Goals. *Sustainability* **2020**, *12*, 1290. [CrossRef]
- 48. Yin, R.K. Case Study Research Design and Methods, 5th ed.; Sage Publications: Thousand Oaks, CA, USA, 2014.
- Corcoran, P.B.; Walker, K.E.; Wals, A.E.J. Case studies, make-your-case studies, and case stories: A critique of case-study methodology in sustainability in higher education. *Environ. Educ. Res.* 2004, 10, 7–21. [CrossRef]
- Kyburz-Graber, R. Case study research on higher education for sustainable development. Epistemological foundation and quality challenges. In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge: London, UK, 2016; pp. 126–141.
- Baxter, P.; Jack, S. Qualitative Case Study Methodology: Study Design and Implementation for Novice Researchers. *Qual. Rep.* 2008, 13, 544–559.
- 52. Bundesministerium für Bildung, Wissenschaft und Forschung. *Universitätsbericht* 2017; Bundesministerium für Bildung, Wissenschaft und Forschung: Wien, Austria, 2018.
- 53. Mayring, P. Qualitative Inhaltsanalyse. In *Grundlagen und Techniken*, 12th ed.; Beltz: Weinheim, Germany; Basel, Switzerland, 2010.
- Stötter, J.; Kromp-Kolb, H.; Körfgen, A.; Allerberger, F.; Lindenthal, T.; Glatz, I.; Lang, R.; Fehr, F.; Bohunovsky, L. Österreichische Universitäten übernehmen Verantwortung: Das Projekt Universitäten und Nachhaltige EntwicklungsZiele (UniNEtZ). *GAIA Ecol. Perspect. Sci. Soc.* 2019, *28*, 163–165. [CrossRef]
- Independent Group of Scientists appointed by the Secretary-General. Global Sustainable Development Report 2019: The Future is Now–Science for Achieving Sustainable Development; United Nations: New York, NY, USA, 2019.
- Lozano, R.; Lukman, R.; Lozano, F.J.; Huisingh, D.; Lambrechts, W. Declarations for sustainability in higher education: Becoming better leaders, through addressing the university system. *J. Clean. Prod.* 2013, 48, 10–19. [CrossRef]
- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]

- Lukman, R.; Glavič, P. What are the key elements of a sustainable university? *Clean Technol. Environ. Policy* 2007, 9, 103–114. [CrossRef]
- 59. DiMaggio, P.J.; Powell, W.W. The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields. *Am. Sociol. Rev.* **1983**, *48*, 147–160. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article Cross-Sector Collaboration in Higher Education Institutions (HEIs): A Critical Analysis of an Urban Sustainability Development Program

Sebastian Mehling and Nina Kolleck *

Department of Education and Psychology, Freie Universität Berlin, 14195 Berlin, Germany; sebastian.mehling@gmx.de

* Correspondence: n.kolleck@fu-berlin.de

Received: 16 May 2019; Accepted: 29 August 2019; Published: 12 September 2019

Abstract: In the last years, a shift in the promotion of sustainable development in Higher Education from a focus on universities' core areas of teaching and research to "whole institution approaches" with an emphasis on the operational management of Higher Education Institutions (HEIs) can be observed in different countries. With the aim to foster sustainability, HEIs have increasingly built cross-sectoral networks, involving not only academics but also practitioners in order to relate sustainability not only to research but also to outreach activities. Although there is an increasing body of literature evaluating such initiatives according to supposedly objective management criteria and indicators, there is still a lack of studies that investigate how the social meaning of knowledge production is (re-)negotiated in and through these partnerships. In this article, we analyze how individuals engaged in a cross-sector partnership make sense of the organizational dilemmas and ambiguities that stem from the complexity of working together across sectors in pursuit of an integrative approach to knowledge production. With the term "sector" we refer to the professional affiliations of the individuals involved in the partnerships, e.g., higher education, administration, formal education or non-governmental organizations. We focus on an illustrative cross-sectoral partnership: The Institute for Sustainable Urban Development (ISU), a collaborative project between Malmö University and Malmö's city administration to facilitate research and planning collaborations between both organizations in respect to furthering sustainable urban (re-)development and higher education in Malmö, Sweden. By employing a constant comparative approach based on Grounded Theory to analyze data collected with focus groups, semi-structured qualitative interviews and document analysis, we claim that rather than entering a partnership with predefined identities, values and sectoral or professional preferences, individuals engage in a narrative struggle about the organizational character of their partnership. Accordingly, an important avenue for investigating cross-sector partnerships is to explore the constructive dilemma of different organizing principles in a cross-sector partnership, and the way people negotiate the boundaries between them. For the cross-sector partnership studied the constructive dilemma for those engaging in it was to separate and link project, organizational and network organizing principles in their work. Implementing whole institutions approaches in order to promote sustainable development in and through HEIs, would accordingly profit much from a deeper analytical investigation of the process of navigating professional identities and organizational narrative(s) in boundary-spanning, cross-sector partnerships.

Keywords: higher education institutions (HEIs); cross-sector collaboration; multi-professional collaboration; transdisciplinary research; narrative analysis; sensemaking; whole institution approach; organizational networks; constant comparative analysis

1. Introduction

Cross-sector partnerships of Higher Education Institutions (HEIs) are in high demand. They promise to produce more excellent and relevant scientific knowledge (e.g., [1], (p. 11); [2], (p. 12)), higher rates and quality of technological innovation in service provision and product development (e.g., [3–7]), more efficient use of resources through collaboration (e.g., [8–11]) and the implementation and diffusion of innovations in Higher Education such as Education for Sustainable Development (ESD) (Authors), to name just a few of the anticipated benefits. Cross-sector partnerships are based on a whole institution approach, which considers that universities operate within complex environments and that all parts of this environment need to be considered when implementing sustainability strategies [12]. Hence, when realizing cross-sector partnerships or collaboration initiatives related to sustainability issues, a great variety of actors need to be involved.

While, however, an increasing body of literature evaluates such partnerships according to supposedly objective management criteria and indicators (e.g., [13,14]), there is still a lack of studies that investigate how the social value or meaning of organizing research and education is (re-)negotiated by people engaging in these partnerships. In this article, we seek to address this gap of research by searching for answers to the question: How do individuals engaged in a cross-sector partnership make sense of the organizational dilemmas and ambiguities that stem from the complexity of working together across sectors in pursuit of an integrative approach to knowledge production?

We do so by collecting data through semi-structured qualitative interviews, focus groups as well as documents (i.e., program documents, partner program documents and literature about the immediate context of the cross-sector partnership) and applying an analytical perspective adapted from sensemaking studies into organization. Techniques of open coding and focused coding following Grounded Theory (GT) are used to analyze the data. With GT we apply a qualitative methodological technique which aims at developing theoretical approaches through analyzing qualitative data. The data analysis is conducted in different coding steps and results in the identification of overall theoretical dimensions.

We define cross-sector partnerships as initiatives that aim at improving knowledge production through the facilitation of exchange relationships and processes between actors associated with different sectors. In doing so, we focus specifically on a cross-sector partnership which involves higher education actors and which follows the overall aim to foster sustainability in higher education and urban development in Malmö, in Sweden. Hence, this study is also important for research in the field of Education for Sustainable Development (ESD). In the last years, ESD has become an important field for HEIs, not least because it plays a fundamental role in achieving the 2030 Development Agenda and related Sustainable Development Goals (SDGs), but also because the concept promises an integration of sustainability into teaching and learning in educational institutions and curricula with the overall aim to empower people to "foresee, face up to and solve the problems that threaten life on our planet" (Authors). As an interdisciplinary approach, ESD supports the skills and knowledge to contribute to sustainable and ethically responsible society. Implementing ESD in HEIs involves pursuing a whole institutions approach which promotes interdisciplinary knowledge and the development of knowledge of local issues and its interrelatedness with global issues. However, our research shows that implementing sustainable development in HEIs not only requires a whole institutions approach, but that such an approach itself should more specifically focus on exploring the dynamic process of negotiating professional and organizational identities within boundary-spanning, cross-sector partnerships.

2. Background: Cross-Sector Partnerships at the Heart of Normative Debates about University Reform in the Context of Education for Sustainable Development

Today, big data, digitalization, climate change, socio-economic globalization and the fragmentation of public spheres have challenged the university's central position in research and (higher) education and its privileged position as an autonomous "republic of science" [15]. Universities are facing

institutional change and instability, which have forced them to re-examine their identity, rules and norms [16] (p. 7). Universities must defend themselves within public debates about their role in society, how they are to be justified and made accountable, how they are legitimized as sustainable actors in society and what kind of relationship they should have with other types of institutions [16]. Cross-sector partnerships are an important arena for these debates about sense and purpose of the university as a central institution for knowledge production in late-modern societies. This is because what emerges in the discourse of cross-sector partnership or collaboration proponents is often an argument about universities needing to engage in cross-sector partnerships with other societal actors in order to become socially, politically and economically more relevant cf. [17,18]. From this perspective, cross-sector partnerships are considered as a new modus of organizing knowledge beyond the university, to co-design their research and training together with social stakeholders, in order to connect research and teaching to "real-life" experiences cf. [19] and to jointly address common societal challenges or solve current problems such as those related to sustainability. When studying cross-sector partnerships and the way they organize research and education, one has to take into consideration debates around organizing and reforming knowledge production in late-modern societies. According to an influential strand in these debates, it is crucial that the university or academy, is modernized, leading it out of the ivory tower and into the center of a vibrant, innovative 'knowledge society' [4,20-22].

While HEIs have become more and more important in the search for a more sustainable future in the last years, many universities are still tackling ESD in a compartmentalized manner, i.e., confining it to specific courses or teaching practices and treating the topic as isolated from research and practice beyond its own institution. From a whole institution approach, instead, HEIs can optimize their ESD practice and their role as agents of change by involving actors from different fields of activity and building collaborative spaces with students, academics, managers and actors from practice to critically reflect sustainability and the implementation of ESD. This is because the whole institutions approach according to D'Andrea and Gosling is an organization-based approach (as opposed to individual or cultural approaches) towards understanding and inducing change in knowledge production and allows therefore to explore research and teaching as part of a complex, inter-disciplinary institution [23]. Instead of focusing on one discipline or another, it takes the whole HEI as the unit of analysis and allows therefore for an "interdisciplinary critical enquiry" (Ibid.), which highlights not only the interaction between disciplines but also between knowledge and the social context of its production and application in an attempt to conceptualize the ongoing transformation and reform of HEIs as a complex process. At the center of this process are questions about the (re-organization) of internal boundaries between disciplines, as much as external boundaries between the HEI and its organizational environment, both of which are at stake when cross sector partnerships between university and non-academic partners are organized.

In this context, the necessity for university reform in general and organizing research and education through cross-sector partnerships in particular is derived from ideas of grassroots democracy, inclusive ways of knowing and more just and liberating modes of organizing knowledge [24–28]. These reforms imply a re-evaluation of practical knowledge, non-scientific expertise and alternative, collaborative and less formalized ways of knowing and learning [29,30]. Trans-disciplinarity, as a principle for cross-sectoral knowledge partnerships, with a broad inclusion of stakeholders from different disciplines, sectors, classes, communities and cultures is often seen as a welcome strategy to answer current calls for more direct citizen involvement in knowledge production in HEIS [22,31,32].

However, there is also a critical stance towards cross-sector partnerships that argues that through engaging more in so-called collaborative projects across sectors, formerly public organizations such as universities would face the danger of being undermined and colonized by certain social interests and economic or political power [33,34]. From this perspective, cross-sector 'stakeholder participation' are further regarded as a euphemism for clientelist and exclusionary coalitions between elite fractions of society, who marginalize lower-class stakeholders and lack public accountability and democratic representation [35–37]. For these critical voices the normative discourse on cross-sector partnerships

provide a tool for transforming political societies and class conflicts into a pacified, collaborative knowledge economy where everything is governed by dictates of growth and innovation, and which is accordingly also linked to technocratic ideas of developing administrative and market-based solutions to fundamentally political problems [38,39].

In this article, we investigate how these different repertoires of meaning within the broader debate about cross-sector partnerships are used, understood and acted upon the main stakeholders involved in cross-sector partnerships, or more specifically, how the conflicts, tensions and dilemmas between the several strands in the debate allow them to intersubjectively construct and negotiate a tentative yet workable common ground. Thus, our aim is to better understand how a set of heterogeneous partners create shared meaning through organizing scientific and educational work. Instead of adding to the many studies of cross-sector partnerships that evaluate them according to predefined and supposedly objective evaluative criteria, this study addresses the need for a more specific understanding how social value gets defined in and through cross-sector partnerships [40]. More specifically, we research cross-sector partnerships as process-in-formation, drawing attention to the narrative dimension of organizing in which actors make sense of the organizational dilemmas and ambiguities that stem from the complexity of working together across sectors. In this way we contribute to a better understanding how professional identities and organizational narratives interconnect within a complex institutions such as a HEI, especially when it engages in cross-sector partnerships beyond its institutional boundaries for the sake of promoting an institutional as well as social transformation towards more sustainable futures. Doing so we supply a more precise application of the whole institutions approach in empirical research, as well as we contribute to a better understanding of the institutional change happening in HEIs under the banner of sustainable development.

3. Theoretical and Methodological Orientation

3.1. Constant Comparative Approach Based on Grounded Theorizing

In order to contribute to a better understanding of how individuals engaged in cross-sector partnership make sense of the organizational dilemmas and ambiguities that stem from the complexity of working together across sectors in pursuit of an integrative approach to knowledge production, this study uses a comparative approach based on Grounded Theory. More specifically, it builds upon a further development of Glaser and Strauss' ideas provided by Goldkuhl and Cronholm, who adapted grounded theorizing and their focus on continuous and mutual refinement of research interest, empirical data and existing theories, in order to arrive at a conceptual treatment of the social phenomenon of interest that is context-sensitive (or has a close fit to the data), but still relates its substantive account to the more abstract and general dimensions of formal theory ([41],p. 194).

In the constant comparative approach to grounded theorizing substantive and formal theoretical concepts are related to each other in mutually constitutive ways, so as to form what Glaser and Strauss have called a middle-range theory [42] (p. 33). The notion of middle-range theory is derived from earlier work by Karl Merton [43], who developed his concept of a theoretical middle-range to argue in favor of special theories with limited conceptual range to deal with delimited aspects of social phenomena, because they would provide fruitful hypotheses that guide further cycles of empirical investigation and theorizing, rather than suggesting theoretical closure ([43], p. 448, p. 457; cf.; [44], (p. 628)). In this way middle-range theories would contribute to an evolving body of theorizing that is grounded in empirical investigations as well as incorporating aspects of formal theory so as to formulate hypotheses and analytical frameworks in close dialogue with established theory to contribute to further cycles of observing, analyzing and theorizing (Ibid. p. 448; [45], (p. 8)).

Following this line of research, this article aims to scrutinize the meaning of important concepts related to cross-sector partnerships, such as knowledge, actors/sectors and partnership from a bottom-up or emic-to-etic (context-sensitive to general) approach to theorizing. However, in opposition to classic, inductive grounded theorizing and more in accordance with later refinement of the research approach

by Corbin and Strauss [46], as well as Fram [47], Charmaz [48] and Goldkuhl and Cronholm [41], we do not develop central theoretical concepts in isolation from prior work, but include them as frame of reference [41].

3.2. Sensemaking

We combine the methodological approach outlined above with the analytical approach called 'sensemaking' developed by Weick [49] and others. More specifically, we apply Taylor and van Every's analytical framework to study cross-sector partnerships in the context of people trying to make sense of their organization by narrating to each other their respective roles, their relationship and modes of exchange around a joint object of value ([50], p. 46).

It is of particular interest how actors use social background knowledge (and the normative weight they carry) to narratively frame and script their communication concerning the organization of their cross-sector collaboration. In order to investigate how actors' personal identities, group relationships and broader social affiliations are linked through communicative sensemaking, we draw on Strauss's [51] conception of organizations as arenas in which actors interact with each other through referencing (and constructing) the broader social worlds they identify with. For the purpose of this study, arenas and social worlds are used as concepts to approach the cross-sector collaboration as an arena in which actors engage each other and make sense of their interaction while linking their activities to a broader social world beyond their immediate situation. However, the broader social world beyond their situation is not understood here as something that exists independently from the situated arena, but is (re-)constructed in and through the interaction and communication taking place within it ([52], p. 207)

Bringing the approach of sensemaking together with Strauss's concepts of arenas and social worlds, a cross-sector partnership can thus be perceived as an arena in which differences and similarities are affirmed, negotiated, contested and/or transformed. This is done by way of actors in the arena interactively, or communicatively negotiating the social meaning of their central objects of value, the modes in which they exchange them, the roles they take or are ascribed within this exchange and the organizational relationships that are constructed to facilitate their interaction and communication.

This meaning-making or sensemaking aspect of organizing a cross-sector partnership is approached within this study as "boundary-making". The very act of organizing a cross-sector partnership helps those involved to define what separates and links the various, conflicting dimensions of their identities (personal, interactional, collective) as part of a broader, normative social story about what separates and links us. Latour [53] called it the story of progress about that "which holds us all together" ([53]). A cross-sector partnership can then be investigated as linked to negotiating the meaning of knowledge (and related objects and concepts) within normative, as well as strategic story-telling about social progress, and the way it aims at affirming and transforming certain conceptual as much as normative boundaries between objects, subjects, modes of exchange, roles and social positions, as well as forms of organizing. In doing so, we consider boundaries as an essential part of keeping the struggle between actors ongoing, so they can keep searching for answers about that which separates and links them in organizing the cross-sector partnership and keeping on organizing it cf. [53]. This struggle about how to define the boundary and communicate about its canonicity as well as their permeability or breakability [54], is what empowers them to think and act socially, as well as to organize. This means to focus the analytical exploration on how organizing a cross-sector partnership enables those involved to draw on conceptual boundaries in and through their interaction and communication about the meaning of their partnership.

4. Methods

4.1. Case Selection

The Institute for Sustainable Urban Development (ISU) in Malmö (Sweden) was selected to answer the research question because it presents a cooperative project between university and city administration to facilitate collaboration between both parent organizations in respect to sustainable urban development. Based on these general objectives ISU engages in various annual activities, most notably organizing networking events, international conferences, and a thesis matchmaking program in which the university students pick a topic relevant for the Malmö administration for their bachelor or master thesis. The results from these theses are then presented at a so-called Urban Strategic Forum between academics and administrators with the hope to spawn more comprehensive collaborative efforts between research, education and urban planning. Altogether, ISU appears to be an illustrative and typical case study to analyze the way, individuals engaged in cross-sector partnerships make sense of the organizational dilemmas and ambiguities that stem from the complexity of working together across sectors in pursuit of an integrative approach to knowledge production.

4.2. Data Collection

Empirical data was collected through 15 interviews with cross-sector partnership core members and their partners to observe the cross-sector partnerships as a collection of individuals and their accounts of the social reality of the cross-sector partnership. We use references to the interviews throughout the analysis and refer to the different interviewees by using an anonymized acronym for each different respondent followed by a number of the paragraph of the interview transcript we are referencing (e.g., BM P46). Furthermore, two focus groups with all the employees of the cross-sector partnership were conducted with the aim to observe the cross-sector partnership as an organizational community. Focus groups are particularly suited to "explore group characteristics and dynamics as relevant constitutive forces in the construction of meaning and the practice of social life" ([55], (p. 902); cf. [56], (p. 315)). The biggest advantage of focus groups is to enable the researcher to observe collective orientations in the process of formation, because in the group conversation individuals are forced to take positions and defend them vis-a-vis others, which allows the dynamic and interactive character of sensemaking to unfold and become visible to the researcher (cf. [57], (p. 582); cf. [58] (p. 197); [59] (p. 294)).

With respect to the data obtained from individual interviews, focus groups also serve to refine and cross-validate (or contextualize) the preliminary analysis derived from these interviews ([60], (p. 31); [61]). Thus, they address the requirement of inductive, comparative theorizing to gradually build theoretical concepts and categories from continuous iterations between data collection, analysis, validation of preliminary concepts and identification of additional data and collection sites necessary for continuing the research cycle until sufficient plausibility (or saturation) of emerging concepts is reached (cf. [48], (p. 89); [62], (p. 189)). However, the focus groups are not referenced directly within the analysis below (as opposed to the interviews), as they have been used more generally to inform the direction of the research and for validation.

Finally, document analysis was conducted of cross-sector partnership program documents, partner program documents and literature about the immediate context of the cross-sector partnership to observe the cross-sector partnership as an organizational text for internal use and outside audiences and to better understand the cross-sector partnership within its immediate spatial, organizational and temporal context. Some of the program documents have been used below in the analytical section as e.g. [63,64], which are program documents and meeting notes as we also specified in the bibliography more clearly. The document analysis of program documents was coded alongside the interviews according to the coding procedure outlined below. It served the purpose of gaining deeper insights and to validate the empirical results gained with focus groups and interview analyses.

4.3. Data Analysis: Open Coding and Focused Coding

The coding procedure involves two coding cycles, which build on each other. These cycles are: initial or open coding and focused coding.

During the initial coding, Ryan and Bernard's technique for identifying themes ([65], (pp. 89–93); [66], (p. 139) is applied. Identifying themes and applying more descriptive (than analytical) coding can be regarded as a good starting point for theorizing, because it allows for thinking in more coherent story lines together and getting a feel for the patterns in language use [66]. Concretely this means that during the open coding cycle we look for repetitions, indigenous concepts, exogenous concepts, metaphors, analogies, transitions, linguistic connectors, dilemmas, establishment of similarity and difference, as well as silences, or the de-emphasize of certain issues. The thematic coding strategy after Ryan and Bernard are thereby used as lenses, meaning they informed where to look and what to look for, rather than resulting in direct coding [65]. For the actual coding in this initial open coding cycle, action-oriented and descriptive codes are used that are closer to the text and preferably derived from the text in vivo, so as preserve the narrative meaning of the coded fragments as much as possible, cf. [48], (p. 120). This process generated initial, descriptive codes of relatively large and coherent segments of the texts, which then were further developed conceptually within memos attached to the coded text passages.

Focused coding is as a more systematical approach to gradually build abstract categories out of the initial descriptive or open codes through constant comparison of sections of the data, codes and memos (and emerging concepts therein) with each other and across texts from interviews, focus groups and documents (cf. [67], (p. 96); [66], (p. 159)). This process especially involves assessing codes according to their conceptual value, i.e., their ability to reveal larger patterns in between statements, and to select and consolidate those that are more promising for further analysis and development in memos [48], (p. 144). It also involves consolidation of codes into groupings of codes (or categories) so as to gain deeper insight into their specific properties, as well as their relationship among each other (which then is further developed through memo writing about categories, re-coding statements on the basis of focused codes). By comparing codes and arranging them into categories, the themes and situations initially focused on are compared with each other and allowed for a more comprehensive understanding of the data. In this fashion a conceptual system of categories is gradually build out of initial observations and preliminary analyses.

5. Findings

5.1. Malmö: A Story of Collaborative Transformation of an Industrial City to a Sustainable KNOWLEDGE City

The Institute for Sustainable Urban Development (ISU) was officially founded in 2007 as a joint, collaborative project between Malmö University and Malmö's city administration to facilitate research and planning collaborations between both organizations in respect to furthering the sustainable urban (re-)development of Malmö (e.g., [68]).

During the 1990s, Malmö faced a deep economic and social crisis with vast numbers of unemployed, civil unrest in its nation-wide notorious banlieues, such as Rosengard, and the rapid closing down of its traditional manufacturing industries; [69] (p. 50). The establishment of the university in the former shipyards of Malmö's central Western Harbor district was a cornerstone in the transformation strategy of the city administration (OZN P11; ID P21; DD P39) (These acronyms indicate a single interviewee, but in order to protect their identity they have been randomized. The P followed by a number indicates the paragraph in the transcript which is referenced). The city's transformation plan strategically focused on sustainable urban development, which was also ingrained in the statutes of the newly founded university. It was to concentrate its research and teaching on the conservation of nature and natural resources and, consequently, all issues relating to sustainable urban development [70].Through its university, Malmö was to become a "multicultural city of knowledge" that would be "attractive,

green, resource and ecologically efficient" [70] and able to connect well-educated people through excellent education, innovative employment opportunities and high-quality (sustainable) housing [71] (pp. 16–17). From the beginning, the relationship between the university and city was very close, whereby the university had a clear mission to contribute to the city's development in a very collaborative, applied and vocational manner.

While Malmö's transformation into a smart and sustainable city is closely linked to the economic crisis in Sweden in the 1990s, it is also influenced by the national Swedish research policy. This changed from having a primarily academic focus to become a governance tool in the efforts to develop/transform Sweden into a "knowledge-based economy" [72], (p. 246). In 1997 a string of reforms were undertaken so as to focus Swedish universities on solving concrete social problems in collaboration with social stakeholders and to match their education to the needs of the labor market (DD P97–99; IZ P13) [72]. At the same time, the core funding of Swedish universities was cut back due to austerity policies. It was substituted through external, competitive funding through research council and sectoral funding agencies, such as the Swedish Foundation for Strategic Research (SSR) (https://strategiska.se/en/) or the Swedish Foundation for Strategic Environmental Research (MISTRA) (https://www.mistra.org/en/). Both are among the largest research funding institutions in Sweden. They fund industrial doctorates, applied research centers at universities and collaborative networks between universities, companies, public agencies and other stakeholders to collaboratively engage in research areas considered of strategic importance for "Sweden's future competitiveness" and its "good living environment", as well as promote opportunities for industrial application of research (SSF, strategiska.se; MISTRA, mistra.org).

5.2. From "Center" to "Institute": Continuously Institutionalizing Collaboration between University and Administration

The history and development of ISU is closely connected to the development of Malmö's transformation story. The idea to institutionalize a collaboration between city administration and university originated with the international exhibition for sustainable housing Bo01 (The name of the exhibition is derived from "Bo", the Swedish word for "to dwell" and the year of its opening 2001 (Austin 2013)) in Western Harbor in 2001–2002 (DD P17, BM P50). This led to the implementation of a project called "Urban Ecological Knowledge and Development Center in Malmö" (called SEKUM) in 2003. Its objective was "to promote sustainable urban development from a local, regional and global perspective as well as to increase cooperation between the different parties [i.e., city administration and university]" ([70] own translation). Its vision was to "gradually develop into a powerful hub and platform for cooperation between industry, universities and colleges, public organizations and other organizations" [70].

This grander vision of facilitating collaboration for making Malmö a smart, successful and sustainable city was accompanied by high hopes for the opportunities that such a center could offer the parties involved. In an internal document [73] it was discussed how SEKUM would enable the city to get excellent knowledge for their city planning (through commissioned theses and later an "Industrial PhD" in cooperation with Malmö University). It would enable the university to increase the quantity of their externally funded research projects, which in turn would increase the quality and relevance of Malmö University College's research and training. It was further hoped that in the longer run SEKUM would gradually develop from a temporary joint project by city administration and the university into a hub and think tank for sustainability in theory and practice. It would not only be financed externally (through selling services, raising membership fees and collecting research funds), but would simultaneously create, finance or support specific development projects by the Malmö administration as well as produce excellent (marketable) research and training about sustainable urban development (Ibid.). After a start-up project phase, it was anticipated that SEKUM would have made major leeway towards the long-term goal of being an independently financed organization that maintained itself by attracting external funding, by providing paid-for training courses and study

trips [73], and by receiving membership fees from the private sectors that would pay for being included in the network [63].

Yet none of these funding models materialized within the initial set-up period of SEKUM (2003–2005)—no external income was generated at all, according to the budgets of 2003–2005. Thus, the premise for continuing SEKUM within a new framework as the Institute for Sustainable Urban Development was to shift operations from the initial "joint project" towards an "association" (förening) which would be able to continue and scale up SEKUM's activities in the long term [74]. This meant on the one hand that the new institute or association would (gradually) cease to be primarily an internal project between city administration and university and instead open up to outside actors, organizations and businesses active in the field of urban sustainable development. The vision guiding this envisaged change was for ISU to become a "nationally and internationally leading competence center for sustainability, and a key player for the transformation of urban areas in the region" [63]. This would also mean that, instead of being dependent on continuous financing from its founding organizations, it would finance itself through membership fees and selling services [63]. The organizational example that was often mentioned as the future ideal for the new institute was MINC (Malmö Incubator), which was (and still is) a city-run, self-financed but non-profit "hub for startups and entrepreneurs". In later phases of the discussion about the new institute, however, a new strategy emerged that shifted significantly away from a commercially successful incubator with a broad network of actors from the business sector toward "an institution of learning and research for sustainable urban development" [75] that would be financed through public funds (such as EU INTERREG Program) and research grants.

Despite this suggested organizational change, the Institute started gradually in 2006 (and was officially founded in 2007) in the same organizational framework that SEKUM operated in, i.e., as a collaborative project between Malmö City and Malmö University [76], (p. 5)—an arrangement that remained in place until the time of the field research in 2014.

5.3. Organizing the Cross-Sector Partnership

From its inauguration ISU has struggled to define its structural framework. On the one hand, it is organized as a project, meaning it does not have "its own organizational number" (BM P25; LI P42). As joint project without legal entity and own organizational status, i.e., tax number, ISU does not have its own budget, staff or office. Instead ISU's staff and accounting is located at the university's department for Innovation and Development ([68] (p. 15); BM P25–27). This does not mean that the department has any influence on the strategic steering and decision-making within ISU: the department is more or less an administrative "paper boss" (BM P26). The ISU Board, consisting of the heads of university and city departments (six in total, three from department in the city administration and three from university departments) and headed by a chairwoman (linked to the central university administration), decide on ISU's focus and activities. This arrangement is based on the joint partnership agreement that was renewed in 2006 for 5 years, and since then on a 3-year basis (cf. [76,77]). ISU's executive director develops a yearly report on its activities of the current and a plan for the next year (BM P13). Its project structure is congruent with the goal of ISU to facilitate bilateral collaborations between the partner organizations, as is also its employment structure of not having its own employees-except of an executive director. Its other staff, the four part-time positions of the so-called "ISU boundary agents", is employed through ISU funds at their home department in either the university or city administration.

The ISU Board is, according to the partnership agreements, the main steering body of ISU. It meets approximately once a month (ONG P17) and its function is to develop ISU's overall strategic program and decide what kind of concrete activities and projects should be focused on in the annual plan, and how accordingly the annual budget of ISU should be allocated (Ibid.; ID P11). Another important task of the board members is to "anchor" the institute in their respective department in either city administration or university (Ibid.). This task is not clearly defined, but for some of the board members it means to involve ISU within specific projects of their departments relating to sustainable urban development (LL P33), while other board members interpret this role more in the sense of making sure
that ISU works in a way that fits with the work tasks and style of the department. In a more general way, though, this task of "anchoring" in their respective departments also simply means for the heads of departments to build a tighter network among themselves, to meet regularly as the ISU Board and to exchange information about what the departments are working on or what kind of project funds they are applying for so as to increase opportunities for cooperation between the departments (ONG P25). In this way ISU forms as a "neutral area" in between the departments:

"And ISU is, I think regarded by us (...) as a rather neutral area. So, if in the Board we often say: 'Let's leave this to ISU', I think we are at the same time saying that this is a question of common interest and no one has the leading part, not the university, nor the city (...)". (DD P29)

However, this quote also sheds light on the difficulty of the Board to assume a leadership position in respect to ISU, when their perspective is to have it as their "neutral meeting ground" and leave things of common interest to ISU, where no one of the city or the university would take the role of "the leading part". At the same time, leaving things of common interest to ISU suggests that indeed ISU would be more than a neutral area, but would at the simultaneously be an entity of its own, that could take over and take the lead on the issues of common interest to the heads of departments. However, this attitude of leaving things to be developed by ISU conflicts with the strategic leadership role of the board. For example, the executive director stated concerning the Board's idea to develop the position of the "boundary agent" within ISU:

"I was not taking part of bringing forward the boundary agent concept. It was put on my lap and I had to take care of it, but I didn't like the framework or the lack of framework I should say. But I should just take care of it and I should just fix it. The Board was like: 'But you are the director, you should do this '—'Yes, but what did you want?'—And nobody was there either. (...) I should just fix it and then the boundary agents, that have been assigned, also come to me and ask: 'What are we gonna do? '—'I don't know what you should do'''. (BM P237)

This problem concerning a strategic decision of the board that subsequently no one took responsibility for, so as to 'leave it to ISU' is thus coupled with pressure on the part of the director and the boundary agents to fix it, to make it work, without exactly knowing what the board's criteria are for determining whether something works or not. On the one hand the vague leadership emanating on the board does create space for the boundary agents to develop their own cross-organizational team. As one boundary agent recalls:

"I think we, when we started this project, the purpose was kind of vague, what we were supposed to do. So, we developed that purpose ourselves, after trying, discussing, thinking about it a lot in our group. And we have a really interesting collaboration and a good collaboration in this boundary agent group, which I think makes us a very strong entity.". (IR P36)

Thus, one could say, because of the lack of leadership and organizational coherence within ISU, the boundary agents were free to develop their collaboration and succeeded to establish a close group identity and professional network across the different sectors and organizations. However, as they are also considered ISU employees they are simultaneously confronted with the expectation to deliver on ISU's envisaged project outcomes as specified in the annual partnership agreement and ISU project plan. The two boundary agents from the city administration were particularly frustrated and admitted that they recently considered stopping to work with ISU:

"I was considering if I want to continue already. This is mainly where I felt there was no common picture. We—as in [the director] as one unit, the steering board as one unit, the boundary agents as one unit—have no common idea, vision, what do we want to do with

ISU, how do we get there? (...) It has happened that from the steering committee's side suddenly it's like: 'No, this is not at all what you're supposed to do!' And then we do not get an explanation what is it they expect from us". (QY P404)

The organizational dilemma demonstrated in this quotation constructs ISU as a self-organizing network of professionals, which, at the same time, is subjected to implicit expectations to produce concrete project outcomes. This dilemma is further expressed by the executive director, when she describes her role as being required to lead without having the authority to do so:

"I would like to become better in how to lead without being a boss. Because, I have to lead but when you are not somebody's boss, it is extremely difficult to lead and to get the person to do something that you want them to do when you are not their boss and you actually don't have anything to say about them, but it is a part of your work task to lead". (BM P172)

This dilemma of ISU being a somewhat leaderless initiative is closely linked to the contradictory identity of ISU being a neutral meeting and networking place between professionals, a tightly budgeted project expected to produce immediately beneficial results for the partner organizations, and a viable organizational entity with clear mission and coherent boundaries. This dilemma culminates in the position of the director, who was specifically chosen by the board because she is considered a networker, rather than a leader:

"You can also see how ISU was working depending on who has been the director of ISU. For example, the first we had this woman [who] was very much involved in building ISU as a sort of independent institution (...) that was more or less independent of these networks inside the university and city administration. It had a life of its own. (...) But on the other hand, we had an organization that from an organizational point of view was working quite well. But no one—even I, who was sitting in the board—knew what was going on. And then we changed the director, so now we have a director with more broker competences (...). She has a very good competence to work in networks, to connect different people with each other and so on". (ONG P96–98)

However, one of the boundary agents states that for him a fundamental problem with ISU is that the director is neither "strongly anchored in the University, nor in the city. She is to some extend free-floating" (JA P129). Although the boundary agents were supposed to anchor ISU more to its parent organizations, they do not consider themselves part of ISU as an organization. Instead, they have developed their group identity in opposition to the organizational identity of ISU and the position of its director:

"The group [of the boundary agents] is very strong, and [the director] is kind of lonely because she's the only one sitting, working in ISU. And we are connected to ISU but we are not working inside of ISU, and I think that also makes it a bit difficult". (IR P36)

The boundary agent's idea of ISU is very much constructed in opposition of the "lonely" and detached director and the organization she stands for. Their vision is that in the longer-term there would be more boundary agents in both organizations and their contact would be more informal and established within their respective organization (IZ P91; RZ). So, in the long-term there would be no need for something like a formal organization like ISU to organize the links between city administration and university. Instead the issues currently linked with and to some degree outsourced to ISU—collaboration across organizational boundaries in order to make Malmö more sustainable—would be more embedded within the values of the parent organization as such, meaning every employee would be considered as a kind of boundary agent and allowed to spend a substantial amount of their work time outside the boundaries of their home organization, to explore ways on how they can work together in order to push for a more sustainable Malmö (IZ P91, P97; QY P24). As one of the boundary agents from the university states:

"I see ISU and all the activities that ISU does as instruments to make people to get together and talk, to establish links and mutual interests and even friendships and that those connections in themselves should work in the future. So, one of the goals of ISU should be, I don't know if that would happen, but to make itself superfluous in a way, that it shouldn't have to be, because connections are already there.". (JA P85)

These remarks in a way show another side of the ISU dilemma between project, organization and network. On the one hand, from the expressed free-floating attitude the ISU process is perceived as horizontal, creative and leaderless. On the other hand, there are strong expectations that ISU should have a clear organizational profile, as well directly deliver concrete project results. This directly contradicts the ideal of ISU to be a place of networking, open conversations and informal communion, where common projects and benefits would emerge effortlessly and naturally, without formal planning and tight performance control. A positive ideal of ISU then is linked to a good, communal, almost familiar climate among the board members (horizontally linking with fellow heads of departments on their professional level), as well as among the boundary agents (horizontally linking with fellow colleagues on their professional level). The negative picture of ISU is that of an organization that is failing as its parts are not vertically integrated. For most interview partners, this image is embodied by the position of the director, who is supposed to lead, while not having the mandate to do so, and should thus lead without being a leader. For others this is embodied in the ISU Board, which does not properly function as a steering body and head of ISU as an organization but is rather a horizontal meeting and networking place for ISUs external partners (the departments of city administration and university).

Both of these perspectives might suggest that ISU in an ironical way is fulfilling its task of creating horizontal links between the mother organizations, because it is failing as an organization and in a way is not able to vertically integrate internally and create boundaries against the organizational outside. Yet, there is a sense of frustration about the lack of forward movement and a clear vision, which is linked to ISU being partly considered and presented as an issue organization and institute pushing forward sustainable urban development in Malmö and expected to work as an independent actor that can be developed by those professionally responsible. At the same time, however, those professionally responsible for ISU (i.e., especially the executive director, but also the boundary agents) do not feel they have the freedom or mandate to do so. They seem to be frustrated by feeling required to lead and develop, while also being expected to adhere and follow the board, which in turn assumes a leadership position without providing the coherent vision or concrete organizational leadership needed, but instead merely exchanges information and loosely defines areas of common interest that then should be developed by ISU (cf. DD P29 above).

5.4. Advantages and Disadvantages of the Fuzzy Profile between Project, Organization and Network

The advantage of such an undetermined structure is perceived by the interviewees to lie in its flexibility and responsiveness, which would allow ISU to adapt its activities to the current and changing needs of the departments, so as to seize on emerging opportunities for collaboration in a timely manner (ISU 2014a; IR P171). However, the disadvantage linked with this structure is described in terms of ISU being governed in an ad-hoc way without clearly formulated organizational boundaries and a committed long-term vision [77]. Instead, as the executive director of ISU states, the quick turn-around of departmental employees being partly financed via ISU and the yearly discussions about what the needs of the departments are leads to a constant, self-referential and inward-looking debate about what to do with ISU and to work on how to work together instead of actually moving forward as an organization (BM P237).

Linked with this perception of pointless debates and the need to constantly explain what ISU is, should or should not do, is a long-standing and widespread sentiment that ISU lacks a clear profile, that it is a "container" or "archipelago" of projects of the heads of departments in the ISU Board. These would use ISU to address their short-term issues and departmental interests, instead of building a

long-term vision and focus for the organizational profile of the institute (LI P42; LI P110). In 2011, an outside evaluator of ISU concluded in his report:

"There is also a strong belief that the ISU's profile has been too fuzzy. It has not been sufficiently clear what ISU stands for and what role it should play. Some say that this is linked to ISU having become too inward-looking. The institute has marketed itself badly. Another factor may be that the concept of an institute creates expectations for something bigger than what you can live up to my opinion". ([76]-own translation)

This statement highlights the disadvantages of ISU being run as an internal collaborative project, but simultaneously forming (or posing) as a somewhat independent institute, an independent issue organization with a focus on advancing knowledge on sustainable urban development. The disadvantage of this is, as the evaluator stated, too high or even false expectations about what ISU can or should do, resulting in a fuzzy profile, thematic inconsistencies and an organizational over-stretch [78]. An interviewee expresses this dilemma in a similar way within the context of ISU's work in respect to sustainable urban development:

"I mean, sustainable urban development, it's a huge area and I think it's important for ISU to have a red thread in what we are doing, because it's easy that you try to work with everything within this area and then nothing will be done of relevance. (...) You have to be very strict on what you're doing. So, you have to learn to say: 'No, ISU can't do this. We are focusing on these questions and will be doing so for another year', or something like that". (DD P79–81)

Thus, in developing its profile, a thematic focus on sustainable urban development would be necessary. This would imply strict rules about the type of projects accepted and carried out by ISU. This kind of approach would create a clearer profile as a thematic institute. Put the other way around, if ISU were not a temporary project but could operate as an institute, it would be easier to develop such a focus, to reject some demands and expectations by being able to say more clearly what ISU stands for in terms of sustainable urban development (LZLZ P197–202). However, such an approach is opposed to the idea of ISU working as an open, inclusive and flexible meeting place for potential collaborators and their project ideas and accordingly, when asked how exactly ISU could better manage expectations, one interviewee answers:

"I don't think you can manage the expectations, actually, because people see us as a kind of a meeting place—'Can't this be something for ISU?', etc.—not realizing that the personal resources are very short. And so, we have a discussion within the board almost every meeting saying: 'Are we doing the right things and how does this new task connect to the others? Is it something ISU should be doing or is it something we could leave to other partners?". (DD P85)

Managing external expectations (and saying 'no' to external audiences or stakeholders) is described by the interview partner as running counter to ISU's image as a meeting place for professional networking. Rather than confronting outside partners and rejecting their ideas, the Board internally argues whether this is something ISU should engage in, and if not, transfer (or divert) it to other partners in its network. In any case, here emphasis is placed in avoiding rejection so as to not threaten ISU's (beneficial) image as an open, inclusive meeting place. However, in the earlier statement by the chairwoman above, being strict and saying no was simultaneously seen as a necessary requirement of ISU being able to navigate the vast thematic field of sustainable urban development coherently and efficiently. The two statements exemplify the ambiguity between ISU as an institute or issue organization and ISU as an open meeting place for projects: The organization should be strict and also reclusive, yet, gain profile and direction, while the meeting place should not manage external expectation, yet rely on its network to transfer project ideas that do not fit to other partners, so as to not lose focus and concentrate on what ISU should actually stand for. The problem of inconsistently labeling a collaborative project as an institute and raising expectations can also be positively understood in the sense that you appear (in the eyes of external audiences especially) to have something bigger, more attractive, capable and qualitatively different (the institute) than what you actually paid for (a project). For example, ISU figures as an independent, neutral host or additional guest for workshops, network events and conferences organized by the university or city administration and is included as a separate entity in university or city administration projects or administrative publications (BM P164). When ISU disseminates information about urban development projects in Malmö, these seem to be coming from a more independent source than if it were issued by the city administration itself. Thus, it might not be an accident or necessarily a sign of weak leadership that ISU remains a hybrid between project and organization, as the blurred boundaries also produce benefits. To borrow a term from psychology, these could be called "secondary gains", i.e., short-term, but specific benefits that accrue from not overcoming a conflict, contradiction or problem. As the executive director sums up the enormous list of adjacent university or administrative projects ISU is engaged in:

"We are valuable because we are neutral. We are not the city and we are not the university, so therefore we are not dangerous or a competitor. (...) We are just a neutral platform and we give away so much for free. We are like a consultant firm they don't have to pay". (BM P164–166)

Being engaged in a multitude of tasks and projects links to the value ISU has by being a "container" for all kinds of projects that the university and the city are involved in. Its special value is also connected to the neutral "institute brand" it wears that can be easily applied to other projects and add the value of sustainability like it did to the city's "The Line" (This project is organized by the city to show Malmö as an innovative business environment and various initiatives, start-up and internet companies connected to sharing-based economic models located along a pedestrian and cycle path linking Malmö center and Western Harbor) with its business focus. The many projects and tasks ISU is involved in and the way its funds and resources are re-directed back to the departments at university and city or adjacent projects of the city and university departments, also serve to "neutralize" ISU as a danger or competitor. Since ISU cannot use funds to build up an independent organization, it remains readily available for utilization as a free consultancy, available brand and service provider to the various endeavors of the parent organizations. As an interviewee from the city administration (and ISU Board) states:

"You could say the main organization or mother organizations they want, they need this organization between, and they want it to be effective and have a high profile, but not too effective and not too high profile. Because suddenly it might be competing with the mother organizations, so it's a balance, always a balance for these types of cross board organizations. And they will always be questioned in the aims and goals and results. That is, I think, built in to the logic in itself". (ID P65)

By keeping ISU, a project, the partners ensure control to question, recalibrate and adjust aims and goals through temporary agreements and part-time employment contracts of departmental employees (rather than having an institute with its own employees). They thereby can still enjoy the benefits of a (seemingly) effective organization that can be used to enhance the profile of the mother organizations endeavors, activities, events. However, the interviewee is also aware that such an approach does inhibit the long-term development of the initiative and curbs the motivation of those working with it:

"When you are a group of people you have to work with your own identity, that's natural. But it very fast becomes a risk to become an irritation for the mother organizations. (...) This is a very important question, I think: How do you secure the commitment of the mother organizations and support from the mother organizations and at the same time, develop the inner life of this organization that it can keep the people that are working here and it gets a known profile and momentum?". (ID P77) What in the first quote by this interviewee was expressed in terms of balancing, might as well be understood in terms of a contradictory expectation: To fulfill the leadership's needs for control and flexible utilization of ISU as a project, while also developing a corporate identity, a long-term strategy, so as to generate the direction and momentum necessary for the development of an independent organizational life of ISU. However, instead of dissolving this contradiction it could also be managed by separating outside appearance from inside substance: having the outside appearance as an (independent, high profile) institute on sustainable urban development, while being internally organized (controlled and bound) as a project for facilitating commitment and collaboration between two parent organizations. Such a concept could be visualized as in Figure 1:



Figure 1. Differing Internal and External Understandings of ISU Structure (Source: own illustration based on interviews and [76]).

Many interviewees point out that the first model is the way ISU should work and the latter would show what should not happen, namely the emergence of an organization detached from its mother organizations and located in between both, blocking their direct interaction or competing with them (ID, ONG, OZM, LI). However, at the same time these two models represent the simultaneousness of ISU's internal project and networking dimension and the external institute dimension of ISU.

5.5. Employing the Fuzzy Profile by Focusing on Learning and Critical Thinking—The Urban Research Day and the Thesis Program/Urban Strategic Forum

Now we want to highlight the ISU's activities or interventions that seem fruitful for developing the different organizational identities of the initiative separately but in concert. First, an annual research conference, Urban Research Day. Here ISU acts as an independent host organization for an event where university academics, students and city administrators meet and discuss sustainable urban development with external experts and interested publics. The other intervention is a combination of a project-oriented cross-organizational workshop format (the so-called Urban Strategic Forum) and a networking format developed by the boundary agents (a thesis match-making program). The first kind of activity develops ISU's external identity as an independent institute, organization or brand, pushing

forward the critical debate on sustainable urban development, while also linking the cross-sector partnership to the greater transformation story of Malmö. The other two activities aim at developing ISU's internal dimension by producing tailor-made knowledge and collaborative projects through establishing a learning and training network spanning across university and administration.

First, an important aspect of the open, more public meeting format of the conference is that it enables ISU to get known among relevant audiences outside the confines of its mother organizations by connecting with Malmö's prestigious transformation. This "brand development" is relevant when it comes to the so-called Urban Research Day, which is organized partly by the university and partly by the city (with help by the ISU director), but with ISU figuring as the principal organizer and host. It is an annual, international conference, organized at the end of each year, the last time (before the field research phase) in December 2013, where the topic was "Culture and Sustainable Urban Development" [78].

Another important aspect of this conference was, that it opened up the issue of sustainable urban development for critical discussion (LL 19). LGQ explains that this Urban Research Day brought about some "really interesting conflicts with the planning department" (LGQ P91) in the city. According to the interview partners this resulted from ISU organizing this event in an open way, inviting critical voices to confront the city administrators and their economic growth-focused "planning culture". They thereby made visible that people involved in the collaborative process of planning a certain industrial area in Malmö had "totally different agendas and objectives" (LGQ P98). She concludes that the incident at the Urban Research Day indicates that one of ISU's important tasks is to be "very open-minded when it comes to allowing for different interpretations and inviting different kinds of people into the debate and into this space that ISU is" (LGQ P100). This would allow conflicts to become visible, which in turn is a necessary prerequisite: "(...) for us to develop a creative collaboration. Since we need to accept that we have different agendas, if not, sustainability will remain an empty signifier that we can fill with whatever." (LGQ P104).

The tense panel on the Urban Research Day was underscored by many critical questions about gentrification (especially from students, LL P19). It not only triggered subsequent inclusion of the cultural department of the city in the collaborative project of the planning department in that Malmö area to be redeveloped (P19), but also led many in the audience to consider this debate as highly relevant and important. Since ISU was organizing it, it sparked interest in its work and in sustainable urban development as a whole. LGQ concludes:

"Targeting these actors that are not normally included in this process, in that way, ISU has kind of provided a parallel, to a certain extent, to the administrative space provided by the municipality. Or, to the academic spaces provided by the university. So, a kind of middle ground. And inviting non-expected agents into this dialogue, it's become political". (LGQ P120)

She highlights the importance of ISU acting as an independent host. This conception is linked with the purpose, identified above, of ISU as an actual, issue-focused institute. Later on, she links this with the issue of whether ISU should be understood as working more informally on building up networks or provide a formal framework and concrete inputs for the creation of joint projects:

"It is extremely key I think, to have these kinds of open platforms, which are not instrumentally targeted towards funding, towards specific goals, or specific impact criteria, whatever. ISU should be much more open. And it's important because we don't know what will be the issues in two years' time. And we need these platforms where we can come together and not react to other people's preformulated questions and so on, but actually formulate our own questions.". (LGQ P138)

"Open" here also means public as opposed to the spaces behind closed doors as an important countermeasure against insider networks, instrumental logics and short-sighted perspectives. However,

for her coming together in an open way also means conflict and accordingly ISU should continue these kind of open public spaces "for encounters where consensus is not already the natural goal, but where there is also room for controversy and debate" (LGQ P46). By inviting unexpected voices and the public, conflicts bring out surprising insights and open up new perspectives on the overall ambition of Malmö's transformation and its story of being a pioneer in sustainable urban development. This in turn strengthens ISU's public profile as a credible, internationally recognized institute and capable force within the field. In the end, though, as LL argued, it was the tangible political scandal of having the growth-focused planning culture of the city's planning department 'exposed' on the ISU stage, which expanded the collaboration effort in the redevelopment area, to include from then on, also the cultural department of the city, as a more competent partner for those issues related to culture as a driving force for sustainable urban development. However, it also raised the interest of administrative and academic staff formerly disinterested in the work of ISU and contributed to ISU's organizational development, because only afterwards the cultural departments of the city and university were systematically included in the ISU board and boundary agent program.

Second, complementing the public conference format raising ISU's profile as an institute are those more internal and cross-organizational intervention formats organized by the boundary agents, i.e., the partner thesis matchmaking and the Urban Strategic Forum. They are organized together with the city departments around key sustainable urban development issues for the city departments. The boundary agents assembled a list of 50 different topics from which students could choose their bachelor or master thesis topic, e.g., how to make Malmö's neighborhoods denser and more mixed-use (BM P61; IZ P29, P31). Subsequently, the boundary agents facilitated contacts and supported the process of the thesis program, meaning they linked students, their supervisors at the university and the city administrators whose questions or issues they explore. In the first year this process led to 20 theses produced by student teams consisting of 50 students ([68]; BM P61). By "matching" the institutions, tutors and students with administrators who have issues in need of research one idea is to produce direct results for the city's planning projects (IZ P37). However, an important objective of this program is to more substantially bring in the academic supervisors of the students into the network between students and administrators. Here the idea is to use the students as a low-threshold and personal way to connect university and city professionals and inspire larger collaborative research projects that could possibly lead to joint applications for research funds and better educational formats for the graduates of Malmö University ((DD P74; IZ P21; LI P18; BM P71). In a way, the student work is seen as a subtle, indirect way to better engage academics with the practical problems of the city administration via education and training (rather than trying to instigate senior research projects directly).

The thesis program was then linked to an Urban Strategic Forum, which is a bi-annual, formal meeting format between university researchers and city administrators introduced in 2012 with the aim to facilitate the identification of common research/planning issues so as contribute to the joint development of projects and funding proposals [79]. The particular Forum in which a selection of student theses was presented was generally lauded (LGQ P76), even called "the best Forum we had" (ONG P66, cf. IR P88).

Seven working groups formed on topics such as "fairer access to the city" or "the dense, green and healthy city" [80]. There is a general hope that these working groups will grow gradually and either directly produce collaborative projects or at least more binding and continuous relationships among researchers and city administrators. Several of those involved describe the key for process as linking the practice and research through setting up encounters between the institutions on a "more grassroots level, between students and civil servants" (LGQ P73), to "use the students as a way to connect" (LI P104), to create personal connections, "to simply meet face-to-face" (BO P125) so as to show that "behind every subject there is a person on either side" (IZ P29).

However, there is a feeling that these working groups would need to be continuously facilitated and supported by ISU so as to ensure that they remain "alive" (BO P108; cf. IR P81). With ISU's limited resources, this is seen as a big challenge. A more realistic assumption might be that ISU will not be able to "follow up on the working group, it has to live by itself" (BM P79). In this respect several interviewees also mention that even though the working groups might not result in formal projects and applications, they would still consider them a success, in that they contributed to establish informal, personal connections between researchers and administrators. Furthermore, they got to know that there is someone on the other side who is also working on the same issues (BO P108). Here one could argue, that ISU's focus on creating informal connections is due to its limited organizational capacity to more substantially control and lead the process that is supposed to result in concrete collaboration results and projects. It, however, also links to ISU's problem to clearly show its organizational impact as the relationships they instigate must live (or die) by themselves, and if they indeed at some later point in time would yield any concrete project results, these would not easily be traced back through the network of collaborators to ISU's activities.

The expectation that ISU should facilitate project organization in a more hands-on way (cf. RI P92–94) links back to the problem of ISU being a project, which is lacking in own, freely usable resources. It also specifies the issue of creating a free space for discussion, where people more fundamentally reflect and ask themselves what they should do together, rather than immediately jumping into the work. As LI argues:

"If the intention with ISU is to develop new project and then to be some kind of breeding ground/place that breeds new projects, then ISU should have a kind of venture fund or time fund (...). Because there's a problem in finding time for sitting down and developing ideas. And (...) it's also some kind of risk, because maybe you're sitting together and you're doing a couple of workshops and then everything ends up in nothing, because lots of ideas won't come together, because they're too complicated and when (...) analyze it you discover: 'it will not work'". (LI 33)

The Urban Strategic Forum and its link to the student thesis program is perhaps a good practice in this respect, as it was generally lauded by those participating. More specifically though, it seems to successfully link more formal, project-focused and also time-efficient aspects of collaboration between professionals and their organizations (the Urban Strategic Forum format), to the longer-term, informal and relationship-focused aspects of the student thesis program. The aspect of informality of the students can here be considered to provide a more personal bridge between the organizations, allowing their professionals to connect for the longer period of time it takes to produce a thesis and enabling both to assume a similar, more personal mentor role (instead of retreating to their different organizational or professional roles). At the same time, the 20 student theses produced within an applied science context facilitated by ISU allows those organizing it to point to outcomes that would not have emerged without its activities. This helps ISU to demonstrate that it can deliver on the expectations that the resources put into ISU should yield a timely return on investment. However, at the same time, it seems important to acknowledge that ISU in its limited scope and with its limited resources can only lay the "breeding-ground" or produce a germ cell for more substantial and higher-quality projects to develop and "live or die" by themselves, i.e., to self-organize gradually and involve administrators and academics in a self-determined, horizontally networked and project-focused way. In the same way, it seems important to acknowledge the open-ended, autonomous character of these network projects or project networks: this means to allow them to fail and grant those engaging in them leeway in terms of time, labor and resources they can spend free from (conventional) organizational performance controlling.

6. Discussion and Conclusions

In accordance with D'Andrea and Gosling's suggestion for a whole institutions approach [23], we explored a central dilemma between identifying as project, organization or network within the change process towards more sustainable HEIs. We did so by researching how individuals engaged in an illustrative cross-sector partnership narratively make sense of the organizational dilemmas and

ambiguities that stem from the complexity of working together across sectors in pursuit of an integrative approach to knowledge production. The meaning of our findings for the whole institution approach in HEIs is manifold. Instead of organically complementing each other in a kind of teleological movement from project via organization to self-organizing and inclusive network, we found in our analysis, that often the different organizational identities as project, organization or network competed with each other and canceled each other out. In the end, the organizers did not end up with a teleological, virtuous development from project, via organization to self-organizing network across sectors, but with an undetermined and aimless movement back-and forth only resulting in a general sense of stagnation and frustration among especially its regular members. This might also be interpreted against the background of the vagueness of the concepts of sustainable development and Education for Sustainable Development (Authors 2011). However, our findings also suggest that the fuzziness of the cross-sector partnership and its indetermination is seen as something that should not necessarily be overcome, but that the state of indeterminacy in various ways contributes to the different goals of project development, organizational development and network development, which-by their nature—involve conflicts, trade-offs and unsolvable organizational dilemmas. Thus, a fuzzy profile, while being often criticized as a hindrance to the further development of the cross sector partnership and as something that is diminishing its potential impact, can also be seen as a (relative) success story as it supplies many "secondary gains" to those involved by not overcoming their differences and organizational dilemmas. These findings extend the dimensions, dilemmas and tensions identified by D'Andrea and Gosling's conception for a whole institutions approach [23].

In this context, three activities of ISU were analyzed more closely, because they suggest a portfolio, which would allow to develop the three different identities of ISU as project, organization and network separately but in concert. First, the annual public conference "Urban Research Day" hosted by ISU, which boosted its public profile, successfully linked the institute to the prestigious Malmö transformation story and also delivered on ISU's organizational expectations to be a viable, publicly relevant and critical institute. Second, the thesis match-making program organized by the ISU boundary agents, which helped to deliver on the project expectation to produce specific outcomes (theses), that have at least the potential to directly benefit the work of the administration, as well as it directly contributes to the universities educational mission. At the same time, it realized the cross-sector partnership's ambition to nurture personal and professional network relationships, because it focusses the professionals on a learning-process via the students. Third, the "Urban Strategic Forum" which brings together academics, students and administrators in a workshop format to further develop their learning network established through the thesis match-making program by providing the breeding ground for project groups to self-organize across organizations.

However, when using this specific research focus and the accompanying set methods there are important limitations.

First, because this study relies on mainly narrative accounts in interviews, focus groups and program documents, it cannot make claims concerning the objective performance of cross-sector partnership. Our results are limited to what the interviewees give for accounts and judgements and how these accounts and judgements allow them to construct their cross-sector partnership as meaningful. Second, because this study relies on investigating only one case of meaning making in and through cross-sector partnerships, the generalizability of the interpretations and claims deduced from the analysis are very limited.

Accordingly, the results presented should not be read as providing general knowledge about the phenomenon of cross-sector partnerships as such, but as providing, theoretically informed patterns identified through conducting interviews and analyzing documents in respect to a unique cross-sector partnership. However, the results presented are grounded in rigorous analysis and systematic and transparent reference to evidence in form of interview and document extracts. In this way they produce context-sensitive knowledge about the specific process of meaning making in a cross-sector partnership,

which provides tentative theoretical concepts that can be used to inform further research and theorizing about other cases of cross-sector partnerships.

However, it seems imperative to not measure the cross-sector partnership as a rational process of efficiently pooling resources and effectively integrating project, organizational and network logics as parts of a smoothly growing, naturally synergetic and expanding organism, yielding evermore benefits for everyone involved. Instead, project, organizational and network logics in a cross-sector partnership need to be acknowledged as constituting substantially different, narrative ways of making sense of the same initiative. They each emphasize different objects of value (tailor-made project results, organizational autonomy and viability, network development), different modes of exchange (professional workshop, public conference, learning program), different roles (professional experts, organizational employer/employee, mentor/student) and organizational relationships (project teams, formal/hierarchical organization, personal/horizontal networks).

This is why we suggest that research in sustainable development in HEIs should more substantially investigate the different professional and organizational identities of the cross-sector partnerships undertaken in the context of a whole institutions approach and more specifically explore the process of narrative boundary-making in between them as that which separates and links the actors in their continuous struggle to make sense of respective professional and organizational identities. We are confident that such research could contribute substantially to better define and implement whole institution approaches in the context of promoting sustainable development in and through HEIs.

However, its explorative investigation into one cross-sector partnership for the promotion of sustainable development in and through a HEI showed that more work is required to understand boundary-making as a narrative process of making sense of whole, i.e., complex institutions such as HEIs. We demonstrated that it is especially worthwhile to focus the whole institution approach more systematically on navigating organizational dilemmas and tensions (as D'Andrea and Gosling suggest [23]) by taking into account how these dilemmas and tensions allow individuals to establish their agency through continuously conversing about dealing with them, and where to draw the line through them, so as to define 'what the story is' and 'what to do about it' [81]. In this way this study suggests to integrate insights from narrative and psychological research in organizational studies into the practice of promoting sustainable development in HEIs through a whole institutions approach. A worthwhile next step in better employing a whole institutions approach in sustainable development of HEIs would be, for example, to integrate Michael White's approach in his work on narrative therapy, in which he worked out a detailed analytical framework and investigative method to explore how people develop ceremonies and cognitive maps to orient themselves and connect to others [82]. Another example is Gavazzi and Fox [83], who use their background in family and marriage therapy to approach partnerships between university and the regional community surrounding it. A further important avenue for future research on the issue of sustainable development in HEIs in general and cross-sector partnerships in context of a whole institutions approach in particular, is the integration of theoretical concepts and methods of cultural psychology and specifically the sub-field of narrative liminality, i.e., the study of boundary-making, border and liminality construction in and through transdisciplinary cross-sector partnerships between universities and non-academic partners. For example, Picione and Valsiner [84] highlight the psychological significance of semiotic borders, demarking separation, differentiation, distinction-making, connection, articulation and relation-enabling. They argue that the border is a narrative tool, which enables actors in organizations to maintain stability and induce transformation at the same time by way of creating an ambiguous and instable "liminal space", and in this way induces creativity, leads to novelty and the creation of new narrative forms of organizing (Ibid.). In a related work Picione and Freda approach the border as a semiotic concept that allows individuals to positions themselves to others and the world by creating a dynamic boundary in terms of necessity, obligation, willingness, possibility, permission, and ability [85]. This study has laid a first path, but future studies on sustainable development in HEIs and the whole institutions approach involving cross-sector partnerships would certainly benefit a lot from more systematically integrating

concepts and approaches from organizational (or group) psychology. More systematically focusing on the narrative construction of professional and organizational identities in a complex, interconnected world, would also allow for cross-sector partnerships to become recognized as critical case for studying how individuals organize their professional selves and organizational relationships in late-modern, increasingly digitalized and globalized knowledge societies in which a radically more sustainable form of development has become essential for survival of human civilization on this planet.

Author Contributions: Conceptualization, S.M. and N.K.; methodology, S.M.; software, X.X.; validation, S.M., Y.Y. and Z.Z.; formal analysis, S.M.; investigation, S.M.; resources, S.M.; data curation, S.M.; writing—original draft preparation, N.K. and S.M.; writing—review and editing, N.K. and S.M.; visualization, S.M.; supervision, N.K.; project administration, S.M.. and N.K.; funding acquisition, S.M..", please turn to the CRediT taxonomy for the term explanation. Authorship must be limited to those who have contributed substantially to the work reported.

Funding: This research was funded by the Freie Universität Berlin.

Acknowledgments: We would like to thank the editors of the Special Issue and the anonymous reviewers for their constructive suggestions and detailed notes and Susanne Lea Helm for her support in revising the article according to the journal's guidelines.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

References

- Perry, B.; May, T. Urban knowledge exchange: Devilish dichotomies and active intermediation. Int. J. Knowl.-Based Dev. 2010, 1, 6–24. [CrossRef]
- Wyatt, S.; Siune, K.; Marcus, E.; Calloni, M.; Felt, U.; Gorski, A.; Grunwald, A.; Rip, A.; Semir, V.D. Challenging Futures of Science in Society, Emerging Trends and Cutting-Edge Issues; EU: Brussels, Belgium, 2009.
- Feldman, M.P.; Florida, R. The Geographic Sources of Innovation: Technological Infrastructure and Product Innovation in the United States. *Ann. Am. Assoc. Geogr.* 1994, 84, 210–229. [CrossRef]
- Etzkowitz, H.; Leydesdorff, L. The Dynamics of Innovation: From National Systems and "Mode 2" to a Triple Helix of University–Industry–Government Relations. *Res. Policy* 2000, 29, 109–123. [CrossRef]
- Carayannis, E.G.; Campbell, D.F.J. Mode 3 Knowledge Production in Quadruple Helix Innovation Systems; Springer: New York, NY, USA, 2012.
- 6. Torugsa, N.; O'Donohue, W. Progress in innovation and knowledge management research: From incremental to transformative innovation. J. Bus. Res. 2016, 69, 1610–1614. [CrossRef]
- Kumar, T.M.V.; Dahiya, B. Smart Economy in Smart Cities. In Smart Economy in Smart Cities: International Collaborative Research: Ottawa, St. Louis, Stuttgart, Bologna, Cape Town, Nairobi, Dakar, Lagos, New Delhi, Varanasi, Vijayawada, Kozhikode, Hong Kong; Kumar, T.M.V., Ed.; Springer: Singapore, 2017; pp. 3–76.
- Hardy, C.; Phillips, N.; Lawrence, T.B. Resources, Knowledge and Influence: The Organizational Effects of Interorganizational Collaboration. J. Manag. Stud. 2003, 40, 321–347. [CrossRef]
- George, G.; Zahra, S.A.; Wood, D.R. The effects of business—University alliances on innovative output and financial performance: A study of publicly traded biotechnology companies. J. Bus. Ventur. 2002, 17, 577–609. [CrossRef]
- Perkmann, M.; Tartari, V.; McKelvey, M.; Autio, E.; Broström, A.; D'Este, P.; Fini, R.; Geuna, A.; Grimaldi, R.; Hughes, A.; et al. Academic engagement and commercialisation: A review of the literature on university—Industry relations. *Res. Policy* 2013, *42*, 423–442. [CrossRef]
- 11. Simeone, L. Design Moves. Translational Processes and Academic Entrepreneurship in Design Labs; Malmö Universitet: Malmø, Sweden, 2016.
- 12. Walsh, L.; Kahn, P. Collaborative Working in Higher Education. The Social Academy; Routledge: New York, NY, USA, 2010.
- Benneworth, P.; Conway, C.; Charles, D.; Humphrey, L.; Younger, P. Characterising Modes of University Engagement with Wider Society. A Literature Review and Survey of Best Practice; Final Report, 10th June 2009; Newcastle University: Newcastle upon Tyne, UK, 2009.
- 14. Bikard, M.; Murray, F.; Gans, J.S. Exploring trade-offs in the organization of scientific work: Collaboration and scientific reward. *Manag. Sci.* 2015, *61*, 1473–1495. [CrossRef]

- 15. Polanyi, M. The republic of science: Its political and economic theory. Minerva 2000, 38, 1–21. [CrossRef]
- Olsen, J.P.; Maassen, P. European debates on the knowledge institution: The modernization of the university at the European level. In *University Dynamics and European Integration*; Maassen, P., Olsen, J.P., Eds.; Springer: Dordrecht, The Netherlands, 2007; pp. 3–22.
- 17. Reale, E.; Primeri, E. *The Transformation of University Institutional and Organizational Boundaries;* Sense Publishers: Rotterdam, The Netherlands; Boston, MA, USA, 2015.
- Schneidewind, U. Nachhaltige Wissenschaft. Plädoyer für Einen Klimawandel im Deutschen Wissenschafts- und Hochschulsystem [Sustainable Science. Plea for Climate Change in the German Science and Higher Education System]; Metropolis: Marburg, Germany, 2009.
- 19. Doberneck, D.M.; Glass, C.R.; Schweitzer, J. From rhetoric to reality: A typology of publically engaged scholarship. *J. High. Educ. Outreach Engagem.* **2010**, *14*, 5–35.
- 20. Weerts, D.J. The university and urban revival: Out of the Ivory Tower and into the streets. *Rev. High. Educ.* **2008**, *31*, 523–524. [CrossRef]
- McKelvey, M.; Holmén, M. Learning to Compete in European Universities: From Social Institution to Knowledge Business; Edward Elgar Publishing: Cheltenham, UK, 2010.
- 22. McGregor, S.L.T.; Volckmann, R. *Transversity. Transdisciplinary Approaches in Higher Education*; Integral Publishers: Tucson, AZ, USA, 2011.
- 23. D'Andrea, V.; Gosling, D. *Improving Teaching and Learning. A Whole Institution Approach*; Society for Research into Higher Education & Open University Press: Maidenhead, UK; New York, NY, USA, 2010.
- 24. Ibáñez-Carrasco, F.; Riaño-Alcalá, P. Organizing community-based research knowledge between universities and communities: Lessons learned. *Community Dev. J.* 2011, 46, 72–88. [CrossRef]
- 25. Reason, P.; Bradbury, H. Handbook of Action Research. Participative Inquiry and Practice; SAGE Publications: London, UK, 2001.
- 26. Stringer, E.T. Action Research, 4th ed.; SAGE Publications: Thousand Oaks, CA, USA, 2013.
- 27. Zuber-Skerritt, O. Action Leadership. Towards a Participatory Paradigm; Springer: Dordrecht, The Netherlands, 2011.
- Greenwood, D.J.; Levin, M. Introduction to Action Research. Social Research for Social Change, 2nd ed.; SAGE Publications: Thousand Oaks, CA, USA, 2007.
- 29. Hajer, M.A.; Wagenaar, H. Deliberative Policy Analysis. Understanding Governance in the Network Society; Cambridge University Press: Cambridge, UK, 2003.
- Fischer, F. Reframing Public Policy. Discursive Politics and Deliberative Practices; Oxford University Press: Oxford, UK, 2003.
- Aboelela, S.W.; Larson, E.; Bakken, S.; Carrasquillo, O.; Formicola, A.; Glied, S.A.; Haas, J.; Gebbie, K.M. Defining interdisciplinary research: Conclusions from a critical review of the literature. *Health Serv. Res.* 2007, 42, 329–346. [CrossRef]
- 32. Dubb, S.; Howard, T. *Linking Colleges to Communities: Engaging the University for Community Development*; The Democracy Collaborative at the University of Maryland: College Park, MD, USA, 2007.
- 33. Calhoun, C. The university and the public good. Thesis Elev. 2006, 84, 7-43. [CrossRef]
- 34. Marginson, S. Putting "public"back into the public university. Thesis Elev. 2006, 84, 44–59. [CrossRef]
- 35. Jörke, D. Bürgerbeiteiligung in der Postdemokratie. APuZ 2011, 1–2, 13–18.
- 36. Boltanski, L.; Chiapello, E. The New Spirit of Capitalism. Int. J. Politics Cult. Soc. 2005, 18, 161–188. [CrossRef]
- 37. Swyngedouw, E. Designing the Post-Political City and the Insurgent Polis; Bedford Press: London, UK, 2011.
- Normand, R. The Changing Epistemic Governance of European Education. The Fabrication of the Homo Academicus Europeanus? Springer International Publishing: Cham, Switzerland, 2016.
- Swyngedouw, E. Governance Innovation and the Citizen: The Janus Face of Governance-beyond-the-State. Urban Stud. 2005, 42, 1991–2006. [CrossRef]
- Ankrah, S.; Al-Tabbaa, O. Universities—Industry collaboration: A systematic review. Scand. J. Manag. 2015, 31, 387–408. [CrossRef]
- Goldkuhl, G.; Cronholm, S. Adding Theoretical Grounding to Grounded Theory: Toward Multi-Grounded Theory. Int. J. Qual. Methods 2010, 9, 187–205. [CrossRef]
- 42. Glaser, B.; Strauss, A.L. *Discovery of Grounded Theory. Strategies for Qualitative Research*, 1st ed.; Taylor and Francis: Somerset, UK, 1967.

- 43. Merton, R.K. On Sociological Theories of the Middle Range 1949. In *Social Theoryand Social Structure*; The Free Press: New York, NY, USA, 1949.
- 44. Geels, F.W. Feelings of Discontent and the Promise of Middle Range Theory for STS. *Sci. Technol. Hum. Values* **2007**, *32*, 627–651. [CrossRef]
- 45. Hedström, P.; Ylikoski, P.K. Analytical Sociology and Rational Choice Theory. In *Analytical Sociology: Actions and Networks*; Manzo, G., Ed.; Wiley: Hoboken, NJ, USA, 2014; pp. 57–70.
- Corbin, J.M.; Strauss, A.L. Basics of Qualitative Research. Techniques and Procedures for Developing Grounded Theory, 4th ed.; SAGE: Los Angeles, CA, USA; London, UK; New Delhi, India; Singapore; Washington, DC, USA; Boston, MA, USA, 2015.
- 47. Fram, S.M. The Constant Comparative Analysis Method Outside of Grounded Theory. *Qual. Rep.* **2013**, *18*, 1–25.
- 48. Charmaz, K. Constructing Grounded Theory, 2nd ed.; SAGE: London, UK, 2014.
- 49. Weick, K.E. Sensemaking in Organizations; SAGE Publications: Thousand Oaks, CA, USA, 1995.
- 50. Taylor, J.R.; van Every, E.J. *The Emergent Organization: Communication as Its Site and Surface*; Routledge: Abingdon, UK, 1999.
- 51. Strauss, A.L. A social world perspective. Stud. Symb. Interact. 1978, 1, 119–128.
- 52. Volbers, J. Theorie und Praxis im Pragmatismus und in der Praxistheorie. In *Praxis Denken*; Springer: Berlin/Heidelberg, Germany, 2015; pp. 193–214.
- 53. Latour, B. The Powers of Association. Sociol. Rev. 1984, 32, 264-280. [CrossRef]
- 54. Bruner, J. The Narrative Construction of Reality. Crit. Ing. 1991, 18, 1–21. [CrossRef]
- 55. Kamberelis, G.; Dimitriadis, G. Focus Groups. From Structured Interviews to Collective Conversations; Routledge: Abingdon, UK, 2013.
- Barbour, R.S. Analyising Focus Groups. In *The SAGE Handbook of Qualitative Data Analysis;* Flick, U., Ed.; SAGE Publications: London, UK, 2013; pp. 313–326.
- Vogl, S. Gruppendiskussion [Group discussion]. In Handbuch Methoden der Empirischen Sozialforschung [Manual Methods of Empirical Social Research]; Baur, N., Blasius, J., Eds.; Springer VS: Wiesbaden, Germany, 2014; pp. 581–596.
- 58. Flick, U. An Introduction to Qualitative Research; SAGE: Los Angeles, CA, USA, 2009.
- Kidd, P.S.; Parshall, M.B. Getting the focus and the group: Enhancing analytical rigor in focus group research. *Qual. Health Res.* 2000, 10, 293–308. [CrossRef]
- 60. Morgan, D.L. Focus Groups as Qualitative Research; SAGE Publications: Thousand Oaks, CA, USA, 1988.
- 61. Morgan, D.L. Focus Groups. Annu. Rev. Sociol. 1996, 22, 129–152. [CrossRef]
- 62. Creswell, J.W. *Research Design. Qualitative, Quantitative, and Mixed Methods Approaches,* 4th ed.; SAGE: Los Angeles, CA, USA, 2014.
- 63. SEKUM. Fran Sekum Till ISU [From Sekum to ISU]; [internal programm document]; SEKUM: Malmö, Sweden, 2004.
- 64. ISU. Verksamhetsplan 2014; [internal programm document]; ISU: Malmö, Sweden, 2014.
- 65. Ryan, G.W.; Bernard, H.R. Techniques to identify themes. Field Methods 2003, 15, 85–109. [CrossRef]
- 66. Saldaña, J. The Coding Manual for Qualitative Researchers; SAGE: Thousand Oaks, CA, USA, 2015.
- Palmberger, M.; Gingrich, A. Qualitative Comparative Practices: Dimensions, Cases and Strategies. In *The* SAGE Handbook of Qualitative Data Analysis; Flick, U., Ed.; SAGE Publications: London, UK, 2013; pp. 94–108.
- 68. ISU. Årsrapport 2013 [Annual Report 2013]; ISU: Malmö, Sweden, 2013.
- 69. Holgersen, S. *The Rise (and Fall?) of Post-Industrial Malmö. Investigations of City-Crisis Dialectics;* Department of Human Geography, Lund University: Lund, Sweden, 2014.
- SEKUM. Minnesanteckningar fran SEKUMs visionsdag 19-11-2003; [Internal Minutes from SEUMs vision day]; SEUM: Malmö, Sweden, 2003.
- 71. Anderson, T. Malmo: A city in transition. Cities 2014, 39, 10-20. [CrossRef]
- Öquist, G.; Benner, M. Why are some nations more successful than others in research impact? A comparison between Denmark and Sweden. In *Incentives and Performance: Governance of Research Organizations*; Welpe, I.M., Wollersheim, J., Ringelhan, S., Osterloh, M., Eds.; Springer International Publishing: Cham, Switzerland, 2015; pp. 241–257.
- 73. SEKUM. Minnesanteckningar fran SEKUM 16.03.2003; [Internal Minutes]; SEKUM: Malmö, Sweden, 2003.

- 74. SEKUM. Verksamhetsplan och Budgetunderlag 2005 [Business Plan and Budget Basis 2005]; [internal programm document; SEKUM: Malmö, Sweden, 2005.
- SEKUM. Minnesantekningar fran SEKUM [Minutes from SEKUM]; [internal minutes]; SEKUM: Malmö, Sweden, 2005.
- Persson, B. ISU—Institutet för Hållbar Stadsutveckling—Utvärdering av Verksamhetens tre Första år. [ISU—Institute for Sustainable Urban Development—Evaluation of the Business's First Three Years]; Larki Lund: Lund, Sweden, 2010.
- Persson, C. Institutet för Hållbar Stadsutveckling—Förslag Till Framtida Verksamhet [Institute for Sustainable Urban Development—Proposal for Future Activities]; ISU—Institutet för Hållbar Stadsutveckling: Malmö, Sweden, 2011.
- 78. ISU. Årsrapport 2013; [internal programm document]; ISU: Malmö, Sweden, 2013.
- 79. ISU. Strategiskt Basdokument 2012 [Strategic Basic Document 2012]; ISU: Malmö, Sweden, 2012.
- 80. ISU. Verksamhetsplan 2014 [Business Plan 2014]; ISU: Malmö, Sweden, 2014.
- Weick, K.E.; Sutcliffe, K.M.; Obstfeld, D. Organizing and the process of sensemaking. Organ. Sci. 2005, 16, 409–421. [CrossRef]
- White, M. Landkarten der Narrativen Therapie [Narrative Therapy Maps]; Carl-Auer-Verlag: Heidelberg, Germany, 2010.
- Gavazzi, S.M.; Fox, M. Understanding campus and community relationships through marriage and family metaphors: A town-gown typology. *Innov. High. Educ.* 2014, 39, 361–374. [CrossRef]
- Picione, R.D.L.; Valsiner, J. Psychological functions of semiotic borders in sense-making: Liminality of narrative processes. *Eur. J. Psychol.* 2017, 13, 532–547. [CrossRef]
- Picione, R.D.L.; Freda, M.F. Borders and modal articulations. Semiotic constructs of sensemaking processes enabling a fecund dialogue between cultural psychology and clinical psychology. *Integr. Psychol. Behav. Sci.* 2016, 50, 29–43. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





MDPI

Assessment of Sustainability Governance in Higher Education Institutions—A Systemic Tool Using a Governance Equalizer

Sebastian Niedlich ^{1,*}, Mara Bauer ², Margarita Doneliene ¹, Larissa Jaeger ², Marco Rieckmann ² and Inka Bormann ¹

- ¹ Department of Education and Psychology, Freie Universität Berlin, Habelschwerdter Allee 45, 14195 Berlin, Germany; margarita.doneliene@fu-berlin.de (M.D.); inka.bormann@fu-berlin.de (I.B.)
- ² Department of Education, Faculty of Education and Social Sciences, University of Vechta, Driverstraße 22, 49377 Vechta, Germany; Mara.Bauer@uni-vechta.de (M.B.); Larissa.Jaeger@uni-vechta.de (L.J.); Marco.Rieckmann@uni-vechta.de (M.R.)
- * Correspondence: sebastian.niedlich@fu-berlin.de

Received: 4 February 2020; Accepted: 25 February 2020; Published: 28 February 2020

Abstract: The paper aims to add to the discussion on sustainability governance in higher education institutions by examining the role of sustainability assessment and introducing an assessment tool inspired by systemic thinking and centered on a 'governance equalizer'. It discusses recent research and argues that the complexity inherent in sustainability governance remains to be addressed adequately. While a number of models and frameworks have been proposed, most of them remain caught between narrow, management-oriented approaches on the one hand, and rather abstract approaches that provide little guidance for improving the field on the other. Sustainability assessment tools represent a potential way to bridge this gap. While there are existing tools which include issues of sustainability governance, these are often limited to aspects that are easily quantifiable and neglect more complex aspects. Against this background, the article proposes an alternative tool to assess sustainability governance in higher education institutions. The tool is based on a multi-case study in Germany and has been tested in a series of workshops. Drawing on the concept of a 'governance equalizer', it focuses on the functional requirements of sustainability governance in five dimensions—politics, profession, organization, knowledge, and the public—and how they are addressed by the HEI. The tool raises the level of abstraction in order to capture complexity, but at the same time keeps sight of governance structures, processes, instruments, and practices. It combines clearly defined criteria that are assessed using carefully developed maturity scales with a focus on stakeholder participation and knowledge.

Keywords: sustainability assessment tool; sustainability governance; higher education institutions; systems theory; governance equalizer; politics; profession; organization; knowledge; public

1. Introduction

'Sustainable development' (SD) is a widely used term with varying meanings [1]. This paper uses the term to describe efforts to integrate environmental and socio-economic issues in order to meet human needs now and in the future. It involves societal reform and transformation, and a reconceptualization of the relationship of the human race with nature. Higher education institutions (HEIs) are frequently counted among the key players in creating a path for society towards sustainable development. It has been pointed out that this requires a fundamental change in the culture, purpose, policy, and practice of HEIs [2] and a transversal, inter- and transdisciplinary approach, integrating the core HEI domains of teaching, research, campus operations, and outreach [3–5].

A key question which arises is therefore how HEIs can coordinate and steer their actions toward sustainable transformation. This question of sustainability 'governance' has been addressed from different angles by a number of studies. While some try to integrate sustainability governance into existing management approaches, others emphasize its complexity and provide more abstract frameworks. Issues of governance also feature in sustainability assessment tools. Such tools appear promising in that they potentially provide a bridge between abstract models and concrete strategies. However, the existing tools have been criticized for addressing only those aspects of governance that lend themselves to quantitative measurement, while other important aspects have been neglected.

Against this background, the article introduces a sustainability governance assessment tool that combines ideas from systemic thinking with the concept of a 'governance equalizer': It focuses on functional requirements of sustainability governance in five dimensions—politics, profession, organization, knowledge, and the public—and how they are addressed by HEIs. The tool thereby raises the level of abstraction in order to capture the complexity of the subject, but at the same time does not lose sight of governance structures, processes, instruments, and practices.

The article starts with a short overview of recent research on sustainability governance in HEIs, followed by a discussion of sustainability assessment tools for use in HEIs and of the role of governance issues in such tools (Sections 2 and 3). Section 4 explains how our own assessment tool was developed in the 'HOCH^{N'} project based on case studies in eleven German HEIs and validated in a series of workshops. Section 5 presents the tool in detail and provides some remarks on how to apply it in practice. The paper concludes by discussing three perceived merits of the tool.

2. Sustainability Governance in Higher Education Institutions

The issue of sustainable development in HEIs continues to gain attention in both practice and research. Although recent studies show progress towards sustainable transformation of HEIs, including signs of holistic and systemic approaches [3,6], they also report that efforts are often compartmentalized, and practical integration of sustainable development goals remains difficult [7,8].

These findings raise the issue of sustainability governance in HEIs. Governance can be defined as "the process of steering society and the economy through collective action and in accordance with common goals" [9] (p. 4). Drawing on this definition, we use the term 'governance' to denote the process of steering HEIs (and related external actors) through collective action toward the common goal of sustainable development. Questions relating to this process have been investigated in a number of recent studies. Such studies can be categorized into three groups: Case studies, analyses of drivers and barriers, and frameworks and models [3].

Initially, research frequently took on the form of institutional case studies, often of a descriptive nature and with a focus on storytelling [10]. While some case studies discuss different approaches and give recommendations on supporting sustainable development in HEIs (e.g. [11,12]), theoretical advancements from this line of research are few, and its potential for improving the field remains largely unexploited [4,13] (see also [3]).

Other research focuses on drivers of and barriers to change in HEIs. Based on a literature review, Verhulst and Lambrechts [14] identify three clusters of barriers: (1) A lack of awareness, (2) the structure of higher education institutions, and (3) a lack of resources. They also point out recurring drivers in the literature on organizational change management, such as organizational culture, empowerment and involvement, and internal communication. Other scholars highlight the role of campus SD champions [15] or "a dedicated team of 'sustainability entrepreneurs', continuous interaction among all HEI actors, the creation of a sustainability office of some kind, a predictable budget, alignment with the topics of the imparted careers, formal networks and open and recurrent communication on sustainability on campus and beyond" [5] (p. 4283). Overall, however, drivers and barriers are often described in a "laundry list style", without allowing a profound understanding of how they interact and change over time [14] (p. 191).

Frameworks and models of change, by comparison, tend to provide a more comprehensive perspective, although approaches and perspectives vary. A number of publications attempt to link the issue of sustainable development to management strategies and practices in HEIs [16–19]. Some of these draw on the Plan-Do-Check-Act (PDCA) cycle and define key steps in the sustainability process. For example, Velazquez et al. [16] emphasize the importance of developing a sustainability vision and mission, setting up a sustainability committee in order to create policies, targets, and objectives, and develop sustainability strategies.

Other scholars identify phases or stages of sustainable development in HEIs without reference to management strategies or approaches [2,5,20]. Instead, their focus lies on emergent processes in HEIs. Hugé, MacLean, and Vargas [5] (p. 4279), for instance, conceive sustainability implementation as a process of social issue maturation involving four phases: Emergence, popularization, formalization into a governance framework, and maturity (uptake of sustainability as a norm). Models such as this convey a less rationalistic image of sustainable development in HEIs; they also leave more room for different, diverging approaches of sustainability governance. However, due to the high degree of abstraction involved, it becomes more difficult to derive recommendations for sustainability implementation from them.

A kind of middle ground can be found in a contribution by Ferrer-Balas, Buckland, and de Mingo [21]. Drawing on systems thinking, they distinguish three interacting dimensions—the framework (changes in culture, institutional structure, and technology), the level of system change (system optimization, system improvement, system renewal), and the actors (cooperation between different stakeholders)—which can be used to establish and evaluate strategic and operational objectives. One of the strengths of this framework is that it is utilized in combination with an assessment tool—an approach that can also be found in other frameworks and models [14] (p. 189). This appears interesting, as such tools can provide a bridge between rather abstract models and frameworks, and practical action to promote sustainability in HEIs.

Another approach is the use of a governance equalizer. This idea is not new. It can be traced back to Clark's notion of "pathways of coordination" in higher education [22]. Schimank [23,24] later used the term 'governance equalizer' in an allusion to the equalizer in a mixing console or a stereo system, where different dimensions can be turned up or down. He focused on how the introduction of New Public Management (NPM) gradually shifted dimensions of governance in higher education—such as academic self-organization or top-down management. This research interest led to rather specific governance dimensions. In order to broaden their applicability to other research, Brüsemeister and colleagues put forward an alternative set of governance dimensions that encompasses five dimensions: Politics, profession, organization, knowledge, and the public (for an overview, see Brüsemeister [25]). The five dimensions were initially derived from empirical findings, but are also supported by sociological and organizational theory [26,27].

Recently the governance equalizer has been applied to SD in HEIs with the five dimensions conceived as functional requirements that sustainability governance must address if it is to work effectively [28,29]. This approach used the governance equalizer as a heuristic framework for the analysis of sustainability governance in HEIs. While it thus remained open to different approaches and contexts, it did not provide criteria and procedures to assess sustainability governance in HEIs. This paper aims to close this gap by introducing a (self-) assessment tool based on the governance equalizer. To this end, the following section takes a closer look at sustainability assessment in HEIs before turning to the development and specifics of the tool.

3. Sustainability Assessment in Higher Education Institutions

Sustainability assessment tools (SATs), which are deemed an essential component of sustainability processes in HEIs by some scholars (e.g. [4,28]), have been in use for some time. Several studies provide analyses and comparisons of these tools [29–33]. They show that different purposes and functions are associated with SATs. We summarize these into two key functions: An analytical or cognitive function,

and a communicative or social function. The analytical/cognitive function comprises monitoring of an HEI with the aim of measuring what is being done to move toward sustainability and how it is done, to draw comparisons across campus, to assess progress and identify performance gaps and good practice, and to provide knowledge for creating strategies and planning. The communicative/social function includes building bridges among different stakeholders and encouraging an HEI's move towards sustainability by communicating goals and experiences, stimulating reflection, advocating policy change, triggering and supporting organizational change processes, and lobbying government and media to promote campus sustainability.

In addition to functions associated with them, extant SATs also differ in the domains they cover as well as in the underlying methodology. Yarime and Tanaka [30] show that education, research, and outreach are not well addressed by most tools, as their focus frequently lies on operational eco-efficiency and governance issues. This focus has been explained by the fact that aspects of governance and operations "are easily observable and manageable, often with quantitative goals and objectives" [30] (p. 74). However, this does not apply equally to all aspects of governance. Governance indicators in SATs frequently refer to visions, strategies, policies, planning, and initiatives. By contrast, informal and more complex issues—such as faculty development in sustainability areas, implementation of policies, multi-stakeholder involvement in decision-making, and inter- and transdisciplinary approaches—have not been addressed adequately [30] (p. 73 f.).

The choice of methodology is thus crucial for SATs. Quantitative approaches alone have been judged insufficient for sustainability assessment purposes because they are unable to capture the interactions within complex systems [34] (p. 2). It has also been criticized that SATs in HEIs predominantly follow a managerialist approach, which aims to incorporate sustainability issues into existing management accounting practices and, in doing so, fails to question underlying unsustainable patterns [4] (p. 1764). SATs, it has been concluded, should not focus on accountability and control, but on stimulating reflection and change, and the use of qualitative indicators is called for to support this aim [28] (p. 818). At the same time, the issue of subjective bias in qualitative assessment needs to be addressed by engaging all stakeholders, defining clear criteria, and insisting on dialogue and consensus among all involved in the assessment [28](p. 819).

4. Tool Development—Background and Project Design

In the following, we introduce a tool for assessing sustainability governance in HEIs. On the one hand the tool builds on the insights obtained from the literature. On the other hand it draws on empirical data from case studies in eleven German HEIs, which were carried out in the project 'Sustainability at Higher Education Institutions: Develop—network—report' [35]. The case studies consisted of 61 face-to-face interviews with different stakeholders (management, sustainability coordinators, student initiatives, technical administration, and academic staff). Inclusion of interview partners followed a selective sampling process. In addition to their time and willingness to participate, criteria for selection included knowledge of and first-hand experience with their institution's sustainability process [36] to ensure that substantial expert knowledge of the sustainability processes of the higher education institutions could be collected. The interviewees were identified and recruited in collaboration with partners within the HOCH-N project and through desktop research. The case studies drew on the concept of a governance equalizer to investigate, inter alia, how the HEIs addressed the functional requirements of sustainability governance (see next section for details).

The interviews were analysed using qualitative content analysis [37]. This included examination of the measures the HEIs took to improve the five dimensions of the governance equalizer, of how progression in each dimension was manifested in institutional arrangements and practices, and of how stakeholders judged the performance of the HEIs in each dimension. By comparing the eleven cases, five levels of progression were defined for each dimension. The findings fed into a preliminary version of the assessment tool, containing a key question and a five-point scale of progression for each governance dimension, along with practical examples for each point on the scales.

The preliminary version of the assessment tool was published as part of a (beta version of a) guide on sustainability governance in HEIs [38]. In a subsequent project phase, the assessment tool was put to the test. To this end, stakeholder workshops were held in several German HEIs which were neither included in the case studies nor in the HOCH-N project. We attempted to include representatives from all stakeholder groups in the HEI, but actual participation from these groups varied. Overall, however, the workshops included feedback from all relevant stakeholder groups. The workshops started out by explaining the five governance dimensions and the assessment scales. Participants were then asked to assess their HEI's performance on each scale and to provide arguments and concrete examples to substantiate their assessment. Furthermore, following the assessment, participants were asked to comment on the comprehensibility and applicability of the tool. The feedback gathered in the validation workshops led to a number of revisions: Comprehensive explanations of the five dimensions were added, definitions of the levels of progression were improved, and additional examples for each point of the scales were included. The description of the tool in the following section reflects these revisions.

5. A Systemic Tool for Assessing Sustainability Governance in HEIs

The design of the assessment tool picks up on some key points discussed above. The tool's main purpose is to support HEIs in their transformation toward sustainability. It aims at providing analytical insight and practical guidance for HEIs while acknowledging the complexity of sustainability governance. In particular, we see the need to ensure an integrated, transversal perspective, to account for the entangled relationships among actors in HEIs, and to recognize the importance of context for sustainability processes and their outcomes. Our approach thus bears a certain proximity to systems thinking (cf. [39] p. 18 ff.). Systems thinking deals with complexity by "increasing the level of abstraction or overview, rather than the conventional reductionist route of examining detail and dividing the issues into smaller parts" [2] (p. 49 f.). The tool follows this prescription to some extent, but attempts to uphold a link with concrete governance structures, processes and instruments.

To this end we approach sustainability governance from a functionalist perspective, as represented by the governance equalizer. As mentioned, the five dimensions represent requirements that sustainability governance must address if it is to work effectively (see also [27,40]). The tool provides a scale for each governance dimension. Some of the criteria and examples in the scales lend themselves to quantification. However, such quantitative indicators alone do not provide sufficient information to assess the state of sustainability governance. Qualitative information—for example, on the quality of multi-stakeholder interaction, the creation of a common understanding of sustainability, or the transfer and use of knowledge—is also required. To allow a comprehensive assessment, the tool uses clearly defined criteria that are assessed using carefully developed maturity scales (cf. [41,42]). It thus provides a means to analyze the existing status of sustainability governance, to identify areas for improvement and, by being applied repeatedly, to assess progress in governance.

This section provides an overview of the five dimensions and how they are accounted for in the assessment tool. Each subsection begins with a definition of the respective governance dimension. This is followed by the five-point scale used to assess progress in the dimension, including definitions, key aspects, and practical examples for each point on the scale. Finally, points of particular interest are discussed, including issues raised in the validation workshops.

5.1. Politics—How is Sustainability Entrenched and Legitimized in the HEI?

This dimension deals with the question of how sustainability can move beyond individual support points and be embedded long-term on the HEI's agenda. The term 'politics' in this context does not (primarily) refer to political institutions outside the HEI, but to actors within the HEI who need to take formal and informal decisions regarding internal goals, structures, procedures, and measures, as well as membership of external networks in the context of sustainability. Collectively binding decisions lend justification to and offer guidance for actions toward sustainability, and provide criteria to judge their success.

On the lowest level of progression in this dimension, support from decision-makers for SD-related activities is merely granted unofficially (see Table 1). The next level includes public, official commitment to the goal of sustainability. On the following levels, sustainability is subsequently codified as an institutional goal (becoming more and more independent from the support of key individuals), broken down to different HEI domains, linked to responsibilities and resources and, finally, operational measures are defined and their implementation and effects assessed.

s'.
;

Level	Definition	Examples
5	The objectives of the sustainability process are translated into binding, operational measures and their implementation and effects are <u>evaluated</u> .	 Agreements on sustainability-related targets have been set and implemented. Procurement agreements include concrete sustainability-related provisions and criteria. Binding decisions to include sustainability in research have been taken. Administrative sustainability units and steering bodies are mandated to approach and involve other HEI staff to pursue sustainability-related goals Sustainability-related evaluation, accountability, reporting, and control instruments and practices are established Sustainability-related auditing/certification takes place. Binding operational measures in different domains have been introduced.
4	The goal of sustainability is broken down to and <u>embedded in different domains</u> .	 Issues of sustainability are included in procurement directives. (Domain-specific) strategies are available and include sustainability-related objectives. Sustainability is included in the HEI's charter. Responsibilities for sustainability issues and tasks have been assigned. Concepts and/or guidelines on sustainability have been developed.
3	Sustainability is codified—in a non-binding form—as a general goal of the HEI.	 Sustainability has been included in the HEI's mission statement and has thus been established as a task independent from individual agents in the HEI. Sustainability features in the HEI's name or in the name of one or more of its organizational units. Agreements about participation/membership in inter-organizational networks among HEIs have been signed by the HEI.
2	Decision-makers in the HEI <u>publicly voice</u> their <u>commitment</u> to the goal of sustainability.	 Public declarations stating the commitment of decision-makers in the HEI to sustainability exist. Decision-makers in the HEI openly support existing initiatives towards sustainability.
1	Individual decision-makers in the HEI recognize and support sustainability-related activities in informal and non-public ways.	- There are informal, non-public declarations of intent and commitment to sustainability.

On the lower levels, the scale emphasizes the role of decision-makers. It is important to clarify that this term does not exclusively refer to top-level management in the HEI, but encompasses all members of the HEI that take part in decisions on goals, priorities, resource allocation, etc. The emphasis on decision-makers echoes the important role that is attributed in parts of the literature to decision-makers in general and of top management in particular. In our view, this role should not be overestimated, however. Managers of course have an important role to play, but while their leadership can be a necessary condition of success, it is hardly sufficient. Instead, successful transformation requires that the goal of sustainability should spread throughout the entire organization and must be translated into concrete decisions and actions. The upper levels of the scale reflect this imperative. Furthermore, it is important that binding decisions should not be confused with centralized, top-down decision-making. Decisions can be arrived at via both hierarchical as well as participative decision-making processes, and, in many instances, broad stakeholder participation may well be more conducive to the sustainability process.

5.2. Profession—How are Different Professional Perspectives and Competencies being Connected?

The dimension 'profession' focuses on the development of an interdisciplinary and transversal understanding of sustainable development in the HEI. The different domains—education, research, campus, and outreach—are marked by different demands, processes, and framework conditions. As a result, they require specific competencies and knowledge, and exhibit specific standards and cultures. Similarly, differences among academic disciplines as well as among external actors related to the HEI can be observed. Moving towards sustainability necessitates a cross-cutting dialogue about what SD should encompass, what principles and standards should apply, and how SD can be integrated in everyday practices in the different domains and disciplines.

On the lowest level of progression, this involves reflections on issues of sustainability by individuals in specific domains or faculties (see Table 2). A next level is reached when groups of individuals in a domain or in a faculty jointly reflect on sustainability. Higher levels include an exchange of ideas and perspectives across domains and/or disciplines, eventually resulting in a common position on sustainability which, ultimately, is reflected in everyday professional actions and joint (interdisciplinary and transversal) activities.

Level	Definition	Examples
5	A common understanding of sustainability is reflected in inter/transdisciplinary and transversal practices within the HEI and beyond, and such practices are a defining trait of the HEI.	 Inter- and transdisciplinary courses and research projects are continuously developed and refined. Sustainability is a mandatory course content for all students at the HEI. Transdisciplinary activities (such as project workshops, Real-World Laboratories) within the HEI and together with external actors are realized on a continuous basis. Sustainability serves as a criterion in appointment procedures. There is a range of trainings on sustainability-related issues that are mandatory for HEI staff. Permanent and temporary academic positions in transdisciplinary research on sustainability have been created.
4	Actors across disciplines and domains have developed a common understanding of sustainability, which they continuously review and revise.	 A common understanding of sustainability for the whole HEI (e.g., in the form of a mission statement) has been developed. A joint transversal position on sustainability has been established. A sustainability strategy exists.
3	A dialogue on sustainability across different domains and across disciplines takes place.	 Different formats of interdisciplinary exchange (such as a research platform) have been established. Conferences and symposia on sustainability-related issues are held. Interdisciplinary lecture series, colloquia, degree courses, and research projects addressing issues of sustainability are carried out.
2	Sustainability issues are addressed collectively within individual faculties or domains.	 Sustainability issues are addressed by environmental management. Teachers at a faculty collaborate to strengthen/include sustainability in the faculty's courses.
1	Individual actors within separate faculties or domains address issues of sustainability.	 Individual researchers or research projects focus their work on sustainability issues. Individual teachers address sustainability in their courses. Individuals from other stakeholder groups work on sustainability-related issues.

Table 2.	Assessment scale	: 'Profession'.

Examples in the 'profession' scale frequently refer to professional practices such as academic courses or environmental action. This is particularly pertinent for Levels 1, 2, and 5. Finding a few of these or similar examples in an HEI, however, can hardly be seen as progression of the HEI as a whole. The assessment must therefore also include the institutional spread of such practices.

In this context, it also needs to be stressed that some of the examples, such as courses focusing on sustainability, are important steps toward an education for sustainable development. In the context of the assessment tool, however, this aspect is not of particular interest. By contrast, changed practices are important because they indicate a transformed professional understanding, which can form the basis for joint action and institutional transformation.

Part of this is the process of aligning different ways of understanding of sustainability. Importantly, this does not mean that a single, unified understanding should be achieved. Both case studies and validation workshops repeatedly showed the need to leave room for diverging professional perspectives. Nonetheless, it is important for HEI stakeholders to come to an agreement that allows joint action. To this end, it might be helpful to settle on a set of normative boundaries within which all stakeholders can pursue their different professional rationales.

5.3. Organization—How are Cooperative Work and Task Performance Made Possible?

Moving HEIs towards sustainability requires breaking down sustainability-related goals so that concrete actions can be taken. This includes the provision of adequate resources and creating structures and procedures that ensure continuous and reliable work. What is more, actions must extend beyond existing organizational boundaries, and interdisciplinary and transversal networks and coordination of activities play an important role. Networking involves actors exchanging views and knowledge and cooperating, whereas coordination aims to ensure coherence and synergies between sustainability-related activities. Overarching coordination does not necessarily mean centralized control, however. Rather, it can also aim to support decentralized initiatives in order to maximize their effectiveness.

At the lowest level of this dimension, individual actors in an HEI take actions towards sustainability (see Table 3). On the subsequent levels, such actions are channeled through projects or other initiatives, structures, and procedures are created to facilitate networking, and resources are provided to coordinate sustainability-related activities. While such provisions are often temporary and rely on specific (competent, motivated, well-connected) persons, at the highest level of this dimension, networking and coordination are established as a permanent function in the HEI and backed by regulations and long-term allocation of resources.

Level	Definition	Examples
5	Firmly established (yet flexible) institutions and processes for the <u>coordination</u> of sustainability-related activities exist <u>on a permanent basis</u> .	 Permanent functions/staff positions to ensure coordination and networking independent of specific individuals have been created. Other sustainability-related tasks and objectives have been codified and permanent staff positions have been created to ensure implementation. Management functions have been expanded to all domains (from environmental to sustainability management). Procurement throughout the institution is based on binding sustainability-related directives.
4	<u>Resources for coordination</u> of sustainability-related activities are provided on a temporary basis.	 A central coordination unit for sustainability-related issues has been established on a temporary basis. Institutions such as green offices or similar contact points have been set up. Temporary posts are in place in the administration for the performance of sustainability-relevant tasks.
3	Structures and procedures exist to facilitate <u>networks</u> among sustainability initiatives in the HEI.	 Networking and dialogue across faculties and domains (possibly including external stakeholders) have been institutionalized, e.g., in the form of round tables, working groups, commissions, or other authorized bodies. Networks and communication platforms to ensure dialogue and cooperation with external actors exist.
2	Sustainability-related <u>actions</u> are <u>taken</u> collectively within separate faculties or domains.	 Decentralized procedures to align sustainability-related actions in the HEI exist (no overarching coordination). Projects and initiatives are carried out independently, without coordination across faculties or domains.
1	Actions towards sustainability are taken by <u>individual actors</u> in the HEI.	- Individual students or academic or administrative staff members carry out sustainability-related activities.

Table 3.	Assessment scale:	'Organization'.
----------	-------------------	-----------------

In total, the examples provided in the scale deal with collective capacity for action and how action can be directed towards sustainability through rules, incentives, etc. Many of the examples refer to structures that need to be in place for this purpose, such as committees, coordination units, or staff positions. The formal existence of such structures by itself, however, does not guarantee effective action. When assessing the 'organization' dimension, it is therefore necessary to reflect how these structures work in practice and to judge if they actually serve their purpose. This would include, for example, analysing the quality of interaction in a committee or the actual mandate of a coordinating unit and the barriers that it might encounter in its work.

5.4. Knowledge—How is the Necessary Knowledge Generated and used Competently?

Sustainable development calls for complex knowledge management. For joint action, actors in HEIs must develop a common understanding of the problems to be addressed and their causes (systems knowledge), they must agree on a judgment of the current situation and set goals for the future (target knowledge), and they must identify ways to solve the problems at hand (transformation knowledge). In addition to technical expertise, this requires knowledge about actors, structures, and processes in the HEI in order to determine the preconditions for successful implementation. Furthermore, it is insufficient for effective sustainability processes to draw upon knowledge in the HEI on an ad hoc basis. Instead, the HEI needs to create ways to continuously identify, generate, disseminate, and utilize knowledge in order to react adequately to emerging problems and facilitate longer-term learning processes. In addition to technical solutions, this calls for participation and networking to support knowledge transfer.

On the first level of progression in this dimension, the relevant knowledge is held by a limited number of individuals (see Table 4). On the next level, the existing knowledge is documented and made available to other actors. While in this case knowledge only flows in one direction, the subsequent levels increasingly include the mutual exchange and joint creation of knowledge. This involves creating opportunities for knowledge transfer, joint problem-solving activities and, ultimately, building the capacity to continuously process and use knowledge to support the sustainability process in the HEI in a longer-term perspective.

It was mentioned before that in assessing structures, the actual working processes associated with them need to be taken into account. This is especially relevant for the 'knowledge' dimension. The structures and procedures listed above are merely the 'containers' that serve to facilitate knowledge work in the HEI. Consequently, the existence of such informal and formal structures, while an important prerequisite, is merely the starting point, and the contents and processes of this knowledge work and their quality should be at the centre of the assessment in this dimension.

Level	Definition	Examples
5	Structures and procedures for continuous joint knowledge work are used to support the sustainability process in the long term.	 Dialogic forms of cooperation (e.g., committees) exist that provide room to work on sustainability-related issues in a long-term perspective (independent of present problems that require short-term solutions). Comprehensive knowledge (inventory, analysis of problems and causes, analysis of actions and their effects) is generated and used to support the management and coordination of the sustainability process. Sustainability reporting is linked to concrete sustainability-related measures and goals.
4	Structures and procedures for joint knowledge work aiming at (short-term) solutions to present problems are in place.	 Dialogic forms of cooperation (e.g., committees) exist that provide room to work on concrete actions toward sustainability (e.g., events, guidelines, courses, or projects). Evaluations of specific activities/measures are carried out. Sustainability reporting provides an analysis of sustainability-related issues and challenges.
3	Opportunities for knowledge exchange are in place.	 Research platforms that enable the sharing of individual knowledge have been created. Conferences and colloquia on sustainability-related issues take place. Transformative, participative, interactive courses focusing on sustainability issues are carried out.
2	Knowledge is <u>documented and made available</u> (unidirectional, without dialogue/exchange).	 A sustainability report is published. Libraries and databases provide sustainability-related literature and information. Sustainability-related training for employees and researchers is available. Individual lectures and seminars on sustainability-related topics are held. Handouts and guidelines on sustainability are provided, e.g., by the administration. Formats such as newsletters or websites on sustainability exist.
1	Relevant knowledge is limited to individuals and/or projects and is not taken up by the HEI.	 Knowledge on sustainability issues is generated individually, e.g., thesis papers or research projects.

Table 4. Assessment scale: 'Knowledge'.

5.5. The Public—How is Awareness of the Need for Sustainable Development Achieved in HEI?

Making sustainability initiatives visible in public is an important part of sustainability governance in HEIs, because doing so creates the opportunity for stakeholders to observe issues, positions, activities, and their results, and to react to them. For instance, actors in HEIs can contribute to increased awareness and participation by demonstrating the need for action, communicating goals and measures, and reporting on progress made. Public attention also helps to reinforce the importance of sustainable development both within and outside the HEI, and to communicate sustainability as part of the HEI's profile in relation to (potential) students and external partners.

On the lowest level, the public dimension involves active communication about sustainability by a small circle of individuals, mostly in their immediate professional surroundings (see Table 5). The second level is reached when individual faculties or other organizational units pursue a targeted communication approach. On the third level, such an approach addresses the whole institution as well as the external public. Ideally, this leads to sustainability becoming a characteristic part of the HEI's identity and, ultimately, the HEI's profile.

Level	Definition	Examples
5	Sustainability is a central distinguishing feature of the HEI, both internally and externally.	 Specific faculties, departments, or the HEI as a whole carry the term sustainability in their names. When new professorships are advertised, they are specified with sustainability in mind. Local sustainability-related debates are taken up and shaped by members of the HEI. Sustainability research and teaching are attractive for students and teachers.
4	Sustainability is a visible part of the <u>HEI's conception</u> .	 Sustainability is included in the HEI's mission statement. The HEI awards sustainability prizes. Sustainability-related courses, lectures, and other events are open to external interested parties (pupils, senior citizens, etc.). A sustainability report is published.
3	A coordinated approach exists to communicate sustainability issues within the <u>whole institution</u> and to the general public.	 Research platforms that enable the sharing of information on sustainability-specific issues have been created. Conferences and colloquia on sustainability-related issues take place. Transformative, participative, interactive courses focusing on sustainability issues are carried out. The HEI's website provides information on its sustainability-related goals and activities.
2	Targeted <u>measures</u> are carried out by organizational units to communicate sustainability issues.	 Sustainability-related issues are addressed in newsletters by faculties, while administrative departments or projects address sustainability issues in newsletters or at specific events, press conferences, etc.
1	A <u>small circle of individuals</u> actively engages in (informal) communication about sustainability issues.	 Sustainability-related information is passed on to those directly affected (e.g., energy-saving in the administration). Committed stakeholders share sustainability-related information with their colleagues.

Table 5. Assessment scale: 'The public'.

Making sustainability part of the HEI's public image was a particular point of discussion in the validation workshops. While the case studies included HEIs that built their entire identity around the issue of sustainability, some workshop participants argued that this was impossible for larger HEIs, which can't focus solely on a single issue and for which it is more difficult to raise awareness of sustainability issues throughout the institution. The revised scale reflects these concerns, but also maintains that incorporating sustainability in an HEI's identity, both internally as well as in its public image, is an important functional requirement for the transformation toward sustainability.

5.6. Working with the Assessment Tool

The main function of the tool does not lie in objective measurement and assessment of sustainability governance, much less in providing the basis for benchmarking among HEIs. Its primary purpose is to enable stakeholders in an HEI to attain a comprehensive picture of sustainability governance that is grounded in systematic analysis, includes different perspectives, and draws on different types of knowledge. In this way, it facilitates dialogue among stakeholders and promotes agreement on both the existing status and the way forward. Furthermore, while some of the information required for the assessment can be obtained from official documents such as strategy papers, sustainability reports, or procurement directives, the knowledge of different stakeholders in the HEI is the key source for a comprehensive assessment.

Against this background, a participatory approach that involves all stakeholder groups in the HEI is recommended. This can take the form of stakeholder workshops (as used in testing the assessment tool) or similar formats. In addition, the test phase showed that it can be helpful to take stock of sustainability governance before carrying out a stakeholder workshop and to present an overview (possibly coupled with a preliminary assessment) to the workshop participants. By doing so, participants are provided with a shared knowledge base and a more structured discussion is facilitated.

The analysis should cover all domains in the HEI (education, research, campus, and outreach). It is not to be expected that sustainability governance will be equally developed in all domains. For example, there might be far-reaching governance structures in place in campus management, but not in education or research. It is therefore important to look at each domain specifically. Some participants in the validation workshops even suggested conducting a separate analysis for each domain. However, we believe that the concept of sustainable development requires a whole-institution approach and that all domains should therefore be included in an integrated analysis.

The assessment process starts out by identifying elements of sustainability governance in the HEI that are relevant for one or several of the five dimensions, and assessing their contribution to those dimensions. This includes the following steps:

- 1. Creating an overview of existing governance elements (gathering information on all elements of sustainability governance).
- 2. For each governance dimension: Checking which elements are relevant (assigning all relevant elements to the respective dimension(s)).
- 3. For all elements assigned to the respective dimension: Evaluating the contribution of each element (assigning a score on the assessment scale).

The examples listed in the assessment scales provide some orientation for identifying and assessing governance elements. It is important to be aware, however, that in the end they are just that—examples. This means that effective sustainability governance does not require that all examples from the scales are in place or that forms of sustainability governance should be limited to these examples. On the contrary, the analysis needs to be open to specific approaches in the HEI that may differ from the examples.

The results of the analysis can be documented by creating a table for each governance dimension that lists all elements of sustainability governance assigned to each point on the scale (using the assessment scales as templates). As above, this can lead to a number of examples at each level of a scale. The analysis then moves on to the next steps:

- 4. Conducting an overall assessment in each dimension (determining a score).
- 5. Identifying the strengths and weaknesses of sustainability governance in the HEI and the requirements for action.

Determining a score for each dimension requires weighing the relative importance of different governance elements, and should be based on discussion among stakeholders. While the assessment scores are thus not objective, they are grounded in the stakeholders' expertise and undergo intersubjective validation. Together, the scores from the five governance dimensions provide a comprehensive assessment of sustainability governance in the HEI. Forms of visualization such as a spider diagram can be used to illuminate the relative strength of the five governance dimensions. It is then up to the stakeholders in the HEI to discuss how the situation should be judged, if one or more governance dimensions needs to be intensified, and how this could be achieved.

6. Discussion

The governance equalizer focuses on the functional requirements that sustainability governance needs to address if it is to work effectively. Framing these requirements in the form of the governance equalizer suggests that HEIs can influence—'turn up' or 'turn down'—the level of each dimension

in their institution—and that turning up one or more dimensions will increase the chances of successful sustainability processes. Elements of governance—structures, processes, instruments, and practices—can thus be analyzed by asking if and how they contribute to these functional requirements. This approach, in our view, has several advantages.

Firstly, by considering different functions, the analysis facilitates a multi-dimensional understanding of governance elements, instead of limiting them to a single purpose. For instance, sustainability reporting can provide an information base ('knowledge') for decisions about a sustainability strategy and to judge its success ('politics'). At the same time, the process of gathering data, assessing progress, and writing the sustainability report could change stakeholder understanding of sustainability and their own role in moving toward it ('profession'). Furthermore, publication of the report could raise overall awareness of sustainability issues ('the public'). Sustainability reporting could thus affect several governance dimensions to different degrees—and its main impact might differ from what would be expected at first glance.

Secondly, by focusing on functions we avoid any overemphasis on individual elements and ask instead if and how the mix of different elements provides a basis for effective governance. For example, parts of the literature highlight the role of sustainability 'champions'. This is also reflected in the assessment tool, as several of the definitions and examples make reference to dedicated and influential individuals. However, the assessment scales also illuminate the complexity of the task of moving HEIs toward sustainability, and demonstrate the multitude of other pieces that also need to be put into place. In the process, rather than depending on individual actors, sustainability has to become institutionally embedded through a number of complementary elements.

Thirdly, the functional perspective offers a way to analyze the divergent governance structures and processes that the HEI creates on the basis of specific institutional cultures (cf. [43]) under a common framework. Instead of looking for 'the best way', the assessment tool leaves room for different approaches to sustainability governance that might be effective (in specific contexts), and thus creates the possibility of identifying 'functional equivalents'. While this can also be done for individual elements of sustainability governance, it applies in particular to the combination of several elements, i.e., the governance mix that underlies an HEI's sustainability process.

The assessment tool aims to create some middle ground between quantitative, objective measurement and the need to capture the more complex aspects of governance. To this end, on the one hand, the analysis is based on clearly defined criteria that are assessed using carefully developed maturity scales (cf. [41]). On the other hand, the tool puts a strong emphasis on the participation of stakeholders in HEIs. These are not just important sources of information—assessing the five governance dimensions ultimately relies on their judgment. We concur with some of the literature that the inclusion of different perspectives in this process represents a way of ensuring its quality, as the chances of avoiding one-sided judgments improve. However, this does not mean that stakeholders need to agree on all issues. In fact, as systems thinking teaches us, it can be useful to deliberately seek out differences in order to achieve a better understanding of the situation [39] (p. 12). In this sense, identifying areas of disagreement can be an important part of the assessment and provide clues for improvements.

Of course, governance in HEIs is only one element of their transformation to sustainability, but one that should support fundamental changes in an HEI's core domains. Sustainable development in these domains can also be supported by sustainability assessment tools. It is the aim of this article to provide the means for a more refined analysis of sustainability governance. This does not mean that other domains should be neglected, however. In fact, it could be both useful and feasible to use more than one assessment tool to support sustainability processes in HEIs [28] (p. 819).

7. Conclusions

This article introduced a sustainability assessment tool centered on five governance dimensions. While these dimensions have been drawn on for analytical purposes, they do, of course, also include

a normative component. They are based on the premise that attaining sustainable development is desirable and that, in order to do so, organizational boundaries need to be bridged, perspectives aligned, overarching coordination ensured, and so on. These assumptions are in line with much of the literature on sustainable development as well as with governance literature in general. Nonetheless, they, too, should be subject to further empirical research. Future insights might therefore require a revision of the functional norms represented by the five dimensions. Moreover, the dimensions and examples were derived exclusively from empirical cases and workshops in German HEIs. The applicability of the assessment tool for HEIs in other countries, especially in the Global South, remains to be empirically investigated.

Author Contributions: Conceptualization, S.N., M.B., M.D., L.J., M.R. and I.B.; methodology, S.N., M.B., M.D., L.J., M.R. and I.B.; validation, S.N., M.B., M.D., L.J., M.R. and I.B.; formal analysis, S.N., M.B., M.D., L.J., M.R. and I.B.; writing—original draft preparation, S.N.; writing—review and editing, S.N., M.B., M.D., L.J., M.R. and I.B.; visualization, S.N., M.B., M.D., L.J., M.R. and I.B.; supervision, M.R. and I.B.; project administration, S.N., M.B. and I.B.; and M.R.; funding acquisition, M.R. and I.B. All authors have read and agree to the published version of the manuscript.

Funding: This research was funded by the Bundesministerium für Bildung und Forschung, Grant/Award Number: 13NKE007A. The article processing charge was funded by the German Research Foundation and the Open Access Publication Fund of the Freie Universität Berlin.

Acknowledgments: We thank two anonymous reviewers for their feedback on an earlier version of this paper.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Hopwood, B.; Mellor, M.; O'Brien, G. Sustainable development: Mapping different approaches. *Sustain. Dev.* 2005, 13, 38–52. [CrossRef]
- Sterling, S. Higher Education, Sustainability, and the Role of Systemic Learning. In *Higher Education and* the Challenge of Sustainability: Problematics, Promise, and Practice; Corcoran, P.B., Wals, A.E.J., Eds.; Kluwer Academic Publishers: Dordrecht, The Netherlands, 2004; pp. 49–70.
- Hoover, E.; Harder, M.K. What lies beneath the surface? The hidden complexities of organizational change for sustainability in higher education. J. Clean. Prod. 2015, 106, 175–188. [CrossRef]
- 4. Arroyo, P. A new taxonomy for examining the multi-role of campus sustainability assessments in organizational change. J. Clean. Prod. 2017, 140, 1763–1774. [CrossRef]
- 5. Hugé, J.; Mac-Lean, C.; Vargas, L. Maturation of sustainability in engineering faculties—From emerging issue to strategy? *J. Clean. Prod.* 2018, *172*, 4277–4285. [CrossRef]
- 6. Wals, A.E.J. Sustainability in higher education in the context of the UN DESD: A review of learning and institutionalization processes. *J. Clean. Prod.* **2014**, *62*, 8–15. [CrossRef]
- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]
- Leal Filho, W.; Shiel, C.; Paço, A.; Mifsud, M.; Veiga Ávila, L.; Londero Brandli, L.; Molthan-Hill, P.; Pace, P.; Azeiteiro, U.M.; Ruiz Vargas, V.; et al. Sustainable Development Goals and sustainability teaching at universities: Falling behind or getting ahead of the pack? J. Clean. Prod. 2019, 232, 285–294. [CrossRef]
- 9. Torfing, J.; Ansell, C. Introduction: Theories of Governance. In *Handbook on Theories of Governance*; Ansell, C., Torfing, J., Eds.; Edward Elgar Publishing: Cheltenham, UK, 2016; pp. 1–17.
- Barth, M.; Rieckmann, M. State of the Art in Research on Higher Education for Sustainable Development. In Routledge Handbook of Higher Education for Sustainable Development; Barth, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge: London, UK, 2016; pp. 100–113.
- 11. Sharp, L. Green campuses: The road from little victories to systemic transformation. *Int. J. Sustain. High. Educ.* 2002, 3, 128–145. [CrossRef]
- 12. Moore, J. Seven recommendations for creating sustainability education at the university level: A guide for change agents. *Int. J. Sustain. High. Educ.* 2005, *6*, 326–339. [CrossRef]
- Corcoran, P.B.; Walker, K.E.; Wals, A.E.J. Case studies, make-your-case studies, and case stories: A critique of case-study methodology in sustainability in higher education. *Environ. Educ. Res.* 2004, 10, 7–21. [CrossRef]

- Verhulst, E.; Lambrechts, W. Fostering the incorporation of sustainable development in higher education. Lessons learned from a change management perspective. J. Clean. Prod. 2015, 106, 189–204. [CrossRef]
- Lozano, R. Incorporation and institutionalization of SD into universities: Breaking through barriers to change. J. Clean. Prod. 2006, 14, 787–796. [CrossRef]
- Velazquez, L.; Munguia, N.; Platt, A.; Taddei, J. Sustainable university: What can be the matter? J. Clean. Prod. 2006, 14, 810–819. [CrossRef]
- Lukman, R.; Glavič, P. What are the key elements of a sustainable university? *Clean. Technol. Environ. Policy* 2007, 9, 103–114. [CrossRef]
- Holmberg, J.; Lundqvist, U.; Svanström, M.; Arehag, M. The university and transformation towards sustainability. The strategy used at Chalmers University of Technology. *Int. J. Sustain. High. Educ.* 2012, 13, 219–231. [CrossRef]
- Holm, T.; Sammalisto, K.; Grindsted, T.S.; Vuorisalo, T. Process framework for identifying sustainability aspects in university curricula and integrating education for sustainable development. J. Clean. Prod. 2015, 106, 164–174. [CrossRef]
- Newman, J. An Organisational Change Management Framework for Sustainability. Greener Manag. Int. 2007, 2007, 65–75. [CrossRef]
- Ferrer-Balas, D.; Buckland, H.; de Mingo, M. Explorations on the University's role in society for sustainable development through a systems transition approach. Case-study of the Technical University of Catalonia (UPC). J. Clean. Prod. 2009, 17, 1075–1085. [CrossRef]
- 22. Clark, B.R. The many pathways of academic coordination. High. Educ. 1979, 8, 251–267. [CrossRef]
- Schimank, U. Welche Chancen und Risiken können unterschiedliche Modelle erweiterter Universitätsau-tonomie für die Forschung und Lehre der Universitäten bringen? In Universitäten im Wettbewerb: Zur Neustrukturierung Österreichischer Universitäten; Titscher, S., Ed.; Hampp: München, Germany, 2000; pp. 94–147.
- Schimank, U. Die Governance-Perspektive: Analytisches Potenzial und anstehende konzeptionelle Fra-gen. In Educational Governance: Handlungskoordination und Steuerung im Bildungssystem, 1. Aufl.; Altrichter, H., Brüsemeister, T., Wissinger, J., Eds.; VS Verlag für Sozialwissenschaften | GWV Fachverlage GmbH Wiesbaden: Wiesbaden, Germany, 2007; pp. 231–260.
- Brüsemeister, T. Vergleichende Analyse mittels Governance-Regler. Eine Synopse. In Handbuch Educational Governance Theorien; Langer, R., Brüsemeister, T., Eds.; Springer Fachmedien Wiesbaden: Wiesbaden, Germany, 2019; pp. 35–49.
- Niedlich, S.; Brüsemeister, T. Modelle regionalen Bildungsmanagements—Ansätze zur Behebung sozialer und bildungsbezogener Ungleichheiten. In *Neue Steuerung—Alte Ungleichheiten? Steuerung und Entwicklung im Bildungssystem*; Dietrich, F., Heinrich, M., Thieme, N., Eds.; Waxmann: Münster, Germany, 2011; pp. 201–218.
- Niedlich, S.; Kummer, B.; Bormann, I.; Rieckmann, M.; Bauer, M. Governance-Regler als Heuristik für die Analyse von Nachhaltigkeitsgovernance an Hochschulen. AP Gov. Arbeitspapier No. 2. 2017. Available online: https://www.hochn.uni-hamburg.de/-downloads/ap2-governance-regler.pdf (accessed on 26 February 2020).
- 28. Berzosa, A.; Bernaldo, M.O.; Fernández-Sanchez, G. Sustainability assessment tools for higher education: An empirical comparative analysis. *J. Clean. Prod.* **2017**, *161*, 812–820. [CrossRef]
- Shriberg, M. Institutional assessment tools for sustainability in higher education: Strengths, weaknesses, and implications for practice and theory. *High. Educ. Policy* 2002, 15, 153–167. [CrossRef]
- Yarime, M.; Tanaka, Y. The Issues and Methodologies in Sustainability Assessment Tools for Higher Education Institutions: A Review of Recent Trends and Future Challenges. J. Educ. Sustain. Dev. 2012, 6, 63–77. [CrossRef]
- Caeiro, S.; Filho, W.L.; Jabbour, C.; Azeiteiro, U.M. Sustainability Assessment Tools in Higher Education Institutions. Mapping Trends and Good Practices Around the World; Springer International Publishing: Cham, Switzerland, 2013.
- 32. Fischer, D.; Jenssen, S.; Tappeser, V. Getting an empirical hold of the sustaina y sustainability assessment tools. *Assess. Eval. High. Educ.* 2015, 40, 785–800. [CrossRef]
- Rammel, C.; Velazquez, L.; Mader, C. Sustainability assessment in higher education institutions: What and how? In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge: London, UK, 2016; pp. 331–346.

- Bond, A.J.; Morrison-Saunders, A. Re-evaluating Sustainability Assessment: Aligning the vision and the practice. *Environ. Impact Assess. Rev.* 2011, 31, 1–7. [CrossRef]
- 35. HOCH-N Sustainability at Higher Education Institutions: Develop—Network—Report. Available online: https://hoch-n.org/en (accessed on 3 February 2020).
- Morse, J.M. Designing funded qualitative research. In *Handbook of Qualitative Research*, 4. Aufl; Denzin, N.K., Ed.; SAGE: Thousand, OK, USA, 1994; pp. 220–235.
- 37. Kuckartz, U. Qualitative Text Analysis. A Guide to Methods, Practice & Using Software; SAGE: London, UK, 2014.
- Bormann, I.; Rieckmann, M.; Bauer, M.; Kummer, B.; Niedlich, S. Sustainability Governance at Higher Education Institutions (beta version). BMBF Project "Sustainability Governance at Universities: Develop—Network—Report (HOCHN)". 2019. Available online: https://www.hochn.uni-hamburg.de/3aktuelles/nachrichten/135-leitfaden-governance-en/hoch-n-guide-sustainability-governance-at-highereducation-institutions.pdf (accessed on 26 February 2020).
- 39. Williams, B.; Hummelbrunner, R. Systems Concepts in Action. A Practitioner's Toolkit; Stanford University Press: Stanford, CA, USA, 2010.
- 40. Bauer, M.; Bormann, I.; Kummer, B.; Niedlich, S.; Rieckmann, M. Sustainability Governance at Universities: Using a Governance Equalizer as a Research Heuristic. *High. Educ. Policy* **2018**, *31*, 491–511. [CrossRef]
- 41. Hussain, T.; Eskildsen, J.; Edgeman, R.; Ismail, M.; Shoukry, A.M.; Gani, S. Imperatives of Sustainable University Excellence: A Conceptual Framework. *Sustainability* **2019**, *11*, 5242. [CrossRef]
- 42. Edgeman, R.; Eskildsen, J. Modeling and Assessing Sustainable Enterprise Excellence. *Bus. Strat. Environ.* 2014, 23, 173–187. [CrossRef]
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, 171, 1–18. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Article

Transfer for Sustainable Development at Higher Education Institutions—Untapped Potential for Education for Sustainable Development and for **Societal Transformation**

Benjamin Nölting *, Heike Molitor, Julian Reimann [†], Jan-Hendrik Skroblin ^{‡,§} and Nadine Dembski

Faculty of Landscape Management and Nature Conservation, Eberswalde University for Sustainable Development, Schicklerstr. 5, 16225 Eberswalde, Germany; heike.molitor@hnee.de (H.M.); julian.reimann@netzwerk-n.org (J.R.); jan-hendrik.skroblin@posteo.de (J.-H.S.); nadine.dembski@hnee.de (N.D.)

* Correspondence: benjamin.noelting@hnee.de

- + Current address: netzwerk n e.V., Berlin, Germany.
- ‡ Current address: Coordinator for development policy of the district office of Spandau, Berlin, Germany.

§ The views expressed in this paper are this author's personal views.

Received: 17 February 2020; Accepted: 2 April 2020; Published: 7 April 2020

Abstract: Higher education institutions (HEIs) are increasingly confronted with societal needs beyond research and teaching. These include sustainable development and technology transfer as well as the practical application of knowledge and ideas. Several HEIs already put sustainable development and transfer into practice. These practitioner-university partnerships comprise a broad range of actors, disciplines, topics, and formats. However, transfer activities that contribute to sustainable development in society still make up only a very small part of HEIs' activities. In response to calls from society as a whole, HEIs could combine transfer and sustainable development more systematically. In this article, we suggest a concept of transfer for sustainable development. The focus is on sustainability transfer in teaching. We used mixed methods for this conceptual work: exploratory workshops, expert interviews, and a case study of transfer in teaching. One of the results presented in this article is a working definition of sustainability transfer at HEIs. In addition, six characteristics for describing sustainability transfer in its various forms are formulated. This conceptualization makes it possible to analyze the diversity of HEIs' sustainability transfer activities, it helps to identify and encourage potential transfer actors at HEIs as well as practitioners, and, thus, tap the full potential of sustainability transfer.

Keywords: sustainable development; higher education institutions; university; transfer; practitioner-university partnership; societal impact; education for sustainable development

1. Introduction: Linking Transfer and Sustainability at Higher Education Institutions

Higher education institutions (HEIs) or universities (which include universities of applied sciences), can make important contributions to sustainable development if they structure their teaching, research, and operational tasks to be sustainable. In keeping with a whole institution approach, the initial focus is on the internal relationships within the organization. This article broadens the viewpoint as it addresses the external relationships of universities, i.e., their interactions with actors outside the university system. The latter are referred to hereinafter as practitioners.

This view from the outside is in line with a trend in recent years that societal actors are placing new demands on universities in Germany that go beyond the core tasks of teaching and research. The intention is therefore to intensify and improve the transfer of knowledge, ideas, and technologies



to society. The goal is to enable knowledge developed at universities to be applied more directly and more rapidly in practice, thereby satisfying mainly specific societal needs. This is also discussed under keywords such as societal impact or third mission [1,2]. Furthermore, universities are called on to become active in the area of sustainable development and to research, teach, and become more sustainable themselves to serve the common good [3].

This article links the areas of transfer and sustainable development focusing on transfer for sustainable development—in short, sustainability transfer. This is seen as having great potential for development because solving sustainability problems requires cooperation between actors from very different sub-systems in society. Universities can contribute theoretical and empirical knowledge, methodological competence, criticism, and reflectivity to achieving this goal. This is associated with a claim by universities to contribute to shaping sustainable development in society through cooperation between practitioners and universities, always understood as an extension and differentiation of the university system, not as a normative guideline for academia [4]. Many stakeholders in universities propose a role change in this process. While they are aligned with the academic system in terms of teaching and research, in the case of sustainability transfer they are dependent on cooperation with practitioners and are one of many societal actors [5]. They operate in a field that is determined by societal problems and demands and where they are confronted with "*real life*". This necessitates an inter and transdisciplinary approach because specialized disciplines cannot sufficiently address the complexity of social-ecological systems. The challenge to scientific thinking that this involves can provide impetus for sustainable development, but there are certain prerequisites.

A broad range of sustainability transfer activities can be observed at HEIs, such as in research and teaching in two cases in Germany: the Leuphana University Lüneburg implemented the *Innovation Incubator* (2009–2015), a research driven project for regional development where about 250 researchers implemented 45 regional projects (https://www.leuphana.de/kooperationen/regional.html); the Eberswalde University for Sustainable Development has established the *InnoForum Ökolandbau Brandenburg*, a regional network of about 80 organic farmers and businesses along the entire organic supply chain, allowing for on farm teaching in the universities' organic farming programs (http://innoforum-brandenburg.de). Other examples include the *School of Sustainability* at the Arizona State University that links academics with practitioners in the community who are implementing sustainability solutions (https://schoolofsustainability.asu.edu/about/school-of-sustainability/) and the *Community Engaged Learning* program of the University of Toronto where students work for 5–7 hours per week with a non-profit organization or social enterprise specialized in front-line service provision, community-based research, community development, or educational work (http://www.newcollege. utoronto.ca/academics/new-college-academic-programs/community-engaged-learning/).

These inspiring examples raise the question of how to grasp the diversity of such practitioner–university partnerships. It is helpful to understand how sustainability transfer works, what universities are already doing, and where development potential exists, but also where there are limitations. How can such cooperation be described, analyzed, and, if necessary, further developed? This paper therefore aims to present an initial conceptual description of sustainability transfer for universities in Germany. This will enable relevant activities at universities to be identified, made visible, and analyzed in detail.

In view of the diversity of sustainability transfer [6], the focus here is on sustainability transfer in teaching so that the conceptual ideas can be tied to a specific area of focus. Several case studies of transfer in teaching point to an enormous potential for concrete, low-threshold contributions to sustainability with students as transfer actors [7–10]. However, such teaching approaches have so far received little attention in HEIs as a whole and their potential remains untapped. Against this background, the article examines the question: How can sustainability transfer at universities be described in conceptual terms using the focal area of teaching as an example?

To answer this question, Section 2 looks at the position of transfer and sustainable development in university practice. The methodological process used for the qualitative empirical investigation is then

outlined (Section 3). Section 4 presents the empirical results, which are used to derive a conceptual description of sustainability transfer (Section 5). Finally, this draft concept is discussed in the context of its implementation in teaching and the whole institution approach (Section 6).

2. Background of Sustainability Transfer—Positioning in University Discourse and Practices

The conceptual description of sustainability transfer is meant to tie in with the practices and discourse of universities with a view to making it compatible with day-to-day university operations. Consequently, the focal areas of transfer and sustainability are explored in the context of (a) discourse on university policy, (b) practical implementation, and (c) scientific approaches relevant to the actors involved. The idea of sustainability transfer is positioned against this background.

2.1. Transfer as a Focal Area

Transfer has gained in importance in *higher education policy discourse*. The German Council for Science and Humanities stresses the practical orientation of transfer, whereby scientific knowledge is to be applied "*as broadly as possible*" through cooperation [11] (p. 35). The German Federal Ministry of Education and Research (BMBF) is also attaching increasing importance to transfer in this sense [12]. An important role is played here by the "*Innovative University*" funding initiative to promote a research-based transfer of ideas, knowledge, and technology from universities of applied sciences, and small and medium-sized universities. Since 2018, 48 universities have received funding in 29 transfer and innovation projects. State ministries also want to improve cooperation between academia and the private sector, policymakers, and civil society; for example, the State of Brandenburg with its transfer strategy [13].

Many individual actors are involved in *practical implementation*, they pursue the most diverse facets of transfer, with a wide range of different partners from practice. Just as diverse are the forms of implementation, such as technology and knowledge transfer, further training, consulting, participation in social and cultural life, participation in policymaking, science communication, contracts with companies, public agencies, and municipalities, etc. [14] (p. 13). At universities of applied sciences, transfer has a long tradition, especially with companies.

Many universities have established transfer offices or transfer centers to provide targeted support for these kinds of activities. The transfer audit of the *Stifterverband für die Deutsche Wissenschaft*, a German organization that seeks to address challenges in science, research, and higher education, in which 33 German universities participated between 2015 and 2018, provided stimulus for discussion of the issue. In the meantime, several universities have defined their own transfer strategies, which were prompted, among other things, by the transfer audit and the *"Innovative University"* funding initiative, which made a transfer strategy a prerequisite for submitting applications.

The understanding of transfer has changed. This is relevant for the *conceptualization* of transfer of universities. The traditional understanding is based on technology transfer, which involves transferring scientific and technical findings from research to companies for use in the production process [14]. The term has been expanded to include knowledge and research transfer, which means the transfer of research findings from all scientific disciplines for use in companies, but also in municipalities, public administration, and civil society actors, including policy consultation. Regional economic clusters and innovation systems around universities and research institutions are examples of this definition of transfer [15]. Against this background, Roessler et al. [14] characterize transfer as reciprocal relationships in which the activities of universities have a direct impact on society and the economy and vice versa, as trends from the private sector and society are reflected in the university [14] (p. 39).

Even broader is the concept of the third mission. Alongside teaching and research, this "third mission" is defined by higher education institutions as interaction with external actors outside the university, which relates to societal needs and which cannot be met by conventional teaching and research alone, but is at least loosely associated with them [1] (p. 18). The third mission includes involvement in society (including cultural, social, and environmental activities), technology and knowledge transfer (including science communication, policy advice), and further education [1,2].
There is a big overlap with similar concepts like the social responsibility of universities, entrepreneurial university, societal collaboration, service learning, living labs, etc. [6]. Trencher et al. criticize that the focus of the third mission is often narrowed down to economic contributions, namely through technology transfer. For this reason, they suggest as a fourth mission the co-creation of sustainability, characterized by collaboration with diverse social actors, to create societal transformation in a specific location, region, or societal sub-sector and based on the values of sustainable development. This fourth mission is more than a mere offshoot of the third mission—it is a differing but compatible mission on its own [6].

2.2. Sustainable Development as a Focal Area—with the Focus on Teaching and Education for Sustainable Development

The *higher education policy discourse* on **sustainable development** is shaped by the United Nations Sustainable Development Goals (SDGs) and the UNESCO World Programme of Action on Education for Sustainable Development (ESD) (2015–2019), which is specified in more detail in the National Action Plan for the implementation of the World Programme of Action for Germany [16] and the recent UNESCO framework "Education for Sustainable Development: Towards achieving the SDGs" (2020–2030) [17].

The BMBF has been supporting sustainability research for about 20 years now, with funding priorities including socio-ecological research, the Research for Sustainability program (FONA), and the Sustainability in Science Initiative (SISI). In this context, the BMBF specifically supports sustainability activities of research organizations (*LENA project*), students (*netzwerk n*), and universities through the collaborative project Sustainability at Higher Education Institutions: HOCH^N, around which a network of sustainability actors from more than 100 German universities has formed in the meantime. As the representative body of higher education institutions in Germany, the German Rectors' Conference (HRK) 2018 declared its commitment to firmly enshrine the goal of sustainability at German universities along with a culture of sustainability [3].

Given the many definitions of sustainability, one challenge in the *practical implementation* of sustainability is to clarify the underlying understanding of the term, to incorporate it into university operations, and make it explicit so that it is accessible for scientific analysis and criticism. Possible starting points could be the guiding sustainability principles of individual universities or a university-wide understanding of sustainability, such as the one developed in the collaborative HOCH^N project [18]. The same applies to the reflective framework for research in social responsibility of non-university research institutes [19]. Another conceivable point of reference are sustainability concepts legitimized through established policies, such as the 17 SDGs of the United Nations or the German Sustainability Strategy.

The central areas of focus for implementing sustainability at HEIs are teaching, research, and transfer [6], as well as the universal themes of governance, operations, and reporting, which are the subject of research at HOCH^N. Just as in the area of transfer, there are many, very diverse initiatives at universities, making an overview difficult.

Since teaching in particular is viewed here as a key area for the practical application of sustainability transfer, ESD will be examined as a central concept. ESD is an international, value-based concept that plays a special role in the process of implementing sustainable development [17,20]. At universities, this concept represents an opportunity for sustainability transfer in teaching, as shown in Figure 1, because "socially responsible science means that HEIs provide the orientation knowledge necessary for social transformation" [21] (p. 51). Sustainability transfer in teaching should work towards sustainable development through ESD. This entails content-related and methodological-didactic requirements for universities, such as dealing with real-life, job, or training-related sustainability problems [22]. Teaching formats, such as projects related to the degree program (learning in real-world situations), are suitable for this purpose. This is where the life and experience of the learners comes into play and

where actual sustainability problems from the real world are addressed or solved with the university actors as equal partners.



Figure 1. Positioning of sustainability transfer in the context of higher education institutions.

These learning experiences are therefore of great importance, as they are acquired in practice (real social contexts) and can be studied or reflected upon in a theoretical framework. This means that the place of learning is not just the university, but also the public space or the working and living environment of the practice partners and the issues they work on. This format of project-based learning allows students a high degree of self-determination (and thus results in a high degree of participation).

Since the world can be described as a complex, interconnected ecological, psychological, societal, and economic system, this requires a systemic approach from multiple perspectives. Many issues and complex themes, such as biodiversity, climate change, poverty alleviation, health, natural resource management, production, and consumption, are closely interwoven and are reflected in the 17 SDGs [23–25].

An important element of ESD is competence-based teaching, which is particularly relevant to sustainability transfer processes [20,26,27]. The following competencies are addressed by sustainability transfer processes:

- Systemic and forward-looking thinking because topics relevant to sustainability in university teaching are complex, often ambiguous, and in some cases, full of uncertainty; many things are interlinked and systemically interrelated;
- Ability to cooperate, negotiate, and participate, because sustainability transfer in teaching requires
 participation in the discussion of sustainable development issues and a shared decision-making process;
- Ability to reflect on personal values and those of others in the context of sustainable development and on one's own role in local communities and (as citizens of the world) the global context, since issues and problems relevant to sustainability do not stop at national borders;
- Empathy, which includes mutual understanding and respect for the needs, ideas, and activities of others.

In summary, the aim of ESD in higher education is to enable students to engage with the major changes of our time from a scientific point of view (understand), to assess the resulting consequences from the perspective of sustainability (evaluate), to rethink the relationships between human beings and nature (change), and to assume responsibility to be able to actively participate in society's transformation toward sustainability [28].

In research, a transdisciplinary sustainability approach is well suited to the topic of transfer [29], as shown in Figure 1. The core element is transdisciplinary exchange with practitioners, where heterogeneous groups of actors shape a common process of learning, impact, and reflection [30,31]. Building on this, transformative sustainability research aims to support transformation processes in the direction of sustainability through innovation and to promote the process of change in society through the scientific development of concrete solutions and innovations in industry and society [32,33]. One interesting approach to transdisciplinary research with a high level of transfer relevance are real-world laboratories where HEIs, municipalities, NGOs, companies, government institutions, associations, etc. work together to find sustainable solutions. Under the guiding principle of sustainable development, they join forces as pioneers of change in real-world laboratories and conduct experiments to develop, test, and research new ideas [34].

2.3. Interim Conclusion

Both the overview of practices and discourse at universities and an evaluation of academic literature show that an understanding of sustainability transfer can draw on both HEI concepts of transfer and sustainable development. Transfer can be described as a partnership between practitioners and universities exchanging technologies, knowledge, and ideas in the focal areas of teaching, research, and a third mission. When the protagonists relate their transfer activities to sustainable development, it can be termed as sustainability transfer. There are even specific approaches that are in line with sustainability transfer in teaching, namely in ESD, and in research, especially transdisciplinary sustainability research. Reliable, as well as innovative, approaches are problem and project-based learning, service learning [8], research-based learning, as well as transdisciplinary research projects (such as action research or real-world laboratories [34]).

Based on this initial evidence, it seems possible to link the university themes of transfer and sustainable development, as shown in Figure 1.

Even though there are interesting and inspiring cases, sustainability transfer has so far hardly been explicitly addressed by HEIs and does not play a significant role in university governance in Germany. Considering the enormous potential for HEIs to contribute to sustainable development through direct interaction with practitioners, it would seem to be a promising endeavor to systemize and conceptualize the notion of sustainability transfer for analyzing the broad range of activities to attain a better understanding of transfer mechanisms and their impacts in the university setting. Such a science-based concept can be used as a reference by the actors involved for further developing their transfer activities.

3. Materials and Methods

The empirical study for the development of a sustainability transfer concept was broken down into three steps: first, an exploration through expert discussions and workshops with actors working in the areas of transfer and sustainability; second, an empirical survey of experts on transfer; and third, an empirical study on the implementation of sustainability transfer in teaching to identify a concrete teaching format.

In line with a heuristic approach, the first step was to develop a working definition and to test its acceptance in workshops. An initial working definition and elements of sustainability transfer were outlined and presented in four workshops in 2017. These workshops specifically addressed transfer actors, including actors from all university levels as well as practitioners from the private sector and public administration. Participants in these workshops therefore included actors from university administration, students, students' self-administration organizations, research and teaching staff, as well as representatives from the private sector and public administration. Established transfer activities, results, and proceedings of implemented projects and potential new formats were discussed. Workshops therefore provided an overview of different existing activities with a regional impact and analyzed to what extent these projects can already be considered to be transfer activities and how they can be developed further. Cooperation with the student association *netzwerk n* was helpful in this process [35]. With the help of examples, aspects of sustainability transfer were explored in greater depth, terms were clarified, and the working definition fine-tuned.

In the second step, categories for the analysis of expert interviews were identified based on this exploratory study (Step 1) and further reading. In addition, questions were asked about the focal areas of teaching, research, and the third mission, although only teaching will be addressed below.

Interviews were conducted with seven transfer experts in April and May 2018. The aim of the interviews was to examine and validate the initial conceptual ideas and to position sustainability transfer in the context of academia and HEIs. The experts were selected on the basis of their publications and the focus of their work with different university and non-university based backgrounds. Two of them manage university transfer offices and four work in the area of HEI research and consultation either at university institutes or in foundations, thus providing an overview of transfer activities inside and outside of universities. One expert works in transdisciplinary sustainability research and teaching. They cover a wide range of expertise and research inside the university in the field of transfer and third mission. The aim was to describe sustainability transfer from different perspectives.

The methodological approach to the guideline-based expert interviews and the qualitative content analysis is based on Gläser and Laudel [36]. The interviews were exploratory, and some were conducted as conversations lasting about one hour. They were transcribed using the MAXQDA tool and analyzed using a category system. The categories were initially developed deductively on the basis of theory and the explorative workshops and then supplemented inductively during the course of the evaluation, resulting in main categories such as basic understanding of transfer, actors, interaction and formats, framework conditions, third mission, and transfer in relation to sustainability. The relevant statements were extracted from the text by means of coding and redundancies were then eliminated. The material was merged in evaluation documents under the main categories to render the interpretation steps transparent. Characteristics and description categories for sustainability transfer were identified on this basis.

In the third step, these key aspects were applied to a case study of sustainability transfer in teaching in order to test and validate the analytical categories. Step three was designed to gain concrete empirical insight into the implementation of sustainability transfer, not as best practice but as a regular case study with strengths and weaknesses. Sustainability transfer was analyzed in a concrete teaching format as part of a master's thesis in 2019. The subject was the module "*Projektarbeit und ganzheitliche Projektgestaltung*" (project work and holistic project design) in the master's degree program Regional Development and Nature Conservation (M.Sc.) at the Eberswalde University for Sustainable Development [37]. In this module, working groups of four to six students design and implement a regional development project together with a practice partner. It is a compulsory module with 12 credits of the European Credit Transfer System.

The following research question was pursued in the master's thesis: How can students, as participants in sustainability transfer, shape sustainable regional development? In total, reports of 17 student projects from the winter semesters 2016/17 and 2017/18 were evaluated, and six of the reports built the data as detailed case studies. These reports consist of (a) a content-related project report and (b) a presentation of the process and project management. Twelve documents from the case studies were thus evaluated using qualitative content analysis. The following criteria, among others, were considered: forms of interaction, methods for structuring interaction, role of students, factors influencing the cooperation process, types of results, and links to sustainability.

In addition, six exploratory interviews were conducted with the practice partners in the six case studies in order to assess the learning and impact potential of the projects on the part of the practice partners.

4. Results for the Description of Sustainability Transfer

This section presents the results of the workshops held to explore and refine the investigative categories for the expert interviews. The results of the interviews and of the in-depth study on the implementation of sustainability transfer in teaching are also presented.

4.1. Exploring the Characteristics of Sustainability Transfer for Investigative Categories

Starting from the approximate understanding of the transfer and sustainability activities of HEIs, as shown in Figure 1, the discussion in the explorative workshops highlighted the following aspects that needed further consideration.

In one workshop with researchers, lecturers, and transfer office staff, a first draft of a description of sustainability transfer was presented and discussed. This basic understanding included the content, goals, and success criteria of transfer. A difficult question was the distinction between transfer and the third mission of universities. Another workshop with practitioners from business and public administration explored the transfer actors. This included a broad range of diverse groups of actors on the practitioner side from large companies to individual citizens, their roles, motivation, and interest in practitioner–university partnerships. In addition, a large variety of forms of interaction and specific cooperation formats in teaching, research, and a third mission were described. Two workshops with students and lecturers were dedicated to transfer in higher education. The discussion explored the interests of students and lecturers in transfer (relating also actors and forms of interaction).

Framework condition was a topic in all four workshops. Potential, restrictions, and preconditions of transfer were discussed, focusing on the general conditions of transfer at HEIs with regard to structures, organizational conditions, relevance for university development, transfer strategies of HEIs, and organizational restrictions. A crucial topic dealt with in all four workshops was how to generally differentiate transfer from transfer for sustainable development. The discussion touched on possible definitions of sustainability and the understanding of sustainability in practitioner–university partnerships, different sustainability goals for transfer activities, the sustainability impact of transfer, varying levels of ambition in terms of sustainable development.

The following investigative categories for the expert interviews were identified in the four workshops:

- Basic understanding of transfer including a differentiation between transfer and third mission.
- Actors and forms of interaction.
- General conditions of transfer.
- Sustainability in transfer.

4.2. Characteristics of Sustainability Transfer

The interviews with the transfer experts were evaluated using the above-mentioned research categories.

4.2.1. Basic Understanding of Transfer and how it Differs from the Third Mission

Several experts adopted a broad understanding of transfer comprising (nearly) all activities of HEIs related to society and non-academic actors (Experts 1, 2, 6). Transfer activities range from intellectual property, and business start-ups to university for children or exhibitions. The concept of transfer has meanwhile been broadened from the traditional understanding of transfer. It started with technology transfer in the 1990s incorporating more and more aspects, such as research and technology transfer, later knowledge transfer, and recently just transfer (Expert 6). Especially the transfer office staff that were interviewed think that quite a lot of transfer activities are taking place at HEIs (Experts 2, 6).

Instead of a one-way transfer from the university to practitioners, the experts emphasize mutual exchange and dialogue as a central feature of transfer, which may encourage the transfer concept (3, 4, 5). As a result, not only technologies, knowledge, and ideas are transferred, but concepts, views, and values are also exchanged in more integrated forms of interaction (Expert 1). Nevertheless, transfer may also include unidirectional activities and formats like public lectures and popular scientific publications (Expert 7). Even though there is a broad variety of transfer activities, not all university members are required to engage in transfer activities. Whether transfer is useful and productive depends on the topic, the disciplines, and the actors (Expert 3).

Universities produce knowledge in research and teaching that can be rendered useful to society through transdisciplinary formats in research and teaching. In these debates, universities can argue that the knowledge they produce is useful to society. In general, all members of HEIs can become transfer actors (Expert 2). To date, universities and transfer offices have focused on transfer linked with research (Experts 2, 6). Transfer in teaching plays a less important role and is not specifically addressed by a transfer office (Expert 2). However, a clear distinction between research and a third mission or teaching and a third mission in relation to transfer is needed (Expert 7).

Transfer activities are often motivated by researchers who are interested in applying their results and putting them into practice. As a result, they contact transfer offices of HEIs (Expert 6). Universities are interested in bringing new products and services that were developed in their organization to the market. Moreover, it is in the university's interest to have these products and services contribute to sustainable development (Expert 2).

In a broader sense, universities' efforts to support social change and transfer can be seen as a means to this end (Expert 2). They have the potential to create social and technical innovations linked to environmental goals by providing arenas for societal discourse and experimentation (Expert 3). In transdisciplinary sustainability research, e.g., in real-world laboratories, research goals are often adopted from practitioners. This is an example of how practice may shape transfer and its aims (Expert 7).

Experts suggest making the goals of transfer transparent because HEIs' activities are financed by public money and they have a responsibility to society (Experts 1, 2). Transparency is thus seen as one success factor (Expert 1). However, concrete goals for transfer activities may also constrict and reduce the scope of action for the transfer partners. For this reason, it may be wise not to overburden the goals of transfer activities but to maintain sensitivity to the actors, the process of cooperation, and the blind spots of academia (Expert 3).

Confronted with a broad range of goals, actors, and topics of transfer, the experts interviewed called for a differentiation of transfer activities. For example, they differentiate between the duration and degree of the institutionalization of cooperation as well as the trust between transfer partners (Expert 1). In general, it seems useful to distinguish between the degrees of complexity of transfer in order to internally analyze the organization's own transfer activities. However, these degrees of complexity do not imply that one form of transfer is rated *"better"* or *"worse"* than another (Expert 1).

When asked about the potential conceptual differences between a third mission and transfer, the transfer experts interviewed admit that there is a significant overlap between the concepts (e.g., Expert 1), nevertheless, they try to differentiate between them. The third mission is seen as an approach to encourage universities to engage with challenges relevant to the region and society as a whole. The third mission aims to motivate the universities to adopt a positive approach to their societal activities, to communicate them more effectively to the outside world, whereas transfer focuses on the central tasks of the university itself, i.e., teaching and research, in order to develop them further by engaging with practice (Experts 4, 5). In line with this thinking, the third mission opens the door for cooperation with practitioners. In addition, it is seen as the broader concept, including the prerequisites and structures, as well as the impacts and consequences of the practitioner–university partnerships, while transfer is restricted solely to the activities of dialogue and providing knowledge (Expert 1). The third mission includes all external relationships of HEIs with society in a broad sense (Expert 6). The term third mission also has a pragmatic aspect. The third mission is easier to link with teaching

and research, while transfer is rooted in the traditional image of a unidirectional dissemination of technologies and knowledge (Experts 4, 5).

However, transfer has many facets, and transfer and the third mission overlap to a large degree, making clear-cut definitions and differentiation difficult (Expert 2). The problem of a clear distinction is not only an academic question, but also a practical question for teaching and research that needs to be resolved for each transfer activity (Experts 4, 5).

4.2.2. Actors and Forms of Interaction

On the HEI side, transfer actors include all members of the university, such as researchers, teachers, students, university management, and administration, e.g., sustainability officers or transfer offices (Experts 1, 2). The non-university partners include companies and economic actors, policymakers, public administration, civil society organizations and initiatives, as well as citizens; in general, all non-academic actors (Expert 2).

HEIs cooperate with companies in different ways. Some companies seek out partnerships, some companies are interested in sharing knowledge, and others have no interest at all. For universities it could be an advantageous strategy to try and acquire companies that are interested, but not yet involved (Expert 2). NGOs are another partner for transfer because they often rely on scientific findings (Expert 3). More sophisticated partnerships include different groups of actors at the same time, including policymakers, entrepreneurs, and civil society groups (Expert 6). The most important transfer partner are still companies, but NGOs and social organizations have increased in number and awareness over the last years. Citizens and the public are not yet represented in practitioner–university partnerships (Expert 2).

The different transfer forms and formats of interactions enable a range of possibilities for cooperation in teaching, research, and the third mission. The form of cooperation between the participants is characterized by the fact that the transfer activities take place in a societal context in which transfer is collaboratively shaped for the mutual benefit of the transfer partners. The topics and contents transferred are determined by the interests of the transfer partners and their scientific and/or practical needs (Expert 1). In this context, there is a scientific link with at least a loose correlation to teaching and research (Experts 4, 5).

Experts suggest differentiating between forms of interaction, for example, between short-term and long-term cooperation or direct and indirect forms of interaction, such as exhibitions or leaflets without face-to-face contact (Expert 1). Further, the side initiating the transfer often defines the logic of the cooperation. When an HEI shapes the partnership, research (or teaching) logic is predominant, but when practitioners are in charge, the partnership is instead geared towards practical outcomes (Expert 7).

The assessments of the transfer experts differed on this point in the interviews. While some considered reciprocity to be a fundamental prerequisite for transfer, in order to identify problems, ascertain expectations, and be able to conduct a process of reflection involving all transfer partners, others emphasized the transfer form of contract research, in which reciprocity plays a lesser role because this transfer is unidirectional (Expert 6). Reciprocal exchange helps to identify the interests and needs of the transfer partner and feedback loops help to adjust mutual expectations and foster learning processes (Experts 2, 6). This encourages multiple perspectives on a problem and the incorporation of different forms of knowledge and expertise (Expert 3), thus stimulating innovation (Expert 1).

A partnership on equal footing accepts different forms of expertise and values different contributions to the transfer cooperation. Mutual respect is a prerequisite for building trust and more complex forms of joint knowledge production (Experts 1, 2, 6). This is a crucial element of transdisciplinary research (Expert 7). However, not every form of transfer is necessarily a reciprocal exchange on equal terms; the formats may vary according to the actors and topics (Expert 7).

4.2.3. General Conditions for Transfer

During the expert interviews it was stressed that the success of the transfer also depends on general framework conditions. A distinction must be made here between the overarching structural conditions which are determined outside of the HEI, e.g., by laws and financing, and the organizational conditions for implementation, which can be shaped by the HEI itself.

University management has a certain amount of freedom to influence the conditions for implementation; it can create structures that coordinate and support transfer activities (Experts 2, 3). Transfer offices in particular play a key role in this respect. They can reliably assume recurring tasks, pool and process knowledge, and establish and maintain contact between potential transfer partners. Transfer offices are thus key points of contact for actors interested in transfer from the university and from practice and are important in establishing contacts and formalizing transfer activities. Last but not least, they can support the transfer partners with suitable methods, formats, and communication (Experts 1, 2, 3, 6). University management and transfer offices may create a culture of facilitation (Experts 1, 7). In addition, some interviewees suggested setting up incentive systems for transfer and creating scope for developing transfer projects and concepts in day-to-day operations alongside teaching and research tasks (Experts 1, 2, 7).

This in turn relates to the overall structural conditions. Policymakers can, for example, provide universities with targeted resources for transfer and transfer offices, establish transfer-friendly funding formats, or raise the standing of transfer (for sustainable development) at universities, so that commitment in this area is also rewarded by additional funds, career opportunities, and reputation for the university (Expert 2).

4.2.4. Sustainability and Sustainability Transfer

Generally, the sustainability aspect of, or even impact on, transfer activities is difficult to assess due to the vagueness of the term sustainability. For this reason, it is important to create our own specific definition of sustainable development and/or refer to a widely accepted definition (Expert 7). Sustainability transfer may, for example, aim to bring about sustainable development in a certain region or municipality (Expert 1). Policymakers call on researchers to define the concept of sustainability more precisely so that it can be implemented through political measures (Expert 6).

The interviewees considered it important here to assess the contribution of sustainability transfer and thus make it tangible. This would also be helpful in assessing the impact of the projects and raising awareness of the effects of the transfer activities (Experts 1, 2, 4, 5). An impact analysis is thus an important feature of sustainability transfer, which can be used on the one hand to assess the impact of transfer projects at an activity and message level and, on the other hand, to enhance the reflection process among those involved in the transfer by creating awareness of their own actions and their impact.

Experts interviewed consider it possible to link transfer explicitly with the concept of sustainability (Expert 6). However, they were also critical of the use of the term sustainability transfer because it restricts transfer activities to a very specific field of transfer that only applies to a small part of HEIs' transfer activities (Expert 1). This specific idea of transfer activities narrows the scope for transfer actors (Expert 2). Some recommended using only the concept of transfer and adopting a multi-pronged approach, with sustainability research being one possible variant (Expert 6).

4.3. Case Study of Sustainability Transfer in Teaching

In the case studies investigated on project work in sustainable regional development, the students first begin by conducting a problem analysis with the involvement of the respective practice partners and thereby established links to sustainability in their projects. When clarifying the transfer task in the conceptual development phase, goals are formulated on the basis of the problem analysis together with practice partners and other relevant project actors, who focus on a contribution to sustainable development in the regional focal area of the practice partners. The students thus first establish a link

to sustainability in the transfer project. Second, they agree on concrete and transparent goals with the practice partners to promote sustainable development.

One quality characteristic of the students' activities in the projects is the active involvement of the practice partners in determining the technical and content-based focus of the project, identifying mutual expectations, and weighing up the options for action, taking into account the general conditions of the module.

The involvement of the practice partners goes beyond clarification of the task and continues throughout the product development process. The intensity of the relationship becomes clear in the process. The students engage in a continuous feedback loop with the practice partners to get feedback for decisions that show the way forward and to incorporate new findings into the clarification of the task and thus adapt the goals. The core element of participation is often workshops that bring together all relevant actors. This is where different requirements are discussed, topics identified, and solution strategies developed. This co-production is particularly highlighted by the practice partners, because the students create a strong impetus that has a transformative effect on the actors' environment in the process. This participation and its impact are mentioned more often as the success of the cooperation project than the actual end product.

The students' perception of their role in the project group and their methodological competence are of particular importance for the feedback and adaptation processes and for the active involvement of relevant actors. The students see themselves as a learning organization in the project groups, which makes them flexible and willing to learn in their interaction with the practice partners and open to changes to goals. They make targeted use of methods that they have learned during their studies in the process. For example, an environment analysis is carried out during project implementation in order to identify and involve relevant actors. The workshop structure is based on facilitation techniques.

The projects in the case studies have a wide range of objectives. They often have a structural and strategic focus—structurally, to give the practice partners concrete options for action so that, for example, decision-making processes in administrative workflows can be improved, and strategically, in order, for example, to identify concrete development opportunities for an organization as part of a socio-ecological transformation.

The concrete end products of student projects are often concepts and tools that usually cannot be implemented or applied with the practice partners during the project. However, they were used after the projects were over; for example, for project applications, to develop guiding principles and improve decision-making processes. The surveyed practice partners described the quality of the end products as innovative and creative, which inspired new ideas. One special learning effect for the students was the opportunity to apply the knowledge and methods acquired through the project-based form of learning. The practitioners benefited from this, as they were able to acquire new knowledge and methods and, in some cases, to develop new attitudes and viewpoints by working together with the students and to incorporate these into their workflows.

5. Evaluation: Conceptual Description of Sustainability Transfer in Relation to Teaching

A concept for sustainability transfer was developed against this empirical background and in line with the research literature. First, a definition of sustainability transfer is presented and then six analytical descriptors are outlined.

Sustainability transfer is defined as a specific form of transfer, namely those practitioner–university partnerships that contribute to or strive for sustainable development in society. A key characteristic is a mutual exchange between university and practitioners, because no group of actors can implement sustainability on its own. However, the degree of complexity and intensity of such practitioner–university partnerships may vary from a more unidirectional transfer (of technologies) to a co-production of knowledge and practical solutions (cf. Section 5.6.). Academia can contribute theoretical and empirical knowledge, methodological competence, criticism, and the ability to reflect. The goals of sustainability transfer are first to make concrete contributions to sustainable development,

such as projects, concepts, or discussions about sustainability. At a minimum, transfer partners should formulate explicit sustainability goals for their respective transfer activities. Further, they should strive to make their societal impact transparent and minimize negative effects as far as possible. Second, it is important to strengthen the capacity of the transfer partners to act. Through interaction, the actors can learn from each other and develop key competencies for sustainability, such as the ability to innovate, systemic thinking, foresight, and strategic and normative competence [38].

Although this definition creates a framework for sustainability transfer, the corresponding practitioner–university partnerships can still cover a wide range [6]. Different transfer activities can be analyzed using the following descriptive characteristics.

5.1. Sustainability Focus of the Transfer Activity

The classification and evaluation of the sustainability focus of transfer, understood as a contribution to sustainable development, is a central requirement for defining sustainability transfer and avoiding arbitrary use of the term. However, there are currently no generally recognized and reliable measurement and evaluation methods that can be used to clearly capture the sustainability impact of academic activities and especially of sustainability transfer [39]. This is why we propose three approaches to the issue of sustainability focus.

First, it can be verified whether or not the transfer partners formulate *sustainability goals* for the respective transfer activity. These goals, which are in the public interest, are normative statements that are not based on scientific research. Nevertheless, scientific arguments can be made, for example, with reference to transdisciplinary sustainability research or ESD. There are two possible approaches to developing sustainability goals. On the one hand, the transfer partners themselves can develop sustainability goals for their specific transfer activity. The reference point can be the transfer actors' understanding of sustainability, which can be seen, for example, in sustainability guidelines, sustainability statements, or societal positioning. On the other hand, externally defined sustainability goals, such as the SDGs, the goals of the German Sustainability Strategy, or goals specific to a certain area, e.g., of the energy transition, can be included. These goals have the advantage that they are accepted by society and have usually been tested from a scientific standpoint. However, they generally have to be incorporated into operations for concrete transfer activities. The discussion of sustainability goals between the transfer partners can culminate in a rationale for the goals, showing the extent to which a contribution to sustainable development can or should be made.

Second, the *sustainability impact* is of great relevance. However, capturing, let alone measuring, these effects is difficult because in complex, real-world situations it is difficult to clearly attribute impacts, because cause–effect relationships are hard to distinguish from other environmental influences and effects can be delayed, shifted regionally and functionally, and can also have (unwanted) side effects [40]. The formulation of sustainability goals is a starting point for capturing the effects, because criteria can be derived from the goals to evaluate the success or failure of transfer activities. For this purpose, various approaches are proposed in the literature, e.g., balancing model [1]; result types [41]; impact forms and effectiveness [39]; impact logic as diagram of a causal model [42].

Third, *minimum and exclusion criteria* for sustainability transfer can be formulated. In the interest of socially responsible research [19], the results of the transfer activities should be freely accessible, especially if the activities carried out by the university are part of a public contract or are financed with public funds (minimum criterion). Furthermore, the transfer activities may not conflict with sustainable development or cause or exacerbate sustainability problems. The SDGs can serve as a point of reference. If the transfer activity endangers the achievement of individual goals, then it is not sustainability transfer (exclusion criterion). In view of the internal inconsistencies of the SDGs, it will be necessary to strike a balance.

5.2. Focal Areas of Sustainability Transfer

The focal areas can be divided into research, teaching, and the third mission. According to the definition of [1], sustainability transfer in the *third mission* refers to those university activities that go beyond the compulsory tasks in teaching and research but are related to them. This then includes all third mission activities that make or aim to make a contribution to sustainable development. In contrast to this, sustainability transfer is explicitly linked with teaching and research here and in line with [5]. The decisive factor is the existence of practitioner–university collaborations in these areas of focus, as shown in Figure 1.

Sustainability transfer in *research* is only briefly described here. It focuses on a joint process of practical application and implementation, which is mutually supported by practitioners and university actors and which is intended to contribute to the solution of specific sustainability problems. This is therefore a sub-aspect of transdisciplinary sustainability research. Specific to sustainability transfer is the link to practical application, i.e., a test, trial, implementation in practice, for which the transfer partners jointly assume responsibility in the form of co-implementation. Since this is a real context, the practical benefit and success of the implementation is extremely important; the university actors cannot pursue their scientific knowledge interests on their own, potentially at the expense of successful implementation [5]. Another benefit for the actors involved are shared evaluations or even a joint reflection process, which can reinforce the learning effects for all participants.

In the transfer of sustainability in *teaching*, practitioners are involved in students' learning processes on sustainability. For example, practitioners report on their work, there are practical tasks, the real world is a place of learning, and students and practice partners work together on sustainability problems. The spectrum ranges from practitioners who come to the lecture hall to the incorporation of teaching in a practical setting.

Sustainability transfer in teaching can be based on the concept of ESD, which supports a clear sustainability focus [23,28]. The incorporation of practical experience strengthens the focus on competencies in particular, which in the case of ESD is aimed at acquiring competencies for influencing the societal impact.

The focus on competencies is promoted by learning in real-world situations because the social context of learning is real and is not a simulation in a lecture hall or seminar. Students shape the world and their environment and learn first-hand that they can make a personal contribution to shaping society, as the empirical study on student sustainability projects shows.

Didactic approaches are problem-based learning, research-based learning, and service learning. Examples are student projects and final theses with transfer partners, dual courses of study, internships, service learning, involvement of transfer partners in teaching, and mentoring and coaching [14].

5.3. Transfer Actors

On the university side, the transfer actors include all university members. The non-university partners include companies and economic actors, policymakers, public administration, civil society organizations and initiatives, as well as citizens. Sustainability transfer is thus shaped by very different actors. Relevant groups of actors for teaching are outlined below.

Transfer in teaching can be unidirectional or reciprocal until the learning process and learning outcomes are reflected on together with the practice partners. Students, teachers, and external partners in practice are involved in the learning process. This can go as far as collaborative efforts to impact society as equal partners. By including practitioners, new teaching–learning contexts are created in which students themselves can become transfer actors [37].

Students put their subject-specific and methodological knowledge to practical use in a real or realistic context and develop an integrated approach to shaping the world they live in.

Practice partners contribute relevant practical problems and gain access to current, research-based knowledge through cooperation. These kinds of teaching partnerships can provide low-threshold access to scientific knowledge, especially for small organizations and initiatives. Practitioners can

benefit from questions and suggestions from students and reflect on their activities from an analytical perspective, which is relevant, for example, for clarifying sustainability goals and priorities.

Teachers create the framework for this learning setting and manage the learning process and the associated communication in the form of learning coaches. They *"translate"* between students and practice. They can improve the quality of their teaching and gain insights into practical sustainability problems which they use as a basis to inspire teaching and research.

5.4. Phases of Sustainability Transfer

Sustainability transfer can ideally be outlined as a four-phase process, whereby considerable fluctuations are possible, and phases can overlap. *"Typical"* tasks and requirements can be allocated to each phase. This makes it possible to get an overview of possible courses of action, facilitates an analysis of the processes, and can show alternatives and options for action to further develop the planning of sustainability transfer. There are four distinct phases [43]:

- Initiation of transfer and identification of actors;
- Concept and goals of the transfer activity;
- Implementation of the transfer activity;
- Compilation and documentation of the results.

In the first phase of initiation, the challenge is to identify the theme and transfer partners for a specific class. The starting point can be problems from practical areas covered by the respective degree programs and modules. In a first meeting, the potential partners must assess the extent to which they agree on the transfer activity they want to undertake. On this basis, they can decide whether cooperation makes sense from their point of view, whether they fit together organizationally, and whether there is enough trust to engage in a joint learning process. Teachers often have the responsibility for this step.

The second phase of conceptual development involves planning the learning process and transfer activities. A suitable way of clarifying the content is to jointly agree on sustainability goals to be achieved in the learning process. Based on shared learning goals, a teaching–learning concept can be developed for which ESD offers a variety of approaches and suggestions for implementation, e.g., case studies, project work, internships, etc. It is important to bear in mind here that the learning processes in sustainability transfer are more open, more difficult for those involved to assess, and often involve a greater (time) investment for students and teachers. Nor can they always be smoothly integrated into curricula and examination regulations.

In the third phase of implementation, the task is to translate analysis, development, or testing of solution approaches into a learning process. The aim is to achieve a learning and development process that is as mutual as possible, in which theory and practice are linked, and scientific methods and concepts are applied. It is important to clarify roles in the various learning and working steps. The roles can change, e.g., students can be both learners and creators of ideas or teachers.

In the fourth phase of results compilation, learning outcomes and experiences are to be documented. Documentation forms the basis for making the sometimes interwoven learning processes transparent. From the teaching point of view, the formal examination is a core element in compiling results. Examination formats based on competencies can be used to compile the students' results in such a way that they can be further used by the practice partners.

5.5. Universal Tasks

Furthermore, two universal tasks are examined which are set as requirements for sustainability transfer across all phases.

The task of *process management* helps to support different groups of actors with different competencies and interests in their cooperation. Organizational structures provide the transfer partners with certainty and the necessary resources. Another task is the internal and external

communication throughout the entire transfer process. It creates transparency of transfer activities and fosters a culture of cooperation. By comparing the tasks in the different phases and the real overall conditions for sustainability transfer, the possibilities, scope for impact, and potential, but also limits, in the implementation of sustainability transfer can be estimated.

As a second task, *reflection* can ensure the sustainability focus of transfer activities. Reflection plays a role in verifying and sharpening the sustainability focus of transfer activities; it helps to identify *"blind spots"* and side effects and to prevent risks. In addition, reflection in all phases can contribute significantly to the quality of the learning and knowledge production. Reflection can be undertaken separately by groups of actors, which is easier to implement. Joint reflection on the transfer process is more complicated but promises more knowledge gain. Through academically guided reflection, common insights and experiences can be identified that go beyond the respective case-related and context-specific results and can lead to transferable insights and experiences. Particularly with regard to the complex challenges of sustainable development, reflection should explicitly incorporate values and emotions. This also applies to learning from mistakes and failures.

5.6. Degrees of Complexity of Sustainability Transfer

Sustainability transfer can be differentiated according to how complex the interaction between university and practitioners is. The differences do not imply any judgement; the levels each have advantages and disadvantages, as shown in Figure 2.



Figure 2. Degrees of complexity of transfer (own diagram).

Provision of knowledge: The basic form of transfer includes all cooperation between the university and external partners. In the foreground is a largely unidirectional transfer of knowledge, technologies, and ideas from university to practice. At its own initiative, the university provides societal actors with findings, knowledge, and products from teaching and research without getting feedback on societal needs and the impact from the practice partners. Such forms of transfer can be carried out with few resources and have a comparatively extensive coverage or reach many people, while the intensity of interaction is low. Examples are exhibitions, popular science and transfer publications, lectures, scientific training, science communication, or technology transfer in companies.

Interaction: The university actors strive to ensure that their transfer activities are effective in practice; this requires input and feedback loops from the practitioners to the university [44]. Universities survey the needs and problems in practice and try to determine, or at least estimate, the impact of a transfer. Reciprocal exchange between university actors and practitioners is thus a key characteristic. Examples are mutual problem descriptions for knowledge-based problem resolution or mutual evaluation of knowledge and technologies in the respective context of activity.

Co-production: If universities want to address sustainability problems and contribute to solving them, then joint knowledge production as equal partners makes sense. All transfer partners contribute their competencies, strengths, and perspectives to joint learning and research processes as part of

co-production. Examples are joint problem descriptions, joint development of guiding principles and sustainability visions, and the development of transformative solutions (e.g., in real-world laboratories).

6. Discussion and Outlook: what does Sustainability Transfer Mean for Universities?

The definition of sustainability transfer provided, and the six descriptive characteristics enable such transfer activities to be systematically identified and described. Visibility can increase appreciation for this form of transfer. The descriptive characteristics also enable empirical analysis and comparison between different forms of sustainability transfer, which in turn allows a more precise positioning in the focal area of universities and increases the academic compatibility.

Nevertheless, the concept needs further specification regarding the distinction between transfer and the third mission. The traditional notion of transfer in the HEIs' context as unidirectional top-down process from university into practice and business regarding technology transfer is problematic. Analyses of transdisciplinary research show that the process of co-design and co-production is crucial when addressing real-world problems and having societal impact [45]. For this reason, it might be productive to speak of a fourth mission of co-creation for sustainability [6]. The context of knowledge generation and especially that of adoption and implementation, the pick-up context, need further consideration with regard to transfer [40]. This implies a stronger focus on the perspective of practitioners.

Further, a more detailed consideration of the broad range of possible instruments and formats for transfer is preferable. In this regard, sustainability transfer can draw on experience with transdisciplinary concepts both from sustainability research and from ESD. For the latter, a service-learning model is an excellent example of how practitioners can communicate societal needs to university and how real-world problems can be integrated in teaching. In a service-learning model, students are expected to provide a direct community or social service that meets real community needs, while supporting purposeful civic learning [8]. Finally, the interviewees underline the usefulness of accepted forms of impact assessment for sustainability transfer that are still lacking.

In our estimation, sustainability transfer in teaching is already being carried out in some places at universities, without this being explicitly stated. By deliberately establishing a link to the concept of sustainability transfer, the potential, but also the limits, can be more clearly identified. In teaching, practitioner–university partnerships can enhance the practical focus of the degree programs, improve professional qualification of students, and support integrated competence orientation during studies [22]. Sustainability transfer can thus be a concrete implementation element of ESD.

The descriptive characteristics of sustainability transfer can be used to reflect on existing activities in teaching. The phases, the universal tasks, and the degrees of complexity help to position current or planned teaching activities and show very different starting points as well as development potential, which can be realized step by step.

Sustainability transfer is linked to a societal need. Students' learning and activities make a difference for society and the specific transfer partners. Students thereby become transfer actors in the context of sustainability and can make a contribution to society on behalf of universities [46]. In this way, sustainability transfer creates opportunities for the university to have an impact on society and to use its competencies in processes to shape society [5].

However, sustainability transfer in teaching depends on certain prerequisites. Resources such as time, space, and money are needed. Teaching concepts must be brought into line with university regulations, such as examination regulations, which requires flexibility on all sides. Moreover, transformative education and ESD are hardly incorporated structurally in teaching programs. There is a gap in HEIs' teaching concepts in Germany regarding ESD and educating change agents for sustainable development [47]. Nevertheless, HEIs are a highly appropriate place to train key competences for future change agents; and sustainability transfer in teaching is a promising approach that allows students to explore, test, experience, and reflect such key competences in a holistic as well as in a realistic way.

These aspects extend beyond sustainability transfer in teaching. In keeping with the whole institution approach, practitioner–university partnerships require that university actors communicate

as coherently as possible with practice actors. Regarding the case study of sustainability transfer in teaching with a focus on regional sustainable development, different practitioners cooperate with different members of the HEI, at least with students and teachers, possibly also with researchers as disciplinary experts and representatives of the administration. In this situation, students have to navigate between scientific rigor and pragmatic real-world solutions. This points to the crucial role of organizational culture in sustainability governance that enables and fosters consistent communication between diverse university actors. In this context, Niedlich et al. emphasize the importance of a holistic orientation based on the university's purpose and its concept of sustainability [48].

When HEIs open themselves up to a certain degree to practitioners and seize on new ideas, they are probably in a better position to contribute to sustainable development effectively. In turn, cooperation between practitioners and universities can have an impact on the internal activities of universities and even on their organizational cultures. One example for this are specific organizational interfaces between the university and practitioners that a reciprocal exchange, as well as communication and cooperation between heterogeneous members of HEIs, with regard to external relationships.

However, university transformation goes beyond organizational changes. Following a whole institution approach requires stimuli from the outside to trigger change within universities. This can also initiate alterations in the organizational culture, supporting a holistic orientation. This may enable a co-existence, an overlap, and a reinforcement of the different missions of HEIS [6]. In this sense, sustainability transfer can be understood as *one* element in sustainability transformation.

Author Contributions: Conceptualization, B.N., N.D., H.M., J.R., J.-H.S.; methodology, B.N., J.R., J.-H.S., N.D.; investigation, J.R., J.-H.S., B.N.; data curation, J.R., J.-H.S.; writing—original draft preparation, B.N., H.M., J.R., J.-H.S.; writing—review and editing, B.N.; supervision, B.N.; project administration, N.D.; funding acquisition, B.N. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the German Federal Ministry of Education and Research (BMBF), grant number 13NKE007L, within the collaboration "*Sustainability at Higher Education Institutions: develop – network – report*" (HOCH^N) and in 2017–2018 by the Ministry of Science, Research and Education of the Federal State of Brandenburg (MWFK).

Acknowledgments: We want to thank our colleagues Kerstin Kräusche, Kerstin Lehmann, Jens Pape, Alexander Pfriem, and Heike Walk for their highly valuable work in the team of the work package "Transfer" at the University of Applied Sciences for Sustainable Development Eberswalde. Thanks to Libby Neumann for very helpful language editing and translation. Our deep gratitude goes to our interview partners. We thank the reviewers as well as the editors of this special issue, Inka Bormann and Marco Rieckmann, for their valuable critique and comments.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Henke, J.; Pasternack, P.; Schmid, S. Third Mission bilanzieren. Die dritte Aufgabe der Hochschulen und ihre öffentliche; Kommunikation; HoF-Handreichungen 8; Institut f
 ür Hochschulforschung (HoF): Halle-Wittenberg, Germany, 2016.
- Henke, J.; Pasternack, P.; Schmid, S. Mission, die dritte. Gesellschaftliche Leistungen der Hochschulen neben Forschung und Lehre: Konzept und Kommunikation der Third Mission; BWV—Berliner Wissenschafts-Verlag: Berlin, Germany, 2017.
- HRK, Hochschulrektorenkonferenz. Für eine Kultur der Nachhaltigkeit. Empfehlungen der 25. Mitgliederversammlung der HRK; Hochschulrektorenkonferenz Mitgliederversammlung: Lüneburg, Germany; Hochschulrektorenkonferenz: Berlin/Bonn, Germany, 2018.
- Grunwald, A. Transformative Wissenschaft—eine neue Ordnung im Wissenschaftsbetrieb? GAIA Ecol. Perspect. Sci. Soc.Soc. 2015, 24, 17–20. [CrossRef]
- 5. Schneidewind, U. Die "Third Mission" zur "First Mission" machen? Die Hochschule 2016, 1, 14–22.
- Trencher, G.; Yarime, M.; McCormick, K.B.; Doll, C.N.H.; Kraines, S.B. Beyond the third mission: Exploring the emerging university function of co-creation for sustainability. *Sci. Public Policy* 2014, *41*, 151–179. [CrossRef]

- Brink, E.; Wamsler, C.; Adolfsson, M.; Axelsson, M.; Beery, T.; Björn, H.; Bramryd, T.; Ekelund, N.; Jephson, T.; Narvelo, W.; et al. On the Road to 'Research Municipalities': Analysing Transdisciplinarity in Municipal Ecosystem Services and Adaptation Planning. *Sustain. Sci.* 2018, *13*, 765–784. [CrossRef] [PubMed]
- Kricsfalusy, V.; George, C.; Reed, M.G. Integrating problem- and project-based learning opportunities: Assessing outcomes of a field course in environment and sustainability. *Environ. Educ. Res.* 2016, 24, 593–610. [CrossRef]
- 9. Papenfuss, J.; Merritt, E. Pedagogical Laboratories: A Case Study of Transformative Sustainability Education in an Ecovillage Context. *Sustainability* **2019**, *11*, 3880. [CrossRef]
- Walk, H.; Luthardt, V.; Nölting, B. Participatory Learning for Transdisciplinary Science in Biosphere Reserves—A Modified Role for Universities. In UNESCO Biosphere Reserves: Supporting Biocultural Diversity, Sustainability and Society; Price, M., Reed, M., Eds.; Earthscan/Routledge: New York, NY, USA; Oxon, UK, 2020; pp. 297–307.
- 11. Wissenschaftsrat. Wissens- und Technologietransfer als Gegenstand institutioneller Strategien; pPosition Paper (Drs. 5665-16); Wissenschaftsrat: Cologne, Germany, 2016.
- BMBF German Federal Ministry of Education and Research. Grundsatzpapier des Bundesministeriums f
 ür Bildung und Forschung zur Wissenschaftskommunikation; BMBF: Berlin, Germany, 2019.
- MWFK. Transferstrategie Brandenburg—Verbesserung der Zusammenarbeit von Wissenschaft mit Wirtschaft, Politik und Zivilgesellschaft; enacted by the cabinet on 5 September 2019; Ministry of Science, Research and Education of the Federal State of Brandenburg, Ed.; MWFK: Potsdam, Germany, 2017.
- Roessler, I.; Duong, S.; Hachmeister, C. Welche Mission haben Hochschulen? Third Mission als Leistung der Fachhochschulen für die und mit der Gesellschaft; CHE gemeinnütziges Centrum für Hochschulentwicklung (working paper 182): Gütersloh, Germany, 2015.
- Warnecke, C. Universitäten und Fachhochschulen im regionalen Innovationssystem. Eine deutschlandweite Betrachtung; RUFIS (Ruhr-Forschungsinstitut f
 ür Innovations- und Strukturpolitik e.V.): Bochum, Germany, 2016.
- 16. BMBF; German Federal Ministry of Education and Research. *Nationaler Aktionsplan Bildung für nachhaltige* Entwicklung. Der deutsche Beitrag zum UNESCO-Weltaktionsprogramm; BMBF: Berlin, Germany, 2017.
- UNESCO United Nations Educational, Scientific and Cultural Organization. General Conference, Framework for the implementation of education for sustainable development (ESD); UNESCO, Ed.; beyond 2019: Paris, France, 2019.
- Vogt, M.; Lütke-Spatz, L.; Weber, C.; Bassen, A.; Bauer, M.; Bormann, I.; Denzler, W.; Geyer, F.; Günther, E.; Jahn, S.; et al. *Nachhaltigkeitsverständnis des Verbundprojekts HOCH^N*; Ludwig-Maximilians-Universität Munich: Munich, Germany, 2020.
- Ferretti, J.; Daedlow, K.; Kopfmüller, J.; Winkelmann, M.; Podhora, A.; Walz, R.; Bertling, J.; Helming, K. Reflexionsrahmen für Forschen in gesellschaftlicher Verantwortung. BMBF-Projekt 'LeNa – Nachhaltigkeitsmanagement in außeruniversitären Forschungsorganisationen'; Leibniz-Gemeinschaft, Helmholtz-Gemeinschaft, Fraunhofer-Gesellschaft: Berlin, Germany, 2016.
- UNESCO United Nations Educational Scientific and Cultural Organization. *Roadmap for Implementing the Global Action Programme on Education for Sustainable;* UNESCO: Paris, France, 2014; Available online: https://unesdoc.unesco.org/ark:/48223/pf0000230514 (accessed on 20 January 2020).
- National Platform. National Action Plan on Education for Sustainable Development. The German Contribution to the UNECSO Global Action Programme; UNECSO: Berlin, Germany, 2017; Available online: https://www. bneportal.de/sites/default/files/downloads/publikationen/BMBF_NAP_BNE_EN_Screen_2.pdf (accessed on 20 January 2020).
- 22. Marcus, J.; Coops, N.C.; Ellis, S.; Robinson, J. Embedding Sustainability Learning Pathways across the University. *Curr. Opin. Environ. Sustain.* **2015**, *16*, 7–13. [CrossRef]
- Molitor, H. Education for sustainable development. In *Humans in the Global Ecosytem. An Introduction to Sustainable Development;* Ibisch, P.L., Molitor, H., Conrad, A., Walk, H., Mihotovic, V., Geyer, J., Eds.; Oekom Verlag: München, Germany, 2019; pp. 333–350.
- 24. Overwien, B.; Rathenow, H.-F. (Eds.) *Globalisierung Fordert Politische Bildung, Politisches Lernen im globalen Kontext*; Farmington Hills: Opladen, Germany, 2009.
- Rieckmann, M. Key themes in Education for Sustainable Development. In *Issues and trends in Education for* Sustainable Development; UNESCO: Paris, France, 2018; pp. 61–84.
- 26. Konrad, T.; Wiek, A.; Barth, M. Embracing conflicts for interpersonal competence development in project-based sustainability courses. *Int. J. Sustain. High. Educ.* **2020**, *21*, 76–96. [CrossRef]

- 27. Rieckmann, M. Learning to transform the world: key competencies in Education for Sustainable Development. In *Issues and trends in Education for Sustainable Development;* UNESCO: Paris, France, 2018; pp. 39–58.
- Bellina, L.; Tegeler, M.K.; Müller-Christ, G.; Potthast, T. Bildung für nachhaltige Entwicklung (BNE) in der Hochschullehre (Betaversion); BMBF-Projekt 'Nachhaltigkeit an Hochschulen: Entwickeln - vernetzen – berichten (Hoch-N): Bremen/Tübingen, Germany, 2018.
- Kates, R.W.; Clark, W.C.; Corell, R.; Hall, J.M.; Jaeger, C.C.; Lowe, I.; McCarthy, J.J.; Schellnhuber, H.J.; Bolin, B.; Dickson, N.M.; et al. Sustainability science. *Science* 2001, 292, 641–642. [CrossRef] [PubMed]
- 30. Defila, R.; Di Giulio, A. (Eds.) *Transdisziplinär forschen Zwischen Ideal und Gelebter Praxis. Hotspots, Geschichten, Wirkungen;* Campus Verlag: Frankfurt, Germany; New York, NY, USA, 2006.
- Lang, D.; Wiek, A.; Bergmann, M.; Stauffacher, M.; Martens, P.; Moll, P.; Swilling, M.; Thomas, C.J. Transdisciplinary research in sustainability science - practice, principles, and challenges. *Sustain. Sci.* 2012, 7, 25–43. [CrossRef]
- Schneidewind, U.; Singer-Brodowski, M. Transformative Wissenschaft. Klimawandel im deutschen Wissenschaftsund Hochschulsystem, 2nd ed.; Metropolis: Marburg, Germany, 2014.
- WBGU, German Advisory Council on Global Change. World in Transition—A Social Contract for Sustainability; Flagship Report; WBGU Secretariat: Berlin, Germany, 2011.
- Wanner, M.; Hilger, A.; Westerkowski, J.; Rose, M.; Stelzer, F.; Schäpke, N. Towards a Cyclical Concept of Real-World Laboratories. *disP - Plan. Rev.* 2018, 54, 94–114. [CrossRef]
- Nölting, B.; Dembski, N.; Dodillet, J.; Holz, J.; Lehmann, K.; Molitor, H.; Pfriem, A.; Reimann, J.; Skroblin, J.-H. *Transfer stärkt Lehre. Wie Nachhaltigkeitstransfer Hochschullehre inspirieren kann*; HNEE (HOCH^N discussion paper at HNE Eberswalde); No. 01; Eberswalde University for Sustainable Development: Eberswalde, Germany, 2018.
- 36. Gläser, J.; Laudel, G. *Experteninterviews und Qualitative Inhaltsanalyse als Instrumente Rekonstruierender Untersuchungen*, 4th ed.; VS Verlag für Sozialwissenschaften: Wiesbaden, Germany, 2010.
- Reimann, J. Nachhaltigkeitstransfer an Hochschulen und der Beitrag f
 ür eine nachhaltige Regionalentwicklung Die Rolle von Studierenden in Projektarbeiten zur Regionalentwicklung; HNEE: Eberswalde, Germany, 2019.
- 38. Wiek, A.; Withycombe, L.; Redman, C.L. Key competencies in sustainability: A reference framework for academic program development. *Sustain. Sci.* 2011, *6*, 203–218. [CrossRef]
- Krainer, L.; Winiwarter, V. Die Universität als Akteurin der transformativen Wissenschaft. GAIA 2016, 25, 110–116. [CrossRef]
- Nagy, E.; Ransiek, A.; Schäfer, M.; Lux, A.; Bergmann, M.; Jahn, T.; Marg, O.; Theiler, L. Transfer as a reciprocal process: How to foster receptivity to results of transdisciplinary research. *Environ. Sci. Policy* 2020, 104, 148–160. [CrossRef]
- Kaufmann-Hayoz, R.; Defila, R.; Di Giulio, A.; Winkelmann, M. Was man sich erhoffen darf Zur gesellschaftlichen Wirkung transdisziplinärer Forschung. In *Transdisziplinär forschen – zwischen Ideal und* gelebter Praxis. Hotspots, Geschichten, Wirkungen; Defila, R., Di Giulio, A., Eds.; Campus Verlag: Frankfurt, Germany; New York, NY, USA, 2016; pp. 289–327.
- 42. Kurz, B.; Kubek, D. Kursbuch Wirkungen. Das Praxishandbuch für Alle, die Gutes Noch Besser tun Wollen, 3rd ed.; PHINEO gAG.: Berlin, Germany, 2015.
- Nölting, B.; Dembski, N.; Kräusche, K.; Lehmann, K.; Molitor, H.; Pape, J.; Pfriem, A.; Reimann, J.; Skroblin, J.-H.; Walk, H. *Transfer für Nachhaltige Entwicklung an Hochschulen*; HNEE: Eberswalde, Germany, 2018.
- Bormann, I. Zwischenräume der Veränderung. Innovationen und ihr Transfer im Feld von Bildung und Erziehung; VS Verlag für Sozialwissenschaften: Wiesbaden, Germany, 2011.
- Lux, A.; Schäfer, M.; Bergmann, M.; Jahn, T.; Marg, O.; Nagy, E.; Ransiek, A.-C.; Theiler, L. Societal effects of transdisciplinary sustainability research - How can they be strengthened during the research process? *Environ. Sci. Policy* 2019, 101, 183–191. [CrossRef]
- 46. Singer-Brodowski, M. Studierende als GestalterInnen einer Hochschulbildung für Nachhaltige Entwicklung. Selbstorganisierte und Problembasierte Nachhaltigkeitskurse und ihr Beitrag zur Überfachlichen Kompetenzentwicklung Studierender; Berliner Wissenschafts-Verlag: Berlin, Germany, 2016.

- 47. Etzkorn, N.; Singer-Brodowski, M. Verankerung von Bildung für nachhaltige Entwicklung im Bildungsbereich Hochschule. In Wegmarken der Transformation. Nationales Monitoring von Bildung für nachhaltige Entwicklung in Deutschland; Brock, A., Haan de, G., Etzkorn, N., Singer-Brodowski, M., Eds.; Verlag Barbara Budrich: Berlin, Germany; Toronto, YTO, Canada, 2018.
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, *5*, 1–18. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article Interdependencies of Culture and Functions of Sustainability Governance at Higher Education Institutions

Mara Bauer^{1,*}, Sebastian Niedlich², Marco Rieckmann¹, Inka Bormann² and Larissa Jaeger¹

- Department of Education, Faculty of Education and Social Sciences, University of Vechta, Driverstraße 22, 49377 Vechta, Germany; marco.rieckmann@uni-vechta.de (M.R.); larissa jaeger@uni-vechta.de (L.J.)
- ² Department of Education and Psychology, Freie Universität Berlin, Habelschwerdter Allee 45,
- 14195 Berlin, Germany; sebastian.niedlich@fu-berlin.de (S.N.); inka.bormann@fu-berlin.de (I.B.) * Correspondence: mara.bauer@uni-vechta.de

Received: 28 February 2020; Accepted: 31 March 2020; Published: 1 April 2020

Abstract: Sustainable development practices in higher education institutions are diverse, with regard not only to the types of challenges that have to be addressed, but also to the forms of sustainability governance adopted by individual higher education institutions. This paper aims to reflect on the aspects of organizational culture that are particularly crucial for the implementation of sustainable practices at higher education institutions. Specifically, it addresses the research question: how do different organizational cultures affect approaches to sustainability governance at higher education institutions (HEIs)? It reflects on data from multi-case studies at eleven German higher education institutions. Four of the cases are analyzed in this paper to draw out the insights they offer on how organizational culture shapes the institutions' approach to sustainable development. A governance equalizer is used as a functional framework for evaluating and discussing the influence of different cultural orientations on sustainability governance. In addition to providing many insights and findings in relation to specific cases, comparison of the different institutions, their governance structures and their cultures of sustainable development helps to emphasize that there is no single cultural factor that can be identified as directly promoting particular governance structures. Rather, there is an active interplay between cultural orientations, which influence, and are also influenced by, the measures deployed. Such influence is not instantly apparent but needs time to develop, and it evolves in a variety of ways as illustrated by the case studies.

Keywords: sustainability governance; higher education; organizational culture; governance equalizer; Germany

1. Introduction

The aspiration of integrating sustainable development (SD) into the structures and processes of higher education institutions (HEIs) while taking account of all their different fields of activity, such as research, teaching, outreach, and operations, has been recognized in practice and research alike as a complex and highly context-dependent task [1–4]. The definitions of sustainable development in the context of higher education institutions remain as rhetorically malleable and as fluid as those in public and political discourse [5]. As a result, SD practices in HEIs have been diverse, not only with regard to the types of challenges that have been addressed, but also in respect of the forms of governance that have been established to coordinate the approach to such challenges (e.g., [6]). Attempts to classify SD implementation structures within HEIs (e.g., [7,8]) have, along with other literature, highlighted the need for a shared commitment to sustainability goals, structures and processes amongst HEI staff, covering teaching, research and operations—in other words, the need for holistic governance [9].

Following that line of argument, HEI governance for sustainability aims to promote and support the changes institutions need to put in place to embed SD in their organizations. It is concerned with all the circumstances, pathways and methods by which sustainability reaches and permeates HEIs. At the core of these processes are the people that make up HEIs. The HEI and its sustainability pathway are essentially comprised of their ideals, vision and engagement, and the ways they communicate and consult [10–12]. This deserves special attention.

Since sustainable development is a complex concept, and HEIs are complex organizations, it is important to be aware of the implications and requirements of a thorough and comprehensive approach to the introduction of SD. For many HEIs, it will presumably mean making a number of fundamental changes within the institution. In this context, the word 'transformation' is not an exaggeration but a prerequisite [13,14]. Cameron and Quinn [15] stress the notion that the best way of implementing such major organizational changes is to take the organizational culture into account and shape it in a way that supports such change. "Although the tools and techniques may be present and the change strategy implemented with vigour, many efforts to improve organizational performance fail because the fundamental culture of the organization – values, ways of thinking, managerial styles, paradigms, approaches to problem solving – remains the same" [15] (p. 11). The failure to achieve organizational change often has ramifications for members of organizations, and could mean that their response to future change initiatives might be even less cooperative [15].

While Cameron and Quinn are concerned with organizational change in general, other research supports their findings in the particular context of sustainable development in HEIs [16,17]. HEIs need to find ways to embed SD into their specific norms and values and into the shared goals of their members, and to create common ground so as to enable their organizations to learn sustainability, learning being essential to the achievement of a paradigm shift [18]. Even though HEIs might be considered experts in learning, this does not necessarily apply to their capacity to learn for and about themselves [17].

As Adams et al. clearly state, "the failure to embed sustainability in HEIs suggests it has failed to become part of the culture" [16] (p. 2). We, therefore, seek to investigate whether this suggestion is supported by the experiences of HEIs, and, more specifically, which kinds of cultural variables feed into which governance structures. Organizational culture in general can be considered an integral part of sustainability governance, since each HEI has a specific mode of engagement with SD depending on its prior experiences, its code of communication, its resources in terms of expertise and time, its recognition of SD as an opportunity for profiling, its perception of public pressure on the HEI, etc. It is, therefore, appropriate to discuss the functions and cultures of sustainability governance at HEIs in greater detail.

Accordingly, this paper explains and brings together two different perspectives on sustainability governance at HEIs: culture and functions. Special attention is paid to the impact that certain organizational cultures have on the development of aspects of functional governance at HEIs, i.e., the structures, processes and environment that are created there. The corresponding research question is: how do different organizational cultures affect approaches to sustainability governance at HEIs?

Against this backdrop, we aim to investigate distinct aspects of organizational culture that are particularly crucial to the implementation of SD at HEIs. In this context, we are taking, as our basis, our earlier work in which we prepared two major frameworks—one on culture [19] and one on functions [2]—and which we will bring together in this paper. We identified two types of cultural orientation that are usually prevalent in HEIs in relation to the attitudes, assumptions and practices that they bring to SD implementation [19]. These are summarized in Section 2.1 and followed in Section 2.2 by an introduction to the governance equalizer, an HEI sustainability governance concept that we used in this paper as a functional framework for evaluating and discussing the significance of the cultural orientations in question. Section 2 thus provides an overview of the frameworks deployed in the analysis that follows.

The foundation for evaluating the various aspects of HEIs' approaches to implementing SD was guided and supplemented by desk research but consisted primarily of extensive empirical research on sustainability governance at eleven German HEIs, which resulted in eleven different case studies. The methodology and research process that eventually led to this paper are described in Section 3. Having laid the groundwork for a combined analysis of the cultural and functional aspects of SD governance in HEIs, Section 4 introduces four of the eleven case studies and investigates how the HEIs' organizational cultures are interlinked with their processes and structures for SD implementation. The subsequent discussion and conclusion highlight the main learning points, merits, limitations, and outstanding or emerging questions.

2. Frameworks on Culture and Functions of Sustainability Governance at HEIs

2.1. Cultural Orientations

Initially, basing ourselves on qualitative content analysis, we derived four cultural dimensions from our interview material, which were then conflated into two overarching cultural orientations that distinguish the way in which HEIs face the challenges of SD: *holistic orientation* and *organizational learning orientation* [19].

Holistic orientation

This category combines the two dimensions *conception of sustainability* and *relevance and scope of organizational change*. Both dimensions emphasize the need to consider different elements simultaneously, and to link them in an integrated approach. Thus, we use the term *holistic* not as a philosophical concept, but to refer to the comprehensiveness of the HEI's orientation towards SD.

Conception of sustainability refers to the understanding that an HEI has of the comprehensiveness of SD. Does the HEI include ecological viability, social justice and economic capacity alongside each other in its long-term responsibilities? A variety of professional perspectives at an HEI might come with a variety of conceptions of sustainability that often focus almost exclusively on one of these dimensions of sustainability instead of taking an integrated approach [20]. The HEI's willingness to combine these perspectives and find common ground to start from can open up new spaces and options for SD that can be discovered and seized [21].

Relevance and scope of organizational change refers to an HEI's idea of where SD should be implemented. Viewing the HEI as a single system whose subsystems such as teaching, research, operations and outreach are given equal footing as fields of activity for SD (whole-institution approach) is deemed crucial [22,23]. As with the *conception of sustainability*, this dimension deals with the question of comprehensiveness. While the first is concerned with the content of SD, the latter considers the context. Taken together, the two dimensions, therefore, represent the holistic ideal inherent in the concept of sustainable development.

Organizational learning orientation

This category combines the dimensions attribution of responsibility and purpose of the higher education institution.

The focus is on the degree to which SD is seen as an issue of organizational development and learning which, firstly, can be represented by the *attribution of responsibility* for SD. In this context, the call for collaborative and co-creative processes receives broad support [10,21,22,24,25]. SD should not be reserved to the institution's elite, but based on the approval and engagement of many [22] and, therefore, handled by means of shared responsibilities and co-determination. This also entails the institution's leadership taking an open-minded approach to the emergence of innovative ideas and unexpected developments in relation to SD from different members of the HEI and to taking SD ideas and goals seriously by fostering discussions [26].

Another aspect of organizational learning derives from the way an HEI perceives itself and its role in the societal context: the *purpose of the higher education institution*. Trencher et al. [27] have coined the term "fourth mission" as a description of the sustainable HEI's role in society: a debater, a networker, and a transformer. HEIs can actively promote co-creation of societal transformation that goes far beyond technology transfer and other economic contributions, which is mostly what the third mission concept has been requesting and seeking to pursue. The idea of a "fourth mission" takes HEIs' role in society further towards active engagement and mutual learning. If an HEI takes on this role, its chances of transforming itself as an institution and achieving a higher level of SD improve. This is especially true when active engagement and mutual learning are paired with participatory processes, since this creates an environment that promotes critical, systemic and future-oriented thinking.

On the basis of our findings, we had assumed that the two orientations had a linear relationship and tended to mutually reinforce each other [19], but, so far, we have not taken a closer look at their implications for the structures, procedures and measures of sustainability governance at HEIs. How are HEIs' sustainability governance set-ups influenced by different cultural orientations? The next section will introduce a functional framework that forms the basis of the subsequent analysis of this question.

2.2. Depicting the Functions of Sustainability Governance through a Governance Equalizer

In order to illustrate governance structures within institutions, de Boer et al. [28] use the image of a piece of audio engineering equipment, known as an equalizer. Instead of base, reverberation or volume, this particular equalizer is equipped with slide controls that display different dimensions of institutional governance and the performance of the institution against each dimension, or functional requirement. In analyzing sustainability governance at HEIs, we draw on Franz and Brüsemeister [29], and have chosen five dimensions for our equalizer: politics, profession, organization, knowledge, and public [2].

The following overview gives a brief outline of the equalizer dimensions which are the result of combining the original heuristic with its representation within our case studies.

1. Politics: How is sustainability implemented and legitimized within and outside the HEI?

This equalizer dimension deals with the question of how sustainability activities are not only selectively supported, but form part of an HEI's long-term development goals. The term politics should not be misunderstood: it does not refer to the state-level measures that set the overall framework for higher education institutions, but to the actions of the HEI members themselves. In order to embed SD, formal and informal decisions and resolutions are required that are binding on as many actors at the HEI as possible. Such definitions give legitimacy to those who are pursuing SD, providing a basis for their actions. They provide goals and standards that act as guidance for sustainability-related activities, and can serve as a basis for reviewing the success of such activities. Collectively binding decisions can be taken both by the hierarchy and as part of participatory processes.

2. Profession: How are professional perspectives and competencies linked?

The dimension of profession is about the gradual development of an interdisciplinary and cross-sectoral understanding of sustainability at an HEI. The different areas of activity at HEIs—teaching, research, and campus management (administration, technology, etc.)—have very different requirements, processes and framework conditions, demand different competencies and specialist knowledge, and have their own standards and 'cultures'. The scientific disciplines also differ considerably in this regard, as do the external actors HEIs are involved with. However, SD requires that internal and external actors work closely together. This facilitates exchanges of views on what SD should include, what principles and standards should be taken into account, and how SD can be integrated into the day-to-day practices of different activities and disciplines.

3. Organization: How are collaborative work and task processing made possible?

In order to encourage SD in HEIs, it is necessary to specify goals and activities that facilitate implementation. This involves providing resources and creating structures and processes that ensure continuous and reliable work. In the context of SD, it is important to ensure long-term action across existing organizational boundaries—between different areas of activity as well as with non-academic actors. To do so, it is necessary to network actors and to coordinate their activities.

4. Knowledge: How is the required knowledge developed and used competently?

SD in higher education requires consideration of the complex issue of *knowledge*. The actors in HEIs have to develop shared ideas about what the problems are and what has caused them (analysis), they have to agree on how the situation is to be evaluated and what the aims will be in the future (goals). It is also important to clarify which measures will be used to solve problems (action). Alongside technical and professional expertise, knowledge of different actors' responsibilities, structures and processes at an HEI plays an important role in clarifying how measures can be best implemented. At the same time, each solution usually entails specific advantages and disadvantages. This means that deciding which measures to implement involves a normative judgment. It is not sufficient, therefore, for the sustainability governance at HEIs to access knowledge only in times of need. Instead, ways must be established to identify, disseminate, and use the relevant knowledge on a continuous basis in order to be able to react flexibly when problems arise and to facilitate longer-term learning processes. This requires more than technical solutions—it also involves collaboration and networking with the aim of supporting exchange and joint knowledge work.

5. Visibility: How is awareness of the need for sustainable development in HEIs created?

Making an HEI's sustainability efforts visible to the internal and external public plays an important role in SD governance, as it provides the opportunity for all the institution's stakeholders to observe and respond to issues and positions as well as activities and their outcomes. Public attention can thus contribute to greater participation in and awareness of SD, such as identifying the need for action, communicating and justifying goals and actions, and reporting on progress. Last but not least, in this way, the importance of sustainability for the HEI can be demonstrated and sustainability can become an important element of an HEI's profile and thus help to attract students or external partners.

We will consider the specific interplay of organizational culture and the implementation of SD at different HEIs against this framework. This paper will, therefore, analyze four cases relating to the ways in which cultural orientation trickles down and can influence the structures and processes that are being established to promote SD, and that then become visible in the governance equalizer. The general assumption that can be made at this point is that a strong cultural orientation towards SD leads to better and more coherent SD governance. How this can take place in different real-world contexts is shown in detail in Section 4.

3. Methodology

This paper is based on the findings of two earlier papers [2,19] that stem from the same research. The first [2] sought to validate the governance equalizer through ten expert interviews; this was then used as a theoretical basis for the evaluation of 61 stakeholder interviews. The second [19] focused on the evolution of categories from this extensive interview material, namely the development of an empirically-based categorization of the characteristics of sustainability governance in HEIs, and further condensation into two cultural orientations. Both the governance equalizer and the two categories of cultural orientations are explained above.

In order to answer the research question for this paper, "How do different organizational cultures affect approaches to sustainability governance at HEIs?", four exemplary HEIs were selected for the purpose of analyzing and illustrating the effect of organizational cultures on governance structures

and processes. The selected cases (described in Section 3.3) are divided into two pairs of HEIs that appeared to offer promising insights into the effect of cultural orientations on sustainability governance after being categorized in Niedlich et al. [19].

The following description of the research process gives an overview of the whole process, and then focuses on the measures that were taken in order to achieve insight into the interplay of cultural orientation and the governance equalizer. By combining the perspectives of culture and functions from the two earlier papers and the frameworks that these established, and feeding them with the four case studies, our aim is to deepen understanding of the significance of organizational culture for SD at HEIs in general, but also with respect to specific functions within HEI SD governance.

3.1. Data Collection

The study that provided the insights outlined above on sustainability governance and organizational culture for SD at HEIs looked at the experiences of, and observations from, eleven German HEIs. These HEIs are all members of the joint research project "Sustainability at Higher Education Institutions: develop-network-report" (HOCH^N) which, in addition to sustainability governance, focuses on research, teaching and operations as well as on the transfer of SD-related knowledge and approaches (see https://hoch-n.org/en).

The research design consisted of face-to-face interviews with different actors at the eleven HEIs. Interview partners were chosen on the basis of selective sampling. In order to gain an in-depth understanding and to incorporate different views and perspectives on sustainability governance, 61 stakeholders from the following groups were interviewed:

- HEI management (11 interviews),
- SD coordinators/commissioners/managers (10 interviews),
- Student initiatives (14 interviews),
- Technical administration (14 interviews), and
- Academic staff (professors and researchers) (12 interviews).

In addition to time and willingness to participate, selection criteria included knowledge and first-hand experience of their institution's sustainability process (cf. [30]). The interviewees were identified and recruited in collaboration with our partners within the HOCH^N network as well as through desktop research. The sample is thus limited by the number of project partners. These provided an interesting sample, as they all were willing to participate in the interviews, had sufficient range of stakeholders involved in the sustainability process at their institutions and hence could provide a variety of perspectives. Although the sample is not representative of all HEIs in Germany, as the participation in the HOCH^N project already implies a bias towards a stronger commitment to sustainability, it shows interesting cases, with very different approaches to the organization of sustainability.

The interviews were conducted using an interview guide consisting of five sections, each containing one key question and supplementary questions. The five sections were:

- 1. Development of the sustainability process;
- 2. Importance of specific actors, structures and processes;
- 3. Sustainability as a guiding principle for the HEI;
- 4. Assessment of the sustainability process;
- 5. Assessment of the stakeholder network by means of the governance equalizer heuristic.

Of the 61 interviewees, 30 were female and 31 were male. As the study was designed on a cross-sectional basis, all individuals were interviewed only once. The average duration of an interview was 47 minutes. All interviews were recorded and transcribed using a simplified approach [31].

3.2. Data Analysis

The material was analyzed in several successive phases. As an initial step, data from the eleven case studies were analyzed using qualitative content analysis [32]. To ensure that data were adequately analyzed and interpreted, all phases of the analysis involved a minimum of two researchers and data analysis and interpretation was validated between researchers using a communicative process (cf. [33]).

First, a qualitative thematic text analysis was conducted, using primary categories derived from the interview guide. After coding data in line with these and other deductive categories from three HEIs, additional categories and subcategories were created. The data from all eleven cases were coded using the revised categories and category-based analysis was undertaken, creating thematic summaries for all coded text segments.

On this basis, case profiles were produced, which, in addition to a timeline for SD processes, included findings on the state of SD activities, initiatives and actors, motives and objectives, views on the societal role of HEIs, influencing factors (structural/institutional, process related, personnel, HEI size, external), and factors of particular interest in relation to specific cases. The case profiles were used to carry out thematic cross-case analysis with the aim of identifying differences and similarities (Figure 1).



Figure 1. Original analysis (focus on sustainability initiatives and governance equalizers) [19].

This led to the identification of four major cultural dimensions of sustainability governance at HEIs [19] characterizing HEIs' conceptions of themselves as institutions as well as their attitudes towards SD (implementation). Each of these cultural dimensions (described briefly in Section 2.1) was then conceptualized as a continuum, ranging from an ideal type of slightest expression to an ideal type of strongest expression. Once every HEI had been located on each of the continuums by means of qualitative content analysis, the results suggested the condensation of the four continuums into two overarching orientations, each comprised of two of the former dimensions (Figure 2).



Figure 2. Subsequent analysis (recoding with a focus on the culture of sustainability governance) [19].

At the end of these two analytical phases, two different perspectives on sustainability governance at HEIs emerged: the governance equalizer, with its requirements relating to the functions of sustainability governance, and the organizational orientations crucial to sustainability governance on a cultural level. In order to probe the interrelation of these perspectives, further examination of the case studies appeared necessary and became the aim of this paper.

For that purpose, two pairs of cases were chosen, and their general sustainability processes were compared within each pair; more importantly, the governance structures that the institutions had set up in light of their organizational cultures were also compared. To facilitate this, the case profiles were

first examined individually for signs in the equalizer dimensions that allowed aspects of the HEIs' cultural orientations to become evident. We then compared the elements that we had identified within each pair to determine the extent to which the different cultures of HEIs had an impact on the design of individual aspects of governance. This acted as the basis for the formulation and discussion of more general assumptions about the interaction of cultures and functions (Figure 3). The reasoning behind the selection of cases is below and is followed by a retracing of the analytical process.



Figure 3. Combined analysis of the cultural orientations and governance equalizer dimensions.

3.3. Selection of Cases for a Comparative Analysis

Cases were selected on the basis of Niedlich et al. [19] and the classification of the cultural orientations at the eleven HEIs presented there. From this standpoint, two pairs of HEIs were identified as particularly interesting. They all stemmed from the center ground of the HEIs under consideration, with one pair showing diverging orientations and the other converging orientations in relation to the regression line (Figure 4) that was established for the complete sample of eleven HEIs. It revealed a clear tendency towards a linear relationship between *holistic* and *learning orientation*.

Cases VI and VIII were chosen here, for they were the ones that deviated the most from the assumption that *holistic orientation* and *organizational learning orientation* are in a linear relationship, but the nature of their divergence was different (note: the numbers of the cases correspond to their use in [19]). Case VI was found to have a more advanced *holistic orientation* than *learning orientation*, while case VIII had a culture that was more inclined to promote mutual and inclusive learning and was less focused on having a comprehensive concept of sustainability and all-pervasive processes [19]. We concluded that comparison of the two cases might reveal how the different orientations affected governance structures.

Cases III and IX both supported our assumption of a mutually reinforcing relationship between *holistic* and *organizational learning orientations* (Figure 4), but they diverged considerably in their expression of this. In fact, they were the two institutions at the opposite ends of a spectrum of HEIs in the center ground of the sample when cultural orientation was analyzed [19]. The cultural orientation of HEI III towards sustainable development was much less evolved than in HEI IX. It was deemed interesting to compare features of the governance equalizer dimensions for each institution as they related to different stages of development in governance.

Figure 4 situates the four HEIs in a matrix between holistic orientation and learning orientation. Since all four cases demonstrated mid-level performance as far as their cultural orientation towards SD was concerned (compared with the other cases in the sample), the field only ranges from relatively low to relatively high. Together, the four cases represent all of the possible combinations of the two orientations in the matrix (HEI VI = high + low; HEI VIII = low + high; HEI III = low + low;

HEI IX = high + high). A practical but somewhat coincidental area of common ground between each pair of HEIs is that their current sustainability processes started at similar times and in similar ways. Additionally, all four of the institutions are financed by the state and not privately funded, which also supports their comparability.



Figure 4. Matrix of selected cases between holistic and learning orientation.

4. Results

A comparative analysis of the two pairs of HEIs provides the following insights in respect of the effects of cultural orientation (*holistic* and *learning orientation*) on the functional requirements of sustainability governance, as illustrated by the sustainability governance equalizer. The two cases constituting each pair are first described separately and then compared with each other.

4.1. Cases VI and VIII: Strong Holistic Orientation and Pronounced Learning Orientation vs. Low Holistic and Distinct Learning Orientation

Cases VI and VIII were used to gain an impression of the extent to which a difference in the degree of the two cultural orientations impacts on governance structures. Case VI had a stronger focus on *holistic orientation* and case VIII had a more pronounced focus on *learning orientation*. While the SD structures and processes that were set up at the two HEIs appear quite similar from the outside, significant differentiation was found. The case descriptions and analyses below explore and draw out, in particular, the differences relating to the manner of structures' outreach into their institution, how autonomous they are and what their conceptual focus is.

4.1.1. Case VI: Strong Holistic and Low Learning Orientation

HEI case no. VI has between 5000 and 10,000 employees and between 25,000 and 50,000 students in 100–200 different study programs with a broad spectrum, and is located in an urban environment. The beginning of its sustainability process can be traced back almost 30 years to the early 1990s when concerns about the scarcity of global resource were raised by campus operations staff. Energy efficiency has been on the construction department's agenda ever since, with other environmental issues

being discussed in a working group specially established for this purpose. From there, sustainability as a concept evolved in HEI VI and was recognized by senior management in recent years as a guiding principle for the whole institution, with the potential to promote and underline its reputation for scientific excellence. A center of expertise was established as the central SD governance body, mediating between senior management and the rest of the HEI. It is comprised of five strategic working groups covering the whole institution and with participation by academic staff, administrative staff and students.

Many activities aimed at creating a sustainable HEI were initiated, including the preparation of a sustainability report and the development of a mission statement. Nevertheless, the HEI's overall SD process is said to be lacking in terms of oversight of the various activities, the transfer of principles into day-to-day practices and broad participation.

Further descriptions of the case are given in Figure 5, which sets the HEI's cultural orientation next to its governance equalizer dimensions. It creates a more comprehensive picture of the HEI's relatively highly developed *holistic orientation* and relatively underdeveloped *learning orientation* in relation to SD. Taking all these aspects into consideration, in what ways they might have influenced each other in the particular context of HEI VI can then be discussed.

SD Cultural Orientations	Governance Equalizer Dimensions
 Learning orientation: The HEI assigns sole authority to senior management in setting the tone for the whole process, has established a centre of expertise that is subject to instructions from above, but seeks to open the process to all HEI members, sees itself as well embedded in society, but its societal memory is ensidered a 	Politics There is an official commitment for SD on the part of senior management, accompanied by strategies and sub strategies and implemented through the center of expertise. Sustainability is embedded in the HEI's statutes and in its general mission statement. \rightarrow rather strong expression
	Profession The center of expertise has initiated a participatory process for the development of a joint definition of sustainability which has brought people from different fields together. Some of whom are also working on interdisciplinary research projects with direct relevance to SD. Sustainability is starting to be actively discussed, but is not yet dealt with in general lectures and seminars. The handling of the HET's sustainability paradigm differs markedly between departments. \rightarrow low to medium expression
'task for the future'. → relatively low expression	Organization Many of the features of HEI VI already mentioned form a comprehensive organizational structure. The center of expertise, the statute and the strategies to promote sustainability are
 Holistic orientation: The HEI has a long tradition of taking environmental considerations into account, and many different conceptions of sustainability among its members, but the centre of expertise works to unite and concretize them, pursues a whole-institution approach, supported by the division of the centre of expertise into different working groups. → relatively high expression 	recognized as reasonable. However, the center of expertises efforts to involve a more diverse group of stakeholders in a structural way has so far not been as fruitful. Administrative procedures have not been aligned with sustainability standards due to a lack of direction, in spite of improved communication with senior management via the center of expertise. \rightarrow medium to strong expression
	<i>Knowledge</i> SD-related knowledge at HEI VI consists of a variety of material that has been published by the center of expertise or the HEI in general. The more extensive examples are the sustainability report and a best practice compilation of different teaching formats for SD. However, there is criticism of the fact that internal knowledge transfer is not sufficiently pervasive and has not reached some key stakeholders preventing them from taking cohesive action. \rightarrow low to medium expression
	Public There is currently some overlap between the knowledge and the public dimensions. Accordingly, the public is addressed, but has not been reached in such a way as to establish general awareness of the HET's sustainability process.

Figure 5. Overview of the cultural orientations and the governance dimensions of higher education institution (HEI) VI.

In light of the cultural orientation, some of the governance equalizer values do not come as a surprise. The HEI's low *learning orientation*, and in particular the fact that responsibility for SD is mainly assigned to senior management can facilitate and accelerate processes such as embedding SD in the mission statement (\rightarrow *politics*), especially when it is paired with a whole-institution approach (\rightarrow *organization*, *profession*). However, in terms of long-term prospects, the success of such a top-down process depends on whether HEI members are willing and able to commit to a mission statement that they were not involved in developing. In order to bring a top-down strategy to life (\rightarrow *politics*, *organization*, *public*), it may be necessary either to allocate additional personnel specific responsibility for achieving the objectives or to re-organize responsibilities, because it is unlikely that staff will have

too much time for unexpected tasks. Otherwise, a low *learning orientation* combined with a high *holistic orientation* towards SD might come in handy when it comes to announcing SD principles for the HEI as a whole, but it does not guarantee that such principles will be eagerly adopted.

As described above, the center of expertise has begun to promote an institution-wide conversation on SD (\rightarrow *profession, knowledge*) because of its belief in holistic transformation, but the power to make real changes to fundamental processes still lies with senior management, which does not take things quite so far. The thoughtfully constructed and broadly established center of expertise, which is striving to coordinate a diligent and coherent process (\rightarrow *organization*), may be in danger of becoming a toothless tiger. This might lead to problems for all the equalizer dimensions, since their development seems to depend on that very structure.

4.1.2. Case VIII: Low Holistic and Profound Learning Orientation

HEI VIII is also located in an urban environment. It also has 25,000–50,000 students across 100–200 different study programs and employs up to 5000 people. Sustainability became an issue at this HEI at the beginning of the 1990s, sparked by employees in the administration who demanded that the institution take meaningful steps to reduce its energy consumption, much like case VI. At HEI VIII, this led to a long process of ISO certification and ultimately resulted in the introduction of the Eco-Management and Audit Scheme (EMAS). Several small sustainability teams were entrusted with the development of energy saving measures at the departmental level; these had a great impact and helped the whole process to evolve.

Nowadays, there is a staff unit for sustainability and energy with approximately ten employees, who are recognized as important drivers of the sustainability process. The concept of sustainability has broadened over time. Through, inter alia, the engagement of a student initiative, the focus has been shifted from campus management resource concerns towards a comprehensive approach including research, teaching and outreach. A research center acted as the key catalyst for sustainable development in research, generally accessible courses were established with a clear relevance for sustainability, and participation in sustainability networks was increased.

As stated above, HEI VIII is an institution with a relatively low *holistic orientation* and a relatively high *learning orientation* towards SD. Therefore, to some extent, it represents the reverse set-up of HEI VI. A closer look at the governance equalizer dimensions for HEI VIII, as summarized in Figure 6, might help to pinpoint the possible effects of the different cultural environment that is indicated on the left side of the figure.

Throughout the five dimensions, the high *learning orientation* seems to have had an impact that is manifested in a collaboratively developed mission statement (\rightarrow *politics, organization, profession, public*) and also in the way the staff unit has been arranged (\rightarrow *organization, knowledge*). It has close links with senior management but is also working with sustainability teams in the HEI's different areas (\rightarrow *profession*). Responsibility for SD is spread between them and not completely centralized, which certainly increases public attention for the topic (\rightarrow *public*). The HEI's *learning orientation,* and specifically its conception of its own role in society as an SD educator, has enabled the development of new concepts of knowledge transfer (\rightarrow *knowledge, public*).

The main gap highlighted by the governance equalizer is insufficient coherence and shared understanding amongst HEI members with regard to SD (\rightarrow *profession*). Against the backdrop of a long tradition of ecological optimization and the ongoing focus on energy and resources in the staff unit, it seems logical that, despite collaborative processes, HEI members are irritated by aspects of the institution's concept of sustainability because the *holistic orientation* is somewhat limited. Nonetheless, the HEI has been able to establish solid overarching governance structures.

SD Cultural Orientations	Governance Equalizer Dimensions
 Learning orientation: The HEI pursues the idea of shared responsibilities, values the students as a great driving force and has identified actors in a range of positions as being part of a collective SD process, has established a steering committee that helps to bring together the different views, expectations and experiences, recognizes senior management commitment and support, but focuses more on the broad mass of HEI members, stresses its role in society and aims for SD especially through its teaching activities. relatively high expression 	Politics The activities are well supported by the HEI's senior management. Despite the fact that sustainability is not regarded as the key aspect of the institution's profile or a core issue, the institution has put in place environmental guidelines and a sustainability mission statement. The large staff unit and EMAS attest to a certain sense of responsibility for the SD process. Hence, the equalizer dimension politics is rather well developed in case VIII. \rightarrow rather strong expression
	Profession This dimension is not as well established. The mission statement was developed by means of a collaborative process and the major student initiative for SD manages to bring together representatives of the different fields of activity. However, HEI members complain about a lack of shared goals or concepts, and the lack of a common language between the disciplines. Given the starting point of the institution's sustainability process, aspects of climate change mitigation and resource savings still receive more attention than other aspects of SD. \rightarrow medium expression
	Organization The HEI has set up solid and effective structures that provide the engaged actors with agency and room for proactive process design. Through arrangements such as the sustainability teams and a steering committee mean that the needs of the individual fields in the HEI can be identified and solutions found. It is only the derivation of measures from the mission statement that has been insufficient. → rather strong expression
 Holistic orientation: The HEI focuses more on ecological aspects of 5D than others due to its history, but is aiming for a more comprehensive approach, has a staff unit for sustainability and energy which is eager to broaden the process and is supported by a steering committee. → relatively low expression 	Knowledge Knowledge and the public are both dimensions that have thus far had medium to low status at HEI VIII so far. Knowledge is evident in a diverse range of experience and theoretical knowledge, found mainly at the staff unit and the research center. However, systematic bundling, preparation and provisioning have not been arranged. Knowledge transfer is enabled through projects such as a program for high school students about sustainability and climate change, and the establishment of a sustainability research network. → medium expression
	Public The institution generally addresses its public is through publicity for events (posters etc.) undertaken by the different actors and through its website, but response levels leave room for improvement. Thanks to the staff unit's affiliation to HEI senior management, the structure is quite well-known. \rightarrow low to medium expression



4.1.3. Comparison of HEIs VI and VIII

The strongest equalizer dimensions for HEIs VI and VIII are *politics* and *organization*. *Profession*, *knowledge* and *public* are all developed to a medium level at both HEIs. At first sight, the different organizational cultures do not appear to have had a particular impact on HEI sustainability governance. However, as the individual case descriptions have shown, the HEIs' SD structures vary with regard to some important details. This is most prominently expressed in the relations identified between the main SD actors and the potential for these relations to be shaped by the culture of the organization (especially its *learning orientation*).

HEI VI with its center of expertise and HEI VIII with its staff unit have set up structures that aim to introduce SD measures into all fields of activity. They both have close links with senior management and are equipped with roughly the same number of staff. However, there are differences in their strategic alignment and integration, i.e., the way functions are organized as shown in the equalizer. These relate to three major aspects, which seem to go hand in hand with the HEIs' cultural orientations: the manner of their outreach into the institution, their degree of autonomy and their conceptual focus.

By 'manner of outreach', we mean the strategies adopted by the two bodies that might help them to gain acceptance for (common) SD activities in the HEI's community. The idea that responsibility for SD should be spread across several pairs of shoulders (\rightarrow *attribution of responsibility*, as part of the *learning orientation*), and that these shoulders need to have some level of autonomy, seems to have driven HEI VIII towards more open and extensive structures. Certainly, the staff unit is the team with primary responsibility; it is also dependent on senior management decisions. But through the sustainability teams in different departments, a unifying steering committee and a good connection with the biggest and highly active student and staff sustainability initiative, they have the potential to shape a coherent and collaborative process that receives a broad welcome. The institution's *learning orientation* has thus enabled reasonable structures in terms of *profession, organization*, and potentially *public*.

This is seemingly less the case in HEI VI, where the working groups are an integral part of the center of expertise, are made up of appointed members and, therefore, act less independently. The structure is not bad per se, but it could be argued that greater awareness of and trust in a broader range of perspectives could help with implementation. This is evident at HEI VI in certain situations where senior management demonstrates a lack of appreciation for the achievements of extensive student engagement. The center of expertise, however, is working on overcoming this problem and is already considered a good contact point for SD issues by other HEI members. These efforts might eventually lead to changes in the HEI's cultural orientations. This would provide an example of how governance structures can influence an HEI's cultural orientation over time.

The conceptual focus refers to the aspects of SD that are being addressed by the staff unit or the center of expertise and their underlying *conceptions of sustainability*. Even though both HEIs originally focused on improving ecological rather than other aspects of sustainability, as the cultural orientations show, HEI VI is now applying a multi-dimensional approach more consistently than HEI VIII. This can be linked to the different staff configurations that are responsible for SD, and their backgrounds. On the one hand, the staff unit at HEI VIII has evolved from a staff unit for energy to a staff unit for sustainability and energy and was, therefore, built around—and still is dominated by—an expertise in energy and resource issues. The center of expertise in HEI VI, on the other hand, was conceived as an interdisciplinary team from the beginning and hence has been better able to identify with a multi-dimensional concept of sustainability.

When looking at overall SD governance at both HEIs, however, this head start has not resulted in a long-term advantage or disadvantage for HEI VI. This is doubtless due to a number of factors. It is suspected, though, that the *conception of sustainability*, which is only one part of the *holistic orientation*, might have less influence on SD governance than the second part, the *relevance and scope of organizational change*. At least regarding the initiation of the process, it seems to be more important for an HEI's SD governance to take an institution-wide approach than have a comprehensive conception of sustainability when structures are created, which can happen at any level and in any part of the institution. After all, the whole-institution approach is an aspect that these two HEIs have in common and that so far seems to have guided them in relation to structures. However, comparison of the two cases shows that the learning orientation needs to be factored in when designing the whole-institution approach and that a holistic orientation alone does not lead to all-encompassing structures and processes.

This last aspect suggests that the discrepancies between the two cultural assumptions, which were found only in HEIs VI and VIII out of the sample of eleven, are not only uncommon, but also, to some extent, counterproductive. A more linear evolution in orientation appears more desirable, although it does not necessarily mean an easy route of SD implementation as the comparison of the next two cases will illustrate.

4.2. Cases III and IX: Low Holistic and Learning Orientation vs. Distinct Learning and Holistic Orientation

Cases III and IX are intended to provide insight into how very different stages of cultural orientation affected governance structures at the HEIs in question. Case III had one of the lowest *learning* and *holistic orientations* and case IX one of the highest. Since they both represented cases where the relationship between *holistic* and *learning orientation* was linear, it was deemed of particular interest to investigate how these individual HEIs at the opposite ends of the proposed cultural orientation spectrum translate their cultures into governance measures. Many differences were identified between the two HEIs, but the most distinctive aspect arising from the comparison was their different approaches towards knowledge work in the context of SD as the following outlines illustrate.

4.2.1. Case III: Low Holistic and Low Learning Orientation

HEI III is an urban HEI with 25,000–50,000 students in 100–200 study programs and employs between 5000 and 10,000 people. The sustainability process was only begun in 2013 with a senior management decision to appoint a sustainability commissioner. Prior to that, a very active student initiative had been the only driver of sustainability-related change at an institutional level.

The sustainability commissioner, who is a professor and, therefore, does not have extra time resources, has been given the support of a research assistant and the task of supervising a 5 year sustainability reporting process with the aim of proposing appropriate measures. By the time the interviews took place, the team had almost finished its inventory and was about to start using it to derive strategies. One central idea is the establishment of a Green Office for process stabilization and continuity.

Apart from this central process, different individual activities are taking place. The abovementioned student initiative and the general student committee organize SD events, promote sustainable consumption on campus and search for innovative mobility solutions. A mixed group of HEI members is focused on making the campus bike friendly. As regards teaching, practical interdisciplinary sustainability courses are being offered.

This description of HEI III paints the picture of an HEI that has set a sustainability process in motion with a comparatively low *learning orientation* towards SD and an initially rather one-dimensional and top–down approach. Figure 7 reflects on some further insights into the HEI's cultural orientation and outlines the governance equalizer dimensions.

SD Cultural Orientations	Governance Equalizer Dimensions
 Learning orientation: The HEI follows the idea that senior management and the sustainability commissioner need to structure and guide the way, is mainly concerned with the sustainability reporting team as the centre of the process, consultation with other HEI members is mostly inventory- related (for the sustainability report) 	Politics Sustainability guidelines are being discussed and the Copernicus Charta was signed early on as an official commitment for SD. \rightarrow rather low expression
	Profession Cooperation between different professions is enhanced through the range of interdisciplinary courses on sustainable development and a certificate that students can receive for attending a certain number of these courses. The sustainability reporting team reaches out to all relevant actors for their inventory and slowly builds relationships with them. But the actors are not actively networked with each other and collaboration is rather accidental. The technical department in particular feels rather isolated, but also claims to have no resources to join in any projects. In addition, there is no common understanding of sustainability. \rightarrow medium expression
 has shown appreciation for the general idea of a broader attribution of responsibility and the transformative potential of an HEI but has not internalized and applied it to its own sustainability process. → relatively low expression 	Organization The active SD structures feeding into the organization dimension are the sustainability commissioner and his team who stay in close contact with senior management via regular meetings. The general student committee is engaged as are some of the teaching staff who organize the SD certificate courses. There used to be a quite influential student initiative that has now become independent of the HEL. In terms of external relations, senior management has joined an alliance with other HEI leaderships where concepts and ideas on the HELs' development, also with regard to sustainability, are regularly exchanged. An external sustainability advisory committee is envisaged. → medium to strong expression
 Holistic orientation: The HEI does not network between the different actors, i.e. the technical staff feels rather isolated, has a sustainability commissioner whose expertise lies in the ecological aspects of SD, the whole process has evolved out of that. → relatively low expression 	Knowledge The generation and aggregation of SD knowledge was a focus of the HEI even before senior management started the official process and it has remained the main concern of the reporting team based on the idea that appropriate strategies can only be developed with a good overview of current data and processes. Decentralized events and the sustainability courses offer HEI members the opportunity to debate different options for action on an individual level. → medium to strong expression
	Public The abovementioned events and other activities organized by the student initiative also seem to be the best way of raising public awareness of the HEI's sustainability ambitions. The reporting team spreads the word slowly by contacting individual actors for their inventory. They have not reached out to a broader public. \rightarrow low expression

Figure 7. Overview of the cultural orientations and governance dimensions of HEI III.

HEI III finds itself at a point in its sustainability process that can easily be traced back to its cultural orientation. The *learning orientation* in particular seems to be clearly reflected in the current governance. Senior management and the sustainability commissioner and his team preferred to learn about their institution before they learned with the institution when they decided that a sustainability report should be produced before any other centralized sustainability activities were pursued so that the latter could be based on reliable information (\rightarrow *profession, organization, knowledge*). It is arguable whether this was a wise or even necessary decision, and, in the long run, this will be illustrated by what the HEI eventually makes of the process that it has started, which it has mostly kept in the presidential office. For now, the activities taking place in addition to that official top–down process such as the sustainability certificate that students can gain—organized by some teaching staff, or other student or professional or technical staff-based initiatives—are trying in vain to find allies in their HEI.

Instead of supporting the building of an SD network within the HEI and starting to share responsibilities, senior management prefers to join external sustainability networks and seek alliances with elite representatives from other HEIs. Both are a potential source of valuable knowledge on SD implementation, but the exclusive involvement of senior management underlines once more the HEI's low *organizational learning orientation*. The establishment of a Green Office (a student-led sustainability bureau with HEI funding) may succeed in breaking with this pattern (\rightarrow *organization*). So long as it is not primarily an extension of the senior management but has some autonomy, it could be a good way to open up the process to the whole HEI community (\rightarrow *profession*, *public*).

Taking the HEI's holistic orientation into consideration, it has certainly had an impact on the ongoing process. The HEI does intend to apply a multi-dimensional approach to sustainability and to integrate it into its different fields of activity, but this concept was formulated only relatively recently. Besides the commissioner and his team, who are responsible for SD across the whole institution (\rightarrow organization), there are hardly any activities that indicate a comprehensive approach is being taken (\rightarrow politics). Due to his area of expertise, the sustainability commissioner mainly takes content-based responsibility for ecological topics. In this case, this has hardly any effect on the process within the HEI as a whole, as there has thus far been virtually no contact with the technical staff of the HEI (\rightarrow profession). The relatively low learning orientation is, therefore, preventing the equally low holistic orientation at HEI III from having an impact on procedures.

4.2.2. Case IX: High Holistic and High Learning Orientation

HEI IX is the only rural HEI in the selection. It has less than 1000 employees, between 5000 and 10,000 students and offers study programs in a narrow course spectrum. In 2013, some committed individuals at HEI IX started submitting requests to senior management for the HEI to promote sustainability. Some responsibility for SD was created in the form of target agreements between the HEI and the federal state. Institution-wide workshops in 2015 introduced further ideas about a sustainable HEI and were followed by the appointment of a sustainability commissioner, also a professor, who was provided with some personnel resources. The sustainability commissioner coordinates a SD working group that has prepared sustainability guidelines that were discussed in another institution-wide workshop including senior management and are being processed by the HEI senate.

The working group, which has a good mix of HEI members, and a student initiative host small events and participate in national sustainability action days. On the administrative side, sustainability considerations have impacted on a number of procurement decisions, and connections have been made between SD and the HEI's policies (e.g., gender equality, family friendliness, inclusion).

HEI IX has the most concise orientation of the four towards sustainability. Figure 8 provides more detailed information on both the cultural orientation and the governance equalizer. The interrelationship between the two is discussed below.

Interestingly, the governance equalizer does not actually portray HEI IX as having advanced sustainability structures and processes, as the organization's culture has suggested. However, that does not necessarily mean that the cultural orientations did not have an impact. Closer examination reveals that the structures and processes that have been implemented or are about to be implemented, especially in relation to *politics* and *organization*, do reflect a relatively high *holistic* and *learning orientation*.

The understanding that actors from all the different groups of stakeholders and fields of activity should have the opportunity to participate in the HEI's sustainability process is clearly expressed in the way the overall process has been executed so far. Several institution-wide workshops were held, which, among other things, resulted in common sustainability guidelines (\rightarrow *politics*, *profession*, *public*). The sustainability commissioner coordinates an open SD working group (\rightarrow *organization*), and student involvement is much appreciated, albeit difficult to maintain. From the very beginning, sustainability was considered to be a multi-dimensional concept, and this is well reflected in the guidelines. The steps may be small, but they set the tone for further development of the HEI's sustainability governance. This is especially true for the *knowledge* dimension, the least prominent equalizer dimension at HEI

IX. This case indicates that participatory processes not only take time, but also require staff resources, with which HEI IX is the least equipped out of the four cases presented here. Apart from that, this HEI certainly has some advantages—shorter distances, a smaller range of disciplines—which facilitate some processes, since it is by far the smallest of the four HEIs.

SD Cultural Orientations	Governance Equalizer Dimensions
 Learning orientation: The HEI has so far undergone an organic and collaborative evolution of SD structures and goals, has established a working group and hosted institution-wide strategy workshops; both support the exchange of view and mutual learning for sustainability, is actively engaged with regional actors and influences 	Politics Sustainability is certainly identified as a relevant issue on the political agenda of HEI IX. The official approval of sustainability guidelines by the senate is underway and target agreements with the federal ministry offer an important framework for initiatives. However, there are complaints about a lack of a clear strategy. → medium expression
	Profession The working group serves as an effective network for people from different fields of activity, and is yet characterized by a variety of individual definitions of sustainability and their application for the HEL SD is encountered in different ways by members of the different disciplines, especially in the context of teaching and research. Nevertheless, there are a number of (practical) seminars with a sustainability focus that are accessible for all students and often provided in collaboration with external partners. The HEI's narrow range of disciplines generally seems to favor more direct exchanges. → medium to strong expression
regional development through various projects. → relatively high expression	Organization A clear course has been set with regard to structure. Further development and implementation are needed. Offers to participate are increasingly recognized and accepted, but student involvement is in question since several key players are leaving the HEL \rightarrow medium expression
 Holistic orientation: The HEI has appointed a sustainability commissioner with an interdisciplinary background and is taking a holistic and collaborative approach to SD, considers all its fields of action as equally relevant to SD. → relatively high expression 	Knowledge This dimension still needs to be strengthened. A considerable amount of research and teaching at HE11X relates to sustainability issues and is constantly producing new Knowledge, but the transfer of this to the HEI itself has not yet been regulated. For now, sustainability actions are based on single, somewhat unrelated ideas rather than systemic priorities. \rightarrow rather low expression
	Public There is a desire for greater public attention for, and knowledge of, the sustainability process particularly from students. There are activities aiming, inter alia, to give the issue greater prominence – such as the strategy workshops, the online presence, a lecture series on sustainability and a sustainability guide produced by the student initiative. Nonetheless, sustainability, or at least the awareness for the HEI's efforts, has by far not reached every corner of the institution. \rightarrow low to medium expression

Figure 8. Overview of the cultural orientations and governance dimensions of HEI IX.

4.2.3. Comparison of HEIs III and IX

At first sight, there are some commonalities in the two institutions' sustainability processes. They started around the same time after small groups or individual HEI members became engaged with sustainability. Once they had been successful in putting sustainability on the HEI's agenda, the senior management team in question decided to appoint a sustainability commissioner to be responsible for coordinating and overseeing the whole process. What appears to be quite different, though, is the conception of how best to set up the process. This is where the cultural orientations divide the two HEIs.

Although in the case of both HEIs, it has been a similarly short time since they initiated their sustainability processes, they have already adopted very different positions. It can be assumed that their cultural orientations have played a role in organizational development towards sustainability. HEI III went straight to action with a clear vision and introduced structures that should form the basis for a well-founded process. From an operational perspective, this makes sense, but the essence of sustainability seems to have been neglected or not fully internalized yet. This means that sustainable development, with its various substantive and methodological components, has either not been fully explored or has been reduced to distinct individual aspects. Even though the will to implement a multi-dimensional understanding of sustainability in all areas of the institution (\rightarrow *holistic orientation*) is generally present, the HEI is still faced with the challenge of transferring this will to the current elite and discrete structures and thus developing them further to make them more accessible. As was the case at HEI IX, the HEI's *organization* and *profession* dimensions would certainly benefit from such a concept.

HEI IX started with somewhat less concrete activity, but active stakeholder participation ensured that there was broad agreement with the central decision by senior management to appoint a sustainability commissioner. This openness signals to the members of the HEI that their engagement is wanted and that it can shape important aspects of SD. This also makes it easier to replicate the organizational culture amongst HEI members and encourage it to evolve collectively. The working group has promoted regular interdisciplinary exchange and collaboration (\rightarrow organizational learning orientation). Nonetheless, there is one equalizer dimension that has been developed notably further at HEI III than HEI IX: *knowledge*.

Maybe unexpectedly, this is also where the differences between the ways the HEIs' governance has been shaped by their cultural orientations are most obvious. The HEIs seem to apply quite different approaches towards knowledge and knowledge work in the context of SD. HEI III, with its focus on knowledge generation, has introduced a structure, namely the sustainability reporting team, which is efficient at gathering data and identifying important contributors for the report. Their knowledge work is a highly centralized and isolated process that does not open up to the potential for common exploration with diverse stakeholders on diverse topics, which is the approach taken by HEI IX. The combination of low holistic orientation and low learning orientation leads HEI III to a very target-oriented, and in that sense efficient, knowledge dimension that, however, will inevitably reach its limits when all information has been gathered. High holistic and learning orientations have helped HEI IX to lay groundwork that gives knowledge the chance to expand at a variety of levels. The downside of this approach is certainly that it takes much more effort in relation to proactive stakeholder engagement and coordination and, therefore, has not yet resulted in a strong knowledge dimension for HEI IX. This comparison of knowledge work captures the distinct styles of SD implementation at both HEIs and corresponds well with the aforementioned aspects. It also highlights the different qualities that can be found within a single equalizer dimension and encourages an integrative approach towards the governance equalizer when introducing governance measures.

Overall, neither of the two HEIs has addressed any of the equalizer dimensions in an all-encompassing way. They have made approximately the same amount of progress, if that is even comparable, and are now at a point where the implementation of concrete and significant measures is required. Their SD processes are relatively young and will need some more time to unfold. It remains to be seen whether they will manage to either seize or change their current functions and culture.

5. Discussion and Conclusions

The comparisons between HEIs VI and VIII and HEIs III and IX have revealed various ways in which the cultural orientations of HEIs are expressed in the functionalities of their SD governance. In each of the individual cases, it was in principle possible to understand the relationship between these two factors. But what general conclusions can be drawn from these observations?

It is certainly a central finding that the cases not only demonstrate an interplay between cultures and functions, but also that the individual cultural orientations interlock during the implementation of structures, etc., and usually produce a common result. It is unlikely to be possible to cite single cultural orientations, be they *holistic* or *learning*, that have clearly and directly affected and shaped individual governance dimensions. This is not least due to the fact that neither cultures nor governance dimensions can always be clearly separated, nor should they be during the consideration of governance measures. Ultimately, HEI sustainability processes are no 'optimized production lines'; they are instead complex social processes that are both, resulting from and leading to social innovations (e.g., [16,27,34]). This is precisely why consideration of organizational cultures is so relevant and why it is worth attempting to identify patterns hampering or supporting fundamental change.

Another essential aspect that emerges from the comparative case analysis is the observation that cultural orientations do not become visible instantaneously and simultaneously in all corners of an HEI. This is in line with approaches of organizational research according to which HEIs may be considered as what is labelled "loosely coupled systems" in which autonomy and cohesion apply at the same time
which allows for testing novel solutions without reaching out to the whole institution right from the beginning [35]. Particularly in large HEIs, there may be "communities within communities" that do not interact as intensely with each other [36] so that a cultural change in one sector does not immediately apply for another one. Apart from this theoretical consideration, the challenge of grasping the cultural orientation of HEIs is partly due to the underlying methodology and material used. For the study reported here, we subjected existing interview material, i.e., material collected for a different purpose, to secondary analysis. Therefore, we did not have access to all the information that would have been desirable to enable the cases to be explored in greater depth. The cultural orientations described here need time to develop and to influence governance structures. This also brings into play what has already been suggested by some of the four cases, namely the possibility that structures and processes implemented may in turn have an impact on cultural orientations, thus giving rise to a feedback loop as part of a living process, as Disterheft et al. [10] have also pointed out with regards to participatory processes in particular.

The analysis has shown, in some cases, that the most direct path from cultures to functions was via the elements assigned to *politics* and/or *organization*. For at least three of the HEIs considered (III, VI and VIII), these governance equalizer dimensions were initially in the foreground. It is quite possible that this reflects a pattern that also appears in other cases from the overall sample. However, there are certainly other ways for SD to get into the HEIs' bloodstreams. For instance, informal and unofficial but significant and profound engagement for SD by a group of actors might favor the development of a collaborative network of HEI members before it seeks to become institutionalized [37]. It, therefore, remains to be seen how successful the approaches will prove to be in the long run and what advantages and disadvantages they each offer. A broader examination of the whole study sample, ideally in combination with a second survey at the same HEIs, could provide insight into this.

This exploratory study, based on a secondary analysis of existing interviews, has identified interesting starting points for further research and, building on Adams et al. [16] and Baker-Shelley et al. [17], has also expanded the understanding of the importance of organizational cultures for sustainable development in HEIs. Real-life examples were used to illustrate potential interdependencies and to draw some general conclusions about how the cultures and functions of HEI governance relate to each other. In particular, it has increased the awareness of the diversity of HEIs' SD processes. No matter how similar some structures may appear, it is not uncommon for details that initially appear inconspicuous to have a significant impact on their success. Both comparisons have made this clear. What was also implied at some points was the need to recognize further factors that influence an HEI's sustainability governance beyond its cultural orientation. Aspects such as the size of the HEI or regional (political, legal and socio-economic) circumstances certainly also need to be factored in and might show some of the SD processes discussed in another light. Some of these aspects are also very likely to have shaped the depicted organizational cultures. However, this opens a new discussion that requires further investigation and exceeds the material of this study.

Overall, the frameworks taken from Bauer et al. [2] and Niedlich et al. [19] have proven themselves to be useful analytical tools. The categories developed provided valuable guidance in the midst of the complexities of HEI SD governance. This combined analysis, enriched by the four case studies, identified some fine mechanisms that can take place at the interplay of culture and functions. At the same time, it enabled a new reflection on the original frameworks themselves. Future research can take this into consideration and also explore other ways of applying them.

In view of the small selection of cases, which is also limited to German HEIs involved in a joint project, the limits of generalizability are quickly reached. The interpretative methodology also imposes certain limitations on the results here. Further investigation on the basis of a larger sample is required in order to verify the findings. Nevertheless, the approach adopted in this paper contributes to the development of instruments that can be applied in practice, e.g., for self-reflection and self-assessment by HEIs (see [38]). It could also serve to further differentiate and operationalize the cultural dimensions for the purpose of standardized surveys or be applied to the development of counselling concepts.

Bearing the initial research question of this paper in mind, it can be stated that an HEI's organizational culture is highly likely to affect its approach to SD governance in crucial ways. The case studies demonstrated essential connections between cultural orientations and the governance approaches of HEIs, as illustrated by the governance equalizer. Many of these connections relate to the specific circumstances of the HEIs in question, i.e., with regard to the individuals involved. Although there will be no other HEI that functions in exactly the same way as one of these four, the perspective offered by this combined analysis of cultures and functions of SD governance at HEIs offers should make it easier to identify trends.

Author Contributions: Conceptualization, M.B., S.N., M.R., I.B. and L.J.; methodology, M.B., S.N., M.R. and I.B.; validation, M.B., S.N., M.R., I.B. and L.J.; formal analysis, M.B., S.N., M.R., I.B. and L.J.; writing—original draft preparation, M.B.; writing—review and editing, M.B., S.N., M.R., I.B. and L.J.; visualization, M.B., S.N., M.R. and I.B.; supervision, M.R. and I.B.; project administration, S.N., M.B. and M.R.; funding acquisition, M.R. and I.B. All authors have read and agree to the published version of the manuscript.

Funding: This research was funded by the Bundesministerium für Bildung und Forschung, Grant/Award Number: 13NKE007A.

Acknowledgments: We thank four anonymous reviewers for their feedback on an earlier version of this paper.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Rinaldi, C.; Cavicchi, A.; Spigarelli, F.; Lacchè, L.; Rubens, A. Universities and smart specialisation strategy: From third mission to sustainable development co-creation. *Int. J. Sustain. Higher. Educ.* 2018, 19, 67–84. [CrossRef]
- Leal Filho, W.; Wu, Y.-C.J.; Brandli, L.L.; Avila, L.V.; Azeiteiro, U.M.; Caeiro, S.; Madruga, L.R. Identifying and overcoming obstacles to the implementation of sustainable development at universities. *J. Integr. Environ. Sci.* 2017, 14, 93–108. [CrossRef]
- 3. Bauer, M.; Bormann, I.; Kummer, B.; Niedlich, S.; Rieckmann, M. Sustainability Governance at Universities: Using a Governance Equalizer as a Research Heuristic. *High. Educ. Policy* **2018**, *31*, 491–511. [CrossRef]
- Awuzie, B.O.; Abuzeinab, A. Modelling Organisational Factors Influencing Sustainable Development Implementation Performance in Higher Education Institutions: An Interpretative Structural Modelling (ISM) Approach. Sustainability 2019, 11, 4312. [CrossRef]
- Weisser, C.R. Defining sustainability in higher education: A rhetorical analysis. *Int. J. Sustain. Higher Educ.* 2017, 18, 1076–1089. [CrossRef]
- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]
- Rath, K.; Schmitt, C.T. Sustainability at Universities: Degrees of Institutionalization for Sustainability at German Higher Education Institutions—A Categorization Pattern. In *Handbook of Theory and Practice of Sustainable Development in Higher Education: Volume 1*; Leal Filho, W., Brandli, L., Castro, P., Newman, J., Eds.; Springer International Publishing: Cham, Switzerland, 2017; pp. 451–470. ISBN 9783319478678.
- Leal Filho, W. Education for Sustainable Development in Higher Education: Reviewing Needs. In *Transformative Approaches to Sustainable Development at Universities*; Leal Filho, W., Ed.; Springer International Publishing: Cham, Switzerland, 2015; pp. 3–12. ISBN 978-3-319-08836-5.
- 9. Mader, C.; Scott, G.; Abdul Razak, D. Effective change management, governance and policy for sustainability transformation in higher education. *Sustain. Account. Manag. Policy J.* **2013**, *4*, 264–284. [CrossRef]
- Spira, F.; Tappeser, V.; Meyer, A. Perspectives on Sustainability Governance from Universities in the USA, UK, and Germany: How do Change Agents Employ Different Tools to Alter Organizational Cultures and Structures? In *Sustainability Assessment Tools in Higher Education Institutions: Mapping Trends and Good Practices around the World*; Caeiro, S., Leal Filho, W., Jabbour, C., Azeiteiro, U.M., Eds.; Springer: Cham, Switzerland, 2013; pp. 175–187. ISBN 978-3-319-02374-8.

- Shriberg, M.P. Sustainability in U.S. Higher Education: Organizational Factors Influencing Campus Environmental Performance and Leadership. Ph.D. Thesis, University of Michigan, Ann Arbor, MI, USA, 2002.
- 12. Disterheft, A.; Caeiro, S.; Azeiteiro, U.M.; Filho, W.L. Sustainable universities a study of critical success factors for participatory approaches. J. Clean. Prod. 2015, 106, 11–21. [CrossRef]
- Scott, G.; Tilbury, D.; Sharp, L.; Deane, E. Turnaround Leadership for Sustainability in Higher Education. Executive Summary; Project Report. 2012. Available online: http://sustainability.edu.au/material/teaching-materials/ turnaround-leadership-sustainability-higher-education-olt-funded-project-and-report/ (accessed on 28 February 2020).
- 14. Moore, J.; Pagani, F.; Quayle, M.; Robinson, J.; Sawada, B.; Spiegelman, G.; van Wynsberghe, R. Recreating the university from within. *Int. J. Sustain. Higher Educ.* 2005, *6*, 65–80. [CrossRef]
- 15. Cameron, K.S.; Quinn, R.E. Diagnosing and Changing Organizational Culture. Based on the Competing Values Framework, Rev. ed.; Jossey-Bass: San Francisco, CA, USA, 2006; ISBN 978-0-7879-8283-6.
- Baker-Shelley, A.; van Zeijl-Rozema, A.; Martens, P. A Conceptual Synthesis of Organisational Transformation: How to Diagnose, and Navigate, Pathways for Sustainability at Universities? J. Clean. Prod. 2017, 145, 262–276. [CrossRef]
- 17. Adams, R.; Martin, S.; Boom, K. University culture and sustainability: Designing and implementing an enabling framework. *J. Clean. Prod.* **2018**, *171*, 434–445. [CrossRef]
- Sterling, S. Higher Education, Sustainability, and the Role of Systemic Learning. In *Higher Education and* the Challenge of Sustainability: Problematics, Promise, and Practice; Corcoran, P.B., Wals, A.E.J., Eds.; Kluwer Academic Publishers: Dordrecht, The Netherlands, 2004; pp. 49–70.
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, 1–18. [CrossRef]
- Wu, Y.-C.J.; Shen, J.-P. Higher education for sustainable development: A systematic review. Int. J. Sustain. High. Educ. 2016, 17, 633–651. [CrossRef]
- Sylvestre, P.; Wright, T.; Sherren, K. Exploring Faculty Conceptualizations of Sustainability in Higher Education: Cultural Barriers to Organizational Change and Potential Resolutions. *J. Educ. Sustain. Dev.* 2013, 7, 223–244. [CrossRef]
- Hoover, E.; Harder, M.K. What lies beneath the surface? The hidden complexities of organizational change for sustainability in higher education. J. Clean. Prod. 2015, 106, 175–188. [CrossRef]
- Singer-Brodowski, M.; Etzkorn, N.; von Seggern, J. One Transformation Path Does Not Fit All—Insights into the Diffusion Processes of Education for Sustainable Development in Different Educational Areas in Germany. *Sustainability* 2019, 11, 269. [CrossRef]
- Leal Filho, W.; Shiel, C.; Paco, A.; Mifsud, M.; Veiga Ávila, L.; Brandli, L.L.; Molthan-Hill, P.; Pace, P.; Azeiteiro, U.M.; Ruiz Vargas, V.; et al. Sustainable development goals and sustainability teaching at universities: Falling behind or getting ahead of the pack? J. Clean. Prod. 2019, 232, 285–294. [CrossRef]
- Holm, T.; Sammalisto, K.; Grindsted, T.S.; Vuorisalo, T. Process framework for identifying sustainability aspects in university curricula and integrating education for sustainable development. J. Clean. Prod. 2015, 106, 164–174. [CrossRef]
- Davis, H.; Goedegebuure, L. Governance for sustainability in higher education. In *Ethics in Higher Education:* Values-Driven Leaders for the Future; Singh, D., Stückelberger, C., Eds.; Globethics.net Education Ethics 1; Globalethics.net: Geneva, Switzerland, 2017; pp. 217–230. Available online: https://www.globethics.net/ documents/4289936/20368389/GE_Education_Ethics_1_isbn9782889311644.pdf (accessed on 31 March 2020).
- Trencher, G.; Yarime, M.; McCormick, K.B.; Doll, C.N.H.; Kraines, S.B. Beyond the third mission: Exploring the emerging university function of co-creation for sustainability. *Sci. Publ. Policy* 2014, *41*, 151–179. [CrossRef]
- De Boer, H.; Enders, J.; Schimank, U. On the Way towards New Public Management? The Governance of University Systems in England, the Netherlands, Austria, and Germany. In *New Forms of Governance in Research Organizations: Disciplinary Approaches, Interfaces and Integration;* Jansen, D., Ed.; Springer: Dordrecht, The Netherlands, 2007; pp. 137–152. ISBN 978-1-4020-5830-1.

- Franz, K.; Brüsemeister, T. Gegenstandsbezogene Analyse sozialer Welten des BNE-Transfers. In Governance-Regime des Transfers von Bildung für Nachhaltige Entwicklung; Bormann, I., Hamborg, Eds.; Springer Fachmedien Wiesbaden: Wiesbaden, Germany, 2016; pp. 67–88. ISBN 978-3-658-13222-4.
- Morse, J.M. Designing funded qualitative research. In *Handbook of Qualitative Research*, 4th ed.; Denzin, N.K., Ed.; SAGE: Thousand Oaks, CA, USA, 1994; pp. 220–235.
- 31. Dresing, T.; Pehl, T. Praxisbuch Interview, Transkription & Analyse. Anleitungen und Regelsysteme für Qualitativ Forschende, 8th ed.; Dr. Dresing und Pehl GmbH: Marburg, Germany, 2018.
- 32. Kuckartz, U. Qualitative Inhaltsanalyse. Methoden, Praxis, Computerunterstützung; Beltz-Juventa: Weinheim, Germany, 2012; ISBN 9783779928157.
- Kyburz-Graber, R. Case Study Research on Higher Education for Sustainable Development. Epistemological Foundation and Quality Challenges. In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Rieckmann, M., Thomas, I., Eds.; Routledge: Abingdon, UK; New York, NY, USA, 2016; pp. 126–141.
- 34. Howaldt, J. New Pathways to Social Change—Creating Impact through Social Innovation Research. In Proceedings of the Conference "Impact of Social Sciences and Humanities for a European Research Agenda Valuation of SSH in mission-oriented research", Vienna, Austria, 28–29 November 2018; pp. 37–48.
- Orton, J.D.; Weick, K.E. Loosely Coupled Systems: A Reconceptualization. Acad. Manag. Rev. 1990, 15, 203–223. [CrossRef]
- Gurbutt, D.J.; Cragg, R.M. A community within a community. In *Culture, Change and Community in Higher Education: Building, Evolving and Re-Building Learning Environments*; Gurbutt, D.J., Cragg, R.M., Eds.; Routledge: Abingdon, UK; New York, NY, USA, 2019; pp. 69–76. ISBN 0367183447.
- 37. Schopp, K.; Bornemann, M.; Potthast, T. The Whole-Institution Approach at the University of Tübingen: Sustainable Development Set in Practice. *Sustainability* **2020**, *12*, 861. [CrossRef]
- Niedlich, S.; Bauer, M.; Doneliene, M.; Jaeger, L.; Rieckmann, M.; Bormann, I. Assessment of Sustainability Governance in Higher Education Institutions—A Systemic Tool Using a Governance Equalizer. *Sustainability* 2020, 12, 1816. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article The Whole-Institution Approach at the University of Tübingen: Sustainable Development Set in Practice

Kerstin Schopp ^{1,*}, Matthias Bornemann ^{1,*} and Thomas Potthast ^{1,2}

- ¹ International Centre for Ethics in the Sciences and Humanities (IZEW), University of Tübingen, Wilhelmstr. 19, 72074 Tübingen, Germany; potthast@uni-tuebingen.de
- ² Chair for Ethics, Theory, and History of the Life Sciences, University of Tübingen, Wilhelmstr. 19, 72074 Tübingen, Germany
- * Correspondence: kerstin.schopp@uni-tuebingen.de (K.S.); matthias.bornemann@uni-tuebingen.de (M.B.)

Received: 17 December 2019; Accepted: 21 January 2020; Published: 23 January 2020

Abstract: In the following paper, we scrutinize understandings and values behind Sustainable Development (SD) in a case study of the University of Tübingen, Germany. In so doing, we adopt the perspective of the whole-institution approach of SD. We do not only analyze documents, but combine our investigations with empirical research on key actors' understandings and values of SD, as well as the competencies and the knowledge to set SD in practice. First, we demonstrate that actors' understandings and the values behind them at the University of Tübingen are in accord with the University of Tübingen, many actors already work in line with the whole-institution approach; this shall be further fostered and strengthened by the Competence Centre for SD. Finally, we demonstrate that both knowledge and competencies are fundamental to act for SD. It is suggested that the University of Tübingen should explicitly adopt the general understanding of SD in the above-mentioned sense, and develop a sustainability strategy, not least in order to support the actors to acquire specific knowledge to reach SD for the whole university. Finally, we discuss the potential and limits of transferring the findings to other Higher Education Institutions (HEI) and the challenges of necessary global perspectives.

Keywords: whole-institution approach; sustainable development; higher education institutions; competencies; knowledge; values; case study

1. Introduction: Topic and Structure of the Article

"Why should I be studying for a future that soon may be no more, when no one is doing anything to save that future?"

(Greta Thunberg, climate activist) [1]

Already in 2009, the German University Presidents' Conference (HRK) decided to implement Sustainable Development (SD) in the fields of research and knowledge transfer, education and teaching and the institution, and to foster Education for SD [2] (p. III). In the same year, mainly initiated by a student initiative, measures for establishing SD at the University of Tübingen, Germany were initiated, e.g., a study program for SD and preparations for an EMAS-certificate (European Management and Audit Scheme). In 2013, the University set up activities aiming at establishing a Competence Centre for Sustainable Development. From its outset, the Competence Centre's main goal is to address SD in a whole-institution approach (see Section 2.2).

The Competence Centre organizes and realizes a wide range of activities across the university, such as network meetings with university staff, and students interested in and working on topics of SD. The Competence Centre's staff regularly evaluates these activities. Therefore, we, as members of the

Centre, already experienced that members of the University of Tübingen have different understandings of what SD is, as well as of how exactly it should be implemented. The Competence Centre is a network node which enables actors to act for SD and, therefore, needs to know the actors' different expectations and needs. Only then will the Centre be able to provide them with the various resources adapted to their needs. Since the different understandings of SD, as well as the knowledge and the competencies of actors at the University of Tübingen differ, it is obviously difficult to plan and implement further activities in the field of SD for the Centre.

Hence, as a reaction to this situation, we have started the research project "SD@UT" which pursued the following research questions:

- 1. What is the understanding of SD of actors, who are involved in different fields of SD at the University of Tübingen?
- 2. How and where shall SD be implemented at the University of Tübingen?
- 3. What kind of knowledge and which competencies do actors need to implement SD in the areas of operations, research, governance, transfer, teaching and education, and sustainability reporting?

In the following, we will present the findings of this research project, structured as follows: First, we will briefly introduce different understandings of SD typically found in the scientific discourse. We will then outline the concept of the whole-institution approach and link it to the German research project HOCH-N, which identifies governance, sustainability reporting, teaching and education, research, operations, and transfer as relevant fields of a Higher Education Institution (HEI) [3]. Thereby, we also consider the Competence Centre's role at the University of Tübingen, its goals as well as its past and future work endeavors.

In the second step, we dive into the Competence Centre's research project that we have briefly introduced above, by considering its starting point.

In the final section, we will discuss options and opportunities to transfer our research results to other (inter)national universities. We will also discuss the need to integrate perspectives of the Global South into these conceptions and explain further research possibilities and needs.

2. State of the Art

In this chapter, we first present several levels and understandings of SD, followed by an overview of the whole-institution approach, as well as the research project HOCH-N's fields of actions [3]. Both form a basis for the Competence Centre's whole-institution approach, which we present in the final part of this chapter.

2.1. Understandings of Sustainable Development

The idea and concept of SD is present in different contexts and on several levels [4] (p. 27). We distinguished four levels and categories, which are, however, only a first classification. These are (i) the political level, (ii) the philosophical level as well as (iii) the personal level. Additionally, there are (iv) models, concepts, and guidelines. We do not want the reader of this paper to understand them separately from each other or to think that they are always redundant, since they smoothly merge into each other. In particular, the boundaries between models, concepts, and guidelines are fluid.

On the political level, the Brundtland Report [5], which was published after four years of work by the World Commission on Environment and Development (WCED), defined SD for the first time in UN history. According to this report, SD "*meets the needs of the present without compromising the ability of future generations to meet their own needs*" [5] (p. 16). Another important political event was the UN Conference on Environment and Development in 1992, which is known as the Earth Summit [6]. It aimed at setting the course for a worldwide SD while taking into account the dependency of human beings on their environment. For this reason, one core element of the Brundtland Report was elaborated: Since SD has to happen within the frame of finite natural resources, the UN brought together environmental and development issues and politics. Out of this idea, finally the eight Millennium Development Goals

(MDGs) [7] were born and adopted in 2000 during the Millennium Summit, and then replaced by the Sustainable Development Goals (SDGs) [8] in 2015.

The philosophical level of SD is connected to the justice-based idea of "the good life", which goes back to the ethics of Aristotle. He argued for the aim of all humans to strive for eudaimonia—happiness or well-being—which requires a good character and, therefore, a sense of responsibility. The philosopher, Hans Jonas, took Aristotle's Eudaimonia, as well as Immanuel Kant's categorical imperative, as basis for his ecological imperative: "*Act in a way that the effects of your action are compatible with the permanence of real human life on earth*!" [9] (p. 36, translated by the authors). These ideas, together with John Rawls' institutional theory of justice [10], surely influenced the Brundtland Report, which deals with inter-and intragenerational justice—the "good life" for everyone at any time.

The personal level is another crucial aspect of SD. However, even though politics often emphasize the importance of people's individual contribution to efforts such as the energy transformation, which shall contribute to SD, the personal level of SD is not yet highlighted in science. Parodi and Tamm state that "one half of the sustainability universe is still mainly unrecognized and unexplored" [11] (p. 1). In times like ours, when pupils demonstrate for their and the world's future, it seems crucial not to forget this dimension of SD, and what an individual can and wants to contribute, as well as how this individual is involved emotionally, culturally or psychologically. Our questionnaires' results demonstrate this level's importance as well (see Section 4.2).

One, albeit inappropriate, model of SD, which is probably best known in German-speaking contexts, is the three-pillar model. According to this model, the economic, environmental, and social domains constitute three equal pillars, which carry the roof of sustainability [12] (p. 13). As mentioned above, models of SD are closely interrelated to concepts and guidelines of SD. These include, for instance, Kopfmüller et al. [13], who developed an integrative concept of SD including its elements, rules, and indicators; Ott & Döring [14], who discussed the concept of a "Strong Sustainability"; Grunwald [15] on the comparison of these concepts; and Ott [16] who strove for embedding the, also in his view, inadequate three-pillar concept of SD into guidelines of Strong Sustainability. Mainly, all of the authors mentioned above, relate their concepts and guidelines to the three-pillar model of SD to improve it further or give it a suitable framework. Nevertheless, we join in the critics of this model with our own critique (see Section 4.2.1).

We point out that the understandings of SD on the different levels can vary within a certain domain. Many actors share some of these understandings; others exist only on an individual level and contradict the well-known ones. Additionally, scholars are divided considering these different meanings. The philosophical level, for instance, has a normative understanding, whereas models are developed to depict something soberly, to make it visible and easy to understand. These multiple understandings build the crucial backdrop against which our research project "SD@UT" (see Section 3) has been conducted. Taking the general inter- and intragenerational justice basis seriously, there are still different understandings and justification levels of SD.

2.2. Understandings of SD of Stakeholders at HEIs

As shown above, there are differing understandings and conceptions of SD in the scientific discourse. When we now focus on a certain group of actors, the question arises, in how far these conceptions are reflected within these groups and which aspects of SD are relevant. There are some international studies asking these questions and concentrating on SD understandings of actors at HEIs [17–23]. Unfortunately, such studies are hard to find in the landscape of German HEIs, and moreover, "there is a shortage of research approaching how faculty and staff perceive their role in relation to sustainability" [22] (p. 46). Almost all of these studies set their focus, more or less, on academic staff or even on teaching staff only. Rarely, other university members, e.g., facilities management directors [20] or university presidents [19] are being interviewed. Almost all of the studies hitherto have been conducted in the context of Education for SD and, therefore, it "is crucial that university faculty and staff have the necessary conditions and competences to provide key SD skills to the students" [22]

(p. 46). Some of these studies also have a focus on evaluation, as they scrutinize the effects of measures (e.g., trainings and workshops) to implement and teach SD within the academic staff [18,21,23].

However, there are similiar findings in these different research approaches. In general, one clear definition of SD does not exist amongst staff members [17] (p. 121), [22] (p. 53). Nevertheless, almost all actors refer to (natural/ecological/environmental) resources and some to (inter- and intragenerational) justice when asked for the meaning(s) of SD [17] (p. 116f.), [18] (p. 229), [19,20] (p. 119f.), [22] (p. 50f.). When first measures of implementing SD at the HEI took place, more reflective aspects, e.g., the values and norms behind SD, are present [21] (p. 34). Consequently, without those measures, some actors have a more superficial understanding of SD, "to keep the concept at a distance and avoid engagement with *it*" [17] (p. 116).

As we will illustrate later (see Section 3.2.), our study has another approach. It examines understandings of SD within the whole of the university and not only amongst lecturers or teaching staff, due to following the idea of the whole-institution approach.

2.3. The Whole-Institution Approach and HOCH-N's Fields of Action

As indicated above, different and complex understandings of SD exist. This complexity is particularly increasing when it comes to applying SD in concrete measures, e.g., in organizations or institutions, as such institutions can themselves already act as societies on a micro-level. At the same time, according to Sibbel, *"it is essential to acknowledge the individual as a part of many social and cultural groups, so this calls for a whole systems approach"* [24] (p. 74) to reach a sustainable society. This whole systems approach combines insights from theories of organizational change as well as livings systems theory [25] (p. 202ff). The idea of organic institutions and structures, of structures facing radical changes while at the same time being highly connected with their surroundings (such as other societal or political actors), is easy to apply to institutions facing SD. This is particularly the case compared to concepts focusing on top-down structures.

Consequently, in 2014, the UNESCO proposed that whole-institution approaches are a method to foster sustainability in learning contexts, because all aspects of the context come in focus and sustainability is implied in all of them [26] (p. 30). The idea of a holistic view on SD in certain institutions is highly linked to Education for SD, which is why this approach is often applied to schools, thereby referred to as a "*whole-school approach*" [26–28]. The process of teaching pupils SD is detached from strict lessons at school and broadened to the school as a whole institution. Therefore, it includes areas such as operations and governance.

In this conception, HEIs function primarily as a place where the educators are educated. A more specialized conception is the "whole-of-university approach" [29]. In transferring the whole-school approach to HEIs, the idea is, again, that a HEI in all its areas has to participate in applying SD. This idea takes into account that the sole education of SD in lectures and seminars is not enough. In contrast to school institutions, the whole-of-university approach adds research as another dimension and stresses the possible links between the different areas: "A whole-of-university approach, however, recognizes that all functions of the institution can benefit from sharing knowledge and that each influences the student learning experience" [29] (p. 57). Notwithstanding, the quote also illustrates some shortcomings in this approach, as the focus point rests on students and their learning of SD.

Authors taking into account the whole-institution approach at the university level equally put their focus on Education for SD (e.g., [30,31]). Our findings are supported by Lozano et al. (2015), who noted a clear focus on topics of education in the scientific literature on "sustainable universities" [32] (p. 4). The same study revealed that members of universities refer to campus operations when asked for SD implementation [32] (p. 9–10). Considering the fact that these respondents know the structures of HEIs, stresses the importance of widening the perspective beyond Education of SD. Therefore, "[f]*uture research could explore in more detail differences between stakeholder groups in HEIs (i.e., students, teaching and non-teaching staff, and relevant external groups)*" [31] (p. 763).

Consequently, the research project HOCH-N follows a different understanding of the whole-institution approach. The research network, which consists of different partner universities in Germany, identified six fields of action where SD can and has to be realized in HEIs. These are sustainability reporting, governance, teaching and education, research, operations, and transfer. These fields of action are highly interconnected and sometimes overlap. Therefore, actors in each field can provide relevant knowledge, experiences, and competencies, and at the same time, profit from relevant knowledge, experiences and competencies of actors from the other fields. For this reason, the focus lies not only on educating students in SD, but on addressing all members of the university as key actors for SD at HEIs. However, implementing a whole-institution approach with its demands of interconnection and holistic outlook is difficult to achieve in practice, even within universities that can be seen as forerunners of implementing SD at HEIs [33] (p. 94). Notwithstanding first trials of pursuing the goal of a sustainable HEI on the systems level via the whole-institution approach [34], there is still a lack of accompanying research studies on preconditions, challenges and outcomes. In our case, at the University of Tübingen, the Competence Centre is working on an implementation of the whole-institution approach and, therefore, has a central role in establishing and fostering this whole-institution approach within the university. We will illustrate the structure and activities of the Competence Centre in the following chapter.

2.4. Knowledge and Competencies to Implement SD at HEIs

When it comes to knowledge resources, which are necessary to implement SD, the Conference of the Swiss Scientific Academies (CASS) [35] refers to the three forms of (i) system knowledge, (ii) target knowledge, and (iii) transformation knowledge. Knowledge on the current situation is referred to as system knowledge. Target knowledge provides information on the condition, which is to be achieved or prevented. Finally, the understanding of transformation knowledge refers to knowledge stocks that indicate the path to the goal [35] (p. 15). Of course, these forms of knowledge are mutually dependent upon another and, at times, are not clearly distinguishable. However, it is obvious that they indicate a possible pathway from the (undesirable) present to the (desirable) future.

With information on the unsatisfying current situation, on how this situation should be, and on how one could reach the goal, system and target knowledge alone will not change the status quo. This is why another resource necessary to act for SD are action-related competencies. We understand "*competencies*" as "*dispositions which individuals need in this environment for acting and self-organisation in various complex contexts and situations*" [36] (p. 129). Accordingly, competencies for SD are competencies which individuals need to, for instance, implement measures to foster SD [37] (p. 205). In order to move from knowledge to application, the actors need competencies to transfer this knowledge into practical actions. In the field of Education for SD, de Haan identifies twelve different competencies and summarizes them under the general term of "*Gestaltungskompetenz*', or 'shaping competence', [which] *means the specific capacity to act and solve problems*" [38] (p. 22). De Haan is one of the main contributors to discourse around Education for SD [39]. However, his approach focuses almost exclusively on the function of HEIs as 'educating the educators'. He does not consider tasks, such as research for SD, which are crucial to HEIs as well.

Authors who also address competencies, but focus on students and (university) courses are Wiek et al. [37], Rieckmann [36], and Lozano et al. [40]. However, with their distinctive perspective, they all remain in the field of Education for SD, even though Wiek et al., as well as Rieckmann, go beyond the 'shaping competence', and Lozano et al. bring together competencies and pedagogical methods. Since HEIs do not solely focus on Education for SD, but do conduct research, follow administrative tasks, and disseminate research results into society—just to name a few of their different tasks—university staff working in these areas might need other competencies than those needed in the field of Education for SD. Rieckmann mentions SD challenges for HEIs which concern all their fields of action [36] (p. 129), but does not link his findings to competencies actors would need to solve these problems. Consequently, HEIs have additional obligations, when it comes to knowledge on and competencies for SD. According to the CASS [35] (p. 21), science is often seen as the 'authority' that is supposed to compile these, the above-mentioned knowledge stocks, and to communicate them into society. Furthermore, it is research for SD that should provide the competencies to engage in implementing SD [41] (p. 2). However, two points are neglected in this analysis. First, this understanding clashes with the transdisciplinary understanding of research for SD [42]. We emphasize that the creation and accumulation of knowledge, as well as the teaching of competencies, are not tasks of science alone. Non-university actors, who represent different societal needs and behalves, are important influencers and shapers of such stocks of knowledge and competencies as well. Since we do not perceive the communication of knowledge and competencies for SD as a purely top-down process, it is crucial to equally engage these actors. Second, science—which, in our case, is represented by the University of Tübingen as HEI—is never isolated from society. All members of a university are citizens as well, who are in direct contact and exchange with their societal environment. Additionally, there is a constant fluctuation of knowledge between science and society, e.g., in the form of scientific reports or discussions.

For this reason, universities have to take their role seriously. They are not only providers for certified knowledge, but also key actors in society. They have to be aware of their political and societal relevance and, therefore, have to act as role models and pioneers on behalf of SD. As much as *"knowledge of people involved and their needs and interests at stake have to be taken into account ...] sustainable development is a socio-political model for societal changes"* [42] (p. 120). Thus, we wanted to understand what kind of knowledge, skills, and competencies are present and especially needed amongst key actors for SD at the University of Tübingen. Additionally, it is crucial to reveal the university's members further expectations and needs. Only if we know them we are able to plan and conceptualize future activities of the Competence Centre in cooperation with other actors of the university. For this reason, we conducted interviews in order to get insights into the university's different fields of action. Hence, we now turn the tables and want to know which knowledge and skills are needed to act for SD in the frame of a university.

3. The Tübingen Case: The Competence Centre and its Research Project

3.1. One University, One Centre, Many Activities and Partners—The Competence Center for Sustainable Development

In 2013, the Ministry of Science, Research and the Arts of the State of Baden-Württemberg decided to fund the concept development and installation of a Competence Centre for Sustainable Development at the University of Tübingen as part of a program "Science for Sustainable Development". The Competence Centre is associated with the International Centre for Ethics in the Sciences and Humanities (IZEW) as one major player for SD, and its main tasks are to bring together all activities in the field of SD at the University of Tübingen, as well as to integrate Education for SD into teaching activities of all curricula. The above-mentioned whole-institution approach (see Section 2.3) provided the starting point for creating the Competence Centre, which addresses researchers, teachers, employees, students, and governance structures at the same time (see Figure 1). The major aim is to combine different actions and measures, which all members of the university carry out and/or implement at all university levels. Additionally, the Competence Centre is the core of an overarching network—not only in the University of Tübingen but also beyond. This approach is based on the idea that a university has a societal responsibility and serves as a role model and blueprint for other societal actors and stakeholders. For this reason, practitioners, for instance from NGOs, are sometimes cooperating in the Centre's activities. Transdisciplinary approaches are fostered but they are still in their early stages. To strengthen its SD networking activities in the university and beyond, the Competence Centre organizes regular network meetings and provides funding for small SD projects, which are initiated and realized by students.



Figure 1. The Competence Centre in the University's Sustainable Development (SD) structure.

The establishment of the Competence Centre is a result of various bottom-up efforts at the University of Tübingen, especially by student initiatives as well as researchers and teachers [43] (p. 210). The efforts and claims of the student initiative "Greening the University e.V." led to the installation of the Board for Sustainable Development [44] in 2010. The Board's work is now coordinated by the Competence Centre. The Board for SD is the main consultant of the president's office of the University of Tübingen concerning all questions on SD. Furthermore, the student initiative's efforts gave way to implementing the EMAS audit at the university in 2011 [45]. The University's Environment and Energy Management section, which is in close contact to the Competence Centre, is responsible for the regularly conducted EMAS audits.

Today, the Competence Centre is active in every field of the University of Tübingen and cooperates with many members of the university. The Centre uses its position to create awareness for topics of SD—within the university as well as for the broader public. The Centre's employees organize public events such as sustainability slams, clothes-trading events or vernissages, as well as workshops for pupils to transfer ideas of and awareness for SD to practical action. Since students and young scientists are driving forces for change and take things into their own hands [46] (p. 6), the Competence Centre aims at enabling students to take over responsibility, to empower themselves, and to foster SD according to their own interests. For this reason, the Centre mentors students who are planning SD projects or want to write their thesis in a field of SD. However, the most important point in this field are the SD network meetings (*"nachhaltig@UniTübingen"* [47]) for all university members to create the room they need for developing new bottom-up ideas. One of the last meetings was organized in cooperation with the Competence Centre and the student representatives in the Board for SD [48].

In summary, the Competence Centre is the university's network nexus point, which brings together people, knowledge, and competencies under the umbrella of SD. It provides contexts and structures to realize SD projects on all university levels and makes people's ideas shine for a better future.

3.2. The Research Project "SD@UT"

Whether a HEI moves forward to implement measures which lead to SD depends on several conditions. On the one hand, one of those characteristics is its understanding of SD, which is the basis for all activities a HEI carries out in the frame of SD [49] (p. 322). For this reason, it may be considered as crucially important for a HEI to develop a common understanding of SD, if it is keen to contributing its part to a sustainable society. On the other hand, "*it is impossible to come up with the right definition*" [50], since so many actors and their various perspectives are involved in the discussion

on working for SD in a large institution like a university. It shall be noted that apart from different details in understanding SD, there are also members of the university who would reject the immediate necessity of considerable transformations for SD in the first place.

This sensitive and controversial field is where the activities of the Competence Centre are located. From its starting point in 2013 until now, the Competence Centre has set up, experienced, and analyzed a broad range of different activities. It turned out that members of the University of Tübingen have different understandings of what SD is and of how exactly it should be implemented. Even though *"Sustainable Development is not an externally defined goal, but an open searching process with heterogeneous target components"* [51] (pp. 16–17, translated by the authors), it is obviously difficult to plan and implement further activities in the field of SD. This problematic situation led to the birth of the Competence Centre's research project, *"Sustainable Development at the University of Tübingen (SD@UT)"*. With this project, we want to shed light on the following research questions already mentioned above:

- 1. What is the understanding of SD of Tübingen University's actors who are involved in different fields of SD?
- 2. How and where shall SD be implemented at the University of Tübingen?
- 3. What kind of knowledge and which competencies do actors need to implement SD in the areas of operations, research, governance, transfer, teaching and education, and sustainability reporting?

3.3. Methods

Since our goal was to include as many actors and their perspectives, insights, and knowledge as possible, a research approach, which is based on a well-differentiated and balanced methodology is appropriate. According to Mikkelsen, "*a wealth of information* [is] *hidden in a variety of sources*" [52] (p. 87). For this reason, we based our methodology on three columns (see Figure 2). These are desktop research, questionnaires, and interviews.

Understandings of Sustainable Development and the Values, Knowledge, and Competencies behind it				
Understandings, Values, Knowledge, Competencies ↑ Desktop Research	Understandings of SD & Values behind SD † Questionnaires	Knowledge & Competencies ↑ Interviews		
Methodology				

Figure 2. SD@UT—Research methodology and research questions.

3.3.1. Desktop Research

One important part of the desktop research was a stakeholder analysis of the actors who are involved in topics of SD at the university, which we did at the beginning of the research project. Whether "stakeholder" is or should be a proper category for individuals of different groups at HEIs shall not be discussed here. We use it as a technical term in a generic sense of social science research, which, in this case, comprises persons active for SD at our university. We identified all members of the university who have already taken part in any activities of the Competence Centre or with whom the Centre has already cooperated. Additionally, we identified individuals who are involved in certain fields of action, for example, research or administration. However, it was also important to include

university members who work in overlapping fields, such as the University Library, science didactics, or the Foreign Language Centre. Finally, we equally set a focus on student groups who want to foster SD at the university. It was our goal to depict not only the HOCH-N fields of action in our work but equally to include 'representatives' for every SDG. With the help of the stakeholder analysis, we could identify university members whose opinions and experiences qualified them to be part of our questionnaire survey. Additionally, a broad literature research and content analysis of these documents supports and informs the discussions and topics revealed in this article.

3.3.2. Questionnaires

In April and May 2019, the Competence Centre identified 76 stakeholders in the field of SD at the University of Tübingen. Student initiatives, professionally active persons, and committed individuals, as well as political groups and representatives of committees were included. Then, the Competence Centre conducted a survey on the understandings and values of SD amongst this wide range of stakeholders. The questionnaire has been developed to obtain an insight on actors' understandings of and values behind SD. For this reason, we framed two types of questions: First, we directly asked about understandings of and values behind SD. Second, we had the idea to ask, in an indirect way, about both of these issues to obtain even more information. We operationalized "values behind SD" through the reasons to act for SD and "understandings of SD" through activities for SD. In our last question, we additionally wanted to learn how we, as members of the Competence Centre, could support the stakeholders with our own further activities. We sent the questionnaire to 76 members of initiatives and university groups, including students, as well as university employees from the fields of research, teaching and education, sustainability reporting, transfer, governance, and operations, who are all working on SD issues. 30 questionnaires were completed and sent back to the Centre, which corresponds to a return rate of 39.5%. Hence, the Competence Centre is able to provide an overview of the actors' understandings of SD and the values behind them. For the analysis of our data, we extracted the core statements of the questionnaires' responses, clustered them, and assigned them to appropriate keywords. We present the results in Section 4.2.

3.3.3. Interviews

In August and September 2019, we conducted six open-guideline face-to-face interviews with representatives of all six HEI areas introduced by the project HOCH-N: sustainability reporting, governance, research, teaching and education, transfer, and operations. We chose our interview partners due to their professional tasks and their personal commitment, as well as their experiences in the field of SD. The interviews consisted of five open questions and gave much room for the interviewees' own perspectives. They were all recorded and transcribed, and took around 10 to 30 min. Our guideline included questions on the importance and necessity of specific knowledge and competencies for SD for the interviewees themselves, other university stakeholders involved in actions for SD, as well as stakeholders who are not active in any area of SD.

We analyzed our data conducting a qualitative content analysis. Our analysis was summarized in categories which we supported by extracted quotes of our interviewees. With the help of our categories and the quotes, we were able to learn in detail about the interviewees' perspectives on their own knowledge and competencies, as well as on knowledge and competencies they wished others to have, who (do not yet) act for SD. We present the interviews' results in Section 4.3.

4. Results

In the following chapter, we first present the status quo of SD actions and institutionalization at the University of Tübingen. Second, we present the questionnaires' findings and demonstrate how they are connected to the Brundtland understanding of SD. Finally, we explain which knowledge and competencies actors need, who are involved in fostering SD at the University of Tübingen, according to our interviewees.

However, before we do so, we present some results of the stakeholder analysis of the actors who are involved in topics of SD at the University of Tübingen.

4.1. Stakeholder Analysis: Actors Involved in Topics of Sustainable Development

Since the University of Tübingen is one of the eleven HEIs, which are involved in the research project HOCH-N and, at the same time, the Competence Centre is set up to implement SD in all parts of the university—striving for a whole-institution approach—we analyzed the measures, which many stakeholders take at the University of Tübingen, to foster SD. Then, we closely considered future potentials in each field of action for the realization of SD at the University of Tübingen.

An analysis of all the fields of action and activities, which stakeholders carry out on behalf of SD at the University of Tübingen, is a challenge in itself and necessarily demands an own research project. For this reason, our analysis is not exhausted and only represents exemplary activities. Table 1 lists some results of our analysis.

Field of Action	Exemplary Activities Carried Out at the University of Tübingen
Research	 Research project: "GLOBUS—Reconsidering European Contributions to Global Justice" [53]. Institute of Evolution and Ecology: main topics: climate change, invasive species, land use and species diversity et al. [54]. Collaborative Research Centre SFB 923: "Bedrohte Ordnungen" [55].
Teaching and Education	 ESD certificate program "Studium Oecologicum" [56]. Courses for students of Environmental Biotechnology: "Sustainable Environmental Biotechnology Systems" [57]. Series of lectures "Gender Justice in Muslim-Christian Readings" [58].
Operations	 Several rewards as "Recycling paper friendly HEI" [59]. University members who come to work by e-bike can charge them for free [60]. The university is using 100% eco-electricity [61].
Governance	 The Advisory Board for Sustainable Development [62]. Representatives of the student body in the Advisory Board for SD [63].
Sustainability reporting	 The University of Tübingen is certified and managed according to the EMAS standards since 2011 [64]. The Competence Centre has, together with 11 other German universities, applied a beta version of the HS-DNK [65]. Every year, the university's environmental manager publishes the environmental declaration [66].
Transfer	 The university is a member of the "Action alliance for a low-waste Tübingen" and supports the ReCup system [67]. The student initiative "Papierpilz" recycles single-sided printed paper [68]. The Faculty of Humanities has university branches in Kyoto, Seoul, and Peking [69]. Sustainability Lecture and Sustainability Award for Bachelor and Master thesis [70,71].

Table 1. SD activities at the University of Tübingen.

In summary, different measures to realize SD on the different levels of the university are set in practice. Some are already institutionalized to a high degree (e.g., the university's EMAS certification

or the Advisory Board for SD) others are project-based activities (e.g., GLOBUS), which will come to an end. All in all, we found that many activities are carried out due to the personal commitment of individual university members. However, to successfully anchor SD on all levels of the university, this commitment has to spill over and every stakeholder of the university has to be committed to SD.

According to Niedlich et al. [72] (p. 3), a HEI needs to change its institutional culture if SD shall be put in practice because SD affects the whole HEI as an organization. They identified four categories of organizational culture as key areas for SD governance at a HEI. We emphasize the fourth, highlighting the importance of a "holistic governance covering all core areas of higher education institutions—teaching, research, operations and outreach" [72] (p. 4). With our findings, we are able to support this argument since it is crucially important for a HEI that the goal to realize SD at the whole university level is commonly pursued bottom-up and top-down.

4.2. The Understanding of Sustainable Development at the University of Tübingen, the Values Behind it, and the Interconnection to the Brundtland Report

4.2.1. Understandings of Sustainable Development at the University of Tübingen

As mentioned above, the Competence Centre has assembled the first university-wide understandings of SD according to the questionnaires' results. Of course, this understanding needs to be reworked and discussed continuously, since foci could change and perspectives are renewed due to new stakeholders, who will join the area of SD at the university. In its mission statement, the University of Tübingen affirms the "maxim of sustainable development" and "regards sustainability as an integral part of research and teaching" [73] (translated by the authors). Provided with this framework, it is essential for all members of the university to understand what SD means.

The Competence Centre's understanding of SD is based on the Brundtland Report of 1987, which also was mentioned several times in the questionnaires. For this reason, we will use it as the core meaning. As outlined above, SD means a development, which "*meets the needs of the present without compromising the ability of future generations to meet their own needs*" [5] (p. 16). At the same time, this development has to and can only happen within the scope of limited technical possibilities and finite natural resources [5]. The three-pillar model of SD is very common and well-known (see Section 2.1), and reasonably criticized by many scholars [13,16,74,75]. As well, in our perspective, it is not suitable for several reasons: (i) It does not depict the complexity of the SD understanding mentioned above, (ii) it ignores exactly those interdependencies between the three pillars of economy, ecology, and society, causing most of the problems and difficulties related with SD, and (iii) it is a static model, which does not demonstrate the process-oriented nature of SD. This theoretical frame of SD has been communicated by the Competence Centre on many occasions, e.g., the networking days "*nachhaltig@UniTuebingen*".

During the questionnaires' analysis, we assigned the core statements of the SD understandings to the key words, which are illustrated in Table 2.

Key Word(s)	Number of Different Statements	Key Word(s)	Number of Different Statements
Consumption Education,	5	Others	2
consciousness, sensitization	14	Preservation of resources	16
Infrastructure	3	Protection, avoidance of damages, preservation	10
Justice	10	Structures and processes	4
Levels of SD	14	Theoretical basis	2
Management	7	Transfer, realization	10

Table 2. Understandings of SD at the University of Tübingen.

In the following, we summarize, categorize, and reproduce the questionnaires' answers on the understandings of SD.

In the questionnaires, stakeholders mentioned several levels of SD; one of them is the personal dimension. At this level, we have the opportunity to change our own lifestyle and influence SD. The societal dimension connects individual and collective actions and brings forward society as well as its processes. Additionally, SD is to be realized on the professional, the institutional, and the political levels where aspects of development politics have to be integrated as well. At the global level, inequalities between humans should be reduced to make the world a better place. Overall, the world should become more just, and inter- and intragenerational interests have to be considered, as well as distributive justice and equality of opportunities. If people pay attention to the environment and all living creatures around them, the highest well-being and the best quality of life for every living being will be achieved.

SD can be realized if people are educated for and sensitive to, as well as conscious of, aspects of SD. These statements are important for HEIs and their research, education, and teaching spheres.

Another important aspect which contributes to SD is the conservation and protection of natural resources. Especially the responsible use of resources, the reduction of pollutant emissions, recycling, and the consequences of livestock farming for the use of resources and human health are important points to consider. The continuous protection of biodiversity and the diversity of species result in resistant ecosystems, which will contribute to the protection of our climate. According to the questionnaires, it is equally important to reduce climate change and to consider the interests of all non-human animals to avoid damages for the environment and nature.

Another aspect is the role of companies, which strive for the common good, and of managers, who use their scope of action for responsible decisions. Furthermore, it is urgent for the economic sector to take its responsibility seriously and to implement CSR strategies such as people, planet & profit (see John Elkington, who introduced the term *"triple bottom line"* in 1994. Further information on what he thinks today are accessible online [76]). The optimization of processes and the consideration of consequences on other regions of the world and future generations contribute to SD, too.

The last aspect, continuously mentioned in the questionnaires, evolves around the individual. Everyone can be a role model, who critically reflects on topics of SD, and thinks of consequences for his/her everyday life, acts responsibly, as well as motivates others. Part of it is certainly to question our own consumption of food and all other products, to live sufficiently, and to avoid a large ecological footprint. Finally, mobility needs to shift from the individual to the public level to create and establish an infrastructure for sustainable actions—not only in the sense of mobility.

4.2.2. Values behind Sustainable Development

The answers, which dealt with the values people associate with SD, were as various as the answers which were given to explain the understandings of SD. Figure 2 gives an overview of the statements on SD values. Some of the values are not values in the classical, philosophical and sociological understandings. For this reason, the key words we assigned them to are not values in this sense. From our perspective, it was crucial not to understand the answers too conceptually because we want every member of the university to find her/his opinion at least partly reflected.

The core value is a "good life for everybody". It is the basis for all other values and, at the same time, refers to the Brundtland Report's moral background of SD (see Figure 3). Values, which we assigned to the key word justice, seem to be an important contribution to such a good life. They rank from social justice, to generational and environmental justice, up to global justice and solidarity.



VALUES BEHIND SUSTAINABLE DEVELOPMENT

Figure 3. Values behind SD; results of the questionnaires.

The key word fairness summarizes values, such as equality and equal rights for men and women. These values show an overlapping in terms of content with justice and structures, especially social security. Other values we assigned to structures are structural change, processuality, movement, and dynamics. In addition to these values, which are based on change, there were also answers pointing to resilience and consistency.

There is a strong relation between the key concepts justice, structures, and esteem. However, values, assigned to esteem have a broader meaning and include respect for all forms of life and nature. Cultural openness can be attributed to these values as well.

We differentiated values assigned to the key concepts structures, esteem, and future. Here, we include durability, the inclusion of long-term consequences and potential, as well as a future-orientation. Future generations play an important role in this cluster, as do challenges, with regard to the future.

There are also values, which are closer related to social action: solidarity, reciprocity, fraternity, compassion, and peace which show the importance of own behavior towards fellow humans (global, current, and in the future).

Closely related to these values are the ones we assigned to responsibility. The scope of responsibility includes nature, environment, as well as future generations, and equally focusses on the present and the future.

The key word ethical practice summarizes answers which are strongly directed towards reflexive actions and which emphasize specific ethical and moral components of SD. Part of these are truthfulness, the ability to criticize oneself, and reflection, as well as ethics as the capability to include the others' needs and to act ethically. The value "not live beyond our means" is equally part of ethical practice and, at the same time, is the link to sufficiency. Here, we included satisfaction, modesty, as well as slowing down. The cluster of sufficiency is closely related to values we assigned to the key word economy. Those emphasize economic aspects and include profitability, public welfare or a full cost accounting for every product. A minimum wage for all employees, as well as fair pricing of loans, link to fairness and justice.

Equally, the concept of "buen vivir" [77] has been mentioned several times. "Buen vivir" originates in Latin America and is a concept for development, giving nature an intrinsic value and highlighting the importance of togetherness of all living beings. We summarized this life in balance with nature and society, and added the Western notion "*humankind as part of the ecosystem*" as well as the sustainable protection of all living creatures, due to their own value of life (for their own sake). This value shows the interconnection to the key words of protection.

Answers we assigned to protection include health, the environment, as well as the protection of nature and climate. The preservation of an environment, which is worthwhile to live in and is capable to live for itself, is closely connected to the protection of resources and resource neutrality. These are part of the key word resources.

We summarized other crucial values to the key word knowledge. These values deal with prudence, care, and mindfulness, as well as consciousness of and education for a careful use of natural resources, for instance. In our opinion, key words as ethical practice, resources, and protection can only be rationalized and then realized, if people know the necessary facts. To take a step further and turn this knowledge to action, everybody needs fun and a personal vocation, which we listed with the key word motivation.

4.2.3. What about the Brundtland Concept of SD?

In summary, there are three key words amongst the answers on understanding of SD, which are, as well, reflected in the values behind SD. These red threads are justice, (preservation of) resources and protection (avoidance of damages, preservation). We summarized them in Table 3.

Understandings of SD [Number of Statements]	Key Word(s)	Values Behind SD [Number of Statements]	
10	Justice	4	
16	(preservation of) Resources	3	
10	Protection (avoidance of damages, preservation)	9	

Table 3. Results of the questionnaires, red threads of SD.

The statements, which we assigned to these key words, show a high divergence in numbers between understandings of and values behind SD. In the category 'understandings of SD', justice, preservation of resources, and protection (avoidance of damages, preservation) include the most (resources) and third most mentioned statements (others). The category 'values behind SD' includes the third most mentioned statements on protection. However, justice and resources show the second and third lowest numbers. In the case of 'justice', this is due to the fact that we assigned slightly different statements to the key words 'fairness' and 'ethical practice'. Nevertheless, these keywords cover similar statements. If we summed them up to the statements of 'justice', this conceptual cloud would include the highest number of statements (17). The same applies to 'resources', which have often been associated with 'future'. If we summed up both numbers of statements, we had the number of 14, which, would be the second highest after the 'justice cloud'.

Since these keywords are represented in the Brundtland understanding of SD as well, we shortly demonstrate how the UN linked justice, resources, and protection to SD. For instance, humanity shall ensure that development is sustainable to meet *"the needs of the present without compromising the ability of future generations to meet their own needs"* [5] (p. 16). Hence, SD is based on the values of inter- and intragenerational justice. Furthermore, the world's poor should *"get their fair share of resources"* [5] (p. 16), which clearly links justice to the distribution of resources. Additionally, according to Brundtland, injustice and environmental degradation as well as poverty and conflict *"interact in complex and potent ways"* [5] (p. 240), which illustrates again the importance to protect and preserve nature and environment. Only if both are not exploited to a non-renewable degree can intragenerational justice be realized. Additionally, the normative claim of justice does not go hand in hand with the three-pillar model of SD but demands an integrative understanding of SD, which we want to realize with the whole-institution approach.

Sustainability 2020, 12, 861

For this reason, our research confirms the suggestion to explicitly use the Brundtland understanding of SD for the whole University of Tübingen.

4.3. Which Knowledge and Competencies are Needed to Realize SD in the University of Tübingen? Transfer of Knowledge—Results of the Interviews

The project SD@UT does not merely analyze the status quo of differing understandings (see Section 4.2.1) and the institutionalization of SD in the University (see Section 4.1). The main goal is to identify concrete measures of how SD comes into practice and how these measures can be optimized. One of the core elements in the understanding of Sustainable Development in the scientific and political discourse is the idea that—besides society and politics—each individual can and should take action to change "unsustainable" conditions and behaviors [11,78]. This requirement was also mentioned in our questionnaires on understandings and values of SD. However, from our point of view, it could be a challenge for each individual to actively involve in practicing SD. It is our hypothesis that actors need appropriate knowledge and necessary skills and competencies to act for SD. For this reason, it might be difficult for those who do not possess—or at least think they do not possess—adequate knowledge or competencies to engage in these activities. To learn if our hypothesis was right or wrong and to obtain insights in knowledge and competencies, stakeholders needed to act for SD working in the six different fields of action at the University of Tübingen, where we carried out six interviews, each with one representative of one field of action.

4.3.1. Proof of the Hypothesis

With the help of the interviews, we wanted to gain more and deeper insights into competencies and knowledge, which are needed to actively foster SD in the context of a university.

All six interviewees agreed on our assumption that specific knowledge and competencies are needed to set SD in practice at all levels of a university. Two of them explained that this particular knowledge can be learnt and accumulates over time. One said that a certain (scientific) background would help and ease this learning process. This knowledge depends, on the one hand, on the field of action the person is fostering measures to act for SD. On the other hand, it depends on the goals which are to be achieved.

4.3.2. Knowledge and Competencies for Each Field of Action

There is specific knowledge, which all representatives have, and competencies they need in their individual field of action.

In the area of sustainability reporting, it is important to know the organizational structures of the university to detect connecting and, maybe, leverage points. Another essential competence is communication with different players at the university and networking with local groups and other HEIs, which implies not only psychological but also strategic competencies. Additionally, it is crucial to strengthen the intrinsic motivation of the university's members and to always keep on maintaining their interest in processes connected to SD. The interviewee feels that these points are important "*that something happens here in the form of concrete projects*" (interview sustainability reporting, min. 04:04, translated by the authors).

The interviewee from the field of governance laid emphasis on the competence to transfer knowledge and to network. It is important to know how to communicate with different stakeholders of the university and to keep on bringing SD topics to the table, which needs courage and diplomatic skills.

In the context of teaching and education, knowledge on topics of SD is essential. For scientific teachers, it is important to be open and to accumulate new knowledge on SD. Another point the interviewee mentioned is to integrate knowledge of the teacher's own research into the seminars. The interaction with scientific colleagues, who are working on similar topics, is an important aspect as well.

The representative person of the field of operations said it is important to learn from scientific disciplines, especially psychology and empirical cultural studies, how to motivate all members of

the university to feel responsible and to not forget the bigger picture. For this field of action, it is important to have the competencies to initiate bottom-up processes and to be involved with student representatives. They are the ones who are able to influence their fellow students and confront them with their own actions' consequences. For this reason, it is essential to network and see synergies, and especially to cooperate with other stakeholders, who are interested in similar topics: *"Those people are everywhere, and if you join them and you work together you really can achieve something"* (interview operations, min. 16:35, translated by the authors). Another crucial point is to be ambitious and set high goals.

In the field of research, the interviewee told us that it is important to develop an own understanding of SD and to gain knowledge on the planetary boundaries of the earth. Then, there are thematic foci an SD scientist can carry out research for, such as climate change, loss of biodiversity or nuclear disarmament. These topics all have practical implications and have to be worked on since they are crucial for the survival of humanity. For this reason, a researcher has to prioritize to actively work on them and not solely discuss them theoretically.

Finally, the representative of the field of transfer laid emphasis on the knowledge of what SD is, and how structures and contexts of SD work in a university. For transfer, it is important to listen to each other carefully and to bring together different ideas as well as to integrate them into concrete projects. Additionally, there is the point of courage to foster topics, which may not be valued by many other actors in society, and to show an intrinsic motivation for change. In the end, for transfer, it is essential to be able to renounce decent things in life and to enjoy challenging oneself, and trying things out.

All interviewees agreed that the knowledge and the competencies they mentioned are finally not specific for the field of SD. On the contrary, people who own these competencies are able to work more productively because "these are basic things you need and they are not necessarily related to Sustainable Development. They are connected to SD, but actually these are human and social competencies you have to own in general" (interview sustainability reporting, min. 06:14, translated by the authors). Another interviewee added that these competencies fit into her desired ideal conception of humans and do not depend on SD.

4.3.3. Knowledge and Competencies to Foster SD in General at a HEI—What Do We Have, What Do We Need?

When we asked the interviewees about the specific knowledge and competencies members of the university involved in fostering SD should have, we received numerous and varying responses.

The person from the field of sustainability reporting emphasized that, on the one hand, a person's interest in support processes is more important than the competencies s/he owns. Especially students need—and do have—motivation to change things. On the other hand, expert knowledge is needed to successfully carry out specific projects.

In the field of governance, it is crucial to find suitable arguments and to know that topics of SD are a red thread, which runs through the whole university. Additional essential competencies are the awareness of other people's needs and interests, as well as the ability to (empathically) deal with other people.

The representative of teaching and education told us that transdisciplinarity is a key aspect of education for SD. People need a sense of responsibility to see and understand the damages their actions are causing: *"People need to understand that it is their business!"* (interview teaching and education, min. 09:07, translated by the authors). On all levels of the university where people make important decisions, they need to understand the financial consequences of non-sustainable management. Additionally, decision-makers need to perceive and understand the responsibility of a university.

In the field of operations, this sense of responsibility is a crucial point as well. However, it is not only the responsibility of the institution university but also a responsibility for the university. For instance, teaching staff should feel positive about being able to use the university's infrastructure, such as rooms, and should have an interest for their students to leave these rooms clean. Then, all members of the university should feel the task to foster SD in society and to act as key agents for SD.

For the area of research, expert knowledge on topics concerning the environment or ecology is essential. Another crucial point is to include non-scientific actors in the sense of transdisciplinary practice and research.

The representative for the field transfer appreciates the ability to first think in a utopic way and, in a second step, think realistically. All university members should be courageous and think big to not restrict themselves during the search for new ideas and solutions. In discussions, the abilities to focus on the topic and to set back one's own questions and interests are very important. This point is related to the competence of being neutral, having self-regulatory skills, and being disciplined.

The last question dealt with the competencies our interviewees have themselves and wished for others, who are not involved in SD activities (yet).

In the area of sustainability reporting, clearly the competence to network and bring knowledge together is key. The representative of governance mentioned characteristics such as motivation, perseverance, the ability to network, and being open-minded. In the area of teaching and education, others should be able to think things together, to broaden their own horizon, and not to remain in rigid structures. She wishes for every member of the university to know and understand "that every action [they take] here may have a consequence somewhere else" (interview teaching and education, min. 16:49, translated by the authors). The interviewee of the field of operations said that she would like all members of the university to get in touch with others to network and to learn continuously. Another wish she had was to recognize the enrichment of the whole process and to experience moments of happiness when they go through this transformation: "I really wish people that they can integrate this [SD] into their lives and experience what I experience, how nice it can be!" (interview operations, min. 23:50, translated by the authors). The representative of the area of research mentioned a cooperative and friendly attitude, fairness, as well as the will to work together. He appreciates the exchange with colleagues in order to learn from each other. Finally, the interviewee from the field of transfer wishes everybody to recognize their own privileges and to reflect on the consequences their actions have for planet earth. No member of the university should stay alone and think of how it will be possible to pass on his or her knowledge to others. In closing, she wishes that everybody had the willpower to change something in every area of his or her own life.

4.3.4. Knowledge and Competencies to Foster SD at a HEI—What Do We Learn?

In summary, all interviewees agreed to our assumption that specific knowledge and competencies are needed to set SD in practice at all levels of a university. There is specific knowledge all representatives have and there are competencies they all need in their individual field of action: networking, keeping sight of the big picture, interaction, and cooperation are the red threads through all the fields of action. This finding is supported by Rieckmann [36] (p. 133), who identified a set of twelve key competencies for SD which include (i) the "competency for cooperation in (heterogeneous) groups", (ii) the "competency for systemic thinking and handling complexity", and (iii) the "competency for empathy and change of perspective".

Hence, on the one hand there are basic competencies, which are needed in all areas of a university as well as specific knowledge and abilities. On the other hand, all interviewees agreed on the fact that the knowledge and competencies, which they need for their own field of action, are not specific SD competencies but facilitate work in general and social abilities people need.

As we view it, this is not as surprising as it seems at first sight. First, the goal in SD every interviewee is striving for is a very broad one. The field of governance did not define it (yet) and the University of Tübingen does not own an explicit sustainability strategy. Consequently, all actors, who are involved in the field of SD, follow their own SD understanding. This is not necessarily 'good' or 'bad'—on the one hand, one could easily lose the big picture; on the other hand, every individual has the freedom to engage actively in his or her favorite topic. Since SD is a process of societal transformation,

the interviewees mentioned knowledge and competencies, which "are somehow helpful for transformation of any kind, or for structural change of any kind" (interview transfer, min. 15:33, translated by the authors). Recently, Niedlich et al. [72] found out that there are significant differences in the conception of SD at the eleven German HEIs they examined. This is due to the lack or existence of a common understanding of SD and an SD mission statement, the way SD discussions are handled and communicated into a HEI, as well as the focus of the SD definition and the targeted SD dimensions [72] (p. 9). For most HEIs Niedlich et al. studied, it is a major challenge to establish a whole-institution approach to SD governance [72] (p. 10). Therefore, it is not surprising that the University of Tübingen, as one of the HOCH-N collaboration partners, provides an example for a HEI which holds no commonly shared conception of SD, and which seems to struggle to find its way to SD governance.

Additionally, the goals are to connect actors and to think of the big picture, as well as to understand the university structures demand for the necessary system, target, and transformation knowledge. According to our interviewees, they have the necessary system and transformation knowledge, or they know with whom they could collaborate to incorporate it. The necessary target knowledge for the specific actions of individuals or groups is also largely available. What is lacking is the target and, even more so, transformative action knowledge to jointly achieve the big picture—a more sustainable university. It is obviously difficult for all members of the university to acquire this kind of knowledge, if there is no sustainability strategy, which leads to a common goal and roadmap, and no statement to explain the university's understanding of SD. Hence, the actors cannot acquire the lacking target knowledge. For this reason, it is even more difficult for them to define necessary competencies they would need. Again, especially for HEIs, it is of utmost importance to understand that knowledge alone will not lead to successful action. Consequently, they mention broader transformation competencies, acting within their own scope, and planning their own actions and activities.

It would be the university's next task to set up an SD-strategy which is (i) explicitly based on the SD understanding following the Brundtland Commission [5], and (ii) will be specified in a participatory process to be organized by the Competence Centre. Then, the university could profit even more from its motivated members, who have the willpower to change something.

5. Discussions and Outlook

We have shown that the understandings of and values behind SD in the scientific and political discourse, as well as at the University of Tübingen, are diverse. However, we were able to demonstrate how actors' understandings and the values behind them at the University of Tübingen are in accordance with the Brundtland understanding of SD. We also showed that at the University of Tübingen many actors already fill the fields of action to foster SD. Nevertheless, there is much room to strengthen and improve activities and their output, as well as to give actors the support and freedom they need to strengthen SD at the University of Tübingen. Finally, we were able to demonstrate that both knowledge and competencies are fundamental to act for SD. Since the University of Tübingen has not yet made an explicit statement on its understanding of SD, it appears to be difficult for the actors to acquire specific knowledge to reach SD for the whole university. For this reason, they possess broad knowledge and competencies, which are crucial to any process of transformation.

Even though existing scientific studies put their focus on teaching staff and not the whole HEI as a unity fostering SD (see Section 2.2), some of our findings on the understandings of SD match these studies' results. Two important issues that were raised repeatedly in the literature [17–20,22] were '(natural/ecological/environmental) resources' and '(inter- and intragenerational) justice'. In our questionnaires' responses, these keywords were raised as well. However, the understandings of SD at the University of Tübingen are far more distinct and nuanced. We assume this is firstly due to the fact that we concentrated on stakeholders who are already involved in thematic fields of SD, and secondly, because we included university members of all fields of action carrying out a research based on the whole-institution approach (see Section 2.3).

In general, we are able to assign the understandings of and values behind SD of stakeholders at Tübingen University to the different contexts and levels of politics, philosophy, and personality, as well as to existing models (see Section 2.1). For instance, they refer to the Brundtland Report (political level), the three pillars model (models), responsible behavior (personal level), and thematic issues related to the 'good life' (philosophical level).

Existing research on university members' competencies for SD equally sets a focus on the field of Education for SD (see Section 2.4). Literature on research for SD competencies at the whole HEI, including different fields of action, does not seem to exist yet. With our results, we demonstrate existing similarities in the perceptions of university members of (i) competencies being key in Education for SD and ii) competencies our interviewees mentioned to be important for setting SD in practice at the whole university level (see Section 4.3.4). As we have outlined, we started from the perspective of a whole-institution approach at the University of Tübingen and combined it with a research project that searches for key actors' understandings and values of SD, as well as the competencies and the knowledge to set SD in practice. With this particular approach, we aimed at looking at the University of Tübingen as a whole. Therefore, our perspective was broad enough to consider all relevant fields of action. At the same time, our perspective was narrow, as we concentrated only on one HEI. For this reason, we cannot simply transfer our results and the approach we chose to other HEIs, although many of our findings resonate very much with shared experiences at other HEIs, e.g., in the HOCH-N network. This is, firstly, because we only asked and interviewed actors at the university that already had a relation to SD. It will be more representative to widen this perspective and include actors who are not yet involved in SD activities. However, this could be the next step for further research.

Secondly, our approach is a special one since, as of now, the University of Tübingen practices SD within ongoing research, education and teaching, and management. Other HEIs emphasize establishing a specific sustainability science or focus on transfer. This means that they act in different contexts with, surely, other settings, opportunities, and results. Thirdly, the German landscape of HEIs is diverse and ranges from comprehensive universities (as Tübingen) and Technical Universities to application-oriented Applied Sciences Universities (Fachhochschulen) and Universities of Teaching (Pädagogische Hochschulen). Following Niedlich et al. [72] (p. 13), and Wals [23] (p. 11) we clearly expect that different orientations of HEIs manifest in different structures. This may result in different foci of knowledge and competencies that the actors at different HEIs have and need, as the relations of the different fields of action may also differ. Further research and comparison of the findings might be very helpful in identifying such differences. As we considered above, the idea of the whole-institution approach is that actors in different areas of an institution can learn from each other. In broadening this approach to a, as we want to call it, 'whole-institutions approach', different institutions can learn, likewise, from each other and adopt fruitful measures in fostering SD.

It is also important to point out that not only our research topic is settled on a certain discourse, but also that the whole discourse itself is a rather Western- or even Germany-centered discourse. However, there are, of course, other concepts, which emerged in the Global South that could improve this discourse. Interestingly, even in our questionnaires, some actors referred to such concepts, e.g., "buen vivir", and integrated these concepts into their own understandings and values of SD. Since SD is a global task, it requires international dialogue [36] (p. 130). Including other concepts broadens the perspective and prevents the danger of a Westernized, or even neo-colonial, point of view, which may include more individual-oriented and technocratic aspects than counter-concepts developed in countries of the Global South. Therefore, these concepts have to be taken into account and the German and Western scientific discourse on SD still lacks a systematic consideration of them.

In the original outline of our research project, we planned to equally integrate perspectives from the Global South and, therefore, sent questionnaires to members of the University of Ghana and the Valley View University in Ghana. Unfortunately, the response rate was very low, so that we could not take these answers into account. However, we managed to get in contact with the "Clean Africa e.V.", an organization that regularly holds summer schools in the Global South, and is right now planning a summer school in Ghana with a focus on the whole-institution approach and SD at Ghanaian universities in summer 2020. We plan the Competence Centre's participation in this summer school to get in contact with key actors of universities in Ghana.

Furthermore, the Competence Centre could profit very much from an intense exchange with key actors at Addis Ababa University in Ethiopia, especially in the person of Professor Workineh Golga Kelbessa. He has already visited the International Centre for Ethics in the Sciences and Humanities several times. In October 2018, the Competence Centre and the research project "Energielabor" have organized a workshop on SD and Environmental Ethics, presenting exemplary perspectives from Africa and Europe, where Professor Kelbessa was one of the keynote speakers.

Additionally, the Competence Centre was involved in a small research project at Lake Kivu, Rwanda, in the summer of 2019 [79]. Together with scientists from the Universities of Heidelberg and Karlsruhe, the Competence Centre examined opportunities to mine methane in Lake Kivu and talked to involved actors on their understandings of SD, and why this form of energy supply was important from their perspective.

Following these different measures, which should just serve as illustrations of how to proceed, we are optimistic that we will manage to integrate perspectives from actors of the Global South in our findings, also in order to broaden the scope of a sustainable University of Tübingen.

Author Contributions: All authors have read and agree to the published version of the manuscript. Conceptualization, M.B., K.S. and T.P.; methodology, K.S., M.B.; validation, K.S., M.B. and T.P.; formal analysis, K.S. and M.B.; investigation, M.B. and K.S.; data curation, K.S. and M.B.; writing—original draft preparation, K.S. and M.B.; writing—review and editing, M.B., K.S. and T.P.; visualization, K.S.; supervision, T.P.; project administration, M.B. and K.S.; funding acquisition, T.P. and K.S.

Funding: This research was supported by the Institutional Strategy of the University of Tübingen (Deutsche Forschungsgemeinschaft, ZUK 63).

Acknowledgments: We want to thank Katharina Wezel, who did an excellent job in analyzing and summarizing SD activities in each field of action at the University of Tübingen. Thanks to Margarita Berg for very helpful language editing. Our deepest gratitude goes to our interview partners and all those who were willing to fill in our questionnaires. We acknowledge support by Open Access Publishing Fund of University of Tübingen. We thank the editors of this special issue, Inka Bormann and Marco Rieckmann, as well as the reviewers for their valuable critique and comments.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Thunberg, G. Fridays for Future. Available online: https://fridaysforfuture.de/ (accessed on 9 September 2019).
- Hochschulrektorenkonferenz (HRK)—Die Stimme der Hochschulen. "Hochschulen für nachhaltige Entwicklung". Erklärung der Hochschulrektorenkonferenz (HRK) und der Deutschen UNESCO-Kommission (DUK) zur Hochschulbildung für nachhaltige Entwicklung; Ein Beitrag zur UN-Dekade "Bildung für nachhaltige Entwicklung"; 2009/2010. Available online: https://www.hrk.de/fileadmin/redaktion/A4/Hochschulen_und_ Nachhaltigkeit_HRK_DUK.pdf (accessed on 21 January 2020).
- HOCH-N. Sustainability at Higher Education Institutions: Develop—Network—Report. Available online: https://www.hochn.uni-hamburg.de/en.html (accessed on 27 August 2019).
- Albiez, M.; Herth, C.; Potthast, T. "Art to stay or Coffee to go?"—Nachhaltige Entwicklung lehren und lernen in der Praxis. In *Turn Knowledge Into Action—Service Learning an der Universität Tübingen*; Müller, F., Ed.; TBHD—Tübinger Beiträge zur Hochschuldidaktik (15,2019,1); Universitätsbibliothek Tübingen: Tübingen, Germany, 2019; pp. 27–32.
- 5. World Commission on Environment and Development (WCED). *Our Common Future (Brundtland-Report);* WCED: Oxford, London, 1987.
- 6. UN Research. UN Documentation: Environment. Available online: https://research.un.org/en/docs/ environment/conferences (accessed on 21 October 2019).
- United Nations. Millennium Development Goals. Available online: https://www.un.org/millenniumgoals/ (accessed on 21 October 2019).

- United Nations. Sustainable Development Goals. Available online: https://sustainabledevelopment.un.org/ ?menu=1300 (accessed on 13 December 2019).
- Jonas, H. Das Prinzip Verantwortung. Versuch einer Ethik f
 ür die technologische Zivilisation, 1st Aufl; Suhrkamp-Taschenbuch 3492; Suhrkamp: Frankfurt am Main, Germany, 1979.
- 10. Rawls, J. A Theory of Justice, original ed.; Harvard University Press: Cambridge, MA, USA, 1971.
- Parodi, O.; Tamm, K. Personal sustainability: Exploring a new field of sustainable development. In *Personal Sustainability: Exploring the Far Side of Sustainable Development*; Parodi, O., Tamm, K., Eds.; Routledge: London, UK; New York, NY, USA, 2019; pp. 1–17.
- Spindler, E.A. Geschichte der Nachhaltigkeit: Vom Werden und Wirken eines beliebten Begriffes. Available online: https://www.nachhaltigkeit.info/media/1326279587phpeJPyvC.pdf (accessed on 21 October 2019).
- Kopfmüller, J.; Brandl, V.; Jörissen, J.; Paetau, M.; Banse, G.; Coenen, R.; Grunwald, A. Nachhaltige Entwicklung integrativ betrachtet. Konstitutive Elemente, Regeln, Indikatoren; Global zukunftsfähige Entwicklung—Perspektiven für Deutschland 1; Sigma: Berlin, Germany, 2001.
- Ott, K.; Döring, R. Theorie und Praxis starker Nachhaltigkeit, 3rd Aufl.; Beiträge zur Theorie und Praxis starker Nachhaltigkeit 1; Metropolis: Marburg, Germany, 2011.
- 15. Grunwald, A. Nachhaltigkeit verstehen. In Arbeiten an der Bedeutung nachhaltiger Entwicklung; Oekom: München, Germany, 2016.
- Ott, K. Leitlinien einer starken Nachhaltigkeit: Ein Vorschlag zur Einbettung des Drei-Säulen-Modells. GAIA 2009, 18, 25–28. [CrossRef]
- Reid, A.; Petocz, P. University Lecturers' Understanding of Sustainability. *High. Educ.* 2006, 51, 105–123. [CrossRef]
- Sammalisto, K.; Lindhqvist, T. Integration of Sustainability in Higher Education: A Study with International Perspectives. *Innov. High. Educ.* 2008, 32, 221–233. [CrossRef]
- 19. Wright, T. University presidents' conceptualizations of sustainability in higher education. *Int. J. Sustain. High. Educ.* **2010**, *11*, 61–73. [CrossRef]
- 20. Wright, T.S.A.; Wilton, H. Facilities management directors' conceptualizations of sustainability in higher education. J. Clean. Prod. 2012, 31, 118–125. [CrossRef]
- 21. Barth, M.; Rieckmann, M. Academic staff development as a catalyst for curriculum change towards education for sustainable development: An output perspective. *J. Clean. Prod.* **2012**, *26*, 28–36. [CrossRef]
- 22. Sammalisto, K.; Sundström, A.; Holm, T. Implementation of sustainability in universities as perceived by faculty and staff a model from a Swedish university. *J. Clean. Prod.* **2015**, *106*, 45–54. [CrossRef]
- 23. Wals, A.E.J. Sustainability in higher education in the context of the UN DESD: A review of learning and institutionalization processes. *J. Clean. Prod.* **2014**, *62*, 8–15. [CrossRef]
- Sibbel, A. Pathways towards sustainability through higher education. Int. J. Sustain. High. Educ. 2009, 10, 68–82. [CrossRef]
- Pittman, J. Living Sustainability through Higher Education: A Whole Systems Design Approach to Organizational Change. In *Higher Education and the Challenge of Sustainability: Problematics, Promise, and Practice*; Corcoran, P.B., Wals, A.E.J., Eds.; Kluwer Academic Publishers: Dordrecht, The Netherlands, 2004; pp. 199–212.
- 26. Buckler, C.; Creech, H. Shaping the Future We Want. UN Decade of Education for Sustainable Development (2005–2014): Final Report; DESD Monitoring and Evaluation; UNESCO: Paris, France, 2014.
- Ferreira, J.-A.; Ryan, L.; Tilbury, D. Whole-School Approaches to Sustainability. A Review of Models for Professional Development in Pre-Service Teacher Education; Australian Research Institute in Education for Sustainability, Macquarie University: North Ryde, NSW, Australian, 2006.
- Mogren, A.; Gericke, N.; Scherp, H.-Å. Whole school approaches to education for sustainable development: A model that links to school improvement. *Environ. Educ. Res.* 2019, 25, 508–531. [CrossRef]
- 29. Mcmillin, J.; Dyball, R. Developing a Whole-of-University Approach to Educating for Sustainability. J. Educ. Sustain. 2009, 3, 55–64. [CrossRef]
- Littledyke, M.; Manolas, E.; Littledyke, R.A. A systems approach to education for sustainability in higher education. Int. J. Sustain. High. Educ. 2013, 14, 367–383. [CrossRef]
- Disterheft, A.; Azeiteiro, U.M.; Leal Filho, W.; Caeiro, S. Participatory processes in sustainable universities what to assess? Int. J. Sustain. High. Educ. 2015, 16, 748–771. [CrossRef]

- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]
- Leal Filho, W.; Wu, Y.-C.J.; Brandli, L.L.; Avila, L.V.; Azeiteiro, U.M.; Caeiro, S.; Madruga, L.R.d.R.G. Identifying and overcoming obstacles to the implementation of sustainable development at universities. *J. Integr. Environ. Sci.* 2017, 14, 93–108. [CrossRef]
- 34. Koester, R.J.; Eflin, J.; Vann, J. Greening of the campus: A whole-systems approach. J. Clean. Prod. 2006, 14, 769–779. [CrossRef]
- 35. Forschung zu Nachhaltigkeit und Globalem Wandel—Wissenschaftspolitische Visionen der Schweizer Forschenden; Forum für Klima und Global Change—Schweizerische Akademie der Naturwissenschaften (CASS (Konferenz der Schweizerischen Wissenschaftlichen Akademien); ProClim (Ed.) CASS/SANW: Bern, Switzerland, 1997.
- 36. Rieckmann, M. Future-oriented higher education: Which key competencies should be fostered through university teaching and learning? *Futures* **2012**, *44*, 127–135. [CrossRef]
- 37. Wiek, A.; Withycombe, L.; Redman, C.L. Key competencies in sustainability: A reference framework for academic program development. *Sustain. Sci.* 2011, *6*, 203–218. [CrossRef]
- 38. De Haan, G. The BLK '21' programme in Germany: A 'Gestaltungskompetenz'-based model for Education for Sustainable Development. *Environ. Educ. Res.* **2006**, *12*, 19–32. [CrossRef]
- De Haan, G. Bildung f
 ür Nachhaltige Entwicklung—ein neues Lern- und Handlungsfeld; UNESCO Heute: Paris, France, 2006; No. 1; pp. 4–8.
- Lozano, R.; Merrill, M.; Sammalisto, K.; Ceulemans, K.; Lozano, F. Connecting Competences and Pedagogical Approaches for Sustainable Development in Higher Education: A Literature Review and Framework Proposal. *Sustainability* 2017, *9*, 1889. [CrossRef]
- 41. UNESCO. Leitlinien für Wissenschaft für nachhaltige Entwicklung in Forschung und Lehre; (Nichtabschließend abgestimmte); Arbeitsübersetzung der Deutschen UNESCO-Kommission: Paris, France, 2017.
- 42. Hirsch Hadorn, G.; Bradley, D.; Pohl, C.; Rist, S.; Wiesmann, U. Implications of transdisciplinarity for sustainability research. *Ecol. Econ.* 2006, *60*, 119–128. [CrossRef]
- Herth, C.; Petrlic, A.; Potthast, T. Lehre heute f
 ür die Herausforderungen von morgen: Studium Oecologicum und Bildung f
 ür Nachhaltige Entwicklung an der Universit
 ät T
 übingen. In Nachhaltigkeit in der Lehre: Eine Herausforderung f
 ür Hochschulen; Leal Filho, W., Ed.; Theorie und Praxis der Nachhaltigkeit; Springer Spektrum: Berlin, Germany, 2018; pp. 208–222.
- 44. University of Tübingen. Board for SD. Available online: https://uni-tuebingen.de/de/79228 (accessed on 23 September 2019).
- 45. University of Tübingen. EMAS Audit. Available online: https://uni-tuebingen.de/de/87762 (accessed on 23 September 2019).
- Geibel, J. Hochschulen in nachhaltiger Entwicklung. In Zukunftsfähige Hochschulen gestalten: Beispiele des Gelingens aus Lehre, Forschung, Betrieb, Governance und Transfer, 2nd ed.; Netzwerk, n.e.V., Ed.; Self-Published: Berlin, Germany, 2018; pp. 6–9.
- Competence Centre for Sustainable Development. Network meetings. Available online: https://unituebingen.de/de/93082 (accessed on 24 September 2019).
- Competence Centre for Sustainable Development. Studentisches Vernetzungstreffen: "Zukunftsfähige Uni—One Friday for Future". Available online: https://uni-tuebingen.de/de/160369 (accessed on 24 September 2019).
- Garnåsjordet, P.A.; Aslaksen, I.; Giampietro, M.; Funtowicz, S.; Ericson, T. Sustainable Development Indicators: From Statistics to Policy. *Environ. Policy Gov.* 2012, 22, 322–336. [CrossRef]
- Bloemertz, L. Alternativen unseres Ernährungssystems—Strukturierungsmöglichkeiten und Potenziale; Presentation on the German Congress for Geography; Christian-Albrechts-University of Kiel: Kiel, Germany, 25–30 September 2019.
- Vogt, M.; Weber, C.F.; Lütke-Spatz, L.; Von Stecker, C. Transfer für nachhaltige Entwicklung an Hochschulen (Beta-Version); BMBF-Projekt "Nachhaltigkeit an Hochschulen: Entwickeln–vernetzen–berichten (HOCH-N)"; HOCH-N: München, Germany, 2018.
- 52. Mikkelsen, B. Methods for Development Work and Research: A New Guide for Practitioners; SAGE Publications India Pvt Ltd.: Panchsheel Enclave, New Delhi, India, 2005.

- University of Tübingen. GLOBUS—Reconsidering European Contributions to Global Justice. Available online: https://uni-tuebingen.de/de/87641 (accessed on 12 December 2019).
- 54. University of Tübingen. Institute of Evolution and Ecology. Available online: https://uni-tuebingen.de/en/ 26486 (accessed on 12 December 2019).
- 55. University of Tübingen. SFB 923: "Bedrohte Ordnungen". Available online: https://bedrohte-ordnungen.de/ (accessed on 12 December 2019).
- 56. University of Tübingen. Studium Oecologicum. Available online: https://uni-tuebingen.de/de/81653 (accessed on 23 September 2019).
- 57. University of Tübingen. Environmental Biotechnology. Available online: https://uni-tuebingen.de/de/97376 (accessed on 12 December 2019).
- University of Tübingen. Zentrum für Islamische Theologie. Available online: https://uni-tuebingen.de/de/ 59992?tx_news_pi1[news]=41174 (accessed on 12 December 2019).
- University of Tübingen. attempto online: Mehrfach ausgezeichnet: Universität Tübingen gilt als "Recyclingpapierfreundliche Hochschule". Available online: https://uni-tuebingen.de/de/62170?tx_news_ pi1[news]=29386 (accessed on 12 December 2019).
- University of Tübingen. Kostenloser Ladestrom für Pedelecs. Available online: https://uni-tuebingen.de/de/ 95382 (accessed on 12 December 2019).
- 61. University of Tübingen. Universität Tübingen hat auf Ökostrom umgestellt. Available online: https://uni-tuebingen.de/de/95382 (accessed on 12 December 2019).
- 62. University of Tübingen. The Advisory Board for Sustainable Development. Available online: https://uni-tuebingen.de/en/79228 (accessed on 12 December 2019).
- 63. Studentische Vertretung im Beirat für Nachhaltige Entwicklung, Uni Tübingen. Available online: https://www.facebook.com/StudierendeBeiratNachhaltigkeit/ (accessed on 12 December 2019).
- University of Tübingen. EMAS an der Universität Tübingen. Available online: https://uni-tuebingen.de/en/ 87762 (accessed on 12 December 2019).
- 65. Competence Centre for Sustainable Development. Deutscher Nachhaltigkeitskodex (DNK). Available online: https://uni-tuebingen.de/de/48701 (accessed on 12 December 2019).
- 66. University of Tübingen. Umwelterklärung 2018. Available online: https://uni-tuebingen.de/en/95382 (accessed on 12 December 2019).
- 67. University of Tübingen. Gegen die Becherflut und überquellende Mülleimer: ReCup in Tübingen. Available online: https://uni-tuebingen.de/en/95382 (accessed on 12 December 2019).
- 68. Papierpilz. Kreatives Prärecycling. Available online: https://papierpilz.wordpress.com/ (accessed on 12 December 2019).
- University of Tübingen. The University branches in East Asia. Available online: https://uni-tuebingen.de/en/ 93407 (accessed on 12 December 2019).
- Competence Centre for Sustainable Development. Sustainability Lecture. Available online: https://unituebingen.de/de/35885 (accessed on 23 September 2019).
- 71. Competence Centre for Sustainable Development. Sustainability Award for Thesis. Available online: https://uni-tuebingen.de/de/32417 (accessed on 23 September 2019).
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019, 171, 434. [CrossRef]
- 73. University of Tübingen. Leitbild. Available online: https://uni-tuebingen.de/de/13046 (accessed on 12 December 2019).
- 74. Döring, R.; Ott, K. Nachhaltigkeitskonzepte. Zeitschrift für Wirtschafts- und Unternehmensethik 2001, 2, 315–339.
- 75. Paech, N. Nachhaltigkeitsprinzipien jenseits des Drei-Säulen-Modells. Nat. und Kultur 2006, 7, 42-62.
- Elkington, J. 25 Years Ago I Coined the Phrase "Triple Bottom Line." Here's Why It's Time to Rethink It. Available online: https://hbr.org/2018/06/25-years-ago-i-coined-the-phrase-triple-bottom-line-heres-whyim-giving-up-on-it (accessed on 23 September 2019).
- 77. Acosta, A. Buen vivir. In Vom Recht auf ein gutes Leben; Oekom: München, Germany, 2015.

- Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderungen (WBGU). Forschung und Bildung für die Transformation. Available online: https://www.wbgu.de/fileadmin/user_upload/wbgu/ publikationen/factsheets/fs5_2012/wbgu_fs5.pdf (accessed on 10 October 2019).
- 79. Competence Centre for Sustainable Development. Energieerzeugung am Kivu-See. Available online: https://uni-tuebingen.de/de/164320 (accessed on 23 September 2019).



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





The Role of Environmental Management Performance in Higher Education Institutions

Nicolas Roos^{1,*}, Xaver Heinicke^{2,*}, Edeltraud Guenther³ and Thomas W. Guenther²

- ¹ Sustainability Management and Environmental Accounting, Faculty of Business Management and Economics, Technische Universitaet Dresden, 01069 Dresden, Germany
- ² Management Accounting and Control, Faculty of Business Management and Economics, Technische Universitaet Dresden, 01069 Dresden, Germany; thomas.guenther@tu-dresden.de
- ³ United Nations University Institute for Integrated Management of Material Fluxes and of Resources Dresden, 01067 Dresden, Germany; guenther@unu.edu
- * Correspondence: nicolas.roos@tu-dresden.de or nicolas.roos@tu-drwesden.de (N.R.); xaver.heinicke@tu-dresden.de (X.H.); Tel.: +49-351-4633-3822 (N.R.)

Received: 28 November 2019; Accepted: 11 January 2020; Published: 16 January 2020

Abstract: Higher education institutions (HEIs) are influential social institutions which educate future decision-makers and shape society as a whole. Motivated by new public management, the proliferation of business tools, and a rising awareness for responsible acting, environmental management has also become a matter for HEIs. Focusing on performance outcomes and improvement based mechanisms leads to a professionalization through the active management of environmental issues. Therefore, the support of management structures is an essential prerequisite when implementing environmental efforts. Thus far, little attention has been dedicated to environmental management performance and steering processes of environmental issues in HEIs, which marks the research gap of this study. This article presents results of a survey on the concept of environmental management performance (EMP) based on Trumpp et al. (2015) aiming to answer the research question of how HEIs conduct environmental objectives, environmental processes, organizational structures, and monitoring. The results show that, as of now, HEIs pursue no common practice when approaching EMP. Nevertheless, two thirds of the respondents show an orientation towards sustainability with particularly high values regarding issues of environmental policy.

Keywords: higher education institutions; sustainability assessment; environmental management performance; German-speaking countries; survey

1. Introduction

Higher education institutions (HEIs), as important societal pillars, play a key role in shaping future generations and paving the way for sustainable development [1,2]. In their role as educational institutions, the creation of knowledge and the promotion of ideas for society are their major undertakings. Besides the core tasks of teaching and research, environmental efforts concerning the institutions' operational performance have gradually emerged. With the conservation of resources, saving energy, or the reduction of waste [3], an integrated understanding of taking responsibility for their own actions has spread within many organizations, including HEIs.

With increased awareness of one's own social responsibility, the wish to fulfill it and to set a good example also rises [1]. As a result, practices dedicated to the systematic management of these tasks, such as environmental management systems, have been established and promoted.

Despite these instruments originally being designed for the business context, they quickly became useful in public organizations such as HEIs. Over time, these became increasingly common as a tool for

MDPI

managing environmental responsibility on campuses of various European countries [4]. The associated thinking of improvement-based mechanisms and performance enhancement instigates a general professionalization of organizational processes, including the handling of organizational responsibility with regard to sustainable action [1]. This occurred in the context of a general transfer of business practices to public administration, resulting in a more holistic approach and professionalization of the organizations [4].

In order to meet these growing challenges, and to meet the requirements of a whole-institution approach, the role of management is becoming increasingly important for the professionalization and control of organizational activities.

Management plays a core role in the allocation of resources, strategic alignment, and the planning, implementation, and evaluation of activities [5,6]. This gives HEIs a greater significance in dealing with ecological sustainability aspects by extending their responsibilities from a purely operational level to a more strategic one. Despite expanding to include management functions, the performance of HEIs regarding ecological aspects is largely excluded from management considerations. Concepts that examine environmental management performance (EMP) in the context of HE remain nonexistent thus far.

As the systematic literature review by Roos and Guenther [7] shows, an examination of EMP has so far played a minimal role in the assessment of (ecological) sustainability at HEIs. Although individual aspects of environmental management at HEIs can be assigned to the dimensions of the EMP according to Trumpp et al. [8], no systematic approach based on performance standards is pursued. For this purpose, a step-by-step integration of control mechanisms within the framework of a holistic institutional approach is suggested, which is oriented towards a growing integration of management capacities at HEIs. However, most of the research to date has been devoted to case studies of other topics, such as the investigation of success factors and obstacles or the implementation of sustainability in HEIs [1,8–10], while others examined the assessment of sustainability engagement [3,11,12]. While this research strand mainly draws on case studies, another research strand, which is represented in more recent research, uses surveys to investigate new insights into implementation efforts to generate sustainability. This research direction is increasingly turning to systematic approaches that examine the management of sustainability and environmental aspects at the management level. Although these approaches are aimed at managing sustainability at an executive level, this research is based on general practice-driven implementation efforts with the aim of initiating engagement. However, approaches examining environmental sustainability from a systematic management performance-driven perspective have not yet been investigated [13–15]. This represents a blind spot in the research on controlling sustainability at the management level and thus marks the research gap in the present work.

On the basis of these insights, the present survey study provides first evidence on the management of environmental sustainability in HEIs along the dimensions of EMP according to Trumpp et al. [8]. Trumpp et al. distinguish between environmental management performance (EMP) and environmental operational performance (EOP). We focused our survey on EMP. The associated research question is therefore dedicated to the following topic:

How do HEIs conduct environmental management along the dimensions of EMP, including environmental policy, environmental objectives, environmental processes, organizational structures, and monitoring?

In our view, this shows a major weakness in the efforts to adopt a whole-institution approach. Although the measurement of EMP is an important aspect, without an overall institutional view of the interrelationships in the network of effects with policies, objectives, structures and processes, their long-term significance and effects remain limited. A task will be to anticipate the growing need for forward-looking management and to perceive EMP as an overall institutional task. Thus, it is a management mission to address and combine the necessary tasks of sustainability management. To answer this research question, the study investigates the present environmental sustainability efforts of HEIs to assess the application of environmental management. Therefore, the survey applied the dimensions of EMP [8] to identify relevant management structures and dimensions. Though the origin of the EMP construct is from the business context, the concept is also applicable for other organizations, such as HEIs [8].

The cross-sectional survey study addressed all public HEIs in German-speaking countries (i.e., Germany, Austria, and Switzerland).

Our study makes two contributions to the literature. First, survey-based research provides the opportunity to investigate relatively complex, multi-faceted phenomena which occur in their natural setting, and surveys are particularly suitable for gathering data on respondents' beliefs, attitudes, and perceptions that drive their behavior [16]. Thus, the survey method allows for mapping current practices in the field, providing insights into intriguing research topics that require further research or that have not yet been studied [16]. The survey study adds to the mostly qualitative research in this area [17,18] reporting on implementation approaches, value-driven behaviors, or discussing factors of success by exploring how HEIs incorporate environmental concerns in their management bodies.

Second, the study expands the existing literature on sustainability in public HEIs by investigating the managerial perspective of engagement. Since current studies on sustainability at HEIs mainly discuss possible pathways of implementation or the assessment of existing practices, the survey contributes to a holistic perspective by examining the management process along the different dimensions of EMP [16]. The study highlights the role of management functions, which enables new perspectives on the often-ignored aspects of systematical internal steering and processing by the top management. This produces a better understanding of objectives and conditions of the successful implementation of sustainability as a cross-cutting function within HEIs.

The remainder of this paper is structured as follows. In the next section, we describe the survey design and the measurement of the constructs. In Section 3, we present the descriptive results. Finally, we discuss our findings and present implications, suggestions for future research, and limitations of our study.

2. Materials and Methods

2.1. Data Collection and Sample

The population comprises all public HEIs in Germany, Austria, and Switzerland with a total number of 339. The HE systems of these three countries are comparable in their composition with a strong state regulation in the HE sector [19]. A standardized questionnaire was developed based on the recommendations of Dillman, Smyth and Christian [20] and pretested with five experienced sustainability practitioners and five experts from the research field. The experts from the HE sustainability practice had recommended a subdivision into "measures" (e.g., energy efficiency) and "fields" of action (e.g., teaching and research). The survey was developed in English to adopt items that had been published in English and afterwards translated into German. The wording of selected items was adjusted to the HE context. In order to ensure consistency in the two languages, an independent translator was asked to back-translate the survey [21]. After revising the questionnaire, we sent it to the person responsible for environmental issues in the individual HEIs in December 2017. We have proceeded, as follows, to select the appropriate respondent. We first tried to contact the person responsible for sustainability. If this was not possible, we searched for a person responsible for environmental issues, then for a person responsible for occupational health and safety or, finally, for a person responsible for public relations with a request to forward the questionnaire to a person responsible for sustainability and environmental issues.

The survey implementation contained the following four steps: (i) initial e-mailing, (ii) first follow-up, (iii) second follow-up, and (iv) third follow-up. The first step involved an e-mail that was sent to every contact person, including a cover letter and the questionnaire as an interactive PDF

form. The first follow-up was a replacement of the questionnaire and contained the same elements as described in the first step but was sent to those who had not answered, while the second and third follow-ups were phone calls.

In total, the survey generated a response of 33 (GER: 27; AUT: 4, CH: 2) usable questionnaires, which were sent exclusively to state HEIs, resulting in a response rate of 9.7%. The response rate is hardly comparable with other studies as surveys in HE on the topic of sustainability are scarce. However, the findings show that sample selection bias is not a major concern [22,23], following the suggestions of Armstrong and Overton [24]. First, in Table 1 we first match the HEI's characteristics of our sample with known attributes of the population (i.e., organizational size, country of origin, and type of HEI). HEIs with less than 7000 students are categorized as small, with at least 7000 and less than 24,000 students as medium, and with at least 24,000 students as large (similar to the classification of Burrell et al. [15]). Using a chi-square difference test, a significant difference in responses is only found for type of HEIs, which means that fewer universities of applied science and more universities responded than expected. Second, in Table 2 we compare latent variable scores between early and late respondents to test for non-response bias following the idea that "less readily" available responses (i.e., late respondents) are equivalent to non-responses. We define early respondents as those who answered at least with the first follow-up. No significant differences were found. For this reason, it is rather unlikely that there is a systemic bias on the basis of differences between early and late respondents. Nevertheless, we are aware of the relative low number of total responses as a limitation of our findings. Response rates around 10% can also be found in other recently published surveys (e.g., Guenther/Heinicke [22] with 11.26%, Bisbe and Malagueno [25] with 13.21%; de Geuser et al. [26] with 9.5%; Libby and Lindsay [27] with 13.6% and 1.5% response rate for two sub-samples). The decline in response rates is also reported in recent methodological papers (e.g., Hiebl and Richter [28]; van der Stede et al. [29]). Thus, we finally decided not to apply statistical tests in our Results section. For the Results section, we use boxplots to display the data of the survey more on an explorative level.

Organi-zational Size	Received	Expected	Type of HEI	Received	Expected	Country of Origin	Received	Expected
Large	6	3.6	University	17	10.9	Germany	27	25.7
Medium	8	8.1	University of - Applied 16 Science	16 22.1	22.1	Austria	4	4.1
Small	19	21.3				22.1	Switzerland	2
Total	3	3	33			3	3	
Chi- square test statistic	1.8	350	5.092			0.4	159	
df	2	2	1			2	2	
<i>p</i> -value	0.3	397	0.024 **			0.7	796	

 Table 1. Non-response test for organizational size, type of HEI (Higher Education Institution), and country of origin.

The Table reports the results of the chi-square statistics for the tests of regional distribution and of type of higher education institution (HEI). * Significant at the 0.1 level. ** Significant at the 0.05 level. *** Significant at the 0.01 level; two-tailed tests.

Latent Constructs	Early Respondents	Late Respondents	Mann-Whitney U Test
Environmental Policy	4.15 (n = 22)	3.75(n = 11)	Z = -0.860, p = 0.390
Environmental Objectives	3.76 (n = 22)	3.45 (n = 11)	Z = -0.748, p = 0.455
Environmental Processes	4.22 (n = 22)	3.33 (n = 11)	Z = -1.514, p = 0.130
Organizational Structure	3.82 (n = 22)	3.55 (n = 11)	Z = -0.404, p = 0.686
Monitoring Systems	3.46 (n = 22)	3.10 (n = 11)	Z = -0.765, p = 0.444
Overall EMP	3.88 (n = 22)	3.42 (n = 11)	Z = -0.860, p = 0.390

Table 2. Non-response test for latent constructs.

The Table shows the construct means of the early and late respondents of our sample for the latent constructs of the structural model. * Significant at the 0.1 level. ** Significant at the 0.05 level. *** Significant at the 0.01 level; two-tailed tests.

2.2. Measures

By addressing persons in charge of environmental management, we aimed to assess the performance of each institution compared to the average of public HEIs of similar size at the public level. Therefore, we refer to the work of Trumpp et al. [8], who define and conceptualize a measurement of environmental management performance that comprises the five dimensions of environmental policy, environmental objectives, environmental processes, organizational structure, and environmental monitoring to pursue a systematization of engagement in order to map the status quo of environmental management performance.

We use 12 items each to measure environmental policy, environmental objectives, and environmental processes; 13 items to measure environmental monitoring; and 2 items to measure organizational structure. All items are based on a seven-point Likert scale with midpoint and endpoints labeled as categories (1 = strongly disagree/not at all; 4 = neither nor; 7 = strongly agree/completely).

In close coordination and cooperation (dialogue, meetings and questionnaire pretest) with experts in operational environmental management at the TU Dresden, a total of four core areas of environmental management were identified in addition to the dimensions according to Trumpp et al., which are important for the investigation of EMP in the context of HEIs and were therefore also considered as independent fields in the questionnaire: (1) Energy efficiency, (2) fresh water consumption, (3) resource consumption, and (4) emission reduction. Furthermore, we developed certain domains within HEIs bearing key management responsibilities in terms of environmental performance. These responsibilities spread over the measures of (5) procurement, (6) facility management, (7) event management, (8) disposal, (9) research and education, (10) mobility, (11) human resource management (HRM), and (12) marketing. With this general structure, the questionnaire portrays the essential fields of action and measures to uncover environmental performance within the dimensions of EMP.

3. Results

For the visualization of our results, we use SPSS 25.0 and MS Excel software programs to create boxplots. Boxplot diagrams allow for a summarization of a dataset along the values of minimum, lower quartile, median, mean value, upper quartile, and maximum. This kind of illustration enables benchmarking and interpreting the results, especially for smaller sample sizes.

The box describes the lower and upper quartile, with the vertical line crossing it as the median. The whiskers connect the quartiles with the minimum (left) and maximum (right). The position of the triangle displays the mean value (see Figure 1). This shows if performance is below or above average. If a value is located within the light grey area or even above, the performance is between the median and the upper quartile. If a value is located in the dark grey area or even below, the opposite is true. The interquartile distance defines the range of the box, whereas the size of the whole boxplot represents the range of all answers given on the question.



Figure 1. Description of boxplot diagrams (own illustration).

For the analysis, we divided the questionnaire into different sections following the dimensions of EMP. For each question, within the sections, we generated a boxplot diagram. The following section describes the results based on the fields of action and measures proposed above.

The first boxplot addressing environmental policy (see Figure 2) shows that energy efficiency is, on average, the most elaborated field of action (mean = 4.64). Policies concerning the reduction of emissions show a slightly below average level. The examination of water consumption and resource utilization deliver average patterns (mean = 4.15 and 4.21, respectively). Turning to HEIs' environmental performance, policy issues concerning waste management show an above mean value (4.79). Considering also the median, disposal policy appears to be a well-elaborated topic. This may be due to legal requirements regulating such matters in German-speaking countries. Concerning mobility, the mean value is 4.21, which is below the median of 5, though the average performance is below average.

When it comes to environmental policies for the case of marketing or HRM, the results tend towards a lower level (1st quartile starts with a value of 1, which means strongly disagree), which is also confirmed by low mean values (3.03 and 3.25, respectively).

For the dimension of environmental objectives (see Figure 3), the measures of energy efficiency (3.94), water consumption (3.58), resource utilization (3.67), and emission reduction (3.48) show a level slightly below average. Consulting the median confirms the trend of a generally only medium adoption in practice. Finally, objectives on energy efficiency show the best average outcome, although the interquartile distance covers a broad range (values between 2 to 6). However, environmental objectives appear to be remarkably well-elaborated concerning disposal (mean = 4.36), as we can see the median reaching a high level (=5.00) compared to the other fields of action. Mobility objectives show a pattern similar to that of the policy dimension, since a mean value of 3.72 lies slightly below the median of 4.00. As was already observable for the dimension of environmental policy, marketing and HRM consistently show a low level of agreement (mean = 3.00 and 3.03, respectively). This becomes even clearer, since the lower quartile starts with the lowest value of 1 (=strongly disagree). Remarkably, event management shows almost similar values of agreement (mean = 3.13).



Figure 2. Results for the dimension environmental policy.

From the boxplot for environmental processes (see Figure 4), we can observe the highest approval rates for energy efficiency improvements (mean = 4.27), although only at a slightly better value. Processes improving the consumption of water (3.91), improving resource efficiency (3.85), or reducing emissions (3.79) show an average agreement. Looking at processes concerning environmental performance, they show a similar level of assessment. The mean values are located between 3.34 (event management) and 4.63 (disposal), which provide no considerable indications corresponding with our observations for the dimension of environmental objectives. The higher mean value and median for waste management again results from legal requirements which regulate mandatory waste disposal. The processes of mobility show a mean value of 4.03 corresponding to the median (value = 4.00). In terms of marketing and HRM, the low values (mean for both = 3.26) reveal a weak level of consent.
Sustainability 2020, 12, 655



Figure 3. Results for the dimension environmental objectives.

The exploration of the organizational structure (Figure 5) follows an alternative design diverging from the interrogative form of the other dimensions. Since this section asks about implementation efforts of HEIs with the aim of revealing certain structures related to environmental topics, the standardized question pattern is unsuitable for providing appropriate results. Therefore, the dimension for organizational structures captures items on training and internal communication. The results show that trainings for employees on environmental topics are generally perceived below average (3.39). The application of adequate internal communication tools for an improved management of environmental topics centers around a mean value of 4.06. Remarkably, the interquartile distance covers a broad spectrum ranging from 2 (=disagree) to 6 (=agree).



Figure 4. Results for the dimension environmental processes.



Figure 5. Results for the dimension organizational structure.

The boxplot in Figure 6, for the dimension of monitoring systems for environmental management performance, indicates that HEIs mainly apply measurement tools for the case of energy efficiency (mean = 4.27). Here we can observe a slightly higher mean value than in other fields of action (water use: 3.67; resource use: 3.24; emission reduction: 3.09). The efficient use of resources, which includes tools for the reduction of emissions and the application of monitoring systems, appears to be slightly underrepresented. In both cases, the lower quartile starts with the value of 1 (=strongly disagree). Regarding the fields of action, HEIs, on average, perform best concerning disposal aspects. With a mean of 4.19 and a median of 5, again this appears to be the topic with the most emphasis, since regulatory requirements force HEIs to adopt certain disposal procedures. Interestingly, measurement tools for green procurement are obviously not in use, as the mean value of 3.00 indicates a general below average adoption. Similar to the low levels of agreement for the dimensions of policies and objectives, the assessment of monitoring for mobility issues is below average (mean = 3.45), which again is below the median of 4.00. Finally, event management (mean = 3.06), marketing (mean = 2.57), and HRM (mean = 2.65) consistently show the same below average performance as the other dimensions. In all these fields, the lower quartile begins with the value of 1 (strongly disagree) and ends with the value of 4 (partly agree). Especially for the case of marketing, the median shows a considerably low level (value = 2.00). The final question on environmental monitoring systems for other relevant fields of HEIs shows a low mean of 2.79.



Figure 6. Results for the dimension monitoring systems.

In summary, the analysis allows deeper insights into the performance dimensions of EMP for HEIs. In the next step, our study examines the mean scores over all items for each EMP dimension (Table 3), which allows for implications on the performance of each single EMP dimension for the HEIs.

Table 3.	EMP (Environmental	Management	Performance) mean scores	for the EMP	dimensions
----------	-------	---------------	------------	-------------	---------------	-------------	------------

Score	Dimension of EMP Environmer		Env.	Env.	Org.	Monitoring
	Policy		Objectives	Processes	Structures	Systems
	Mean Value	4.0	3.6	3.9	3.7	3.3

Apparently, the dimension environmental policy shows the highest mean value of 4.00. This shows the general commitment of HEIs to proclaim sustainability engagement on an institutional level. A similar value can be found for the dimension environmental processes (mean = 3.9). This supports a certain anchoring of the environmental engagement within the institutions. The organizational structures with a mean of 3.7 are directly interrelated to these processes and objectives (mean = 3.6), consequently followed by monitoring systems (mean = 3.3). In sum, the relative moderate commitment for EMP of HEIs may be due to different emphasis of the HEI's top management for environmental issues in general over other issues at the HEI as well as individual commitment within the HEI's top management.

A further split of the results for the different EMP dimensions by size in Table 4 delivers an even more detailed picture of the performance of the HEIs.

Dimension of EMP Size	Environmental Policy	Env. Objectives	Env. Processes	Org. Structures	Monitoring Systems	Total EMP
Small Size	3.79	3.44	3.59	3.76	3.08	3.52
Medium Size	3.90	3.43	4.20	3.69	3.69	3.78
Large Size	4.88	4.65	4.67	3.67	3.70	4.28

Table 4. EMP mean scores and HEI Size.

Evidently, large institutions perform, on average, better than small or medium ones. Going from large to small, the study observes a slope of mean values in almost every dimension, whereas the gap between large and medium-sized institutions shows to be more considerable than between medium and small ones. Not surprisingly, the mean value of total EMP also reflects these size effects.

Furthermore, when dividing the sample into two groups of high and low performers, we examine what makes the difference between good and bad performance in terms of EMP. For this case, the mean EMP scores for each EMP dimension are taken and aggregated to one total EMP score of each HEI. We classify HEIs as high performers when the mean score value is greater than 5.00 and as low performers for mean score values of less than 3.00. Cases between 3.00 and 5.00 are categorized as medium [30].

In total, the study identified 7 high performers, 11 low performers, and 15 medium performers. Based on this categorization of HEIs, an additional in-depth analysis of each category was conducted.

Publicly available information on the integration of environmental topics in HEIs' policies, as well as objectives, structures, processes or monitoring on the websites of the HEIs, was examined for further information on environmentally relevant performance characteristics in order to enrich the existing data from the survey to find out more about further efforts on (environmental) sustainability within the curriculum, research projects or voluntary self-obligations. Furthermore, the study sought proof of institutional anchoring within management bodies or formal commitments in the published statements on mission, values, or strategy. In addition, this information was matched with a possible assignment within the EMAS register to see if an institution's performance is based on external validation.

For the case of high-performing HEIs, the study identified a continuous orientation towards sustainability in research and teaching programs. In approximately 70% of the cases, we could find

evidence for institutionalization within the institution's structures, resulting in a mean value of 5.7 for this specific EMP dimension. The strongest mean value could be observed for the dimension environmental policy (6.3), followed by environmental processes (5.9) and environmental objectives (5.8). Interestingly, environmental monitoring appears to be an outlier, since the mean value of 5.1 lies noticeably below the other dimensions. Four of these seven institutions are also registered with EMAS. Differing from this pattern, two cases have high total EMP scores but no publicly available information on certain activities is provided. The mean value of high performers over all EMP dimensions is 5.8.

In the case of low-performing HEIs, the examination on the publicly available information delivered no further information on sustainability activities, neither in research and teaching, nor for the institutional context. Remarkably, almost two-thirds of the low performing HEIs are rather small, with only 10% of low-performing institutions being large. As was already observable within the high-performing HEIs, structures (2.2) show the highest adoption, followed by policy (2.0) and processes (1.9). Unlike the high-performer group, the mean monitoring performance (1.8) of low performers exceeds the performance on environmental objectives (1.7). Interestingly, we also found evidence for engagement in research or teaching or even both, which obviously did not result from an increased environmental management performance of the institution as such (cases 15 and 18). The mean value of low-performing institutions over all EMP dimensions is 1.6.

In the group of medium-performing HEIs, the results show a heterogeneous picture. Almost two-thirds of the medium-performers show an orientation towards sustainability, at least in research and teaching, in institutional bodies, or even both. The mean value for policy shows the highest value (4.4), followed by processes (4.4) and objectives (4.0). In contrast to the other performing groups, organizational structures (3.9) seem to be underdeveloped, while monitoring (3.6) appears to be relatively less common. Remarkably, in two cases, the study found no indication for actual engagement, which could explain the average results (cases 21 and 24). In contrast, the study also finds cases showing activities in research and teaching and in institutional anchoring (cases 1, 22 and 33). Surprisingly, none of the medium cases have undertaken EMAS registration. The mean value of medium performers is 4.0, congruent with the median value of our scale.

4. Discussion

This survey study examines the incorporation and operationalization of environmental issues in HEIs' management and answers the research question of how HEIs conduct environmental management along the dimensions of EMP, including environmental policy, environmental objectives, environmental processes, organizational structures, and monitoring. The empirical data provide insights into the status quo of adoption and examine HEIs' environmental performance along the EMP dimensions of policy, objectives, structures, processes, and monitoring. The results show that environmental sustainability issues in fact play a role within HEIs' management, though the distribution across the different dimensions differs noticeably. The difference in performance levels between environmental objectives and environmental policy indicates a gap in the translation of (voluntary) commitments into clearly defined environmental objectives. The decoupling of these two aspects shows that although there is a basic awareness of the need to take environmental concerns into account, this awareness remains stuck on an abstract level and thus calls into question the actual will to commit oneself more, which ultimately leaves the impression of lip service. When looking at a whole-institutional approach, it becomes clear that although there is a willingness to assume responsibility at management level, this does not appear to be perceived as a core task of HEI action. In this understanding, efforts to establish supporting structures, processes or control mechanisms, in an effort to improve EMP, seem less promising in the long term.

Including the low average performance in terms of monitoring systems, a possible misfit with regard to effective management becomes noticeable (You cannot manage what you do not measure), which might be due to mutual reinforcement in conjunction with weak goal setting.

Despite great efforts with a total of five rounds, only a relatively low response rate could be achieved among the HEIs surveyed. Although we exhausted all possibilities from a methodological point of view [28], we were not able to increase the response rate significantly. This may be due to the fact that there are no central contacts in the HEIs with regard to EMP. Since this survey was developed together with practical experts from the HE environmental management along the usual cascade of responsibilities for environmental issues in HEIs, the usual persons in charge seem only partially informed about the surveyed topics for the cross-section of HEIs. This leads to the conclusion that concrete contact persons, if any, are difficult to identify for the broad field of sustainability management. This may be due to a lack of consent on the managerial level, suggesting that holistic responsibility has not yet been regulated within the organization. This leads to the assumption that the spread of the cross-sectional function of sustainability management in HEIs has so far been low and a silo thinking (e.g., facility management, operations, focus on environmental or waste topics) according to defined responsibilities prevails.

With regard to the adoption of ecological sustainability in HEIs, another reason may be the lack of consent at the management level, which can be derived from a lack of relevance to the topic. There is some competitive pressure on HEIs with regard to integrating sustainability in research and teaching performance, but sustainability aspects, so far, are of minor relevance. Possible reasons could be that the public financing of HEIs through target agreements contains few or no sustainability targets. This means that the ministries do not tie any sustainability goals to the general budget of the public HEIs, thus the topic of sustainability plays a subordinate role. A similar result is reported in Heinicke and Guenther [31] for the implementation of management control systems. The authors argue that the fulfillment of political and legal requirements is of highest priority in order to secure funding for the HEI and it suppresses the adoption of management systems if not explicitly demanded or connected with funding.

Another reason for the relatively low and rather diverse adoption rates in our survey may be that sustainability management in HEIs is driven by the individual engagement of single players within the HEIs. This is confirmed by experiences of the authors in the research project HOCH-N. The HOCH-N project consists of a research network of 13 German universities as well as a number of committed institutions, organizations, and individuals from the academic environment who are dedicated to the question of sustainable development in the HE context. Respondents that are part of this network show a markedly sharper ecological sustainability profile in terms of policies, goals, structures, processes, and monitoring compared to HEIs that are not involved.

Our study *contributes* to the discussion of sustainability in HEIs in two ways. First, the study provides a survey-driven empirical approach for the systematization and relevance of environmental management aspects for determining the EMP of HEIs.

The vast majority of studies on environmental aspects of sustainability at HEIs, to date, have investigated success factors for the implementation of sustainability in the institution on a case-by-case basis. Due to the high contextual specifics, a generalization of the results has been possible only to a limited extent. The investigation of the introduction of sustainability from the ground up follows different development paths, mostly relating to the voluntary commitment of motivated actors. The present study allows to exceed the limitations of institutional specifics and to achieve a degree of generalizability. Furthermore, the study is oriented towards mechanisms of control as they are established in the corporate context as well as in new public management. Thus, the approach ties in with existing management structures and allows adopting a whole-institution approach from a management perspective. In addition to the examination of general success factors, the study shows where, from an organizational perspective, challenges for the implementation of sustainability can exist, and provides indications of how these can be mastered with the help of a performance-oriented approach.

The present study expands existing empirical research by investigating, by way of a broad survey, the occurrence of EMP in the context of HEIs. To the best of our knowledge, there are currently no other survey studies which have assessed the EMP of management bodies for the case of HEIs. Based on our full sample of HEIs from German-speaking countries, with no major concerns of response bias, the study expands the relevant literature by clarifying the importance of environmental management performance as a holistic (management) approach across the dimensions of EMP and its relevance to the measurement and management of environmental aspects at HEIs. It is evident that an organization-wide approach to managing environmental performance cannot be viewed solely at management or operational level, but must be implemented as a bipolar flow, flanked by voluntary engagement and stakeholder participation bottom-up and top-down in the organization if it is to be successful.

In view of the results of the study, it may also be useful to include aspects of organizational culture in the consideration. As Niedlich et al. [32] have shown in their study, organizational culture plays a crucial role in HEI's governance for the implementation of sustainability aspects. Cultural orientations at management level can thus act as a stimulus or barrier to the success or failure of implementation. With regard to EMP, it might be useful to consider these aspects of organizational culture.

In summary, the analysis shows an average moderate level of EMP and an interrelation between an increased EMP and ambitions for sustainable research and/or teaching.

Environmental policy shows, on average, the strongest emphasis, whereas monitoring systems tend to have received lesser attention. Furthermore, according to our results, the study discloses an association between the environmental performance level for EMP and an institution's size. Analyzing distinct fields of action and measures of environmental performance shows a moderate performance on disposal topics (mean value 4.49), energy efficiency (4.10), and facility management (4.04) issues. On the downside, the results show weak performance for marketing (2.96), HRM (3.05), and event management (3.25) topics.

A stronger performance in the field of waste disposal topics might be caused by the presence of distinct legal requirements regulating the waste management of (public) organizations and firms.

High performance levels in both functions of energy efficiency and facility management might result from a remarkable overlap in their practical execution. Since energy efficiency may also be included in an environmentally sustainable facility management, their mutual affection under the lens of practical operations appears to be evident. Another reason for these results might be due to low thresholds for taking action. In both cases, the implementation and operation of optimization potentials requires relatively short decision-making processes or top management commitment, but mainly personal engagement and willingness to contribute from the person(s) in charge of the function.

Remarkably, there is a missing entanglement with procurement issues (3.67) since maintenance requires appropriate supplies of operating resources. This actually contrasts with the performance results in the dimension of environmental processes (see Figure 6). Interestingly, more than half of the responding institutions have staff in charge of environmental (sustainability) issues (regardless of their size). According to our results, it seems that those functions are predominantly assigned to the HEI's facility office, since the study measured a strong EMP centered on central functions (especially facility management and disposal). However, this shows that an institutionalization of engagement does not lead automatically to an increase of the HEI's overall environmental management performance, since distinct fields of action and measures appear to perform weaker than others. This might also arise from a general lack of comprehensive systems of sustainability objectives, which require a holistic approach that includes all EMP dimensions.

As a second contribution, the study expands the current literature on sustainability in public HEIs by exploring the role of management for the implementation of (environmental) sustainability efforts. The application of EMP enables a new perspective for the understanding of aims and conditions of the successful implementation of a holistic sustainability approach within HEIs.

With a moderate adoption of environmental policies, processes and organizational structures, and a low dissemination of monitoring systems, EMP does not seem to have supported the change of perspective from an output orientation (e.g., reduction of waste and energy) to an outcome orientation for societal welfare (role model for students and researchers). With a general willingness to become active at the top management level, the operational practice (so far) is not sufficiently capable of fulfilling this commitment.

As the results show, environmental management is an emerging issue for HEIs in their endeavor to fulfill their social mission and reduce negative impacts of research and teaching. With the apparent willingness of management bodies to adapt environmental sustainability into their policies and structures, a first step towards the implementation of environmental management has already been made. However, the existing structures also must cover operational practices to pursue a holistic approach to implementing sustainability across the institution, thereby ensuring the lasting success of these efforts.

5. Concluding Remarks

Based on these findings, the study proposes distinct *implications* for both decision makers (e.g., administrators, environmental management officers, and the HEI's top management) and researchers. First, the pledge of engagement within the policy must be translated into certain goals guiding environmental officers and enabling the assessment of activities through monitoring systems, including the distribution of (financial) resources for the appropriate allocation of means to implement sustainability. Importantly, measuring the allocation and use of means is crucial for building long-term structures enabling continuous (improvement) processes. Applying monitoring systems in that context not only contributes to assessing the EMP, but also enhances transparency and broadcasts information on the progress of implementation and improvement, improving the social credibility needed to pursue the intended policies and objectives.

Second, the results suggest the charge of environmental sustainability within HEIs is complex. It seems that the success of certain efforts depends on the setting of the person(s) in charge of these issues. Apparently, environmental sustainability seems to be located within the facility division in most cases. This may be a logical consequence of the importance of this field of action, which, however, likewise neglects other fields. A possible solution might be the implementation of cross-sectional posts that can manage and control all relevant fields of action and measures of environmental sustainability simultaneously. Another option describes participatory approaches supporting existing structures to implement policies and objectives by voluntary engagement or shared governance. Since distinct stakeholder groups, such as students or staff members, are directly affected by HEIs and therefore have a legitimate claim to participate, their involvement on a voluntary basis could have the potential to foster environmental engagement and integrate relevant target groups into decision-making processes.

The aforementioned points to the third implication on continuity and impact. Since our study observes a weak anchoring of environmental objectives and an even weaker embedding of monitoring and control mechanisms, the measurement and assessment performance of environmental issues under the lens of EMP remains limited to short or mid-term effects. Generating impacts should actually be the standard for engagement, but appropriate indicators or measurement tools are still missing. Nevertheless, the fact that HEIs have established structures suggests a positive development and shows their willingness to contribute and shoulder responsibility. Turning to impact measurement, the question arises as to how HEIs can pursue purpose through their actions.

As many other surveys, our study also has *limitations* providing avenues for future research. One major limitation in the generalizability of the results lays in the fact that only German-speaking and state-run HEIs are taken into account. Public sponsorship and state funding are a decisive differentiating factor in comparison with other countries and also in contrast to private HEIs. Future research may assess whether our findings also hold for other regions and HEI settings. Another major limitation is the relatively low total number of responses which does not allow for statistical testing of our findings. Thus, interpretation of results has to be conducted carefully. In addition, the unit response tests indicate that the sample contains more universities than universities of applied science than expected, which could bias our findings. Therefore, future research could examine the EMP for other

countries or HEI cultures (e.g., UK or US). Furthermore, future researchers could draw a comparison between private and public institutions. A replication of our HE setting in other global regions, as well as a comparison with other educational systems (e.g., North America, Australia, and Asia) could also be an interesting endeavor. It is also conceivable to expand our survey to performance issues in teaching and research in relation to the operational performance of an HEI to assess the overall sustainability performance of an HEI.

Another future task will be to define an HEI's responsibilities to planet and people beyond the core business of teaching and research. This opens the field of whether the terms sustainability and responsibility can be regarded as complementary or as competing terms associated with different functions. Thus, a future research question could cover whether it is useful to introduce an overarching sustainability (performance) management on all dimensions of the triple bottom line or if treating each dimension (i.e., environmental and social sustainability) separately is more appropriate. Based on the results, HEIs are moving in the right direction, even though there is a considerable potential for institutions to improve further, and ensure this progress continues long into the future.

Author Contributions: Conceptualization, N.R., X.H., E.G., and T.W.G.; methodology, X.H. and N.R.; software, X.H. and N.R.; validation, X.H. and N.R.; formal analysis, X.H. and N.R.; resources, E.G. and T.W.G.; data curation, N.R. and X.H.; writing—original draft preparation, N.R. and X.H.; writing—review and editing, N.R., X.H., E.G., and T.W.G.; visualization, X.H. and N.R.; supervision, E.G. and T.W.G.; project administration, N.R. All authors have read and agree to the published version of the manuscript.

Funding: This research received no external funding.

Acknowledgments: We thank the editor and reviewers for their valuable and helpful feedback on this publication.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Amaral, L.P.; Martins, N.; Gouveia, J.B. Quest for a sustainable university: A review. Int. J. Sustain. High. Educ. 2015, 16, 155–172. [CrossRef]
- Figueiró, P.S.; Raufflet, E. Sustainability in higher education: A systematic review with focus on management education. J. Clean. Prod. 2015, 106, 22–33. [CrossRef]
- Alshuwaikhat, H.M.; Abubakar, I. An integrated approach to achieving campus sustainability: Assessment of the current campus environmental management practices. J. Clean. Prod. 2008, 16, 1777–1785. [CrossRef]
- Disterheft, A.; da Silva, S.S.F.; Ramos, M.R.; Azeiteiro, U.M.d. Environmental Management Systems (EMS) implementation processes and practices in European higher education institutions—Top-down versus participatory approaches. J. Clean. Prod. 2012, 31, 80–90. [CrossRef]
- Sammalisto, K.; Sundström, A.; Holm, T. Implementation of sustainability in universities as perceived by faculty and staff – A model from a Swedish university. J. Clean. Prod. 2015, 106, 45–54. [CrossRef]
- Brinkhurst, M.; Rose, P.; Maurice, G.; Ackerman, J.D. Achieving campus sustainability: Top-down, bottom-up, or neither? Int. J. Sustain. High. Educ. 2011, 12, 338–354. [CrossRef]
- Roos, N.; Guenther, E. Sustainability Management Control Systems in Higher Education Institutions. From Measurement to Management. *IJSHE* 2020, 21. [CrossRef]
- Trumpp, C.; Endrikat, J.; Zopf, C.; Guenther, E. Definition, Conceptualization, and Measurement of Corporate Environmental Performance: A Critical Examination of a Multidimensional Construct. J. Bus. Ethics 2015, 126, 185–204. [CrossRef]
- Clarke, A.; Kouri, R. Choosing an appropriate university or college environmental management system. J. Clean. Prod. 2009, 17, 971–984. [CrossRef]
- Leal Filho, W.; Wu, Y.C.J.; Brandli, L.L.; Avila, L.V.; Azeiteiro, U.M.; Caeiro, S.; Madruga, L.R.D.R.G. Identifying and overcoming obstacles to the implementation of sustainable development at universities. *J. Integr. Environ. Sci.* 2017, 14, 93–108. [CrossRef]
- Lozano, R. Incorporation and institutionalization of SD into universities: Breaking through barriers to change. J. Clean. Prod. 2006, 14, 787–796. [CrossRef]
- Savely, S.M.; Carson, A.I.; Delclos, G.L. An environmental management system implementation model for U.S. colleges and universities. *J. Clean. Prod.* 2007, *15*, 660–670. [CrossRef]

- Berzosa, A.; Bernaldo, M.O.; Fernández-Sanchez, G. Sustainability assessment tools for higher education: An empirical comparative analysis. J. Clean. Prod. 2017, 161, 812–820. [CrossRef]
- Lozano, R. A tool for a Graphical Assessment of Sustainability in Universities (GASU). J. Clean. Prod. 2006, 14, 963–972. [CrossRef]
- Burrell, D.N.; Anderson, M.; Bessette, D.; Dawson, M. A Contemporary Evaluation of Universities and Sustainability Strategic Planning. *Rev. Manag. Innov. Creat.* 2011, 4, 65–80.
- Speklé, R.F.; Widener, S.K. Challenging Issues in Survey Research: Discussion and Suggestions. J. Manag. Account. Res. 2018, 30, 3–21. [CrossRef]
- Posner, S.M.; Stuart, R. Understanding and advancing campus sustainability using a systems framework. Int. J. Sustain. High. Educ. 2013, 14, 264–277. [CrossRef]
- Ribeiro, M.M.; Hoover, E.; Burford, G.; Buchebner, J.; Lindenthal, T. Values as a bridge between sustainability and institutional assessment: A case study from BOKU University. *Int. J. Sustain. High. Educ.* 2015, 17, 40–53. [CrossRef]
- 19. Kehm, B.M.; Lanzendorf, U. The impacts of university management on academic work: Reform experiences in Austria and Germany. *Manag. Rev.* 2007, *18*, 153–173. [CrossRef]
- Dillman, D.A.; Smyth, J.D.; Christian, L.M. Internet, phone, mail and mixed-mode surveys: The tailored design method. *Reis* 2016, 154, 161–176.
- 21. Brislin, R.W. Back-translation for cross-cultural research. J. Cross-Cult. Psychol. 1970, 1, 185–216. [CrossRef]
- Guenther, T.W.; Heinicke, A. Relationships among types of use, levels of sophistication, and organizational outcomes of performance measurement systems: The crucial role of design choices. *Manag. Account. Res.* 2019, 42, 1–25. [CrossRef]
- 23. Van der Stede, W.A.; Young, S.M.; Chen, C.X. Assessing the quality of evidence in empirical management accounting research: The case of survey studies. *Account. Organ. Soc.* 2005, 30, 655–684. [CrossRef]
- 24. Armstrong, J.S.; Overton, T.S. Estimating nonresponse bias in mail surveys. J. Mark. Res. 1977, 14, 396–402. [CrossRef]
- 25. Bisbe, J.; Malagueño, R. Using strategic performance measurement systems for strategy formulation: Does it work in dynamic environments? *Manag. Account. Res.* **2012**, *23*, 296–311. [CrossRef]
- 26. De Geuser, F.; Mooraj, S.; Oyon, D. Does the balanced scorecard add value? Empirical evidence on its effect on performance. *Eur. Account. Rev.* 2009, *18*, 93–122. [CrossRef]
- 27. Libby, T.; Lindsay, R.M. Beyond budgeting or budgeting reconsidered? A survey of North-American budgeting practice. *Manag. Account. Res.* 2010, 21, 56–75. [CrossRef]
- Hiebl, M.R.W.; Richter, J.F. Response Rates in Management Accounting Survey Research. J. Manag. Account. Res. 2018, 30, 59–79. [CrossRef]
- Van der Stede, W.A.; Young, S.M.; Chen, C.X. Doing management accounting survey research. *Handb. Manag.* Account. Res. 2006, 1, 445–478.
- Henri, J. Taxonomy of performance measurement systems. In Advances in Management Accounting; Elsevier: Amsterdam, The Netherlands, 2009; Volume 17, pp. 247–288.
- 31. Heinicke, X.; Guenther, T.W. The Role of Management Controls in the Higher Education Sector: An Investigation of Different Perceptions. *Eur. Account. Rev.* **2019**, 1–50. [CrossRef]
- Niedlich, S.; Kummer, B.; Bauer, M.; Rieckmann, M.; Bormann, I. Cultures of sustainability governance in higher education institutions: A multi-case study of dimensions and implications. *High. Educ. Q.* 2019. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article Bridging Intellectual Capital, Sustainable Development and Quality of Life in Higher Education Institutions

Eugénia de Matos Pedro ^{1,*}, João Leitão ^{1,2} and Helena Alves ¹

- ¹ Faculty of Human and Social Sciences, NECE–Research Center in Business Sciences, University of Beira Interior, 6200-001 Covilhã, Portugal; jleitao@ubi.pt (J.L.); halves@ubi.pt (H.A.)
- ² CEG-IST and ICS, University of Lisbon, 1649-004 Lisboa, Portugal
- * Correspondence: eugenia@ubi.pt

Received: 19 November 2019; Accepted: 2 January 2020; Published: 8 January 2020

Abstract: This paper analyses the relationship between the intellectual capital of higher education institutions (HEIs) and their sustainable development practices, and assesses whether higher education institutions' sustainable development practices are related to their stakeholders' quality of life. Using a structural equation model, two model specifications are estimated, gathering primary data from a convenience sample composed of 738 full-time students and 587 teachers/researchers at seven Portuguese higher education institutions. The findings reveal that intellectual capital influences sustainable development practices directly and positively, whereas sustainable development practices influence students' quality of life in a significant way, although the same is not verified for teachers/researchers. These findings provide insightful implications for policy-making and intellectual capital management for practices in higher education institutions; firstly, by showing that the sustainable development concept is associated with HEIs' practices of economic, environmental, social and organisational sustainability; secondly, by concluding that public Portuguese HEIs need to improve the social dimension of their sustainable development practices, and here there may be room for improvement in the institution through better and more proficient social engagement that is more directed to the challenges of sustainability and social change; and thirdly, by showing that the inclusion of better sustainable practices has repercussions on the quality of life of all stakeholders.

Keywords: higher education institutions; intellectual capital; performance; quality of life; sustainable development

1. Introduction

The different ways in which organisations, cities, regions and countries manage and introduce intellectual capital (IC) practices have been found to be a decisive factor, not only for their reputation, competitiveness and wealth, but also in raising their sustainability, focusing on citizens' quality of life (QoL) and contributing to a more sustainable and balanced society [1]. Despite the remaining gap and lack of information in the literature about how intellectual capital and sustainability influence each other from the practitioners' perspective, researchers' theoretical perspectives have shown how IC and sustainability are closely related [2]. For example, it was revealed that a country's knowledge assets and intangible assets have significant implications for its future value, inasmuch as they represent a source of skills and competences considered essential for national economic growth, the development of human capital and promotion of QoL [3]. Adding to the previous statements, knowledge, creativity and innovation have become the main factors stimulating social and economic development, reinforcing the role of IC in generating sustainable growth and development [4].

IC plays an important role in determining regional competitiveness, being even more important in regard to sustainable regional competitiveness, and it is assumed that the most important intellectual resources are those that contribute to the creation of a competitive advantage and so result in an improved economic situation of a region [5]. For example, Dal Mas [6] demonstrated the relationship between IC and sustainability from a practitioner's point of view, supporting the interlink between IC and sustainable regional competitiveness. In addition, it increases society's awareness of sustainability, defines a region's legal and institutional environment, expresses relationships between the various stakeholders and, as a whole, creates the basis for forming sustainable competitive advantage. According to Malhotra [7], the intangible assets of a country have significant implications for the future national value, because they represent a source of the skills and competences considered essential to national economic growth, human development and QoL.

Higher education institutions (HEIs) are also part of this premise, as they are experiencing the challenges of sustainability, which is increasingly recognised as an essential driver for the development of sustainable societies [8], and also contributing to the QoL of their stakeholders and the populations where they are located [9]. Several challenges have been faced by HEIs, namely regarding budget reductions, which imply the implementation of efficiency and cost reduction logics, as well as adopting new community welfare promotion practices to improve their own quality of academic life (QAL). In this way, HEIs reinforce the attractiveness and retention of human and financial resources, which will positively contribute to the sustainability of these institutions. HEIs have shown a growing commitment to sustainable development (SD) through their mission statements, support and agreements, as well as through the effective implementation of the initiatives and practices of SD [10]. In recent years, some studies have been carried out regarding HEIs' involvement in the implementation of SD practices, e.g., [10–12]. These practices are linked to different dimensions (e.g., economic, environmental, social and organisational) and are integrated into the main activities of HEIs, namely teaching, research, operations, social commitments and culture [10].

Bearing in mind the publication by UNESCO [13] for education institutions, including HEIs, it is recommended that all their processes should be based on sustainability principles. In specific terms, for SD practices in HEIs to be more effective, according to UNESCO, the institution as a whole has to be transformed. Such a whole-institution approach aims at mainstreaming sustainability into all aspects of the education institution, which involves rethinking the curriculum, campus operations, organizational activities, culture, student participation, leadership and management, community relationships and research [14].

In the literature, the concepts of sustainability and SD are commonly considered interchangeable, and sometimes as equivalent, e.g., [15,16]. This study, in the same line as stated in [17], assumes that sustainability is a principle, while SD relates to a social process involving choices and decisions towards sustainability. In other words, SD is the means to achieve sustainability, which is the final, long-term objective [16]. Therefore, HEIs have a fundamental role to perform in implementing and leading SD initiatives through the institutions' internal policies and practices.

Following this line of research, having as a vision the whole-institution approach mentioned above, the objective of this study is to fill this research gap in the perceptions of the stakeholders of Portuguese HEIs in relation to their SD practices, and, in turn, investigate how these practices can contribute to these stakeholders' QoL. Regarding HEIs' IC, there is still room and a need to understand how the management of IC can be articulated according to the existing SD practices in these institutions, so that the latter can function as mechanisms to identify existing gaps in HEIs' strategic reports and plans to be filled in the short and long-term.

Considering the above and the stakeholders' perspective, this study aims to fulfil the following objectives: (i) ascertain whether there is a direct, positive and significant relationship between HEIs' IC and their SD practices; and (ii) check whether HEIs' SD practices are directly, positively and significantly related to their stakeholders' QoL, thereby shedding light on a new perspective of the ongoing research on IC.

Considering the importance of IC, and having found no studies so far relating HEIs' IC with their own SD practices and QoL, considering stakeholders' perception, this study proposes to analyse SD practices, through economic, social, environmental and organizational dimensions, and the QoL of HEIs' stakeholders (students, teachers/researchers) through: (i) students' quality of academic life (QAL); and (ii) through the quality of work life (QWL) of teachers/researchers. This approach is of interest to both the scientific community and to HEI managers, as it is an innovative and relevant subject, never before studied, and may lead to better results for students and greater motivation among their collaborators.

To fulfil these aims, there is, first, a brief overview of the evolution of the literature; firstly, on IC in HEIs and the relationship with SD practices, and secondly on the relationship between SD practices in HEIs and QoL. Then, two models are presented and tested through quantitative analysis, gathering primary data from students and teachers/researchers at seven Portuguese HEIs, using a structural equation model (SEM) and the partial least-squares (PLS) method in order to verify the robustness of those relationships. Finally, the conclusions, implications and limitations of the study are elaborated.

2. Intellectual Capital of HEIs and Sustainable Development Practices

As entities involved in the creation and spread of knowledge, HEIs have been taking on a more entrepreneurial role, involving networking and international collaboration, and are increasingly more articulated in regard to critical issues of sustainability and social change, as stated by [18]. These authors highlight the fact of this idea being in line with the perspective of the fourth stage of IC, i.e., the creation of knowledge focused on the ecosystem. Studies related to this stage defend a change in approach to understanding the drivers of wealth creation, based on a balance of intellectual and financial measures, in order to create a more holistic vision of the national innovation capacity and the renewal of society and politics [19].

Therefore, monitoring IC is a way of measuring and controlling intangible and fundamental elements for these organisations [7], at the same time as ensuring SD. Similarly, the management of IC and its importance in HEIs are examples of issues studied by various authors, e.g., [20], as well as the association between HEIs' IC and sustainability [7], and how SD can be integrated into HEIs' practices [21].

Many HEIs have begun to incorporate SD practices into their systems and a variety of sustainability assessment tools have been developed to support HEIs in systematically measuring, auditing, benchmarking, and communicating SD efforts to their stakeholders [22]. As an example, it may refer to the Association for the Advancement of Sustainability in Higher Education (AASHE), which began in 2006. AASHE empowers higher education faculties, administrators, staff and students to be effective change agents and drivers of sustainability innovation [23]. This association developed the Sustainability Tracking, Assessment and Rating System (STARS), which is a framework for colleges and universities for measuring their own sustainability and it is the product of an extensive stakeholder engagement process. This approach fits also with the key elements for whole-institution approaches mentioned by UNESCO [13], as it allows the HEI, together with its stakeholders (e.g., teachers/researchers and students), to jointly develop the vision and strategic plan to implement SD practices in the whole institution.

However, despite the role played by HEIs in promoting SD being recognised as essential, e.g., [24,25], with examples of SD practices in different dimensions (e.g., environmental, economic, social and organisational) worldwide and integrated in HEIs' main activities (e.g., education, research, operations, social involvement and governance/culture) [26], some articles, e.g., [10,21,27] have pointed out that SD practices vary considerably from one HEI to another, and as for results found for SD practices implemented in Portuguese HEIs specifically, this is still at an early stage.

The potential impacts on HEIs' SD are based on practices related to economic growth, changes in social and business practices, social cohesion, contributions to climate change, sustainable human behaviours and urban development [22]. The most explored dimensions related to SD, in the HEI context, are environmental, social/cultural, economic and organisational/educational/political. These dimensions are integrated into activities related to teaching, research, campus operations, community actions, assessment and the drawing up of reports [10]. Some authors identify three fundamental dimensions of SD: economic, social, and environmental [11,28]. However, it is increasingly common to find other dimensions, such as organisational, e.g., [29,30] and cultural, e.g., [29,30]. In the specific case of HEIs, the following dimensions were proposed in regard to the implementation of SD practices: environmental, economic, social/cultural, and organisational/educational/political, e.g., [10,24,31]. This study considers that SD practices operate in the following four dimensions: economic, social, environmental, and organisational.

In the same vein of [10], the economic dimension of SD involves practices of economic viability and considers economic needs (e.g., concern about economic performance, plans and actions to improve energy efficiency, and budgeting for practices that promote sustainable development). The social/cultural dimension concerns the actions of an organisation's human resources or the surrounding community (e.g., policies to promote equality and diversity, developing and participating in recreational, cultural or sporting activities, concerns and initiatives regarding social inclusion and scientific initiatives directed towards the outside community). The environmental dimension proposes including environmental concerns and practices in the institution's strategy (e.g., constructing sustainable buildings on campus, separating waste and sending it for recycling and equipment to generate renewable energy). Finally, the organisational dimension concerns how institutions mould their behaviour and values, and how the different stakeholders perceive and if they are satisfied with approaches and objectives related to sustainable development (e.g., declarations and statements on the HEI's views and formal documents on values, strategy, transparency in governance and ethical commitments).

Some authors highlight the importance of stakeholders' perceptions in research related to HEIs' SD, e.g., [12,32]. The discussion on sustainability is based on stakeholder theory [17]. Stakeholder theory aims to analyse the relationship between an organization and the economic and social actors (individually or collectively) that affect, are affected by, and have interests in the procedural and substantive aspects of corporate activities. The management principles of stakeholder theory are reflected in the new model of HEIs' governance through the presence of different internal and external stakeholders in the various management organs [33]. In addition, the stakeholder satisfaction affects organisations' competitiveness and image, with stakeholders' needs and expectations affecting the organisation's management system [34]. The same author concluded that a wide understanding and incorporation of these needs in the management system can contribute to achieving the objectives proposed and increasing stakeholders' QoL.

HEIs' IC can be one of the key elements in promoting SD [7], and in its generalized expression, the SD concept represents an evolutionary coordination of various concerns linked to well-being, such as social, cultural, economic and environmental concerns [35]. Furthermore, these authors emphasize that sustainable behaviour is conceived as actions that contribute to the QoL of current and future generations. HEIs' IC is identified in various studies as a composite of human capital, structural capital and relational capital, e.g., [36,37]. This capital approach differentiates from the one presented in the scope of the theory of capital developed by Pierre Bourdieu, since the latter considers other types of capital, such as economic capital, cultural capital, social capital and symbolic capital [38]. In this scope, it deserves to be outlined that, in the case of social capital, Bourdieu [38] refers to networks as a form of social capital, but also incorporates the nature of culture and how it is reproduced and transformed, as well as how it connects to social stratification and the reproduction and exercise of power, which is connected with the mode of how human capital evolves in the scope of social systems, as a heritage and a reproductive mechanism of social stratification. In this study, HEIs' IC stems from the triad of capitals; human, structural, and relational, having as reference the studies of [36,37].

In the HEI context, human capital is the sum of explicit and tacit knowledge held by all the human resources existing in the institution (teaching, research and development, management, directing

and administrative staff in all services), acquired through both formal and non-formal education and the training processes included in their activities [20,39,40]. According to the vision expressed in [41], human capital can play an important role in SD practices, through the intermediation between the various stakeholders and regional actors, through the demonstration of good practices such as developing management activities, strategic planning, construction projects, minimizing waste and practices of energy efficiency and sustainability, and responsible purchasing programmes, and through good, environmentally-friendly initiatives with an impact on the campus. Leaders can offer incentives to recognise and reward staff for becoming involved in groups leading SD in the academic and regional community. Notably, [10] concluded that, in general, Portuguese HEIs value and stimulate professional and personal development (e.g., vocational training, academic training), in order to ensure the adoption of good practices within the institution.

Additionally, in the HEI context, structural capital includes all explicit knowledge interrelated with the internal processes of the promotion, communication, and management of scientific and technical knowledge in the organisation, which spans both organisational aspects (operating environments derived from the interactions between research management and the organisation of processes, organisational routines, corporate culture and values, and internal procedures, within the scope of quality and information systems, among others), and technological aspects (technological resources available in the university, such as bibliographic and documentary resources, archives, technical developments, patents, licenses, software, and databases, among others) [39,42]. For example, in [43], the structural capital was related to the SD practices towards the improvement of some organisational processes and practices, such as structural improvements based on new technologies (databases, intellectual property) and organizational culture based on the management of environmental sustainability practices. However, in [10], it was stressed that, until now, in Portuguese HEIs the focus has been on processes related to the separation of waste and its forwarding for recycling and plans to reduce the production of waste (e.g., paper, plastic, metal, oils, batteries), so as to ensure SD.

Concerning relational capital in the HEI context, this reflects the extensive collection of economic, political and institutional relationships that have been built up and maintained between HEIs and their non-academic partners (companies, non-governmental organisations, local government and society in general), as well as the perceptions others hold of the institution in terms of its image, attractiveness, trustworthiness and security, among others [40,42].

Relational capital is the connector between the HEI and its various stakeholders, partners, firms, institutions, etc. In [44], the importance of relational capital for SD was revealed, in that it stimulates people's participative and cooperative capacity and makes them responsible for community development, through promotion and interaction between people, structures and institutions, sustained by mutual trust, tolerance and cooperation, as well as mutual respect, civility and participation. Initiatives related to SD in education, research, operations and the outside community help HEIs to respond to various challenges, attracting resources, lowering costs, promoting more effective management and tackling new challenges in society [45]. All this will also contribute to a more positive image of the HEI, attracting more students, promoting quality and excellence, and thereby contributing to the HEI's internationalization.

Considering the above, the following research hypothesis is formulated:

Hypothesis 1 (H1). *HEIs' IC has a positive and significant influence on the institution's sustainable development practices.*

3. Sustainable Development Practices and Quality of Life

The idea that economic development must be sustainable implies recognising the basic idea that natural resources are scarce and limited, therefore accepting that different socio-economic activities must be restrained [46]. However, according to the same authors, the concept extends ideologically to

the cultural and social relations involved in SD processes, including those affecting social well-being and QoL.

The concept of sustainability emphasizes the idea of human behaviours that allow individuals in the present and future to satisfy their needs without exceeding nature's capacity to recover the resources extracted from it [47,48]. These human behaviours involve psychological tendencies and behaviours that show concern about conditions in the physical environment and the completeness of the social environment [46].

In turn, QoL covers a number of indicators portraying various environmental, social, economic and subjective factors [49]. Therefore, a better QoL can be achieved in societies that enjoy a well preserved and constructed natural environment, as well as good governance and good levels of physical health, economy and subjective well-being, as highlighted by [46]. The same authors conclude that, consequently, the interactions between human beings and their physical and social environment should create high levels of satisfaction with these factors, besides well-being and happiness, if these interactions are pro-sustainable-that is, if they are committed with aspects concerning sustainability issues in their daily lives, such as shared value, social welfare and environmentally friendly practices. Furthermore, Moser [50] claimed that a pro-sustainable relationship with the social and physical environment results in satisfying humans' needs and conserving that same environment. Taking care of the environment, conserving and preserving it, is a commitment that all organizations will be urged to make in the short term because it raises the QoL of individuals in the workplace (microenvironment) and those who inhabit the global space (macroenvironment) [51]. SD practices imply the improvement of QoL through satisfaction with many aspects of life, such as education, justice, community participation and recreation [52]. Thus, environmental, cultural and economic factors can interfere with the degree of satisfaction with life, especially if biological needs, safety aspects, social aspects, and psychological aspects have been minimally affected [53].

HEIs contribute to SD through their teaching, research, extension and management practices [54]. Following the statements of [55], a sustainable HEI is one that values the quality of teaching, implements practices aimed at improving the quality of academic life (QAL) and is concerned about managing the use of natural resources. Therefore, in the perspective expressed in [56], HEIs should integrate the principles and practices of sustainability, as that vision and institutional orientation is revealed to be important in undertaking a necessary process of awareness among the academic community and to help decision-making, planning and operational processes.

The psychology of sustainability and SD [57] looks at sustainability not only in terms of the ecological and socio-economic environment, but also in terms of improving everyone's QoL, as mentioned in [58]. In this line of thought, the same authors highlighted that it is essential to analyse the quality of working life (QWL), as professional activity plays a fundamental role in determining employees' physical and mental health and well-being. Similarly, in [59] it was claimed that SD can only materialize in work environments that promote employees' well-being.

As an indicator of well-being, QoL is, today, also an extremely important factor [9], as in its wider sense it involves the components of individuals' lives related to their financial situation, health, interaction with the environment, social relations, affective life, leisure, satisfaction with life and other aspects. QoL is a concept that has inspired much research in the last few decades and had a strong influence on social and political trends applied to various fields, such as urban and regional planning, health promotion and also in social and economic investigation [60].

The literature available in this field can be divided into two types of studies: (i) the studies that consider QoL as a set of purely economic factors (GDP per capita, cost of living, employment, scale economies, etc.), determinants of the growth, decline and competitiveness of organisations [61]; and (ii) as a set of non-economic factors, as a subject of research in the quality of academic life of students (QAL) (satisfaction with services, emotions felt in campus, etc.), e.g., [62–68], or as a factor for assessing quality of work life (needs for satisfaction in a physical and emotional line) (QWL) [69–73].

Some studies demonstrate the relationship between HEIs' IC and SD [7,22,74], and between HEIs' IC and QoL [9,75], but after checking some recent literature reviews regarding IC [19] and searching the most renowned databases (e.g., Web of Science and Scopus), there were no studies which aimed to simultaneously analyse IC, SD and QoL in HEIs, considering their stakeholders' perceptions.

The SD practices can be related to QoL. For example, the social dimension of SD in the HEI context is associated with the quality of work and the quality of life in the academic community [10]; SD in HEIs is a type of development that ensures individuals' QoL through the conservation and preservation of the environment [54]. For example, in [50] it is stated that problems related to noise and environmental pollution are frequently mentioned by individuals as threats to their QoL [76]. If HEIs promote SD practices on campus, such as noise reduction, diminishing the use of paper and recycling campaigns by providing containers for this purpose, they can contribute to greater satisfaction among students.

Recent years have witnessed an exponential increase in the number of studies on QoL in educational environments in relation to the different individuals and groups therein [63,67,68,77–79], more specifically in the areas of students' QAL, e.g., [64–68] and the QWL of teachers and researchers [70,71]. QAL can be assessed in terms of feelings of global satisfaction with the student's experience of life at university [80]. QAL concerns the degree of need for satisfaction and the experiences that create positive emotions in the context of university life experienced by students [81]. Furthermore, the QAL corresponds to a sub-domain of QoL in general, expressed through the satisfaction revealed with the domain of university life [63,80]. These same authors conceptualized QAL as students' general feeling of satisfaction with the experience of university life through the presence of positive sentiments and the absence of negative ones.

QAL has also been measured as a composite of cognitive assessment, i.e., satisfaction of needs in life in the HEI, and affective assessment, referring to positive and negative affective experiences occurring throughout the period of studies at the HEI [64–68] This study adopts the view proposed in [80] regarding QAL, for whom this is defined according to the global feelings of satisfaction a student experiences in relation to university life. As QAL is measured through the determinants of satisfaction with HEI life [68], these SD practices are expected to contribute positively to students' QAL.

According to some studies, SD has a relationship with QoL [46], and with QAL [10]. If QAL measures students' QoL in the HEI context, then QAL can supposedly be affected by HEIs' SD practices. As mentioned, no studies are known to relate HEIs' sustainable development to students' QAL. For greater understanding of this connection, the following research hypothesis is formulated:

Hypothesis 2 (H2). Sustainable development practices in HEIs have a positive and significant influence on students' QAL.

QWL considers the organisational environment according to a wide range of needs for staff well-being at the workplace [69,82]. QWL has a multi-layered, dynamic structure covering different concepts such as safety at work, reward systems, workflows, opportunities for educational and work development, and participation in decision-making processes [70].

In [69], it was stated that QWL describes human resources' satisfaction of seven principal needs, namely health and safety, economic and family, social, esteem, self-updating, knowledge and aesthetic needs. However, in [80], these measures were conceptualized for updating QWL in terms of satisfaction composed of two sets of needs. Firstly, we have the composite of satisfaction of lower order needs, which includes satisfaction of health and safety needs, as well as satisfaction with economic and family needs. Secondly, we consider the composite of higher order needs, which includes satisfaction of social, esteem, self-updating, knowledge and aesthetic needs. These authors argued that the examination of the relative effectiveness of higher and lower order needs helps to prioritize the satisfaction of workers' needs. This method was also validated in [71].

QWL has been studied and defined by various authors e.g., [69,80,83,84]. However, the present study focuses specifically on HEIs, using the definition proposed in [85] for conceptualizing QWL—that

is, the staff's satisfaction with a variety of needs through resources, activities and results arising from participation in the workplace; and for measures of QWL, the updated needs proposed by the first authors were stated in [86]. Various studies in the field of QWL dealt with HEI teachers/researchers [70,71,73], but so far no studies are known to have dealt with the relationship between sustainable development practices in HEIs and the QWL of teachers/researchers, and so this is an innovative approach with potential interest for both researchers and HEI managers.

As already stated, SD affects QoL [46]. For example, recycling paper and other office material can make people feel that they are contributing to improving the state of the planet, and, as such, feel prouder of the place they work in, i.e., greater satisfaction and, therefore, a better QWL. A similar feeling is hoped for when the HEI contributes proactively to the balanced development of society through actions of social responsibility. A widely used definition of social responsibility for SD is that of the World Business Council [87], according to which corporate social responsibility is organisations' continued commitment to behave ethically and contribute to economic development, improving the QoL of the workforce and their families, as well as that of the local community and society in general. For example, in [88], the social dimension of SD was positioned as a motivational factor for the staff working in the organisation. Therefore, teachers and researchers' involvement in actions to help the community will make them feel better and consequently have a better QWL.

If QWL is people's response or affective reaction to the organisational system [89] and measures teachers and researchers' QoL in the HEI context, then QWL can supposedly be affected by HEIs' sustainable development. To deepen the understanding of this connection, the following research hypothesis is considered:

Hypothesis 3 (H3). Sustainable development practices in HEIs have a positive and significant influence on the QWL of teachers and researchers.

Considering the literature review and the research hypotheses formulated, two models of analysis are proposed in Figure 1. Model 1 is concerned with students' perceptions, and Model 2 relates to teachers and researchers' perceptions.



Figure 1. Cont.



Figure 1. Influence of higher education institutions' (HEIs) intellectual capital (IC) on their own sustainable development practices and on students' quality of academic life (QAL) and teachers/researchers' quality of working life (QWL). Source: Own elaboration.

4. Research Methodology

With the motivation of accomplishing the objectives, this study was analytical and correlational, because it sought to explore the variables and the relationships between them, and it was cross-sectional because the samples were taken in a single period. The purpose of the study was descriptive because it aimed to discriminate the determining factors possibly associated with the phenomenon under study [90]. Through a quantitative, objectivist and, therefore, deductive approach, this research was supported by models built on results and previous research, with quantitative indicators collected through a questionnaire.

4.1. Unit of Analysis

The subject of study corresponded to the entirety of the diverse internal and external stakeholders of Portuguese State HEIs. Based on [91], students and teachers/researchers were selected for this study, given their importance and relevance for the study objectives. The selection of this population was justified as it ensured a diversified sample with the representation of one HEI per region NUTS II level (The Nomenclature of Territorial Units for Statistics (NUTS) is developed by Eurostat, and employed in Portugal for statistical purposes (https://ec.europa.eu/eurostat) and considered the entirety of these seven HEIs as a suitable laboratory to test the effects of IC on QAL and QWL. Due to limitations, in terms of data access, the sample's design incorporated seven HEIs, for the total number of seven NUTS II regions in Portugal, in order to ensure the total geographical coverture of Portugal, including five regions from continental Portugal: North, Centre, Metropolitan area of Lisbon, Alentejo, and Algarve, and also two autonomous regions: Madeira and Azores.

The Portuguese higher education system (public and private) is a binary system, where we can find the university education that is oriented towards the supply of solid scientific formation, joining efforts and competences of teaching and research units and the polytechnic education that is concentrated especially on applied sciences, vocational training and advanced, professionally-oriented technical training. In the current study, the decision was taken to focus on public university education, since they correspond to the dominant share of institutions providing higher education and research services in Portugal; this option facilitated the comparative analysis and representativeness of the results.

4.2. Instrument for Data Collection and Variables

Quantitative data were collected through a Questionnaire A, for students, and a Questionnaire B, for teachers/researchers, resorting to structured, closed questions. A seven-point Likert scale was used for the answers. This scale seems to be the most correct in this study as the respondents build acceptance levels according to their experiences and social influences, giving the opportunity to give clear answers instead of neutral or ambiguous answers. This type of scale has already been used in similar studies related to IC, e.g., [20], QAL, e.g., [68], and QWL, e.g., [71].

Both questionnaires were pre-tested to ensure that all the questions were understood and accepted in the same way by all respondents. Subsequently, some of the items of both questionnaires were adapted to improve comprehension.

4.2.1. Variables for IC

IC was measured considering the dominant triad formed of human capital (HC), structural capital (SC) and relational capital (RC), in line with the multidimensional analysis suggested in [19]. To determine the type of IC indicators, the methodological design proposed in [92] was followed. The 32 key indicators used for IC are presented in Supplementary Materials Annex 2.

4.2.2. Variables for Sustainable Development

The variables to measure the SD practices in HEIs are based on the study developed in [10]. The dimensions used are economic, environmental, social and organisational (see Supplementary Materials Annex 2).

4.2.3. Variables for QAL

Concerning QAL, as mentioned in the literature review, previous studies such as [64,65,68] were the cornerstones.

For the cognitive component, the scale proposed in [64,65] was adopted; and for the affective component, the criterion adopted in [64,65] was used, resorting to the scale proposed in [93]. Both criteria (cognitive component, affective component) have already been used and validated in previous studies, such as [67,68] (see Supplementary Materials Annex 2).

4.2.4. Variables for QWL

As for QWL, several studies were considered, e.g., [69,71,80], incorporating the correspondent adjustments (see Supplementary Materials Annex 2).

4.3. Sample and Data Collection Procedure

The definitive sample was collected between November 2017 and February 2018, in two phases. In the first phase, the questionnaires were sent by e-mail to seven Portuguese HEIs (see Table 1), via the Communication and Image Department at the University of Beira Interior. This e-mail, containing a link to the questionnaire, explained the purpose of the study, ensuring that participation was voluntary, anonymous and confidential.

In the second phase, as the first phase did not result in a representative sample, some paper questionnaires were administered in the classroom. The potential bias of students' non-response was assessed through *t*-tests, with no significant differences being observed between the two groups.

The participants in this study were 749 students and 587 teachers/researchers, having eliminated eleven student questionnaires as they were not correctly completed. The final sample comprised 738 students and 587 teachers/researchers (Supplementary Materials Annex 3 shows the sample characterisation with distribution of respondent students and teachers/researchers by HEI, area of study, gender and age group).

Table 1. HEIs, geographical area of location, weight by HEI, total of students (S) and teachers/researchers (T/R) samples.

HEIs	Region	Weigh (%	t HEIs 6)	Total S Colle	bample ected	Optimum Sample Size *	
	(NUTS II)	S	T/R	S	T/R	S	T/R
ISCTE-Instituto U. Lisboa	Metropolitan Area of Lisbon	16.67	13.1	118	77	109	70
U Açores	Autonomous Region of Açores	5.29	6.6	48	39	35	30
U. Algarve	Algarve	14.61	10.7	98	64	96	64
U. Beira Interior	Centre	12.86	17.6	132	105	84	105
U. Évora	Alentejo	12.16	13.8	88	82	80	83
U. Madeira	Autonomous Region of Madeira	5.27	6,8	35	41	35	41
U. Minho	North	33.14	31.4	219	179	217	173
Total		100	100	738	587	656	566

Legend: S = Students; T/R = Teachers/Researchers. * The optimal sample size to be collected at each participating HEI was determined for a confidence level of 99% and considering a sampling error of 5%, as proposed by [94]. Source: Own elaboration.

5. Presentation and Discussion of the Results

Prior to the analysis of the evidence provided by Model 1 and Model 2, the descriptive statistics of the variables studied were contemplated, as well as the distribution of the mean values in relation to students and teachers/researchers, which was found to be quite homogeneous in both models. The correlational relation between the control variables was also analysed. The results, presented in Table 2, show that the distribution of the mean values is quite homogeneous and all the correlations are statistically significant (p < 0.01), with values below or very close to 0.750, not indicating potential problems of autocorrelation.

Table 2. Descriptive statistics and correlation between variables Model 1 and Model 2.

Model 1	Human Capital	Structural Capital	Relational Capital	Economic	Environmental	Social	Organisational	QAL
Human capital	1							
Structural capital	0.716 **	1						
Relational capital	0.729 **	0.835 **	1					
Economic	0.584 **	0.731 **	0.731 **	1				
Environmental	0.560 **	0.618 **	0.651 **	0.627 **	1			
Social	0.492 **	0.594 **	0.582 **	0.650 **	0.591 **	1		
Organisational	0.694 **	0.742 **	0.804 **	0.697 **	0.690 **	0.572 **	1	
QAL	0.586 **	0.532 **	0.577 **	0.444 **	0.468 **	0.407 **	0.609 **	1
Average	4.836	4.752	4.905	4.634	5.240	4.890	5.119	4.836
Variance	0.560	0.962	1.005	1.145	1.430	1.542	1.470	0.560
Model 2	Human Capital	Structural Capital	Relational Capital	Economic	Environmental	Social	Organisational	QWL
Human capital	1							
Structural capital	0.736 **	1						
Relational capital	0.688 **	0.824 **	1					
Economic	0.652 **	0.737 **	0.796 **	1				
Environmental	0.473 **	0.619 **	0.698 **	0.678 **	1			
Social	0.435 **	0.545 **	0.603 **	0.582 **	0.528 **	1		
Organisational	0.422 **	0.569 **	0.586 **	0.610 **	0.582 **	0.565 **	1	
QAL	0.349 **	0.391 **	0.391 **	0.398 **	0.404 **	0.228 **	0.284 **	1
Average	4.256	4.187	4.450	4.802	5.040	4.600	4.377	4.736
Variance	0.705	0.995	0.931	1.133	1.782	2.036	1.321	0.929

** The correlation is significant at the level of 0.01 (2 extremities). Source: Own elaboration.

The data were analysed using a selected specification of a structural equation model (SEM), using the partial least squares (PLS) method, SEM–PLS. Considering the statement presented by Hair et al. [95], the PLS assumes no distribution to the data and is relatively robust against distribution deviations. However, the same authors stated that researchers should still examine PLS–SEM results

carefully when distributions deviate substantially from normal. In accordance with this, absolute skewness and/or kurtosis values of greater than one are indicative of non-normal data. Taking into account what was mentioned by Hair el al. [95], in this case, regarding skewness and kurtosis statistics, they do not provide evidence of a non-normal distribution. In both models, the kurtosis and skewness values of the indicators are within the acceptable range of -1 and +1. The only exception is the ORG indicator, in M1, which has a skewness of -1.113 and a kurtosis of 1.365, and thus exhibits a slight degree of non-normality. However, as the degree of skewness and kurtosis is not severe and because ORG is one of four indicators measuring the (reflexive) SD construct, this deviation from normality is not considered a problem and the indicator is retained.

The variance inflation factor (VIF) was also used to diagnose collinearity, and it was found that the variance value of each indicator is no higher than 2.7, signalling no potential multicollinearity issues.

5.1. Model Estimation

According to the procedures defined in [95], SEM–PLS is used mainly to develop theories in exploratory studies focusing on explaining the variance in dependent variables when examining the model. SmartPLS (v 3.2.7) software [96] was used to estimate the parameters, using bootstrapping of 5000 samples to obtain their significance [95].

The PLS model was assessed in three stages: (i) assessment of the global model was determined; (ii) the reliability/validity of the measurement model was checked; and (iii) the meaning of the paths (relations between constructs) within the structural model was assessed [97].

The initial measurement model of this study denotes reflexive characteristics (see Supplementary Materials Annex 2), containing two multidimensional constructs (second-order constructs) and nine latent variables (first-order constructs) that cannot be observed or measured directly, and can only be inferred through their observable variables, i.e., the forty-five indicators (see Supplementary Materials Annex 1).

After determining the values and adjusting the constructs of QAL and QWL, considering the literature review, the two models proposed were analysed. As in both models there is a second-order construct, this analysis will follow a two-step approach as recommended by [98], that is: (i) treatment of M1 and M2 only with the first-order constructs (models M1a and M2a); and (ii) treatment of the models incorporating the aggregate scores as an indicator of the second-order constructs (Models M1b and M2b).

Stage 1: Treatment of Models M1a and M2a. In this stage, the global model and measurement model will be analysed.

Assessment of the global model requires the use of quality adjustment measures. After estimating the two models (M1a and M2a) using SmartPls [96], it was found necessary to adjust both models, since the values presented did not agree with recommendations in the literature of reference. It was found adequate to drop the indicators with the smallest loading values that were detracting from the result. The models were estimated until the standardized root mean square residual (SRMR) value in both models reached the cut-off value of >0.08 [99]. Regarding M1a, the indicators of HC3, HC5, HC6, HC7, HC8, HC9, SC7, RC2, RC3 and RC8 were withdrawn; and from M2a the indicators of HC1, HC3, HC4, HC5, HC7, HC8, HC9, HC10, SC1, SC2, SC3, SC7, SC8, SC10, RC1, RC2, RC3 and RC4 were withdrawn. Figure 2 shows the final M1a and M2a models.



Figure 2. Models M1a and M2a adjusted only with the first-order constructs relations. Source: Own elaboration.

As observed in Table 3, the original SRMR value in both models was <0.08 [99] and all the deviations were insignificant because 95% of the bootstrap quantile (HI95) of the value of the three measures were greater than the original values [97].

Fit Measures	Origina	al Value	H	[95
in measures	M1a	M2a	M1a	M2a
SRMR	0.059	0.069	0.064	0.077
dULS	1.866	1.444	2.171	1.790
dG	0.630	0.825	0.703	0.893

Table 3. Quality of adjustment (estimated model and saturated model).

Legend: SRMR: Standardized root mean square residual; dULS: unweighted least squares discrepancy; dG: geodesic discrepancy. Source: Own elaboration.

In the assessment of the measurement model, only the reflexive indicators will be analysed, because the models do not have formative indicators. All the measures have as reference the recent studies of [100,101].

In the analysis of reflexive indicators, consideration should be given to: (i) reflexive indicator loadings; (ii) internal consistency reliability; (iii) convergent validity; and (iv) assessment of discriminant validity.

Regarding the loading values (see Table 4) of M1a and M2a, all the indicators are seen to present values above 0.70, as recommended, except for one indicator from M2a. However, as this indicator is close to 0.70, we decided to retain it, in agreement with the recommendation present in [102,103] by considering that it is necessary in the model.

Table 4 also presents the results of the analysis of internal consistency reliability, as well as the Cronbach alpha value. Interpretation of the coefficients of these analyses should also not present values under 0.70 (or 0.60 in exploratory research). All the variables also satisfy the requirements of the Dijkstra–Henseler indicator (ρA) (*rho_A*), since the values obtained by calculating the indicator are above the reference of 0.70.

The assessment of convergent validity is through the average variance extracted (AVE), which must be equal to or above 0.50. The result, presented in the same table, shows that the AVE value agrees with the literature of reference, i.e., above 0.5.

The discriminant validity is better detected through the calculation of the heterotrait–monotrait (HTMT) ratio, which for conceptually similar constructs must be HTMT < 0.90, and, for conceptually different constructs, HTMT < 0.85. Table 5 confirms that the result of this last analysis also agrees with the authors' recommendation, except for two values in both models that are very close to 0.90. In addition to these guidelines, to complement this result, researchers can formally test whether the HTMT value is significantly lower than unity (1) using bootstrapping, and in both cases Table 6 confirms that the result of this last analysis also agrees with these authors' recommendations, i.e., no interval has the value of one.

		M1a						M2a			
Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	pc	AVE	Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	Pc	AVE
HC		0.815	0.830	0.870	0.573	HC		0.727	0.734	0.848	0.654
HC1	0.742					HC2	0.869				
HC2	0.729					HC6	0.679				
HC4	0.733					HC11	0.863				
HC10	0.820					SC		0.852	0.855	0.895	0.631
HC11	0.757					SC11	0.747				
SC		0.910	0.912	0.925	0.554	SC4	0.821				
RC1	0.781					SC5	0.834				
RC10	0.774					SC6	0.854				
RC4	0.720					SC9	0.705				
RC5	0.801					RC		0.862	0.867	0.897	0.591
RC6	0.791					RC10	0.799				
RC7	0.801					RC5	0.814				
RC9	0.804					RC6	0.771				

 Table 4. Analysis of the measuring model (loadings, internal consistency and reliability, Dijkstra–Henseler indicator, composite reliability and average variance extracted (AVE) of models M1a and M2a.

		M1a						M2a			
Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	pc	AVE	Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	Pc	AVE
RC		0.894	0.896	0.917	0.612	RC7	0.755				
RC1	0.729					RC8	0.703				
RC2	0.701					RC9	0.768				
RC3	0.780					ECO		0.760	0.807	0.861	0.674
RC4	0.702					ECO1	0.797				
RC5	0.735					ECO2	0.894				
RC6	0.760					ECO3	0.767				
RC7	0.745					ENV	1.000				
RC8	0.810					SOC	1.000	1.000	1.000	1.000	1.000
RC9	0.749					ORG					
RC10	0.724					ORG1	0.864				
ECO		0.820	0.838	0.892	0.734	ORG2	0.841				
ECO1	0.882					ORG3	0.884				
ECO2	0.885					ORG4	0.805				
ECO3	0.801					QWL	1.000	1.000	1.000	1.000	1.000
ENV	1.000	1.000	1.000	1.000	1.000						
SOC	1.000	1.000	1.000	1.000	1.000						
ORG	0.921	0.922	0.944	0.808	0.921						
ORG1	0.908										
ORG2	0.907										
ORG3	0.897										
ORG4	0.883										
QAL	1.000	1.000	1.000	1.000	1.000						

Table 4. Cont.

Source: Own elaboration.

Table 5. Heterotrait-monotrait (HTMT) ratio of models M1a and M2a.

M1a	HC	SC	RC	ECO	ENV	SOC	ORG	QAL
HC								
SC	0.905							
RC	0.854	0.925						
ECO	0.776	0.844	0.852					
ENV	0.555	0.652	0.685	0.694				
SOC	0.526	0.870	0.608	0.720	0.591			
ORG	0.715	0.814	0.623	0.801	0.719	0.597		
QAL	0.238	0.263	0.280	0.186	0.220	0.168	0.300	
M2a	HC	SC	RC	ECO	ENV	SOC	ORG	QAL
HC								
SC	0.901							
RC	0.888	0.906						
ECO	0.565	0.676	0.689					
ENV	0.672	0.709	0.738	0.673				
SOC	0.559	0.563	0.640	0.652	0.527			
ORG	0.818	0.835	0.883	0.754	0.731	0.626		
QWL	0.059	0.065	0.093	0.068	0.116	0.033	0.092	

Source: Own elaboration.

Table 6. Heterotrait-monotrait ((HTMT) ratio of models M1a a	nd M2a using bootstrapping.
----------------------------------	------------------------------	-----------------------------

Variables M1a	Original Sample (O)	Sample Mean (M)	2.5%	97.5%	Original Sample (O)	Sample Mean (M)	Bias	2.5%	97.5%
HC -> ECO	0.103	0.105	0.018	0.197	0.103	0.105	0.002	0.016	0.195
HC -> ENV	-0.039	-0.040	-0.139	0.061	-0.039	-0.040	-0.001	-0.139	0.061
HC -> SOC	-0.009	-0.008	-0.114	0.098	-0.009	-0.008	0.000	-0.114	0.098
HC -> ORG	0.000	0.000	-0.080	0.076	0.000	0.000	0.000	-0.081	0.075
SC -> ECO	0.341	0.342	0.229	0.450	0.341	0.342	0.001	0.227	0.447
Structural Capital -> Environmental	0.306	0.307	0.177	0.434	0.306	0.307	0.001	0.173	0.430
Structural Capital -> Organisational	0.292	0.293	0.179	0.410	0.292	0.293	0.001	0.177	0.406
Structural Capital -> Social	0.379	0.380	0.232	0.521	0.379	0.380	0.001	0.227	0.518
Relational Capital -> Economic	0.374	0.372	0.269	0.476	0.374	0.372	-0.002	0.274	0.480
Relational Capital -> Environmental	0.422	0.421	0.303	0.541	0.422	0.421	-0.001	0.305	0.542
Relational Capital -> Organizational	0.548	0.547	0.442	0.650	0.548	0.547	-0.001	0.444	0.650
Relational Capital -> Social	0.266	0.265	0.132	0.396	0.266	0.265	-0.001	0.134	0.399
ECO -> QAL	-0.081	-0.080	-0.208	0.052	-0.081	-0.080	0.001	-0.211	0.049
ENV -> QAL	0.056	0.056	-0.039	0.152	0.056	0.056	0.000	-0.039	0.152
SOC -> QAL	0.017	0.018	-0.072	0.106	0.017	0.018	0.001	-0.074	0.104
ORG -> QAL	0.297	0.297	0.184	0.408	0.297	0.297	0.000	0.184	0.407
Variables M2a	Original Sample (O)	Sample Mean (M)	2.5%	97.5%	Original Sample (O)	Sample Mean (M)	Bias	2.5%	97.5%
Economic -> Quality of Work Life	-0.005	-0.005	-0.102	0.094	-0.005	-0.005	0.000	-0.103	0.093
Environmental -> Quality of Work Life	0.120	0.120	0.015	0.226	0.120	0.120	0.000	0.016	0.226
Human Capital -> Economic	-0.122	-0.121	-0.227	-0.020	-0.122	-0.121	0.001	-0.229	-0.022
Human Capital -> Environmental	0.038	0.039	-0.061	0.140	0.038	0.039	0.001	-0.061	0.138
Human Capital -> Organizational	0.090	0.091	-0.002	0.184	0.090	0.091	0.001	-0.003	0.182
Human Capital -> Social	0.066	0.067	-0.053	0.188	0.066	0.067	0.001	-0.055	0.186
Organisational -> Quality of Work Life	0.034	0.034	-0.084	0.152	0.034	0.034	0.000	-0.084	0.152
Relational Capital -> Economic	0.379	0.380	0.270	0.489	0.379	0.380	0.001	0.267	0.486
Relational Capital -> Environmental	0.461	0.462	0.372	0.547	0.461	0.462	0.001	0.370	0.546
Relational Capital -> Organisational	0.522	0.522	0.438	0.599	0.522	0.522	0.000	0.434	0.596
Relational Capital -> Social	0.471	0.469	0.355	0.572	0.471	0.469	-0.001	0.358	0.574
Social -> Quality of Work Life	-0.047	-0.047	-0.158	0.066	-0.047	-0.047	0.001	-0.160	0.065
Structural Capital -> Economic	0.360	0.360	0.240	0.477	0.360	0.360	0.000	0.239	0.476
Structural Capital -> Environmental	0.266	0.265	0.156	0.370	0.266	0.265	-0.002	0.158	0.371
Structural Capital -> Organisational	0.241	0.241	0.139	0.345	0.241	0.241	0.000	0.140	0.345
Structural Capital -> Social	0.100	0.100	-0.024	0.228	0.100	0.100	0.001	-0.025	0.227

Source: Own elaboration.

Stage 2: Treatment of Models M1b and M2b. As the proposed model adopts a different nomological structure, as suggested in [98] after calculating the results of the first order model (Models M1a and M2a), the measurement model of the second order models needs to be tested (Models M1b and M2b). The second order constructs (intellectual capital and sustainable development) incorporate the respective score of the first order dimension produced by SmartPLS [96]. After this stage, the structural model can be estimated [103].

For the measurement model, the procedure is exactly as in Stage 1. Analysis of Table 7 confirms that all the values are within the established parameters (>0.70) or very close to that value. The same table presents the results of the analysis of internal consistency and reliability, as well as Cronbach's alpha and AVE values. According to the literature of reference mentioned in Stage 1, all the values are within normality.

As for the heterotrait–monotrait (HTMT) ratio, the values are also within normality (see Tables 8 and 9).

		M1b						M2b			
Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	pc	AVE	Variable/ Indicator	Loading	Cronb. Alpha	Rho_A	Pc	AVE
IC		0.889	0.896	0.931	0.818	IC		0.879	0.890	0.925	0.805
HC	0.879					HC	0.876				
SC	0.926					SC	0.917				
RC	0.908					RC	0.897				
SD		0.858	0.870	0.903	0.700	SD		0.848	0.866	0.897	0.686
ECO	0.859					ECO	0.812				
ENV	0.823					ENV	0.853				
SOC	0.796					SOC	0.768				
ORG	0.869					ORG	0.876				
QAL	1.000	1.000	1.000	1.000	1.000	QAL	1.000	1.000	1.000	1.000	1.000

Table 7. Measuring model (loadings, internal consistency and reliability, Dijkstra-Henseler indicator, composite reliability and AVE) of models M1b and M2b.

Source: Own elaboration.

Table 8. Heterotrait-monotrait (HTMT) ratio of models M1b and M2b.

M1b	IC	SD	QWL	M2b	IC	SD	QWL		
IC				IC					
SD	0.887			SD	0.881				
QWL	0.282	0.259		QWL	0.070	0.100			

Source: Own elaboration.

Table 9. Heterotrait-monotrait (HTMT) ratio of models M1b and M2b using bootstrapping.

Variables M2a	Original Sample (O)	Sample Mean (M)	2.5%	97.5%	Original Sample (O)	Sample Mean (M)	Bias	2.5%	97.5%
IC -> SD	0.786	0.786	0.752	0.817	0.786	0.786	0.000	0.750	0.816
SD -> QAL	0.245	0.246	0.175	0.315	0.245	0.246	0.001	0.172	0.312
Variables M2b	Original Sample (O)	Sample Mean (M)	2.5%	97.5%	Original Sample (O)	Sample Mean (M)	Bias	2.5%	97.5%
IC -> SD SD -> QWL	0.779 0.097	0.779 0.097	0.753 0.032	0.803 0.160	0.779 0.097	0.779 0.097	0.001 0.000	0.751 0.031	0.802 0.159

Source: Own elaboration.

With no formative indicators to analyse, the structural model is assessed below.

5.2. Assessment of the Structural Model

Primary assessment of the structural model is carried out considering two assessment criteria, namely the determination coefficient statistic (\mathbb{R}^2), which measures the degree of model adjustment, and the statistical significances of the path coefficients [100,101]. As analysing structural equations through the PLS method consists of maximizing the value of the explained variance of the endogenous latent variables, the \mathbb{R}^2 value of the constructs should present a high value [100,101].

Regarding the estimation of the effect size (f^2) , according to [104] the reference values are: $0.02 \le f^2 < 0.15$: small effect; $0.15 \le f^2 < 0.35$: moderate effect; $f^2 \ge 0.35$: large effect.

The Stone–Geisser (Q^2) test is used as a criterion to measure the predictive relevance of the reflexive dependent constructs [105]. As in f^2 , values of 0.02, 0.15 and 0.35 indicate that an exogenous construction has small, moderate or large predictive relevance in a given endogenous construction.

Analysing the values presented in Table 10, the results confirm that the structural model of both models presents acceptable predictive relevance (R^2) for SD and weak for QAL and QWL, and that the values, also presented in this table for f^2 and Q^2 , are in accordance with the above-mentioned criteria.

Variables	R ²		f	2	Q ²		
	M1b	M2b	M1b	M2b	M1b	M2b	
IC			1.617 ***	1.541 ***			
SD	0.618	0.606	0.064 **	0.009 *	0.403 ***	0.388 ***	
QAL/QWL	0.060	0.009			0.054 *	0.008 *	

Table 10. Determination coefficient (R^2), estimate of the size effects (f^2), and predictive relevance (Q^2) of models M1b and M2b.

Legend: * $0.02 \le f^2/Q^2 < 0.15$: small. ** $0.15 \le f^2/Q^2 < 0.35$: moderate *** $f^2/Q^2 \ge 0.35$: large. Source: Own elaboration.

Concerning the robustness of the path coefficients, the reference value is above 0.2 [103,106]. The observation of Table 11 reveals that all the coefficients present a value above 0.2, meaning that there is robustness in the relationships tested, except for SD -> QWAL (p = 0.097). Considering the estimated values of the coefficients and corresponding t values, there is good adjustment of the data used to estimate the model and test the hypotheses studied, in terms of structural relations. The final models are presented in Figure 3.

Table 11. Robustness of the coefficients and level of significance of the structural relations of models M1b and M2b.

Structural Relations	Estimated Value		Sample Mean		Standard Deviation		t-Value	
	M1b	M2b	M1b	M2b	M1b	M2b	M1b	M2b
$H_1: \text{IC} \rightarrow \text{SD}$	0.786 †	0.779 †	0.786	0.779	0.017	0.015	47.328 ***	51.124 ***
H_2/H_3 : SD -> QAL/QWAL	0.245 †	0.097	0.246	0.097	0.036	0.039	6.804 ***	2.473 **

Legend: \dagger = Robustness of the coefficient because the value obtained is above 0.2. ** = level of significance 5% (>1.96) *** = level of significance 1% (>2.58). Source: Own elaboration.



Figure 3. Complete final structural models of M1b and M2b, and the respective weights and loadings. Source: Own elaboration.

M1b

5.3. Contrasting Literature and Empirical Findings

Regarding hypothesis 1, HEIs' IC has a direct and positive influence on HEIs' SD practices, and so this hypothesis is not rejected in either model (M1: 0.786; and M2: 0.779). Indeed, HEIs' IC is found to promote SD through the different dimensions of sustainability studied (economic, environmental, social and organisational), which is in line with the global idea proposed in [8] that HEIs are agents of change for sustainability, being associated with the pressing challenges society faces related to accelerated environmental changes, the shortage of resources, increased inequality and injustice, as well as rapid technological change and social change. These results are also in line with other studies [24,25] that recognised HEIs' IC as essential in promoting SD, with examples of SD practices in environmental, economic, social and organisational dimensions being integrated into activities related to education, research, operations, social involvement and governance/culture worldwide [26].

Concerning the dimensions of IC, these are very balanced, finding a higher value in both models for SC (M1: 0.926; and M2: 0.917). The perception of SC, linked essentially to physical structures and the campus, seems to be the one both students and teachers/researchers give most importance to. This result coincides with those of other authors, e.g., [107], for whom SC is the most important part of IC because it serves as a vehicle to convert staff's personal knowledge into value. In addition, HC has the lowest value (M1: 0.879 and M2: 0.876). This difference, and also considering that the HC indicators presented the greatest problems in the models, with some of them being withdrawn, can mean that neither students nor teachers/researchers may be sufficiently well informed about their HEI's human resource system, and there may be inefficient management of these resources if it does not reveal their importance and staff's competences for the institution's good functioning.

As for the dimensions of SD, these results are in line with the state of the art, e.g., [10,22], which shows that the SD concept is associated equally with HEIs' practices of economic, environmental, social and organisational sustainability. Strangely, the social dimension, for both students and teachers/researchers, is the least robust one (M1: 0.796 and M2: 0.768), although positive and quite significant. Recovering [10], which found that Portuguese HEIs are mainly engaged in the social dimension of SD practices, contradicts the result obtained here somewhat, inasmuch as this study was made considering the perception of students and teachers/researchers. Perhaps HEIs are not sending out the right image in relation to the social dimension and/or respondents are giving greater importance to the other dimensions. This fact may be associated with the SD practices still in phases closely linked to planning, as previously advocated in [45].

Regarding hypothesis 2, according to which SD practices in HEIs have a direct and positive influence on students' QAL, this was not rejected either (M1b: 0.245). There is evidence for the presence of SD practices in HEIs that interact in students' lives, inasmuch as they are perceived by the latter and are part of their concerns. As mentioned by [46], sustainable behaviours contribute to quality of life in more instances than expected and students' perceptions of the dimensions associated with SD practices are very important for them to feel secure in both the present and future. These results strengthen the idea that SD practices are related to QAL, through the satisfaction with the experiences that create positive emotions in the context of university life experienced by students [62–68]. These satisfaction can be observed, as mentioned by [50], through several improvements in the campus environment (noise reduction, less use of paper, recycling campaigns, etc.), and in this way HEIs can contribute to greater satisfaction among students.

Concerning the result found for hypothesis 3, HEIs' SD practices have a direct and positive influence on the QWL of teachers/researchers. Despite this influence being positive and significant, for a 5% level reason the associated hypothesis is rejected. This result partially contradicts some authors, e.g., [46], who argued that SD affects QoL. Bearing in mind that QWL is a more specific construct, related to needs for satisfaction in the workplace, and not in a general way, this analogy must be done carefully because more evidence found in the HEI work context is needed.

The difference found between hypothesis 2 and hypothesis 3 sheds new light on the interesting fact that young people are more aware of, and perhaps more concerned about, matters related

to sustainability, and consequently about the future, compared to what is found in teachers' and researchers' perceptions. Therefore, the formers' perception of what is done in relation to SD in their HEI is revealed to be greater.

All in all, the results now obtained can be applied to the practice by HEIs managers through more visible sustainable efforts, building bridges within HEIs between IC, SD and QoL that will lead stakeholders to recognise the institution's sustainability efforts. For this they need, as mentioned in [23], to generate new ideas, to engage the HEIs' human resources in sustainability, promoting a better QoL, to create a baseline for continuous improvement, to inform strategic planning and budgeting, to integrate sustainability into the curriculum, to make real progress towards sustainability, and to be part of a global community involved in sustainability purposes.

6. Concluding Remarks and Future Research

This paper focuses on the influence of HEIs' IC (HC, SC, RC) on HEIs' SD, and on the influence of HEIs' SD on stakeholders' QoL (QAL of students and QWL of teachers/researchers), formulating three hypotheses for test purposes. To respond to the proposed objectives, after determining the state-of-the-art, a quantitative analysis was performed by collecting primary data and using a structural equation model and the PLS method. SEM–PLS supported two of the three previously formulated hypotheses.

The results obtained are important contributions to the literature on IC through the ratification of new evidences for theory, as they confirm empirically that, firstly, a positive and significant relationship exists between HEIs' IC and HEIs' SD, and secondly, a positive and significant relationship exists between HEIs' SD and students' QAL. Regarding the influence of HEIs' SD on teachers/researchers' QWL, no empirical evidence was found of a robust relationship between these two constructs, suggesting there may be other variables that are not being considered and which could possibly change this result, and so new, more thorough research in this field is suggested. Therefore, this type of relationship, never before studied, opens new theoretical horizons and new perspectives for further study and research in this area. The results indicate that IC (HC, SC and RC) has a positive and significant influence on HEIs' (economic, environmental, social and organisational) SD, since hypothesis 1 was not rejected in either model. That is, through the perception that students and teachers/researchers have of the IC and SD of their HEI, it is concluded that IC is directly and positively related to that institution's SD. These results are consistent with previous evidence [7]. HEIs should approach their IC as a whole, since all dimensions are revealed to be important. However, attention is drawn to the fact that HC is the one where there may be more room for improvement, since it had least weight in IC. Considering the results obtained in this study, for IC to produce an even greater impact on SD, HEIs should create and implement strategies towards continuous improvement of their human resources, as by devoting more attention to their human resources they can have greater empowerment and thereby influence HEIs' SD even more. This conclusion ratifies [108], which stated that human capital is an indicator of value creation that can be used to help formulate organisational strategy, provide a basis for evaluation and allocate some resources in the HEI context.

The results of this study are also in line with the previous concluding remarks found in [10], who revealed that Portuguese HEIs are beginning to give relevance to all the dimensions of SD and include them in their strategic plans, communication strategies and various policies. Nevertheless, it stands out that the social dimension has the lowest value (M1: 0.796 and M2: 0.768), and there may be room for a better positioning of HEIs through better and more proficient social engagement, as mentioned in [10], oriented towards the increasingly urgent challenges of sustainability, associated with rapid change and increased complexity and social unrest.

As for the relationship between HEIs' sustainability and QoL, there is evidence of its existence, supported by finding a positive and significant relationship between HEIs' SD and QAL. It is, therefore, underlined that HEIs have a fundamental role in promoting SD and their leaders' efforts are vital in achieving the goals associated with SD. HEIs must recognise their importance and responsibility,

not only in terms of pro-sustainability education but also by including measures of SD that have repercussions for the QoL of their stakeholders and that of the region's population and the country. As demonstrated in several studies, notably in the Spanish context, e.g., [31], it is important to develop policy statements, in order to increase sustainability practices in Portuguese HEIS [10].

HEIs should pay attention to how they manage their IC, creating value not only for the institution itself through the contribution to SD but also creating value for the QoL of their students and teachers/researchers, developing these points that may possibly be more connected to the latter's QWL. In addition, the efforts of HEI leaders should focus on achieving SD goals, and the actions promoted by these institutions should be in line with the perceptions of all their stakeholders.

Referred to as an implication, given the importance of transforming the education institution as a whole, the priority action areas undergo transformations at the level of information, because more and better information should be given about what happens in SD practices in HEIs, providing information which is accessible to all. However, and from the results obtained, perception is seen to be different depending on the stakeholder, and so SD practices should be monitored on a regular basis and the reports should be provided in such a way that everyone understands their content, using simple and accessible language. As noted by UNESCO [13], education institutions are encouraged to implement sustainability strategies and plans with institution-wide approaches, taking into account some key elements such as inter-institutional networks that facilitate mutual support, such as peer-to-peer learning on a whole-institution scale, and increase the visibility of the approach to promote it as a role model for change and adaptation.

This study also provides practical implications for stakeholders: (i) HEIs must satisfy students' needs and emotions, fostering QAL through a better engagement in sustainability activities, by integrating sustainability into the academic curriculum, and by giving more information at a higher quality about what is happening within the HEI concerning SD; and (ii) HEIs must develop some support infrastructures that allow managers to track which sustainability satisfaction needs (QWL) teachers/researchers may have, so that institutions can develop strategies leading to SD while enhancing human resources' satisfaction needs within the employer institution. For example, those needs may be related to social responsibility, and so can be addressed through the greater dissemination of the activities that the institution develops and/or intends to develop, and through specific educational training, that can contribute to both personal enrichment and a greater competence in knowledge transfer to their peers and/or students.

Regarding the limitations of this study, firstly the fact that various indicators from the initial model were eliminated, especially concerning HC, and this elimination may have limited our results. However, despite this, the final model presented very significant and conclusive results, allowing for very useful conclusions to be drawn and the non-rejection of two research hypotheses.

Secondly, the sample was confined to Portuguese HEIs and, therefore, these results cannot be generalized to HEIs in other countries.

Another aspect associated to the representativeness is the fact that the sample is related only to public university education. As Portugal has public and private HEIs and a binary system, as mentioned before, it would be worthwhile to have selected private HEIs and the polytechnic institutions. Therefore, representativeness is limited, and the results of the study cannot be generalized to the entire Portuguese higher education system. Nevertheless, based on the Portuguese public university education system, the sample was representative of the reality under study, since each institution was located in a different region at the NUTS II level.

Thirdly, HEI stakeholders were represented by only students and teachers/researchers. However, in studies made in other HEIs, the top management and/or leadership are almost always the ones surveyed.

As mentioned, the difference in the results found for QAL and QWL is a serious, sustained warning based on new empirical evidence, that young people are more aware of issues related to sustainability than teachers and researchers, since the former denote an high level of perception concerning sustainability issues, due to previous engagement in education programmes, which raised

their social consciousness on the need for change and addressing sustainability issues associated with climate change, social inequality and common well-being, which they tend to value as change mechanisms that can contribute to improving their quality of life, including the academic context and society as a whole. Here, the age factor may have some relevance, in that young people have been found to show greater concern about sustainability and the future of the planet. Nevertheless, the role attributable to pro-sustainability education from an early age can no longer be ignored, including the economic, social, environmental and organisational dimensions, as this can make all the difference in the inter-related cycles of learning and performance throughout life.

With this final motivation, a window of opportunity opens to make future comparative studies based on the age factor and pro-sustainability education factor, since we believe that both can be determinant for the development of successful SD practices, in the HEI context in particular, and society in general. Future research avenues can be explored by developing studies focusing especially on HC, aiming to test disaggregated measures and indicators of this critical asset. Adding to this, cross-country comparisons are suggested in light of the whole-institution approach, in order to assess the role played by "organisational inertia", in terms of potential resistance to change involving the adoption of a whole-institution sustainability vision and the implementation of SD practices at the institutional level. It would be also of interest to deepen the scarce knowledge on IC in HEIs by contrasting the perceptions of the governance board and the students concerning the different activities of this type of knowledge institution, which play a significant role in educating proactive citizens regarding sustainable development and quality of life, with a clear vision of social impact.

Supplementary Materials: The following are available online at http://www.mdpi.com/2071-1050/12/2/479/s1, Table S1. List of constructs and respective indicators referring to models 1 and 2.

Author Contributions: E.d.M.P., J.L. and H.A. contributed to the design and implementation of the research, to the analysis of the results and to the writing of the manuscript. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by the FCT—Fundação para a Ciência e a Tecnologia, in the scope of the research activities developed at the PTDC/EGE-OGE/29926/2017.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Matos, F.; Vairinhos, V.; Maurício, P.; Leif, S. Introduction. In *Intellectual Capital Management as an Indicator* of Sustainability; Matos, F., Vairinhos, V., Maurício, P., Leif, S., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 1–10.
- Mas, F.D. The Relationship Between Intellectual Capital and Sustainability: An Analysis of Practitioner's Thought. In *Intellectual Capital Management as an Indicator of Sustainability*; Matos, F., Vairinhos, V., Maurício, P., Leif, S., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 11–24.
- Malhotra, Y. Measuring Knowledge Assets of a Nation: Knowledge Systems for Development. In United Nations Advisory Meeting of the Department of Economic and Social Affairs Division for Public Administration and Development Management; United Nations: New York, NY, USA, 2003; p. 52.
- Cabrita, M.R.; Cabrita, C. The Role of Creative Industries in Stimulating Intellectual Capital in Cities and Regions. In 2nd European Conference on Intellectual Capital; ISCTE Lisbon University Institute and Polytechnic Institute of Leiria: Lisbon, Portugal, 2010; pp. 171–179.
- Januškaite, V.; Užiene, L. Intellectual Capital as a Factor of Sustainable Regional Competitiveness. Sustainability 2018, 10, 4848. [CrossRef]
- Dal Mas, F. The Relationship Between Intellectual Capital and Sustainability: An Analysis of Practitioner's Thought. In *Intellectual Capital Management as a Driver of Sustainability*; Matos, F., Vairinhos, V., Selig, P.M., Edvinsson, L., Eds.; Springer International Publishing: Cham, Switzerland, 2019.
- Silva, T.M.; Ferreira, A. Intellectual Capital Sustainability in Brazilian Public Higher Education. In *Intellectual Capital Management as an Indicator of Sustainability*; Matos, F., Vairinhos, V., Maurício, P., Leif, S., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 177–203. [CrossRef]

- Stephens, J.C.; Hernandez, M.E.; Román, M.; Graham, A.C.; Scholz, R.W. Higher Education as a Change Agent for Sustainability in Different Cultures and Contexts. *Int. J. Sustain. High. Educ.* 2008, 9, 317–338. [CrossRef]
- Pedro, E.M. Ensaios Sobre Capital Intelectual Das Instituições de Ensino Superior: Taxonomia, Mensuração e Impactos; University of Beira Interior: Covilhã, Portugal, 2018.
- Aleixo, A.M.; Azeiteiro, U.; Leal, S. The Implementation of Sustainability Practices in Portuguese Higher Education Institutions. *Int. J. Sustain. High. Educ.* 2018, 19, 146–178. [CrossRef]
- 11. Sammalisto, K.; Sundstrom, A.; Holm, T. Implementation of Sustainability in Universities as Perceived by Faculty and Staff e a Model from a Swedish University. J. Clean. Prod. 2015, 106, 45–54. [CrossRef]
- 12. Too, L.; Bajracharya, B. Sustainable Campus: Engaging the Community in Sustainability. *Int. J. Sustain. High. Educ.* **2015**, *16*, 57–71. [CrossRef]
- UNESCO. Education for Sustainable Development Goals: Learning Objectives; United Nations Educational, Scientific and Cultural Organization: Paris, France, 2017.
- 14. UNESCO. Shaping the Future We Want UN Decade of Education for Sustainable Development. 2005-2014 Final Report; UNESCO: Luxembourg, 2014. [CrossRef]
- Sartori, S.; Latrônico, F.; Campos, L. Sustainability and Sustainable Development: A Taxonomy in the Field of Literature. *Ambient. Soc.* 2014, 17, 1–22. [CrossRef]
- Olawumi, T.O.; Chan, D.W.M. A Scientometric Review of Global Research on Sustainability and Sustainable Development. J. Clean. Prod. 2018, 183, 231–250. [CrossRef]
- Silva Junior, A.; Martins-Silva, P.O.; Vasconcelos, K.C.A.; Silva, V.C.; Melo, M.R.; Dumer, M.C.R. Sustainability Indicators for the Management of Brazilian Higher Education Institutions. *BAR Brazil. Adm. Rev.* 2018, 15, 1–22. [CrossRef]
- Frondizi, R.; Fantauzzi, C.; Colasanti, N.; Fiorani, G. The Evaluation of Universities' Third Mission and Intellectual Capital: Theoretical Analysis and Application to Italy. *Sustainability* 2019, 11, 3455. [CrossRef]
- Pedro, E.; Leitão, J.; Alves, H. Back to the Future of Intellectual Capital Research: A Systematic Literature Review. Manag. Decis. 2018, 56, 2502–2583. [CrossRef]
- Ramírez-Córcoles, Y.; Peñalver, J.; Ponce, Á. Intellectual Capital in Spanish Public Universities: Stakeholders' Information Needs. J. Intellect. Cap. 2011, 12, 356–376. [CrossRef]
- Aleixo, A.M.; Azeiteiro, U.M.; Leal, S. Confronting the Challenges in Integrating Sustainable Development in a Curriculum: The Case of the Civil Engineering. In *Challenges in Higher Education for Sustainability, Management and Industrial Engineering;* Davim, J.P., Filho, W.L., Eds.; Springer International Publishing: Cham, Switzerland, 2016; pp. 159–187. [CrossRef]
- 22. Findler, F.; Schönherr, N.; Lozano, R.; Stacherl, B. Assessing the Impacts of Higher Education Institutions on Sustainable Development-an Analysis of Tools and Indicators. *Sustainability* **2018**, *11*, 59. [CrossRef]
- 23. AASHE. The Association for the Advancement of Sustainability in Higher Education. Available online: https://www.aashe.org/ (accessed on 4 November 2019).
- 24. Lozano, R. The State of Sustainability Reporting in Universities. *Int. J. Sustain. High. Educ.* 2011, 12, 67–78. [CrossRef]
- Lozano, R.; Lozano, F.J.; Mulder, K.; Huisingh, D.; Waas, T. Advancing Higher Education for Sustainable Development: International Insights and Critical Reflections. J. Clean. Prod. 2013, 48, 3–9. [CrossRef]
- De Amorim, W.A.C.; Fischer, R.M.; Comini, G.M.; Rodrigues, J. Capital Humano e Sustentabilidade: Uma Proposta de Abordagem Multidisciplinar. *Gestão Reg.* 2015, 31, 151–163. [CrossRef]
- Niedlich, S.; Rieckmann, M.; Kummer, B.; Bormann, I. Cultures of Sustainability Governance in Higher Education Institutions: A Multi-Case Study of Dimensions and Implications. *High. Educ. Q.* 2019, 1–18. [CrossRef]
- Amaral, L.P.; Martins, N.; Gouveia, J.B. Quest for a Sustainable University: A Review. Int. J. Sustain. High. Educ. 2015, 16, 155–172. [CrossRef]
- Disterheft, A.; Caeiro, S.; Azeiteiro, U.M.; Filho, W.L. Sustainability Science and Education for Sustainable Development in Universities: A Way for Transition. In *Sustainability Assessment Tools in Higher Education Institutions: Mapping Trends and Good Practices Around the World*; Caeiro, S., Filho, W.L., Jabbour, C., Azeiteiro, U.M., Eds.; Springer International Publishing: Cardiff, UK, 2013; pp. 3–28.

- Leal Filho, W.; Manolas, E.; Pace, P. The Future We Want: Key Issues on Sustainable Development in Higher Education after Rio and the UN Decade of Education for Sustainable Development. *Int. J. Sustain. High. Educ.* 2015, 16, 112–129. [CrossRef]
- Larrán, J.M.; Madueño, J.H.; Calzado, Y.; Andrades, J. A Proposal for Measuring Sustainability in Universities: A Case Study of Spain. Int. J. Sustain. High. Educ. 2016, 17, 671–697. [CrossRef]
- Vagnoni, E.; Cavicchi, C. An Exploratory Study of Sustainable Development at Italian Universities. Int. J. Sustain. High. Educ. 2015, 16, 217–236. [CrossRef]
- Lourenço, R.T.; Mano, M. Os Stakeholders e as Instituições de Ensino Superior. In Proceedings of the 4a Conferencia FORGES, Luanda e Lubango, Angola, 2014; pp. 1–16.
- Cvetic, T.; Arsovski, S.; Stefanovic, M.; Djordjevic, A. Impact of Standardized Management Systems on Quality of Workinng Life. In 1st International Conference on Quality of Life; University of Kragujevac: Kragujevac, Serbia, 2016; pp. 215–220.
- Tapia-Fonllem, C.; Fraijo-Sing, B.; Corral-Verdugo, V.; Valdez, A.O. Education for Sustainable Development in Higher Education Institutions: Its Influence on the pro-Sustainability Orientation of Mexican Students. SAGE Open 2017, 7, 1–15. [CrossRef]
- 36. Veltri, S.; Mastroleo, G.; Schaffhauser-Linzatti, M. Measuring Intellectual Capital in the University Sector Using a Fuzzy Logic Expert System. *Knowl. Manag. Res. Pract.* **2014**, *12*, 175–192. [CrossRef]
- Secundo, G.; Elena-Perez, S.; Martinaitis, Ž.; Leitner, K.-H. An Intellectual Capital Maturity Model (ICMM) to Improve Strategic Management in European Universities. J. Intellect. Cap. 2015, 16, 390–418. [CrossRef]
- 38. Thiry-Cherques, H.R. Pierre Bourdieu: The Theory in Practice. Rev. Adm. Pública 2006, 40, 27–53. [CrossRef]
- Ramírez-Córcoles, Y.; Gordillo, S.B. Recognition and Measurement of Intellectual Capital in Spanish Universities. J. Intellect. Cap. 2014, 15, 173–188. [CrossRef]
- Veltri, S.; Silvestri, A. The Free State University Integrated Reporting: A Critical Consideration. J. Intellect. Cap. 2015, 16, 443–462. [CrossRef]
- Puukka, J. Higher Education for Sustainable Development Mobilising Higher Education for Sustainable Development—Lessons Learnt from the OECD Study. Proc. Int. Barcelona Conf. High. Educ. High. Educ. Sustain. Dev. GUNI Glob. Univ. Netw. Innov. 2008, 7, 1–26.
- Ramírez-Córcoles, Y.; Tejada, Á.; Baidez, A.; Montero, F. Managing Intellectual Capital: Conceptual Background and Application for European Higher Education Institutions. In Proceedings of the 6th International Conference on Knowledge Management in Asia Pacific, Shanghai, China, 10 October 2012; pp. 1–8.
- Tonial, G.; Cassol, A.; Selig, P.M.; Giugliani, E. Intellectual Capital Management and Sustainability Activities in Brazilian Organizations: A Case Study. In *Intellectual Capital Management as a Driver of Sustainability*; Matos, F., Vairinhos, V., Selig, P.M., Edvinsson, L., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 119–138.
- Duque, E. Capital Social Como Instrumento de Desenvolvimento Sustentáve. Configurações 2013, 11, 189–201. [CrossRef]
- Aleixo, A.M.; Leal, S.; Azeiteiro, U.M. Conceptualization of Sustainable Higher Education Institutions, Roles, Barriers, and Challenges for Sustainability: An Exploratory Study in Portugal. J. Clean. Prod. 2018, 172, 1664–1667. [CrossRef]
- Tapia-fonllem, C.; Corral-verdugo, V.; Fraijo-sing, B. Sustainable Behavior and Quality of Life. In Handbook of Environmental Psychology and Quality of Life Research; Fleury-Bahi, G., Pol, E., Navarro, O., Eds.; Springer: Cham, Switzerland, 2017; pp. 173–184. [CrossRef]
- Fergus, A.H.T.; Rowney, J.I.A. Sustainable Development: Lost Meaning and Opportunity? J. Bussines Ethics 2005, 60, 17–25. [CrossRef]
- Keiner, M. Re-Emphasizing Sustainable Development-the Concept of 'Evolutionability'. *Environ. Dev. Sustain.* 2004, 6, 379–392. [CrossRef]
- 49. Diener, E.; Suh, E. Measuring Quality of Life: Economic, Social, and Subjective Indicators. *Soc. Indic. Res.* **1997**, 40, 189–216. [CrossRef]
- Moser, G. Quality of Life and Sustainability: Toward Person-Environment Congruity. J. Environ. Psychol. 2009, 29, 351–357. [CrossRef]
- 51. Rodrigues, R.R.J. Quality of Work Life and Sustainable Environmental Actions. *Rev. Psicol. Saúde* 2015, *7*, 177–185.

- 52. Beck, W.; Van der Masesn, L.; Walker, A. The Social Quality of Europe; Policy Press: Bristol, UK, 1998.
- Júnior, A.F.; Ferreira, M.B.R. Life Quality and World Sustainable Development: Perspectives to Survival. *Rev. Bras. Ciências Saúde* 2006, 7, 82–91.
- 54. Machado, N.S.; Weber, J.; Silveira, A.; Petarnella, L. Higher Education and Sustainability: Understanding of the Managers of a Higher Education Institution. *Rev. Ciências Adm.* **2018**, *20*, 42–54. [CrossRef]
- 55. Madeira, A.C.F.D. *Indicadores de Sustentabilidade Para Instituições de Ensino Superior;* University of Porto: Porto, Portugal, 2008.
- Silva, A.A.M.; Claro, J.A.S.; Costa, L.V.; Tinoco, J.E.P. Gestão Ambiental e Universidade: O Estudo de Caso Do Programa Metodista Sustentável. *Desenvolv. Questão.* 2013, 13, 146–177.
- Di Fabio, A. Constructing and Managing Personal Project, Career Project, Life Project: The Challenge of Sustainability. In *Invited Lecture at the Seminar Organized by the Faculty of Health Sciences*; Hokkaido University: Sapporo, Japan, 2016.
- Ariza-Montes, A.; Giorgi, G.; Hernández-Perlines, F.; Fiz-Perez, J. Decent Work as a Necessary Condition for Sustainable Well-Being. A Tale of Pi(i)Gs and Farmers. *Sustainability* 2019, 11, 1051. [CrossRef]
- 59. Di Fabio, A.; Kenny, M.E. From Decent Work to Decent Lives: Positive Self and Relational Management (PS&RM) in the Twenty-First Century. *Front. Psychol.* **2016**, *7*, 1–12. [CrossRef]
- Turkoglu, H. Sustainable Development and Quality of Urban Life. *Procedia Soc. Behav. Sci.* 2015, 202, 10–14. [CrossRef]
- Belbute, J.; Marques, J.; Gomes, C.; Carvalho, J.; Castro, E.; Pinto, M.; Gomes, P. Qualidade de Vida—Uma Análise à Escala Local, 3rd ed.; Universidade de Évora: Évora, Portugal, 2009; Available online: https://dspace.uevora.pt/rdpc/bitstream/10174/3168/1/Qual%200%20valor%20da%20qualidade% 20de%20vida%20-rdspace%20-%20JBelbute.pdf (accessed on 31 December 2019).
- Sirgy, M.J.; Grzeskowiak, S.; Rahtz, D. Quality of College Life (QCL) of Students: Developing and Validating a Measure of Well-Being. Soc. Indic. Res. 2007, 80, 343–360. [CrossRef]
- Sirgy, M.J.; Lee, D.-J.; Grzeskowiak, S.; Yu, G.B.; Webb, D.; El-Hasan, K.; Jesus Garcia Vega, J.; Ekici, A.; Johar, J.S.; Krishen, A.; et al. Quality of College Life (QCL) of Students: Further Validation of a Measure of Well-Being. Soc. Indic. Res. 2010, 99, 375–390. [CrossRef]
- 64. Yu, G.; Lee, D.-J. A Model of Quality of College Life (QCL) of Students in Korea. *Soc. Indic. Res.* 2008, *87*, 269–285. [CrossRef]
- 65. Yu, G.B.; Kim, J.-H. Testing the Mediating Effect of the Quality of College Life in the Student Satisfaction and Student Loyalty Relationship. *Appl. Res. Qual. Life* **2008**, *3*, 1–21. [CrossRef]
- Arslan, S.; Akkas, O.A. Quality of College Life (QCL) of Students in Turkey: Students' Life Satisfaction and Identification. Soc. Indic. Res. 2014, 115, 869–884. [CrossRef]
- Pedro, E.; Alves, H.; Leitão, J. Evaluating Students' Quality of Academic Life: Using Data in a Structural Model Approach. In Using Data to Improve Higher Education Research, Policy and Practice; Menon, M.E., Terkla, D.G., Gibbs, P., Eds.; Sense Publishers: Rotterdam, The Netherlands, 2014; pp. 93–118. [CrossRef]
- Pedro, E.; Leitão, J.; Alves, H. Does the Quality of Academic Life Matter for Students' Performance, Loyalty and University Recommendation? *Appl. Res. Qual. Life* 2016, 11, 293–316. [CrossRef]
- 69. Sirgy, M.J.; Efraty, D.; Siegel, P.; Lee, D.A. New Measure of Quality of Work Life (Qowl) Based on Need Satisfaction and Spillover Theories'. *Soc. Indic. Res.* **2001**, *55*, 241–302. [CrossRef]
- Mirkamali, S.M.; Thani, F.N. A Study on the Quality of Work Life (QWL) among Faculty Members of University of Tehran (UT) and Sharif University of Technology (SUT). *Procedia Soc. Behav. Sci.* 2011, 29, 179–187. [CrossRef]
- Parsa, B.; Idris, K.B.; Samah, B.B.A.; Wahat, N.W.B.A.; Parsa, P. Relationship between Quality of Work Life and Career Advancement among Iranian Academics. *Proceedia Soc. Behav. Sci.* 2014, 152, 108–111. [CrossRef]
- Castilha, F.A.; Couto, J.G.; Fernandes Filho, J. University Professors Quality of Life in The West of Parana State, Brazil. FIEP Bull. Online 2015, 85, 981–990. [CrossRef]
- Kermansaravi, F.; Navidian, A.; Navabi Rigi, S.; Yaghoubinia, F. The Relationship Between Quality of Work Life and Job Satisfaction of Faculty Members in Zahedan University of Medical Sciences. *Glob. J. Health Sci.* 2015, 7, 228–234. [CrossRef] [PubMed]
- 74. Intellectual Capital Management as a Driver of Sustainability; Matos, F.; Vairinhos, V.; Selig, P.M.; Edvinsson, L. (Eds.) Springer International Publishing: Cham, Switzerland, 2019. [CrossRef]
- 75. Pedro, E.M.; Leitão, J.; Alves, H. Quality of Academic Life and Regional Growth. In Proceedings of the 24th Conference APDR Congress 'Intellectual Capital and Regional Development: New Landscapes and Challenges for Space Planning, Covilhã, Portugal, 6–7 July 2017; pp. 906–914, ISBN 978-989-8780-05-8.
- Rizk, C. Citadins, Votre Quartier Est-Il Agréable à Vivre? INSEE-première no. 934; National Institute of Statistics and Economic Studies: Paris, France, 2003.
- Henning, M.A.; Krägeloh, C.U.; Hawken, S.J.; Doherty, I.; Zhao, Y.; Shulruf, B. Motivation to Learn, Quality of Life and Estimated Academic Achievement: Medical Students Studying in New Zealand. *Med. Sci. Educ.* 2011, 21, 142–150. [CrossRef]
- Pinto, G.M.C.; Moraes, J.; Pedroso, B.; Pilatti, L.A.; Picinin, C.T. Quality of Life of Docents of Graduate Production Engineering: Case Study of Professors of Brazilian South. FIEP Bull. 2017, 87, 43–47. [CrossRef]
- De Pedro, E.M.; Alves, H.; Leitão, J. Does the Quality of Academic Life Mediate the Satisfaction, Loyalty and Recommendation of HEI Students? *Int. J. Educ. Manag.* 2018, 32, 881–900. [CrossRef]
- Lee, D.-J.; Singhapakdi, A.; Sirgy, M.J. Further Validation of a Need-Based Quality-of-Work-Life (QWL) Measure: Evidence from Marketing Practitioners. *Appl. Res. Qual. Life* 2007, *2*, 273–287. [CrossRef]
- El Hassan, K. Quality of College Life (QCL): Validation of a Measure of Student Well-Being in the Middle East. Int. J. Educ. Psychol. Assess. 2011, 8, 12–22. [CrossRef]
- 82. Cascio, W.F. Managing Human Resources, 5th ed.; Boston Irwin McGraw-Hill: Boston, MA, USA, 1998.
- Walton, R.E. Criteria for Quality of Working Life. In *The Quality of Working Life*; Davis, L.E., Cherns, A.B., Eds.; The Free Press: New York, NY, USA, 1975; pp. 91–104.
- Cole, D.C.; Robson, L.S.; Lemieux-Charles, L.; McGuire, W.; Sicotte, C.; Champagne, F. Quality of Working Life Indicators in Canadian Health Care Organizations: A Tool for Healthy, Health Care Workplaces? Occup. Med. Chic. Ill. 2005, 55, 54–59. [CrossRef]
- Sirgy, M.; Cornwell, T. Further Validation of the Sirgy et Al.'s Measure of Community Quality of Life. Soc. Indic. Res. 2001, 56, 125–143. [CrossRef]
- Lee, J.J.; Rice, C. Welcome to America? International Student Perceptions of Discrimination. *High. Educ.* 2007, 53, 381–409. [CrossRef]
- Holme, R.; Watts, P. Corporate Social Responsibility: Making Good Business Sense (World Business Council for Sustainable Development, Geneve); World Business Council for Sustainable Development: Geneva, Switzerland, 2000.
- Drumwright, M.E. Socially Responsible Organizational Buying: Environmental Concern as a Noneconomic Buying Criterion. J. Mark. 1994, 58, 1–19. [CrossRef]
- Sink, D.S. Productivity Management: Planning, Evaluation, Control. and Improvement; John Wiley and Sons, Inc.: New York, NY, USA, 1985.
- 90. Fortin, M.F. O Processo de Investigação: Da Conceção à Realização; Lusociência Ed.: Loures, Portugal, 1999.
- Mainardes, E.; Alves, H.; Raposo, M. Identifying Stakeholders in a Portuguese University: A Case Study La Identificación de Los Stakeholders En Una Universidad Portuguesa. *Rev. Educ.* 2013, 362, 10–122.
- 92. Pedro, E.; Leitão, J.; Alves, H. The Intellectual Capital of Higher Education Institutions Operationalizing Measurement through a Strategic Prospective Lens. J. Intellect. Cap. 2019. [CrossRef]
- 93. Diener, E.; Smith, H.; Fujita, F. The Personality Structure of Affect. J. Pers. Soc. Psychol. 1995, 69, 130–141. [CrossRef]
- Smith, S.M. Determining Sample Size: How to Ensure you Get the Correct Sample Size. Available online: https://www.qualtrics.com (accessed on 19 August 2017).
- 95. Hair, J.F.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M. A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), 2nd ed.; Sage Publications, Inc.: Los Angeles, CA, USA, 2017.
- 96. Ringle, C.M.; Wende, S.; Becker, J.-M. SmartPLS 3; SmartPLS: Bönningstedt, Germany, 2015.
- 97. Henseler, J.; Hubona, G.; Ray, P.A. Using PLS Path Modeling in New Technology Research: Updated Guidelines. *Ind. Manag. Data Syst.* 2016, *116*, 2–20. [CrossRef]
- Wright, R.; Campbell, D.; Thatcher, J.; Roberts, N. Operationalizing Multidimensional Constructs in Structural Equation Modeling: Recommendations for IS Research. *Commun. Assoc. Inf. Syst. Vol.* 2012, 30, 367–412. [CrossRef]
- 99. Hu, L.-T.; Bentler, P.M. Fit Indices in Covariance Structure Modeling: Sensitivity to Underparameterized Model Misspecification. *Psychol. Methods* **1998**, *3*, 424–453. [CrossRef]

- Hair, J.F.; Risher, J.J.; Sarstedt, M.; Ringle, C.M. When to Use and How to Report the Results of PLS-SEM. Eur. Bus. Rev. 2019, 31, 2–24. [CrossRef]
- 101. Hair, J.F.; Sarstedt, M.; Ringle, C.M. Rethinking Some of the Rethinking of Partial Least Squares. *Eur. J. Mark.* 2019. [CrossRef]
- Sarkar, M.; Echambadi, R.; Cavusgil, S.T.; Aulakh, P.S. The Influence of Complementarity, Compatibility, and Relationship Capital on Alliance Performance. J. Acad. Mark. Sci. 2001, 29, 358–373. [CrossRef]
- Hair, J.F.; Ringle, C.M.; Sarstedt, M. PLS-SEM: Indeed a Silver Bullet. J. Mark. Theory Pract. 2011, 19, 139–152. [CrossRef]
- 104. Cohen, J. Statistical Power Analysis for the Behavioral Sciences, 2nd ed.; Erlbaum: Hillsdale, NJ, USA, 1988.
- 105. Barroso Castro, C.; Cepeda Carrión, G.; Roldan Salgueiro, J.L. Investigar En Economía de La Empresa ¿partial Least Squares o Modelos Basados En La Covarianza? In XIX Congreso Anual y XV Congreso Hispano Francés de AEDEM; Universidad de Sevilla: Sevilla, Spain, 2005; Volume 1, pp. 625–634.
- Chin, W.W. The Partial Least Squares Approach to Structural Equation Modeling, in Marcoulides. Modern Methods Business Res. 1998, 295, 295–336.
- Martínez-Torres, M.R. A Procedure to Design a Structural and Measurement Model of Intellectual Capital: An Exploratory Study. Inf. Manag. 2006, 43, 617–626. [CrossRef]
- Liu, C.C. Developing Measures of Value Creation at Private Universities. Int. J. Manag. Educ. 2007, 1, 86–99. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Article

'A Nut We Have Officially yet to Crack': Forcing the Attention of Athletic Departments Toward Sustainability Through Shared Governance

Martin Barrett ^{1,*}, Kyle S. Bunds ², Jonathan M. Casper ², Michael B. Edwards ², D. Scott Showalter ³ and Gareth J. Jones ⁴

- ¹ Department of Kinesiology and Recreation, Frostburg State University, Frostburg, MD 21532, USA
- ² Department of Parks, Recreation and Tourism Management, North Carolina State University, Raleigh, NC 27695, USA; ksbunds@ncsu.edu (K.S.B.); jmcasper@ncsu.edu (J.M.C.); mbedward@ncsu.edu (M.B.E.)
- ³ Poole College of Management, North Carolina State University, Raleigh, NC 27695, USA; dsshowal@ncsu.edu
- ⁴ School of Sport, Tourism and Hospitality Management, Temple University, Philadelphia, PA 19122, USA; gareth.jones@temple.edu
- * Correspondence: mbarrett@frostburg.edu

Received: 19 June 2019; Accepted: 16 September 2019; Published: 23 September 2019

Abstract: In many ways, intercollegiate athletics represents the 'sustainable' front porch of higher education. The high-visibility, high-impact nature of elite-level college athletics make athletic departments a central player in the sustainable development journey. However, not all athletic departments respond to this responsibility, nor are all responses uniformly successful. According to national reporting frameworks, an increasing number of universities in the United States are choosing to involve their athletic departments in university-level sustainability governance structures, but the benefits and limitations of this remain unclear. Using the theory of loosely coupled systems, and more specifically, the voice of compensations (which views loose coupling as an unsatisfactory state), the purpose of this paper is to explore perceptions of athletic department engagement in shared sustainability governance, and, thus, a whole-of-institution approach. Semi-structured interviews with sustainability office personnel were conducted and analyzed, and the findings imply that shared sustainability governance has the potential to focus the attention of athletic departments toward sustainability, as well as to reaffirm shared values. Yet, to maximize the impact of athletic departments toward the sustainable development goals of a university, sustainability office personnel suggest the deployment of additional change levers, in a multi-dimensional fashion, as supplementary coupling mechanisms. These would include more rigorous sustainability goals (top-down), continued collaboration on 'low-hanging fruit' initiatives (lateral), student-athlete engagement (bottom-up), and the development of an internal sustainability framework (inside-out).

Keywords: athletic departments; higher education; sustainability; loose coupling; shared governance; United States

1. Introduction

Executing strategy in universities is a complex matter, as the institutions themselves are likened to "multi-headed monsters, each with unique recipes for success" [1] (p. 28). Moreover, basic units within universities have fairly independent decision-making power over their activities [2], and freedom to self-determine policies and priorities [3]. In essence, complex organizations like universities display varying levels of coupling across a number of domains where interdependent elements vary in the number and strength of their interdependencies [4]. In fact, formal structures in educational

MDP

organizations are understood as being decoupled from technical activities and outcomes as a means of maintaining support in a pluralistic environment [5]. Accordingly, some decisions in universities requiring specialized expertise (e.g., teaching and research) can only be made by individual professors, while others can come from central administrators as well as from collective and interactive processes [6]. Unsurprisingly, an extensive body of literature in higher education has focused on the contentious relationships between university administrators and faculty [7–10].

As an alternative and more contemporary example of coupling in United States higher education, a number of scholars note the widening financial and strategic chasm between universities and intercollegiate athletic departments [11–14]. For example, on one hand, universities who compete at the highest level of intercollegiate athletic competition regard their athletic departments as "auxiliary activities', responsible for generating most of the revenue to cover their costs through ticket sales, licensing, and broadcasting contracts" [15] (p. 87). On the other hand, five out of every six of the top athletic departments successfully secure scarce general funding to intercollegiate sports through student fees [16]. On one hand, the very nature of intercollegiate athletic means athletic departments embody values and work towards goals that are different from academic units [15]. On the other hand, administrators leverage athletic programs to serve as the highly visible 'front porch' of universities [17]. Such antagonistic examples depict what Orton and Weick [4] define as a loosely coupled, rather than decoupled, context. As such, athletic departments, for all of their perceived autonomy, simultaneously demonstrate responsiveness and distinctiveness with the university writ large across multiple domains.

In the sustainability domain, previous empirical work has supported the willingness of athletic departments to prioritize [18,19] and adopt greener practices [20]. This greening movement within intercollegiate athletics follows an increasing number of universities worldwide that are recognizing the importance of sustainable development [1,21] and aligning themselves with the principles of sustainability [22]. As such, sustainability in intercollegiate athletics demonstrates a level of responsiveness with the university more generally. Conversely, athletic departments are also utilizing their unique resources, such as high profile status and marketing platforms [18], to develop a more distinctive approach to sustainability efforts in comparison with other university departments and units. As a result, the simultaneous responsiveness and distinctiveness of athletic department sustainability action implies that the university–athletics relationship is, again, loosely coupled.

Loose coupling is often perceived as an unsatisfactory state that must be reversed, and this perspective is referred to as the voice of compensations [4]. In terms of corporate social responsibility more generally, Asif, Searcy, Zutshi, and Fisscher [23] suggest that managing stakeholder interests in isolation places pressure on organizational resources (e.g., personnel, time, and money), and leads to an ineffective use of such resources. Therefore, sustainability delivery is most effective within functionally integrative organizational structures [24], or tightly coupled systems. To compensate more generally for a lack of integration, calls for greater accountability in higher education have facilitated a move toward tight coupling and narrow control [25], which has been accompanied by frequent shifts in governance arrangements [26]. In the sustainability domain, it is at the organizational level where the charge is to create and establish campus-wide policies, objectives, and targets as well as prove the main decision-making function [27], and, thus, compensate for the lack of integration across the institution as a whole. To this end, Adams [28] puts forward a framework for managing sustainability in higher education, and these governance arrangements include a number of essential elements, including an empowered senior person responsible for sustainability, and a collaborative approach across senior leaders.

Athletic departments look to compensate for a lack of a coordinated and organized approach to sustainability through the creation of cross-functional 'green teams' [18,19,29,30]. Green teams are voluntary and formal collaborative arrangements between relevant internal and external stakeholders that can help advance an organization's sustainability initiatives [30]. Moreover, a green team is a planning and policy making team whose impact extends into organization operations and the external world of its stakeholders [29]. However, previous research suggests the capacity in which athletic

departments serve on green teams as well as the exact mission of these teams remains unclear [18]. To date, there is a scarcity of research focusing on athletic department involvement, specifically in university-level sustainability planning and coordination. As a result, an opportunity exists to understand the perceived benefits of athletic department contributions to shared governance models.

The purpose of this study is to describe athletic department involvement in shared sustainability governance. Furthermore, the purpose of this study is supported by two specific research questions:

Research Question 1. How are universities forcing the attention of athletic departments toward sustainability through engagement in shared sustainability governance?

Research Question 2. *How is athletic department engagement in shared sustainability governance compensating for loose coupling with the university overall?*

Accordingly, this study examines sustainability governance at universities across the United States through semi-structured interviews with sustainability office personnel. The result is a better understanding of how governance mechanisms compensate for the loose coupling between athletic departments and the wider university system. The remainder of this paper is organized as follows. First, the context and background of strategic sustainability management in higher education and intercollegiate athletics is summarized. Then, the theoretical foundations are built through the perspective of loosely coupled systems and the role of shared governance as a coupling mechanism. The methods are then explained before a combined findings and discussion section is presented. Finally, the study concludes with practical implications and suggestions for future research.

2. Research Context

2.1. Sustainability Management in Higher Education

Universities are integrating sustainability as a core strategic principle across campuses [31] through the development and implementation of comprehensive plans for setting sustainability goals [32]. Arguably, this strategic commitment began in 2006, when 12 college and university presidents initiated the American College and University Presidents' Climate Commitment (ACUPCC) [33]. The ACUPCC ignited a sustainability movement with 400 schools making the commitment in the initial period of charter membership between December 2006 and September 2007 [34]. The commitment itself made clear the role higher education has in leading climate and sustainability action for the sake of students and society [33].

Universities and colleges have since taken significant green strides in demonstrating sustainability in practice, thanks in part to the 2008 Higher Education Sustainability Act and the establishment of the University Sustainability Grants Program [35]. More recently, the Association for the Advancement of Sustainability in Higher Education (AASHE), through their 2018 annual conference titled 'Global Goals: Rising to the Challenge', began championing the critical role higher education must play in achieving the United Nations Sustainable Development Goals. Accordingly, the role of universities in advancing sustainability is twofold: as an institution that needs to be changed and/or as a potential change agent [36]. For example, universities are adopting a number of infrastructural and operational business practices to reduce their ecological footprint. Such efforts include waste management strategies like recycling and composting [37] and the reduction of greenhouse gas emissions [38]. Additionally, universities are operationalizing their potential as sustainability change agents by integrating sustainability into the curriculum, student and faculty life, and the wider community [39]. However, in higher education's twofold sustainability responsibility, a tendency exists for universities to focus more so on how higher education can change internally [36]. Dyer and Dyer [40] claim this tendency is reinforced by the ACUPCC, which creates a framework wherein institutions commit to future successes in the form of climate neutrality, then backcast to the present before creating plans to move toward the desired future state. Salviono, Franzoni, and Casnna [41] consider university governance with such a limited propensity towards sustainability as an inefficient condition to promote

change. In fact, while universities recognize their role in leading societal change toward a more sustainable world, many have not implemented whole-institution change [42].

To address limitations in university sustainability strategy, a number of frameworks exist to encourage best practice approaches to sustainability management in the higher education context. Adams [28] put forward a framework for managing sustainability in universities that includes a number of essential and desired elements. Among the essential elements are the visible support of a president/vice-chancellor and governing body, pro-active senior leadership, an empowered senior person responsible for sustainability, and a collaborative approach across senior leaders. Additional desirable elements in Adams' framework include an advisory board of external expertise, policies that incorporate sustainability, a stakeholder engagement strategy, and alignment with existing national quality frameworks. Epstein's [43] Corporate Sustainability Model considers drivers of sustainability performance (inputs), actions managers can take to affect performance (processes), and consequences of those actions (outputs). In terms of managerial actions, Epstein advocates the importance of committed leadership, strategy development, organizational design that embeds sustainability across all units, and the alignment of systems to coordinate activities and motivate employees. Such frameworks are warranted due to the proliferation of strategic approaches to sustainability at the university level, which is evidenced through emerging green campus indexes like The Princeton Review's 'Top 50 Green Colleges' and the Sierra Club's 'Cool Schools' ranking for the greenest colleges and universities.

2.2. Sustainability Management in Intercollegiate Athletics

In intercollegiate athletics, sustainability decision-making is highly differentiated between institutions. At the lower levels of intercollegiate athletic competition (i.e., Division III), sustainability decision-making is a function of an athletic director [19], whereas, at the elite level (Division I), decision-making occurs predominantly through associate athletic directors [18]. Yet, athletic department personnel can lack the environmental skills or training to make effective decisions [44], which necessitates a process of resource exchange in order for athletic departments to, at least, initiate sustainability projects. Often, and without a direct mandate from a higher administrative power, collaborative relationships develop through outreach from sustainability office personnel to the athletic department [44]. Such collaborative efforts are particularly effective in developing sustainability initiatives focusing on public external events (e.g., tailgate and facility/stadium recycling) [44]. Accordingly, not only are athletic departments somewhat reactive in their commitment to sustainability, but these collaborative efforts are predicated on interpersonal relationships as opposed to institutional commitments.

Where athletic departments are involved in structures beyond dyadic interpersonal ties (e.g., in green teams), the scope and remit of these groups show considerable variation. For example, of the two athletic departments involved in environmental committees in the Pfahl, Casper, Trendafilova, McCullough, and Nguyen [44] study, one is involved in a working group that "makes recommendations that are then filtered up to the athletics administration" (p. 40) and the other is a higher-level council comprising 20–25 campus units. Such variation is likely pervasive across the higher education sector. Casper, Pfahl, and McSherry [18] found 41% of Division I institutions had athletic department representation on campus-wide sustainability teams, while Casper and Pfahl [19] found 28% of Division III institutions with similar involvement. Nevertheless, McCullough et al. [30] suggest that green teams provide a bridge with university-wide goals through formal, joint decision-making processes.

3. Theoretical Background

3.1. Loosely Coupled Systems

Any location in an organization, be it a department, unit, office, or working group, contains interdependent elements that vary in the number and strength of their interdependencies [4]. Moreover, these interdependent elements represent shared variables that influence the degree of interaction, or

coupling, between specific locations within an organization or broader systems [45]. In other words, coupling refers to "the degree to which events within one part of a system are felt by other parts of that system" [46] (p. 245). Glassman [45] suggests that with coupling, two qualifying variables are necessary. First, coupling requires time to be isolated as a special variable, as any judgment on an organization or system is relevant only during a given interval of time. Second, provision must be made for the presence of interactions between variables of varying strengths. So, as opposed to being a dichotomizing concept, coupling has the flexibility to reflect complexity of organizations as living systems. For example, Orton and Weick [4] suggest that organizations can be tightly coupled (i.e., where they demonstrate responsiveness without distinctiveness), decoupled (i.e., where they demonstrate distinctiveness and responsiveness simultaneously).

Loose coupling allows organizations to "follow their idiosyncratic learning processes while retaining some degree of responsiveness" [47] (p. 1028). Importantly, Weick [48] presents educational organizations as loosely coupled exemplars and, in doing so, demonstrates the process by which soft structures guide the loose assemblages of diverse educational organizations (e.g., schools) to develop similar meanings across time. Specifically, Weick [48] depicts the relationship between a counselor's office and the principal's office, both of which retain some level of identity and separateness, but remain attached under the overarching umbrella of the school. Yet, Goldspink [49] suggests that educational reform within a loosely coupled context is incompatible with formal or bureaucratic control, and that different forms of management are necessary to realize their distinct advantages. As a result, higher education experiences frequent shifts in governance arrangements, including the reshuffling of authority and responsibilities [26]. In the context of United States higher education, calls for more accountability have contributed to a move toward tight coupling with narrow control [25].

The main advantage of loose coupling as a durable concept is summarized by Orton and Weick [4] in the following passage:

... loose coupling allows theorists to posit that any system, in an organizational location, can act on both a technical level, which is closed to outside forces (coupling produces stability), and an institutional level, which is open to outside forces (looseness produces flexibility). (p. 205)

However, Orton and Weick [10] suggest that the application of loose coupling is often confused. Specifically, organizational theorists succumb to not distinguishing between dialectical interpretation (i.e., based on the degrees of responsiveness and distinctiveness, organizations are either non-coupled, tightly coupled, or loosely coupled) and the unidimensional interpretation of loose coupling (i.e., loose coupling is the end point of a sliding scale between tightly coupled and loosely coupled).

3.2. Shared Governance as a Compensation for Loose Coupling

In response to the varied use of the concept, Orton and Weick [4] present a reconceptualization of loose coupling by identifying five distinct voices in the related literature. The voice of compensation suggests that loose coupling is an unsatisfactory condition that should be reversed [4]. Tierney and Minor [50] suggest attempts to tighten loose coupling within university settings centered around effective governance. Therefore, using a voice of compensation, governance arrangements present an opportunity to rectify loose coupling as an unsatisfactory condition through enhanced leadership, focused attention, and shared values [4]. Specifically, Orton and Weick [4] refer to the role of leadership (strong or subtle) in unifying goals, the role of targets in focusing attention, and the role of reaffirming shared values in rectifying losses in control.

Governance gives overall direction to an organization, and oversees and controls management actions through accountability and regulation [51]. As such, shared governance allows various groups of people a share in these key decision-making processes [52]. Shared governance also allows certain groups to exercise primary responsibility for specific areas of decision-making [52]. According to

Lechuga [53], shared governance is a unique characteristic of the United States higher education system. Cramer and Mozlin [54] suggest the modern roots of shared governance in higher education stem back to the 1966 Joint Statement on Government of Colleges and Universities. More specifically, the joint statement defines the main campus constituents as the governing board members, administrators, faculty members, students, and other persons. Therefore, shared governance in higher education refers to the structures and processes through which these constituents participate in the development of policies, and in decision-making that affects the institution [55]. Bejou and Bejou [56] offer a similar definition and suggest that shared governance must focus on informed decision-making, transparency, and open lines of communication between and among all components of the university community, while highlighting the important need for accountability, mutual respect, and trust within this participatory system.

Lapworth [57] claims models of shared governance fall under one of two categories: corporate models, with a focus on governing bodies, and consensual models, which focus on the roles of others. Alternatively, Yanko, Hardt, and Bradstock [58] have identified four different models of shared governance: unit-based, councilor, administrative, and congressional (see Figure 1). In unit-based shared governance models, each unit establishes its own system, which therein creates multiple models within a single institution. In councilor models of shared governance, a coordinating council directs activities on a more aggregate, departmental level, and the unit councils replicate the coordinating council structure. In congressional models of shared governance, all staff belong to a congress, which draws direct comparisons to the federal government. Furthermore, in congressional models of shared governance, separate practice and management structures exist, and a forum integrates the work of the councils overseeing these parallel structures.



Figure 1. Models of shared governance (adapted from Yanko et al. [58]).

Importantly, models of shared governance operate at different organizational levels, which influences extent of authority, decision-making, and participation [59]. Unit-based governance, intuitively, operates at the unit level. The unit level does, however, hold different meanings across varying organizational contexts. For example, for an institution such as a university or college, the unit can often denote the department, as opposed to an institution such as a hospital where the unit can be interpreted as a functional area or team. Yet, when governance is restricted to the unit-based level it is incompletely shared [60]. Alternatively, councilor and administrative models of shared governance operate at a more organizational level. Specifically within the context of higher education, it is at the organizational level where the charge is to create and establish campus-wide policies, objectives, and targets as well as to prove the main decision-making function [27]. The challenge with shared governance models is that organizations are complex and dynamic, which means that for these models to be effective they need to be diverse and flexible where individual institutions can adopt models to suit their character and needs [57].

Evidence exists to support the effectiveness of shared governance across various settings. Generally speaking, shared governance can reduce information asymmetry [61], promote a common understanding of goals, help prevent misunderstandings [62], and assist in coordinating goals and skills between partners [63]. Additionally, Bstieler, Hemmert, and Barczak [64] suggest that shared governance can be a bonding mechanism that helps partners to adapt to new circumstances—for instance, in the context of this study, an emerging organizational priority such as sustainability. In the context of nursing, which is where much of the literature of shared governance exists, outcomes of moving to a governance model include improved financial performance, employee satisfaction, and retention [65]. For university–industry partnerships, evidence points to the importance of shared governance and intellectual property policies as significant facilitators of trusting relationships and successful outcomes [66]. Moreover, Bstieler et al. [63] interpret trust formation as an outcome of parties collectively working through challenges, negotiating mutual adaptations, and making project-related decisions.

Considering how sustainability delivery is most effective within functionally integrative organizational structures [24], an opportunity exists to deductively combine the theory of loosely coupled systems and the concept of shared governance in the context of higher education sustainability. Adams' [28] framework for managing sustainability in universities posits effective sustainability integration as an outcome of a collaborative approach. Adams' framework also posits effective sustainability integration as dependent on the senior person responsibility for sustainability being empowered to lead. The juxtaposition of a collaborative approach with the empowerment of individuals to make decisions renders Adams' framework consistent with how Olson [52] understands the role of shared governance in decision-making. Together, shared governance and effective sustainability integration provide a relevant analytical framework to understand the exact types of sustainability governance structures to which athletic departments are involved. Shared governance also yields a number of benefits such as reducing asymmetry [61], promoting common understanding [62], and facilitating coordination [63], which are all means by which systems become increasingly coupled. Therefore, in understanding shared governance as a compensation for loose coupling, these concepts offer a basis for analyzing the benefits of involving athletic departments in such structures.

4. Methods

4.1. Research Design

This study applied a descriptive research design to establish a "relatively concrete description" [67] (p. 1) of athletic department involvement in organizational-level models of shared sustainability governance. The uncertainty surrounding athletic department involvement in models of shared governance beyond the unit level justified a descriptive research design with the multi-faceted goal of describing situations and events, examining why patterns in these situations and events exist, and determining what these patterns imply [68]. Participant interviews were used as the primary research method. Specifically, the participant interviews followed a semi-structured format, which allowed for interviews to cover the same topics along with some flexibility to probe other related topics [69]. The semi-structured interviews enabled the elicitation of subjective experiences relating to athletic department involvement in shared sustainability governance.

4.2. Study Population

For this study, the population comprised degree-granting institutions in the United States committed to advancing sustainability, which also included a National Collegiate Athletic Association (NCAA)-affiliated athletic program. To access this population, sampling was restricted to universities participating in the AASHE Sustainability Tracking, Assessment and Rating System (STARS), which is a transparent, self-reporting framework for colleges and universities to measure their sustainability performance. Furthermore, STARS is considered "the singular consensus-based tool for use by colleges

and universities" [70] (p. 100), and all completed reports are publicly accessible online. In December 2018, at the time of sampling, 228 universities out of a total of 1027 active STARS participants met the inclusion criterion of having an active report at the Bronze level or higher and a NCAA affiliated athletic program. A further review of STARS reports identified the number of universities who involve their athletic departments in organizational-level models of shared sustainability governance. Specifically, involvement was understood through responses within the 'PA-1: Sustainability Coordination' credit category where universities were asked to list sustainability committee members, to which a total of 32 institutions emerged. Figure 2 outlines the steps taken to arrive at a target population. [71]



Figure 2. Target Population.

Furthermore, the target population included a majority of public institutions (22 public and 10 private) with NCAA Division I athletic programs (28 Division I, two Division II, and two Division III) and STARS certification at the Gold level (19 Gold and 13 Silver).

4.3. Data Collection

Semi-structured interviews were conducted with university sustainability personnel from 12 of the 32 universities identified in the target population. The sample of 12 universities was approximately representative of the target population with a majority of public institutions (10 public and two private), NCAA Division I athletic programs (11 Division I and one Division II), and STARS certification at the Gold level (seven Gold and five Silver). The institutions were sampled using non-specific emphasis with an opportunistic strategy [72]. First, the researchers took advantage of the network of sustainability practitioners at their host institutions within the target population, which yielded five research participants. Second, the researchers made contact with all of the remaining institutions by phone or e-mail with a view to securing additional research participants. The recruitment of research participants conformed to a number of protective principles, which were vetted by the institutional review board at the lead author's host institution, including the right to participate and cease participation, the full disclosure of the study context, and the right to privacy [73]. Ultimately, the sample of 12 enabled a point of data saturation in which the ability to obtain additional new information was attained [74].

The justification for using sustainability office personnel as interview participations was based on their knowledge of and experience with the phenomenon of interest [75]. Specifically, the frequent interactions of sustainability office personnel with shared sustainability governance furnished them with the subjective knowledge from which to construct a descriptive understanding of the situation. While 12 institutions were sampled, 13 participants agreeing to be interviewed (i.e., two representatives from the same institution agreed to participate in a joint interview). Of these 13 participants, seven held positions at the director level, three at the coordinator level, one at the manager level, one at the officer level, and one held a title of assistant to a senior administrative officer. Interviews were

conducted via telephone and lasted on average 25 minutes each. All interviews were conducted over a four-week period across the months of February and March 2019.

The interview guide was developed using the literature on sustainability management in higher education, loosely coupled systems, and shared governance. Specifically, the interview guide was separated into two parts in line with the two research questions. For example, the essential and desired elements of Adams' [28] framework for managing sustainability in higher education guided a number of questions in the interview guide relating to exactly what structures athletic departments were engaged in. Similarly, the benefits of shared governance (see [48–51]) guided the questions relating to how the interview participations perceived athletics to benefit from engagement in shared sustainability governance structures, which were also framed using the voice of compensations [4]. As such, the interview guide made use of direct and indirect questioning. The use of indirect questioning was seen as a key strategy for ensuring data credibility through informant honesty [76], which helped mitigate the effects of social desirability bias [77] by asking the participants (sustainability office personnel) to respond from the perspective of another group (athletic department). Furthermore, credibility was sought through a process of early familiarity with the culture of participating organizations [76], which included a broader review of each institution's STARS report and other relevant strategic documents posted online.

4.4. Data Analysis

Interviews were transcribed by the researchers and converted to a text format, which generated 41 single-spaced pages of data. Data analysis was guided by an analytical framework derived from literature on sustainability management in higher education, loosely coupled systems, and shared governance. To start, the researchers used the two primary research questions as a basis for structural coding, which allowed for further coding based on comparable segments [78]. Next, both data segments underwent a process of thematic analysis in keeping with Braun and Clarke's [79] methods, which allowed for the researcher to play an active role in the creation of themes in the data. For the first data segment, elements of Adams' [28] framework for managing sustainability in higher education along with Olson's [52] characteristics of shared governance were used as a coding frame from which to categorize athletic department involvement in shared sustainability governance structures. For the second data segment, the coding frame was derived from the literature on loosely coupled systems as well as the benefits of shared governance (i.e., trust, bonding, reducing asymmetry, goal coordination, etc.). Finally, the data segment relating to the second research question underwent descriptive coding, which was intended to augment and build off the deductive analytical framework to capture any unexpected topics.

The coding of transcripts was conducted using NVivo 12 software.

5. Findings and Discussion

The findings and discussion section is divided into four sub-sections. First, the shared sustainability governance engagement is discussed, which not only covers how athletic departments come to be involved, but also what systems of shared sustainability governance athletic departments are becoming involved in specifically, why they are involved at this level, and who is representing athletic departments on shared governance coordinating councils. Second, the compensatory outcomes of athletic department involvement in shared sustainability governance are discussed with a particular focus on how shared governance is facilitating tighter coupling. Third, the limitations of shared sustainability governance as a coupling mechanism are presented. Fourth, the limitations of the study are discussed.

5.1. Engaging Athletics in Shared Sustainability Governance

The respondents described shared sustainability governance within higher education mostly as interactions among and between either departments or subunits of departments. For example, one

respondent described the coordinating council for sustainability as having "representation from across the institution" and "every department and business unit". However, what is also evident is interactions between departments and individuals who represent ideas and intentions. Specifically, in referring to the representation of the coordinating council for sustainability, another respondent stated:

It's primarily staff, facilities, campus planning and development, academic affairs, so the deans of all of the colleges on campus or their designate is represented, and then a representative from housing and residential life, and then those students who are designate environmental senators serve on the committee along with a handful of other students that have various roles on campus.

Shared sustainability governance appears to reflect departmental interactions through the interactions between individuals who represent those departments. On one hand, shared sustainability governance calls for representative involvement, rather than total, which is a logistical necessity considering the vastness of university campuses. However, on the other hand, the presence of individuals representing ideas and intentions also appears to be a matter of organizational structure and hierarchy. As an example, one respondent described how the coordinating council included a president, directors, faculty, and students. In other words, the president and directors represent their particular position and directorates by way of seniority, whereas faculty and student members of the shared sustainability governance system likely represent their colleagues and peers through a personal or professional interest in the subject matter.

Further context provided by the respondents into the decision-making authority of the coordinating council reveal some variation in committee purpose and function. Some sustainability office personnel discussed their coordinating council as an entity that drives campus sustainability strategy through the identification of very specific areas for the university to target, which is consistent with Orton and Weick's [4] claim that a careful selection of targets can compensate for loose coupling. As an example, one respondent shared how the coordinating council "weighed in heavily" on a number of campus issues, which also highlights how the coordinating council leverages an enhanced leadership status to influence other organizational levels and components [4]. Furthermore, another respondent shared how the university has been using the coordinating council "as governance for decisions as we move forward," which involved having input into changing university-wide goals as well as making changes to the campus sustainability plan.

However, some coordinating councils play a less authoritative role in campus sustainability matters. For example, one respondent described the purpose of the council as a "sounding board and a vetting entity" for the initiatives of the office of sustainability, while another described the role of the council to advise the administration on what can be done around sustainability. As a result, some shared sustainability governance systems reflect a more typical definition of shared governance that includes a balance between corporate and collegial approaches [53], and between participatory and primary decision-making responsibilities [52]. Alternatively, other systems appear to have less of an orientation around accountability, whereby efforts simply look to facilitate purposeful behavior towards the attainment of a goal or final state [80].

First and foremost, the justification for including athletic departments in shared sustainability governance is a matter of visibility. For example, according to one respondent, athletics is "the public face of the university". More specifically, another respondent claimed athletics to be "one of the most visible activities for any university that has Division I athletics". The high visibility of athletics in these universities presents an opportunity for sustainability to engage new audiences, which, according to one respondent, includes a "broad brush of stakeholders that traditionally the university research outcomes would not necessarily reach". As such, intercollegiate athletics occupies a unique position as a university department, which another respondent made clear by suggesting how "people have an affinity to the university as a result of athletics" and that "if anyone is going to help with promoting sustainability and creating outreach and awareness, athletics is a big component of that".

Subsequently, athletics provides a very public gateway for various stakeholders to gain a sense of the university's commitment to sustainability. As one respondent stated, "athletics has become the face of sustainability in many ways". This respondent further elaborated on this connection with the following statement:

I liken this to a job I had in the private sector where we were committed to sustainability, but it didn't have a face until we tied it to one of the more popular products. Once we tied sustainability to athletics that gave us a front door to open conversations with a number of different stakeholders.

Skyttner [81] contends that the introduction of an input that is both unique and time critical may permit a semi-organized system to organize itself and to grow. In essence, the introduction of athletics as an input into the sustainable development journey of the university has had a similar effect. Furthermore, the majority opinion of sustainability office personnel suggests how athletic departments possess the potential to leverage their position as the 'front porch' of the university [17] to become the 'sustainable front porch'. As one respondent stated that "certainly how the public sees us it's really vital", and continued on to elaborate that, proportionately, "... even a smaller initiative in athletics could make a big difference". Ultimately, athletic departments serve in a sustainability ambassadorial role on university campuses.

Second, some shared sustainability governance systems involve their athletic departments, and simultaneously exclude other departments on campus, due to athletics' status as a high-impact partner. For example, one respondent considered intercollegiate athletics to be "a major function of the university". More specifically, another respondent claimed:

Athletics has two primary areas where they overlap with sustainability—one is greenhouse gas emissions, and the other is materials.... So by engaging the Athletics Director we are reaching an important subset of campus that is having an impact in those sustainability areas.

Moreover, this impact according to another respondent includes a "big role" within waste production and resource consumption. As such, combining high visibility with a high impact as grounds to include athletics in shared sustainability governance helps universities in their quest to meet their twofold sustainability responsibility, which Stephens, Hernandez, Román, Graham, and Scholz [36] describe in terms of sustainable operational change and sustainability change promotion.

Despite the compelling case for the inclusion of athletics in coordinating councils, much variation exists in exactly who represents the athletic department at this level. At one university, the sustainability office representative identifies the Athletic Director as the coordinating council appointee for athletics, but, in the same breath, explains how appointees are free to delegate who actually attends the formal meetings. A number of other sustainability office personnel have shared how coordinating council representation sits with the Director of Athletics, which is somewhat contradictory to how Casper et al. [18] present sustainability decision-making within Division I programs as occurring predominantly at levels below that of the Athletics Director. However, in other cases, athletic department representation has taken the form of a more facilities-based approach (e.g., Director of Facilities, Director for Internal Operations, Stadium Manager, etc.). In the case of the latter, athletic department representation in shared sustainability governance potentially falls foul of how Stephens et al. [36] claim universities often focus disproportionately on the way higher education can change internally, rather than realizing the potential of higher education as a sustainability change agent.

While athletic department representatives appear to vary across campuses in title, seniority, and role, the rationale for appointment or identification of representatives shares much similarity across different contexts. Specifically, personal interest, experience, and passion for sustainability often provide a justifiable means for inclusion in the coordinating council. For example, one respondent described how "a personal interest in sustainability" is useful "no matter what role you have". Likewise, another respondent described their athletics representative as someone who "was familiar with LEED

[Leadership in Energy and Environmental Design, green building rating system] and sustainability initiatives in general" and who "came in a few years ago, kind of charged, fired up about those kinds of ideas".

In terms of how these athletic department representatives become involved in shared sustainability governance, some athletics representatives are appointed, while others become involved voluntarily. As an example, one respondent described the process of appointment, and stated that "every year the executive vice president comes up with a list of new [coordinating council] members and we do that in conversation with the committee co-chairs". Another respondent discussed how the appointment of representatives helps create a routine, which, in turn, "helps to make sure they are attending our meetings and are engaged in our sub-committees and working groups". Yet, such efforts demonstrate what Orton and Weick [4] consider to be focused attention, whereby sustainability office personnel who are empowered by senior accountable officers are forcefully creating orderly contingencies among university departments and campus constituents.

Alternatively, athletic department representatives are sometimes groomed for voluntary coordinating council involvement through their prior involvement in on-the-ground sustainability initiatives. For example, one respondent described the following scenario:

So we did have someone within athletics two years ago who had more experience working with green athletic programs at other universities. He helped us to implement green soccer games, we connected him with our composting and catering on campus, and worked really hard to make sure there was as little waste at games as possible.

Therefore, the typical approach appears to focus on engaging representatives who already have a basic knowledge and awareness of sustainability. Furthermore, just as Pfahl et al. [44] identify how collaborative processes are initiated by sustainability office personnel, the identification of suitable coordinating council representatives from within athletics often lies with the sustainability office.

Regardless of the nature of the forceful attention applied by systems of shared sustainability governance, sustainability office personnel emphasize the importance of formalizing the commitment beyond the level of the individual. For example, as one respondent claimed that "it is very important to find ways to institutionalize what we're doing so that when people leave everything doesn't fall apart". Similarly, another respondent described how informal commitments are problematic, especially when coupled with a perceived high staff turnover within athletics, stating: "You're making a little bit of progress in terms of getting some people on board, and then again, unfortunately he left athletics." Additionally, the interviews suggest not all of what is disclosed on the STARS report is the case in practice. Specifically, three of the 12 institutions included within the sample referred to the athletic department no longer being a member of their university-wide sustainability committee whether that be due to personnel changes or a shift in the approach of the coordinating council.

The sample of 12 universities included within this study also implemented different variations of shared governance. While all 12 universities convened a coordinating council, only seven universities implemented a subcommittee structure that fed into the coordinating council, thus adopting a true councilor model of shared governance. Additionally, seven of the 12 universities implemented an appointee system for shared governance representatives, while six preferred to secure shared governance representation through voluntary commitments. Most of the university-level councils were coordinating councils, while others were advisory councils. Similarly, most of the councils involved athletics within the coordinating council, while others were involved within thematic subcommittees. On one hand, these findings reify previous research (see [20]), which provides evidence of the varying scopes and remits of environmental committees. On the other hand, however, the variation in shared sustainability governance systems supports Lapworth's [57] proposition that effective models of shared governance require diversity and flexibility to suit context-specific character and needs.

5.2. Shared Sustainability Governance as a Compensation for Loose Coupling

Sustainability office personnel share a number of benefits to engage athletic departments in shared sustainability governance. Functionally, even at the university level, shared sustainability governance serves as a conduit for collaboration whereby a diverse group of people and departments come together to create sustainable solutions. One respondent explained how the functional capability of the coordinating council and the resulting diverse thinking, in many ways, encapsulates the very notion of sustainability "in a nutshell". Other practical examples have also been shared. For example, one respondent was able to draw on scenarios where the athletic department had tabled possible solar and stormwater management projects through the coordinating council, which sparked a number of possible inter-departmental collaborations. In essence, this respondent felt that engagement in the coordinating council was helping the athletic department with their future sustainability plans. Again, the fact that university-level models of governance provide opportunities to table department-specific problems is another example of the varying scopes and remits of environmental committees [18,44]. Furthermore, the fact that sustainability problems are being tabled in strategic-level meetings provides evidence of the loose coupling of sustainability practices across university departments.

Beyond the direct effects of collaboration, shared sustainability governance is seen as critical to mitigating the loss of control that the decentralized, loosely coupled structures within higher education promote. Specifically, one sustainability office representative stated:

A lot of times athletics can feel like its own separate world because they are funded differently and all that. But they are still a part of the university. So I think having them be a part of a larger sustainability committee helps them realize being bold and savvy should translate over to athletics as well.

Accordingly, these findings reify the work of McCullough, Kellison, and Wendling [30], who suggest cross-functional sustainability teams provide a bridge with university-wide goals through formal, joint decision-making processes. Similarly, shared governance provides an opportunity for universities to operationalize sustainability as a legitimate bonding mechanism, which Bstieller et al. [64] suggest is made possible by how shared governance enables organizations to adapt to new circumstances. After all, as one respondent claimed, athletic departments are "not a separate wing that can go off on their own".

Athletic department involvement in shared sustainability governance enables, as Orton and Weick [4] suggest, a genuine reaffirmation of shared values. As an example, one respondent claimed how securing commitments from the athletic department to the coordinating council "shows that they value the campus values" and are "trying to support a larger vision of the university". Moreover, another respondent described the sense of having "a unified presence" where all major functions of the university were "on the same page" as the main benefit of engaging the athletic department within the coordinating council. Subsequently, shared sustainability governance likely creates assumptions among non-athletic department staff that attendance is in some way indicative of Meyer and Rowan's [82] notion of "the logic of confidence and good faith", which creates the assumption that departments are performing their tasks and responding to their responsibilities correctly.

Sustainability office personnel also consider how a shared vision helps athletic departments, and other university departments for that matter, in identifying the potential synergies and leverage points of sustainability. In turn, these opportunities foster a sense of motivation toward sustainability action, which one respondent summarized as an emergent perspective on the "direction for how and what type of action to take". Furthermore, another respondent suggested the coordinating council provides the athletic department with the drive to "get ahead and stay ahead" with peers on campus, which leverages the competitive edge that is synonymous with athletics competition. As such, shared sustainability governance in this context is active in promoting a common understanding of goals [62], as well as coordinating goals and skills between partners [63]. Additionally, reaffirming shared values

is critical to creating a level of responsiveness to sustainability, which, in this domain, supports coupling across the university.

Sharing values and perspectives is not one-directional. In fact, a number of respondents discussed how having the athletic department represented in the coordinating committee facilitates a deeper appreciation of the unique challenges faced by athletic programs. The following passage from one of the respondents embodies this mutually beneficial process:

We understand athletics has special needs and concerns, but we might not fully understand what those concerns are. So to have them at the table, it's great. So for example, just turf, our baseball fields are turf. So for sustainability it reduces water I guess and pesticide use because it's not real grass. But that wasn't their only reason for getting it, it was because we can't compete with southern schools because it's cold and shitty, you know, up here in spring. So we have to have artificial turf in order for our fields not to flood.

Similarly, another respondent discussed how a lack of involvement by several campus constituents with the athletic department, including students, faculty, and staff, often leads to misinformed perceptions of decision-making within athletics, to which this same respondent claimed:

Perhaps there may be an actual reason why the stadium lights are on all night long and there's no game. Maybe there's an actual reason. Like they're doing construction in there or something like that. And those kind of things come up in meeting and people were actually getting an answer, and they were like 'Right. OK. That makes sense.'

Accordingly, shared sustainability governance is active in reducing information asymmetry [61] and helping to prevent misunderstandings between governance members [62]. As one respondent summarized, "it makes the athletic department more personable", which provides a platform from which to build trusting relationships and, potentially, successful outcomes [66].

Yet, overall, shared sustainability governance appears to compensate for the decentralized nature of university structures, which makes sustainability a discretionary endeavor. To do so, shared sustainability governance enhances leadership around specific targets by reaffirming with the athletic department the value of sustainability.

5.3. 'A Perfect Storm': Supplementing Shared Sustainability Governance

Despite athletic department involvement in shared sustainability governance furnishing a number of compensatory benefits, the benefits are neither uniform nor absolute. As an example, collaboration was understood as a benefit by some, but by others collaboration was mostly one-sided. On the topic of collaboration between the athletic department and other university departments or campus constituents, one respondent replied that most of these collaborations are "people reaching out and engaging them [the athletic department] and it's not happening the other way yet". Furthermore, another respondent expressed their frustration by stating how they were "getting participation on a campus-wide basis from athletics, but not seeing a whole lot within the athletics department just yet". So, there is a sense that athletic departments are, in some ways, honoring the commitment to attending coordinating council meetings, but not necessarily converting this commitment into action. In this sense, the shared sustainability governance structures represent an epistemic community where actors share basic causal beliefs and normative values [83]. However, as Newig, Günther, and Pahl-Wostl [84] argue, participation in these communities does not necessarily involve the same level of interest in the problem at stake, which in this case is sustainability.

Similarly, the coercive nature of appointment-based engagement in shared sustainability governance is likely a step too far for some athletic departments. As one respondent explained:

They see the benefit, but I think it's still at the point where we're making them do something. So I think it's been a long time where I think that's where we're perceived as we're telling them to do something rather than working with them. Herein lies one of the challenges presented by forcing attention toward specific targets or goals within a decentralized system. By forcing athletic department attention toward sustainability, athletics personnel perhaps feel constrained to operate within the parameters of the university. Fittingly, another sustainability office representative used a sporting metaphor to effectively relate this thought by suggesting that the athletic department was "sort of playing the game and they're on the field. I'm not sure if they've really realized what the end goal is". Such a quandary leaves some sustainability office personnel increasingly frustrated in their attempts to help athletic departments realize the full extent to which sustainability creates value. To this end, one respondent claimed that "it is a nut that we have officially yet to crack and even with athletics being a part of the committee on sustainability they weren't necessarily actively pushing for sustainable practices in athletics".

Ultimately, the overriding consensus among sustainability office personnel interviewed in this study was that shared sustainability governance involvement only stands to enhance commitment when used in concert with other change levers. This sentiment was most accurately explained by one respondent who described a scenario whereby the university has "support from the top down and the bottom up, and then you have some champions in the middle that helps to connect that. It's rare to find that, to find all of those factors together in the same institution, a perfect storm." Figure 3 is a visual representation of the change levers the respondents disclosed, which either they have successfully deployed or that they feel would be a success given the complexities of engaging athletic departments in the sustainability journey.





Casper and Pfahl [19] suggest "environmental activities blur lines between distinct management functions making it both a top-down and bottom-up issue at the same time" (p. 24). Figure 3 embodies the notion of the 'perfect storm' by organizing the disclosed change levers through not only a top-down and bottom-up approach, but a multidimensional, all-angles approach. Specifically, there are strategic moves that the university could enforce in a top-down manner, or whereby a critical mass could focus attention through bottom-up endeavors. Likewise, there are strategic moves that the university could introduce to incentivize lateral collaboration or involvement in sustainability, as well as measures the athletic department could activate from an inside-out perspective. As such, the identification of these supplemental change levers represent the "nonobvious sources of order that administrators can use to

influence dispersed organizations" [4] (p. 211). Table 1 provides additional context and supporting quotations for each of the proposed change levers.

Lever	Direction	Representative Quotation
Rigorous		Until all sustainability is just incorporated into every single unit or until out climate commitments have some sort of teeth behind them and saying you must do it.
sustainability goals	Top-down	I think the opportunity we've had is just applying these increasingly rigorous sustainability goals for all facilities to athletics, which sort of brings them along in a way that maybe they haven't called out for themselves.
Leadership directives	Top-down	My hope would be that the department's priorities are synced with the university's priorities, and values, and one of those is sustainability. So it is my hope that would filter down.
'Low-hanging fruit'	Lateral	I think there's an interest within waste management and recycling and composting. I say that just because that's the thing we spend the most time talking about and folks seem to get the most excited about. We do have a lot of low-hanging fruit opportunities there within that realm. So that might be one sort of focal area where I can sort of see a little bit more traction with them.
Voluntary initiatives	Lateral	We are also working on a program that is in-house that allows departments to sign up to make their travel carbon neutral or by helping the university to reduce greenhouse emissions through efficiency projects on campus. We are hoping that by using programs like this it will allow departments and events to advertise the fact that they are doing something that is carbon neutral and get credit for that, that will apply pressure, peer pressure.
Internal framework	Inside-out	I would say the responsibility of the university writ large with sustainability would be to create a culture where somebody on the inside of athletics says, 'you know what, sustainability is really part of what we do and we need to call it out here.'
		I think having a framework for them can sort of clearly define their expectations, their strategies, to make athletics more sustainable.
Leadership support	Inside-out	I would think it would have to be our Athletics Director wanting sustainability to be part of what they do. And that might happen through conversations with our President But if it became part of the leadership team and one of their goals was around sustainability, then I could imagine a greater involvement with athletics.
		But I think it they are going to go all in, then it's going to take that top level leadership to say, 'look, this is really important'. We want a culture of winning and winning games, but we're not just winning on the field – we're winning off the field.
Sustainability coordinator	Inside-out	I think it would take dedicated time from the current staff. We can help support from the outside, but with larger departments on campus there are certain sub-cultures in these departments. And so until those sub-cultures recognize sustainability as part of their daily practices, I think any change that comes from the outside will only be temporary.
		For the athletics department to become more sustainable and really push that, honestly it's having someone that's dedicated 24/7 to the operations of our athletics department.
Sponsorship	Inside-out	The other would be sponsorship. Nothing has come to fruition yet, but that is something that if it did happen, that would be a much easier way for athletics, especially the operations, to really make it a priority. That would be a game changer.
Student-athlete	P. 44	Two years ago I had a student working on zero waste. They worked a lot on reducing waste and raising awareness on campus Our Athletics Director was engaged in that and he helped and advocated for building composting into some of the athletics events and recycling. So that connection, that student-to-director at a smaller scale and not as part of the large group was probably more impactful than coming to the large group.
action	Bottom-up	I think student interest goes a long, long way. And so if some of the sports teams are interested in this, saying that it is a huge problem for the future of the sport, that would go a long way. I think that goes farther than anything I can say or do as a staff member.
		I think on a college campus the more the student are invested in, involved in something, the better chance of success we have.

Table 1. Levers for forcing athletic department attention toward sustainability.

5.4. Limitations of the Study

Using AASHE STARS as a means by which to identify cases of athletic department engagement in shared sustainability governance structures presents challenges in reaching the population of interest. First, using AASHE STARS in the context of this study did limit the sample population to universities who chose to complete the reporting process. Second, AASHE STARS, like the majority of assessment-based reporting processes, provides a snapshot in time, and what this study revealed was that shared sustainability governance structures, models, and membership are in a constant state of flux.

In addition, compensations proved a relevant voice to use within the loosely coupled context of athletic departments and wider university sustainability goals and priorities, as most respondents within this study considered this level of coupling to be a sub-optimal state. Yet, perhaps loose coupling in this context is more advantageous than disadvantageous. As one respondent claimed: "The university is so diffuse. There is a lot of autonomy. Ideas are coming from departments. How can we do x, y and z? There is no top-down pressure being applied." In such a scenario, the coordinating council empowers the representatives to go back to their departments and say this is worthwhile, rather than returning simply to communicate and provide an update on the university's overall strategy and goals. Furthermore, Haas [83] suggests the best institutional structure for dealing with complex policy environments is, in fact, loose and decentralized, but also dense networks. Haas [83] suggests network density enables the quick relay of information, and that the loosely coupled and decentralized nature of such networks allows the inactivity of one actor not to jeopardize the efforts of the whole. Accordingly, while the voice of compensations provides a sufficiently narrow theoretical guide for this study, under-utilizing other loose coupling voices (e.g., the voice of direct effects or organizational outcomes) presents a limitation in terms of the insights generated.

6. Conclusions

Theoretically, this study provides further evidence of the dialectical nature of loose coupling within the higher education context, and, more specifically, between a more contemporary coupling context than the historical focus on interdependencies between administrators and faculty. The study supports how athletic departments demonstrate at least a basic level of engagement in sustainability that mirrors the wider university (i.e., determinacy). Yet, the varying levels of commitment from athletic departments to shared governance structures demonstrate a simultaneous level of independence between athletics and the university more generally. Athletic departments highlight the challenges universities face in reflecting institutional commitments across all aspects of the campus, which the Rio+20 People's Sustainability Treaty on Higher Education [85] suggests spans campus management, curriculum, research, and student and community engagement activities. As such, athletic departments are a key internal stakeholder for universities to engage in order to achieve a whole-of-institution approach and commitment to sustainability.

The insight generated through this study points to a number of practical implications for sustainability office personnel. Likewise, such implications are potentially useful for athletic department staff who identify as sustainability leads, and who wish to create greater buy-in internally from colleagues. First, justification should be made for athletic department involvement in shared sustainability governance. For example, does the athletic department boast high visibility as well as being a high-impact campus department? If yes, then athletics likely deserves a seat at the table as a critical cog in the twofold responsibility of universities to sustainability (i.e., as an operation to be changed, and as a change agent). Second (and dependent upon the model of shared governance being deployed), is where exactly the athletics departments are engaged in the system. Athletics, as understood through several of the respondents within this study, has the potential to be the 'sustainable front porch' of the university. However, depending on the context, there might be more merit in engaging the athletics department through a thematic subcommittee where tighter coupling exists, and where greater leverage points and synergies can be pitched to optimize levels of engagement.

Third, identifying the most appropriate person to represent the athletics department is crucial, as it is this individual who will serve as a more formal bridge with the university's sustainability goals. Fourth (and connected to the third practical implication), is the extent to which the shared sustainability governance system will focus the attention of the athletic department. In other words, who will represent the athletic department and will this representative be appointed or approached with a request to voluntary join the coordinating council and/or thematic subcommittee? Such decisions are not solely a matter for athletic departments. However, appointing the Director of Athletics to the coordinating council does not automatically amount to an institutionalized commitment. Rather, sustainability office personnel should match as best they can the personal interest and passion for sustainability of existing athletic department staff with an appropriate means by which to make this engagement routine. Recognizing the relatively high turnover of staff within athletics positions, those charged with heading up shared sustainability governance systems should perhaps focus on succession planning. For example, campuses that address leadership continuity as well as providing training and advice to new officers are more successful at governance [86].

Finally, and most importantly, efforts to engage athletic departments in shared sustainability governance should be supplemented by other change levers. Ultimately, the aspirational condition is to achieve a 'perfect storm' that calls for a combination of top-down, bottom-up, lateral, or inside-out tactics. Yet, while this study suggests that securing an institutionalized commitment is a complex matter, further research is necessary to tease out the importance and effectiveness of multi-dimensional efforts to engage athletics departments in sustainability action. For example, an in-depth case study of a university whose athletics department is immersed in this multi-dimensional approach would elucidate the relative importance of each of the strategic levers. More specifically, a scarcity of research exists that understands the role of student athletes as catalysts for athletics department sustainability action, and further qualitative research could generate a rich understanding of the role of sustainability in the dual identity of student athletes on university campuses in the United States.

Author Contributions: Conceptualization, M.B.; data curation M.B.; formal analysis, M.B.; methodology, M.B.; supervision, K.S.B.; writing – original draft, M.B.; writing – review and editing, K.S.B., J.M.C., M.B.E, D.S.S. and G.J.J.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Krizek, K.J.; Newport, D.; White, J.; Townsend, A.R. Higher education's sustainability imperative: How to practically respond? *Int. J. Sustain. High. Educ.* **2012**, *13*, 19–33. [CrossRef]
- Reponen, T. Is leadership possible at loosely coupled organizations such as universities? *High. Educ. Policy* 1999, 12, 237–244. [CrossRef]
- OECD. Changing patterns of governance in higher education. In *Education Policy Analysis*; OECD: Paris, France, 2003; Available online: https://www.oecd.org/education/skills-beyond-school/35747684.pdf (accessed on 12 September 2019).
- Orton, J.D.; Weick, K.E. Loosely coupled systems: A reconceptualization. Acad. Manag. Rev. 1990, 15, 203–223. [CrossRef]
- Meyer, J.W.; Rowan, B. The structure of educational organizations. In *Schooling in a Social Context*, 3rd ed.; Ballantine, J.H., Spade, J.Z., Eds.; Pine Forge Press: Lose Angeles, CA, USA, 2008; pp. 217–226.
- Hardy, C.; Langley, A.; Mintzberg, H.; Rose, J. Strategy formation in the university setting. *Rev. High. Educ.* 1983, 6, 407–433. [CrossRef]
- 7. Birnbaum, R. How Academic Leadership Works; Jossey-Bass: San Francisco, CA, USA, 1992.
- 8. Blackburn, R.T.; Lawrence, J.H. Faculty at Work; The John Hopkins Press: Baltimore, MD, USA, 1995.
- Del Favero, M.; Bray, N.J. Herding cats and big dogs: Tensions in the faculty-administrator relationship. In *Higher Education: Handbook of Theory and Research*; Paulsen, M.B., Perna, L.W., Eds.; Springer: Dordrecht, The Netherlands, 2010; Volume 25.

- Peterson, M.W.; White, T.H. Faculty and administrator perceptions of their environments: Different views of different models of organization? *Res. High. Educ.* 1992, 33, 177–204. [CrossRef]
- Chandler, J.C., III. Bridging the Chasm: Emerging Model of Leadership in Intercollegiate Athletics Governance. Ph.D. Theis, The University of Las Vegas Nevada, Las Vegas, NV, USA, December 2011. Available online: https://digitalscholarship.unlv.edu/cgi/viewcontent.cgi?referer=https://scholar.google.com/ &httpsredir=1&article=2375&context=thesesdissertations (accessed on 12 September 2019).
- 12. Hanford, G.H.; Greenberg, M. We should speak the "awful truth" about college sports ... or does the public like the status quo? *Chron. High. Educ.* **2003**, *49*, 10–13.
- 13. Sack, A.L. College sport and the student athlete. J. Sport Soc. Issues 1987, 11, 31-48. [CrossRef]
- 14. Suggs, W. Old challenges and new opportunities for studying the financial aspects of intercollegiate athletics. *N. Dir. High. Educ.* **2009**, *148*, 11–22. [CrossRef]
- 15. Duderstadt, J.J. Intercollegiate Athletics and the American University: A University President's Perspective; The University of Michigan Press: Ann Arbor, MI, USA, 2009.
- Sanderson, A.R.; Siegfried, J.J. The case for paying college athletes. J. Econ. Perspect. 2015, 29, 115–138. [CrossRef]
- Bass, J.R.; Schaeperkoetter, C.C.; Bunds, K.S. *The 'Front Porch': Examining the Increasing Interconnection of* University and Athletic Department Funding; ASHE higher education report; John Wiley & Sons: Hoboken, NJ, USA, 2015; Volume 41.
- Casper, J.M.; Pfahl, M.E.; McSherry, M. Athletics department awareness and action regarding the environment: A study of NCAA athletics department sustainability practices. J. Sport Manag. 2012, 26, 11–29. [CrossRef]
- Casper, J.M.; Pfahl, M.E. Environment sustainability practices in U.S. NCAA Division II athletics departments. Int. J. Event Manag. Res. 2015, 10, 12–36.
- 20. Casper, J.M.; Pfahl, M.E.; McCullough, B.P. Intercollegiate sport and the environment: Examining fan engagement based on athletics department sustainability efforts. J. Issues Intercoll. Athl. 2014, 7, 65–91.
- Ferrer-Balas, D.; Adachi, J.; Banas, S.; Davidson, C.I.; Hoshikoshi, A.; Mishra, A.; Motodoa, Y.; Onga, M.; Ostwald, M. An international comparative analysis of sustainability transformation across seven universities. *Int. J. Sustain. High. Educ.* 2008, *9*, 295–316. [CrossRef]
- 22. Adams, R.; Martin, S.; Boom, K. University culture and sustainability: Designing and implementing an enabling framework. *J. Clean. Prod.* **2018**, *171*, 434–445. [CrossRef]
- Asif, M.; Searcy, C.; Zutshi, A.; Fisscher, O.A.M. An integrated management systems approach to corporate social responsibility. J. Clean. Prod. 2013, 56, 7–17. [CrossRef]
- 24. Viebahn, P. An environmental management model for universities: From environmental guidelines to staff involvement. J. Clean. Prod. 2002, 10, 3–12. [CrossRef]
- Meyer, H.-D.; Rowan, B. Institutional analysis and the study of education. In *The New Institutionalism in Education*; Meyer, H.-D., Rowan, B., Eds.; SUNY Press: Albany, NY, USA, 2006; pp. 1–13.
- 26. De Boer, H.F.; Enders, J.; Leisyte, L. Public sector reform in Dutch higher education: The organizational transformation of the university. *Public Adm.* **2007**, *85*, 27–46. [CrossRef]
- Velazquez, L.; Munguia, N.; Platt, A.; Taddei, J. Sustainable university: What can be the matter? J. Clean. Prod. 2006, 14, 810–819. [CrossRef]
- Adams, C.A. Sustainability reporting and performance management in universities: Challenges and benefits. Sustain. Account. Manag. Policy J. 2013, 4, 384–392. [CrossRef]
- Pfahl, M.E. Strategic issues associated with the development of internal sustainability teams in sport and recreation organization: A framework for action and sustainable environmental performance. *Int. J. Sport Manag. Rec. Tour.* 2010, *6*, 37–61. [CrossRef]
- McCullough, B.P.; Kellison, T.; Wendling, E. Formation and function of a collegiate athletics sustainability committee. J. Amat. Sport 2018, 4, 52–81. [CrossRef]
- ISCN Secretariat. International Sustainable Campus Network (ISCN): Taking Your Campus Sustainability Program to a Global Stage. In Proceedings of the 4th UNICA Green Academic Footprint Workshop, Brussels, Belgium, 27–28 March 2014; Available online: http://www.unica-network.eu/sites/default/files/ ISCN%20Overview_20140327_for%20UGAF_cg.pdf (accessed on 12 September 2019).
- 32. Swearingen White, S. Campus sustainability plans in the United States: Where, what, and how to evaluate? Int. J. Sustain. High. Educ. 2014, 15, 228–241. [CrossRef]
- 33. Second Nature. History. Available online: https://secondnature.org/history/ (accessed on 12 September 2019).

- 34. Swearingen White, S. Early participation in the American College and University Presidents' Climate Commitment. *Int. J. Sustain. High. Educ.* 2009, *10*, 215–227. [CrossRef]
- Tilbury, D. Higher education for sustainability: A global overview of commitment and progress. *High. Educ. World* 2011, 4, 18–28.
- Stephens, J.C.; Hernandez, M.E.; Román, M.; Graham, A.C.; Scholz, R.W. Higher education as a change agent for sustainability in different cultures and contexts. *Int. J. Sustain. High. Educ.* 2008, *9*, 317–338. [CrossRef]
- Hottle, T.A.; Bilec, M.M.; Brown, N.R.; Landis, A.E. Toward zero waste: Composting and recycling for sustainable venue based events. *Waste Manag.* 2015, *38*, 86–94. [CrossRef]
- 38. Klein-Banai, C.; Theis, T.L. An urban university's ecological footprint and the effect of climate change. *Ecol. Indic.* **2011**, *11*, 857–860. [CrossRef]
- Dmochowski, J.E.; Garofalo, D.; Fisher, S.; Greene, A.; Gambogi, D. Integrating sustainability across the university curriculum. *Int. J. Sustain. High. Educ.* 2016, 17, 652–670. [CrossRef]
- 40. Dyer, G.; Dyer, M. Strategic leadership for sustainability by higher education: The American College & University Presidents' Climate Commitment. J. Clean. Prod. 2017, 140, 111–116.
- 41. Salvioni, D.; Franzoni, S.; Cassna, R. Sustainability in the higher education system: An opportunity to improve quality and image. *Sustainability* **2017**, *9*, 914–942. [CrossRef]
- 42. Ralph, M.; Stubbs, W. Integrating environmental sustainability into universities. *High. Educ.* 2014, 67, 71–90. [CrossRef]
- 43. Epstein, M.J. Implementing corporate sustainability: Measuring and managing social and environmental impacts. *Strateg. Financ. January* **2008**, 25–31.
- Pfahl, M.E.; Casper, J.M.; Trendafilova, S.; McCullough, B.P.; Nguyen, S.N. Crossing boundaries: An examination of sustainability department and athletics department collaboration regarding environmental issues. *Commun. Sport* 2015, *3*, 27–56. [CrossRef]
- 45. Glassman, R.B. Persistence and loose coupling in living systems. Behav. Sci. 1973, 18, 83–98. [CrossRef]
- Bossert, S.; Dwyer, D.; Rowan, B.; Lee, G. The instructional management role of the principal. *Educ. Adm. Q.* 1982, 18, 34–64. [CrossRef]
- 47. Brusoni, S.; Prencipe, A. Managing knowledge in loosely coupled networks: Exploring the links between product and knowledge dynamics. *J. Manag. Stud.* **2001**, *38*, 1019–1035. [CrossRef]
- 48. Weick, K.E. Educational organizations as loosely coupled systems. Adm. Sci. Q. 1976, 21, 1–19. [CrossRef]
- Goldspink, C. Rehtinking educational reform: A loosely coupled and complex systems perspective. Educ. Manag. Adm. Leadersh. 2007, 35, 27–50. [CrossRef]
- Tierney, W.G.; Minor, J.T. A cultural perspective on communications and governance. In *Restructuring Shared Governance in Higher Education: New Directions for Higher Education*; Tierney, W.G., Lechuga, V.M., Eds.; Jossey-Bass: San Francisco, CA, USA, 2004; pp. 85–94.
- 51. Tricker, R.I. International Corporate Governance; Prentice Hall: Englewood Cliffs, NJ, USA, 1984.
- Olson, G.A. Exactly What Is 'Shared Governance'? The Chronicle of Higher Education: Washington, DC, USA; Available online: https://www.chronicle.com/article/Exactly-What-Is-Shared/47065 (accessed on 12 September 2019).
- Lechuga, V.M. Exploring current issues on shared governance. In *Restructuring Shared Governance in Higher Education*; Tierney, W.G., Lechuga, V.M., Eds.; Jossey-Bass: San Francisco, CA, USA, 2004; pp. 95–98.
- Cramer, S.F.; Mozlin, R. Sharing shared governance: The benefits of systemness. In Shared Governance in Higher Education: Demands, Transitions, Transformations; Cramer, S.F., Ed.; SUNY Press: Albany, NY, USA, 2017; pp. 89–116.
- SUNY Shared Governance Transformation Team. Preamble to Shared Governance Toolkit Guidelines. 2011. Available online: https://system.suny.edu/media/suny/content-assets/documents/powerofsuny/the-processof-the-power-of-suny/SharedGovernance_20110408.pdf (accessed on 12 September 2019).
- 56. Bejou, D.; Bejou, A. Shared governance and punctuated equilibrium in higher education: The case for student recruitment, retention, and graduation. *J. Relatsh. Market.* **2012**, *11*, 248–258. [CrossRef]
- Lapworth, S. Arresting decline in shared governance: Towards a flexible model for academic participation. *High. Educ. Q.* 2004, *58*, 299–314. [CrossRef]
- Yanko, J.R.; Hardt, M.; Bradstock, J. The clinical nurse specialist role in shared governance. *Crit. Care Nurs. Q.* 1995, 18, 86–90. [CrossRef]
- 59. Bernreuter, M. The other side of shared governance. J. Nurs. Adm. 1993, 23, 12-14. [CrossRef]

- 60. Hess, J.R. Shared governance: Innovation or imitation? Nurs. Econ. 1994, 12, 28-34.
- 61. Saxton, T. The effects of partner and relationship characteristics on alliance outcomes. *Acad. Manag. J.* **1997**, 40, 443–461.
- 62. Asmus, D.; Griffin, J. Harnessing the power of your suppliers. *McKinsey Q.* 1993, 3, 63–79.
- Lawson, B.; Peterson, K.J.; Cousins, P.D.; Handfield, R.B. Knowledge sharing in interorganizational product development teams: The effect of formal and informal socialization mechanisms. *J. Prod. Innov. Manag.* 2009, 26, 156–172. [CrossRef]
- Bstieler, L.; Hemmert, M.; Barczak, G. Trust formation in university-industry collaborations in the U.S. biotechnology industry: IP policies, shared governance, and champions. *J. Prod. Innov. Manag.* 2014, 32, 111–121. [CrossRef]
- 65. Anthony, M.K. Shared governance models: The theory, practice, and evidence. Online J. Nurs. Issues 2004, 9, 138–157.
- Burnside, B.; Witkin, L. Forging successful university-industry collaborations. *Res.-Technol. Manag.* 2008, 51, 26–30. [CrossRef]
- 67. De Vaus, D.A.; de Vaus, D. Research Design in Social Research; Sage: Thousand Oaks, CA, USA, 2001.
- 68. Babbie, E. The Practice of Social Research, 13th ed.; Cengage Learning: Boston, MA, USA, 2012.
- 69. Remler, D.K.; Van Ryzin, G.G. *Research Methods in Practice: Strategies for Description and Causation;* Sage: Thousand Oaks, CA, USA, 2011.
- White, G.B.; Koester, R.J. STARS and GRI: Tools for campus greening strategies and prioritizations. Sustainability 2012, 5, 100–106. [CrossRef]
- NCES. Digest of Education Statistics: 2016. Available online: https://nces.ed.gov/programs/digest/d16/ch_3. asp (accessed on 12 September 2019).
- Palinkas, L.A.; Horwitz, S.M.; Green, C.A.; Wisdom, J.P.; Duan, N.; Hoagwood, K. Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Adm. Policy Ment. Health* 2015, 42, 533–544. [CrossRef]
- Barrow, J.M.; Khandhar, P.B. Research Ethics. Available online: https://www.ncbi.nlm.nih.gov/books/ NBK459281/ (accessed on 12 September 2019).
- 74. Guest, G.; Bunce, A.; Johnson, L. How many interviews are enough? An experiment with data saturation and variability. *Field Methods* **2006**, *18*, 59–82. [CrossRef]
- 75. Creswell, J.W.; Plano Clark, V.L. *Designing and Conducting Mixed Methods Research*, 3rd ed.; Sage: Thousand Oaks, CA, USA, 2011.
- Shenton, A.K. Strategies for ensuring trustworthiness in qualitative research projects. *Educ. Inf.* 2004, 22, 63–75. [CrossRef]
- 77. Fisher, R.J. Social desirability bias and the validity of indirect questioning. J. Consum. Res. **1993**, 20, 303–315. [CrossRef]
- 78. Saldaña, J. The Coding Manual for Qualitative Researchers; Sage: Thousand Oaks, CA, USA, 2015.
- 79. Braun, V.; Clarke, V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- 80. Van Gigch, J.P. Applied General Systems Theory, 2nd ed.; Harper: New York, NY, USA, 1978.
- 81. Skyttner, L. General Systems Theory: Ideas and Applications; World Scientific: River Edge, NJ, USA, 2001.
- Meyer, J.W.; Rowan, B. Institutionalized organizations: Formal structure as myth and ceremony. *Am. J. Sociol.* 1977, 83, 340–363. [CrossRef]
- 83. Haas, P.M. Addressing the global governance deficit. Global Environ. Polit. 2004, 4, 1–15. [CrossRef]
- 84. Newig, J.; Günther, D.; Pahl-Wostl, C. Synapses in the network: Learning in governance networks in the context of environmental management. *Ecol. Soc.* **2010**, *15*, 24. [CrossRef]
- Rio+20. People's Sustainability Treaty on Higher Education. In Proceedings of the United Nations Conference on Sustainable Development, Rio de Janeiro, Brazil, 20–22 June 2012; Available online: https: //sustainabilitytreaties.org/draft-treaties/higher-education/ (accessed on 12 September 2019).
- 86. Lee, B. Campus leaders and campus senates. In *Faculty in Governance: The Role of Senates and Joint Committees in Academic Decision Making;* Birnbaum, R., Ed.; Jossey-Bass: San Francisco, CA, USA, 1991.



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Article



Modelling Organisational Factors Influencing Sustainable Development Implementation Performance in Higher Education Institutions: An Interpretative Structural Modelling (ISM) Approach

Bankole Osita Awuzie^{1,*} and Amal Abuzeinab²

- ¹ Department of Built Environment, Central University of Technology, Bloemfontein 9301, South Africa
- ² Institute of Architecture, De Montfort University (DMU), Leicester LE1 9BH, UK
- * Correspondence: bawuzie@cut.ac.za

Received: 29 May 2019; Accepted: 23 July 2019; Published: 9 August 2019

Abstract: Globally, higher education institutions (HEIs) have continued to record varied sustainable development (SD) implementation performances. This variance has been attributed to the presence of certain organisational factors. Whereas previous studies have successfully identified the factors influencing SD implementation performance in HEIs, few studies have attempted to explore the relationship between these factors and the influence of such a relationship on the management of SD implementation in HEIs. This is the objective of this study. Understandably, knowledge of such relationships will facilitate the development of appropriate frameworks for managing SD implementation in HEIs. Relying on a case study of a South African University of Technology (SAUoT), this study elicits data through a focus group discussion session. An interpretative structural modelling (ISM) focus group protocol indicating extant pair-wise relationships between identified organisational factor categories was extensively discussed. The emergent data was recorded, transcribed verbatim and subsequently analysed. The findings suggest that communication was critical to the prevalence of other factors, hence indicating its centrality to the effective management of SD implementation in HEIs. These findings will guide implementing agents in HEIs towards developing appropriate mechanisms for communicating SD implementation strategies.

Keywords: higher education institutions; implementation; organisational factors; sustainable development; interpretative structural modelling (ISM)

1. Introduction

Based on their time-honoured role in shaping and enabling the attainment of society's aspirations, higher education institutions (HEIs) are fast assuming leadership positions in championing societal transformation towards sustainability. Consequently, a meteoric rise in the attention being accorded to sustainable development [1] in the aftermath of 'Our Common Future' report of the Brundtland Commission by HEIs has been observed [2]. However, Zutshi and Creed [3] trace the earliest instance of the HEI-sustainability/sustainable development (SD) nexus to the United Nations Conference on the Human Environment in 1972, otherwise known as the Stockholm Declaration. They point out that this nexus was focused on environmental sustainability. However, the Talloires declaration marked a defining moment for SD implementation in HEIs given its global acceptance by university leadership [3]. Since then, HEIs have shown concern about the incorporation of SD into their core activities [4,5]. However, such interest has been traced to the traditional roles of HEIs as change agents [6,7] among increasing expectations from society for them to share knowledge created therein with relevant stakeholders [3]. Such transfer of knowledge will contribute to the expected transformation of

entities situated beyond HEI boundaries, if communicated to appropriate quarters [8]. To buttress the contributions of HEIs to society, Ngo and Trinh [9] have argued for the centrality of HEIs not just in the modernisation of society, but also for the development and provision of cultural centres which have formed the bedrock for the physical development of cities. This much has been attested by the plethora of SD advocacies, declarations, and partnerships to which a multiplicity of HEIs have signed on to since the post-Talloires era [2,10].

Significant improvements in SD uptake among HEIs across the globe have been reported [11,12]. However, varied SD implementation performances have also been observed [13,14]. Whereas some HEIs have reported successful SD implementation, others have posted underwhelming performances. In some other instances, SD implementation has remained largely underreported [3,4]. This appears to be the case for South African HEIs. As such, a comprehensive assessment of the South African HEI SD implementation performance has become difficult [15]. Undoubtedly, such an assessment will not only contribute to improving understanding of the efforts being made by HEIs, but also lead to the identification of factors influencing the SD implementation performance. Further to this, there is an imminent need for the extant relationships between these factors and the influence of such relationships on the implementation performance to be gauged. Enabling such understanding will facilitate the development of an optimal SD implementation framework within HEIs as managers and implementation agents alike can focus on the aspects that will facilitate optimal implementation and, perhaps, have a significant impact on the implementation performance.

This study was informed by this gap, especially within the South African HEI context. Relevant literature highlights the significant contributions made by organisational factors to the variance experienced in SD performance implementation within HEIs [16,17]. Accordingly, any attempt at addressing SD implementation performance within HEIs will require a thorough understanding of the organisational factors, the extant inter-relationships and the influence thereof, on SD implementation. This is central to the scope of this study.

To achieve its main aim, this study will be structured accordingly. First, a review of the literature on the role of HEIs as SD champions in contemporary society and the nature of organisational factors influencing SD implementation in HEIs will be presented. In the second section, a description of the case study research design employed in the study will be provided. Furthermore, an exposé on the study context, a South African University of Technology (SAUoT), will be given. Additionally, in this section, the modalities behind the use of interpretative structural modelling (ISM) as an analytic tool will be presented. This will be followed by an account of the development of the ISM model in the fourth section, as well as a detailed presentation of the findings as they concern the development of the ISM-based model. A discussion of the findings will be provided in the fifth section of the paper. Finally, the study concludes in the sixth section.

2. Theoretical Perspective

2.1. HEIs and Sustainable Development

The twin concepts of sustainability and SD continue to dominate the global societal development discourse. Developmental patterns are consistently being aligned with the sustainable development goals (SDGs) and the inherent milestones promoted by the United Nations (UN) and adopted by signatory countries of the UN SDG charter. Due to the lack of a widely accepted definition and multiplicity of views concerning their actual connotation, notwithstanding [18], sustainability and SD have remained recurring constants in the scheme of things. This study aligns itself with the position outlined by Boström [19], wherein sustainability was described as a state of utopia, whereas SD consisted of the steps required to arrive at that state. Therefore, within the boundaries of HEIs, the desire to achieve a sustainable university status can be related to sustainability and a utopian state, whilst the strategy and processes being implemented have to conform to SD ethos. Accordingly, within society and the domain of societal sustainability aspirations, HEIs are expected to contribute through

their internal processes, with the product of such processes and engagement with society shaping SD strides towards the implementation of sustainability [20–26].

An examination of existing literature traces the evolution of the SD theme in HEIs through the various declarations and partnerships that commenced with the Stockholm Declaration of 1972 [2,3,5]. Lozano et al., [2] have demonstrated a long-standing commitment of HEIs to leading the drive for SD. This commitment is supposedly premised on the traditional role of HEIs as new knowledge creators [27]. The significance of HEIs to the SD challenge is reinforced by the fact that 14 out of 18 declarations pertaining to education are targeted at HEIs. HEIs are associated with the advancement of knowledge frontiers, hence society's expectation for them to provide leadership for SD [28]. They are expected to ensure that the next generation of professionals are equipped with the necessary skills to oversee the implementation of SD across various societal and organisational facets [12,29–35]. According to Stephens, Hernandez [29], HEIs are expected to show leadership by example, serving as models of sustainable practice for the society to emulate. Furthermore, Lozano et al., [2] have mentioned a surge in the number of HEI signatories. Several scholars have highlighted the efforts being made across the global HEI community in championing SD implementation within and beyond their institutions. For instance, Hugé, Block [36], in an assessment of SD implementation in HEIs, observed the slow integration of sustainability into academic research. They proposed actions for achieving this objective, having noted the criticality of such incorporation for societal sustainability aspirations. In another study, Ngo and Trinh [9] investigated the manner through which HEIs provide the intellectual requirements of their environs using the university-city complex model. The point of departure in this study was the notion that HEIs were responsible for the creation of knowledge that can be subsequently applied to city development in a manner corresponding with society's SD aspirations. Another contribution of HEIs to the production of sustainability knowledge for society's benefit was highlighted by Trencher, Nagao [24]. According to them, society stands to benefit from the co-creation of sustainability knowledge with HEIs. However, HEIs have to be adequately prepared to carry such responsibility. Sedlacek [37] has articulated the place of universities in engendering sustainable development within societal boundaries. According to Sedlacek, HEIs contribute to SD at an individual and societal level through research, the education of individuals and the supposed influence on governance. It is expected that, as sustainability champions, HEIs can influence the mindset of individuals and policymakers towards SD [38]. Furthermore, the contributions of HEIs towards energy efficiency and conservation through the design and subsequent development of sustainable campus improvement programmes are outlined in Faghihi, Hessami [39]. Concerning the incorporation of a sustainability ethos into the curriculum, scholars across different studies have shown how HEIs are transforming their curriculum across various disciplines towards being pro-SD in nature, through an assessment of the extant curriculum and subsequent modifications [40-47].

From the above, it can be deduced that HEIs have indeed taken the lead to support society's sustainability aspirations at different levels and through numerous means available to them. Despite this, 28 years after the first declaration directly targeted at HEIs, there has been a lack of reports detailing successful SD implementation across various HEIs, particularly among signatories of these declarations [2]. This under-reporting, a severe incidence across the developing world, has been attributed to an underwhelming implementation performance as HEIs which have made a success of SD implementation have often drawn attention to their achievements [48]. In addition, the seeming failure of HEIs to provide leadership has been blamed as the reason for the poor societal uptake of the agenda in these countries. Nevertheless, certain attributes of HEIs have been blamed for this. Salient amongst these attributes is the HEIs' renowned resistance to change and innovation [6,49] and the discipline-centric nature of these institutions, which leads to knowledge compartmentalisation. Compartmentalisation is considered hostile to the creation of relevant knowledge for resolving SD-related challenges, hence the recent agitation surrounding the adoption of inter-and transdisciplinary research approaches [23,50]. The inability of HEIs to deliver on the expectations of society on

sustainability and SD has been attributed to a plethora of factors [16,22,51]. These factors are covered in more detail in the next section.

2.2. Organisational Factors and SD Implementation

Richardson and Lynes [51] have attempted to categorise the factors influencing SD implementation in HEIs into two broad categories, namely organisational and financial factors. This study will focus on organisational factors because they arguably outweigh other factors. Table 1 chronicles various factors according to their categories, as sourced from a review of relevant literature. From Table 1, it can be deduced that the organisational factors outweigh the financial factors and, accordingly, significantly influence implementation. As such, managing the influence of this category of factors will yield positive changes on the SD implementation curve. These factors are highlighted in Table 1. For clarity, the organisational factors are classified as Collaboration, Leadership, Communication, Knowledge, Behavioural, and Physical factors.

Category	Factors	Authors
Organisational	Collaboration (Stakeholder Collaboration/Staff Commitment/Student Partnerships/Collaborative Decision-making/Presence of Silos)	Lozano-García, Huisingh [14], Ralph and Stubbs [14], Stafford [17], Sharp [52], Shriberg [27], Sharp [17], Cortese [6], McMillin and Dyball [13]
0	Leadership (Strategic Vision/Support from Top Level Management /Visionary Leadership) Incentive Structure/Connectors to Society/Coordination Units and Projects/Sustainability Champions/Organizational Structure/Societal Pressure)	Ferrer-Balas, Adachi [53], Velazquez, Munguia [16], Luo and Yang [49], Sharp [52], Velazquez, Munguia [4]
	Communication (Communication of the Sustainability concept)	Luo and Yang [49], Sharp [52], Djordjevic and Cotton [54], Franz-Balsen and Heinrichs [55]
	Knowledge (Degree of Innovativeness/Understanding/ Awareness/Experience/Skills)	Ferrer-Balas, Lozano [49], Velazquez, Munguia [52], Luo and Yang [42], Ralph and Stubbs [11]
	Behavioural (Appreciation of the Value of Outreach Activities within Academia/Level of Freedom exercised by Faculty Members/Desire to Change)	Ferrer-Balas, Adachi [53], Ferrer-Balas, Lozano [56], Luo and Yang [49], Ralph and Stubbs [12], Shribeg [27]
	Physical (Organizational Size)	Stafford [17] Ferrer-Balas, Adachi [53]
Financial	Finance (Financial Constraints/Consideration of Life-Cycle Savings During Budget Modelling/Source of Funding)	Luo and Yang [49], Stafford [17], Velazquez, Munguia [4], Ralph and Stubbs [12], Ferrer-Balas, Lozano [56], Ferrer-Balas, Adachi [53]

Table 1. Factors Influencing sustainable development (SD) Implementation in higher education.

Source: Awuzie and Emuze [48].

Having identified the factors capable of influencing the SD implementation performance in HEIs, it becomes pertinent to explore the interrelationships existing between these factors. Also imperative is the need to establish the influence of such interrelationships on the SD implementation performance within HEIs. It is to be expected that this understanding will engender the successful development of

an appropriate mechanism for managing SD implementation effectively in the HEIs for the benefit of the wider society. This is what subsequent parts of this paper will concern themselves with, albeit via a case study of the context of a South African University of Technology (SAUoT) and the institution's sustainable university (SU) aspirations.

3. Materials and Methods

The objective of this study is to explore the interrelationship existing between organisational factors influencing the SD implementation performance within an HEI context, relying on an SAUoT exemplar. Accordingly, the choice of the case study research design came naturally to the authors due to the design's reputation for enabling an understanding of a phenomenon within its context. In this case, the choice of method facilitated not just the identification of the organisational factors affecting SD implementation within SAUoT from the perspective of relevant parties, but also enabled the determination of the relationships between these factors and the influence thereof, on implementation performance within the case study. Additionally, the case study design makes the use of a multiplicity of data collection and analysis techniques necessary during the data elicitation and analysis stages of the research project [57,58].

In this study, data was collected using a focus group discussion. Focus group discussions have been described as an appropriate medium for eliciting the opinions of a small group of stakeholders concerning any phenomenon, with a facilitator steering, discussions accordingly [59,60]. According to Hugé, Block [36], this data elicitation technique is specifically designed to elicit information concerning people's preferences, opinions, and values as it pertains to a given topic. However, Kitzinger [60] was quick to add that the success of the focus group was largely dependent on the facilitation skills of the facilitator, as well as the selection of the focus group panel. According to Kitzinger, the wrong facilitation will yield responses that are irrelevant to the scope of the study. A lack of consideration of the power dynamics existing between the discussants during recruitment, as well as the absence of a strategy for dealing with this, if it exists, was capable of undermining whatever benefits were expected. This is necessarily so as these discussions are often held in a permissive, convivial environment, allowing for free and unhindered interaction between discussants.

Discussants of the focus group were purposively selected based on their roles in SD implementation at SAUoT. Care was taken to cover all facets of engagement with SD implementation. However, students were not enlisted at this point, representing a probable limitation of the study; this was intentional as the authors opined that a smaller sample of implementing agents was appropriate for the study. The lead author acted as the facilitator and was assisted by the second author at different intervals. Table 2 provides a description of the focus group discussants' demographics.

No	ID	Job Sector
1	L	Lecturer
2	FS1	Facilities
3	RF	Research Fellow
4	PS	Procurement Staff
5	L2	Lecturer
6	FS2	Facilities

Table 2. Focus Group Discussants' Profile.

In total, six discussants participated in the focus group besides the authors. The session lasted for approximately two and half hours. With the permission of the discussants, the session was recorded and subsequently transcribed. The discussions at this point centred on the interpretative structural modelling (ISM) protocol indicating a pairwise relationship between the identified organisational factors; see Table 3. As indicative of the protocol utilised, an ISM methodology was deployed when

analysing the data emanating from the transcripts. This culminated in the development of an ISM-based model showing the relationships between the identified factors.

	Organisational Factors	Description
1	Collaboration	Stakeholder collaboration to attain a common purpose such as SD implementation
2	Leadership	Willingness from upper management to buy-in and drive SD implementation and willingness to lead the HEI towards the attainment of an SU status
3	Communication	The ability of HEIs to effectively communicate its SD agenda to all stakeholders
4	Knowledge	The ability of an organisation to create and share knowledge
5	Behavioural	Human behaviour, reforming the individual, change in attitude
6	Physical	The size of an organisation

Table 3.	A Descri	ption of O	Prganisational	Factors.
----------	----------	------------	-----------------------	----------

Adapted from Awuzie and Emuze [48].

3.1. Study Context: SAUoT

In 2010/2011, SAUoT commenced a transformational journey towards becoming a sustainable university of technology (SUoT). This transformation was built around the following context-specific features: its place as a South African public institution and its nature as a university of technology (UoT). The former makes it imperative for SAUoT to adopt and support the national commitments and development aspirations of the government and citizenry of the South African nation, especially as it concerns making contributions in science, technology transfer, and education. The latter is concerned with the UoT's institutional context.

SAUoT 's resolve in achieving an SUoT status is discernible, particularly given its development of a sustainability implementation framework. Furthermore, the HEI has inaugurated a Sustainable Development Working Group with a mandate to monitor and co-ordinate the various SD projects. These gestures signal its move from strategy adoption and articulation to actual implementation.

Obviously, it is one of the few HEIs within South Africa that has developed such an implementation framework. It is the intention of this study to identify the motivating factors (drivers) behind the HEI's resolve to embark upon SD implementation. It is believed that an identification of these drivers will promote the development of a social ontology among various stakeholders of the implementation exercise and thus enable a positive attitudinal change amongst them and provide a framework for other HEIs embarking on this path.

3.2. Interpretive Structural Modelling (ISM)

ISM is a qualitative and interpretive method used to generate solutions for complex problems. ISM is a valuable management tool because it identifies the relevant importance of each variable with reference to the problem under consideration [61,62].

ISM is used to identify and structure a relationship between variables that define a problem. The main objectives of ISM are as follows:

- 1. To identify and rank the relationship between variables;
- 2. To discuss the managerial implications of the outcome to aid decision makers.

The various steps involved in the ISM method are extracted from [63–65] and are as follows:

Step 1. Identification of variables relevant to the problem. This can be done through secondary data or primary data, such as that collected through interviews, surveys or focus groups;

Step 2. Establishing a contextual relationship type, such as influence or drive, depending on the problem;

Step 3. Development of a structural self-interaction matrix (SSIM) by a pair-wise comparison. This step will be carried out by experts on the problem context. The participants must decide upon the pairwise relationship between the variables. The existence of a relation between any two variables (i and j) and the associated direction of the relation is questioned. Four symbols are used to denote the direction of the relationship between the variables i and j:

V—for the relation from i to j, but not in both directions i \longrightarrow j; A—for the relation from j to I, but not in both directions j \longrightarrow i; X—for both direction relations from i to j and j to i, as well as i \longleftrightarrow j;

O—if the relation between the variables does not appear to be valid i \leftarrow i:

Step 4. A reachability matrix (RM) is developed from the SSIM and the matrix is checked for transitivity. The transitivity of the contextual relation is a basic assumption made in ISM. It states that if a variable X is related to Y and Y is related to Z, then X is necessarily related to Z. The reachability matrix is a binary matrix since the entries V, A, X, and O of the SSIM are converted into 1 and 0 as follows: V, X = 1 & A, O = 0;

Step 5. Classification of variables based on their driving and dependence power using MICMAC (matriced' impacts croises-multipication applique' and classment) analysis;

Step 6. The reachability matrix obtained in step 4 is partitioned into different levels;

Step 7. Based on the relationships given above in the reachability matrix, a directed graph is drawn, and the transitive links are removed;

Step 8. The ISM model developed in step 7 is reviewed to check for conceptual inconsistency and necessary modifications are made.

4. ISM Model Development

In this section, a rendition of the steps taken towards model development based on the ISM methodology is provided.

4.1. Identification of Organisational Factors Affecting Sustainable Development in a South African University

The factors have been identified in previous research by combining the literature and a qualitative single case study research design in the same study context [48]. The final organizational factors resulted from a series of semi-structured interview sessions held with purposively selected interviewees. The identified factors are presented in Table 3 below with a brief description.

4.2. Developing SSIM for Organisational Factors

Existing pair-wise relationships were identified via a focus group with six stakeholders from SAUoT, as detailed previously. A contextual relationship of "influence" was chosen for the ISM focus group protocol, as reported in Table 4.

	Pairwise Relationship	Type of Relationship	Response
1	Collaboration-Leadership	Do collaboration factors influence leadership factors?	Yes
2	Collaboration-Communication	Do collaboration factors influence Communication factors?	No
3	Collaboration-Knowledge	Do collaboration factors influence Knowledge factors?	Yes
4	Collaboration-Behavioural	Do collaboration factors influence Behavioural factors?	Yes
5	Collaboration-Physical	Do collaboration factors influence physical factors?	Yes
6	Leadership-Collaboration	Do leadership factors influence collaboration factors?	Yes
7	Leadership-Communication	Do leadership factors influence Communication factors?	Yes
8	Leadership-knowledge	Do leadership factors influence knowledge factors?	Yes
9	Leadership-Behavioural	Do leadership factors influence Behavioural factors?	Yes
10	Leadership-Physical	Do leadership factors influence Physical factors?	Yes
11	Communication-Collaboration	Do Communication factors influence Collaboration factors?	Yes
12	Communication-Leadership	Do Communication factors influence Leadership factors?	Yes
13	Communication-knowledge	Do Communication factors influence knowledge factors?	Yes
14	Communication-Behavioural	Do Communication factors influence Behavioural factors?	Yes
15	Communication-Physical	Do Communication factors influence Physical factors?	Yes
16	Knowledge-Collaboration	Do Knowledge factors influence Collaboration factors?	Yes
17	Knowledge-Leadership	Do Knowledge factors influence Leadership factors?	Yes
18	Knowledge-Communication	Do Knowledge factors influence Communication factors?	Yes
19	Knowledge-Behavioural	Do Knowledge factors influence Behavioural factors?	Yes
20	Knowledge-Physical	Do Knowledge factors influence Physical factors?	Yes
21	Behavioural-Collaboration	Do Behavioural factors influence Collaboration factors?	Yes
22	Behavioural-Leadership	Do Behavioural factors influence Leadership factors?	Yes
23	Behavioural-Communication	Do Behavioural factors influence Communication factors?	Yes
24	Behavioural-knowledge	Do Behavioural factors influence knowledge factors?	Yes
25	Behavioural-Physical	Do Behavioural factors influence Physical factors?	Yes
26	Physical-Collaboration	Do Physical factors influence Collaboration factors?	Yes
27	Physical-Leadership	Do Physical factors influence Leadership factors?	Yes
28	Physical-Communication	Do Physical factors influence Communication factors?	Yes
29	Physical-Knowledge	Do Physical factors influence Knowledge factors?	Yes
30	Physical-Behavioural	Do Physical factors influence Behavioural factors?	Yes

Table 4. Interpretative su detarar modelning (1514) i ocus Group i rotoce	Table 4. Inter	pretative structural	l modelling	(ISM)	Focus	Group	Protocol.
--	----------------	----------------------	-------------	-------	-------	-------	-----------

Table 4 presents pair-wise relationships and the response of each relationship by the stakeholders. This step will inform the next steps in the development of the SSIM matrix as explained below.

Four symbols were used to denote the direction of the relationship between any two organisational factors (i and j):

- 1. V: factor i will influence factor j, but not in both directions;
- 2. A: factor j will influence factor I, but not in both directions;
- 3. X: factor i and j will influence each other; and
- 4. O: factor i and j are unrelated.

Table 5 presents the SSIM with different symbols relevant to each pair-wise relationship.

No.	Organisational Factors	1	2	3	4	5	6
1	Collaboration		Х	А	Х	Х	Х
2	Leadership			Х	Х	Х	Х
3	Communication				Х	Х	Х
4	Knowledge					Х	Х
5	Behavioural						Х
6	Physical						

Table 5. Structural self-interaction matrix (SSIM).

Table 5 builds on Table 4's responses and uses the logic above to construct the SSIM matrix.

From the matrix above, it was clear that all the organisational factors were related, and therefore, we did not use the symbol (O), indicating the absence of a relationship. The contextual relationship between organisational factors was obtained from the participants as detailed in Table 4.

4.3. Developing RM from SSIM

The RM was obtained by converting the SSIM into a binary matrix by substituting V, A, X, and O with 1 and 0 as per the case. The rules for the substitution of 1s and 0s are the following:

- 1. If the (i, j) entry in the SSIM is V, then the (i, j) entry in the reachability matrix becomes 1 and the (j, i) entry becomes 0;
- 2. If the (i, j) entry in the SSIM is A, then the (i, j) entry in the reachability matrix becomes 0 and the (j, i) entry becomes 1;
- 3. If the (i, j) entry in the SSIM is X, then the (i, j) entry in the reachability matrix becomes 1 and the (j, i) entry also becomes 1;
- 4. If the (i, j) entry in the SSIM is O, then the (i, j) entry in the reachability matrix becomes 0 and the (j, i) entry also becomes 0.

Following these rules, the RM for the organisational factors is shown in Table 6.

No.	Organisational Factors	1	2	3	4	5	6
1	Collaboration	1	1	0	1	1	1
2	Leadership	1	1	1	1	1	1
3	Communication	1	1	1	1	1	1
4	Knowledge	1	1	1	1	1	1
5	Behavioural	1	1	1	1	1	1
6	Physical	1	1	1	1	1	1

Table 6. RM matrix.

Table 6 uses the rule of binary entries to replace symbols V, A, X, and O with 1 and 0. Therefore, Table 6 converts Table 5 into a binary matrix.

Table 7 presents the final RM. As there is no transitivity, the RM matrix and final RM will be the same. Table 6 has been represented to show the calculation of driving and dependence power as an important step of the ISM method. Hence, it has been re-named as Table 7 because it includes this calculation. The driving power of an organisational factor is the total number of factors, including itself, that it may influence. The dependence power is the total number of factors that may help in alleviating it. Based on the driving and dependence power, the organisational factor can be classified into four clusters: autonomous, dependent, linkage, and independent/driver barriers. This classification and its implications are explained in more detail in the next section.

No.	Organisational Factors	1	2	3	4	5	6	Driver Power
1	Collaboration	1	1	0	1	1	1	5
2	Leadership	1	1	1	1	1	1	6
3	Communication	1	1	1	1	1	1	6
4	Knowledge	1	1	1	1	1	1	6
5	Behavioural	1	1	1	1	1	1	6
6	Physical	1	1	1	1	1	1	6
	Dependence	6	6	5	6	6	6	35/35

Table 7. Final RM.

4.4. Classification of Organisational Factors: MICMAC Analysis

Based on the driver power and dependence power generated in Table 6, the organisational factors were classified into four clusters, as shown in Figure 1 below.



Figure 1. Driving power and dependence diagram.

The major findings of this classification (Figure 1) were as follows:

- 1. The diagram indicated that there is no factor that comes under an autonomous cluster. Autonomous factors generally appear as weak drivers, as well as being weakly dependent, and are relatively disconnected from the system. These factors do not have much influence on the other factors of the system;
- 2. No dependent factors. The dependent factors mean other factors need to be addressed and removed before their removal;

- 3. All factors were within the linkage cluster. Linkage barriers have a strong driving power, as well as strong dependence. These factors are unstable because any action on them will influence others and have a feedback effect on themselves;
- 4. No factors within the driver cluster. Driver factors will have strong driving power, but weak dependence power. Driver factors need to be addressed first and they can influence all other factors.

4.5. Partitioning the RM into Different Levels

From the final RM, the reachability and antecedent set for each factor were derived and the intersection of these sets was then identified, as presented in Table 8. The factor for which the reachability and the intersection sets were the same in the first iteration was assigned as the top-level element in the ISM hierarchy. Similarly, levels were identified for other factors by duplication of this process. Once the level was identified for a factor, it was discarded from the list of remaining factors. Table 8 presents the first iteration, which showed that five factors out of six were found in the first level. Therefore, the remaining factor was in the second level. These two levels helped in developing the ISM model in the final step.

Organisational Factors	Reachability Set	Antecedent Set Intersect	Intersection Set	Level
1	1, 2, 4, 5, 6	1, 2, 3, 4, 5, 6	1, 2, 4, 5, 6	1st
2	2, 1, 3, 4, 5, 6	2, 1, 3, 4, 5, 6	2, 1, 3, 4, 5, 6	1st
3	3, 1, 2, 4, 5, 6	3, 2, 4, 5, 6	3, 2, 4, 5, 6	
4	4, 1, 2, 3, 5, 6	4, 1, 2, 3, 5, 6	4, 1, 2, 3, 5, 6	1st
5	5, 1, 2, 3, 4, 6	5, 1, 2, 3, 4, 6	5, 1, 2, 3, 4, 6	1st
6	6, 1, 2, 3, 4, 5	6, 1, 2, 3, 4, 5	6, 1, 2, 3, 4, 5	1st

lable o. meranom i	Table 8.	Iteration	1
--------------------	----------	-----------	---

4.6. Developing the ISM Model for Organisational Factors

From Table 6, all factors but communications were found at level one. Therefore, they will be positioned at the top-level of the ISM hierarchy. The final ISM model for organisational factors is shown in Figure 2 below. The arrow direction indicates the relationship between the different factors. For example, the relationship between collaboration and leadership factors was a two-way relationship. Therefore, an arrow pointing in both directions was used to denote this relationship. Conversely, the relationship between communication and collaboration factors only occurred in one direction, in which the former could influence the latter. Therefore, an arrow pointing from the communication to collaboration factor was used.



Figure 2. Interpretative structural modelling (ISM)-based model for organisational factors influencing the sustainable development (SD) implementation performance.
It was found from Figure 2 that communication (factor 3) was a significant organisational factor affecting SD implementation as it came in the base level of the ISM model. However, the ISM model only two levels to indicate the close relationship between these factors.

5. Discussion

From the ISM-based model shown in Figure 2, a close relationship between the various organisational factors which have been previously identified as influencing the SD implementation performance at the SAUoT was observed. The positioning of these factors on two levels, levels 1 and 2, highlights this closeness. The model serves as an example of operational management of HEIs to complement teaching and research for SD to contribute to a whole-institute approach to SD [13].

The salience of communication as a critical factor capable of influencing other organisational factors has been reinforced. This notion corresponds to findings from similar studies, which have sought to investigate the SD implementation both within HEIs and in other organisations [3,21,66,67]. For instance, Zutshi and Creed [3] cite instances where HEIs have been admonished to ensure that the communication of their sustainability themes remains at the forefront of their institutional communication strategies. Furthermore, they allude to happenings in the United Kingdom, United States, and Australian HEI context, where such suggestions are also being observed. Still focusing on the utility of communication protocols, Zutshi and Creed [3] maintain that an evaluation of the effectiveness of communication protocols can only be based on the actions taken by recipients of such information as it is reflective of their interpretation of the information passed on through such protocols. By implication, poor communication protocols will lead to wrong interpretations by the stakeholders, an act which can undermine other factors identified in Figure 1. In their contribution, Adomssent, Godemann [8] reiterate the importance of communication and participation in securing the optimal implementation of SD in universities. According to them, communication is usually deployed towards achieving consensus from stakeholders within the HEI context and beyond concerning the developmental pattern that has to be adopted to achieve sustainability, as well as the integral processes thereof. Franz-Balsen and Heinrichs [48], whilst observing the lack of studies looking into sustainability communication management within HEIs at the time, reiterated the significance of all types of communication in engendering effective SD implementation. They maintained that the "vision of a sustainable university is ideally generated in a mutual communication process and is continuously elaborated, there stimulating structural changes as well as individual and collective development" (431). This emphasises the influence of effective communication structures on the breaking down of extant silos which often trigger resistance to change and participation apathy among stakeholders, which are factors recognised as severe impediments to smooth SD implementation in HEIs [16,48].

At SAUoT, previous studies have indicated that the communication of SD implementation plans and processes has proceeded in a top-down manner, with instructions and aspirations being handed down to implementing agents [68]. Additionally, a subsequent study discovered an absence of a common understanding of what sustainability and SD entailed from an SAUoT perspective, as well as what the stakeholders stood to benefit from the transition towards an SU status [69]. Without a common ontology in sight, optimal SD implementation will remain an abandoned idea. No doubt, the setting up of effective communication protocols within SAUoT will lead to an improvement of this situation, as it has been noted that the implementation of SD in the institution has been fraught with participation apathy in most spheres. It has been a case of 'their sustainability not ours', i.e., a lack of ownership in most instances. This working group launched the 'Sustainable Development Working Group' (SDWG), which evolved as a means of breaking down extant discipline-oriented mentality which had led to the growth of knowledge silos within CUT and enabled the effective communication of SAUoT's SD ideals to relevant implementation agents. The group had an adequate representation from all stakeholder groups in SAUoT and was steered by the office of the Deputy Vice-Chancellor, Research, Innovation and Engagement. However, the continued apathy towards participation in SD implementation implies a seemingly lackluster performance of this working

group as the silos have continued to persist. The group's mode of operation is being associated with another form of top-down implementation communication structure. The group is supposed to guide implementing agents within the institution in such a manner as to achieve conformity with the contents of a pre-determined sustainability development implementation framework. Therefore, this mode of operation is considered a contradiction for optimal communication.

The inadequacies of such top-down communication protocols are perceived as authoritarian and incapable of stirring individual and collective interest in SD implementation [61]. In her study into SD implementation across 30 HEIs, Sharp [11] advocated for the adoption of person-to-person communication and dialogue, as well as improved listening skills on the part of the agents. Studies like Disterheft, Caeiro [70] have highlighted the need for the adoption of participatory approaches in driving SD implementation. However, such implementation approaches can only yield positive outcomes with the support of effective communication. Djordjevic and Cotton [54] have identified barriers which serve to undermine the communication process of sustainability and SD in HEIs and these barriers do not largely differ from what is obtained within the study context. These barriers range from the complexity and non-contextual orientation of the message being communicated and a lack of the same understanding between the sender and recipient concerning the contents of the message, to information overload, thus leading to noisy channels for information sharing, top-down communication, and overt-reliance on electronic communication, often at the expense of face-to-face communication etc. Therefore, it is possible to state that effective communication structures are imperative for successful levels of SD implementation to be achieved in HEIs, like the SAUoT being understudied. According to Zutshi and Creed [3], the Talloires declaration was particularly emphatic concerning the relevance of communication in driving SD adoption and implementation. To reinforce this position, they observed that the aspects pertaining to communication constituted 60% of the recommended actions stated in that declaration. However, according to them, a meagre 0.2% of the signatories of that declaration have effectively communicated SD through their institutional websites. Speaking from a conventional organisational perspective, Siano, Conte [71] have described sustainability communication as a necessary platform for showcasing an organisation's sustainability commitment, whilst also facilitating communication of the reasons behind their sustainability ideals, thus allowing for an appreciation of the alignment between the organisational projects and corporate image. This is the situation in the HEI organisational context as HEIs need to showcase their sustainable development ideals through effective communication protocols as this will bring about an increased commitment from a diverse range of stakeholders.

Notwithstanding the significance of communication to SD implementation, it is evident from Figure 1 that factors such as leadership, behaviour, knowledge, and physical factors are all influenced by communication in a bi-directional manner. This much affirmed by Mohamad, Kadir [72] in their study on the importance of the heartware in engendering SD implementation in HEIs. Heartware in this instance was exercised through the presence of shared values concerning sustainability among stakeholders, which in turn, inspire voluntary action and adaptive governance for the resolution of any conflicts hindering implementation [72]. The organisational factors were identified as impacting the heartware aspect, in addition to the software and hardware aspects. Therefore, it is necessary that these factors be considered in tandem with the communication structures within the HEI.

6. Research Implications

The ISM-based model developed will assist the management of the SAUoT to re-focus their implementation efforts accordingly. Ideally, such efforts will take into consideration the re-design of the sustainability communication strategy of the HEI. The model provides a step-by-step guide to start solving the problem. The SAUoT can embark on a wider consultation to identify effective means of communication. The university should identify common and specific modes of communication, taking into consideration the current communication culture and preference. ISM-based model is focused on

the operational management of SAUoT to complement its existing teaching and research strategies for SD.

Decision makers will need to appreciate the close relationship between these factors, and they will need to invest significant resources to tackle these factors. The findings from this study will enable SAUoT to prioritise and allocate resources for implementing SD more effectively.

7. Conclusions

The implementation of SD findings from this study highlights the extant relationship between the various organisational factors influencing the SD implementation performance in an HEI. An ISM methodology has been used to show these relationships based on the views of implementing agents within the context of SAUoT. Based on evidence from the emergent ISM-based model, it was observed that the factors shared close relationships and influenced each other to a large extent, except for the collaboration–communication pairwise relationship, where the relationship was observed as being uni-directional instead of bi-directional, as is the case in other pairwise relationships evaluated. However, the critical nature of sustainability communication was deduced. Accordingly, the absence of the effective communication of SD will serve to undermine all the other efforts of the distinct implementing agents, as reiterated in similar studies. The case is no different to the case investigated herein.

The study relied on a single case study: the SAUoT. As such, the conclusions arrived at are reflective of this case. Although this can be regarded as a limitation of this study, it must be noted that the scope of this study was to understand the relationships between identified factors influencing the SD implementation performance in the SAUoT context. Additionally, it sought to highlight the utility of the ISM in establishing this relationship. Although both objectives have been achieved, it should be stated that the conclusions reached herein are to be subjected to further generalisation in subsequent studies. Such studies will seek to compare HEIs in different geographic and economic contexts.

Author Contributions: Conceptualisation, B.O.A.; data curation, A.A.; investigation, B.O.A. and A.A.; design of ISM methodology, A.A.; project administration, B.O.A.; Writing—original draft, B.O.A.; writing—review & editing, A.A.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Geissdoerfer, M.; Savaget, P.; Bocken, N.M.; Hultink, E.J. The Circular Economy—A new sustainability paradigm? J. Clean. Prod. 2017, 143, 757–768. [CrossRef]
- Lozano, R.; Lukman, R.; Lozano, F.J.; Huisingh, D.; Lambrechts, W. Declarations for sustainability in higher education: Becoming better leaders, through addressing the university system. J. Clean. Prod. 2013, 48, 10–19. [CrossRef]
- Zutshi, A.; Creed, D.A. Declaring Talloires: Profile of sustainability communications in Australian signatory universities. J. Clean. Prod. 2018, 187, 687–698. [CrossRef]
- 4. Velazquez, L.; Munguía, N.; Platt, A.; Taddei, J. Sustainable university: What can be the matter? *J. Clean. Prod.* **2006**, *14*, 810–819. [CrossRef]
- Michelsen, G. Policy, politics and polity in higher education for sustainable development. In *Routledge* Handbook of Higher Education for Sustainable Development; Barth, M., Michelsen, G., Thomas, I., Rieckmann, M., Eds.; Routledge: London, UK, 2016; pp. 40–55.
- Cortese, A.D. The critical role of higher education in creating a sustainable future. *Plan. High. Educ.* 2003, 31, 15–22.
- Wright, T.; Horst, N. Exploring the ambiguity: What faculty leaders really think of sustainability in higher education. Int. J. Sustain. High. Educ. 2013, 14, 209–227. [CrossRef]
- Godemann, J.; Michelsen, G. Transferability of approaches to sustainable development at universities as a challenge. *Int. J. Sustain. High. Educ.* 2007, *8*, 385–402.

- Ngo, L.M.; Trinh, T.A. A University-city Complex, a Model for Sustainable Development: A Case Study in Vietnam. *Procedia Eng.* 2016, 142, 92–99. [CrossRef]
- Wright, T. The evolution of sustainability declarations in higher education. In *Higher Education and the Challenge of Sustainability: Problematics, Promise, and Practice;* Corcoran, P.E., Wals, A.E.J., Eds.; Springer: Amsterdam, The Netherlands, 2004; pp. 7–19.
- Sharp, L. Higher education: The quest for the sustainable campus. Sustain. Sci. Pract. Policy 2009, 5, 1–8. [CrossRef]
- 12. Ralph, M.; Stubbs, W. Integrating environmental sustainability into universities. *High. Educ.* 2014, 67, 71–90. [CrossRef]
- McMillin, J.; Dyball, R. Developing a whole-of-university approach to educating for sustainability linking curriculum, research and sustainable campus operations. J. Educ. Sustain. Dev. 2009, 3, 55–64. [CrossRef]
- 14. Lozano-García, F.J.; Huisingh, D.; Delgado-Fabián, M. An interconnected approach to incorporate sustainable development at Tecnológico de Monterrey. *Int. J. Sustain. High. Educ.* **2009**, *10*, 318–333. [CrossRef]
- Purushottam, N.; Rwelamila, P. Green campus initiatives as projects: Can creating conducive internal university project environment a key to success? In Proceedings of the 31st Annual ARCOM Conference, Lincoln, UK, 7–9 September 2015; Association of Researchers in Construction Management: London, UK, 2015.
- Velazquez, L.; Munguia, N.; Sanchez, M. Deterring sustainability in higher education institutions: An appraisal of the factors which influence sustainability in higher education institutions. *Int. J. Sustain. High. Educ.* 2005, *6*, 383–391. [CrossRef]
- 17. Stafford, S.L. How green is your campus? An analysis of the factors that drive universities to embrace sustainability. *Contemp. Econ. Policy* **2011**, *29*, 337–356. [CrossRef]
- Waas, T.; Hugé, J.; Verbruggen, A.; Wright, T. Sustainable Development: A Bird's Eye View. Sustainability 2011, 3, 1637–1661. [CrossRef]
- 19. Boström, M. A missing pillar? Challenges in theorizing and practicing social sustainability: Introduction to the special issue. *Sustain. Sci. Pract. Policy* **2012**, *8*, 3–14. [CrossRef]
- Friman, M.; Schreiber, D.; Syrjänen, R.; Kokkonen, E.; Mutanen, A.; Salminen, J. Steering sustainable development in higher education—Outcomes from Brazil and Finland. *J. Clean. Prod.* 2018, 186, 364–372. [CrossRef]
- Ramos, T.B.; Caeiro, S.; Van Hoof, B.; Lozano, R.; Huisingh, D.; Ceulemans, K. Experiences from the implementation of sustainable development in higher education institutions: Environmental Management for Sustainable Universities. J. Clean. Prod. 2015, 106, 3–10. [CrossRef]
- 22. Sammalisto, K.; Sundström, A.; Holm, T. Implementation of sustainability in universities as perceived by faculty and staff—A model from a Swedish university. *J. Clean. Prod.* **2015**, *106*, 45–54. [CrossRef]
- Trencher, G.; Nagao, M.; Chen, C.; Ichiki, K.; Sadayoshi, T.; Kinai, M.; Kamitani, M.; Nakamura, S.; Yamauchi, A.; Yarime, M. Implementing Sustainability Co-Creation between Universities and Society: A Typology-Based Understanding. *Sustainability* 2017, *9*, 594. [CrossRef]
- 24. Khalili, N.R.; Duecker, S.; Ashton, W.; Chavez, F. From cleaner production to sustainable development: The role of academia. J. Clean. Prod. 2015, 96, 30–43. [CrossRef]
- 25. Filho, W.L. About the Role of Universities and Their Contribution to Sustainable Development. *High. Educ. Policy* **2011**, *24*, 427–438. [CrossRef]
- Godemann, J.; Bebbington, J.; Herzig, C.; Moon, J. Higher education and sustainable development: Exploring possibilities for organisational change. *Account. Audit. Account. J.* 2014, 27, 218–233. [CrossRef]
- Shriberg, M.P. Sustainability in US Higher Education: Organizational Factors Influencing Campus Environmental Performance and Leadership. Ph.D. Thesis, The University of Michigan, Ann Arbor, MI, USA, 2002.
- Stephens, J.C.; Hernandez, M.E.; Román, M.; Graham, A.C.; Scholz, R.W. Higher education as a change agent for sustainability in different cultures and contexts. *Int. J. Sustain. High. Educ.* 2008, *9*, 317–338. [CrossRef]
- Aktas, C.B.; Whelan, R.; Stoffer, H.; Todd, E.; Kern, C.L. Developing a university-wide course on sustainability: A critical evaluation of planning and implementation. *J. Clean. Prod.* 2015, 106, 216–221. [CrossRef]

- 30. Karatzoglou, B. An in-depth literature review of the evolving roles and contributions of universities to Education for Sustainable Development. *J. Clean. Prod.* **2013**, *49*, 44–53. [CrossRef]
- Lozano, R.; Barreiro-Gen, M.; Lozano, F.J.; Sammalisto, K. Teaching Sustainability in European Higher Education Institutions: Assessing the Connections between Competences and Pedagogical Approaches. *Sustainability* 2019, 11, 1602. [CrossRef]
- Dlouhá, J.; Heras, R.; Mulà, I.; Salgado, F.P.; Henderson, L. Competences to Address SDGs in Higher Education—A Reflection on the Equilibrium between Systemic and Personal Approaches to Achieve Transformative Action. *Sustainability* 2019, *11*, 3664. [CrossRef]
- Wiek, A.; Withycombe, L.; Redman, C.L. Key competencies in sustainability: a reference framework for academic program development. *Sustain. Sci.* 2011, *6*, 203–218. [CrossRef]
- Rieckmann, M. Future-oriented higher education: Which key competencies should be fostered through university teaching and learning? *Futures* 2012, 44, 127–135. [CrossRef]
- Barth, M.; Godemann, J.; Rieckmann, M.; Stoltenberg, U. Developing key competencies for sustainable development in higher education. *Int. J. Sustain. High. Educ.* 2007, 8, 416–430. [CrossRef]
- Hugé, J.; Block, T.; Waas, T.; Wright, T.; Dahdouh-Guebas, F. How to walk the talk? Developing actions for sustainability in academic research. J. Clean. Prod. 2016, 137, 83–92. [CrossRef]
- Sedlacek, S. The role of universities in fostering sustainable development at the regional level. *J. Clean. Prod.* 2013, 48, 74–84. [CrossRef]
- Ferrer-Balas, D.; Buckland, H.; De Mingo, M. Explorations on the University's role in society for sustainable development through a systems transition approach. Case-study of the Technical University of Catalonia (UPC). J. Clean. Prod. 2009, 17, 1075–1085. [CrossRef]
- Faghihi, V.; Hessami, A.R.; Ford, D.N. Sustainable campus improvement program design using energy efficiency and conservation. J. Clean. Prod. 2015, 107, 400–409. [CrossRef]
- Mulder, K.F.; Segalas, J.; Ferrer-Balas, D. How to educate engineers for/in sustainable development: Ten years of discussion, remaining challenges. *Int. J. Sustain. High. Educ.* 2012, 13, 211–218. [CrossRef]
- Holm, T.; Sammalisto, K.; Grindsted, T.S.; Vuorisalo, T. Process framework for identifying sustainability aspects in university curricula and integrating education for sustainable development. J. Clean. Prod. 2015, 106, 164–174. [CrossRef]
- Lambrechts, W.; Mulà, I.; Ceulemans, K.; Molderez, I.; Gaeremynck, V. The integration of competences for sustainable development in higher education: An analysis of bachelor programs in management. *J. Clean. Prod.* 2013, 48, 65–73. [CrossRef]
- Lozano, F.J.; Lozano, R. Developing the curriculum for a new Bachelor's degree in Engineering for Sustainable Development. J. Clean. Prod. 2014, 64, 136–146. [CrossRef]
- 44. Lozano, R.; Watson, M.K. Chemistry Education for Sustainability: Assessing the chemistry curricula at Cardiff University. *Educ. Quím.* 2013, 24, 184–192. [CrossRef]
- 45. Lozano, R. Diffusion of sustainable development in universities' curricula: An empirical example from Cardiff University. J. Clean. Prod. 2010, 18, 637–644. [CrossRef]
- Cai, X. Overviews of Campus Sustainability Projects at Illinois: Opportunities for Education and Research. Ph.D. Thesis, Institute for Sustainability, Energy, and Environment, University of Illinois at Urbana-Champaign, Urbana-Champaign, IL, USA, 2017.
- Thomas, I. Challenges for implementation of education for sustainable development in higher education institutions. In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Thomas, I., Rieckmann, M., Eds.; Routledge: London, UK, 2016; pp. 40–55.
- Awuzie, B.; Emuze, F. An identification of organizational factors affecting sustainable development in a South African University. In Proceedings of the 4th Construction Management Conference, Port Elizabeth, South Africa, 29 November–1 December 2015.
- Luo, H.; Yang, J. Overcoming organisational resistance to sustainability innovations in Australian universities. In Proceedings of the 12th Annual Australasian Campuses Towards Sustainability, Queensland, Australia, 26–28 September 2012.

- 50. Biberhofer, P.; Rammel, C. Transdisciplinary learning and teaching as answers to urban sustainability challenges. *Int. J. Sustain. High. Educ.* 2017, *18*, 63–83. [CrossRef]
- Richardson, G.R.; Lynes, J.K. Institutional motivations and barriers to the construction of green buildings on campus: A case study of the University of Waterloo, Ontario. *Int. J. Sustain. High. Educ.* 2007, *8*, 339–354. [CrossRef]
- 52. Sharp, L. Green campuses: The road from little victories to systemic transformation. *Int. J. Sustain. High. Educ.* 2002, 3, 128–145. [CrossRef]
- Adachi, J.; Banáš, S.; Davidson, C.; Hoshikoshi, A.; Mishra, A.; Motodoa, Y.; Onga, M.; Ostwald, M.; Ferrer-Balas, D.; Ferrer-Balas, D. An international comparative analysis of sustainability transformation across seven universities. *Int. J. Sustain. High. Educ.* 2008, *9*, 295–316.
- Djordjevic, A.; Cotton, D. Communicating the sustainability message in higher education institutions. *Int. J. Sustain. High. Educ.* 2011, 12, 381–394. [CrossRef]
- 55. Heinrichs, H.; Franz-Balsen, A.; Franz-Balsen, A. Managing sustainability communication on campus: Experiences from Lüneburg. *Int. J. Sustain. High. Educ.* **2007**, *8*, 431–445.
- Ferrer-Balas, D.; Lozano, R.; Huisingh, D.; Buckland, H.; Ysern, P.; Zilahy, G. Going beyond the rhetoric: System-wide changes in universities for sustainable societies. J. Clean. Prod. 2010, 18, 607–610. [CrossRef]
- Corcoran, P.B.; Walker, K.E.; Wals, A.E.J. Case studies, make-your-case studies, and case stories: a critique of case-study methodology in sustainability in higher education. *Environ. Educ. Res.* 2004, 10, 7–21. [CrossRef]
- Kyburz-Graber, R. Case study research on higher education for sustainable development: epistemological foundation and quality challenges. In *Routledge Handbook of Higher Education for Sustainable Development*; Barth, M., Michelsen, G., Thomas, I., Rieckmann, M., Eds.; Routledge: London, UK, 2016; pp. 40–55.
- Stewart, D.W.; Shamdasani, P.N. Focus Groups: Theory and Practice; Sage Publications: Thousand Oaks, CA, USA, 2014; Volume 20.
- Kitzinger, J. The methodology of Focus Groups: The importance of interaction between research participants. Sociol. Health Illn. 1994, 16, 103–121. [CrossRef]
- Shen, L.; Song, X.; Wu, Y.; Liao, S.; Zhang, X.; Xiangnan, S. Interpretive Structural Modeling based factor analysis on the implementation of Emission Trading System in the Chinese building sector. *J. Clean. Prod.* 2016, 127, 214–227. [CrossRef]
- Pfohl, H.-C.; Gallus, P.; Thomas, D.; Pfohl, H. Interpretive structural modeling of supply chain risks. Int. J. Phys. Distrib. Logist. Manag. 2011, 41, 839–859. [CrossRef]
- Attri, R.; Dev, N.; Sharma, V. Interpretive structural modelling (ISM) approach: An overview. *Res. J. Manag. Sci.* 2013, 2, 3–8.
- 64. Ravi, V.; Shankar, R. Analysis of interactions among the barriers of reverse logistics. *Technol. Forecast. Soc. Chang.* **2005**, *72*, 1011–1029. [CrossRef]
- Shahabadkar, P. Deployment of Interpretive Structural Modelling Methodology in Supply Chain Management—An overview. Int. J. Ind. Eng. Prod. Res. 2012, 23, 195–205.
- Lozano, R.; Ceulemans, K.; Alonso-Almeida, M.; Huisingh, D.; Lozano, F.J.; Waas, T.; Lambrechts, W.; Lukman, R.; Hugé, J. A review of commitment and implementation of sustainable development in higher education: Results from a worldwide survey. J. Clean. Prod. 2015, 108, 1–18. [CrossRef]
- Saleh, A.A.; Mohammed, A.H.; Abdullah, M.N. Critical Success Factors for Sustainable University: A Framework from the Energy Management View. *Proceedia Soc. Behav. Sci.* 2015, 172, 503–510. [CrossRef]
- 68. Awuzie, B.; Emuze, F.A.; Ngowi, A. Critical Success Factors for Smart and Sustainable Facilities Management in a South African University of Technology. In Proceedings of the 2015 Smart and Sustainable Built Environment (SASBE) Conference, Pretoria, South Africa, 9–11 December 2015; Pretoria CIB/CSIR/University of Pretoria: Pretoria, South Africa, 2015.
- Awuzie, B.; Emuze, F.; Ngowi, A. Towards a Social Ontology on Sustainable Development in CUT: Understanding Stakeholder Perceptions. In *Handbook of Theory and Practice of Sustainable Development in Higher Education*; Leal Filho, W., Skanavis, C., do Paço, A., Rogers, J., Kuznetsova, O., Castro, P., Eds.; Springer: Cham, Switzerland, 2017; pp. 425–439.
- Disterheft, A.; Caeiro, S.; Azeiteiro, U.M.; Filho, W.L. Sustainable universities—A study of critical success factors for participatory approaches. J. Clean. Prod. 2015, 106, 11–21. [CrossRef]

- 71. Siano, A.; Conte, F.; Amabile, S.; Vollero, A.; Piciocchi, P. Communicating Sustainability: An Operational Model for Evaluating Corporate Websites. *Sustainability* **2016**, *8*, 950. [CrossRef]
- Mohamad, Z.F.; Kadir, S.N.; Nasaruddin, A.; Sakai, N.; Zuki, F.M.; Hussein, H.; Sulaiman, A.H.; Salleh, M.S. Heartware as a Driver for Campus Sustainability: Insights from an Action-oriented Exploratory Case Study. J. Clean. Prod. 2018, 196, 1086–1096. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Sustainability Editorial Office E-mail: sustainability@mdpi.com www.mdpi.com/journal/sustainability



MDPI St. Alban-Anlage 66 4052 Basel Switzerland

Tel: +41 61 683 77 34 Fax: +41 61 302 89 18

www.mdpi.com

